BANK REGULATION IMPLICATIONS FOR MANAGING ACCOUNTING QUALITY RISK: A BASEL AND IFRS PERSPECTIVE

TANUJA DULMINI DOMINICK MAHINDA BADDEVITHANA

A thesis submitted in partial fulfillment of the requirement of the University of Greenwich for the Degree of Doctor of Philosophy

November 2012

DECLARATION

I certify that this work has not been accepted in substance for any degree, and is not concurrently being submitted for any degree other than that of the degree of Doctor of Philosophy being studied at the University of Greenwich. I also declare that this work is the result of my own investigations except where otherwise identified by references and that I have not plagiarised the work of others.

Signed by:

Student:	ftre
Date:	30 Nov 2012
Supervisor:	hn
Date:	3 Dec 2012

To my parents

ACKNOWLEDGEMENTS

I wish to extend my sincere gratitude to Dr. Aleksandar Stojanovic for his supervision, continual guidance and advice throughout this research and thesis. I am also grateful to Professor Zeljko Sevic for his supervision and guidance. I would also like to extend my sincere gratitude to Dr. Venkata Chinthalapati who supervised this research and thesis. I would like to thank Mr. John Smullen who supervised this thesis and helped shape the contents during its early stages. I am grateful to Dr. Lesley Catchpowle for her help and guidance in the initial structuring of the thesis and for her assistance at the start of my degree. I would also like to thank Ms. Gillian Haxell who helped with many administrative aspects with regard to my degree.

I would like to thank Dr. Priyanka Baddevithana and Mr. Duminda Baddevithana for their invaluable guidance while proof reading this thesis, and also Ms. Sajini Baddevithana and Mrs. Henrietta Kumarapperuma who responded to my call for assistance to proof read many parts of this thesis.

I wish to extend my gratitude to all my teachers in general, research peers and my professional colleagues for providing assistance and encouragement in many meaningful ways. I would like to thank my friends who provided moral support whenever possible.

Finally, I wish to thank my parents Dr. Mahinda and Mrs. Neliya Baddevithana for their patience and support during this work. To all my family, who at many occasions assisted and gave encouragement to this work, I am deeply grateful.

While acknowledging all who contributed to my research and thesis in so many ways, any errors and omissions that may remain in this thesis are mine alone.

GLOSSARY

This section presents some of the main symbols, terms and abbreviations applied in this thesis.

*	The super-scripted number is at the 95% confidence level (also termed the 95% confidence interval). This is interpreted also as being at the 0.05 significance level or being significant at 5%.
**	The super-scripted number is at the 99% confidence level (also termed the 99% confidence interval). This is interpreted also as being at the 0.01 significance level or being significant at 1%.
Accounting Quality	The characteristics of accounting information signifying qualities that include being relevant, reliable and un-biased.
Accounting Quality Risk	The absolute measure of the difference between the change in an accounting total variable and the change in the market price variable.
	Accounting quality risk is related to accounting quality in that a high level of accounting quality translates to a low level of exposure to accounting quality risk. Conversely, a low level of accounting quality translates to a high level of exposure to accounting quality risk.
Accounting to Market Price Relative Delta	For a firm, the measure of the difference between the change in the accounting total variable and the change in the market price variable for one unit of time.
	For a portfolio or a sample of firms, this is the measure of the difference between the change in the accounting total variable and the change in the market price variable for one unit of time averaged at the firm level. It provides a measure of the level of <i>accounting quality</i> , and is a direct measure of the level of exposure to <i>accounting quality risk</i> .
Accounting Total(s)	The Financial Statement totals generally reported in the Balance Sheet, Income Statement and Cash Flow Statement.

Accounting Value-at-Risk	The measure of the maximum potential change in an accounting total based variable for a specified probability over a specified time horizon.
	In addition, accounting Value-at-Risk is also referred to as the Accounting total Value-at-Risk, or Change in Accounting total Value-at-Risk. The accounting Value-at-Risk measures estimated in this study include the following: Total Equity Value-at-Risk using the total equity change variable, Total Assets Value-at-Risk using the total assets change variable, Total Liabilities Value-at-Risk using the total liabilities change variable and Net Income Value-at-Risk using the net income change variable.
Adjusted Market Price	The Market Price of a firm's equity share (stock) recorded at a specified time from a regulated financial exchange that has been adjusted for capital events (also termed corporate actions). See Appendix Y for a list of capital events that can potentially adjust an equity share's (a stock's) market price.
Adjusted Price	See Adjusted Market Price.
ASB	The Accounting Standards Board The accounting standards setting body for the United Kingdom.
Basel	The Basel Accord or Basel regulations. The standards applied for national and regional banking regulations. The term Basel, in a regulatory context, applies to the regulations specified as Basel I, Basel II, Basel II.5 and Basel III.
BM	Book-to-Market ratio BM may also be written as B/M.
Book Equity	Total Shareholders' Equity adjusted for taxes and preferred stock. See Fama and French 2008, page 2979.
Book Equity Value	See Book Equity.
Book Value	See Book Equity.
Capital Events	See Corporate Actions.
Close Market Price	The market price of a firm's equity share (stock) recorded at the closing time of the exchange the shares are traded on. The 'Close' signifies the end of a trading day.

Company	See Firm.
Control group	Sample of firms from the London Stock Exchange (LSE) that conduct banking related activities and did not adopt the IASB's IFRS accounting standards on 1st January 2005.
Corporate Actions	Actions or events that adjust the Market Price of a firm's equity share (stock).
d(E/S)	Relative change in Total Shareholders' Equity per Common Shares Outstanding
dA	Log change in Total Assets
dBM	Relative change in Book-to-Market Value
dE	Relative change in Total Shareholders' Equity
dI	Relative change in Net Income
Difference Component	A measure of the movement of one variable relative to the movement of another variable in unit time. Provides a measure of <i>valuation bias</i> or <i>information bias</i> . Measures the level of difference quantified by the relative delta measure (see Relative Delta).
Difference or Dispersion	The magnitude of movement of one variable in an opposite direction to another variable.
Distress and Distress Risk	See Financial Distress Risk.
dL	Log change in Total Liabilities
dln (BM)	Log change in Book-to-Market ratio
dln(E)	Log change in Total Shareholders' Equity
dln(E/S)	Log change in Total Shareholders' Equity per Common Outstanding Shares
dln(I)	Log change in Net Income
dln(MB)	Log change in Market-to-Book ratio

dM	Log change in Market Price This is the same as Market Price Return or Market Return, also termed the <i>Price Return</i> or <i>Return</i> .
dMB	Relative change in Market-to-Book ratio
dMV	Log change in Market Value
dS	Log change in Number of Common Shares Outstanding
End-of-Time Horizon Breach Count	The <i>binary count</i> of the event where a Value-at-Risk level is breached by the market price return on the actual time horizon date. The time horizon date is specified as the date that corresponds to the number of days specified by the <i>time</i> <i>horizon</i> counted forward from the current date.
Equity Share	One part portion of equity held by a firm, generally apportioned for a financial instrument traded on a regulated exchange.
Explicit Accounting Role	The role specified in this study that describes the effect financial accounting has within the risk management framework. This role is observed when the accounting function reports information that is accurate and without bias when compared to comparable information observed in the financial markets.
Financial Distress	See Financial Distress Risk.
Financial Distress Risk	The measure of the difference between the change in the total equity variable and the change in the market price variable.
	In addition, the measure of the difference between the change in accounting total variables and the change in the market price variable.
	Financial distress risk is also specified in literature to be an effect from a high book-to-market ratio.
Financial Reporting Quality	See Accounting Quality.
Firm	An organisation or a group that is a going concern and is established for the purpose of conducting business.
Framework	A system of measures or models applied with the aim of addressing a specified objective.

Growth Stock	A financial equity share (stock) of a firm with a low book-to- market ratio.
Historical Value-At- Risk Actual	Unless otherwise stated, the <i>Historical Value-At-Risk Actual</i> measure specifies the <i>market price return Value-at-Risk</i> measured using the historical simulation Value-at-Risk approach. This measure is calculated at the 95% confidence level, and at the 250-day time horizon using 301 historical 250-day market price returns.
	Historical Value-At-Risk Actual is also referred to as the market price return Historical Value-at-Risk Actual.
IASB	International Accounting Standards Board
IFRS	International Financial Reporting Standards The reporting standards that are applied as a part of the IASB accounting standards framework.
Implicit Accounting Role	The effect financial accounting has within the risk management framework when the accounting function reports information that exhibits a different level when compared to comparable information observed in the financial markets.
In-Time Horizon Breach Count	The number of times a Value-at-Risk level is breached by the market price return within a specified time horizon.
Key Accounting Totals	The Balance Sheet measures: Total Shareholders' Equity, Total Assets, Total Liabilities, and the Income Statement measure: Net Income.
Legacy System	A <i>dated</i> mainframe accounting system. This term may also apply to any system that is recognised to require re-engineering or significant upgrading as part of a normal or special course of events.
LHS	Left-Hand Side
ln	Natural logarithm to the base e
Log Return	See Market Price Return.
log_e	See <i>ln</i> .

Μ	Market price variable. Has the same definition as P . Applied when the change in the market price variable dM is used (where dM is also termed the market price return).
Market Price	The price of one equity share (stock) for a firm that is traded on a regulated financial exchange and is observed or recorded at a specified time.
Market Price Return	The change in market price defined relative to some initial market price (JPMorgan and Reuter 1996).
Market Price Return Value-at-Risk	See Market Price Value-at-Risk.
Market Price Value- at-Risk	Value-at-Risk (see Value-at-Risk) calculated by using the market price return variable, also referred to as the <i>market price return Value-at-Risk</i> and also as the <i>Historical Value-at-Risk Actual</i> .
Market Return	See Market Price Return.
Measure	A mathematical rule applied to quantify a specified level.
Model	A single measure or a system of measures based on a specified measurement rule.
Obs.	Average number of possible <i>Observations</i> , rounded to the nearest whole number.
Obs N	Average number of possible <i>Observations</i> that represent the number of firms in the sample, rounded to the nearest whole number.
Obs T	Average number of possible <i>Observations</i> that represent the number of time periods observed, rounded to the nearest whole number.
Open Market Price	The market price of a firm's equity share (stock) recorded at the opening time of the exchange the shares are traded on. 'Open' signifies the start of a trading day.
Р	See Market Price.
Primary sample	Sample comprising the five banks listed in the London Stock Exchange (LSE) banking sector.
R^2	Coefficient of determination

Regulatory Relative Delta	A model proposed to monitor and specify adjustment levels to the Basel minimum regulatory capital level.
Relative Delta	The measure of the difference between the change in one variable and the change in another variable. Applied in this study to represent the measure of the difference between the change in an accounting total variable and the change in the market price variable.
	Provides a measure of the level of <i>accounting quality</i> , and is a direct measure of the level of exposure to <i>accounting quality risk</i> . The <i>relative delta</i> measure may also be referred to by the terms <i>delta relative</i> , <i>delta difference</i> or <i>difference delta</i> .
Return	See Market Price Return.
RHS	Right-Hand Side
SD	Standard Deviation
SE	Standard Error
Secondary sample	Sample of firms from the London Stock Exchange (LSE) that conduct banking related activities and adopted the IASB's IFRS accounting standards on 1st January 2005 and for the years thereafter.
Share	See Equity Share.
SSE	Error Sum of Squares
SST	Total Sum of Squares
Stock	One equity share.
TA	Total Assets amount.
TBM	Total Book-to-Market ratio amount.
TE	Total Shareholders' Equity amount.
TI	Total Net Income amount. In a general sense, Net Income may also be referred to as earnings.
TL	Total Liabilities amount.
TMB	Total Market-to-Book ratio amount.
TMV	Total Market Value amount.

Total Equity	Total Shareholders' Equity measured by applying the prevailing accounting standards.	
Total Equity Book	See Book Equity.	
Total variable	The variable that represents a <i>total</i> value recorded on a specified date. This variable does not represent a change variable such as those calculated using the relative or log change forms.	
TS	Total number of Shares outstanding.	
UK GAAP	The United Kingdom Generally Accepted Accounting Practice	
V	Value-at-Risk variable or Historical Value-at-Risk Actual variable.	
V95	Historical Value-at-Risk Actual variable at the 95% confidence level.	
Value-at-Risk	The measure of the maximum potential change in value of a financial item or instrument for a specified probability over a specified time horizon.	
Value Relevance	A measure of the level of accuracy that a firm's accounting information reflects in its value, when compared to an observable valuation from a financial market based measure. For a firm an example of a financial market based measure with a high level of observability would be its equity share price observed on a regulated financial market.	
Value Stock	A financial equity share (stock) of a firm with a high book-to- market ratio.	
VaR	See Value-at-Risk.	
VaR 001	Value-at-Risk at the 99.9% confidence level, that is, at the 0.001 significance level.	
VaR 01	Value-at-Risk at the 99% confidence level, that is, at the 0.01 significance level.	
VaR 05	Value-at-Risk at the 95% confidence level, that is, at the 0.05 significance level.	
VOL	Volume of trade, specified by the number of a firm's shares traded on a regulated financial exchange during a specified time period, generally 1 trading day.	

ABSTRACT

This thesis examines whether accounting quality, measured as the difference between accounting and market price change, had an impact on the five primary UK banks that adopted IASB's IFRS accounting standards in 2005.

The findings reveal that the changes in accounting standards resulted in the banks experiencing decreased levels of accounting quality and increased levels of exposure to financial distress risk in the period 2005 to 2008, compared to the pre-adoption period of 1992 to 2004. These findings are corroborated when examining a secondary sample of banking related firms that also adopted the same standards in 2005. A control group that did not adopt these standards exhibited an opposite trend, recording a comparative increase in accounting quality from 2005 to 2008. For all firms tested, the 1-day market price Value-at-Risk (VaR) levels increased year-on-year from 2005 to 2009, with VaR breaches during March and May 2006. These firms, for the 2005 to 2009 period, also displayed increased levels of financial market volatility. Importantly, examining the banks' Basel capital requirements, it is implied that their levels increased after 2005. These findings, in general, contribute to extending literature that focuses on the accounting standards change.

One of the findings from this examination is that contrary to the European Commission's 2002 (EC 2002) and IASB's (IASB 2009) expectation to strengthen the efficient functioning of the European and global financial markets, in the UK the banking sector's investor uncertainty increased significantly during 2005 to 2009.

Another finding relates to the measurements applied in this research. Changes in accounting quality and the Basel minimum capital requirement are examined by applying two measures systemised as the *relative delta* and the *regulatory relative delta* respectively. Both function by quantifying differences between accounting totals and market price. It is discovered that these measures, *accounting VaR*, and the technology framework, as introduced in this study, have potential benefits and regulatory implications. These are aimed at facilitating the mitigation of risks that impact on accounting quality.

CONTENTS

DECENTION	ГЮМ	ii
ACKNOWL	EDGEMENTS	iv
GLOSSARY	ζ	v
ABSTRACT	۲	xiii
CONTENTS	5	xiv
LIST OF TA	ABLES	xxi
LIST OF FI	GURES	xxiv
Chapter 1	Introduction	1
1.1	Background	4
1.2	Research Problem	10
1.3	Overview of the Research Methodology	14
1.4	Thesis Organisation	
1.5	Thesis Contributions	20
1.6	Summary	22
	L'Anna Anna Dan ànna	24
Chapter 2	Literature Review	
Chapter 2 2.1	Chapter Introduction	24
2.1 2.2	Chapter Introduction Research Background	24 24 24
2.1 2.2 2.3	Chapter Introduction Research Background Background to the 2005 Accounting Change	
2.1 2.2 2.3 2.4	Chapter Introduction Research Background Background to the 2005 Accounting Change Risk Measurement.	
2.1 2.2 2.3 2.4 2.5	Chapter Introduction Research Background Background to the 2005 Accounting Change Risk Measurement Standard Deviation, CAPM, and the Book-to-Market Ratio	
2.1 2.2 2.3 2.4 2.5	Chapter Introduction Research Background Background to the 2005 Accounting Change Risk Measurement Standard Deviation, CAPM, and the Book-to-Market Ratio 2.5.1 The Capital Asset Pricing Model	
2.1 2.2 2.3 2.4 2.5	Chapter Introduction	
2.1 2.2 2.3 2.4 2.5 2.6	Chapter Introduction Research Background Background to the 2005 Accounting Change Risk Measurement Standard Deviation, CAPM, and the Book-to-Market Ratio 2.5.1 The Capital Asset Pricing Model 2.5.2 Book-to-Market Ratio Book Delta to Market Delta	24 24 24 30 37 38 38 38 39 40
2.1 2.2 2.3 2.4 2.5 2.6 2.7	Chapter Introduction Research Background Background to the 2005 Accounting Change Risk Measurement Standard Deviation, CAPM, and the Book-to-Market Ratio 2.5.1 The Capital Asset Pricing Model 2.5.2 Book-to-Market Ratio Book Delta to Market Delta Book Delta to Market Delta and the Relative Delta Measure	24 24 24 30 37 38 38 38 39 40 42
2.1 2.2 2.3 2.4 2.5 2.6 2.7 2.8	Chapter Introduction Research Background Background to the 2005 Accounting Change Risk Measurement Standard Deviation, CAPM, and the Book-to-Market Ratio 2.5.1 The Capital Asset Pricing Model 2.5.2 Book-to-Market Ratio Book Delta to Market Delta Book Delta to Market Delta and the Relative Delta Measure Accounting Totals	24 24 24 30 37 38 38 38 39 40 42 43
2.1 2.2 2.3 2.4 2.5 2.6 2.7 2.8	Chapter Introduction Research Background Background to the 2005 Accounting Change Risk Measurement Standard Deviation, CAPM, and the Book-to-Market Ratio 2.5.1 The Capital Asset Pricing Model 2.5.2 Book-to-Market Ratio Book Delta to Market Delta Book Delta to Market Delta and the Relative Delta Measure Accounting Totals 2.8.1 Accounting Equation	
2.1 2.2 2.3 2.4 2.5 2.6 2.7 2.8	Chapter Introduction Research Background Background to the 2005 Accounting Change Risk Measurement Standard Deviation, CAPM, and the Book-to-Market Ratio 2.5.1 The Capital Asset Pricing Model 2.5.2 Book-to-Market Ratio Book Delta to Market Delta Book Delta to Market Delta and the Relative Delta Measure Accounting Totals 2.8.1 Accounting Equation 2.8.2 Total Shareholders' Equity	

	2.8.4	Total Liabilities	45
	2.8.5	Net Income	45
2.9	Accounting Quality		46
	2.9.1	Accounting Quality and Value Relevance	47
	2.9.2	Accounting Quality, Value Relevance and the Relative Delta	
		Measure	47
	2.9.3	Relationship Between Accounting Quality, Accounting Qual	ity
		Risk, Value Relevance, Difference Component, Relative Del	ta
		Measure, Explicit and Implicit Roles of Accounting, Financia	al
		Distress Risk and Information Bias	48
2.10	Basel	Risk Management Framework and Risk Measurement	49
	2.10.1	The Basel Capital Requirement Formula	50
	2.10.2	Basel Risk Management Framework and Value-at-Risk	52
2.11	The V	Value-at-Risk Measure	53
2.12	Value	e-at-Risk Models	54
	2.12.1	Variance-Covariance Model	56
	2.12.2	Historical Simulation Model	57
	2.12.3	Monte Carlo Simulation Model	59
	2.12.4	Stress Testing	61
	2.12.5	Backtesting	61
2.13	Conte	emporary Technology Framework	61
2.14	Sumr	nary	66
Chapter 3	Rese	arch Methodology	67
3.1	Chap	ter Introduction	67
3.2	Resea	arch Question	68
	3.2.1	Research Question Examination	69
	3.2.2	The Fama and French Book-to-Market Ratio and Financial	
		Distress Risk	70
	3.2.3	The Fama and French Book-to-Market Ratio and the Differen	nce
		Component	72
	3.2.4	The Relative Delta Measure	78
	3.2.5	Key Accounting Elements and the Difference Component an	d
		Relative Delta	79

3.3	Null	Hypothesis and Detailed Null Hypotheses
	3.3.1	Null Hypothesis
	3.3.2	Detailed Null Hypotheses
3.4	Rese	arch Hypothesis and Detailed Research Hypotheses
	3.4.1	Research Hypothesis
	3.4.2	Detailed Research Hypotheses
3.5	Main	variables
3.6	Samp	ple Selection Criteria
	3.6.1	Control Group Selection Criteria
	3.6.2	Sample and Control Group Selection
	3.6.3	Primary Sample Banks
	3.6.4	Primary Sample Constraint
	3.6.5	Secondary Sample
	3.6.6	Primary Sample Firm Selection
	3.6.7	Secondary Sample Firm Selection
	3.6.8	Sample Firms Surveyed
	3.6.9	Control Group Firm Selection
3.7	Infor	mation and Data Collection102
	3.7.1	Information Collection
	3.7.2	Market and Accounting Data Collection102
	3.7.3	Data Sources
	3.7.4	GDP Indicators Variable Data Collection
	3.7.5	General Market Indicator Variables Data Collection104
	3.7.6	Market and Accounting Variables Data Collection104
	3.7.7	Time Series Interval Selection
3.8	Intro	duction to the Null Hypothesis Tests and Analysis107
3.9	Null	Hypothesis Test Details using the Change in Key Accounting Totals
	and M	Market Price Return
	3.9.1	Time Series Regressions for Change in Key Accounting Totals and
		Market Price Return
	3.9.2	Relative Delta and Change in Key Accounting Totals and Market
		Price Return
	3.9.3	Regressions for Relative Delta and Historical Value-at-Risk Actual
		– Null Hypothesis Evaluation118
	3.9.4	Accounting Value-at-Risk119

	3.9.5	Relationship Between Accounting Value-at-Risk and Market Price
		Return Value-at-Risk and Relative Delta
	3.9.6	Regulatory Capital Estimates
	3.9.7	UK Bank Regressions for Relative Delta and Historical Value-at-
		Risk Actual
3.10	Sum	mary
Chapter 4	Anal	lysis and Results129
4.1	Chap	oter Introduction
	4.1.1	Samples and Control Group Setup130
	4.1.2	Chapter Organisation
4.2	Anal	ysis and Summary Results for Relative Percentage Change of
	Aver	rages for Before to/and After 2005
	4.2.1	Percentage Change Calculations for Summary Results
	4.2.2	Market Price Returns Before and After 2005 – Examination of
		Material Significance
	4.2.3	Average Value-at-Risk Levels Before and After 2005 –
		Examination of Material Significance145
	4.2.4	GDP and General Market Conditions Before and After 2005 –
		Examination of Statistical Significance149
	4.2.5	Selected Variable Totals Before and After 2005 – Examination of
		Material Significance
	4.2.6	Null Hypotheses Test using Relative Delta Before to/and After
		2005 – Examination of Material Significance 154
	4.2.7	Null Hypotheses Test using Regression Analysis for Relative
		Delta and Value-at-Risk Before to/and After 2005 - Examination
		of Statistical Significance158
4.3	Regr	essions for Change in Accounting Totals and Market Price Return
	4.3.1	Regressions for Change in Total Equity and Market Price Return
	4.3.2	Regressions for Change in Total Equity per Share and Market
		Price Return

	4.3.3	Regressions for Change in Total Assets and Market Price Return
	4.3.4	Regressions for Change in Total Liabilities and Market Price
	4.3.5	Regressions for Change in Net Income and Market Price Return 188
4.4	Sign	ificance of Differences for Change in Accounting Totals and Market
	Price	Return Regression Slope Coefficients
4.5	Rela	tive Delta Measures: Changes in Accounting Totals to Market Price
	Retu	rns
4.6	Regr	essions for Accounting Total to Market Price Relative Delta and
	Histo	prical Value-at-Risk Actual
	4.6.1	Regressions for Total Equity to Market Price Relative Delta and
		Historical Value-at-Risk Actual
	4.6.2	Regressions for Total Assets to Market Price Relative Delta and
		Historical Value-at-Risk Actual
	4.6.3	Regressions for Total Liabilities to Market Price Relative Delta
		and Historical Value-at-Risk Actual
	4.6.4	Regressions for Net income to Market Price Relative Delta and
		Historical Value-at-Risk Actual
4.7	Acco	ounting Value-at-Risk Estimates and Relationships
	4.7.1	Estimates for Accounting Value-at-Risk
	4.7.2	Relationship Between Accounting Value-at-Risk, Historical
		Value-at-Risk Actual and Relative Delta
4.8	Rela	tive Delta Measures for UK Banks
4.9	Estin	nates of Accounting Value-at-Risk for UK Banks
4.10	Regu	latory Relative Delta Framework: Estimates for Minimum
	Regu	latory Capital for UK Banks
4.11	Regu	latory Capital Measurement and the Technology Framework . 272
4.12	Sum	mary
Chapter 5	Con	clusion
5.1	Conc	clusive Statement Analysis

5.2	Extension to the Fama and French Time Dependent Treatment for the		
	Book-to-Market Ratio	292	
5.3	Fama and French Baseline Regression and its Inference to this Sta	udy	
5.4	Regulatory and Firmwide Technology Framework Findings	292	
5.5	Policy Implications	294	
5.6	Managerial Implications	296	
5.7	Technology Implications	296	
5.8	Statistical Implications	297	
5.9	Thesis Limitations	298	
5.10	Summary	299	
Chapter 6	Future Work	301	
6.1	Chapter Introduction	301	
6.2	Recommendations for Future Work	301	
6.3	Summary	307	
Chapter 7	References	308	
Appendix A	The Fama and French Time Dependent Book-to-Market Ratio		
	Treatment and the Relative Delta	321	
Appendix B	Book Delta and Accounting Delta to Market Delta	323	
Appendix C	The Relative Delta Measure	326	
Appendix D	Relationship Between Financial Distress Risk, Difference Com	ponent	
	and Relative Delta	333	
Appendix E	Accounting to Market Price Relative Delta Definitions	335	
Appendix F	Approach for Change in Accounting Totals and Market Price F	Return	
	Regressions	340	
Appendix G	Approach for Accounting to Market Price Relative Delta and V	/alue-	
	at-Risk Regressions	347	
Appendix H	Derivation and Proof of Accounting Value-at-Risk	349	
Appendix I	Accounting Value-at-Risk	352	
Appendix J	Accounting Value-at-Risk Market Price Value-at-Risk and Re	lative	
	recounting value at rush, market rice value at rush and re	iuti v e	

Appendix K	Regulatory Capital and Regulatory Value-at-Risk
Appendix L	Adjustment to the Minimum Basel Regulatory Capital for UK Banks
	using the Regulatory Relative Delta
Appendix M	Quantitative Analytical Approaches Applied in this Study
Appendix N	Value-at-Risk Implementation Details
Appendix O	Quantitative Analytical Methods Implementation Details
Appendix P	Analysis and Results for Market Price Returns, Value-at-Risk, GDP
	and General Market Conditions
Appendix Q	The Fama and French Baseline Regressions
Appendix R	Analysis and Results for Book-to-Market Ratio, Key Market Totals,
	Key Accounting Totals and Historical Value-At-Risk Actual 562
Appendix S	Analysis and Results for Changes in - Book-to-Market Ratio, Key
	Market Totals and Key Accounting Totals579
Appendix T	Analysis and Results for Change in Market Value and Market Price
	Return Regressions
Appendix U	Analysis and Results for Change in Accounting and Change in Market
	Price Cross-Sectional Regressions
Appendix V	Analysis and Results for Accounting to Market Price Relative Delta
	and Value-at-Risk using Cross-Sectional Regressions for Samples and
	Control Group and using Time Series Regressions for UK Banks
Appendix W	Primary Sample and Secondary Sample Data
Appendix X	Control Group Data
Appendix Y	Capital Events that Adjust Market Price720
Appendix Z	Calculated and Omitted Data

LIST OF TABLES

Table 3.1	Main Variable Selection	87
Table 3.2	Main Variable Specifications	88
Table 3.3	Banking Institutions Listed in the London Stock Exchange (LSE)	93
Table 3.4	Primary and Secondary Sample Selection Criteria	94
Table 3.5	Control Group Selection Criteria	96
Table 3.6	Primary and Secondary Sample Firms	97
Table 3.7	Control Group Firms	100
Table 3.8	Data Collection: Data Description and Data Sources	103
Table 3.9	Time Series Ranges	106
Table 3.10	Time Series Periods	107
Table 3.11	Accounting and Market Price Regulatory Relative Delta	125
Table 4.1	Summary Results: Relative Percentage Change of Averages for Market	
	Price Return Distribution and Variance-Covariance Value-at-Risk	
	Descriptive Statistics for Before To/And After 2005	140
Table 4.2	Summary Results: Relative Percentage Change of Averages for Value-	at-
	Risk Models for Before And After 2005	145
Table 4.3	Summary Results: Relative Percentage Change of Averages for Description	otive
	Statistics for the Totals and Historical Value-at-Risk Actual Variables f	or
	Before And After 2005	151
Table 4.4	Summary Results: Relative Percentage Change of Averages for the Rel	ative
	Delta Variable for Before To/And After 2005	154
Table 4.5	Summary Results: Regression Results for the Relative Delta Variable a	nd
	Historical Value-at-Risk Actual Variable for Before To/And After 2005	5
		159
Table 4.6	Time Series Regression Analysis for Change in Total Equity and Marko	et
	Price Return	164
Table 4.7	Time Series Regression Analysis for Change in Total Equity per Share	and
	Market Price Return	170
Table 4.8	Time Series Regression Analysis for Change in Total Assets and Marke	et
	Price Return	176
Table 4.9	Time Series Regression Analysis for Change in Total Liabilities and M	arket
	Price Return	182

Table 4.10	Time Series Regression Analysis for Change in Net Income and Market
	Price Return
Table 4.11	t-Test for Significance of Differences Within and Between Samples and
	Control Group for Change in Accounting Totals and Market Price Return
	Regression Slopes from Table 4.6 to Table 4.10
Table 4.12	Percentage Changes for Relative Delta Measure Results from Table 4.13
Table 4.13	Relative Delta Measures from Change in Accounting Totals and Market
	Price Return
Table 4.14	Time Series Regression Analysis for Total Equity to Market Price Relative
	Delta and Historical Value-at-Risk Actual
Table 4.15	Time Series Regression Analysis for Total Assets to Market Price Relative
	Delta and Historical Value-at-Risk Actual
Table 4.16	Time Series Regression Analysis for Total Liabilities to Market Price
	Relative Delta and Historical Value-at-Risk Actual
Table 4.17	Time Series Regression Analysis for Net Income to Market Price Relative
	Delta and Historical Value-at-Risk Actual
Table 4.18	Percentage Changes for Accounting Value-at-Risk Results from Table 4.19
Table 4.19	Estimates for Accounting Value-at-Risk from Relative Delta and Historical
	Value-at-Risk Actual
Table 4.20	Relationship Between the Difference in Accounting Value-at-Risk and
	Historical Value-at-Risk Actual to Relative Delta
Table 4.21	Percentage Changes for Relative Delta Measure Results for UK Banks from
	Table 4.22
Table 4.22	Relative Delta Measures for UK Banks
Table 4.23	Percentage Changes for Accounting Value-at-Risk Results for UK Banks
	from Table 4.24
Table 4.24	Estimates of Accounting Value-at-Risk from Relative Delta and Historical
	Value-at-Risk Actual for UK Banks
Table 4.25	Adjusted Minimum Basel Regulatory Capital from Total Equity for UK
	Banks
Table 4.26	Percentage Changes for Regulatory Relative Delta Results for UK Banks
	from Table 4.27

Table 4.27	Regulatory Relative Delta Framework: Estimates for Minimum Regulatory
	Capital for UK Banks from the Maximum Absolute Relative Delta Adjusted
	to Specified Threshold Levels
Table 5.1	Conclusions for Market Price Returns, Market Price Return Volatility and
	Market Price Return Value-at-Risk
Table 5.2	Conclusions for GDP and General Market Conditions to the Samples and
	the Control Group Market Price Returns and Value-at-Risk
Table 5.3	Conclusions for Null Hypothesis Tests Based on Accounting Totals283
Table 5.4	Conclusions for Null Hypothesis Tests Based on the Change in Accounting
	Totals to Change in Market Price
Table 5.5	Conclusions for Null Hypothesis Tests Based on Accounting to Market
	Price Relative Delta
Table 5.6	Conclusions for Null Hypothesis Tests Based on Accounting to Market
	Price Relative Delta for UK Banks
Table 5.7	Conclusions for Null Hypothesis Tests Based on Relative Delta and Market
	Price Value-at-Risk
Table 5.8	Conclusions for Null Hypothesis Tests Based on Relative Delta and Market
	Price Value-at-Risk for UK Banks
Table 5.9	Conclusions for Null Hypothesis Tests Based on Accounting Value-at-Risk
	and Market Price Value-at-Risk
Table 5.10	Conclusions for Null Hypothesis Tests Based on Accounting Value-at-Risk
	and Market Price Value-at-Risk for UK Banks
Table 5.11	Conclusions for Null Hypothesis Tests Based on Regulatory Relative
	Relative Delta for UK Banks

LIST OF FIGURES

Figure 1.1	Plots of 1-Day Market Price Return and 1-Day Value-at-Risk for UK Banks
	from 2004 to 2009
Figure 1.2	Plots of 1-Day Market Price Return and 1-Day Value-at-Risk for UK Banks
	from 1995 to 2009
Figure 1.3	Plots of 30-Day Market Price Return and 30-Day Value-at-Risk for UK
	Banks from 1995 to 2009
Figure 1.4	Plots of 250-Day Market Price Return and 250-Day Value-at-Risk for UK
	Banks from 1997 to 2009
Figure 1.5	Plots of 500-Day Market Price Return and 500-Day Value-at-Risk for UK
	Banks from 2000 to 2009
Figure 2.1	Bank Technology Framework Overview
Figure 4.1	Proposed Bank Technology Framework Overview

Chapter 1 Introduction

The role financial accounting plays in risk management may be contended as being intricate in nature, in the sense that accounting may find itself playing two roles. The first role, by definition, is the generally accepted explicit role where accounting methods, specified by accounting standards, are applied to report information that can help investors manage potential exposure to financial risk. The accounting methods classified under this category would include the approaches: fair value accounting; deferral accounting; and, accrual accounting (Ongkrutaraksa 1999). The American Bankers Association (ABA) President and CEO (from 2005 to 2010) Edward Yingling expresses this explicit role to be analogous to the metaphor that accounting policy is accurately *measuring the temperature of a flame* (Leone 2008). In this metaphorical context, the flame's *temperature* is considered in this study to be analogous to a *firm's value*.

The second role accounting plays is a more subtle implicit role and is intrinsic to the first role. This is where the accounting policy applied to help facilitate economic decision making that aim to manage and mitigate financial risk, may inadvertently report *biased* information. This *biased information*, its level specified in this study as a measure of exposure to *accounting quality risk*, could potentially have the effect to introduce *financial risk* that is exogenous to the normal functioning of the financial system. Expressed by Yingling this implicit accounting role would be analogous to the metaphor that accounting policy is *adding fuel to a fire, and thus increasing the temperature of the flame that it's measuring*¹ (Leone 2008). This implicit accounting policy is *reducing fuel to a fire, and thus decreasing the temperature of the flame that it's measuring* the temperature of the flame that it second fuel to the difficulty in measuring its effects (IASBI 2008), may be contended as being overlooked within the contemporary

¹ Reported by Leone (2008), Yingling's statement with regards to *adding fuel to a fire*, contextually refers to outside influences adding to or *distorting* the temperature of a flame from its true level. That is, for example, the accounting standards applied may overstate or understate a valuation for a firm when compared to its *true value*. Yingling's statement may also be interpreted as referring to the market valuation of a firm, such as its market price, becoming exposed to increased levels of volatility than would otherwise be experienced if another accounting approach was applied.

firmwide risk management and the regulatory risk measurement frameworks, including the *Basel risk management framework* (Basel 2011).

This research is concerned with the interrelated nature of both the explicit and the implicit accounting roles. However, this study aims to determine accounting policy performance from the basis of accounting's explicit role. If deviations from this normal explicit function are exhibited, even during volatile market conditions, then this study considers such a period of observed deviation as evidence of accounting's implicit role.

As its focus, this study takes the before and after effects arising from the important 2005 accounting standards change on the five major banks registered with the London Stock Exchange's (LSE's) United Kingdom (UK) banking sector². The 2005 accounting change witnessed UK banks and other firms throughout Europe adopt the International Accounting Standards Board (IASB) framework's internationally focused accounting standards. The IASB's accounting standards are entitled the International Financial Reporting Standards (IFRS)³. The 2005 accounting change⁴ has become the focus of much recent debate as one of the most significant financial accounting changes in modern history (Armstrong, Barth, Jagolinzer, and Riedl 2010).

This study examines how well the function of accounting played its explicit role after the 2005 accounting change when compared with its function before the change. The role this change played in reducing financial risk is examined by approaches that include comparing the levels of market price return Value-at-Risk and accounting Value-at-Risk for before and after the accounting change. In the process of its

² This study categorises banks to be in the UK banking sector if they are listed in the London Stock Exchange (LSE) *Banks* Industry Sector. As at January 2012 there are five UK banks listed in the LSE's Banks sector. The five banks are presented in Table 3.3 and are grouped and referred in this study also by the terms: *banks*, *UK banks*, *banks in the UK banking sector* and the *Primary sample*. LSE information sourced from: http://www.londonstockexchange.com [accessed 15th May 2012].

³ This study makes reference to the IASB framework's accounting standards also as the IASB accounting standards and the IFRS accounting standards.

⁴ The 2005 accounting standards referred to in this study are based on the 2005 financial statements published by the UK banks listed in the LSE (HSBC 2006, Barclays 2006, RBS 2006, Lloyds 2006, Standard Chartered 2006). In addition, the IFRS official accounting standards pronouncements issued on the 1st of January 2008 and 2009 (IASB 2008, 2009) are also referenced. Other than changes to implementation dates for some IFRS accounting standards, the referenced pronouncements maintain a faithful representation of the IFRS accounting standards applied from 2005 to 2009. Any changes to the IFRS pronouncements after 2005 and during 2009 are not considered significant to the findings reported in this study.

investigation, this study develops a measurement system for the currently un-recognised effects from the implicit role that accounting may have played within the firmwide and regulatory risk management frameworks. This measurement system is developed for the firmwide risk management framework by extending the Fama and French (2008) time dependent treatment of the book-to-market ratio. This system is developed with the aim to report the *difference* between accounting totals and market price change variables. In addition, it is developed to report levels of exposure to financial distress risk and to provide a direct measure of the levels of exposure to accounting quality risk. The measurement system developed in this study, for reasons of brevity, is termed the relative delta. The details of which will be discussed in the later sections. The measurement at the regulatory level extends this relative delta measure to produce the regulatory relative delta. The regulatory relative delta measure is developed with the objective to extend the current Basel III (Basel 2011) bank regulatory framework to adjust its minimum capital requirement to effects that may arise from accounting's implicit role. In addition, this study addresses the limitations of the current technology framework to implement such a measurement system.

This study's methodological approach is guided by the hypothesis based deductive methodology, termed the hypothetico-deductive methodology, and the falsification theory⁵ (Popper 1992). Based on a research question, this study specifies a null hypothesis and applies its methodological approach to establish evidence to verify or falsify its null hypothesis⁶. The research question this study asks and the null hypothesis this study tests are both introduced in Sections 1.2 and 1.3 respectively. To test the null hypothesis, material and statistical significance is examined for market price and accounting variables for banks in the UK banking sector for before and after the 2005 accounting change, within the time range 1992 to 2009. Furthermore, the Gross Domestic Product variables and general market variables are tested to deduce general economic and market conditions during the same time range. In order to test the selected variables for material and statistical significance, this study applies a number of

⁵ Hypothesis based deduction combined with the falsification theory is a scientific method that was first formally prescribed by Professor Sir Karl Raimund Popper (Popper 1992).

⁶ The methodological approach this study applies is guided by the falsification theory that aims to build evidence to falsify a set hypothesis. This study also discusses aspects of building evidence to verify or *corroborate* its null hypothesis. Although, this study discusses both aspects of verifying and falsifying the null hypothesis, the tests conducted and the conclusions arrived at are guided by the level of evidence that falsifies the null hypothesis. This approach is consistent with the falsification theory (or the falsification methodological approach) prescribed by Popper (1992).

quantitative analytical methods. These methods include descriptive statistics, correlation analysis, cross-sectional and time series regression, and relative percentage change analysis. The variables tested include market price return, Value-at-Risk (VaR), total shareholders' equity, total assets, total liabilities, net income, the developed relative delta and regulatory relative delta measures.

1.1 Background

Under European Community (EC) regulations, the five large banking institutions registered in LSE's UK banking sector in 2005 observed a change in accounting standards. Before the change, banks applied the accounting standards accepted under the UK GAAP⁷ (United Kingdom Generally Accepted Accounting Practice). The accounting standards accepted under the UK GAAP are the standards issued by the Accounting Standard Board (ASB). The accounting standards applied by these banks after 2005, and for the subsequent years, were the IASB accounting standards.

The purpose of the 2005 accounting change was to help improve and strengthen the efficient functioning of European capital markets (EC 2002, IASB 2009). With this aim, firms in the UK and throughout Europe changed from regionally developed sets of accounting standards accepted under the local GAAP (Generally Accepted Accounting Practice), to the IASB's IFRS accounting standards. The major problem that has arisen regarding this accounting change is that, on the one hand the EC regulations endorsed the accounting change in 2005 to improve capital market functioning, and on the other hand, Platikanova and Nobes (2006), Paananen and Lin (2009) and Morais and Curto (2008) reveal that contrary to the European Community and IASB expectations, samples of firms in Europe that adopted these new standards in 2005, soon after exhibited increased levels of market price volatility. In addition, their findings reveal that the adopting firms examined exhibit evidence of decreased accounting data value relevance resulting in lower accounting quality. Section 2.9 presents the approach applied in this study to characterise accounting quality⁸ and quantify value relevance.

⁷ The term GAAP may also be referred to as Generally Accepted Accounting Principles and Generally Accepted Accounting Policies.

⁸ In order to maintain continuity with current literature, this thesis makes reference to the term *accounting quality*. However, the level of *accounting quality* may be translated as a firm's level of exposure to *accounting quality risk*. Where, *high accounting quality* translates to a *low level of exposure to accounting quality risk*. Conversely, *low accounting quality* translates to a *high level of exposure to accounting quality risk*.

With the aim to examine the levels of volatility exhibited by the UK banks during and after the 2005 accounting change, Figure 1.1 plots the stock market price returns from 2004 to 2009 for the banks in the UK banking sector. Figure 1.1 generally shows that the level of market price return volatility increases and Value-at-Risk levels become breached during March and May 2006 and during March 2007. Figure 1.1 also exhibits increasing volatility levels and significant Value-at-Risk breaches near the start of July 2008 and then continuing into the first half of 2009.



Figure 1.1 Plots of 1-Day Market Price Return and 1-Day Value-at-Risk for UK Banks from 2004 to 2009

Plot of the 1-day market price returns average for the banks in the LSE's UK banking sector (solid line). Plot of 1-day Value-at-Risk at the 95% confidence level (dashed line), from January 2004 to November 2009, using Monte Carlo Simulation Value-at-Risk modelled from 800 1-day historical market price returns with 1 million and one simulations.

Figures 1.2 to 1.5 provide, for the same LSE registered banks, charts with average Value-at-Risk plots for selected time horizons with corresponding market price returns. They also show that the banks, for the 2005 to 2009 time range, experienced market price return volatility. In addition, the plots show that there was volatility before 2005. However, these plots indicate that volatility levels were more significant after 2005 when compared to before. Examining Figures 1.1 to 1.5, purely on the basis to inform the firmwide risk management and banking regulatory frameworks, it would be difficult to argue against such a significantly gradual increase in volatility levels after March 2006 from being subject to further investigation.





Plot of the 1-day market price returns average for the banks in the LSE's UK banking sector (solid line). Plot of 1-day Value-at-Risk at the 95% confidence level (dashed line) and at the 99.9% confidence level (dotted line), from July 1995 to November 2009, using Monte Carlo Simulation Value-at-Risk modelled from 800 1-day historical market price returns with 1 million and one simulations.





Plot of the 30-day market price returns average for the banks in the LSE's UK banking sector (solid line). Plot of 30-day Value-at-Risk at the 95% confidence level (dashed line) and at the 99.9% confidence level (dotted line), from October 1995 to November 2009, using a Monte Carlo Simulation Value-at-Risk modelled from 800 30-day historical market price returns with 1 million and one simulations.





Plot of the 250-day market price returns average for the banks in the LSE's UK banking sector (solid line). Plot of 250-day Value-at-Risk at the 95% confidence level (dashed line) and at the 99.9% confidence level (dotted line), from July 1997 to November 2009, using a Monte Carlo Simulation Value-at-Risk modelled from 800 250-day historical market price returns with 1 million and one simulations.





Plot of the 500-day market price returns average for the banks in the LSE's UK banking sector (solid line). Plot of 500-day Value-at-Risk at the 95% confidence level (long dashed line), at the 99.9% confidence level (dot-dash line), at the 95% confidence level (dashed line) and at the 99.9% confidence level (dotted line), from January 2000 to November 2009, using a Monte Carlo Simulation Value-at-Risk modelled from 500 500-day and 800 500-day historical market price returns with 1 million and one simulations.

1.2 Research Problem

Observing Figures 1.1 to 1.5, as stated earlier, it is evident that significant levels of volatility are exhibited after the 2005 accounting change. This study investigates the problem of these increased levels of volatility, with the important aspect that such levels being contrary to the previously stated EC (2002) and IASB (2009) expectations. Specifically this study expects, at a minimum, market prices to follow the *semi-strong form* of the efficient market model (Fama 1965, 1970). That is to say, this study expects stock market price information to reflect accurately published accounting information. Within the context of risk management, this view is consistent with financial accounting deduced to having played the explicit accounting role.

It is noted that the levels of volatility shown in Figure 1.1 after July 2008 has infamously been attributed to the financial market conditions during September 2008, pertaining to the *visible* and well publicised *Chapter 11* filing by Lehman Brothers Holdings. The wider debate that has taken place recently has focused on the *invisible* cause of the volatility, and that the 2008 volatility is as a consequence to a complex sequence of cause and effect reactions arising from mortgage based derivative product trading and sales activity. These reactions resulting in individual bank and insurer book positions in these *collateralised* derivative products becoming exposed to uncertainty, and subsequently caused the capital markets to react to these exposures in the form of market price volatility.

Although, concerns had also been aired and evidence exhibited from European firms, that the application of the new accounting standards in 2005 would create volatility in the financial markets, it is difficult to decipher the specific role accounting took in the observed UK bank volatility shown in Figure 1.1. As stated previously, the general view this study takes is that accounting played the normally assumed explicit role and reported the observed volatility. Importantly, accounting's explicit role is expected to report the increase in volatility levels exhibited during 2008. It is from this view that the study examines deviations from this explicit role, and the level of these deviations quantifying the level of accounting's implicit role. This study deduces that if the level of *deviation* is significant, that is, if the level of the implicit role is significant, then the

level of investor uncertainty would increase with the resulting financial market volatility.

From this basis, several explanations may be considered for the cause of the significant levels of volatility exhibited after 2005 in Figures 1.1 to 1.5. The first explanation, especially for before the second half of 2008, is that it is due to *normal* capital market activity and has no connection with the 2005 accounting change. The second explanation is that the accounting change provided a more transparent view of the bank's accounting figures with subsequent investor reactions creating the observed volatility. The first and second explanations are as a consequence of the explicit role of accounting process applied after 2005 presented information that when compared to other available information is found to be *unexpected* or *biased*. This explanation is consistent with the implicit role of accounting with an insignificant level of influence from the explicit role. The fourth explanation is that an interrelated combination of the three above explanations has taken effect at differing levels.

As stated previously, this study approaches the investigation for the observed volatility from the basis of the accounting change. The previously stated EC (2002) and IASB (2009) expectations for the new accounting standards to exhibit a purely explicit role and *report only the temperature of the flame*, is first investigated by deducing if the observed volatility is purely as a consequence of normal capital market activity. Maintaining the remit for the accounting function as a purely explicit role, this study aims to deduce how effectively the accounting data reflected the market price variations. It is expected that closer the accounting data reports the observed market price changes, the more relevant and reliable would be the accounting data and therefore the higher the accounting quality than otherwise.

Aligned with the EC (2002) and IASB (2009) expectations, this study expects that after the 2005 accounting change there will be an increase in value relevance. It is expected that even during periods of increased levels of financial market price volatility that the new accounting standards remit will be evident by it exhibiting a purely explicit role; reporting results that reflect stock market prices even during volatile periods. Ideally, it is expected for the banks to show an increase in the level of value relevance, and thus higher *accounting quality*, after the 2005 accounting change when compared to before. It is important to note that this study specifies and performs tests (see Chapter 3), and examines its results (see Chapter 4) and concludes its findings (see Chapter 5) based on evidence produced using the time variable. Specifically, the results are produced on a yearly basis and also based on three time periods. The three time periods are: 1992 to 2004 that produces results for before the 2005 IASB accounting change, and 1992 to 2007 and 1992 to 2008 that produces results for after the accounting change (see Chapter 3, Section 3.7.7). The latter two time periods examine the years 2007 and 2008 that exhibit increasing levels of market price volatility (see Figure 1.1).

This analysis approach, based on yearly and time range periods, provides a control for *time* by using the *time* variable, and thus effectively also provides a control for periods of observed market price volatility. As stated earlier in this section, during increased levels of market price volatility, it is expected that accounting information will exhibit an explicit role and reflect this increase in volatility (EC 2002, IASB 2009). Therefore, the results presented in this study that examine accounting quality, expects at a minimum, accounting quality levels to be similar after the 2005 accounting change compared to before. This expectation is maintained in this study even during periods of increased levels of financial market volatility as indicated from 2007 to 2009 in Figure 1.1.

To progress with this investigation, at this juncture it is imperative that the general methodological measurement approach applied in this study to measure accounting quality, and thus provide for a firm a measure of its level of exposure to accounting quality risk, is carefully introduced.

The general premise that this study applies in its investigation is that the accounting totals: total shareholders' equity, total assets, total liabilities and net income, provide information as to a measure of a firm's value from the financial statement's perspective. This study considers the accounting totals to provide a measure of the metaphorical *temperature of the flame*, where, as stated previously the *flame* is considered analogous to the *firm's value*. The general expectation is that the accounting standards applied by a firm in the preparation of its financial statements, and principally from an examination of the accounting totals, would provide an investor with a measure of the firm's value that is *accurate* and free from *bias*. In addition, the measure of the *change in an accounting total* for a given time period, termed in this study as the *accounting delta*,
provides time relative information to quantify the accounting related growth of the firm's value over time. This study applies this *accounting delta* measure as its *accounting valuation* for the banks. However, this does not cover the full extent of the proposed premise.

In applying such an approach, a methodological question arises that asks: how is it possible to determine if the *accounting delta* reports a firm's value that is accurate and free from bias? Extending the Fama and French (2008) treatment of the book-to-market ratio, this question is addressed by introducing a second valuation for the same firm reported from the financial market's perspective. This valuation being simply the firm's stock market share price, termed the *market price*. This study considers the market price to be metaphorically analogous to the measure of the *true temperature of the flame*. In order to scale a firm's market price to its accounting delta, the change in market price is measured for the same time period applied to evaluate the accounting delta. The change in market price is termed the *market price delta* or the *market price return* measure or simply as the *market delta*.

Now, measuring for a firm, the change of one of its accounting totals, evaluated by the *accounting delta*, and comparing that change with the change in the market price, evaluated by the *market price delta*, in theory would provide investors with a measure of the firm's information bias, or more accurately its *valuation bias*⁹. The *valuation bias* is specified as the difference between the accounting valuation (*accounting delta*) and the financial market valuation (*market price delta*). This *valuation bias* is termed in this study for brevity as the *difference* or the *difference component*¹⁰ (see Sections 3.2.3 and 3.2.5 that detail the measurement of the *difference component*). Thus, the greater the level of *difference* between the accounting delta and the market price delta, the greater the firm's accounting and market price *valuation bias*. Conversely, the lower this level of *difference*, the lower the firm's *valuation bias*.

Directing focus on the *delta* measures, this study terms the *difference* between the accounting delta and the market price delta as the *accounting totals to market price*

⁹ The term *valuation bias* applied in this study refers to *information bias* unless detailed otherwise.

¹⁰ The term *difference* or *difference component* applied in this study represents a vector measure (measuring the magnitude and direction) calculated from the difference between one variable moving in one direction and another variable moving in another direction.

relative delta or, as the *accounting to market price relative delta*. This level of *difference*, measured by the *difference component*, now simply translates to the level reported by the previously introduced *relative delta measure*.

Based on this approach, specifying the level of *difference* to be a measure of value relevance and thus *accounting quality*, it would be reasonable to expect the level of *difference* to be lower for higher quality value relevant accounting data than otherwise. Applying this same reasoning to determine accounting quality for the accounting change, it would be reasonable to expect that the level of *difference* for the adopting banks is the same or less (tending to zero) after 2005 when compared to the *difference* before. In a metaphorical sense, it is expected that the accounting totals would provide a more accurate measure of the *flame's temperature* after the 2005 change when compared to before. Even during volatile market behaviour, such an expectation would be close to the ideal sought by the EC regulations and the IASB accounting framework. On this basis, this study directs its research question to ask:

Does accounting quality improve for UK banks that adopted the IFRS accounting standards in 2005? That is, does the difference between the change in accounting totals and the change in market price decrease for UK banks that adopted the IFRS accounting standards in 2005 when compared to before the adoption?

1.3 Overview of the Research Methodology

In order to achieve this study's aims and objectives in accordance with the research question, the following null hypothesis is tested:

Accounting quality in UK banks was not affected by the adoption of the IFRS accounting standards in 2005. For the UK banks that adopted the new accounting standards in 2005, there is no significant difference between the change in accounting totals and the change in market price after 2005 when compared to before.

To test the null hypothesis this study applies the earlier declared hypothesis based deductive methodology, combined with the falsification theory (Popper 1992).

Within this methodological framework, the null hypothesis is developed into four

detailed null hypotheses. The *detailed null hypotheses* test the *difference between the change in accounting totals and the change in market price* by applying the relative delta measure for the accounting totals: total shareholders' equity, total assets, total liabilities and net income with the market price (see Section 3.3.2). The aim being to gather evidence from the four *detailed null hypotheses* to provide strength to either verify or falsify the null hypothesis.

To test the *detailed null hypotheses* the sample of the five main banks listed in the LSE bank sector, termed the *Primary sample*, is surveyed and data collected from 1992 to 2009. Market price data is collected on a daily basis, and accounting total information is collected on a yearly basis.

With the objective to increase the Primary sample a second sample, termed the *Secondary sample*, is selected and surveyed. The Secondary sample consisted of 28 banking related financial services firms listed in the LSE that also changed to the IASB accounting standards in 2005. After quality assurance and data refining, the number of firms tested in the Secondary sample comprised 11 firms. The tests that are applied to the Primary sample are also applied to the Secondary sample, and also to a sample that consists of both the primary and the secondary sample firms, termed the *Primary and Secondary sample*.

As a control for the 2005 accounting change, 20 banking related financial services firms listed in the LSE, termed the *Control group*, are selected and surveyed. The important contrast between the *Primary sample firms* and the *Secondary sample firms* and the *Control group firms* is that the Control group firms did not adopt the new accounting standards in 2005. After quality assurance and data refining the number of firms tested in the Control group consisted of 12 firms. With the objective to determine evidence of materially and statistically significant reactions before and after the 2005 accounting change, the tests applied to the Primary sample firms and the Secondary sample firms are applied to the Control group.

To support this research's null hypothesis, it is expected that the tests applied to the Primary sample and the Secondary sample firms would produce results that exhibit similar levels for the *difference component* after the 2005 accounting change compared

to before. In addition, the samples are expected to exhibit similar, or ideally lower, *differences* compared to the Control group after the 2005 accounting change.

Guided by this general analysis of results that is based on sample reactions for before and after the 2005 accounting change, and the sample reactions compared to the control group, this study proceeds as follows.

For the Primary sample, the Secondary sample, the Primary and Secondary sample and the Control group this study gathers evidence based on the market price returns and Value-at-Risk reactions for before and after 2005. First, for the samples and the control group¹¹ market price return and distribution characteristics are analysed together with the variance-covariance Value-at-Risk levels. Then, the Value-at-Risk levels are measured by applying the three primary VaR models: variance-covariance, historical and Monte Carlo simulation. From these measurements, the historical Value-at-Risk model is tested and the Historical Value-at-Risk Actual measure defined. Unless otherwise specified, for the remainder of the tests, the Historical Value-at-Risk Actual that is calculated using a *zero-day* time horizon is applied to represent this study's measure of *market price Value-at-Risk*.

To determine the market price return and Value-at-Risk reactions of the samples and the control group to economic conditions, the Gross Domestic Product (GDP) indicator is selected and its levels are analysed using time series regression. To determine reactions to general market conditions selected stock market indices, selected short-term interest rates, selected treasury Government bond rates and selected foreign exchange rates are tested with the samples and the control group market price returns and Value-at-Risk, again using time series regression.

After this series of tests, the study proceeds by introducing the accounting totals. The accounting totals are tested to determine levels of material change before and after 2005 compared to both the market price levels and Value-at-Risk levels. In addition, statistical significance is evaluated using correlation analysis. Then the change in

¹¹ The term 'samples' is used in this study to reference the three samples: the Primary sample, the Secondary sample and the Primary and Secondary sample. The terms 'samples and control group', 'the samples and the control group', 'samples' and 'control group' are at times contextually applied in this study primarily as technical terms.

accounting totals (accounting *delta*) is tested with the change in market price (market price *delta*) and Value-at-Risk to determine material and statistical significance, where statistical significance is again evaluated using correlation analysis.

The accounting delta and market price delta are then analysed using cross-sectional and time series regressions to test for statistically significant reactions for before and after the 2005 change. The study then proceeds to test and compare the levels of *difference* produced from the accounting to market price relative delta measure for before and after 2005. The object of this test is to determine the level of material significance using the levels of the *difference component*. The relative delta is also analysed for each bank in the Primary sample to determine the individual bank's accounting to market price reactions for before and after the accounting change. Importantly, this latter test is performed with the objective to provide direct evidence to strengthen or weaken the detailed null hypotheses.

The relative delta measures are then tested with Value-at-Risk using time series and cross-sectional regressions to determine the statistical significance for the detailed null hypotheses. Time series regressions are also used to determine each individual bank's accounting to market price relative delta reactions to Value-at-Risk. Given the nature of the relative delta measure, it would be expected that to verify a single detailed null hypothesis, a statistically significant reaction between the corresponding relative delta measure and Value-at-Risk after the 2005 accounting change would not be evident.

The *relative delta measure* and the *market price Value-at-Risk* are evaluated to provide an estimate for the *accounting Value-at-Risk* levels. The study then examines Value-at-Risk levels for specified time periods by comparing the *accounting Value-at-Risk* estimates with the *market price Value-at-Risk*. The expectation would be that both the accounting Value-at-Risk and market price Value-at-Risk would report similar levels for before the 2005 change when compared to after the change. The accounting Valueat-Risk and market price Value-at-Risk levels are also analysed at the individual bank level. This analysis is performed to determine if the Value-at-Risk remain at equivalent levels for the banks before and after the accounting change.

Finally, the *regulatory relative delta* is evaluated for each bank. The *regulatory relative delta* is computed based on a series of threshold levels to determine the level of

adjustment applied to the Basel (2011) 8% minimum regulatory capital requirement¹². The adjusted levels are then analysed for before and after 2005 to determine whether the minimum capital has adjusted up or down. If improved accounting quality is to be evident after 2005 then this study expects that the adjusted minimum capital requirement levels for the banks to be less after 2005 compared to before.

1.4 Thesis Organisation

This thesis, in Chapter 2, provides a literature review that introduces its motivations and previous research conducted in the area of this study. The background to the 2005 accounting change is introduced with its motivations and aims. The studies that have tested the accounting change before the adoption event and its effects during and after this event are introduced. Literature surrounding risk management is presented from the point of view of developing the characteristics of the relative delta measure with reference to the Fama and French (2008) time dependent treatment of the book-tomarket ratio. The financial statement accounting totals measured in this thesis are introduced. The relationships between accounting quality, value relevance, the relative delta measure and other key measures applied in this study are then formally introduced. Chapter 2 then proceeds to introduce the Basel risk management framework from the perspective of risk weighted assets and its application of the Value-at-Risk risk measurement approach. The Basel framework is referenced as a prelude to introducing the *relative delta* measure in a regulatory capital context, referred to as the *regulatory* relative delta measurement framework. The variance-covariance, historical and Monte Carlo simulation Value-at-Risk approaches that are applied in this study are then discussed in detail. The connection between Value-at-Risk and stress testing, and backtesting are introduced. The chapter then proceeds to discuss the technology requirements necessary to implement the relative delta measure and the regulatory relative delta measure as well as the accounting Value-at-Risk measure.

Guided by the hypothetico-deductive falsification methodological approach, Chapter 3 details the research methodology applied in this study. Chapter 3 proceeds by detailing and examining the research question. The null hypothesis tested in this study is

¹² From 1992 to 2009, the maximum time range applied for analysis in this study, the total minimum capital requirement under the Basel I and Basel II regulatory frameworks was at 8% (Basel 1998, Basel 2006). This 8% minimum total capital requirement remains in effect for the Basel III framework from 2013 until 2019 (Basel 2011).

specified. With the aim to provide a framework to test the null hypothesis the set of *detailed null hypotheses* referred to earlier in this chapter are also specified. Then the sample selection criteria are detailed, and the banks and firms selected for the Primary sample, the Secondary sample and the Control group are presented. The data collection and time range criteria are then specified. Chapter 3 then proceeds to detail the quantitative and analytical methods applied in this study to test the null hypothesis.

Chapter 4 presents the analysis and results for the samples and the control group. This chapter generally presents relative percentage changes based on the analysis and results for before and after the 2005 accounting change. The first part of Chapter 4 presents analysis and summary results based on detailed results for the market price return, Value-at-Risk and the relative delta measure. In addition, an analysis of summary results for tests based on GDP and general market conditions is presented. The second part of the chapter presents analysis and detailed results based on the *change in accounting totals, change in market price* and the *relative delta*. Analysis and detailed results for the banks in the Primary sample are presented based on the regulatory relative delta measure. Chapter 4 concludes by proposing a bank technology framework to enable the measurement of the *relative delta, regulatory relative delta* and *accounting Value-at-Risk*.

Chapter 5 presents this study's conclusion from the basis of a series of conclusive statements that test the level of verifiability or falsifiability of the null hypothesis. In addition, the development and application of the relative delta and regulatory relative delta measures are noted as an extension to the Fama and French (2008) time dependent treatment of the book-to-market ratio. The Fama and French (2008) baseline regression test and its historical inference to this study are also presented. Thereafter, concluding discussions are presented regarding the technology framework proposed in this study to inform the financial system. Chapter 5 then presents implications that result from this thesis. Possible limitations to this thesis are then discussed. This chapter then concludes by presenting an axiom based on the findings from this research.

Chapter 6 proposes future work to extend this research, and future work that may be considered in this study's research area.

Chapter 7 provides the references for this research.

1.5 Thesis Contributions

This thesis proposes a number of distinct contributions to knowledge by extending the literature surrounding the 2005 accounting change, and extending banking regulatory policy. This research also proposes recommendations to extend firmwide risk management policy and financial accounting policy. In addition, recommendations are proposed to extend the banking technology framework.

Current research addresses the effects of the 2005 IASB accounting standards adoption event for European firms. This thesis extends this current research by focusing on the UK banking sector.

This study introduces a specification for an accounting Value-at-Risk measurement in Section 3.9.4, Appendix H and Appendix I. In Appendix K, it is proposed that this measure together with the market price Value-at-Risk measure contribute to the current regulatory and firmwide risk management practice. This study proposes that the measure of the accounting Value-at-Risk, when compared to the market price Value-at-Risk, provides a well tested framework to inform the risk management practice of *bias* between information from the accounting function and information sourced from the financial markets.

This study extends the Fama and French (2008) time dependent treatment of the bookto-market ratio by developing the *accounting to market price relative delta* measure. The relative delta measure is introduced in this chapter and specified in Section 2.7 and Appendix C and developed in Section 3.9.2, Appendix B and Appendix E. The relative delta measure is proposed in this study to measure for a firm or a portfolio of firms its level of accounting data value relevance, and thus its level of accounting quality. The relative delta measure is proposed to also measure for a firm or a portfolio its level of exposure to financial distress risk arising from the change in total shareholders' equity and the change in market price for a specified time period. This study then extends this measurement to address *financial distress risk* arising from the relationship between the change in market price and the change in the accounting totals: total assets, total liabilities and net income. In effect, this study proposes a system of relative delta measures that monitor and determine the level of exposure to financial distress risk for a firm or a portfolio of firms. Importantly, it is proposed that the relative delta measure is applied to provide information to direct investigation to equity, assets, liabilities or net income items to mitigate financial distress risk and accounting quality risk.

This study proposes to contribute to regulatory policy and to firmwide risk management practices by recommending the implementation of the system of relative delta measures at the regulatory and firm level. In keeping with the characteristics of the system of relative deltas, the purpose of implementing such a system would be to inform regulators and risk controllers, the level of a firm's exposure to financial distress risk, as well as accounting quality risk. In doing so, it would provide information to direct action to mitigate such risks. The regulatory relative delta, that is introduced and specified in Section 3.9.6, Appendix K and Appendix L, is also proposed in this study to contribute to the regulatory policy, specifically to the Basel regulatory framework. The regulatory relative delta is proposed to provide a mechanism to monitor changes to the minimum regulatory capital level. Section 3.9.6 shows that the *regulatory relative delta* extends the *relative delta* by specifying a special implementation of the system of *relative delta* measures.

The relative delta and regulatory relative delta measures are proposed in this study to inform the regulatory and firmwide risk management levels. However, it is evident that the measures may also be proposed to make a contribution to accounting policy, in that the measures are monitored by accounting standards authorities to determine how the financial markets react to the interpretation and application of adopted financial accounting standards.

This study proposes to contribute to the regulatory and banking technology framework. Section 4.11 and Figure 4.1 presents a proposed technology architecture to measure the accounting to market price relative delta, regulatory relative delta and the accounting Value-at-Risk. The technology framework is specified to enable the updating of accounting information on a real-time or close to real time basis. Such an architecture would provide accounting information at a frequency that is comparable to the *currently available* real-time market price updates.

This study proposes to also contribute to the regression analysis approach by introducing in Appendix M.5.2 a further measurement to assess the accuracy of the coefficient of determination R^2 . The coefficient R^2 for a regression model measures how

well the independent variable explains the dependent variable. This research determines that for a regression model to be accurate, the *independent variable* and the *residue variable* must be randomly distributed. Utilising the Pearson product-moment correlation coefficient, this research proposes and measures the correlation between the independent variable and the residue variable to evaluate the relationship's randomness. If the correlation coefficient is approximately between 0.5 and -0.5 then it is concluded that the R^2 measure is a valid measure for the regression model's strength. If the correlation coefficient is significantly above or below zero (greater than 0.5 or less than -0.5) the independent variable's and the residue variable's relationship may not be random, and therefore further study of the regression relationship may be considered. This research attempts to implement, with limited scope, this proposed correlation measure to determine the accuracy of the reported R^2 coefficient.

In addition, a contribution to the risk management practice is proposed by introducing a new algorithm and smoothing formula to calculate the standard normal inverse - cumulative distribution function (CDF). The Monte Carlo simulation Value-at-Risk measure is dependent on an algorithm to generate a *random* stochastic variable. This stochastic variable is primarily based on a random number between 0 (zero) and 1 and modelled using the standard normal inverse CDF. Recent work has determined that there is no closed form solution for the standard normal inverse CDF. Detailed in Appendix N.6.2 and N.6.3, this research specifies and develops, for the Monte Carlo simulation Value-at-Risk model, a new algorithm and smoothing formula to calculate the standard normal inverse CDF. This standard normal inverse CDF is developed to generate a stochastic variable with mean zero and standard deviation of 1.0.

1.6 Summary

This chapter has introduced the explicit and implicit roles that *accounting* plays in the general context of risk management. This chapter also highlights the difficulty in recognising and quantifying accounting's implicit role, this role being identified as creating the problems of reduced accounting quality and market price volatility. The 2005 IASB financial accounting standards adoption by the UK banks is introduced as the event this study examines. Such an examination is introduced to determine the level of the explicit role that accounting played after the accounting change compared to before. This study examines deviation from this explicit role to be evidence of

accounting's implicit role. In addition, this study deduces evidence to determine if this implicit role shows any relationship to risks that are exogenous to those *normally* faced by the banks examined. The principle measurements applied in this study are then named. The relative delta measure is introduced from its development by extending the Fama and French (2008) time dependent treatment of the book-to-market ratio. The accounting Value-at-Risk and regulatory relative delta measures are also introduced together with an introduction to the methodological approach this study applies.

This chapter has introduced the background to the research problem. Literature that revealed for European firms a reduction in accounting quality after adopting the new accounting standards in 2005 are also introduced. Attention is drawn to this reduction as evidence for the implicit role that accounting plays. Charts are also presented that show for the banks' market price returns, materially significant levels of volatility and Valueat-Risk breaches after the 2005 accounting change. The research problem is formally introduced, in that the requirements from the EC and the IASB for the 2005 accounting change are observed to be falling short of expectations. From the context of this problem, the research question is introduced. Then the null hypothesis and a more detailed overview of the methodology applied to address the research question are presented. This chapter provides a summary of the thesis organisation and presents an overview of the contributions that this research expects to make to knowledge. The areas this research expects to contribute to include the following frameworks: regulatory and firmwide risk management, accounting standards, and regulatory and banking technology. In addition, this study intends to contribute to the regression analysis approach.

Chapter 2 Literature Review

2.1 Chapter Introduction

This chapter presents the background and a review of literature that addresses this research. First, an overview of relevant background literature is presented. Then a background and review of literature examining the change in accounting standards that took place in 2005 from the UK GAAP's ASB accounting standards to the IASB's IFRS accounting standards is provided. Thereafter, this chapter provides a review of literature for measurement of market risk, followed by an examination of the book-to-market ratio and then formally introduces the *relative delta measure*. The relational characteristics of accounting quality, value relevance, the relative delta measure and other key measures applied in this study are then formally introduced. The literature pertaining to the Basel regulatory framework and the Value-at-Risk measure is then reviewed. This chapter concludes by reviewing the potential effect of the Value-at-Risk and the *relative delta measure* on the current banking and regulatory technology framework.

2.2 Research Background

The purpose of accounting as accepted by the accounting profession, Beaver, Kettler and Scholes (1970) observe, is to facilitate investor decision making. The IASB states that the principal purpose of *the accounting framework* is to present information that is most useful for economic decision making (IASB 2011). It could be contended that with the power to affect the actions of economic agents, the rules of accounting that govern reporting of financial statements must be unbiased and neutral to both financial markets and its agents. If neutrality is not evident, it may be maintained that, economic agents, financial markets, financial instruments, and ultimately governments and regions, may become exposed to unnecessary risks.

Exposure to risks are also compounded by the fact that financial markets are extremely dynamic, in that new markets open, new financial instrument products are designed and created, and new clients begin dealing, trading and borrowing (Banks 2002, Merton 1995, Dosi and Moretto 2003, Wang, Wen and Yang 2010, Culp 2010). With improvements in computer systems and telecommunication technology, advances in

finance theory, deregulation and globalisation, this dynamism will continue to accelerate (Banks 2002, Merton 1995, Merton and Bodie 1996¹³). Firms creating, innovating or using new instruments or employing new risk and financing techniques are likely to remain active in new structures (Banks 2002). Determined by Merton and Bodie (1996), a fundamental part of the infrastructure that will require significant change to accommodate this future financial innovation is the financial accounting system.

Aimed at improving the financial system, the IASB was formed in 1973 (IASB 2009, Alfredson, Leo, Picker, Pacter, Radford and Wise 2007). The IASB is aware of the increase in the number and complexity of existing and new financial instrument products traded in the financial markets (IASB1 2008). The IASB also acknowledges that issues exist within the financial system to recognise and measure some complex financial instruments, resulting in accounting standards that are unable to report effectively financial instruments (IASB1 2008). Asserted by Merton and Bodie (1996), an inability for a financial accounting system to report financial instruments has the potential to affect significantly the financial markets, the risk management system and the banking regulatory system.

Merton and Bodie (1996) stated that one of the key public-policy issues facing the global financial system and the international regulatory and supervisory architecture is the requirement to recognise and measure all financial instruments on the balance sheet. The Chartered Financial Analysts (CFA) Institute also maintained that any standard on accounting for derivative financial instruments and hedging activities, should apply to all derivative instruments, and that these derivatives should have a method to be recognised on the balance sheet (CFA 1997).

Merton and Bodie assert that banks and other financial institutions are increasingly having large and varied exposures to financial instruments that are 'off-the balance sheet' (Merton and Bodie 1996, p 19). They note that although off-balance sheet contracts like interest-rate swaps and futures contracts have no initial value, they can have an immediate and significant impact on the risk exposure of the various assets and

¹³ The referenced paper is based on Chapters 1 and 8 of '*The Global Financial System: A Functional Perspective*' by Dwight B. Crane, Zvi Bodie, Kenneth A. Froot, Scott P. Mason, Robert C. Merton, Andre F. Perold, Erik Sirri and Peter Tufano (1995).

liabilities that are on the balance sheet. They determine that firms may use off-balance sheet contractual arrangements such as derivative financial instruments, to protect information from outsiders. At times for some firms, protecting information may be a primary motive, but the more frequent and widespread reason that these '*zero-value*' contractuals are off-balance sheet is that the accounting framework lacks a system to disclose them (Merton and Bodie 1996, p 19).

In addition to Merton and Bodies' (1996) work, off-balance sheet accounting has been the focus of much attention and is referenced by works that include Guan and Chen, W (1998), Monson (2001), O'Haver (2003), Barlas, Madison, Randall and Verschoor (2003), Humphreys (2004), Freire and Figueiredo (2005), Coffee (2005), Cunningham (2005), Chen, Y (2006), Corner (2006), Langley (2006), Mountain (2008), and Lander and Auger (2008).

An article based on the prepared remarks by Randall Kroszner, the Federal Reserve Board governor (from 2006 to 2009) addressing the Global Association of Risk Management Professionals (SI 2008), affirmed the importance of the work on offbalance sheet reporting. Kroszner stated that a firm's true risk exposure requires examining financial instrument risks that are reported on the accounting balance sheet, but also instruments that are reported off-balance sheet. Kroszner determines that offbalance sheet accounting of risk exposures are sometimes more difficult to identify and often not so easy to quantify. As stated by Merton and Bodie (1996), major changes in accounting structure and methodology are required to address such financial instrument accounting inadequacies. In particular, Merton and Bodie declare, financial accounting needs fundamental revisions and requires a specialised new branch called 'risk accounting' (Merton and Bodie 1996, page 19).

As stated earlier, during the financial reporting periods, prior to and including the year ending 31st of December 2004, UK firms prepared and presented financial statements in accordance with the accounting standards accepted under the UK GAAP. Beginning 1st of January 2005, UK firms adopted and reported financial statements in accordance with the IASB framework's IFRS accounting standards (IASB 2009). UK firms under mandatory European Union Regulations adopted the IASB framework's accounting standards for subsequent reporting periods (EC 2002, IASB 2009, Elliot, B and Elliot, J 2008).

On the 1st of January 2005, firms within the banking and finance industries applied four key IFRS accounting standards principally aimed at financial instrument reporting in financial statements. These standards were applied as part of the transitional provisions under the IASB's IFRS accounting standard titled *First-time Adoption of International Financial Reporting Standards*, referred to as IFRS 1 (IASB 2009). The four key accounting standards applied are the following:

- International Financial Reporting Standard 4: *Insurance Contracts*, referred to as IFRS 4.
- International Financial Reporting Standard 7: Financial Instruments: Disclosures, referred to as IFRS 7.
- 3) International Accounting Standard 32: *Financial Instruments: Presentation*, referred to as IAS 32.
- International Accounting Standard 39: *Financial Instruments: Recognition and Measurement*, referred to as IAS 39¹⁴.

Reported by Walton (2004) prior to the IASB accounting standards adoption event in 2005, dissenting views from European adopters existed. The primary basis of opposition, Walton (2004) and Armstrong et al. (2010) observed, was concern that addressed the requirements for reporting financial instrument under the IASB's IAS 39 standard. The CFA Institute, in recognition of financial instrument accounting issues, strongly supported the IASB's 2008 (IASB1 2008) discussion paper entitled *Reducing Complexity in Reporting Financial Instruments* (Wild and Poole 2008).

During the IASB accounting standards adoption in 2005, studies conducted by Platikanova and Nobes (2006), Paananen and Lin (2009) and Morais and Curto (2008) corroborated the concerns reported by Walton (2004), IASB1 (2008) and Wild and

¹⁴ In November 2009 and October 2010 portions of the IAS 39 accounting standard were replaced by the International Financial Reporting Standard 9: *Financial Instruments*, referred to as IFRS 9. It is the intent of the IASB to replace ultimately the IAS 39 standard in its entirety with the IFRS 9 standard (IASB 2011).

Poole (2008). In that, they observed a decrease in the value relevance of accounting information after the adoption of the IASB accounting standards in 2005.

Examination by Walton (2004) prior to the IASB accounting standards adoption in 2005, determined that one primary difference between the IASB accounting standards and the previous standards was the recognition and measurement of derivative financial instruments at *fair value*. Preceding reporting standards maintained a more static accounting model for derivative accounting whilst the IASB accounting standards advocate a more dynamic recognition of derivative instruments based on fair value accounting (Walton 2004).

The fair value accounting method based on *market value*, or more accurately *market* price, has its supporters (IASB 2009, Walton and Aerts 2006, Leone 2008), as well as those who felt that fair value does not necessarily represent an un-biased view (Walton and Aerts 2006, Cheney 2008, Leone 2008). The latter concern may be highlighted by the supposition that prices quoted on financial markets may fluctuate rapidly on a day to day basis, and to utilise market based valuation in the accounting system would cause 'gratuitous instability', instead of 'reflecting long term economic value' (Walton and Aerts 2006, p 476). Cheney (2008) also states that fair value accounting used for reporting some financial instruments, has given rise to a global controversy in that the fair value method may be causing volatility in securities and credit markets. Trade groups representing the international banking community, as reported by Leone (2008), had also expressed concerns regarding using only the fair value method of accounting, preferring a mixed-attribute model of reporting. The mixed-attribute accounting model, applies the fair value method for financial instruments held for trading purposes, and historical cost calculations for assets and liabilities held-to-maturity, or based around long-term trading (Leone 2008).

However, reported in the same article by Leone (2008) a poll by the CFA Institute found that 74% of investment professionals thought fair value requirements for reporting financial instruments improved general market integrity. Meanwhile, 79% of investors polled found the fair value method improved transparency, and contributed to the investor understanding of the reporting institution's risk profile.

The IASB stipulates financial instruments are to be valued using the fair value method under the *fair value option* criteria, and also that financial instruments used for hedging activity are to be treated under the practice of hedge accounting (IASB 2009). However, as reported by Walton and Aerts (2006), Cheney (2008) and Leone (2008), such attempts at implementing the fair value and the hedge accounting methods have been met with critical opposition. In that, the fair value treatment being attributed to increases in financial instrument price volatility levels.

From the risk management perspective, it may be contended that firmwide risk management systems and approaches are not tuned to specifically deal with financial risks attributed to effects that arise from the application of, and changes in, accounting standards. The book-to-market effect specified by Fama and French (1992, 2008), Peterkort and Nielsen (2005) and Bodie, Kane and Marcus (2002) and the book-tomarket delta effect detailed by Fama and French (2008) and developed in this study as the relative delta measure, have provided evidence to be an effective measure of a firm's exposure to financial distress risk. However, the book-to-market effect assigned to emanate from the level of difference attributed to the accounting measure - book equity (or, as applied in this study, *total equity*), and the market measure - stock market price, has not received attention and thus does not feature prominently in the contemporary risk management framework. Contemporary risk management systems are tuned to be more reactive to market risks arising from stock market price risks, interest rate risks and foreign exchange risks (JPMorgan Chase and Reuters 1996). The assertion for risk management systems to also monitor and manage financial distress risk may be strengthened by examining evidence reported by Platikanova and Nobes (2006) and Paananen and Lin (2009) that strongly suggest stock market price volatility may be reactive to financial distress risk.

The regulatory framework is however better aligned to monitor and manage risks arising from the accounting and market relationship. The Basel (2011) regulatory framework is currently aimed at being highly reactive to credit risks and market risks as well as operational risks. The Basel regulatory framework centers on risk weights that are applied to *balance sheet* assets. The risk weights act the same way as risk premiums and adjust asset values based on the asset instrument category. Although the Basel regulatory framework has a stronger relation to accounting information than firmwide

risk management systems, it does not adjust for potential risks arising from the accounting information itself.

Contrary to the expectations of the processes and frameworks implemented by accounting and regulatory bodies (EC 2002, IASB 2009), the studies by Platikanova and Nobes (2006), Paananen and Lin (2009), and Morais and Curto (2008) highlight that a deficiency in accounting information arose in the financial system during the IASB accounting standards adoption event in 2005. A deficiency observed to be maintained for the subsequent years (Paananen and Lin 2009).

This study strengthens its justification to test its null hypothesis presented in Section 3.3.1, with the view to determine if the financial system deficiencies declared by Merton and Bodie (1996), the CFA Institute (1997), Walton and Aerts (2006), Cheney (2008) and Leone (2008), and the financial system deficiencies observed by Platikanova and Nobes (2006), Paananen and Lin (2009), and Morais and Curto (2008) are evident for the UK banks after the 2005 accounting change when compared to before.

The following sections provide a background and a further review of literature that examine the change in accounting standards that took place in 2005 from the UK GAAP's ASB accounting standards to the IASB's IFRS accounting standards. This chapter then provides a review of literature for measurement of market risk, followed by an examination of the book-to-market ratio. The development of the relative delta measure and the relational characteristics of accounting quality to measures applied in this study are then formally introduced. The literature pertaining to the Value-at-Risk measure is then reviewed. Thereafter literature for the Basel regulatory framework is reviewed and the application of the relative delta measure to the framework's minimum capital requirement is developed. The chapter concludes by reviewing the potential impact of the relative delta measure on the current banking and regulatory technology framework.

2.3 Background to the 2005 Accounting Change

In January 2005, the major banking firms registered with the LSE observed a change in accounting standards that governed the preparation and presentation of consolidated financial statements (IASB 2009). This change was applicable to banking firms and

other organisations registered with the European Community's regulated capital markets, under the 2002 European Commission regulation entitled 'Regulation (EC) No. 1606/2002 of The European Parliament and of The Council of 19 July 2002 on the Application of International Accounting Standards' (EC 2002).

The 2002 European Commission regulations (EC 2002) ruled that firms listed in regulated capital markets within the European Community and governed by the regional laws change from current applied accounting standards to the IASB's IFRS¹⁵ accounting standards, at the latest by 2005. That is, firms had to prepare financial statements for the 2005 fiscal year in accordance with the IASB accounting rules.

The aim of the 2002 European commission regulation concentrated on improving two innately related areas:

- 1) Make improvements to the efficient functioning of regulated financial capital markets at the European Community member state regional level.
- As a result of improvements to the efficient functioning of regulated financial capital markets at the regional level, improve economic performance at the European Community level.

The objectives of the European Commission's 2002 regulations were to be met by implementing a legal framework that included employing a more rule based set of accounting standards, which would differ from regional accounting standards based on best practices. Compared to regional accounting standards, the new standards were aimed at enhancing the current levels of intra-firm transparency and inter-firm comparability of accounting information.

The new standards would be recognised to have enhanced transparency if they reported information that is relevant, reliable and understandable (IASB 2011). The IASB (2011) specifies information to have the quality of relevance where information reported is free

¹⁵ The EC (2002) regulations pronounced that the International Accounting Standards Board (IASB) is the new name assigned to the International Accounting Standards Committee (IASC), and the International Financial Reporting Standards (IFRS) is the new name assigned to the International Accounting Standards (IAS). Both name changes taking effect from 1st April 2001.

from bias, free from material error, and based on transactions that have already taken place. The quality of reliability is characterised by information that is reported on future transactions that have a high likelihood of occurring. The quality of understandability is characterised by information that can be interpreted easily by decision makers with reasonable business knowledge. However, the IASB (2011) asserts that complex information that is relevant and has the potential to assist decision making should not be excluded from the financial statement.

The aims of the new accounting standards were also to present information that would enhance the ability of accounting figures to be comparable through time, and between firms. These aims were to be met by the objective of having a single set of accounting standards to be applied by all regulated firms within the European Community, with the potential for the standards to be applied at the global level.

The European Commission asserted that by applying the IASB accounting regulatory framework and the adoption of the IASB IFRS accounting standards at the Europe level in 2005 and at an international level thereafter, the aims of the 2002 regulation for enhanced transparency and comparability of financial statements would be met (EC 2002, IASB 2009).

Armstrong et al. (2010), Ramanna and Sletten (2009), Beuselinck, Joos, Khurana and Van der Meulen (2010), Horton, Serafeim and Serafeim (2013) and the earlier referenced studies by Platikanova and Nobes (2006), Paananen and Lin (2009) and Morais and Curto (2008) have examined the 2005 accounting change from differing perspectives.

The study by Armstrong et al. (2010) tested the reaction of firms listed in European stock markets to events that took place prior to, but was related to the 2005 accounting change. The study found evidence that firms' stock market reactions to these *accounting change related events* corroborated investor sentiments. In that, the accounting change would bring about higher quality of information, and thus a decrease in information asymmetry. These benefits were attributed to the surveyed investors' high level of expectation for the new standards reported by Leone (2008). Armstrong et al. (2010) also found positive sentiments attributed to expected benefits from enhanced

information comparability between firms. However, contrary to expectations, the study found evidence that firms domiciled in code-law based countries showed market reactions that evidenced a more resistant view to the accounting change than did firms domiciled in common-law based countries.

Ramanna and Sletten (2009) analysed the sentiment to the accounting change in non-European countries from 2002 to 2007. The study finds evidence that the larger economies are resistant to give up standards-setting authority to a single international body, such as the IASB. The study also found that the benefits gained from the accounting change would first increase and then decrease the levels of influence from domestic governing institutions. The study also states that countries are more likely to make the accounting change if trade partners and geographically regional countries adopt the IASB accounting standards.

Beuselinck et al. (2010) measured information quality for European firms from 2003 to 2007 by examining analyst earnings forecasts for firms that adopted the IASB accounting standards. The study finds a reduction in earnings forecast errors especially after 2006. Further, the study finds the largest improvement in forecasts was exhibited by analysts that examined firms in more than one country. Horton et al. (2013) also measured information quality, however, for firms in 46 countries¹⁶ from 2001 to 2007. The study, same as the Beuselinck et al.'s (2010) examination, measured information quality from analyst earnings forecasts for IASB accounting standards adopters. The study finds a reduction of earnings forecast errors and suggests a greater level of forecast accuracy for mandatory IASB accounting standards adopting firms when compared to voluntary adopters. The findings from both Beuselinck et al (2010) and Horton et al. (2013) suggest an increase in forecasted earnings accuracy for firms that adopted the IASB accounting standards. However, Beuselinck et al. (2010) find that consensus amongst analysts for a firm's forecasted earnings remained unchanged irrespective of the accounting standard the firm applied. Horton et al. (2013) report that earnings consensus declined for firms that mandatorily adopted the IASB accounting standards.

¹⁶ Hong Kong is included in Horton et al. (2013) sample of 46 countries. However, they note that Hong Kong is a Special Administrative Region (SAR) of the People's Republic of China.

The studies conducted by Platikanova and Nobes (2006), Paananen and Lin (2009) and Morais and Curto (2008) concentrated on the reaction of firms listed in European stock markets. The study conducted by Platikanova and Nobes (2006) concentrated on reactions at the firm level during the accounting change, while the studies from Paananen and Lin (2009) and Morais and Curto (2008) concentrated on market reactions before and after the 2005 accounting change.

The study by Platikanova and Nobes (2006) found evidence that firms listed in European stock markets exhibited a decrease in value relevance and an increase in information asymmetry during 2005, the year of the accounting change. They report that during 2005, firms in the UK experienced an exceptional increase in stock market price volatility levels compared to years 2003 and 2004, the increase in volatility levels they attributed to information asymmetry. The study by Paananen and Lin (2009), when concentrating on firms listed in the German stock market, found evidence of a decrease in value relevance during and also after the 2005 change. The study by Morais and Curto (2008) that concentrated on firms listed in the Portuguese stock market, also found evidence of a decrease in value relevance of a decrease in value relevance after the 2005 accounting change.

Based on the evidence produced from studies that analysed firms in the European stock markets, it is difficult to determine if the European Commission's objectives have been met during the 2005 IASB accounting standards implementation phase. The principal objectives of the Commission being to enhance accounting information transparency and comparability.

It may be contended that there was little doubt regarding the high level of expectation for the 2005 accounting standards to meet the set objectives. This is reflected by Armstrong et al. (2010) who consider the 2005 accounting change to be one of the most significant accounting changes in recent years. The significance attached to the 2005 *event* may remain for some years, with considerable research and debate directed at the planned transparency and comparability aims for the accounting standards' convergence on a global level.

As referred to earlier, the importance attached to the 2005 accounting change is further heightened by reports of dissenting views that existed prior to the change (Walton 2004), with concerns reaching the highest government levels (Armstrong et al. 2010).

Walton (2004) reported that opposition to the new standards came primarily from European advisors and technical expert groups. Serious concerns were also noted from European banks. The primary basis of opposition, Armstrong et al. (2010) and Walton (2004) observed, addressed the requirements for reporting financial instruments under the IASB's IAS 39 standard. As detailed by Walton (2004), the IAS 39 accounting standard, entitled '*International Accounting Standard 39: Financial Instruments: Recognition and Measurement*' (IASB 2009 pp. 1983 - 2080) was criticised for lacking the required qualities to provide a true and fair view of financial instruments. It was claimed that this standard did not represent accurately the effects of hedging strategies used by banks.

Subsequent to the 2005 accounting change, and in view of evolving issues, the IASB published the earlier referenced discussion paper entitled, '*Reducing Complexity in Reporting Financial Instruments*' (IASB1 2008). The aim of the discussion paper signalled IASB's attempt to start taking steps to gain further understanding of issues with financial instrument accounting. The IASB discussion paper, according to Wild and Poole (2008), is in response to the widely held view that the IASB accounting standard's requirements for reporting financial instruments favoured the fair value method of accounting as a long-term solution, and the standards were difficult to understand, interpret and apply. As referred to earlier, the Chartered Financial Analyst (CFA) Institute, in recognition of financial instrument accounting issues, strongly supported IASB's 2008 discussion paper (Wild and Poole 2008).

From this discussion, it is evident that the expected benefits from adopting the IASB's IFRS accounting standards would be improved transparency and comparability resulting in greater financial market stability and higher levels of accounting quality (EC 2002, IASB 2009, Leone 2008). The concern, however, is that firms that adopt the IASB accounting standards would experience increased levels of stock market price fluctuations and thus experience higher levels of exposure to stock market price volatility than otherwise (Walton and Aerts 2006, Walton 2004, Cheney 2008, IASB1 2008, Wild and Poole 2008, Leone 2008). The empirical examination of these theories by Platikanova and Nobes (2006), Paananen and Lin (2009) and Morais and Curto (2008) suggest that European firms that adopted the IASB accounting standards in 2005 experienced increased levels of exposure to stock market price volatility and a deterioration in accounting quality after the adoption compared to before. From the

basis of this discussion, this study expects the link between the IASB's IFRS accounting standards adoption and improved transparency, and also improved comparability, to be evidenced by higher levels of accounting quality than otherwise (EC 2002, IASB 2009 and Leone 2008). As discussed earlier, accounting quality is examined in this study by applying the relative delta measure that is developed from the time dependent treatment of the book-to-market ratio (Fama and French 2008). The development of the relative delta measure is discussed in the following sections and its methodological approach is presented in Chapter 3, Section 3.9.2.

To reiterate, given the evidence from the reviewed literature for European firms' stock market reactions before and after the accounting standards change, together with the dissenting views reported from European banking firms. Such evidence would add strength to the expectation that the UK banking sector may be an important area to research during the 2005 accounting change. This study aims to extend the research conducted in this area by effectively testing its null hypothesis. That is, principally analysing the effects between accounting totals and the market price before and after the 2005 accounting change for UK banks in the time range 1992 to 2009.

As stated earlier, this research aims to adapt the Fama and French (2008) time dependent treatment of the book-to-market ratio to develop the *relative delta measure*. As introduced, the relative delta measure is aimed at addressing this study's null hypothesis by measuring, for a given time period, how key accounting totals changed relative to the change in market price.

From the Fama and French (1992) and Peterkort and Nielsen (2005) evidence for the book-to-market ratio's risk measurement properties, and from the Fama and French (2008) evidence, the level of difference evaluated using the relative delta would also provide for a firm its level of exposure to financial distress risk. This study measures a bank's level of exposure to financial distress risk by using both the book-to-market ratio, and the relative delta. This study also compares both measures to the Value-at-Risk measure. The Value-at-Risk measurement model is endorsed by the IASB (2012) and by the Basel III accord (Basel 2011) as an accepted approach to measuring *financial risk* within a firm's risk management process. This study extensively applies the Value-at-Risk measure to the market price variable in order to determine the level of stock market price risk for the UK banking sector. This study also applies the Value-at-Risk

measure to the accounting total variable by developing an *accounting Value-at-Risk*. This study develops the accounting Value-at-Risk by first developing the relative delta measure and then applying the relative delta to the market price return Value-at-Risk (see Appendix H).

Having a measure for Value-at-Risk based on both the accounting total variables and the market price variable would enable extending the null hypothesis test by examining, for the UK banks, the difference in the level of accounting Value-at-Risk to the market price Value-at-Risk¹⁷ before and after the 2005 accounting change. To support the null hypothesis it would be expected that both the accounting and market price Value-at-Risk would exhibit similar levels for before and after the 2005 change.

The derivation for a measure of Value-at-Risk for a key accounting total would involve an additional step of scaling or adjusting the market price return Value-at-Risk with the accounting total and market price relative delta (see equation (3.44)).

The relative delta measure is presented in Sections 2.7 and 3.9.2, and the accounting Value-at-Risk measure is presented in Section 3.9.4. The following sections examine literature that relates to the development of these measures.

2.4 Risk Measurement

A logical approach to the development of the relative delta measure and then the accounting Value-at-Risk would be to examine literature based on the measurement of risk from a market perspective. This part of the literature review begins by examining work that specified the standard deviation measure to be a fundamental measure of risk. From the development of the standard deviation, a brief introduction to the well-known capital asset pricing model is provided followed by a review of the book-to-market ratio. This study considers the book-to-market ratio to be an important measure in the development of the relative delta measure. It is then expected to review the Value-at-Risk measure with the objective to providing a detailed examination of the literature surrounding its development and use. A review of the Value-at-Risk measure is also

¹⁷ This study refers to the *market price Value-at-Risk* also as the *market price return Value-at-Risk* and also as the *Historical Value-at-Risk Actual*.

presented by detailing literature that contributed to the development of the Monte Carlo simulation Value-at-Risk model.

2.5 Standard Deviation, CAPM, and the Book-to-Market Ratio

It may be contended that Bernoulli was the first to specify the association between the standard deviation measure and its application in quantifying the risk of financial loss during a financial transaction between two parties. His 1738 work entitled the Specimen Theoriae Novae de Mensura Sortis'¹⁸ (Bernoulli 1738) proposed that a measure of financial risk of loss during a transaction for an observed price variable should be a measure of both its average and its deviation from its average. The latter is the specification for the contemporary statistical standard deviation measure. According to Bernoulli the measure of potential loss from a financial transaction, measured using the standard deviation measure, may also provide a measure of its gain. Markowitz (1952), in his work on modern portfolio theory extended the use of the standard deviation measure by specifying it as a direct measure of financial risk. He proposed that the standard deviation, also termed the standard error or volatility, for a given portfolio of instruments could be reduced to an optimal minimum, thus reducing its risk, while at the same time increasing the level of its financial gain, that is, its return¹⁹. This increase in returns efficiency he proposed is possible by diversifying the risk to return profile of the instruments contained in the portfolio. Markowitz termed this theory the E-V rule or the Expected returns to Variance of returns rule, where either the variance or its square root, the standard deviation, may be used as the measure of risk.

2.5.1 The Capital Asset Pricing Model

Markowitz's theoretical development of the E-V rule has had a profound influence on the way modern financial markets approach the measurement of financial risk and return. The most noted development from the E-V rule is the capital assets pricing model, more popularly known by the acronym CAPM (Jensen, Black and Scholes 1972). The development of the Capital Asset Pricing Model has been influenced by the

¹⁸ This study makes reference to Bernoulli's 1738 work based on the translation from Latin to English under the title 'Exposition of a New Theory on the Measurement of Risk' from the Journal Econometrica published by the Econometric Society in 1953.

¹⁹ Details for financial return, from the perspective of market price return, are presented in Section 3.9.2 and Appendix M.7.

work of Sharpe (1964), Treynor (1962, French, C 2003), Lintner (1965) and Mossin (1966).

Specified by Jensen et al. (1972) the capital asset pricing model comprises a straightline model such as the linear regression model. The equation consists of a dependent variable represented by the expected returns for a selected financial instrument. The independent variable theoretically may consist of adjusted expected returns from all available stocks traded in the financial market. Characteristically the independent variable is selected as an adjusted market portfolio (Jensen et al. 1972). The stock market portfolio is adjusted by a risk free rate. The risk free rate is represented by the interest rate or yield from an interest rate instrument such as a Government bond that has a similar maturity as the holding period for the model's dependent variable instrument. The slope from the implemented straight-line equation provides a measure of the selected financial instrument's risk premium and expected return. The slope is termed the *beta*, and symbolized by the Greek letter β . The intercept from the regression equation represents the risk free interest rate.

2.5.2 Book-to-Market Ratio

Using variables that are not entirely familiar to the Asset Pricing Models, but with the same objective, the literature examination of the book-to-market ratio²⁰ (BM) has centered on how well it can measure a firm's risk and return (Fama and French 1992). The book-to-market ratio has been extensively treated in the work of Banz (1981), Bhandari (1988), and Fama and French (1992, 1993, 2008). For a firm's *book*-to-*market* ratio measure, the literature specifies *book* to be the firm's *book value* measured using *book equity* that is based on the accounting measure *total shareholders' equity*, and *market* as a firm's *market value*²¹. Where, the *market value* is calculated by multiplying the number of the firm's shares traded or outstanding in the market by the current market price of those shares.

The attention given to the book-to-market ratio is justifiable as it combines the accounting measure of a firm's value, the book value, to its financial market's measure

²⁰ This study refers to the book-to-market ratio interchangeably also as BM.

²¹ Market Value is also referred to as Market Capitalisation by Fama and French (2008).

of value, the market value. Attention further grew for the BM when Fama and French (1992, 1993) evaluated that the BM measure provided a more accurate measure of portfolio returns for a single time period than the capital asset pricing models beta measure. Fama and French (1992, 1993, 2008) termed stocks with high BM as value stocks, and stocks with low BM as growth stocks. Empirically they evaluated that the measure of a stock's BM is a direct reflection of its level of financial risk. They evaluated that value stocks with a high BM would be in *financial distress* and exhibit lower earnings, that is a lower net income accounting total, and thus exhibit a greater level of financial distress risk when compared to low BM growth stocks.

Studies conducted by Fama and French (1992, 2008), Rosenberg, Reid, and Lanstein (1985), Lakonishok, Shleifer, and Vishny (1994) found evidence that the *high-risk* high BM value stocks had higher average returns compared to low BM growth stocks. The increase in returns may be explained as the expected investor compensation for carrying *high-risk* high BM value stocks. Even though Fama and French (1992, 1993) evaluated the BM to be a better measure for a firm's expected return compared to the CAPM beta, Vuolteenaho (2002) and Fama and French (2008) recognised that the BM is a *noisy* measure of expected returns for portfolios of stocks.

2.6 Book Delta to Market Delta

Fama and French (2008) and Vuolteenaho (2002) attribute the *noisy* behaviour exhibited by a portfolio's average BM measure, when used to estimate expected returns, was as a result of investors reacting to expected cash flows. They determine that this investor expectation is due to the expected dividend payments from the portfolio's individual stocks.

Fama and French (2006) tested an array of variables, including accounting variables based on financial statement accounting totals, in an attempt to mitigate the observed BM noise and thus improve the BM estimates of average cross-sectional stock returns. However, they reported limited success.

Fama and French (2008) tested the time dependent progress, i.e. the *delta*, of the individual BM component variables, to determine whether it would improve predictions

for cross-sectional returns. The Fama and French (2008) development of the BM and *book delta* to *market delta* measure takes the following decomposition:

$$BM_t = BM_{t-k} + [dB_{t-k,t} - dM_{t-k,t}]$$
(2.1)

Where BM_t is the log^{22} of the BM at time t and is expressed as, BM_{t-k} : the log of the BM at time t - k plus the difference between $dB_{t-k,t}$: the change or *delta* in the log of book equity from time t - k to t, and $dM_{t-k,t}$: the change or *delta* in the log market price for the same time period t - k to t. In addition, this study refers to *dB* as the *book delta* and *dM* as the *market delta*.

Fama and French (2008) found evidence that examining the BM components, *book delta* to *market delta*, presented in equation (2.1) did improve estimates of expected returns.

The term *book delta* to *market delta* is applied in this study to describe the component expressed inside the square brackets in equation (2.1). It is also the addition of this component to the contemporary book-to-market ratio that this study terms *the time dependent treatment of the book-to-market ratio*, and is referred to in Section 1.2 to introduce the development of the relative delta measure. Defining this component for a single time unit specifies the following expression (see also Appendix A):

$$[dB_{t-1,t} - dM_{t-1,t}]$$

This measurement component is applied in Section 2.7 to develop the relative delta measure (see also Appendix A to Appendix D). The relative delta measure is then specified in Sections 2.9.2 and 2.9.3 as a measure of accounting quality and thus as a measure of the level of exposure to accounting quality risk.

²² The term log is used to specify the natural logarithm to the base e and may also be denoted as ln(X), where (X) is the termed being logged.

2.7 Book Delta to Market Delta and the Relative Delta Measure

The evidence found by Fama and French (2008) that the change in book equity, referred to in this study also as the *book delta*, and the change in market price, referred to in this study also as the *market price delta*, provides an enhanced measure of expected cross-sectional stock returns and provides the foundation for this study's development of the relative *delta measure*. The development of the relative delta measure is presented in Appendix B. The specification for the relative delta measure is presented in Appendix C.

This study extends the Fama and French (2008) time dependent treatment of the bookto-market measure exhibited in equation (2.1) in four distinct ways:

1) The book-to-market ratio form specified in equation (2.1) is a measure of the scalar book value to market value. The relative delta extends this form by focusing on the delta component. That is:

$$\delta_{t-1,t} = \left[dE_{t-1,t} - dM_{t-1,t} \right] \tag{2.2}$$

Where for time t - 1 to t: $\delta_{t-1,t}$ is specified as the *total equity to market price relative delta*, $dE_{t-1,t}$ is the log change in *total equity* and $dM_{t-1,t}$ is the log change in market price. The relative delta measure shown in equation (2.2) measures at time t the *difference component* for time t - 1 to t. Therefore, the *difference component* is specified in this equation as the measure of the difference between the log change in the accounting totals measure, total equity, and the log change in market price.

2) Fama and French (2008) calculate *book equity* primarily by applying the following data items from the data provider Compustat: total assets minus liabilities, plus deferred taxes and investment tax credit, minus liquidating, redemption or carrying value of preferred stock²³. To analyse the full effect of the financial statement's accounting totals, this study replaces book equity with the total shareholder's equity (*total equity*) financial statement accounting total item. In effect, maintaining the total assets minus

²³ Specified in Fama and French (2008, p 2979).

liabilities measurement component from the Fama and French (2008) book equity measure.

3) The Fama and French (2008) approach to the book equity component, in equation (2.2), is calculated principally by the *change in the book equity per share* measure. Although, this study does perform tests applying the *total equity per share* measure, the relative delta measure extends the Fama and French (2008) measure by specifying the *change in total equity*, in comparison to the Fama and French (2008) *change in book equity*. To assist the development of the relative delta measure, the latter accounting standards terminology, i.e. *total equity*, is applied in this study in place of *book equity*. Details of the Fama and French approach and this study's approach to equation (2.2) are presented in Section 3.2.3.

4) The relative delta measure extends the Fama and French (2008) book equity treatment by examining total equity, and the additional accounting totals: total assets, total liabilities and net income.

The following section reviews the balance sheet and income statement accounting totals examined in this study.

2.8 Accounting Totals

Discussed earlier, Fama and French (2008) found evidence that constructing a measure using the difference between the change in book equity and that of the change in market price provides explanatory power for expected returns and thus also a measure of perceived financial distress risk. This study extends the essence of this measure by specifying a system of measures that quantifies the levels of difference based on the accounting equation. This system of measures is collectively termed the *accounting to market price relative delta measure*.

2.8.1 Accounting Equation

The accounting equation takes the general form:

$$Total Shareholders' Equity = Total Assets - Total Liabilities$$
(2.3)

The term *Total Shareholders' Equity*, in equation (2.3), is the accounting measure applied in this study in the same way as the Fama and French (2008) book equity measure. Total shareholders' equity, total assets and total liabilities are reported in the principal financial statement generally termed the balance sheet. Another key measure is net income and is reported in the key financial statement generally termed the income statement.

Before and after the accounting change, for each of the key financial statement totals: total shareholders' equity, total assets, total liabilities and net income, this study measures their material change as well as their change relative to the change in market price. The latter measurement is the referred *relative delta measure*. These are the principal measures this study applies to test its null hypothesis.

From the basis of definitions presented by the IASB (2011), the following sections present the key financial statement totals applied in this study and referred to as the accounting totals.

2.8.2 Total Shareholders' Equity

From the accounting equation, the balance sheet reports the total shareholders' equity to be the difference between the measurement of asset items and the measurement of liability items. The IASB (2011) specifies the *total shareholders' equity amount* as a balance sheet element that is the residual interest in assets after the deduction of all liabilities.

The IASB (2011) sub classifies the total shareholders' equity amount to include an equity reserve item termed retained earnings. The retained earnings figure includes amounts that have been credited from tax liabilities and amounts reserved under regulatory ordinance and legal ordinance. The amount reserved under regulatory ordinance is a key measure in the Basel (2011) regulatory standards where it forms a part of what is termed the *regulatory capital*. The Basel standards specify that the measure of regulatory capital together with, what are termed *risk weighted assets*, provide a basis to measure a firm's capital adequacy, that is, the level of capital available to meet the likelihood of financial loss. Within the Basel framework, a firm's capital adequacy is measured by the Basel capital requirement formula. The level of

capital adequacy reported by this formula ultimately provides a measure of a firm's credit risk. The Basel capital requirement formula is presented in Section 2.10.

2.8.3 Total Assets

The balance sheet reports total assets to be the amount recorded for the sum total of all active past transactions that represent an inflow of economic benefit to a firm. The IASB (2011) specifies assets as a balance sheet element, and a single *asset amount* to be an item that results in an inflow of resources that embody economic benefit from the item to the firm.

IASB (2011) specifies that for a transaction to be recognised as an asset amount on the balance sheet, the transaction will have a high probability of future economic benefit to flow from the item to the firm, and the item's cost or value can be reliably measured. The asset would be recognised on the balance sheet until it is determined that future economic benefit from the item is improbable.

2.8.4 Total Liabilities

Total liabilities are reported on the balance sheet when a firm's sum total of active past transactions represents the present obligations that must be satisfied. The IASB (2011) specifies liabilities as a balance sheet element, and a single *liability amount* to be an item that satisfies a present obligation with the result of an outflow of resources that embody economic benefit to the item from the firm.

The IASB (2011) specifies that for a transaction to be recognised as a liability amount on the balance sheet, the transaction will have a high probability of future economic benefit to flow to the item from the firm.

2.8.5 Net Income

Net Income is reported in the income statement and relates to the financial performance or the profitability of a firm during the past tax or fiscal year. That is, the monetary amount that has flowed into the firm during the financial year. Generally, the net income figure is represented by the residual monetary amount of revenue after the deduction of all expenses. The IASB (2011) specifies net income as an income element, and a single *net income amount* as an item that represents an increase in the economic benefit to a firm.²⁴

Net Income in this study is considered as the amount remaining at the financial year-end after deducting income tax expenses and other expenses. In the banking template for the financial accounting income statement, the net income amount used in this research is generally referred to as the *profit that is attributed to a firm's shareholders*.

The reviewed accounting totals together with the market price are the principal variables applied in this study to determine accounting quality and thus its null hypothesis. This study quantifies accounting quality by measuring the difference between the key financial statement accounting totals and the market price before and after the 2005 accounting change. To measure this difference (or value relevance), this study adopts a system of *relative delta measures* developed by extending the Fama and French (2008) measure for the change in book equity to the change in market price (see Sections 3.2.4 and 3.2.5). This system of measures includes the previously introduced *Total Equity to Market Price Relative Delta*, and extends the measurement principle to include the other key financial statement accounting totals. The system of relative delta measures used in this study are the: *Total Equity to Market Price Relative Delta*, *Total Liabilities to Market Price Relative Delta* and *Net Income to Market Price Relative Delta*. The next section formally introduces and reviews *accounting quality* and its relation to *value relevance, relative delta* and other key measures referenced in this study.

2.9 Accounting Quality

Accounting quality is determined by Barth, Landsman and Lang (2008), Morais and Curto (2008), Paananen and Lin (2009) and Liu, C, Yao, Hu and Liu, L (2011) to be a function of value relevance, and earnings quality. Where, earnings quality is specified by the level of earnings management and loss recognition. They determine that an

 $^{^{24}}$ The financial statement reports the net income amounts based on the amount of capital that has been earned by a firm from the beginning of a fiscal year to the end of that fiscal year. That is, the capital attributed to the 12 months that comprises the fiscal year. The fiscal year is also referred to as the tax year. From 2002 to 2008 this research studies the effect of the individual yearly net income amounts for the sample firms and control group based on the last 6 months prior to the end of the fiscal year. For the purpose of reference, Appendix R.1.2 presents for the primary sample a material analysis of both the 6 month and 12 month net income amounts.

improvement in accounting quality is attributed to an increase in value relevance, decrease in earnings management, or a reduction in the time taken for losses to be recognised. Although the general approach in determining improved accounting quality shows evidence that it relies upon quantifying both value relevance and earnings quality, Platikanova and Nobes (2006) specifies this improved quality to be predominantly a function of an increase in value relevance.

This section details the associative relationship between accounting quality and the *measures*: value relevance, relative delta, the explicit and implicit roles of accounting, financial distress risk, and the level of *information bias* between accounting and market price data elements.

2.9.1 Accounting Quality and Value Relevance

The determination of accounting quality by measuring value relevance is specified by Amir, Harris and Venuti (1993), Barth et al. (2008), Morais and Curto (2008), Paananen and Lin (2009) and Liu, C et al. (2011) to be a function of the relationship between the accounting total net income and market price, and book equity and market price. They prescribe that the stronger the positive relationship for both the accounting measures net income and book equity to that of market price the higher the value relevance and therefore the greater the quality of the reported accounting data.

Platikanova and Nobes (2006) specify accounting quality by also utilising market price levels. However, they do so in order to determine the absence of volatility from the market price's bid-ask spread for a specified time period. If volatility is absent then it is concluded that publicly available accounting data is value relevant and exhibit a high level of accounting quality than if volatility had been present.

2.9.2 Accounting Quality, Value Relevance and the Relative Delta Measure

The relative delta measure is introduced in Section 1.1 and specified in Section 2.7 and Appendix C and is applied in this study as a measure of the difference between accounting and market price change variables. The relative delta measure, based on its origins to improve estimates of stock returns and measure financial distress risk (Fama and French 1992, 2008, Peterkort and Nielsen 2005, Bodie et al. 2002) is founded on the same principles as the measure of value relevance and thus the determination of the

level of accounting quality detailed by Amir et al. (1993), Barth et al. (2008), Morais and Curto (2008), Paananen and Lin (2009) and Liu, C et al. (2011). That is, the relative delta measure and the value relevance measurement approaches detailed in the referenced literature are fundamentally based on the level of the relationship between accounting and market price variables.

The use of the relative delta to test this study's null hypothesis by evaluating the level of value relevance, and therefore quantifying the level of accounting quality, is specified in Section 3.3.1. In Section 1.1, Section 2.7 and Appendix C the relative delta measure is defined such that the closer its evaluation to zero, due to a smaller difference between the accounting and market price change variables, the greater is the value relevance and thus the greater the accounting quality of the tested accounting data. The converse is true when the relative delta measures a greater difference between the change variables. This higher level of relative delta indicates a lower level of value relevance and thus characterises a lower level of accounting quality.

2.9.3 Relationship Between Accounting Quality, Accounting Quality Risk, Value Relevance, Difference Component, Relative Delta Measure, Explicit and Implicit Roles of Accounting, Financial Distress Risk and Information Bias

As stated, the relative delta measure is generally specified as the difference between the change in accounting totals and the change in market price (see Section 2.7 and Appendix C). In Sections 1.2 and 3.2.3 to 3.2.5, the relative delta by quantifying the difference between these change variables is detailed to be the direct measure of the *difference component*. The relative delta measure in Section 1.1 and Section 2.9.2 has also been detailed to reflect value relevance and thus accounting quality. It must also be asserted that the relative delta measure, and thus the measures of value relevance and accounting quality, is also related to the role accounting plays in helping to mitigate financial risk. Introduced in Section 1.1, the closer the relative delta evaluates to zero, the greater the explicit role that accounting takes within the context of financial risk management. This effect indicates a high level of value relevance and thus a high level of accounting quality, and also indicates a low level of exposure to accounting quality risk.

On the contrary, as elucidated in Section 1.1, the higher the level of the relative delta
measure as a consequence of a greater difference between the accounting and market price change variables, the lower the level of accounting's explicit role and the greater its implicit role. Specified in Section 1.1, Section 3.2.2 and Appendix C.2 this effect is also a measure of *financial distress risk*. In that, the higher the level of the relative delta measure, that is the higher the level of the *difference component*, the greater is the level of exposure to financial distress risk. In addition, this effect is also a reflection of an increase in *information bias* between the accounting and market price data elements. This general effect indicates a lower level of value relevance and thus a lower level of accounting quality, and also indicates a higher level of exposure to accounting quality risk, than would otherwise be determined.

2.10 Basel Risk Management Framework and Risk Measurement

The measure of a firm's *accounting quality, relative delta, financial distress risk* and *information bias*, although measured and treated the same in this study, are not currently prescribed measures of risk within the Basel III bank regulatory framework. The Basel I (Basel 1998), Basel II (Basel 2004, 2006), Basel II.5 (Basel 2009) and Basel III (Basel 2010, 2011) regulatory frameworks, stipulate that pre-specified risk weights must be applied to categories of currently held assets recognised, measured and reported in a firm's balance sheet as well as items that are off-balance sheet (Basel 2011). The level of the weighting applied is dependent on the class of the recognised assets, with a 0 (zero) risk weight applied to the *price level* or value of assets such as cash, and higher risk weights applied to the price of derivative financial instruments such as, swaps, futures, forwards and options. Higher risk weights are also applied to complex securitisations such as structured investment vehicles and products such as credit default swaps.

The prices applied to a firm's assets and items recognised on and off its balance sheet, that include complex financial instruments such as derivatives, are based on valuation methods and models. The valuation methods and models are regulated by national financial supervisory bodies such as the Prudential Regulation Authority (PRA) and the Financial Conduct Authority (FCA)²⁵ for the UK and the Securities and Exchange

²⁵ The UK financial supervisory and regulatory bodies: the Prudential Regulation Authority (PRA) and the Financial Conduct Authority (FCA), were formed on 1st April 2013 as successors to the Financial Services Authority (FSA).

Commission (SEC) for the United States of America (USA). The valuation methods, primarily the fair value method, used after the 2005 accounting change, although under the supervision of domestic financial regulatory bodies, were specified by the IASB accounting standards setting authority. Reviewed in Section 2.2, the fair value approach to valuation has come under much scrutiny, and as stated, is the topic of a discussion paper produced by the IASB (IASB1 2008). It is apparent that the value of a firm's assets, and also its equity and liabilities elements, recognised in the Basle regulatory framework is governed by the valuation approach applied by the firm, irrespective of concerns regarding that approach. Based on this reasoning it may be contended that a requirement to indicate, or even adjust, the valuation of a firm's assets to levels of financial distress risk and information bias captured by the relative delta would be aligned at a minimum to supervisory and regulatory requirements. This study examines this valuation adjustment from the perspective of the Basel capital requirement formula.

2.10.1 The Basel Capital Requirement Formula

The Basel capital requirement formula that represents a firm's capital adequacy is specified by formula (2.4). The Basel regulatory framework specifies that regional regulatory bodies and central banks should stipulate to financial firms, under their supervision, to measure and report asset prices for both on and off balance sheet instruments. Presented in equation (2.4), for a firm, the measure of asset prices for on and off balance sheet instruments specifies the *risk weighted assets (RWA)* denominator component of the Basel capital requirement formula.

The Basel capital requirement formula takes the following form:

$$Capital Requirement = \frac{Total Regulatory Capital}{RWA}$$
(2.4)

Where the capital requirement formula in equation (2.4) represents for a firm its level of capital adequacy, and is calculated using the ratio of the firm's total regulatory capital to its risk weighted assets (*RWA*).

The *total regulatory capital* component in equation (2.4) comprises the measure for a firm's assets that are free from any *measurable risk*, and include reserves and disclosed and *non-disclosed* collateral (Basel 2011). As a base line, the Basel capital requirement

formula stipulates that a firm should have 8% or greater of non-risk regulatory capital to 100% of risk weighted assets. Thus, firms with capital requirement levels measuring 8% or greater are classified as having adequate capital to cover losses resulting from risky assets. That is, a firm would have a level of *total regulatory capital* that is the same or greater than the 8% threshold when compared to 100% of the firm's *risk weighted assets*. Firms with a capital requirement level less than 8% are alerted to be at risk and thus would incur penalties until their capital base is raised to and above the 8% minimum.²⁶

From the Basel capital requirement formula it would be seen that the less the level of risk perceived by a firm's assets, computed for the *risk weighted assets* component, the less the level of *total regulatory capital* that would be needed to meet or exceed the 8% capital requirement. The pre-set risk weights applied to asset classes to compute the *risk* weighted assets component under the Basel capital formula are seen as restrictive by some firms and seen to disfavour financial instruments issued by low credit risk firms (Gup 2004). This was the requirement specified in the original Basel I Accord that would enforce the application of the same pre-set risk weight for the same category of instrument from a low credit risk firm compared to a high credit risk firm. This onesize-fits all characteristic of the initial Basel regulations was designed to provide a simple framework that all firms could apply, irrespective of the level of sophistication of their internal risk management framework. For firms with more sophisticated risk management frameworks the Basel II and III regulations allow application of internal risk weights to compute the capital requirement's risk weighted assets component. The computation and subsequent application of internal risk weights is conditional upon the applying firm providing extensive documentation on its risk measurement and management policies, processes and methods to its regional regulatory bodies.

The most popular risk measurement method endorsed by the Basel (2011) regulations to compute internal risk weights is the *Value-at-Risk* measure. Application of the Value-at-

²⁶ The Basel III regulations specify 8% of risk weighted assets to be the 'Minimum Total Capital' requirement (Basel 2011, p 69). To increase capital levels from this 8%, the Basel III regulations specify a 'Capital conservation buffer' (Basel 2011, p 54) and a discretionary 'Countercyclical buffer' (Basel 2011, p 57). These two financial buffers, in effect, raise the *minimum regulatory capital requirement* from 8% to 8.625% on the 1st of January 2016 and steadily increases year-on-year to 10.5% on the 1st of January 2019 (Basel 2011).

Risk risk measurement approach is also permitted by the IASB (2012) to provide *sensitivity analysis* for a firm's exposure to market risks during the financial year.²⁷

2.10.2 Basel Risk Management Framework and Value-at-Risk

The Value-at-Risk measure was primarily developed to measure the risk for all assets held by a firm, thus providing a measure of the firm's total financial risk exposure. However, the Value-at-Risk measure has become better known for its ability to provide a measure for market risk. That is, the ability to provide measures for a firm's stock market price risk, interest rate risk and currency risk. Previously it was stated that the Basel regulatory framework does not provide a risk adjustment such as the relative *delta measure*, to account for the difference between the changes in a firm's reported key financial accounting totals and the change in its stock market price. From the perspective of the Basel capital requirement it was alluded to earlier that such a risk adjustment, and thus the resulting capital requirement adjustment, could be contended to be in keeping with the essence of the Basel framework. In Appendix J, this risk adjustment based on the relative delta is related to the Value-at-Risk measurement approach. Further, in Section 3.9.6 and Appendix K, this study proposes the extending of the current Basel capital requirement policy to include such a risk adjustment. This study introduces, also in Section 3.9.6, an application of the accounting total to market price relative delta measure. It is proposed that this introduced measure adjusts the Basel capital requirement formula to take account of significant differences between a firm's accounting based value and its market based value.

The reference to the Value-at-Risk measure within the Basel regulatory and the IASB accounting frameworks should alone warrant its detailed review. In addition, the application and associative relationship between Value-at-Risk and the relative delta measure detailed in this study, adds strength to a review of its origins and its methods of implementation. Given this prominence, the Value-at-Risk measure is reviewed in the following sections and detailed in Appendix N.

²⁷ The IASB specify market risk to be: interest rate risk, currency risk and *other price risk*. The IASB specify *other price risk* to be the risk that a financial instrument may become exposed to, as a result of, for example, changes in stock market prices or commodity prices, that is, risks other than interest rate risk or currency risk.

2.11 The Value-at-Risk Measure

As stated earlier, the Value-at-Risk measure was primarily developed to measure the risk for all assets held by a firm, thus providing for a firm its total financial risk exposure. From these beginnings, Value-at-Risk has become better known for the ability to measure market risk. That is, the ability to estimate for a firm its level of exposure to stock market price risk, interest rate risk and currency risk.

The Value-at-Risk measure is specified by JPMorgan Chase & Co.²⁸, the firm attributed to the contemporary Value-at-Risk definition, by the following (JPMorgan Chase and Reuters 1996):

The measure of the maximum potential change in value of a financial instrument for a specified probability over a specified time horizon

Tracing the origins of the Value-at-Risk measure, Dowd (2003) reports that during the late 1970s and 1980s, methods and models were being developed internally by firms to measure the levels of risk attributed to derivatives and other financial instruments. The objective of these methods and models was to provide an aggregate measure of the risks faced by the whole institution, termed firmwide risk²⁹. To emerge from this quest for a measure of firmwide risk during the 1980s was the Value-at-Risk measure. Morgan Stanley Capital International (MSCI), the acquirer in 2010 of RiskMetrics - a firm branched from JPMorgan Chase & Co., credits the origin of the Value-at-Risk measure to banks from the USA during the 1980s (MSCI 1999).

MSCI states that the stimulus for the development of the Value-at-Risk measure was as a response to the developing derivative markets. MSCI claimed that traditional measures such as duration analysis were inadequate to managing a firm's risks attributed to derivative financial instruments. In addition, the move by banks and other financial institutions to revalue derivative instruments from prices in observable

²⁸ JPMorgan Chase & Co. was formed in the year 2000 when Chase Manhattan Corporation merged with J.P. Morgan & Co.. The contemporary Value-at-Risk measure was developed in the 1980s by J.P. Morgan & Co. prior to the merger with Chase Manhattan Corporation. However, for completeness, this study makes reference to JPMorgan Chase & Co. when referring to either firm prior to the merger.

²⁹ Morgan Stanley Capital International (MSCI 1999) and Dowd (2003) use the term 'firmwide risk' to refer to an entire firm's risk exposure.

markets, a process known as marking-to-market (or mark-to-market), proved well suited to the Value-at-Risk model. This suitability has proved to stimulate further the adoption of the Value-at-Risk approach (MSCI 1999).

It was at the request of the JPMorgan Chase & Co. chairman (1980 to 1986) Sir Dennis Weatherstone during the 1980s that the contemporary structure for the Value-at-Risk measure emerged. Dowd (2003) reports that Sir Weatherstone requested his staff to provide him with a daily one page document reporting the risk and potential losses for the firm's whole trading portfolio over the following 24 hours. To meet this request the JPMorgan Chase & Co. staff developed a system based on the Value-at-Risk model. This model was used to measure the risk faced by the bank's individual trading positions and to provide also an aggregated single measure to represent the entire bank's total risk exposure or its firmwide risk. This report became famously known as the 4:15 report because it was presented after the close of trading at 4:15 pm each day.

As suggested earlier, the Value-at-Risk measure, together with the relative delta measure, is central to fulfilling the objectives of this research. This study applies Valueat-Risk primarily to determine for the banking firms their stock market price Value-at-Risk. This study also applies the Value-at-Risk measurement approach to test further its null hypothesis and in so doing also tests the *relative delta* measure in a firmwide risk and regulatory capital context.

2.12 Value-at-Risk Models

The general Value-at-Risk model is considered for application in estimating the risk of future losses for a single financial instrument as well as portfolios of financial instruments. Dowd (2003) specifies the Value-at-Risk approach applied for a single financial instrument to consist of the following procedure:

 A number of possible future prices are estimated or generated for a financial instrument for a predetermined length of time that a firm expects to hold that instrument. This predetermined length of time is termed the time horizon or the holding period for the instrument.

- 2) From the number of possible future prices estimated or generated, the financial instrument's corresponding possible future price *returns* are computed.
- 3) From these possible future returns, a *distribution* for the financial instrument's price returns is constructed.
- 4) For the desired confidence level, usually at the 95% or 99% level, that is the 0.05 or 0.01 level of probability significance respectively, the instrument's price returns distribution is examined and the return corresponding to the desired confidence level is read.

The return that is read from this distribution is the measure for the financial instrument's Value-at-Risk for the given level of confidence and given time horizon. For example, if a financial instrument's Value-at-Risk were to be specified at the 95% confidence level (0.05 significance level) for a 250-day time horizon, then moving from the highest to the lowest returns on the constructed distribution, the 5th percentile return would be the specified measure for the instrument's Value-at-Risk. Thus, this measure would be interpreted by a practitioner to represent, at a 95% level of confidence, that the financial instrument's return would not go below the return represented by this Value-at-Risk level over the next 250 days.

Hull (2009) specifies that a firm could measure Value-at-Risk with time horizons that are less than a day. JPMorgan Chase and Reuters (1996) specify a firm would typically measure Value-at-Risk with a 1-day time horizon. The Basel regulations (Basel 2011) specify that a firm should measure Value-at-Risk for a time horizon determined by the margin or holding characteristics of the traded instrument.

This study presents Value-at-Risk results for the 1-day, 250-day and 500-day time horizons. In addition, a *0-day time horizon* Value-at-Risk, termed the *Historical Value-at-Risk Actual*, is specified. The Value-at-Risk confidence level used in this study is the 95% level, that is, at the 0.05 significance level. However, a comparison of confidence levels based on the 95%, 99% and 99.9% levels is provided.

Inspecting the earlier stated step 1) in this section, when a financial instrument's future price returns are estimated or generated using *historical price movements*, MSCI (1999)

specifies that 2 years, approximately 500 business days, of past price movements should be used. Basel (2011) regulations specify that 3 years, approximately 800 business days, of past price movements should be used. This study estimates and generates price returns based on both 500 and 800 days historical price movements.

Detailed by JPMorgan Chase and Reuters (1996), Dowd (2003), and Hull (2009) there are three principal Value-at-Risk models a firm could apply. These models being:

- 1) The variance-covariance model
- 2) The historical simulation $model^{30}$
- 3) The Monte Carlo simulation model

To measure a financial instrument's Value-at-Risk, the variance-covariance model measures Value-at-Risk based on a primary statistical measure for the instrument's price. This primary statistical measure is the standard deviation. The historical simulation model measures Value-at-Risk based on the instrument's actual historical price information. The Monte Carlo simulation model measures Value-at-Risk based on a *price* or *returns model* that is applied to simulate the financial instrument's possible future price movements. This study applies all three Value-at-Risk models. Sections 2.12.1 to 2.12.3, presented next, provides details for each of these principal VaR model *constructions* at the 95% and 99% confidence levels. In addition, each models advantages and disadvantages are considered.

2.12.1 Variance-Covariance Model

The Value-at-Risk measure for a financial instrument computed with the variancecovariance model, detailed by JPMorgan Chase and Reuters (1996) and Dowd (2003) involves a process based on the standard deviation statistical measure. A normal probability distribution, specified by the distribution's statistical mean and standard deviation, is assumed for the financial instrument's future price movements. This study applies standard deviation estimates based on historical returns. These returns are computed using the required time horizon applied to past market prices.

³⁰ The historical simulation model applied to measure Value-at-Risk is also referred to as the historical Value-at-Risk model.

Once the standard deviation from the market price returns distribution has been estimated, it is simply scaled by the probability that represents the desired confidence level. For a 95% confidence level (that represents the 0.05 probability level, this being the same as the 0.05 significance level), the standard deviation is multiplied by -1.65. For a 99% confidence level (that represents the 0.01 probability level, this being the same as the 0.01 significance level), this standard deviation is multiplied by -2.33. For a normal distribution, these values: -1.65 and -2.33, are measures of the *number of standard deviations from the statistical mean* that represent probabilities 0.05 and 0.01 respectively. Thus, multiplying the standard deviation by -1.65 and -2.33 would provide a measure for the instrument's 5th and 1st percentile returns respectively. For the specified time horizon, these market price returns at the 5th and 1st percentiles represent the instrument's variance-covariance Value-at-Risk measure at the 95% and 99% confidence levels respectively. Details of the variance-covariance Value-at-Risk approach applied in this study are detailed in Appendix N.3.

MSCI (1999) and Dowd (2003) detail the advantages in applying the variancecovariance Value-at-Risk model. These advantages include the model's intuitive nature and its ability to measure the change in an instrument's Value-at-Risk incrementally. The limitation is that the model assumes a financial instrument's market price returns to be normally distributed. Such an assumption ignores the presence of significant skewness in the returns distribution that may result in potentially misleading results. This model is stated to be unsuitable when applied to measure Value-at-Risk for option derivative instruments and other instruments with non-linear payment profiles. In addition, this model is not suited when modelling extreme events, such as event scenarios applied in stress testing.

2.12.2 Historical Simulation Model

The Value-at-Risk measure for a financial instrument computed with the historical simulation model, detailed by JPMorgan Chase and Reuters (1996) and Dowd (2003), involves a process based on the financial instrument's actual historical market price return *distribution*. This model considers that future price movements would be similar to those that have occurred in the past. This Value-at-Risk model is specified by calculating a number of historical market price returns for a given *time horizon*. For example, if a firm would like to hold a financial instrument for 250 days, represented by

a Value-at-Risk time horizon of 250 days, then a number of the instrument's past prices that are 250 days apart are applied on a *rolling basis* to compute the instrument's historical returns. From these historical returns, the model constructs a distribution. This distribution is effectively the returns arranged, or *sorted*, from the highest to the lowest level.

From this distribution, the sorted return that represents the desired confidence level would be read. For a 95% confidence level, that represents the 0.05 probability level, the actual market price return at the 5th percentile would be read from the distribution. For a 99% confidence level, that represents the 0.01 probability level, the actual return at the 1st percentile would be read. For example, if a firm applies the historical simulation Value-at-Risk model by constructing a distribution based on 101 historical market price return observations, then this 5th percentile would be the 96th level when moving along the distribution from the highest to the lowest sorted returns. Then, the 1st percentile would be the 100th lowest level. This example of the historical Value-at-Risk model computes 101 returns to account for the 0th percentile observation, represented by the 101st (lowermost) level. For the specified time horizon, the financial instrument's market price returns read at the 5th and the 1st percentiles represent the instrument's historical simulation Value-at-Risk measure at the 95% and 99% confidence levels respectively.

The previous example applied 101 historical market price return observations to construct a distribution. However, as referred to earlier and in Appendix N, MSCI (1999) specifies approximately 500 days of returns to construct the distribution. The Basel (2011) specification applies approximately 800 days. To test its null hypothesis, this study applies both the MSCI 500 days, and the Basel 800 days historical market price return observations to construct the distribution. Details of the historical simulation Value-at-Risk approach are presented in Appendix N.4.

MSCI (1999) and Dowd (2003) detail the attractions for applying the historical simulation Value-at-Risk model. These include the model's intuitive nature, ease of use when interpreting its results, ease of implementation, not dependent on the *normal distribution* and has the ability to handle all financial instrument positions including derivative positions. The limitations for this model include the unavailability of historical data for some instruments, its distribution is fully dependent on historical

datasets, it has problems with *too short* or *too long* time horizon estimation periods, and it is unable to model future changes in instrument structures.

2.12.3 Monte Carlo Simulation Model

The Value-at-Risk measure for a financial instrument computed with the Monte Carlo simulation model is detailed by JPMorgan and Reuters (1996), Dowd (2003), and Hull (2009). This model applies a future market price return distribution that is constructed using a simulation model. This simulation process considers that a probability distribution for future price movements can be generated by simulating *near-random* stochastic market prices or market price returns.

Metropolis and Ulam (1949) named this simulation process the *Monte Carlo method*. The origin of the Monte Carlo method traces to a particle motion detailed by Brown (1828). Brown examined the movement of pollen particles through the air and determined that the path taken by a particle was as a consequence of its interactions with rapidly moving smaller particles. This *particle path motion* reported by Hull (2009) is famously known as the Brownian motion.

Metropolis and Ulam (1949) found that they were unable to model the path behaviour of some atomic particles using classical mathematical techniques. They propositioned that a simulation method based on the Brownian motion would provide a particle path model that overcame any classical mathematical limitations. To model a particle's motion Metropolis and Ulam proposed an iterative simulation process. This process specified a stochastic model that applied a random probability level between 0 and 1 to a *well-defined* mathematical model. They proposed that the process is repeated, using incremental time steps for a desired time horizon, to produce a single possible particle path or branch. They then proposed that this process should be repeated many times to produce a finite branching graph that represents the simulated particle paths. They concluded that if it were possible for the process to be repeated near or to infinity then the resulting space of branches represented a probability distribution that converges to a statistical mean representing the particle's true value. This would be similar to saying that if a fair coin is tossed an infinite number of times then the chance for the coin landing head side up would converge to a statistical mean value of 0.5.

Applying the same Monte Carlo simulation principle to the path that a financial instrument's price is expected to take during a specified time horizon, would result in a similar finite branching graph. Such a graph would provide a simulated distribution for the instrument's possible prices that may be arranged, or *sorted*, from the highest to the lowest price, or its highest to lowest calculated price returns.

Applying a nearly identical process to the historical Value-at-Risk model, the sorted price return that represents the desired confidence level would be read. For a 95% confidence level, that represents the 0.05 probability level, the simulated price return representing the 5th percentile would be read from the distribution. For a 99% confidence level, that represents the 0.01 probability level, the simulated return at the 1st percentile would be read. For example, if a firm applies the Monte Carlo simulation Value-at-Risk model by constructing a distribution based on 101 simulated price returns, then the 5th percentile return would be the 96th level when moving along the distribution from the highest to the lowest sorted return. Then, the 1st percentile return would simply be the 100th lowest level. This model simulates 101 returns to account for the 0th percentile observation, represented by the 101st (lowermost) level. For the specified time horizon, the financial instrument's returns read at the 5th and 1st percentiles represent the instrument's Monte Carlo simulation Value-at-Risk measure at the 95% and the 99% confidence levels respectively.

The previous example applied 101 simulated price returns to construct a distribution. To test its null hypothesis, this study applies approximately 1 million simulated price returns for the distribution. Details for this Monte Carlo simulation Value-at-Risk approach are presented in Appendix N.5.

MSCI (1999) and Dowd (2003) detail the attractions for applying the Monte Carlo simulation Value-at-Risk model. These include the model's power and flexibility to measure virtually any financial position, it has the ability to measure exotic options and complex positions, and it is not dependent on the *normal distribution*. The limitations for this model are that it is not intuitive, it is difficult to explain, implementation requires computer processing time, implementation is currently costly, and it's results are dependent on the applied mathematical model and stochastic process.

2.12.4 Stress Testing

Hull (2009) specifies stress testing to be a process where a financial instrument or portfolios of financial instruments are evaluated to determine the size of potential losses in response to scenarios based on extreme conditions. These include conditions such as extreme financial market movements. The Value-at-Risk measure is adapted for stress testing by supplying the model with the parameters specified from events that have already taken place. These parameters may also be specified by management based on global economic uncertainties. In addition to the parameters supplied, during stress testing the Value-at-Risk model would be taken beyond its normal probability significance levels.

According to Dowd (2003), the advantages in applying this stress test process include the ability to handle any specific scenario, ability to model extreme events, ability to model what-if scenarios, provide assistance with the risk management function, and the process works well with the Monte Carlo simulation model. According to Dowd (2003) and Abbink (2011) the disadvantages with this process are that it is reliant on how accurately scenarios are specified, and implementation may be complex and difficult.

2.12.5 Backtesting

Backtesting applied to a model, or a *system* used for decision making, is a process that monitors the validity of the model, or system, by comparing its predicted results with actual results. Hull (2009) states that backtesting is a process that involves comparing a financial instrument's predicted Value-at-Risk with its actual market price movements once the prediction period has passed. For a Value-at-Risk model, the backtesting process provides a measure of its accuracy, and also provides a framework to calibrate a particular model. This study applies backtesting to produce the yearly results presented in Table P.2. Details of the backtesting approach applied in this study are presented in Appendix N.7.

2.13 Contemporary Technology Framework

Stated by Banks (2002) and Merton (1995), financial instruments continue to grow in both numbers and complexity of structures. It may be debated that a need to maintain an understanding of complex financial instruments is a critical requirement to develop and implement adaptable and effective accounting, risk, regulatory and technology processes and frameworks, and the corresponding technology interfaces and platforms.

Concentrating on risk management technologies, they have been criticised to be not fully developed (CFA 1997). This lack of risk management systems development has resulted in firms having a restrictive firmwide risk management and disclosing framework. Although noted to exhibit improvement, on the whole, there is criticism regarding technology. In that, technology systems being introduced to manage accounting, risk management and regulatory frameworks need radical re-engineering to accommodate new financial instruments and structures (Merton and Bodie 1996, IASB 2009). A major aspect of this re-engineering would be to recognise effectively and measure risks associated with these instruments and structures. As alluded to in the following paragraphs, these findings are further strengthened during the course of this study's investigations.

To build evidence to either verify or falsify the null hypothesis, this study has developed the relative delta measure by extending the time dependent book-to-market measure specified by Fama and French (2008). The accounting totals and market price based relative delta measure, introduced in Section 2.7 and specified in Section 3.2.4 and Appendix C, has been applied in this study to effectively provide estimates for the difference between the Value-at-Risk calculated from the change in accounting total variables and the Value-at-Risk calculated from the market price return variable (see Section 3.9.5). The accounting and market price based relative delta measure, in the context of its relation to the Value-at-Risk measure, has also been applied to estimate how its varying levels affect the Basel Regulatory capital requirement (see Section 3.9.6).

Following on from this, it may be contended, that by observing the book-to-market ratio's accounting and market components, and their relation to the level of financial distress risk, the accounting discipline has a role in risk management. In addition, within the context of risk management as prescribed by the Basel (2011) regulatory framework, the capital requirement formula that is utilised to manage risk is founded primarily on approaches applied to calculate accounting totals. This study itself emphasises the role of accounting within a risk management context by the development of the relative delta measure. The relative delta measure, built on

62

accounting and market price variables, as stated, is developed to test its null hypothesis. This study also proposes that the relative delta measure, applied to measure a firm's exposure to financial distress risk, incorporates and informs at the firmwide level, and also at the national and international regulatory risk management levels. It is evident, based on the discussion thus far, that the relative measurement of accounting total variables to the market price variable would alone provide a beneficial measure of financial distress risk within both the regulatory and firmwide risk management frameworks.

To implement such a regulatory and risk management framework requires a resolution to a timing limitation. This limitation arises from the disparity between the measurement frequency of accounting variables to that of market price variables. Market prices are reported virtually on a real-time basis; however, audited accounting totals are reported on a yearly basis. Ideally, to allow the accurate and timely capture of accounting information and market price information, both variables would require updating and measuring at the same frequency. To measure *relative Value-at-Risk levels* or relative delta levels it would be ideal to have a measurement frequency that is *realtime* or as close to real-time as possible.

The current accounting and risk management technology frameworks reflect this timing issue. Although risk management systems have come under criticism (CFA 1997); Jorion (2009) reports that risk management systems have advanced to such a level that they can generally be updated on a *real-time* basis with current market price and modelled price information. With accounting systems, once items are posted, prices for those items are difficult to update on a real-time basis. This is mainly due to the underlying legacy mainframe accounting systems. Banks (2002) evaluates that in general, technology processes implemented in the mid 1990s are rigid and focused only on what needed to be done at the time with little alternative for product change.

From the perspective of process components, a schematic overview of a current characteristic technology framework applied in a bank is presented in Figure 2.1.

Figure 2.1 details that, at time t = T, denoted by t_T (Transaction/Trade time), financial instruments are updated on a real-time basis with current prices. The priced financial instruments are available to the trade process and the risk management process at time

 t_T . If an instrument is traded at time t_T , the transaction is registered at the trade process at time t_T , and the transacted item's price and other details are passed to the settlement process. The settlement process would then generally check the transacted instrument's transaction details, and register its traded price and details at the trade time, T. The settlement of the transacted instrument may be instantaneous or it may take several days, incurring a delay of time S, denoted by t_S (Settlement time). Once the transaction details pass the settlement process the instrument's price and details are posted to and recognised at a quality control process or accounting process at time t = T plus S, denoted by t_{T+S} . Once transaction details are posted to the accounting process, it is difficult for the current accounting systems to attach additional price changes for the transacted instrument or item. Thus, the price applied to the item within the accounting system generally remains at the price level attached after settlement at the time T plus S.

Presented in Section 4.11 and shown in Figure 4.1, this study proposes a technology framework for a bank that principally applies a *real-time* pricing process to the accounting process. Such a technology framework is proposed to allow the measurement of both the accounting total variables and the market price variables on a *real-time* or close to real-time basis.

Figure 2.1Bank Technology Framework OverviewCurrent banking technology framework for financial instrument and financial item: pricing, trading, risk management, and accounting.



2.14 Summary

This chapter has presented a review of the Fama and French (2008) time dependent book-to-market ratio treatment, and provided this study's specifications for its extension that is termed the relative delta measure. This chapter discusses recent accounting standard developments focusing on the IASB accounting standards, and discusses literature that subjects the 2005 IASB's IFRS accounting standards implementation. The IASB accounting standards are reviewed to lay a foundation to extend current literature. It is proposed to extend current literature by testing UK listed banks' accounting quality reactions before and after the 2005 IFRS accounting implementation. These accounting quality reactions specified as being generally quantified by the proposed relative delta measure. The important relationships between accounting quality, accounting quality risk, value relevance, the relative delta measure and other key measures applied in this study have been formally introduced. The contemporary risk measurement methods, and the Value-at-Risk approach and its models are reviewed together with stress testing and backtesting approaches. The Basel regulatory framework is reviewed from the context of its minimum capital requirement. The Basel regulatory framework's relation to Value-at-Risk is introduced in order to lay the foundation to apply the relative delta measure to gauge and adjust for financial distress risk at the regulatory level. This chapter also presents an overview regarding technology considerations to implement what this study proposes as the accounting Value-at-Risk³¹, and the proposed relative delta measure and its application at the regulatory level.

³¹ This study proposes the measurement of the *change in accounting totals Value-at-Risk* that applies the key accounting total *change variables* and is termed the *accounting totals Value-at-Risk*. This same term also refers to the *accounting totals Value-at-Risk* that applies the *total variable* in place of the *change variable*.

Chapter 3 Research Methodology

3.1 Chapter Introduction

This chapter introduces the methodology applied in this research that is guided by the hypothesis based deductive methodology, termed the hypothetico-deductive methodology, combined with the falsification theory (Popper 1992).

The methodological approach first details the research question that this study examines. The research question, presented in Section 3.2, is then framed as a null hypothesis. This null hypothesis is presented in Section 3.3.1. The null hypothesis is then examined by gathering evidence from the tests presented in summary in Section 3.8 and presented in detail in Section 3.9 and Appendix M to Appendix R. The results from these tests are then assessed to determine if they provide evidence to confirm the expectations of conclusive statements presented in Chapter 5. The evidence from these tested statements is then assessed to deduce whether they *add strength to verify* the null hypothesis or if they *add strength to falsify* or weaken the null hypothesis.

One of the primary measures used to test the null hypothesis is the *relative delta measure*. Sections 2.7 and 3.2.4 details the development of the relative delta measurement system from the extension of the Fama and French (2008) treatment of the book-to-market ratio. This extension maintains the relative delta to be a measure of financial distress risk. In Section 2.9, this measure is presented also to reflect levels of accounting quality and information bias. Specified in Section 3.9.6 and Appendix K, the relative delta is applied to determine the role the accounting standards played in the context of the Basel risk management framework.

To test deductively the null hypothesis this chapter specifies the quantitative analytical measures applied in this study. These measures include the Value-at-Risk measure, descriptive statistics, percentage changes, material significance of *difference* - calculated by applying the introduced *relative delta measure*, and material significance of changes to the Basel regulatory capital requirement - calculated by applying the introduced *relative delta measure*. This study principally applies the ordinary-least-squares time series and cross-sectional regression analysis approaches

and the Pearson correlation measure to determine levels of statistical significance for selected tests applied to its null hypothesis.

This chapter specifies the sample survey selection criteria and the time series ranges applied in this study, and the selected firms for survey are presented. The criteria for the selection of a control group are specified and the control group firms are also presented. This chapter also presents the information and data collection methods used in this study.

To deduce the effects of general economic and market conditions during the 2005 accounting change, the tests applied to the UK Gross Domestic Product figures, general market variables and market and accounting variables are presented in this chapter. These tests are conducted in addition to those aimed directly at testing the null hypothesis.

3.2 Research Question

The research question pertaining to this study is:

Does accounting quality improve for UK banks that adopted the IFRS accounting standards in 2005? That is, does the difference between the change in accounting totals and the change in market price decrease for UK banks that adopted the IFRS accounting standards in 2005 when compared to before the adoption?

Refining the research question further to address each of the key financial statement accounting variables, the research question poses the following *supplementary questions*:

For the UK banks, from before to after 2005, what was the change in the level of difference between:

1) The change in the total equity variable and the change in the market price variable?

2) The change in the total assets variable and the change in the market price variable?

3) The change in the total liabilities variable and the change in the market price variable?

4) The change in the net income variable and the change in the market price variable?

3.2.1 Research Question Examination

In a general sense, the research question asks - for the UK banks that changed accounting standards in 2005, what was the level of difference between the change in key financial statement accounting variables and the change in the market price variable before and after the accounting change? Importantly, the question asks - were the before and after 2005 levels of *difference* between the change variables significant?

As stated, the motivation for this study's research question is to determine the effects of the 2005 accounting change on the accounting and market price variable relationship. The general approach this study applies to quantify this relationship between accounting values and the market price value is attributed to the book-to-market ratio effect detailed by Fama and French (1992, 2008), Peterkort and Nielsen (2005) and Bodie et al. (2002). They detail that the book-to-market ratio effect determines that firms with a high bookto-market ratio exhibit a high level of exposure to financial distress risk, and firms with a low book-to-market ratio exhibit a lower level of financial distress risk. If this effect were related to the measure of difference between the change in key accounting total variables and the change in the market price variable then such an effect would provide an analysis of a bank's level of exposure to financial distress risk. Moving full circle, if the financial distress risk effect were considered to be directly proportional to the level of difference between the accounting and market price change variables for the UK banks, it would be possible to determine the levels of financial distress risk exposure the banks experienced before and after the 2005 accounting change. Aligning to the research question, and from the relationships presented in Section 2.9, this measure of the level of difference between the change variables would also translate to the level of accounting quality experienced by the banks before and after the 2005 change.

It must be noted that although this study applies the interpretation from Fama and French (1992, 2008), Peterkort and Nielsen (2005) and Bodie et al. (2002) that high book-to-market value firms exhibit exposures to financial distress risk, Dichev (1998),

Griffin and Lemmon (2002), and Zaretsky and Zumwalt (2008) find that low book-tomarket value firms exhibit exposures to distress risk. As stated, this study primarily applies the former interpretation of financial distress risk. However, both approaches support this study's definition of accounting quality, where measures of both high and low levels of book-to-market value translate to a decrease in the level of accounting quality, and thus also translate to an increase in the level of exposure to accounting quality risk. This relationship between accounting quality, the book-to-market ratio and financial distress risk is quantified by the difference component and the relative delta measure (see Appendix C, Table C.1 to Table C.3 and Appendix D, Table D.1). The relationship between accounting quality, financial distress risk, the difference component and the relative delta measure is presented in Section 2.9.3. In addition, Section 2.9.3 presents the relationship between the relative delta measure and the difference component, and their relationship to the measure of information bias (IASB 2011) between the accounting and market price data elements.

Having introduced this general measurement approach, a testable null hypothesis that addresses the research question must determine the level of difference between the change variables, and the significance attached to this level of difference for before and after the accounting standards adoption. To measure and thus examine this level of difference this study extends the Fama and French (2008) structure for the time dependent treatment of the book-to-market ratio. This extension is discussed in Section 2.7, and gradually developed from Sections 3.2.2 to 3.2.5 that starts from the book-to-market ratio's measure of financial distress risk and then onto its relation to the *difference component*. This *difference component*, strengthened by its relationship to the relative delta and importantly to the measure of accounting quality (see Section 2.9.3), is specified in Sections 3.2.4 and 3.2.5. In these sections the *difference component* is specified in relation to the relative delta measure, prior to the relative delta's extensive application in testing this study's null hypothesis specified in Section 3.3.1.

3.2.2 The Fama and French Book-to-Market Ratio and Financial Distress Risk

From the Fama and French (1992, 2008) findings, it is evidenced that a firm's book-tomarket ratio provides a measure of its future returns and its future level of exposure to financial distress risk. Fama and French (2008) also provide evidence that a firm's evolution of the book-to-market ratio contains information that improves predictions for expected returns and thus also improves predictions for a firm's level of exposure to financial distress risk (see Chapter 1, Sections 2.1, 2.3, 3.2.1, Appendix A, Appendix B and Appendix C).

As alluded to earlier, to examine this study's research question and permit the direct analysis of its null hypothesis, this research develops the *relative delta measure* by extending the Fama and French time dependant treatment for the book-to-market ratio (Fama and French 2008).

The relative delta measure develops the *book-to-market effect* detailed by Fama and French (1992, 2008), Peterkort and Nielsen (2005) and Bodie et al. (2002). The book-to-market effect relies on the book-to-market ratio to equate to *one* when there is no inherent perception of financial distress related risk, and above or below *one* when a perception of financial distress risk is present or *increasingly not* present respectively. For a firm's financial stock instrument, the level by which the book-to-market measure is above or below *one*, respectively determines whether the financial instrument is undervalued or overvalued (Fama and French 1992, 2008, Peterkort and Nielsen 2005). Aligned to the components of the book-to-market ratio, this study specifies the book value component to be the measure of change in an *accounting total variable* for a specified time period, and the market price component to be the measure of the change in the *market price variable* for the same specified time period (Fama and French 2008) this study determines the following relationships:

$$dAC_{t-1,t} = ln\left(\frac{AC_t}{AC_{t-1}}\right)$$
(3.1)

$$dM_{t-1,t} = ln\left(\frac{P_t}{P_{t-1}}\right) \tag{3.2}$$

Where:

 $dAC_{t-1,t}$ = change in the accounting variable between period t and t-1 $dM_{t-1,t}$ = change in the market price variable between the period t and t-1 ln = natural logarithm to the base e AC_t = measure of the accounting variable at time t AC_{t-1} = measure of the accounting variable at time t-1 P_t = measure of the market price variable at time t P_{t-1} = measure of the market price variable at time t - 1t = time index

Equations (3.1) and (3.2) are then combined to provide a relative measure of the change in the accounting variable for a given time period to the change in the market price variable for the same time period. Combining equation (3.1) and (3.2) defines:

$$\delta_{t-1,t} = dAC_{t-1,t} - dM_{t-1,t} \tag{3.3}$$

Where:

 $\delta_{t-1,t}$ = relative delta: change in the accounting variable for a unit change in the market price variable between the period *t* and t-1 $dAC_{t-1,t}$ = change in the accounting variable between the period *t* and t-1 $dM_{t-1,t}$ = change in the market price variable between the period *t* and t-1

t = time index

It is possible to apply now the definition of the book value to market value relationship specified by Fama and French (1992, 2008), Peterkort and Nielsen (2005) and Bodie et al. (2002), to that of the relative delta measure specified in equation (3.3). It is so that if there is no perception of financial distress related risk for a firm's financial stock instrument, then the relative delta measure specified in equation (3.3) must equate to zero (= 0). If financial distress related risk is present then equation (3.3) may be evaluated as a measure for a firm's level of exposure to financial distress risk. In addition, given the relationships presented in Section 2.9.3 and 3.2.1 (see also Appendix D, Table D.1) the level of financial distress risk measure using the relative delta measure in equation (3.3) also translates to a measure of a firm's accounting quality.

3.2.3 The Fama and French Book-to-Market Ratio and the Difference Component

The relation between the book-to-market effect and the relative delta measure applied in this study to test its null hypothesis has been discussed in Section 3.2.2. In addition, Section 3.2.2 and Appendix C, and Table C.1, have introduced the relation between the book-to-market effect, financial distress risk and the relative delta measure.

As a prelude and foundation for Section 3.2.4, this section presents the details for the development of the relative delta measure from the Fama and French (2008) specification for the time *change* component from the book-to-market ratio. This section also presents the specification for the application of the relative delta measure as a quantitative analytical method to measure the difference between the change in the key accounting total variables and the change in the market price variable.

Fama and French (2008) develop the *change* of the book-to-market ratio (BM) by applying the following relationship:

$$BM_{t} = BM_{t-k} + \left[dB_{t-k,t} - dM_{t-k,t} \right]$$
(3.4)

Where:

 BM_t = natural log of book-to-market ratio at time t BM_{t-k} = natural log of book-to-market ratio at time t - k B = book equity per share measure for a firm M = market price for 1 share of the firm's stock $dB_{t-k,t}$ = natural log change in book equity value from time t - k to time t $dM_{t-k,t}$ = natural log change in market price from time t - k to time t t = time index k = chronological time interval

The term $[dB_{t-k,t} - dM_{t-k,t}]$ is the component that measures the *change*, or the *difference*, of the book-to-market ratio. This *change component*, or *difference component*, of the book-to-market ratio Fama and French (2008) specify as the *difference* between the change in book equity, dB, and the change in market price, dM, for the time period t - k to t.

Specifying this *difference component* from equation (3.4) gives the following equation:

$$\delta BM_{t-k,t} = dB_{t-k,t} - dM_{t-k,t} \tag{3.5}$$

Where:

 $\delta BM_{t-k,t}$ = difference between the change in the book equity value and the change in the market price from time t - k to time t

 $dB_{t-k,t}$ = natural log change in book equity value from time t - k to time t

 $dM_{t-k,t}$ = natural log change in market price from time t - k to time t

- t = time index
- k = chronological time interval

Fama and French (2008) specify the construction of the BM_t ratio in equation (3.4) using approaches: i) the book equity value per total market value, presented in equation (3.6); and ii) the book equity value per share per unit stock price, presented in equation (3.7):

$$BM_t = \frac{B_t}{(S_t \times M_t)} \tag{3.6}$$

Where:

 BM_t = book-to-market ratio at time t specified using book equity to market value $(S_t \times M_t)$ = market value³² at time t

S = total number of a firm's issued shares outstanding in the stock market

B = book equity value

M= stock market price of 1 share for the firm's stock

 S_t = total number of the firm's issued shares outstanding in the stock market at time t

 B_t = book equity value at time t

 M_t = stock market price of 1 share for the firm's stock at time t

t = time index

$$BM_t = \frac{B_t}{S_t} \times \frac{1}{M_t} \tag{3.7}$$

Where:

 BM_t = book-to-market ratio at time t specified using book equity per share to stock market price

 $\frac{B_t}{S_t}$ = book equity per share

S = total number of a firm's issued shares outstanding in the stock market

B = book equity value

M = stock market price for 1 share of the firm's stock

 S_t = total number of the firm's issued shares outstanding in the stock market at time t

³² Fama and French (2008) refer to *market value* as *market capitalisation* or *market cap*.

 B_t = book equity value at time t

 M_t = stock market price of 1 share for the firm's stock at time t

t = time index

Rewriting equation (3.4), using the book-to-market specification presented in equation (3.6) that examines the book-to-market ratio using the *market* value variable, gives:

$$BM_{t} = BM_{t-k} + \left[dB_{t-k,t} - d(S \times M)_{t-k,t} \right]$$
(3.8)

Where:

 BM_t = natural log book-to-market ratio at time t

 BM_{t-k} = natural log book-to-market ratio at time t - k

S = total number of a firm's issued shares outstanding in the stock market

B = book equity value

M = stock market price for 1 share of the firm's stock

 $dB_{t-k,t}$ = natural log change in book equity value from time t - k to time t

 $d(S \times M)_{t-k,t}$ = natural log change in market value from time t - k to time t

t = time index

k = chronological time interval

Rewriting equation (3.8) in natural log form and expanding for time periods t and t - k, gives:

$$ln\left(\frac{B_t}{M_t}\right) = ln\left(\frac{B_{t-k}}{M_{t-k}}\right) + \left[ln\left(\frac{B_t}{B_{t-k}}\right) - ln\left(\frac{\{S \times M\}_t}{\{S \times M\}_{t-k}}\right)\right]$$
(3.9)

Where:

 $ln\left(\frac{B_t}{M_t}\right) = \text{natural log book-to-market ratio at time } t$ $ln\left(\frac{B_{t-k}}{M_{t-k}}\right) = \text{natural log book-to-market ratio at time } t - k$ $ln\left(\frac{B_t}{B_{t-k}}\right) = \text{natural log change in book equity value from time } t - k \text{ to time } t$ $ln\left(\frac{\{S \times M\}_t}{\{S \times M\}_{t-k}}\right) = \text{natural log change market value from time } t - k \text{ to time } t$ ln = natural logarithm to the base e k = chronological time interval t = time index

Specifying the *difference component* from equation (3.5) using equation (3.9) gives:

$$\delta BM_{t-k,t} = ln\left(\frac{B_t}{B_{t-k}}\right) - ln\left(\frac{\{S \times M\}_t}{\{S \times M\}_{t-k}}\right)$$
(3.10)

Where:

 $\delta BM_{t-k,t}$ = difference between the change in the book equity value and the change in the market price from time t - k to time t. The other parameter definitions are the same as equation (3.9).

Rewriting equation (3.4), using the book-to-market ratio specification presented in equation (3.7) to examine now the book-to-market ratio approach that uses the *equity per share* variable, gives:

$$BM_{t} = BM_{t-k} + \left[d(B/S)_{t-k,t} - dM_{t-k,t}\right]$$
(3.11)

Where:

 BM_t = natural log book-to-market ratio at time t

 BM_{t-k} = natural log book-to-market ratio at time t - k

S = total number of a firm's issued shares outstanding in the stock market

B = book equity value

M = stock market price for 1 share of the firm's stock

 $dM_{t-k,t}$ = natural log change in market price (market price return) from time t - k to time t

 $d(B/S)_{t-k,t}$ = natural log change in equity per share from time t - k to time t

t = time index

k = chronological time interval

Rewriting equation (3.11) in natural log form and expanding for time period t - k to t, gives:

$$ln\left(\frac{B_t}{M_t}\right) = ln\left(\frac{B_{t-k}}{M_{t-k}}\right) + \left[ln\left(\frac{\{B/S\}_t}{\{B/S\}_{t-k}}\right) - ln\left(\frac{M_t}{M_{t-k}}\right)\right]$$
(3.12)

Where:

 $ln\left(\frac{B_t}{M_t}\right) =$ natural log book-to-market ratio at time t

 $ln\left(\frac{B_{t-k}}{M_{t-k}}\right) = \text{natural log book-to-market ratio at time } t - k$ $ln\left(\frac{\{B/S\}_t}{\{B/S\}_{t-k}}\right) = \text{natural log change in book equity per share from time } t - k \text{ to time } t$ $ln\left(\frac{M_t}{M_{t-k}}\right) = \text{natural log change in market price from time } t - k \text{ to time } t$

Specifying the *difference component* from equation (3.5) using equation (3.12) gives:

$$\delta BM_{t-k,t} = \ln\left(\frac{\{B/S\}_t}{\{B/S\}_{t-k}}\right) - \ln\left(\frac{M_t}{M_{t-k}}\right)$$
(3.13)

Where:

 $\delta BM_{t-k,t}$ = difference between the change in the book equity value and the change in the market price from time t - k to time t. The other parameter definitions are the same as equation (3.12).

Specifying the book-to-market ratio's *difference component* by applying the market value based measure presented by equation (3.10) or the book equity per share measure presented by equation (3.13) generally leads to the same result (Fama and French 2008).

This study analyses the before and after 2005 book-to-market ratio effects by applying both the market value and the equity per share measurement approaches. The market value effect is analysed by determining its relationship with the market price variable (see Appendix T). The equity per share effect is analysed by applying equation (3.12) to determine the relationship between the equity per share component and the market price component (see Section 4.3.2).

The *difference* measurements specified in equation (3.10) and equation (3.13), do not react to a firm's market price adjustments from events such as share issuance or share buybacks, events that effectively change the firm's number of outstanding shares in the market. However, the relative delta difference measurement specified in Section 2.7, criterion 3, applies the change in the total equity variable that is sensitive to the changes in outstanding shares. To reflect this effect from changes in shares on the measure of difference between the change in the book equity variable and the change in the market price variable, this study extends the Fama and French (2008) specification by introducing the following relationship:

$$\delta E M_{t-k,t} = \left[d E_{t-k,t} - d M_{t-k,t} \right] \tag{3.14}$$

Where:

 $\delta EM_{t-k,t}$ = difference between the change in the total equity variable to the stock market price variable from time t - k to t

E = total equity representing the balance sheet measure of the total shareholders' equity for a firm

M = stock market price for 1 share of the firm's stock

 $dE_{t-k,t}$ = natural log change in the total equity variable from time t - k to time t $dM_{t-k,t}$ = natural log change in the market price variable from time t - k to time tt = time index

k = chronological time interval

Rewriting equation (3.14) in natural log form gives:

$$\delta E M_{t-k,t} = \left[ln \left(\frac{E_t}{E_{t-k}} \right) - ln \left(\frac{M_t}{M_{t-k}} \right) \right]$$
(3.15)

Where:

 $\delta EM_{t-k,t}$ = difference between the change in the total equity variable to the stock market price variable from time t - k to t

- E_t = total equity at time t
- M_t = market price at time t

 E_{t-k} = total equity at time t - k

 M_{t-k} = market price at time t - k

ln = natural logarithm to the base e

t = time index

k = chronological time interval

3.2.4 The Relative Delta Measure

From the Fama and French (2008) relative change component specified in equation (3.5) and its development to the relative delta measure specified in equation (3.14), the relative delta measure for the general accounting total change to the market price

change is specified by (note that equation (3.16) specifies the same relationship as equation (3.3)):

$$\delta_{t-1,t} = dAC_{t-1,t} - dM_{t-1,t} \tag{3.16}$$

Where:

 $\delta_{t-1,t}$ = relative delta measure between the period t and t-1 $dAC_{t-1,t}$ = change in the accounting variable between the period t and t-1 $dM_{t-1,t}$ = change in the market price variable between the period t and t-1t = time index

The relative delta measure is developed by this study to assist in addressing its research question. As stated earlier, it does so by providing a framework to test the null hypothesis specified in Section 3.3.1. That is, the relative delta measure is applied to determine the change to the levels of difference between the change in key accounting variables and the change in the market price variable, effectively measuring the *difference component*, for before and after the 2005 accounting change. The relative delta measure is expected to align in the same way to the measure for financial distress risk, as does the specification for the book-to-market ratio detailed by Fama and French (2008, 1992), Peterkort and Nielsen (2005) and Bodie et al. (2002).

3.2.5 Key Accounting Elements and the Difference Component and Relative Delta

From the general relative delta form presented in equation (3.16), this section presents the development of the individual accounting totals to market price relative delta measures. These relative delta measures are applied to test the detailed null hypotheses presented in Section 3.3.2. The evidence from these tests is applied to test the null hypothesis presented in Section 3.3.1.

The key accounting total *elements*³³ from the financial statement that are applied in this study and specified by the IASB (2012) are:

³³ The term accounting *element* is used in this study to maintain continuity with the IASB (2008, 2009, 2011, 2012) specification, and has the same general meaning as the term accounting variable.

- 1) Total Shareholders' Equity (*Total Equity*)
- 2) Total Assets
- 3) Total Liabilities
- 4) Net Income

Reproducing the form presented in equation (2.3), the elements total shareholders' equity, total assets and total liabilities are presented on the balance sheet financial statement and maintain the following generalised relationship (IASB 2012):

$$Equity = Assets - Liabilities \tag{3.17}$$

The net income element is presented on the income statement, and represents the change in the level of the assets and liabilities elements for the financial year and thus also provides a guide for total equity for that financial year. The net income element has a direct relation to equity other than for contributions from, and distributions to, equity participants (IASB 2012).

Applying the Fama and French (2008) notation and restating equation (3.14) (with different bracketing) specifies the following set of equations. The set of equations are applied to determine the difference between the change in the key accounting variables and the change in the market price variable and is specified as follows:

$$\delta E M_{t-k,t} = (d E_{t-k,t} - d M_{t-k,t})$$
(3.18)

$$\delta AM_{t-k,t} = (dA_{t-k,t} - dM_{t-k,t})$$
(3.19)

$$\delta LM_{t-k,t} = (dL_{t-k,t} - dM_{t-k,t})$$
(3.20)

$$\delta IM_{t-k,t} = (dI_{t-k,t} - dM_{t-k,t})$$
(3.21)

Where in equations (3.18) to (3.21):

 $\delta EM_{t-k,t}$ = difference between the change in the total shareholders' equity variable and the stock market price variable from time t - k to t $\delta AM_{t-k,t}$ = difference between the change in the total assets variable and the stock market price variable from time t - k to t

 $\delta LM_{t-k,t}$ = difference between the change in the total liabilities variable and the stock market price variable from time t - k to t

 $\delta IM_{t-k,t}$ = difference between the change in the net income variable to the stock market price variable from time t - k to t

E = balance sheet measure of the total shareholders' equity for a firm

A = balance sheet measure of the total assets for the firm

L = balance sheet measure of the total liabilities for the firm

I = income statement measure of net income for the firm

M = stock market price for 1 share of the firm's stock

 $dE_{t-k,t}$ = natural log change in the total shareholders' equity variable from time t - k to time t

 $dA_{t-k,t}$ = natural log change in the total assets variable from time t - k to time t

 $dL_{t-k,t}$ = natural log change in the total liabilities variable from time t - k to time t

 $dI_{t-k,t}$ = natural log change in the total net income variable from time t - k to time t

 $dM_{t-k,t}$ = natural log change in the market price variable from time t - k to time t

ln = natural logarithm to the base e

t = time index

k = chronological time interval

This study terms equations (3.18), (3.19), (3.20) and (3.21) as the *relative delta measures* and are developed in Appendix C to Appendix E and specified in Sections E.1.1, E.1.2, E.1.3 and E.1.4.

3.3 Null Hypothesis and Detailed Null Hypotheses

To address the research question presented in Section 3.2, this study's research design states and tests the following null hypothesis and a system of detailed null hypotheses. These hypotheses are specified by referring to the *difference component* and its measurement framework, the relative delta, specified in Sections 3.2.4 and 3.2.5.

3.3.1 Null Hypothesis

This study structures its approach to address the research question by formulating a *general* null hypothesis, H_0 . The *general* null hypothesis is specified as:

 H_0 : Accounting quality in UK banks was not affected by the adoption of the IFRS accounting standards in 2005. For the UK banks that adopted the new accounting standards in 2005, there is no significant difference between the change in accounting totals and the change in market price after 2005 when compared to the level of difference before.

The null hypothesis, H_0 , is specified to determine the following hypothetical relationship:

$$H_0: \delta_{t \le 2004} = \delta_{t \ge 2005} \tag{3.22}$$

Where for the banks in the UK banking sector:

 H_0 = difference between the key accounting variables and market price variable null hypothesis

 $\delta_{t \le 2004} = \delta_{t \ge 2005}$ = difference between the change in key accounting variables and the change in the market price variable for before and after the 2005 accounting change $t \le 2004$ = measure of time in years that specify the time interval 1994 to 2004 $t \ge 2005$ = measure of time in years that specify the time interval 2005 to 2008

To test the null hypothesis a set of detailed null hypotheses are specified that are founded on the supplementary research questions presented in Section 3.2.

3.3.2 Detailed Null Hypotheses

Founded on and aligned to the supplementary research questions presented in Section 3.2, and derived from the null hypothesis specified in Section 3.3.1, a set of *detailed null hypotheses* are specified and tested by applying the relative delta *measures* presented in equations (3.18) to (3.21) in Section 3.2.5. The set of detailed null hypotheses is specified to take the following form:

For UK banks, from before to after 2005, there was no significant change to the level of difference between:

- 1) $H_{0:E}$: the change in the total equity variable to the change in the market price variable.
- 2) $H_{0:A}$: the change in the total assets variable to the change in the market price variable.
- *3) H*_{0:*L*}: *the change in the total liabilities variable to the change in the market price variable.*
- 4) $H_{0:I}$ the change in the net income variable to the change in the market price variable.

The *detailed null hypotheses* aim to provide a structure to test the *null hypothesis*, and provide tests based on each key accounting variable to the market price variable. Derived from the relative delta equations (3.18), (3.19), (3.20) and (3.21) the set of detailed null hypotheses are specified to determine the following hypothetical relationships:

$$H_{0:E}:\delta EM_{t \le 2004} = \delta EM_{t \ge 2005}$$
(3.23)

$$H_{0:A}:\delta AM_{t \le 2004} = \delta AM_{t \ge 2005}$$
(3.24)

$$H_{0:L}:\delta LM_{t \le 2004} = \delta LM_{t \ge 2005} \tag{3.25}$$

$$H_{0:I}:\delta IM_{t \le 2004} = \delta IM_{t \ge 2005} \tag{3.26}$$

Where for the banks in the UK banking sector:

 $H_{0:E}$ = total shareholders' equity to market price detailed null hypothesis

 $H_{0:A}$ = total assets to market price detailed null hypothesis

 $H_{0:L}$ = total liabilities to market price detailed null hypothesis

 $H_{0:I}$ = net income to market price detailed null hypothesis

 $\delta EM_{t \le 2004} = \delta EM_{t \ge 2005}$ = difference between the change in the total shareholders' equity variable and the market price variable for before and after the 2005 accounting change

 $\delta AM_{t \le 2004} = \delta AM_{t \ge 2005}$ = difference between the change in the total assets variable and the market price variable for before and after the 2005 accounting change $\delta LM_{t \le 2004} = \delta LM_{t \ge 2005}$ = difference between the change in the total liabilities variable and the market price variable for before and after the 2005 accounting change $\delta IM_{t \le 2004} = \delta IM_{t \ge 2005}$ = difference between the change in the net income variable and the market price variable for before and after the 2005 accounting change $t \le 2004$ = measure of time in years that specify the time interval 1994 to 2004 $t \ge 2005$ = measure of time in years that specify the time interval 2005 to 2008

3.4 Research Hypothesis and Detailed Research Hypotheses

From the tests applied, if it is revealed that the null hypothesis is falsified then it is concluded that an alternative hypothesis is accepted. This alternative hypothesis is termed the research hypothesis. This section presents this research hypothesis and also the detailed research hypotheses that correspond to the null hypothesis and the detailed null hypotheses presented in Section 3.3, respectively.

3.4.1 Research Hypothesis

This study's research hypothesis, H_a , that is aligned to the research question specified in Section 3.2 and directed by the null hypothesis specified in Section 3.3.1, takes the following form:

 H_a : Accounting quality in UK banks was affected after the adoption of the IFRS accounting standards in 2005. For the UK banks that adopted the new accounting standards in 2005, there was a significant difference between the change in accounting totals and the change in market price after 2005 when compared to before.

The research hypothesis, H_a , is specified to determine the following hypothetical relationship:

$$H_a: \delta_{t \le 2004} \neq \delta_{t \ge 2005} \tag{3.27}$$
Where for the banks in the UK banking sector:

 H_a = difference between the key accounting variables and market price variable research hypothesis

 $\delta_{t \le 2004} \neq \delta_{t \ge 2005}$ = difference between the change in key accounting variables and the change in the market price variable for before and after the 2005 accounting change $t \le 2004$ = measure of time in years that specify the time interval 1994 to 2004 $t \ge 2005$ = measure of time in years that specify the time interval 2005 to 2008

The research hypothesis, H_a , is the alternative hypothesis that is accepted if this research is unable to significantly verify its null hypothesis. The null hypothesis's level of verifiability is determined principally on evidence from tests conducted on the set of detailed null hypotheses.

The relationship between the research hypothesis to that of the null hypothesis is maintained between a set of *detailed research hypotheses* to the set of *detailed null hypotheses* that is specified in Section 3.3.2. The *detailed research hypotheses* specify a set of alternative hypotheses that are accepted if this research fails to verify its corresponding *detailed null hypotheses*.

3.4.2 Detailed Research Hypotheses

The detailed research hypotheses, directed by the set of detailed null hypotheses specified in Section 3.3.2, take the following form:

For UK banks, from before to after 2005, there was a significant change to the level of difference between:

- 1) $H_{a:E}$: the change in the total equity variable to the change in the market price variable.
- *2) H*_{a:A}: the change in the total assets variable to the change in the market price variable.
- *3) H*_{a:L}: the change in the total liabilities variable to the change in the market price variable.

4) $H_{a:I}$ the change in the net income variable to the change in the market price variable.

As stated earlier, and to emphasise, the set of detailed research hypotheses are the set of alternative hypotheses that are accepted if this research is unable to significantly verify its corresponding detailed null hypotheses specified in equations (3.23), (3.24), (3.25), and (3.26) presented in Section 3.3.2. Based on the relative delta relationships specified in equations (3.18), (3.19), (3.20) and (3.21) the set of detailed research hypotheses specify the following hypothetical relationships:

$$H_{a:E}:\delta EM_{t \le 2004} \neq \delta EM_{t \ge 2005} \tag{3.28}$$

$$H_{a:A}:\delta AM_{t \le 2004} \neq \delta AM_{t \ge 2005} \tag{3.29}$$

$$H_{a:L}:\delta LM_{t \le 2004} \neq \delta LM_{t \ge 2005}$$
(3.30)

$$H_{a:I}:\delta IM_{t \le 2004} \neq \delta IM_{t \ge 2005}$$
(3.31)

Where for the banks in the UK banking sector:

 $H_{a:E}$ = difference between the total shareholders' equity and market price detailed research hypothesis

 $H_{a:A}$ = difference between the total assets and market price detailed research hypothesis $H_{a:L}$ = difference between the total liabilities and market price detailed research hypothesis

 $H_{a:I}$ = difference between the net income and market price detailed research hypothesis $\delta EM_{t \le 2004} \neq \delta EM_{t \ge 2005}$ = difference between the change in the total shareholders' equity variable and the market price variable for before and after the 2005 accounting change

 $\delta AM_{t \le 2004} \neq \delta AM_{t \ge 2005}$ = difference between the change in the total assets variable and the market price variable for before and after the 2005 accounting change $\delta LM_{t \le 2004} \neq \delta LM_{t \ge 2005}$ = difference between the change in the total liabilities variable and the market price variable for before and after the 2005 accounting change $\delta IM_{t \le 2004} \neq \delta IM_{t \ge 2005}$ = difference between the change in the net income variable and the market price variable for before and after the 2005 accounting change $t \le 2004$ = measure of time in years that specify the time interval 1994 to 2004 $t \ge 2005$ = measure of time in years that specify the time interval 2005 to 2008

3.5 Main Variables

The main variables selected in this study to test its null hypothesis are presented in the following Table 3.1:

alternative names that are used to refer to the variable presented.					
Variable Description	Variable Name				
	Market Price also referred to as the: Adjusted Close Price				
Market variables	Change in Market Price also referred to as the: Market Price Return				
	Value-at-Risk				
	Total Shareholders' Equity also referred to as the: Total Equity				
variables	Total Assets				
	Total Liabilities				
	Net Income				

The table shows the main variables tested. The Variable Description column presents the description for the variables presented in the Variable Name column. The Variable Name column presents the variable names, and where specified,

The variables presented in Table 3.1 are detailed in Table 3.2. Table 3.2 specifies *additional* main variables that are applied in this research to test its null hypothesis. In addition, Table 3.2 presents the derivation relationships between these variables.

Table 3.2Main Variable Specifications

The table shows the specifications for the main variables tested. The Variable Name column specifies the descriptive name used to reference the variable. The Symbol column specifies the main symbol used for the variable. The Alternative Symbol column lists other symbols used to specify that same variable. The Derived From column presents the variables, denoted by the variable symbols, used to derive the specified variable. The Units column specifies the units for the referenced variable.

Variable Name	Symbol	Alternative Symbol	Derived From	Units
Adjusted Close Price	Р	-	-	GBPX [€]
Market Price Return (Change in Market Price)	М	-	Р	-
Market Price Value-at-Risk	V _{dM}	$V, V_{M,}$ VaR05	М	-
Total Equity	TE	_	-	GBP 1 MM [§]
Total Assets	TA	-	-	GBP 1 MM [§]
Total Liabilities	TL	-	-	GBP 1 MM [§]
Net Income	TI	-	-	GBP 1 MM [§]
General Accounting Variable	AC	-	-	-
Change in Total Equity	Ε	-	TE	-
Change in Total Assets	Α	-	ТА	-
Change in Total Liabilities	L	-	TL	-
Change in Net Income	Ι	-	TI	-
General Relative Delta	δ	-	-	-
Equity to Market Price Relative Delta	δΕΜ	E - M	E and M	-
Assets to Market Price Relative Delta	δΑΜ	A - M	A and M	-
Liabilities to Market Price Relative Delta	δLM	L - M	L and M	-
Net Income to Market Price Relative Delta	δΙΜ	I - M	I and M	-
General Accounting Value-at-Risk	V _{dAC}	V _{AC}	δAC and V_{dM}	-
Equity Value-at-Risk	V_{dE}	V_E	δEM and V_{dM}	-
Assets Value-at-Risk	V_{dA}	V_A	δAM and V_{dM}	-
Liabilities Value-at-Risk	V_{dL}	V_L	δLM and V_{dM}	-
Net Income Value-at-Risk	V_{dI}	V_I	δIM and V_{dM}	-
			δΕΜ	
Regulatory Relative Delta	ΔR	-	δΑΜ	%
			δLM δIM	

Table notes: [€] GBPX specifies the scale: 0.01 GBP (i.e. pence). [§] GPB 1 MM specifies the scale: 1,000,000 GBP.

The tests performed using the variables in Table 3.2 are presented in summary in Section 3.8 and specified in detail in Section 3.9 and Appendix M, Appendix N and Appendix O. From these tests, the variable relationships that are examined to improve evaluation for this research's null hypothesis are next emphasised by referencing the variables in Table 3.2.

- The Market Price Return variable, derived from the market price variable, is used to evaluate descriptive statistics for the market price return variable's distribution on a yearly basis from 1993 to 2009.
- The Value-at-Risk variable is applied and backtested to determine the frequency of breaches on a yearly basis from 1993 to 2009.
- The Market Price Return variable and the Value-at-Risk variable are tested with selected UK Gross Domestic Product and financial market indices; from 1994 to 2008 using descriptive statistics, correlation analysis and time series regression analysis.
- The Market Price Return variable is tested with the change in accounting total variables: Change in Total Equity, Change in Total Assets, Change in Total Liabilities, and Change in Net Income; from 1994 to 2008 using descriptive statistics, correlation analysis and cross-sectional and time series regression analysis.
- Material significance is evaluated from the Relative Delta Measure on a yearly basis from 1994 to 2008. That is, the change in the accounting total variables: Change in Total Equity, Change in Total Assets, Change in Total Liabilities and Change in Net Income; to the Change in the Market Price variable.
- The Relative Delta Measure is tested with the Value-at-Risk variable from 1994 to 2008 using descriptive statistics, correlation analysis and cross-sectional and time series regression analysis.

3.6 Sample Selection Criteria

To build corroborative evidence to verify the null hypothesis specified in Section 3.3.1, the null hypothesis itself and the *detailed null hypotheses*, specified in Section 3.3.2, are tested. To test these hypotheses a number of firms are selected to represent the population of interest. The sample of firms is selected for the *total* time series range specified in Section 3.7.7.

Guided by the null hypothesis specified in Section 3.3.1, the criteria for the sample selection include banks that operate in the financial sector. This selection criterion is justified, in that, this research examines how UK banks' accounting quality reacted to the 2005 accounting change. Therefore, it stands to reason that banks are selected, as they are this study's population of interest. In addition, it may be contended that banks are the primary institutions that originate and transact financial instruments and other financial instrument products (Bodie et al. 2002). Examination of financial statements prepared and presented by these firms should reflect how financial instruments and also other financial items, were reported during and after the transition event from the accounting standards accepted under UK GAAP to the IASB standards in 2005 (IASB 2009).

To avoid strong sample selection bias, as specified by Fama and French (1992), this research uses the following selection criteria to produce the sample of firms to survey:

- 1) The firm is classified by the LSE as operating within the banking industry.
- The firm's primary source of revenue is generated from financial instrument transactions.
- 3) The firm prepared and presented financial statements under UK GAAP until the year 2005, prior to the IASB accounting standards adoption.
- 4) The firm prepared and presented financial statements by adopting the IASB's IFRS accounting standards from the year 2005 to year 2008.
- 5) The firm has observable annually reported financial information that is original

i.e. not restated.

- 6) The firm has observable financial accounting information that is original and not restated from the year 2005 to 2008.
- The firm reports its annual (fiscal year) financial information for the 31st day in the month of December each year.
- The firm reports its interim financial information for the 30th day in the month of June each year.³⁴
- 9) The firm reports, or can be converted to report, financial statement element items: Shareholders' Equity, Total Assets, Total Liabilities, and Net Profit (Profit after tax); with the Great British Pound (GBP) as the base reporting currency.
- 10) The firm's shares are registered with the LSE.
- 11) The firm's share price, or market price, is recorded with GBP as the base reporting currency.
- 12) The firm has observable financial market information from the year 2005 to 2008.

3.6.1 Control Group Selection Criteria

The control group firms are selected based on the same characteristics as the sample firms, as specified in Section 3.6, however, with one important difference that relates to criterion 4. The important difference is that the control group firms maintained its accounting practice under the UK GAAP's accepted accounting standards, prior to, and after, the IASB accounting standards adoption event in 2005. This is opposed to the sample firms that changed reporting standards from those accepted under UK GAAP to the IASB standards after 2005.

³⁴ Interim information is not analysed in this thesis. However, criterion 8 is included for completeness.

The purpose of the control group is to provide a controlled test environment to help deduce the significance of the selected sample's material and statistical evidence for before and after the 2005 accounting change. It is considered that if the results for sample tests, for before and after this change, materially and statistically match the results produced from the control group, it would provide strong evidence that the observed effects are unlikely to be reactions directly attributable to the accounting change. If there is a lack of significance between the samples and the control group results, for before and after 2005, it may be contended that the results are related to the accounting change as opposed to other effects such as those arising from general economic or market conditions.

3.6.2 Sample and Control Group Selection

The firms selected for the sample are determined based on the criteria specified in Section 3.6. Emphasising criterion 4, the sample selected consists of firms that prepared and presented financial statements by adopting the IASB accounting standards from the year 2005 to year 2008.

As discussed in Section 3.6.1, the firms for the control group are selected on much the same basis as the sample firms, matching the criteria specified in Section 3.6. As stated, the control group does not observe criterion 4 specified in Section 3.6. The control group of firms selected consists of firms that prepared and presented financial statements under the UK GAAP accounting standards from the year 2005 to year 2008 without adopting the IASB accounting standards.

With regards to reporting dates, the sample selection criterion 7 specified in Section 3.6 states that firms selected in a sample must report annual financial information on the 31st day in the month of December each year. This criterion is adjusted, so that, if a sample firm's or a control group firm's annual reporting month is recorded between July and December, the financial data is categorised for that reporting year. If this firm's annual reporting month is recorded between January and June, the financial data is categorised for the previous reporting year.

3.6.3 Primary Sample Banks

The LSE, in March 2010, categorised five banks in the Financial Times Stock Exchange (FTSE) 100 Index in the banking institution category denoted as the *Industry sector: Banks*³⁵. The five banks are presented in Table 3.3.

1 adie 5.5 Banking Institutions Listed in the London Stock Exchange (L	Table 3.3	Banking Institutions	Listed in the I	London Stock Exchange ((LSE)
--	-----------	-----------------------------	-----------------	-------------------------	-------

The table shows the banks listed in the LSE's Industry Sector: Banks, as at March 2010. The Bank Name column presents the banks name. The LSE Symbol column presents the LSE ticker symbol for the bank presented in the Bank Name column.

Bank Name	LSE Symbol
HSBC HOLDINGS PLC	HSBA
BARCLAYS PLC	BARC
ROYAL BANK OF SCOTLAND GROUP PLC (THE)	RBS
LLOYDS BANKING GROUP PLC	LLOY
STANDARD CHARTERED PLC	STAN

The primary sample in Table 3.3 consists of banking firms that are listed in the LSE's FTSE 100 firms' index, referred to as the FTSE 100 Index, and FTSE 350 firms' index, referred to as the FTSE 350 Index.

3.6.4 Primary Sample Constraint

The number of primary sample banking institutions listed in the LSE numbered five firms, and as specified by Byrne (2002) represents the population of interest, for this study's examination.

The question may arise as to the credibility of results produced from a population comprising five banks.³⁶ However, it may be argued from the Berkowitz and O'Brien's

³⁵ The Primary Sample banks registered on the LSE's FTSE 100 *Industry sector: Banks* is located at: http://www.londonstockexchange.com [accessed 15th May 2012].

³⁶ This study expects to test its null hypothesis based on evidence from the population of five banks in Table 3.3, tested both at the sample and at the individual bank levels. It is expected that any significant effects from any one bank when tested at the sample level would become evident from the results produced from tests at the individual bank level.

(2002) sample of six banks, examined over a time period that is less than applied in this study, that it is possible to produce credible and robust results based on such a refined sample size. Although given this evidence, to increase the robustness of the results from the five primary sample banks, presented in Table 3.3, a secondary sample is drawn and tested from the LSE comprising banking related firms categorised within the financial services industry.

3.6.5 Secondary Sample

The primary sample is increased by searching for banking related firms listed in the LSE. The electronic information publishing firm, The Bureau van Dijk Electronic Publishing, referred to as BvDEP or BvD, is used to perform company searches for the secondary sample. The product utilised to search for firms for this sample was the BvDEP information database product Orbis (BvDEP 2010).

3.6.6 Primary Sample Firm Selection

It was stated in Section 3.6.3 that the primary banks are recognised as listed in the LSE. However, the BvDEP ORBIS database product is used to select formally the primary sample banks based on the selection criteria presented in Table 3.4. The result, based on the selected criteria, produced the five banks in the primary population presented in Table 3.3.

Table 3.4	Primary and	Secondary S	Sample Selectio	n Criteria
-----------	-------------	-------------	-----------------	------------

The table shows the sample selection criteria for the primary sample and the secondary sample, sourced from the Bureau van Dijk Electronic Publishing ORBIS database. The Criteria Description column presents the main database search criterion selected. The Criteria column presents the search criterion selected from the main criterion.

Criteria Description	Criteria
Type of Companies (Entities)	Banks
Main Exchange	London Stock Exchange
Accounting Template	Banks
Accounting Practice	IFRS

3.6.7 Secondary Sample Firm Selection

In addition to the formal selection of the primary banks listed in the LSE, the BvDEP ORBIS database product is used to select the secondary sample firms, based on the same selection criteria presented in Table 3.4. The result based on these criteria, in addition to the five primary banks, produced an additional 28 firms.

The search criteria presented in Table 3.4 recognises these 28 firms as banks. In addition, the results produced by the BvDEP ORBIS database product reference these firms to be recognised by the EC (NACE 2008) as those related to banking activities, in particular financial and insurance activities. However, the LSE categorises these firms as either financial services firms or equity investment instrument firms. To maintain a general naming convention, firms that are produced in addition to the five primary sample banks, when applying a search criteria such as that presented in Table 3.4, are referred to in this study as both banking related firms or banking related financial services firms.

3.6.8 Sample Firms Surveyed

The total number of firms selected as the sample to be surveyed totaled 33, and comprises the five banking firms selected for the primary sample, and 28 firms selected for the secondary sample.

The 33 firms surveyed, produced from the selection criteria presented in Table 3.4, are presented in Table 3.6. After quality assurance, 16 firms are selected to test the study's null hypothesis. The 17 firms were excluded after refining due to missing data. These firms were from the secondary sample grouping.

The firms selected for the primary and secondary sample are the five banks that make up the primary sample, and the 11 banking related financial services firms that make up the secondary sample. To denote the selected firms sample categorisation each firm is labeled as either 'Primary' or 'Secondary' in the Sample column in Table 3.6. The accounting totals and corresponding market data collected for the primary sample and the secondary sample are presented in Appendix W.

3.6.9 Control Group Firm Selection

The BvDEP ORBIS database is also used to select firms to provide a control group. Table 3.5 presents the selection criteria used to select the control group of firms.

The difference with the selection criteria for the control group compared to the sample firms, as stated earlier, is that the control group of firms did not adopt the IASB accounting standards. Instead, during the 2005 adoption event, the control group prepared and presented financial statements in accordance with the UK GAAP's accepted accounting standards.

The selection criteria presented in Table 3.5 produced 20 firms. The 20 firms selected for the control group are presented in Table 3.7.

Table 3.5	Control Group Selection Criteria
-----------	---

The table shows the Control group sample selection criteria, sourced from the Bureau van Dijk Electronic Publishing ORBIS database. The Criteria Description column presents the main database search criterion selected. The Criteria column presents the search criterion selected from the main criterion.

Criteria Description	Criteria
Type of Companies (Entities)	Banks
Main Exchange	London Stock Exchange
Accounting Template	Banks
Accounting Practice	Local GAAP

After quality assurance of the 20 control group firms, 12 firms are selected to assist with the null hypothesis analysis. The eight firms excluded from the 20 were due to missing data. The firms chosen to represent the control group are labeled 'Control' in the Group column presented in Table 3.7. The accounting totals and corresponding market data collected for the control group firms are presented in Appendix X.

Table 3.6Primary and Secondary Sample Firms

The table shows the Primary sample and the Secondary sample firms selected from the Bureau van Dijk Electronic Publishing ORBIS database. The Firm Name column presents the name of the sample firm. The ISIN (International Securities Identification Number) column and the Ticker Symbol column present unique identifiers for the respective firms presented in the Firm Name column. The Accounting Standards column presents the accounting standard applied by the firm from 1st January 2005, where IASB refers to the International Accounting Standard Board's IFRS accounting standards. The Main Exchange column presents the firm's LSE trading platform; the trading platforms being: SETS (Stock Exchange Electronic Trading Service), and SEAQ (Stock Exchange Automated Quotations). The FTSE Index column presents the firm's LSE index listing. The Sample column presents the firm's sample category applied in this study.

Firm Name	ISIN	Ticker Symbol	Accounting Standard	Main Exchange	FTSE Index	Sample
HSBC HOLDINGS PLC	GB0005405286	HSBA	IASB	London Stock Exchange (SETS)	FTSE 100 and FTSE 350	Primary
BARCLAYS PLC	GB0031348658	BARC	IASB	London Stock Exchange (SETS)	FTSE 100 and FTSE 350	Primary
ROYAL BANK OF SCOTLAND GROUP PLC (THE)	GB0007547838	RBS	IASB	London Stock Exchange (SETS)	FTSE 100 and FTSE 350	Primary
LLOYDS BANKING GROUP PLC	GB0008706128	LLOY	IASB	London Stock Exchange (SETS)	FTSE 100 and FTSE 350	Primary
STANDARD CHARTERED PLC	GB0004082847	STAN	IASB	London Stock Exchange (SETS)	FTSE 100 and FTSE 350	Primary
MAN GROUP PLC	GB00B28KQ186	EMG	IASB	London Stock Exchange (SETS)	FTSE 350	
ICAP PLC	GB0033872168	IAP	IASB	London Stock Exchange (SETS)	FTSE 350	
CATTLES PLC	GB0001803666	СТТ	IASB	London Stock Exchange (SETS)		
INVESTEC PLC	GB00B17BBQ50	INVP	IASB	London Stock Exchange (SETS)	FTSE 350	

Table 3.6 (Continued)							
Firm Name	ISIN	Ticker Symbol	Accounting Standard	Main Exchange	FTSE Index	Sample	
SCHRODERS PLC	GB0002405495	SDR	IASB	London Stock Exchange (SETS)	FTSE 350	Secondary	
PROVIDENT FINANCIAL PLC	GB00B1Z4ST84	PFG	IASB	London Stock Exchange (SETS)	FTSE 350	Secondary	
CLOSE BROTHERS GROUP PLC	GB0007668071	CBG	IASB	London Stock Exchange (SETS)	FTSE 350	Secondary	
INTERNATIONAL PERSONAL FINANCE PLC	GB00B1YKG049	IPF	IASB	London Stock Exchange (SETS)	FTSE 350		
ABERDEEN ASSET MANAGEMENT PLC	GB0000031285	ADN	IASB	London Stock Exchange (SETS)	FTSE 350	Secondary	
INTERMEDIATE CAPITAL GROUP PLC	GB0004564430	ICP	IASB	London Stock Exchange (SETS)	FTSE 350		
BREWIN DOLPHIN HOLDINGS PLC	GB0001765816	BRW	IASB	London Stock Exchange (SETS)	FTSE 350		
PARAGON GROUP OF COMPANIES PLC	GB00B2NGPM57	PAG	IASB	London Stock Exchange (SETS)	FTSE 350	Secondary	
RATHBONE BROTHERS PLC	GB0002148343	RAT	IASB	London Stock Exchange (SETS)	FTSE 350	Secondary	
EVOLUTION GROUP PLC (THE)	GB0030221864	EVG	IASB	London Stock Exchange (SETS)	FTSE 350		
ARBUTHNOT BANKING GROUP PLC	GB0007922338	ARBB	IASB	London Stock Exchange (SEAQ)	FTSE AIM All- Share Constituents		
LONDON CAPITAL GROUP HOLDINGS PLC	GB00B0RHGY93	LCG	IASB	London Stock Exchange (SEAQ)	FTSE AIM All- Share Constituents		

Table 3.6 (Continued)							
Firm Name	ISIN	Ticker Symbol	Accounting Standard	Main Exchange	FTSE Index	Sample	
EUROPEAN ISLAMIC INVESTMENT BANK PLC	GB00B126GW60	EIIB	IASB	London Stock Exchange (SETS)	FTSE AIM All- Share Constituents		
ISLAMIC BANK OF BRITAIN PLC	GB00B02KNV97	IBB	IASB	London Stock Exchange (SETS)	FTSE AIM All- Share Constituents		
JUPITER PRIMADONA GROWTH TRUST PLC	GB0007033763	JPG	IASB	London Stock Exchange (SEAQ)			
TULLETT PREBON PLC	GB00B1H0DZ51	TLPR	IASB	London Stock Exchange (SETS)	FTSE 350		
POLAR CAPITAL TECHNOLOGY TRUST PLC	GB0004220025	РСТ	IASB	London Stock Exchange (SETS)			
DUNEDIN ENTERPRISE INVESTMENT TRUST PLC	GB0005776561	DNE	IASB	London Stock Exchange (SETS)			
ELECTRA PRIVATE EQUITY PLC	GB0003085445	ELTA	IASB	London Stock Exchange (SETS)		Secondary	
BANKERS INVESTMENT TRUST PLC	GB0000767003	BNKR	IASB	London Stock Exchange (SETS)		Secondary	
WITAN INVESTMENT TRUST PLC	GB0009744060	WTAN	IASB	London Stock Exchange (SETS)		Secondary	
RIT CAPITAL PARTNERS PLC	GB0007366395	RCP	IASB	London Stock Exchange (SETS)		Secondary	
ALLIANCE TRUST PLC	GB00B11V7W98	ATST	IASB	London Stock Exchange (SETS)		Secondary	
3I GROUP PLC	GB00B1YW4409	III	IASB	London Stock Exchange (SETS)	FTSE 350		

Table note: The information in this table is as of 10th December 2009.

Table 3.7Control Group Firms

The table shows the Control Group firms selected from the Bureau van Dijk Electronic Publishing ORBIS database. The Firm Name column presents the name of the Control group firm. The ISIN (International Securities Identification Number) column and the Ticker Symbol column present unique identifiers for the respective firms presented in the Firm Name column. The Accounting Standards column presents the accounting standards applied by the firm from 1st January 2005, where UK GAAP refers to the standards accepted under the United Kingdom Generally Accepted Accounting Practice. The Main Exchange column presents the firm's LSE trading platform; the trading platforms being: SETS (Stock Exchange Electronic Trading Service), and SEAQ (Stock Exchange Automated Quotations). The FTSE Index column presents the firm's LSE index listing. The Group column indicates the firm's category applied in this study.

ISIN	Ticker Symbol	Accounting Standard	Main Exchange	Group
GB0000485838	BGFD	UK GAAP	London Stock Exchange (SETS)	
GB0006667470	PHI	UK GAAP	London Stock Exchange (SEAQ)	Control
GB0006450703	NVT	UK GAAP	London Stock Exchange (SEAQ)	
GB0031152027	NTN	UK GAAP	London Stock Exchange (SEAQ)	
GB0005356430	NTV	UK GAAP	London Stock Exchange (SEAQ)	
GB0000706274	BGS	UK GAAP	London Stock Exchange (SEAQ)	Control
GB0002093689	NNA	UK GAAP	London Stock Exchange (SEAQ)	
GB00B08S4K30	NRI	UK GAAP	London Stock Exchange (SEAQ)	
	ISIN GB0000485838 GB0006667470 GB0006450703 GB00031152027 GB0005356430 GB0000706274 GB0002093689 GB00B08S4K30	ISIN Ticker Symbol GB0000485838 BGFD GB0006667470 PHI GB0006450703 NVT GB00031152027 NTN GB0005356430 NTV GB0000706274 BGS GB0002093689 NNA GB00B08S4K30 NRI	ISINTicker SymbolAccounting StandardGB0000485838BGFDUK GAAPGB0006667470PHIUK GAAPGB0006450703NVTUK GAAPGB0031152027NTNUK GAAPGB0005356430NTVUK GAAPGB0000706274BGSUK GAAPGB0002093689NNAUK GAAPGB00B08S4K30NRIUK GAAP	ISINTicker SymbolAccounting StandardMain ExchangeGB0000485838BGFDUK GAAPLondon Stock Exchange (SETS)GB0006667470PHIUK GAAPLondon Stock Exchange (SEAQ)GB0006450703NVTUK GAAPLondon Stock Exchange (SEAQ)GB0031152027NTNUK GAAPLondon Stock Exchange (SEAQ)GB0005356430NTVUK GAAPLondon Stock Exchange (SEAQ)GB0000706274BGSUK GAAPLondon Stock Exchange (SEAQ)GB0002093689NNAUK GAAPLondon Stock Exchange (SEAQ)GB00B08S4K30NRIUK GAAPLondon Stock Exchange (SEAQ)

Firm Name	ISIN	Ticker Symbol	Accounting Standard	Main Exchange	Group
MID WYND INTERNATIONAL INVESTMENT	GB0005893838	MWY	UK GAAP	London Stock Exchange (SEAQ)	
DUNEDIN SMALLER COMPANIES INVESTMENT TRUST	GB00B1GCL258	DNDL	UK GAAP	London Stock Exchange (SETS)	Control
MURRAY INTERNATIONAL TRUST PLC	GB0006111909	MYI	UK GAAP	London Stock Exchange (SETS)	Control
EDINBURGH WORLDWIDE INVESTMENT TRUST PLC	GB0002916335	EWI	UK GAAP	London Stock Exchange (SETS)	
THROGMORTON TRUST PLC	GB0008910555	THRG	UK GAAP	London Stock Exchange (SETS)	Control
BRITISH ASSETS TRUST PLC	GB0001297562	BSET	UK GAAP	London Stock Exchange (SETS)	Control
EDINBURGH INVESTMENT TRUST PLC (THE)	GB0003052338	EDIN	UK GAAP	London Stock Exchange (SETS)	Control
SCOTTISH INVESTMENT TRUST PLC	GB0007826091	SCIN	UK GAAP	London Stock Exchange (SETS)	Control
MONKS INVESTMENT TRUST PLC	GB0030517261	MNKS	UK GAAP	London Stock Exchange (SETS)	Control
MERCANTILE INVESTMENT TRUST PLC (THE)	GB0005794036	MRC	UK GAAP	London Stock Exchange (SETS)	Control
FOREIGN & COLONIAL INVESTMENT TRUST PLC (THE)	GB0003466074	FRCL	UK GAAP	London Stock Exchange (SETS)	Control
SCOTTISH MORTGAGE INVESTMENT TRUST PLC	GB0007838849	SMT	UK GAAP	London Stock Exchange (SETS)	Control

Table note: The information in this table is as of 10th December 2009.

3.7 Information and Data Collection

Information and data collection methods and data sources used in this study are presented in this section. Information and data are gathered in the subject areas of: financial instruments, accounting standards and methods, risk monitoring and management methods, and the financial regulatory framework and the technology framework. In addition, this section presents the variables and time ranges selected for data gathering. The time ranges applied for data analysis is also presented.

3.7.1 Information Collection

The following sources were used for information collection purposes in this study:

- Current and past financial statement reports.
- Electronic databases from data providers.
- Published articles in the press and relevant magazines.
- On-line published articles and relevant magazines from reputable authorities.
- Articles, papers and books in the following areas:
 - Business and Finance
 - Accounting
 - Risk management
 - Regulation, supervision and governance
 - Technology
 - Science, mathematics and statistics
 - Other relevant articles and papers and books

3.7.2 Market and Accounting Data Collection

The following sources were used for data collection purposes in this study:

- Current and past financial information from financial statement reports.
- Financial information and data from regulatory authorities.
- Financial information from exchanges.
- Financial and market data from data providers.
- Institution information from regulatory authorities.

- Institution information from *information and market data* providers.
- Institution information from exchanges.

3.7.3 Data Sources

Table 3.8 presents a summary of the categories for the data collected with the corresponding data sources.

Table 3.8 Data Collection: Data Description and Data Sources

The table shows the types of data collected and the respective data sources. The Data Description column presents the categories of data collected. The Data Sources column presents the source used to collect data for the respective data categories.

Data Description	Data Source
General Economic Indicators	Office for National Statistics ^a
	1) Bank of England ^b
General Market Indicators	2) Thomson One Banker product
Market data	Thomson Reuters Datastream database product
Accounting data	1) Thomson Reuters World Source database product
	2) Orbis database product
	3) Financial Statements
Table motor:	

Table notes:

^aData was sourced from the Office for National Statistics located at www.statistics.gov.uk, and was sourced from the 28th to 31st March 2010. ^bData was sourced from the Bank of England located at www.bankofengland.co.uk, and was sourced from the 28th to 31st March 2010.

3.7.4 GDP Indicators Variable Data Collection

Determined by Ang, Piazzesi and Wei (2006) a measure of economic growth is directly related to the measure of the Gross Domestic Product (GDP) in the economy of interest. In order to establish the general UK economic conditions from 1994 to 2008, the All Production and Business Services GDP sector data sets are collected on a quarterly basis for this time period. See Section 3.8 for how the GDP variable is applied to test this study's null hypothesis.

3.7.5 General Market Indicator Variables Data Collection

To determine the general market conditions from 1994 to 2008 data is collected for this time period for selected market indices, interest rate and currency indices variables.

To provide analysis of general stock market conditions from 1992 to 2008 data is collected on a daily basis for the London Stock Exchange Indices: FTSE 350, FTSE 100, FTSE ALL shares and some selected indices from around the world.

To provide analysis of the general interest rate market condition from 1992 to 2008 the London Interbank Offered Rate (LIBOR) rates data and the UK government bond, referred to as the UK Gilt government bond, yield data are collected on a daily basis. The LIBOR rates are chosen to provide an analysis of the UK financial markets' response to short-term interest rates, and the UK Government bond yields are chosen to provide the markets' response to longer-term interest rates.

To provide analysis of general currency market conditions from 1992 to 2008 data is collected on a daily basis for the currency pairs: European EURO and GBP, United States Dollar (USD) and GBP, and Japanese Yen and GBP.

See Section 3.8 for how the market indicator variables are applied to test this study's null hypothesis.

3.7.6 Market and Accounting Variables Data Collection

Market data is collected to analyse the market data variations from 1992 to 2009. The primary market variable of interest in this study is the market price variable. However, with the aim to provide additional market variable analysis, data are also collected for the market value and the book-to-market ratio variables. With a view to maintain data integrity from the source, the book-to-market ratio variable data is collected and also presented in this study as its mathematical reciprocal, termed the *market-to-book* ratio.

For each firm in the samples and the control group, market data is collected from 1992 to 2009 on a daily basis for the following financial market variables:

Market Price Market Value *Market-to-Book* Volume Common Shares Outstanding

Accounting data is collected to analyse the accounting data variations from 1992 to 2008. For each firm in the samples and the control group, accounting data is collected for this time period on a yearly basis for the following accounting variables:

Total Shareholders' Equity Total Assets Total Liabilities Net Income

3.7.7 Time Series Interval Selection

Table 3.9 presents the 1992 to 2009 time range applied in this research. In addition to yearly time periods that are in this time range, three time interval ranges are selected for analysis. Table 3.9 also presents the overlapping years that correspond to the time interval ranges applied in this study.

Table 3.9Time Series Ranges

The table shows the years and the time ranges tested. The main Time Ranges column presents the full 1992 to 2009 time range tested. The Total column presents the 1994 to 2008 time range, indicated by the bordered years. The Range 1 column presents the 1994 to 2004 time period applied to test for before the 2005 accounting change, indicated by the bordered years The Range 2 column presents the 1994 to 2007 time period, indicated by the bordered years.

Time Ranges				
Total	Range 1	Range 2		
1992	1992	1992		
1993	1993	1993		
1994	1994	1994		
1995	1995	1995		
1996	1996	1996		
1997	1997	1997		
1998	1998	1998		
1999	1999	1999		
2000	2000	2000		
2001	2001	2001		
2002	2002	2002		
2003	2003	2003		
2004	2004	2004		
2005	2005	2005		
2006	2006	2006		
2007	2007	2007		
2008	2008	2008		
2009	2009	2009		

Table note: The highlighted row identifies the samples IFRS accounting standards adoption year.

The criteria for the selection of the three time ranges specified in Table 3.9 are to provide a set comparative time series regressions that allow the analysis of selected variable reactions, for before and after the 2005 accounting change. The three time

ranges presented in Table 3.9, that specify a total time period (1994 to 2008) and the two sub-time periods that are applied in time series regressions, are presented in summary in Table 3.10.

The table shows the time ranges tested. The Time Period column presents the time range descriptions. The Years: t to T column presents the time period tested in years. The Number of Years Observed, T column presents the count of the years observed in each time range.			
Time Period	Years: t to T ^a	Number of Observed Years, <i>T</i>	
Total Time Period	1994 to 2008	15	
Sub-Time Period 1	1994 to 2004	11	

Table 3.10Time Series Periods

Table notes: ^a The starting year and maximum ending year for the time series periods are from 1994 to 2008. However, analysis of descriptive statistics on a year-by-year basis is performed from years 1992 to 2009.

1994 to 2007

14

From Table 3.10 the total time period, from 1994 to 2008, is selected to provide regressions that analyse selected variable relationships for the time period that *includes* the 2005 accounting change year. The first sub-time period 1, from 1994 to 2004, provides regressions that analyse variable relationships *just prior* to the 2005 change. The second sub-time period 2, from 1994 to 2007, provides regressions that analyse variable relationships *just prior* to the 2005 change. The second sub-time period 2, from 1994 to 2007, provides regressions that analyse variable relationships for the time period that includes the 2005 accounting change year. However, sub-time period 2 provides a control for effects observed during the 2008 year by excluding that year (see Section 1.2).

3.8 Introduction to the Null Hypothesis Tests and Analysis

Sub-Time Period 2

This thesis discusses the theory that improved accounting quality is associated with firms that adopt the IFRS accounting standards (see Chapter 2, Section 2.2 and 2.3). This theory is supported by the EC (2002) and IASB (2011) and is empirically tested by Platikanova and Nobes (2006), Paananen and Lin (2009) and Morais and Curto (2008) for sample firms in Europe after the Europe-wide first-time adoption of the IFRS accounting standards in 2005. This study examines this theory from the context of the

research question presented in Section 3.2 that asks, 'Does accounting quality improve for UK banks that adopted the IFRS accounting standards in 2005?'.

To determine the level of evidence to support this theory, the null hypothesis presented in Section 3.3.1 and specified by equation (3.22) is examined using the tests introduced in this section. To examine the level of verifiability (or conversely, the level of *falsifiability*) for the null hypothesis, the tests introduced in this section are aligned to determine conclusions based on a series of testable statements. This series of testable statements are presented in Table 5.1 to Table 5.11 under the *Statement Tested* column. For the UK banks after 2005 compared to before, and in general when compared to the Control group, it is ideally expected that the results from these tests to show: the market price returns increased; market price volatility and Value-at-Risk levels decreased; accounting quality, measured using the relative delta measure, improved and thus the level of exposure to accounting quality risk decreased (see Section 2.9.3); the relative delta measure does not exhibit explanatory power for Value-at-Risk; and, the level of capital that would need to be provisioned for the Basel minimum capital requirement reduced.

The specific approach that this study applies to test its null hypothesis is to conduct tests that support directly either the null hypothesis itself, specified in equation (3.22), or the null hypotheses specified in equations (3.23) to (3.26). The evidence from tests directed to evaluate these detailed null hypotheses is analysed to determine if their results add strength to verify or falsify the null hypothesis.

This section next introduces the null hypothesis tests conducted. Appendix M, Appendix N and Appendix O provide details for these tests, together with Section 3.9 that provides details of the tests conducted using the change in the accounting total variables and the change in the market price variable.

Market Price Return Distribution and Variance-Covariance Value-at-Risk Analysis

The descriptive statistics detailed in Appendix M are calculated for 300 historical 1-day market price returns to report distribution mean, standard deviation of distribution means, skewness, kurtosis, and excess kurtosis based on time series panel data. For the

same data, the variance-covariance Value-at-Risk variable calculated using the model detailed in Appendix N.3 is presented and analysed.

This analysis is conducted to determine the levels of market price returns, market price return volatility and the variance-covariance Value-at-Risk for before and after 2005.

Value-at-Risk Analysis

The market price return Value-at-Risk measure, detailed in Appendix N, for the 1-day, 250-day and 500-day time horizons at the 95% confidence level is analysed. The Value-at-Risk models analysed are the:

- 1) variance-covariance
- 2) historical simulation
- 3) Monte Carlo simulation

The variance-covariance model is detailed in Appendix N.3, the historical simulation model, referred to also as the historical model, is detailed in Appendix N.4 and the Monte Carlo simulation model is detailed in Appendix N.5.

The three Value-at-Risk models are tested to determine the number of times the market price return variable was breached. That is, the number of times the actual market price return variable became less than the Value-at-Risk measure before or on the day that specifies the Value-at-Risk time-horizon. The models are tested for breaches by applying the backtesting approach specified by JPMorgan Chase and Reuters (1996) and Hull (2009) and presented in Appendix N.7. The specific approaches applied in this study are the *in-time horizon backtest* as described in Appendix N.7.3.

These analyses are conducted to determine the levels of Value-at-Risk for before and after 2005.

Historical Value-at-Risk Actual for the 1-Day, 250-Day and 500-Day Time Horizons at the 95% Confidence Level Analysis

To determine the Historical Value-at-Risk Actual variable's time horizon characteristics, the 250-day Historical Value-at-Risk Actual measure specified in Appendix N.8, is analysed. This measure is analysed with the 1-day and 500-day Historical Value-at-Risk Actual measurements. The Historical Value-at-Risk Actual variable's time horizon characteristics are analysed using the descriptive statistics specified in Appendix O.1.

The Historical Value-at-Risk Actual measurements are also graphically plotted with the sample mean market price returns and the standard deviation of sample mean returns. The plots are made for the 1-day and 250-day Historical Value-at-Risk Actual based on 300 and 800 historical returns, and the 500-day measure based on 500 and 800 historical returns.

Regressions for Historical Value-at-Risk Actual at the 99% and 99.9% Confidence Levels and the 95% Confidence Level Analysis

To determine the relation between the Historical Value-at-Risk Actual variable, for the 99% and 99.9% confidence levels and the 95% level, the cross-sectional and time series regressions specified in Appendix O.2 are performed. The Historical Value-at-Risk Actual variable tested is measured using 300 sequential historical observations of 250-day market price returns.

Market Price Returns and Historical Value-at-Risk Analysis

To determine the relation between the market price return variable and the Historical Value-at-Risk Actual variable, the cross-sectional and time series regressions specified in Appendix O.3 are performed. The Historical Value-at-Risk Actual variable tested is measured using 300 sequential historical observations of 250-day market price returns.

GDP and Market Indices Analysis

To test the relation before and after the 2005 accounting change between the GDP and the market price variable, and Market Indices and the market price variable, the time series regressions specified in Appendix O.4 are performed.

To determine the relation before and after the 2005 accounting change between the GDP Value-at-Risk and the market price Value-at-Risk variable, and Market Indices Value-at-Risk and the market price Value-at-Risk variable, the time series regressions specified in Appendix O.5 are performed.

Baseline Market Value and Book-to-Market Ratio Analysis

The Fama and French (2008) baseline regression that tests the relation between the market value and the book-to-market ratio to that of expected returns is tested using the regression specified in Appendix O.5, equation (O.15). The regression applies the t-statistic measure prescribed in Fama and MacBeth (1973) and specified in Appendix O.6, equation (O.16).

Total Variables and Value-at-Risk Analysis

To determine the relationship between the book-to-market ratio, market price, market value, volume of shares traded, number of shares outstanding and key accounting total variables (termed collectively as the *total variables*) to the Value-at-Risk variable, the descriptive statistic analysis specified in Appendix O.7 is applied.

To test the correlation between these total variables and the Value-at-Risk variable, the Pearson correlation analysis specified in Appendix O.8 is applied.

Focusing on the accounting total variables, the results from these variables are applied to provide a measure of the levels of material significance before and after the 2005 accounting change. The results from the accounting total analysis are applied to test directly the detailed null hypotheses specified in Section 3.3.2. The results from the other total variables are analysed to determine if they provide any significant evidence to test the null hypothesis.

Change in Total Variables and Value-at-Risk Analysis

To determine the relationship between the yearly changes for the book-to-market ratio, market price (that is the yearly market price return), market value, the number of share outstanding, and key accounting total variables to the Value-at-Risk variable, the descriptive statistic analysis specified in Appendix O.9 is applied.

To test the correlation between the yearly change variables and the Value-at-Risk variable the Pearson correlation analysis specified in Appendix O.10 is applied.

Market Value and Market Price Return Analysis

To test the relation before and after the 2005 accounting change between yearly changes in the market value variable to the market price variable (market price return variable) the cross-sectional and time series regressions specified in Appendix O.11 are performed.

Change in Key Accounting Totals and Market Price Return Analysis

To test the relation before and after the 2005 accounting change between the change in the key accounting variables: total equity, total assets, total liabilities and net income, and the market price return variable, the time series regressions specified in Section 3.9.1 and the cross-sectional regression specified in Appendix O.12 are performed.

The results from these tests provide a measure of the levels of statistical significance between the change in the accounting variables and the market price return variable for before and after the 2005 change. The results from this analysis are applied to test directly the detailed null hypotheses introduced in Section 3.3.2.

Change in Key Accounting Totals and Market Price Return Analysis – The Relative Delta Measure

To measure the difference between the change in the accounting variable and the market price return variable, that is to measure the *accounting to market price relative delta*, the approach specified in Section 3.9.2 is applied.

The accounting to market price relative delta measure, specified in Section 3.9.2, provides a measure of material significance that is applied to test directly the detailed null hypotheses introduced in Section 3.3.2.

Relative Delta to Value-at-Risk Analysis

To test the relation between the change in key accounting and market price return variables to the Value-at-Risk variable for before and after the 2005 accounting change,

the time series regressions specified in Section 3.9.3 and cross-sectional regressions specified in Appendix O.13 are performed.

These tests provide a measure of the levels of statistical significance between the *accounting to market price relative delta* variable and the *Value-at-Risk* variable for before and after the accounting change. The results from this analysis are applied to test directly the detailed null hypotheses introduced in Section 3.3.2.

Accounting Value-at-Risk Analysis

To estimate the Value-at-Risk for the key accounting variables, the measure developed in Section 3.9.4 is applied.

The accounting Value-at-Risk together with the market price Value-at-Risk provides a measure of the difference between the Value-at-Risk levels before and after the 2005 accounting change. The differences in the Value-at-Risk measures are applied to test *indirectly* the detailed null hypotheses introduced in Section 3.3.2.

Accounting Value-at-Risk, Market Price Return Value-at-Risk and Relative Delta Analysis

To determine the relationship between the measures accounting Value-at-Risk, market price return Value-at-Risk and the accounting to market price relative delta, the relationship specified in Section 3.9.5 is applied. From this relationship it is determined that the difference between the accounting Value-at-Risk and the market price return Value-at-Risk relates directly to the relative delta measure.

Relative Delta and Regulatory Capital Analysis

To estimate the change to the 8% minimum regulatory capital requirement specified by the Basel Capital Accord (Basel 2011) a threshold based capital adjustment measure is introduced in Section 3.9.6. This adjustment is calculated by applying the regulatory relative delta measure proposed in Section 3.9.6.

This adjustment is applied to determine its effect on regulatory capital for before and after the 2005 accounting change. The purpose of this test is to provide direct evidence

to verify the null hypothesis specified in equation (3.22).

UK Banks Relative Delta and Value-at-Risk Analysis

To test the relation between the accounting to market price relative delta variable and the Value-at-Risk variable for before and after the 2005 accounting change for each bank in the UK banking sector, the time series regression specified in Section 3.9.7 is performed.

This test provides for each UK bank, a measure of the levels of statistical significance between the accounting to market price relative delta variables and the Value-at-Risk variable for before and after the accounting change. The results from this analysis are applied to test directly the detailed null hypotheses introduced in Section 3.3.2.

3.9 Null Hypothesis Test Details using the Change in Key Accounting Totals and Market Price Return

The previous section, Section 3.8, introduced the tests applied in this study to examine its research question, presented in Section 3.2, by testing the null hypothesis presented in Section 3.3.1. This section presents details of the specific null hypothesis tests that are introduced in Section 3.8 that determine the level of verifiability (or conversely, the level of falsifiability) by examining the relationship between the change in key accounting variables and the change in the market price variable. This change in the market price variable is also termed the market price return variable.

3.9.1 Time Series Regressions for Change in Key Accounting Totals and Market Price Return

This section presents details of the test introduced in Section 3.8 with the title: Change in Key Accounting Totals and Market Price Return Analysis.

The time series regression applied to determine the relationship between the change in key accounting variables³⁷ and the market price return variable is specified by the following:

³⁷ The change in accounting variables is calculated using the approach detailed in Section 3.9.2.

$$d\overline{M}_{N:t-1,t} = a_T + b_T d\overline{AC}_{N:t-1,t} + e_{N:t-1,t}$$
(3.32)

Where:

 $d\overline{M}_{N:t-1,t}$ = mean market price return at time t for N sample firms

 $d\overline{AC}_{N:t-1,t}$ = average change in key accounting total variable at time *t* for *N* sample firms

 b_T = regression slope

 a_T = regression intercept

 $e_{N:t-1,t}$ = error term representing the difference between the actual dependent variable values to the calculated dependent variable values using the regression equation

N = average number of sample firms

T = maximum year count at the sample level

t =time series index: t = 1, 2, ..., T

This regression approach³⁸ is presented in a generalised form for each key accounting total and market price return in Appendix F.2.

3.9.2 Relative Delta and Change in Key Accounting Totals and Market Price Return

This section presents details of the test introduced in Section 3.8 with the title: Change in Key Accounting Totals and Market Price Return Analysis – The Relative Delta Measure.

The log change equation (M.31), or for negative variable values the relative change formula specified in equation (M.30) in Appendix M, is applied to the accounting variables to measure the level of change for the time period: t - 1 to t. The formula applied at the firm level to calculate the change in *positive variable values* is given by:

³⁸ The regression form specified in equation (3.32) may also be specified using the regression model form: $dM_{N,t} = \beta_0 + \beta_1 dAC_{N,t} + e_{N,t}$ – where: $dM_{N,t}$ is equivalent to $d\overline{M}_{N:t-1,t}$ = the mean market price return at time t for N sample firms; β_0 is equivalent to a_T = the regression intercept; β_1 is equivalent to b_T = the regression slope, and $e_{N,t}$ is equivalent to e_T = the regression error. The regressions specified in this study may be interpreted in a similar way using either form.

$$dAC_{t-1,t} = ln\left(\frac{AC_t}{AC_{t-1}}\right)$$
(3.33)

Where:

 $dAC_{t-1,t} = \log$ change in the accounting total variable from time t - 1 to t ln = natural logarithm to the base e $AC_t =$ accounting total variable value at time t $AC_{t-1} =$ accounting total variable value at time t - 1t = time index

For negative variable values, the following formula is applied:

$$dAC_{t-1,t} = \frac{AC_t - AC_{t-1}}{AC_{t-1}}$$
(3.34)

Where:

 $dAC_{t-1,t}$ = relative change in the accounting total variable from time t - 1 to t AC_t = accounting total variable value at time t AC_{t-1} = accounting total variable value at time t - 1t = time index

The log change equation (M.31) in Appendix M to determine the market price change or the market price return is applied using the following formula:

$$dM_{t-1,t} = ln\left(\frac{P_t}{P_{t-1}}\right)$$
 (3.35)

Where:

 $dM_{t-1,t} = \log$ market price change or market price return from time t - 1 to t

ln = natural logarithm to the base e

 P_t = market price at time t

 P_{t-1} = market price at time t - 1

t = time index

From equation (3.33), or equation (3.34) where negative total values are present, and equation (3.35), this study specifies the measure termed the relative delta. This relative delta measure, specified in Section 3.2.4, measures the difference between the change in

the accounting total variables for a given time period and the change in the market price variable for the same time period. As stated in Section 2.7, extended from the Fama and French (2008) time dependent treatment of the book-to-market ratio, the formula for the accounting to market price relative delta at the firm level is given by:

$$\delta_{i,t-1,t} = dAC_{i,t-1,t} - dM_{i,t-1,t} \tag{3.36}$$

Where:

 $\delta_{i,t-1,t}$ = relative delta measure to measure the difference between the change in the accounting variable and the change in the market price variable from time t - 1 to t for the *i*th firm in *N* sample firms

 $dAC_{i,t-1,t}$ = change in the accounting total variable from time t - 1 to t for the i^{th} firm in N sample firms

 $dM_{i,t-1,t}$ = log change in the market price (market price return) from time t - 1 to t for the i^{th} firm in N sample firms

- i =firm index: i = 1, 2, ..., N
- t = time index

Equation (3.36) is also expressed as:

$$\delta_{i:t-1,t} = (dAC - dM)_{i:t-1,t} \tag{3.37}$$

Where the parameter definitions are the same as equation (3.36) where: the terms in brackets $(dAC - dM)_{i:t-1,t} = dAC_{i,t-1,t} - dM_{i,t-1,t}$

At the sample level the relative delta formula presented in equation (3.36) is given by:

$$\delta_{N,t-1,t} = dAC_{N,t-1,t} - dM_{N,t-1,t}$$
(3.38)

Where:

 $\delta_{N,t-1,t}$ = sample relative delta measure to measure the difference between the sample average change in the accounting variable and the sample average change in the market price variable from time t - 1 to t for N sample firms

 $dAC_{N,t-1,t}$ = sample accounting total variable change from time t - 1 to t for N sample firms

 $dM_{N,t-1,t}$ = sample log market price variable change from time t - 1 to t for N sample firms

N = number of sample firms

t = time index

Equation (3.38) is also expressed as:

$$\delta_{N:t-1,t} = (\overline{dAC - dM})_{N:i,t-1,t}$$
(3.39)

Where the parameter definitions are the same as equation (3.38); where: $(\overline{dAC} - d\overline{M})_{N:i,t-1,t} = \frac{1}{N} \sum_{i=1}^{N} (dAC_{i,t-1,t} - dM_{i,t-1,t})$

3.9.3 Regressions for Relative Delta and Historical Value-at-Risk Actual – Null Hypothesis Evaluation

This section presents details of the test introduced in Section 3.8 with the title: Relative Delta to Value-at-Risk Analysis.

To determine the relationship between the relative delta variable and the Historical Value-at-Risk Actual variable, the time series regressions detailed in this section and the cross-sectional regressions in Appendix O.13 are performed.

To strengthen evidence to verify this research's null hypothesis, it would be expected that the regressive relationship between the relative delta and Value-at-Risk variables for before the 2005 accounting change would be maintained after this change. It would ideally be expected that the relative delta measure would not exhibit a significant relationship to Value-at-Risk levels after 2005. Such a relationship, especially during periods of market price volatility, would signify that the change in the key accounting totals varied in the same direction with the same magnitude as the change in the market price (resulting in a zero or negligible net effect). This effect would show that the relative delta variations were not the same as risk exposures measured using Value-at-Risk.

Time Series Regressions for Relative Delta and Historical Value-at-Risk Actual – Null Hypothesis evaluation

The time series regression applied to determine the relationship between the relative delta variable and the Historical Value-at-Risk Actual variable is specified by the following:

$$\bar{V}_{N:t} = a_T + b_T (\overline{dAC - dM})_{N:t-1,t} + e_{N:t}$$
 (3.40)

Where:

 $\overline{V}_{N:t}$ = mean *Historical Value-At-Risk Actual* at time *t* for *N* sample firms at the 95% confidence level for the 250-day time horizon

 $(\overline{dAC} - dM)_{N:t-1,t}$ = mean relative delta measure: average natural log change in the key accounting variable to the market price return variable at time *t* for *N* sample firms b_T = regression slope

 a_T = regression intercept

 $e_{N:t}$ = error term representing the difference between the actual dependent variable values to the calculated dependent variable values using the regression equation

N = average number of sample firms

T = maximum year count at the sample level

t =time series index: t = 1, 2, ..., T

This regression approach is presented in a generalised form for each key accounting total and market price return in Appendix G.

3.9.4 Accounting Value-at-Risk

This section presents details of the test introduced in Section 3.8 with the title: Accounting Value-at-Risk Analysis.

From the book-to-market ratio specified by Fama and French (1992, 2008), Peterkort and Nielsen (2005), and Bodie et al. (2002), for a *riskless state* the change in the book equity value is matched by the change in the market price.

That is:

$$dB_{t-1,t} = dM_{t-1,t}$$

or:
$$(dB_{t-1,t} - dM_{t-1,t}) = 0$$

(3.41)

Where:

 $dB_{t-1,t}$ = change in book equity from time t - 1 to t $dM_{t-1,t}$ = change in market price from time t - 1 to t

For a *riskless state* this would also mean that a Value-at-Risk measure derived from changes in the book equity variable, dB, must change in the same direction and with the same magnitude as the Value-at-Risk measure derived from the market price return variable, dM.

That is:

$$V_{dB:t-1,t} = V_{dM:t-1,t}$$
(3.42)

Where:

 $V_{dB:t-1,t}$ = Value-at-Risk calculated from the change in the book equity variable for the time interval t - 1 to t

 $V_{dM:t-1,t}$ = Value-at-Risk calculated from the change in the market price variable for the time interval t - 1 to t

Introducing to the right-hand side (RHS) of equation (3.42) the component $(dB_{t-1,t} - dM_{t-1,t})$ from equation (3.41) gives:

$$V_{dB:t-1,t} = (dB_{t-1,t} - dM_{t-1,t}) + V_{dM:t-1,t}$$
(3.43)

Where the parameter definitions in (3.43) are the same as equations (3.41) and (3.42).

Equation (3.43) is tested in this study to estimate the Value-at-Risk for the key accounting variables using the following general formula:

$$V_{dAC:t} = (dAC_{t-1,t} - dM_{t-1,t}) + V_{dM:t}$$
(3.44)
Where:

 $V_{dAC:t}$ = Value-at-Risk estimate for the key accounting total variable at time t $V_{dM:t}^{39}$ = Value-at-Risk derived from the change in the market price variable at time t $(dAC_{t-1,t} - dM_{t-1,t})$ = level of difference between the change in the key accounting total variable and the change in the market price variable or the market price return (this is the accounting to market price relative delta)

3.9.5 Relationship Between Accounting Value-at-Risk and Market Price Return Value-at-Risk and Relative Delta

This section presents details of the test introduced in Section 3.8 with the title: Accounting Value-at-Risk, Market Price Return Value-at-Risk and Relative Delta Analysis.

Rearranging equation (3.44) from Section 3.9.4 gives:

$$V_{dAC:t} - V_{dM:t} = (dAC_{t-1,t} - dM_{t-1,t})$$
(3.45)

Where:

 $V_{dAC:t}$ = Value-at-Risk estimate for the key accounting total variable at time t $V_{dM:t}$ = Value-at-Risk derived from the change in the market price variable at time t $(dAC_{t-1,t} - dM_{t-1,t})$ = level of difference between the change in the key accounting total variable and the change in the market price variable

As stated previously, the term $(dAC_{t-1,t} - dM_{t-1,t})$ is the relative delta measure and is specified in Section 3.2.4. From equation (3.45), it is shown that the difference between the Value-at-Risk from the accounting variable and the Value-at-Risk from the market price variable would be the same as the difference between the change in the accounting and the change in the market price variables. That is, this difference between the accounting and the market price return Value-at-Risk variables is signified by the measure of the accounting to market price relative delta.

³⁹ For this study $V_{dM:t}$ is represented by the *Historical Value-At-Risk Actual* measure at the 95% confidence level, and at the 250-day time horizon.

3.9.6 Regulatory Capital Estimates

This section presents details of the test introduced in Section 3.8 with the title: Relative Delta and Regulatory Capital Analysis.

To estimate the change to the minimum regulatory capital requirement specified by the Basel Capital Accord for before and after the 2005 accounting change, this section develops the *regulatory relative delta* measurement framework. This section develops this framework from its introduction and description provided in Section 2.10 and Appendix K respectively.

The minimum regulatory capital specified by Basel (2011) is set at 8%. This requirement states that the Total Capital, that is Tier 1 Capital plus Tier 2 Capital, must be at a minimum 8% of the total *risk weighted assets* value at all times.

That is:

$$Capital Requirement \ge \frac{Total Regulatory Capital}{Risk Weighted Assets} \ge 8\%$$
(3.46)

Coding expression (3.46), gives:

$$\overline{\omega}_t \ge \frac{\vartheta_t}{\omega_t} \ge 8\% \tag{3.47}$$

Where:

 ϖ_t = minimum capital requirement at time t

 ϑ_t = total regulatory capital at time t

 ω_t = risk weighted assets at time t

t = time index

If the risk weights applied to the risk weighted assets, ω , increase by 1% then to maintain the minimum 8% proportion for the capital requirement, the total regulatory capital, ϑ , would also have to increase by a *relative* minimum of 1%.

As stated in Section 2.10, at present, the measure of financial distress risk captured by the book-to-market ratio as has been studied by Fama and French (2008, 1992) and Peterkort and Nielsen (2005) has not been integrated into the Basel (2011) framework.

This study proposes that the level of financial distress risk that is captured by the bookto-market ratio can be applied to adjust risk weighted assets, ω , thus also reflecting an adjustment to the regulatory capital requirement, ϖ . Such an approach is proposed in this research to apply within the Basel regulatory framework to measure a firm's level of exposure to financial distress risk. Applying the book-to-market ratio to capture a firm's level of exposure to financial distress risk would firstly extend the categories of risk captured within the Basel framework and secondly strengthen the principal approach to monitoring financial distress risk proposed in this study. The proposed measurement approach is to monitor the difference between the change in the accounting measure of a firm's value (by measuring the firm's change in book equity or its *total equity*), and its change in the financial market's valuation (by measuring the firm's change in market price).

Such a measure would be applied to equation (3.47) using the following formula:

$$\varpi_t \ge \frac{\vartheta_t}{\omega_t \times (1 + [BM_t])} \ge 8\% \tag{3.48}$$

Where:

 ϖ_t = minimum capital requirement at time t ϑ_t = total regulatory capital at time t ω_t = risk weighted assets at time t BM_t = book-to-market ratio at time t $(1 + [BM_t])$ = percentage (%) adjustment to risk weighted assets at time tt = time index

The BM measure, while adjusting for a firm's exposure to financial distress risk, is reported in Section 2.6 to be a noisy measure for a portfolio of firms for any given time t (Fama and French 2008). Adapting the Fama and French (2008) change in the book equity value to the change in market price, and applying only positive changes, would provide an alternative measure of a firm's exposure to financial distress risk. Applying this approach equation (3.48) then becomes:

$$\varpi_t \ge \frac{\vartheta_t}{\omega_t \times (1 + [dB_{t-1,t} - dM_{t-1,t}])} \ge 8\%$$
(3.49)

Where:

 ϖ_t = minimum capital requirement at time t ϑ_t = total regulatory capital at time t ω_t = risk weighted assets at time t $dB_{t-1,t}$ = change in book equity value from time t - 1 to t $dM_{t-1,t}$ = change in market price from time t - 1 to t $(1 + [dB_{t-1,t} - dM_{t-1,t}])$ = percentage (%) adjustment to risk weighted assets at time tt = time index

Applying the adjustments specified in equation (3.49) at a minimum would provide a measure of the exposure to financial distress risk evaluated from the positive difference between the change variables, where the change in the book equity level is greater than the change in the market price.

This study proposes that the regulatory capital be adjusted not only for the positive changes, but also for the absolute change, that is, the *unsigned change*, from the change in book equity (or *total equity*) to the change in market price based on a particular threshold. Such that, if the difference between the two variables increase above a specified threshold then only would it have any effect on the regulatory capital. Such an approach expects that the accounting book equity (or *total equity*) variable should closely follow the movements of the market price variable.

In addition to monitoring a firm's book equity (or *total equity*) for a given time period, the change in other key accounting variables to the change in the market price variable would allow the regulatory capital levels to be adjusted to the variable pairs that exhibit the greatest level of difference. Such an approach, that this study proposes, would incentivise and direct a firm to take action to determine the cause for this difference. Based on Section 2.7, criterion 2, applying the *total equity* measure instead of book equity and monitoring the *total equity* change, or total asset, or total liabilities, or net income changes to the market price change is in effect monitoring and applying the accounting to market price relative delta measures specified in Section 3.2.5.

Applying such a regulatory adjustment using the relative delta measures would give:

$$\varpi_t \ge \frac{\vartheta_t}{\omega_t \times (1 + \Delta R_{t-1,t})} \ge 8\%$$
(3.50)

Where:

 ϖ_t = minimum capital requirement at time t

 ϑ_t = total regulatory capital at time t

 ω_t = risk weighted assets at time t

 $\Delta R_{t-1,t}$ = percentage (%) adjustment to risk weighted assets at time t

t = time index

This study introduces, applies and proposes the use of the $\Delta R_{t-1,t}$, termed the regulatory relative delta measure. The regulatory relative delta measure quantifies the unsigned change or difference between the change in key accounting variables, $dAC_{t-1,t}$ and the change in the market price variable, $dM_{t-1,t}$. This study proposes that the largest relative delta level based on the key accounting variables is chosen as the basis for the adjustment to the Basel capital requirement. The four measures proposed for this adjustment are presented in Table 3.11.

Table 3.11 Accounting and Market Price Regulatory Relative Delta

The table shows the absolute *accounting to market price relative delta* measures applied to calculate the proposed *regulatory relative delta* measure, ΔR , for time period t - 1 to t. $dE_{t-1,t}$ is the change in total equity, $dA_{t-1,t}$ is the change in total assets, $dL_{t-1,t}$ is the change in total liabilities, $dI_{t-1,t}$ is the change in net income, $dM_{t-1,t}$ is the market price return, and t is the time index. | | brackets signify the absolute or unsigned value for the bracketed expression.

Proposed $\Delta R_{t-1,t}$ Measure

 $\begin{aligned} \left| dE_{t-1,t} - dM_{t-1,t} \right| \\ \left| dA_{t-1,t} - dM_{t-1,t} \right| \\ \left| dL_{t-1,t} - dM_{t-1,t} \right| \\ \left| dI_{t-1,t} - dM_{t-1,t} \right| \end{aligned}$

Based on the absolute maximum accounting relative delta measures presented in Table 3.11, the regulatory relative delta measure that this study applies is specified by the following general formula:

$$\Delta R_{t-1,t} = \left| dAC_{t-1,t} - dM_{t-1,t} \right|_{max}$$
(3.51)

Where:

 $\Delta R_{t-1,t} = \text{regulatory relative delta measure}$ $\left| dAC_{t-1,t} - dM_{t-1,t} \right|_{max} = \text{absolute maximum relative delta measure}$ t = time index

In equation (3.50) applying the regulatory relative delta equation (3.51) to the regulatory capital, ϑ , provides a measure of how much the regulatory capital must change to maintain the 8% minimum, i.e.

Adjusted Regulatory Capital =
$$\vartheta_t \times (1 + \Delta R_{t-1,t})$$
 (3.52)

Equation (3.52) would be the result of the adjustment made to the risk weighted assets, ω , that is:

Adjusted Risk Weighted Assets =
$$\omega_t \times (1 + \Delta R_{t-1,t})$$
 (3.53)

An intuitive implementation of equation (3.52) to determine the adjusted level of regulatory capital, ϑ_t , is to apply the regulatory relative delta adjustment directly to the 8% minimum capital requirement, such that:

Adjusted Regulatory Capital =
$$8\% \times (1 + \Delta R_{t-1,t})$$
 (3.54)

Equation (3.54) is applied in this study for the UK banks to determine the change in the level of regulatory capital before and after the 2005 accounting change. In addition, this equation is implemented at various thresholds specified by the system of statements in expression (3.55).

 $if \ \Delta R_{t-1,t} \leq \lambda \ then$ $Minimum \ (Min.) Regulatory \ Capital = 8\%$ $if \ \Delta R_{t-1,t} > \lambda \ then$ (3.55)

Adjusted Min. Regulatory Capital = $8\% \times (1 + \Delta R_{t-1,t})$

Where:

 $\Delta R_{t-1,t} = \text{regulatory relative delta}$ $\lambda = \text{threshold level}$ t = time index

From expression (3.55) the regulatory relative delta measure, $\Delta R_{t-1,t}$, would be monitored for the largest absolute accounting total to market price relative delta level, specified in Table 3.11. If the regulatory relative delta level is the same or less than the threshold level, λ , then the unchanged 8% minimum regulatory capital would be applied for the risk weighted assets, ω . This condition is specified by the conditional expression, $\Delta R_{t-1,t} \leq \lambda$, in expression (3.55).

However, if the regulatory relative delta level exceeds the threshold, λ , then the regulatory relative delta adjustment would be applied to change the 8% minimum capital. This condition is specified by the second conditional expression, $\Delta R_{t-1,t} > \lambda$, in expression (3.55).

3.9.7 UK Bank Regressions for Relative Delta and Historical Value-at-Risk Actual

Building on the sample level regression analysis specified in section 3.9.3, this section specifies tests applied at the individual primary bank level. For each primary bank, to determine the relationship between the relative delta variable and the Value-at-Risk Actual, time series regression is applied.

To strengthen evidence to verify this research's null hypothesis at the bank level, applying the same reasoning from Section 3.9.3, it would be expected that the regressive relationship between the relative delta and Value-at-Risk variables for before the 2005 accounting change would be maintained after the change. The ideal

expectation would be that the relative delta measure would not exhibit a significant relationship to Value-at-Risk levels after 2005. From Section 3.9.3 such a relationship would signify that the relative delta variations were not the same as risk exposures measured using Value-at-Risk.

The time series regression applied at the bank level to determine the relationship between the relative delta variable and the Value-at-Risk Actual variable is specified by the following:

$$V_{t} = a_{T} + b_{T}(dAC - dM)_{t-1,t} + e_{t}$$
(3.56)

Where:

 V_t = *Historical Value-At-Risk Actual* at time *t* for a primary sample bank at the 95% confidence level for the 250-day time horizon

 $(dAC - dM)_{t-1,t}$ = relative delta measure: average natural log change in the key accounting variable to the market price return variable at time *t* for a primary sample bank

 b_T = regression slope

 a_T = regression intercept

 e_t = error term representing the difference between the actual dependent variable values to the calculated dependent variable values using the regression equation

T = maximum time series year count

t =time series index: t = 1, 2, ..., T

3.10 Summary

This chapter has formally presented and refined this study's research question. The null hypothesis has been formally presented and a series of detailed null hypotheses specified. Details regarding the methodological approach and quantitative analytical methods that this study applies to test its null hypothesis have been specified. The construction for the extension of the Fama and French (2008) time dependent treatment of the book-to-market ratio, termed the relative delta measure, is detailed for its application to test the null hypothesis. This chapter has also provided the construction details for the regulatory relative delta measurement framework that is also applied to test the null hypothesis.

Chapter 4 Analysis and Results

4.1 Chapter Introduction

This chapter presents and analyses results produced to examine the null hypothesis presented in Section 3.3.1. The null hypothesis expects accounting quality (see Section 2.9) in UK banks that adopted the IASB accounting standards in 2005 to remain unchanged after this adoption. The framework for the tests applied and examination of the results is guided by the hypothetico-deductive methodological approach and falsification theory. These tests and the quantitative analytical methods aligned to this methodological framework are presented and referred to in Chapter 3 (see Chapter 3, Section 3.8, for an introduction to the tests conducted to produce the results presented in this chapter). The general analysis presented in this chapter reports and examines material and statistical significance to build conclusive evidence to determine the level of verifiability, or conversely falsifiability, of this study's null hypothesis. As stated in Chapter 3, the results from these tests are assessed to determine if they provide evidence to confirm the expectations of the conclusive statements presented in Chapter 5. Ultimately, it is the evidence from these tested statements that this study assesses to deduce whether they add strength to verify the null hypothesis, or if they add strength to *falsify* or weaken the null hypothesis.

To gather evidence to test the null hypothesis, as previously stated, quantitative analytical methods are applied for before and after the 2005 accounting change. These analytical methods are applied to determine the relationship between selected market and accounting variables for banks that changed accounting standards in 2005 and are registered with the LSE's UK banking sector. To support the null hypothesis this study expects the banks to exhibit levels of a *difference component*, specified as the measure of the difference between the change in key accounting total variables and the market price return variable (see Sections 3.2.3 to 3.2.5), that remain at the same level after the 2005 accounting change when compared to before. Such an observation, based on the relationship alluded to between the *difference component* and accounting quality in Section 2.9.3, would indicate that there was no effect to accounting quality, after the accounting change. Ideally, based on the research question (see Section 3.2) and from the general financial system's perspective (Basel 2011, IASB 2009), it would be

expected for this *difference component* to be significantly less after the accounting change, and thus reflect an improvement in accounting quality, and also reflect a decrease in the level of exposure to accounting quality risk (see Section 2.9.3), after 2005.

The principal bank variables tested are the market price, the change in market price (market price return), the market price return Value-at-Risk, the key accounting total variables: total equity, total assets, total liabilities and net income, and this study's derived variables: the relative delta, the accounting Value-at-Risk and the regulatory relative delta.

As stated earlier, the null hypothesis test results presented in this chapter concentrate and rely on material and statistical significance. The analytical methods applied to quantify significance for selected variables include descriptive statistics, distribution analysis, correlation analysis and regression analysis. These analytical methods applied, and results presented and analysed in this chapter, are for the time range 1992 to 2009. The analysis and discussions principally concentrate on changes in material and statistical significance for results before 2005, and for results after 2005. Within this 1992 to 2009 time range, results are presented and analysed on a yearly basis. In addition, this study applies the time range approach detailed in Jensen et al. (1972) and Fama and French (2008), by specifying the time series range 1994 to 2008 and subperiods within this time range - 1994 to 2004 and 1994 to 2007 (see Section 3.7.7). The time series ranges assist this study by increasing its observation for the bank sample by pooling the number of firms in the sample by the number of observations. In effect, this time series approach provides $N \times T$ observations, where N is the number of firms in a sample or group and T the number of time series years.⁴⁰

4.1.1 Samples and Control Group Setup

The analytical methods applied in this study as mentioned previously include regression analysis. The specific regression models applied in this study are the cross-sectional regression model specified by Fama and French (2008), and the time series regression

⁴⁰ It is noted that in the result tables where the number of pooled observations are stated, the number of observations may differ by 1 or 2 observations when compared to the $N \times T$ formula. This difference is due primarily to adjustments from rounding effects, and omission of items. Omitted items are presented in Appendix Z. This study considers these adjustments not to impact its results significantly.

model specified by Jensen et al. (1972). When examining the application in this study of these regression approaches when analysing their results, both Fama and French (2006, 2008) and Jensen et al. (1972) find that the cross-sectional regression approach, applied to test variables to better estimate average expected portfolio risk premiums and returns, is subjected to regression slope estimate errors. They find that a regression slope for an entire portfolio is affected by the mechanism for the linear regression model itself. In that, the linear regression model specifies for a relationship a model based on how closely its slope can fit equidistantly from the modelled variable data points. If, at a given instance in time, a proportion of portfolio returns were linearly distributed in clusters of groups at a distance above and below its modeled regression slope, the slope would provide a measure that would be biased towards the average returns for the portfolio. To have a contemporary regression that is less biased and be able to improve estimates of expected returns for firms in such a portfolio, first the firms in the entire portfolio would need to be grouped into those that have returns above the slope and those that have returns below the slope. Then two regressions would be performed, one for each group. In this design, the slopes produced by the resulting models would better reflect the expected risk premiums and returns for those concentrated firms compared to the slope for the entire portfolio.

To determine variables that would help to better estimate expected returns, Fama and French (2008) grouped firms registered with the New York Stock Exchange (NYSE) into two groups; firms with above and, firms with below the 20th percentile level of market capitalisation, where market capitalisation is specified to be analogous to the market value measure. Cross-sectional regression was then applied to the two individual groups, effectively two portfolios. Although Fama and French (2008), for a given time instance, does apply cross-sectional regression to each market capitalised group, the significance for each regression slope is, however, calculated by applying the Fama and McBeth (1973) *t-statistic* that is based on a time series average of cross-sectional regression slopes.

Jensen et al. (1972) grouped NYSE firms based on monthly returns into 10 equal size groups, effectively 10 portfolios, with each portfolio containing contiguous and increasing returns. These returns are averaged on a monthly basis for each portfolio. Time series regression is then applied to each portfolio's average returns to estimate the individual portfolios expected risk premiums and returns compared to a market portfolio.

The grouping process applied by Fama and French (2008) and Jensen et al. (1972) in essence is applied in this study for regression and other analytical methods with the motivation to examine its null hypothesis. From the sample selection criteria specified in Section 3.2.4, this study approaches the grouping effect in two ways. Firstly, to test the null hypothesis at group level, this study treats banks in the LSE banking sector as a clustered group. This study tests, for before and after the 2005 accounting change, selected variables from this primary group as a portfolio in isolation from other firms in the LSE. Then a second *portfolio* of banking related financial services firms that apply the bank accounting template and also changed accounting standards in 2005, is selected from the LSE. To corroborate the validity for the primary *portfolio* test results, the same primary group tests are also applied to the *closely clustered* secondary group. To determine the general bank and banking related firms reactions to the 2005 change, the primary group is combined with the secondary group. The same tests applied to the primary group are applied to this primary and secondary grouping. Secondly, to refine the null hypothesis tests, the primary group is ungrouped to its constituent firms. Time series regression is then applied to selected variables at the individual bank level to determine variable reactions before and after 2005. This study formally specifies the primary group as the Primary sample and the secondary group as the Secondary sample, and the primary and secondary grouping collectively as the Primary and Secondary sample. The firms selected for the Primary sample and Secondary sample are presented in Table 3.6.

Examining the change from before to after the 2005 accounting adoption of the Primary sample and the Secondary sample *difference component*, measured from the difference between the change in accounting total variables and the change in the market price variable, would provide strong evidence to accept or reject this research's null hypothesis. That is, if accounting quality measured by the average level of the *difference component* from the change variables remained at the same level after 2005, this would be considered as direct evidence to strengthen the case to accept the null hypothesis. On the contrary, if a materially or statistically significant change in this *difference component* after 2005 is evidenced, as discussed in Section 1.2, it may be contended that this increase in *difference* is as a consequence of either normal economic

and/or general market forces, or deterioration in accounting quality, or a combination of both.

To isolate the Primary sample and the Secondary sample to effects from general economic and general market variable reactions, this study introduces a control group. This control group is selected from a sample of firms that did not change accounting standards in 2005 but applied the UK GAAP's ASB accounting standards approach for before and the subsequent years after 2005. These control group firms are selected from the LSE and are banking related financial services firms that apply the bank accounting template. This study refers to the selected control group of firms formally as the Control group. The Control group selection criteria are specified in Section 3.6.1 and the Control group firms are presented in Table 3.7. Tests applied to the Primary sample and the Secondary sample are also applied to the Control group to determine if any significant material and statistical reactions, to general economic and market conditions, are observed for the Primary sample and the Secondary sample compared to the Control group. The same tests are applied to the samples and the control group to determine also if any significant material and statistical changes to the *difference component* were observed for the Primary sample and the Secondary sample compared to the Control group. This study applies such a test approach to effectively control for the 2005 accounting change.

The samples and the control group reactions to economic and general market conditions are tested using the change in the market price (market price return) and Value-at-Risk variables. These samples and control group variables are tested with corresponding *change* and Value-at-Risk variables of selected Gross Domestic Product (GDP) sector indices, stock market indices, interest rate indicators and foreign exchange rates.

4.1.2 Chapter Organisation

Generally, the results sections in this chapter are structured to introduce first the results tables and the figures examined in that section. The actual result tables and figures are then presented. A single result table⁴¹ is generally organised into sample and control

⁴¹ The tables in this chapter tabulate numerical results to varying degrees of decimal point significance. Zeros or hyphens exhibited as an entire result are considered below the generally reported decimal point significance or cannot be substantiated. Such entries are not considered significant to the findings of this research.

group result panels. Where, in general, the first panel presents results for the Primary and Secondary sample, the second panel presents results for the Control group, where presented, the third panel presents results for the Primary Sample, and where presented, the fourth panel presents results for the Secondary sample. When presenting figures, this same *samples and control group* sequence is generally adopted.

The results reported in the tables and figures are then examined and analytically discussed. This analysis concentrates on evaluating evidence for the presence or the non-presence of material and statistical variable significance for before and after 2005. The analysis generally follows the pattern of examining first the Primary and Secondary sample results, then the Control group results, then when presented may be followed by an analysis of the Primary sample results.

To emphasise the analytical approach presented in this chapter, for the key variables introduced and tested in each results section an analysis is presented. This analysis examines whether the results provide evidence to support this study's theoretical discussion (see Chapter 3, Section 3.2) and null hypothesis (see Chapter 3, Section 3.3.1), that discusses whether accounting quality improved for UK banks that adopted the IFRS accounting standards in 2005 (see also Chapter 2, Sections 2.2 and 2.3). The analysis presented, at the samples and control group level and also at the individual bank level, examines variable reactions for before the 2005 accounting change compared to the reactions after the change. The findings from this analysis are then applied to test the null hypothesis by examining a series of testable conclusive statements. This series of conclusive statements are presented in Chapter 5, Table 5.1 to Table 5.11 under the *Statement Tested* column. From this basis and for reasons of brevity, it is expected that the variable analysis presented and referred to in this chapter is linked to, and interpreted within, this general theoretical and methodological framework.

The following provides a summary description of the sections presented in this chapter. In addition, a description of the tests, introduced and referred to in Section 3.8, that are applied to produce the results is presented together with references to relevant results from the appendices. Section 4.2 presents summary results and analysis based on relative percentage changes. These percentage changes are based on results presented at a detailed level in Table P.1, Table P.2, Table R.4, Table 4.13, and Table 4.24. Section 4.2.1 details the formulas applied to produce the percentage changes based on two averaging approaches applied to the detailed results.

Section 4.2.2 presents relative percentage change results and analysis for before and after 2005 for yearly market price return descriptive statistics. In addition, percentage change results and analysis for distribution descriptive statistics are presented. These percentage change levels are based on the results from Appendix P.1.

Section 4.2.3 presents relative percentage change results and analysis for before and after 2005 from the variance-covariance, historical and Monte Carlo simulation Valueat-Risk backtested results presented in Appendix P.2. These results are based on selected Value-at-Risk time horizons at the 95% confidence level and calculated for a selected number of historical and simulated market price return observations. From the basis of the analysis presented in Section 4.2.3 and Appendix P.2, Appendix P.3 presents results for the Historical Value-at-Risk Actual measure⁴² that is tested at the 95% confidence level to determine its time horizon characteristics. Appendix P.4 presents results for the 250-day time horizon for this Value-at-Risk measure tested to determine its confidence level characteristics. Appendix P.5 tests this Historical Value-at-Risk Actual variable with the market price return variable to determine their levels of significance.

Section 4.2.4 provides the analysis for the reactions to economic and market conditions, for the samples and the control group market price return variable and the Historical Value-at-Risk Actual variable. The economic condition reaction analysis is provided with reference to tests with GDP levels. Analysis of the market conditions is provided with reference to tests with stock market indices, selected short and long-term benchmark interest rates, and selected foreign exchange currency pair rates. The analysis presented in Section 4.2.4 is based on the results and analyses presented in Appendix P.6 to Appendix P.15.

⁴² The Historical Value-at-Risk Actual measure is specified in Appendix N.8.

Section 4.2.5 presents relative percentage change results to determine how the market price return variable and the Historical Value-at-Risk Actual variable reacted with selected variables that include the book-to-market ratio, market value and key accounting total variables: total equity, total assets, total liabilities and net income. The tests are conducted on fiscal year-end totals. These percentage change levels reflect the descriptive statistics and correlation results presented in Appendix R and Appendix S.

Section 4.2.6 presents relative percentage change results for before and after 2005 for the relative delta variables applied to quantify the level of difference between the change in key accounting totals and market price return on a yearly basis. These percentage change levels are based on the results and analyses presented in Sections 4.5 and 4.8.

Section 4.2.7 presents relative percentage changes for before and after 2005 from the regression results based on the Historical Value-at-Risk Actual variable and the relative delta variable. These percentage change levels are based on the results and analyses presented in Sections 4.6 and V.5.

Section 4.3 presents the time series regression results for the relationship between the change in key accounting totals and market price return. Appendix U presents the cross-section regression results for the same relationship on a yearly basis.

Section 4.4 presents, for the samples and control group, t-test results that measure the statistical significance of differences between the key accounting totals and market price return regression slope coefficients presented in Section 4.3. The significance of differences tests are conducted for slopes within and between the samples and control group.

Section 4.5 presents results for the relative delta variables applied to quantify the difference between the change in key accounting totals and market price return on a yearly basis.

Section 4.6 presents time series regression results for the Historical Value-at-Risk Actual variable reactions to changes in the relative delta variable. Appendix V.1 to Appendix V.4 presents the cross-section regression results for the same relationship on a yearly basis. Appendix V.5 presents the time series regression results for the same relationship at the individual bank level.

Sections 4.7.1 and 4.7.2 present estimates for the *accounting Value-at-Risk* measure and its relation to the *market price return Historical Value-at-Risk Actual* and the *relative delta*.

Section 4.8 presents relative delta variable results for the banks on an individual basis. These results quantify the difference between the change in key accounting totals and market price return on a yearly basis.

Section 4.9 presents at the individual bank level, estimates of the *accounting Value-at-Risk* measure and its relation to the market price return Historical Value-at-Risk Actual variable and the *relative delta* variable.

Section 4.10 presents the adjustment to the Basel minimum capital requirement at the individual bank level from the proposed regulatory relative delta framework. This framework is specified in Section 3.9.6 and described in Appendix K.

Section 4.11 proposes a conceptual bank technology framework at the firmwide supervisory and regulatory reporting level. This framework is proposed to address data considerations in order to report the measures accounting Value-at-Risk, relative delta, and the regulatory relative delta.

Section 4.12 presents a summary of the analysis and the main findings from the results in this chapter.

4.2 Analysis and Summary Results for Relative Percentage Change of Averages for Before to/and After 2005

This section reports and analyses summary results that are produced from *detailed* results presented in Table 4.6 to Table 4.27 and the tables in Appendix P to Appendix R. These detailed results are considered central to the examination of this study's null hypothesis. The summary results in Table 4.1 to Table 4.4 are based on percentage changes that are presented by first applying an averaging process to the selected

detailed results (see Section 4.2.1). These results consist of percentage change analysis for *before to after 2005* and/or for *before and after 2005*. The latter approach is applied to provide greater significance to the results recorded after 2005, this the year of the accounting change. The results for *before to after 2005* are calculated by applying equation (4.1). The results for *before and after 2005* are calculated by applying equation (4.2). These approaches differ from the percentage change analysis in Sections 4.3 to 4.10 and Appendix P to Appendix V that applies percentage changes to *non-averaged* detailed results.

4.2.1 Percentage Change Calculations for Summary Results

The following percentage change calculations are applied to present the summary results in Table 4.1 to Table 4.4.

Before to After 2005 Relative Percentage Change of Averages

Table 4.1 Panel A and Table 4.4 Panel A present results that compare variable reactions for before and after 2005 by applying the following percentage change formula:

$$\Gamma_{L,U} = \frac{\bar{\Gamma}_U - \bar{\Gamma}_L}{\bar{\Gamma}_L} \times 100 \tag{4.1}$$

Where:

 $\Gamma_{L,U}$ = relative percentage change from the Lower year average, $\overline{\Gamma}_L$, to Upper year average, $\overline{\Gamma}_U$.

 $\overline{\Gamma}_L$ = Lower year average calculated by taking the average for the year range starting from year 1992 to the ending year 2004. Similar to $\overline{\Gamma}_U$ the start year applied may be 1993 or 1994.

 $\overline{\Gamma}_U$ = Upper year average calculated by taking the average for the year range starting from year 1992 to the ending year: 2005, 2006, 2007, 2008, or 2009. Depending on the results analysed, the start year applied for analysis may be 1993 or 1994.

Before and After 2005 Relative Percentage Change of Averages

This section presents results in Table 4.1 Panel B to Table 4.3 and Table 4.4 Panel B that compare variable reactions for before and after 2005 by applying the percentage change formula shown in equation (4.2). This formula specifies an *upper bound* year

that starts at the 2005 year compared to the 1992 starting year applied in equation (4.1). This formula is specified by:

$$\Gamma_{L,u} = \frac{\bar{\Gamma}_u - \bar{\Gamma}_L}{\bar{\Gamma}_L} \times 100 \tag{4.2}$$

Where:

 $\Gamma_{L,u}$ = relative percentage change from the Lower year average, $\overline{\Gamma}_L$, to Upper year average, $\overline{\Gamma}_u$.

 $\overline{\Gamma}_L$ = Lower year average calculated by taking the average for the year range starting from year 1992 to the ending year 2004. Depending on the results analysed, the start year applied for analysis may be 1993 or 1994.

 $\overline{\Gamma}_u$ = Upper year average calculated by taking the average for the year range starting from year 2005 to the ending year: 2005, 2006, 2007, 2008, or 2009.

As indicated earlier, this approach specified in equation (4.2) provides greater significance to the variable levels recorded for the 2005 year and after, for its percentage changes, than does equation (4.1). It is expected that both formulas, when applied to analyse the same variables, will infer the same general pattern. As evident when comparing the results in Table 4.1 Panel A (reported using equation (4.1)) and Table 4.1 Panel B (reported using equation (4.2)) that equation (4.2) generally *magnifies* the percentage change results reported using formula (4.1).

4.2.2 Market Price Returns Before and After 2005 – Examination of Material Significance

The results presented in Table 4.1 show percentage changes for the average samples and control group variables: market price returns, variance-covariance Value-at-Risk and distribution statistics, for before and after 2005. The average percentage changes presented in this table are produced from the results in Table P.1. The difference between the results presented in Table 4.1 Panel A and Panel B is that Panel A applies equation (4.1) and Panel B applies the *more 2005 sensitive* equation (4.2).

Table 4.1Summary Results: Relative Percentage Change of Averages for
Market Price Return Distribution and Variance-Covariance Value-at-Risk
Descriptive Statistics for Before To/And After 2005

The Table 4.1 Panel A columns present the relative percentage changes of averaged results for *before 2005 to after 2005* calculated using equation (4.1). The Table 4.1 Panel B columns present the relative percentage changes of averaged results for *before 2005 and after 2005* calculated using equation (4.2). Table 4.1 Panel A and Panel B columns and sub-panels present the following:

The Variable column presents the following sub-panels for the samples and the control group respectively: the Return sub-panel presents the market price return percentage changes; the Variance-Covariance VaR sub-panel presents market price return variance-covariance Value-at-Risk percentage changes; the Dist. sub-panel presents the 300-day market price return distribution percentage changes; the Skewness sub-panel presents the distribution skew percentage changes; the Kurtosis sub-panel presents the distribution kurtosis percentage changes.

The Sample and Control Group column presents the samples and the control group result subpanels for the variables presented in the Variable column.

The Percentage Change of Averages (%) column presents the following columns: the Mean column presents the percentage changes of the mean for the variables presented in the Variable column; the Standard Deviation of Sample Means column presents the percentage changes of the standard deviation of means for the variables presented in the Variable column; the 1993 to 2004 column represents the year range applied to calculate the lower bound average in equation (4.1) and (4.2); in Table 4.1 Panel A, the 1993 to 2007, 1993 to 2008 and 1993 to 2009 columns represent the year ranges applied to calculate the upper bound averages in equation (4.1); in Table 4.1 Panel B the 2005 to 2007, 2005 to 2008 and 2005 to 2009 columns represent the year ranges applied to calculate the upper bound averages in equation (4.2).

Table 4.1 (Continued)

PANEL A Percentage Changes for Before To After 2005

		Percentage Change of Averages (%)											
	Sample and Control		Mean		Standard Deviation of Sample Means								
Variable	Group	1993 to 2004											
		1993 to 2007	1993 to 2008	1993 to 2009	1993 to 2007	1993 to 2008	1993 to 2009						
Return	Primary and Secondary Control Primary Secondary	-9.43 31.20 -18.87 -3.26	-49.06 -31.00 -67.25 -38.95	-36.07 -1.18 -58.24 -22.85	-0.92 0.25 -6.09 2.22	8.03 0.63 9.24 6.25	10.74 0.07 16.62 7.84						
Variance- Covariance VaR	Primary and Secondary Control Primary Secondary	3.69 1.36 8.45 0.78	-0.56 -1.94 5.31 -4.15	-10.60 -9.31 -9.12 -11.49	-10.00 -8.20 -8.86 -9.74	-4.77 -11.48 -4.11 -3.37	5.88 -10.90 24.20 -1.36						
Dist.	Primary and Secondary Control Primary Secondary	-0.31 40.69 -9.71 6.67	-23.85 13.79 -34.64 -16.25	-43.53 -14.81 -68.74 -28.24	-5.78 -7.80 -11.58 -4.00	4.17 -5.93 -3.29 7.50	12.16 -7.88 15.17 5.10						
Skewness	Primary and Secondary Control Primary Secondary	-22.59 -8.74 14.25 -20.29	-29.24 -4.11 25.97 -28.33	-38.25 -0.66 6.45 -29.52	-11.25 4.26 -12.57 -9.39	-12.13 3.06 -16.18 -9.14	-10.07 3.39 -3.50 -12.94						
Kurtosis	Primary and Secondary Control Primary Secondary	-7.90 1.16 -5.99 -8.46	-8.65 -0.23 -6.48 -9.28	-7.97 -1.00 2.60 -11.04	-15.05 -1.91 -16.36 -14.28	-16.24 -6.07 -20.06 -14.55	-14.37 -8.53 -8.24 -17.71						
Excess Kurtosis	Primary and Secondary Control Primary Secondary	-10.92 1.97 -9.71 -11.21	-11.96 -0.40 -10.50 -12.30	-11.01 -1.71 4.22 -14.63	-15.05 -1.91 -16.36 -14.28	-16.24 -6.07 -20.06 -14.55	-14.37 -8.53 -8.24 -17.71						

Panel note:

Percentage changes calculated using equation (4.1).

Percentage Changes for Before And After 2005													
		Percentage Change of Averages (%)											
Variabla	Sample and Control		Mean		Standa Sa	Standard Deviation of Sample Means							
v arrable	Group	1993 to 2004											
		2005 to 2007	2005 to 2008	2005 to 2009	2005 to 2007	2005 to 2008	2005 to 2009						
Return	Primary and Secondary Control Primary Secondary	-47.17 156.00 -94.37 -16.28	-196.23 -124.00 -269.01 -155.81	-122.64 -4.00 -198.03 -77.67	-4.59 1.27 -30.43 11.11	32.11 2.53 36.96 25.00	36.51 0.25 56.52 26.67						
VaR	Primary and Secondary Control Primary Secondary	18.47 6.80 42.26 3.89	-2.23 -7.77 21.26 -16.61	-36.05 -31.65 -31.02 -39.08	-50.00 -40.98 -44.30 -48.72	-19.08 -45.90 -16.46 -13.46	20.00 -37.05 82.28 -4.62						
Dist.	Primary and Secondary Control Primary Secondary	-1.54 203.45 -48.57 33.33	-95.38 55.17 -138.57 -65.00	-148.00 -50.34 -233.71 -96.00	-28.89 -38.98 -57.89 -20.00	16.67 -23.73 -13.16 30.00	41.33 -26.78 51.58 17.33						
Skewness	Primary and Secondary Control Primary Secondary	-112.97 -43.69 71.24 -101.47	-116.94 -16.46 103.90 -113.30	-130.06 -2.24 21.92 -100.35	-56.26 21.28 -62.85 -46.96	-48.50 12.22 -64.71 -36.55	-34.24 11.52 -11.91 -44.00						
Kurtosis	Primary and Secondary Control Primary Secondary	-39.52 5.82 -29.97 -42.30	-34.62 -0.93 -25.92 -37.13	-27.10 -3.40 8.85 -37.54	-75.26 -9.54 -81.82 -71.41	-64.97 -24.27 -80.25 -58.20	-48.84 -29.01 -28.03 -60.20						
Excess Kurtosis	Primary and Secondary Control Primary Secondary	-54.60 9.83 -48.54 -56.04	-47.82 -1.61 -41.98 -49.19	-37.44 -5.81 14.34 -49.73	-75.26 -9.54 -81.82 -71.41	-64.97 -24.27 -80.25 -58.20	-48.84 -29.01 -28.03 -60.20						

Table 4.1 (Continued)

PANEL B

Panel note: Percentage changes calculated using equation (4.2).

Table note: Summary percentage change results are produced from the detailed results presented in Table P.1.

Analysis

Table 4.1 Panel A and Panel B show that, in the respective Returns sub-panel, the average 1-day yearly returns for the Primary and Secondary sample experienced a materially significant decrease after 2005 when compared to before. The Control group registered higher returns in 2007, however, for 2008 and 2009 followed a similar returns profile to the Primary and Secondary sample. The standard deviation of sample means for the Primary and Secondary sample shows that the difference in the levels of returns from each firm increased in 2008 and 2009. The evidence from Table 4.1 suggest that the pricing levels for firms within the Primary banking sample, that is for the banks in the UK banking sector, generally became more varied after 2005 than before.

Table 4.1 Panel A and Panel B also present, in the respective Variance-Covariance VaR sub-panel, the samples and the control group average yearly variance-covariance Valueat-Risk for the 1-day time horizon at the 95% confidence level. For the Value-at-Risk results presented, a larger negative number specifies an increase in Value-at-Risk. Examining the results in Table 4.1 shows that a general increase in variance-covariance Value-at-Risk was registered after 2005. This suggests an increase in average market price volatility levels for the samples and the control group after 2005, specifically for years 2008 and continuing in to 2009.

Table 4.1 Panel A and Panel B also present the distribution statistics based on the 300 historical 1-day returns applied to determine the variance-covariance Value-at-Risk results presented in the Variance-Covariance VaR sub-panel. The Dist. sub-panel presents the samples and the control group average distribution mean for the 300 historical 1-day returns. The distribution results exhibit varying positive and negative average means after 2005.

The Skewness sub-panel presents the degree of skewness for the distribution and shows that the Primary and Secondary sample become increasingly left skewed (negatively skewed). This is due to the characteristics from the Secondary sample distribution that exhibited higher sample positive returns after 2005. However, the Primary sample shows an increasing right skewness (positive skewness) from 2005 to 2008, then exhibiting movement towards the normal distribution in 2009. This movement resultant from increased negative returns during 2008 and then *decreased* negative returns in

2009. The Control group shows a decreasing negative skewness, exhibiting a move towards the normal distribution after 2005, and thus showing decreasing positive returns after 2005.

The Excess Kurtosis sub-panel that presents the Kurtosis results minus 3 (-3), after averaging, shows that the Primary and Secondary sample exhibits a decrease in excess kurtosis after 2005, suggesting a decreasing leptokurtic distribution. This distribution shows that the returns are generally distributed at a *flatter* level around the mean after 2005 when compared to before. This observed excess kurtosis appears to be influenced by the distribution characteristics of the Secondary sample. The Primary sample also shows a movement towards a more flat distribution in 2007 and 2008, and a movement towards a more narrowly peaked leptokurtic distribution in 2009. The Control group generally exhibits a decreased leptokurtic distribution after 2005.

4.2.3 Average Value-at-Risk Levels Before and After 2005 – Examination of Material Significance

Table 4.2 presents relative percentage changes for the Value-at-Risk models for before and after 2005 from results presented in Table P.2. Analysis of the Table 4.2 results presented in this section refers to evidence from the Historical Value-at-Risk Actual variable charts presented in Figures P.1 to P.24.

Table 4.2Summary Results: Relative Percentage Change of Averages for
Value-at-Risk Models for Before And After 2005

The Table 4.2 columns present the relative percentage changes of averaged results for *before 2005 and after 2005* calculated using equation (4.2). The Table 4.2 columns and panels present the following:

The Value-at-Risk Model column presents the Value-at-Risk model applied to the Primary and Secondary sample and the Control group for the 1-day, 250-day and 500-day time horizon panels.

The Sample and Control Group column presents the samples and the control group result panels.

The Time Horizon [*Year Range*] column presents the year ranges applied to calculate the lower bound average in equation (4.2).

The Percentage Change of Averages (%) column presents the following columns: the 1993 to 2004 column repeats the Time Horizon [*Year Range*] column's maximum year range applied to calculate the lower bound average in equation (4.2); the 2005 to 2007, 2005 to 2008 and 2005 to 2009 columns represent the year ranges applied to calculate the upper bound averages in equation (4.2); the Returns column presents the market price return percentage changes; the VaR column presents the Value-at-Risk percentage changes for the backtest results based on the number of times the market price return breached the Value-at-Risk level within the specified time horizon; and the End H. (*End-of-time Horizon breach*) column presents backtest results reproduced from Table P.2 and indicates if the actual market price return variable breached the Value-at-Risk level at the time horizon date.

			Time	Time Percentage Change of Averages (%)													
	Value-at-Risk	Sample and Control	Horizon	20n 1993 to 2004 (Maximum Year Range before 2005)													
	Model	Group	[Year]		2	005 to 2007	7		2	2005 to 200	8			2005 to 200	19		
			[Range]	Returns	VaR	In H.	End H.	Returns	VaR	In H.	End H.	Returns	VaR	In H.	End H.		
	Variance- Covariance	Primary and Secondary Control	1 Davi	-47.17 156.00	18.47 6.80	-	-	-196.23 -124.00	-2.33 -7.77	-	-	-122.64 -4.00	-36.05 -31.65	-	-		
	Historical	Primary and Secondary Control	$\begin{bmatrix} 1993 \ to \end{bmatrix}$	-47.17 156.00	17.74 10.76	-	- -	-196.23 -124.00	-2.01 -6.85	-	-	-122.64 -4.00	-35.45 -32.79	-	-		
	Monte Carlo Simulation	Primary and Secondary Control	1 2004 1	-47.17 156.00	20.92 2.84	-	-	-196.23 -124.00	-3.00 -13.80	-	-	-122.64 -4.00	-38.93 -40.53	-	-		
	Variance- Covariance	Primary and Secondary Control	250 Day	59.28 495.96	20.02 -1.74	-86.62 -97.64	-	-104.49 154.21	23.77 5.29	1.28 -53.42	1 1	-171.38 -1.68	15.91 2.29	91.24 18.87	1 1		
	Historical	Primary and Secondary Control	$\begin{bmatrix} 1995 \ to \end{bmatrix}$	59.28 495.96	54.65 77.67	-39.07 -51.14	1 -	-104.49 154.21	58.58 85.65	15.55 -16.55	1 1	-171.38 -1.68	8.45 57.66	40.25 10.60	1 1		
	Monte Carlo Simulation	Primary and Secondary Control	1 2004 1	59.28 495.96	50.65 70.07	-38.69 -55.97	1 -	-104.49 154.21	53.76 75.16	13.99 -40.25	1 1	-171.38 -1.68	8.94 53.02	42.49 -16.98	1 1		
	Variance- Covariance	Primary and Secondary Control	500 Dec	230.34 2,459.80	-29.56 -35.61	-83.38 -91.57	-	38.90 1,603.68	-13.77 -20.77	-60.21 -84.19	1 -	-141.89 737.25	-1.25 -8.39	27.73 -39.91	1 1		
	Historical	Primary and Secondary Control	$\begin{bmatrix} 1998 \ to \end{bmatrix}$	230.34 2,459.80	-1,332.15 -212.52	-96.64 -98.74	-	38.90 1,603.68	-924.51 -108.03	-67.72 -73.62	1 1	-141.89 737.25	-642.44 -38.88	-19.71 -28.48	1 1		
	Monte Carlo Simulation	Primary and Secondary Control	ı 2004 J	230.34 2,459.80	-633.39 -283.17	-62.70 -94.07	-	38.90 1,603.68	-429.12 -162.15	-45.19 -49.32	1 1	-141.89 737.25	-289.36 -77.82	-37.42 -8.93	1 1		

Table 4.2 (Continued)

Table notes:

Percentage changes calculated using equation (4.2). Summary percentage change results are produced from the detailed results presented in Table P.2.

146

Analysis

Table 4.2 presents the three Value-at-Risk models applied in this study. The Returns column in this table shows the change after 2005 compared to before for average yearly returns calculated using the specified number of historical time horizon days. It is noted that the returns for the 1-day variance-covariance Value-at-Risk model for the Primary and Secondary sample and the Control group are the same changes presented in Table 4.1 Panel B. In addition, the End H. column presents a coding where a 1 registers a yearly end-of-time horizon breach. The results presented in this column are reproduced from Table P.2 and do not make the relative comparison to before 2005.

For the 1-day time horizon, Table 4.2 shows that the three Value-at-Risk models did not register significant breaches⁴³ based on yearly averages. However, examining the charts presented in Figures P.1, P.2, P.7, P.8, P.13, P.14, P.19 and P.20, it is evident that observable materially significant breaches did take place after 2005 for the Historical Value-at-Risk Actual variable. This evidence is visually corroborated by the charts presented in Figures 1.1 and 1.2 that uses the Monte Carlo Simulation Value-at Risk model. Examining Table 4.2 and Table P.2 (Panels A and B), the three models show that the Value-at-Risk levels generally increased significantly for both the Primary and Secondary sample and the Control group after 2005 until 2009.

For the 250-day time horizon, Table 4.2 exhibits a decrease in the average yearly Valueat-Risk for both the Primary and Secondary sample and the Control group after 2005. This decrease is registered for the years 2007, 2008 and 2009. However, examining the percentage change of averages for the 2005 year, a materially significant percentage increase (-26.2%) was registered for the Variance-Covariance Value-at-Risk model (see Table P.2). Examining Table P.2 (Panels C and D), the three models show that the Value-at-Risk levels generally decreased after 2005 until 2007 and then increased significantly from 2008 to 2009.

For the 250-day *in-time horizon breaches*, Table 4.2 shows that generally after 2005 there was an increase in the number of breaches for the Primary and Secondary sample.

⁴³ For the 1-day time horizon both the *in-time horizon breaches* and *end-of-time horizon breaches* have the same meaning.

Changes registered for 2009 exhibit a materially significant number of breaches after 2005 for all three models. The Control group after 2005 consistently registered lower levels of breaches compared to the Primary and Secondary sample. The *end-of-time horizon breaches* show that the three models registered breaches for both the Primary and Secondary sample and the Control group after 2005. It is noted that in Table P.2 an end-of-time horizon breach was registered for the Control group from the historical and Monte Carlo simulation models prior to 2005, in 2001.

The 500-day time horizon in Table 4.2 shows that, for both the Primary and Secondary sample and the Control group, the average yearly Value-at-Risk exhibit materially significant increases after 2005. These increases are registered by the three models with the variance-covariance model exhibiting a lower level of increase.

For the 500-day *in-time horizon breaches*, Table 4.2 shows that generally after 2005 there was a decrease in the average number of breaches for both the Primary and Secondary sample and the Control group. The *end-of-time horizon breaches* show that the three models also registered breaches after 2005. Table P.2 (Panel E), shows that end-of-time horizon breaches were registered for the historical and Monte Carlo simulation models prior to 2005, in 2002 and 2003.

Table 4.2, although showing summary results, loses some information due to rounding effects from arithmetic averaging. As stated earlier in this section, a more detailed graphical interpretation of the Table 4.2 results can be observed from the Value-at-Risk charts presented in Figures 1.1 and 1.2, and Figures P.1 to P.24. In addition, these results can also be observed in Figures 1.3 to 1.5.

Examining Table 4.2 and Table P.2, the three models show similar Value-at-Risk patterns for both the Primary and Secondary sample and the Control group after 2005. When examining the actual Value-at-Risk levels, it is difficult to draw a common conclusion from the three models. However, a highly significant result is the 500-day Value-at-Risk that generally shows firms in the Primary and Secondary sample became exposed to a significantly greater level of Value-at-Risk after 2005 when compared to the firms in the Control group.

4.2.4 GDP and General Market Conditions Before and After 2005 – Examination of Statistical Significance

This study has tested the market price return variable and Value-at-Risk variable to determine how the UK banks reacted to selected GDP sectors and general market indices before and after 2005. The general market indices were chosen to reflect selected domestic and world stock indices levels, selected short-term and long-term interest rates, and selected foreign exchange rates. To provide an analysis of the GDP sectors and general market reactions, the results presented in the tables from Appendix P.6 to P.15 are directly referenced.

Analysis

The results presented for the return variable in Table P.8, and for the Value-at-Risk variable in Table P.9, the sample panels generally exhibit evidence that the banks returns and Value-at-Risk reacted with statistical significance after 2005 in relation to the selected GDP sectors. A similar pattern of significance was exhibited with the selected LIBOR rates and foreign exchange currency pairs after 2005 compared to before. The results with the LIBOR variables are presented in Table P.13 and Table P.14, and foreign exchange currency pair variable results are presented in Table P.17 and Table P.18. The selected stock indices reacted differently exhibiting much closer levels of significance when comparing before and after 2005. However, maintaining the same pattern as exhibited by the other indicators tested, the stock indices also show statistical significance after 2005 compared to before. Results for the selected stock indices returns are presented in Table P.11, and Value-at-Risk results in Table P.12. Presented in Table P.15, the statistical evidence from the GILT Government benchmark bond rate tests suggest a deficiency in significant reactions registered before and after 2005 with the market price return variable. However, Table P.16 shows that generally the GILTS bond Value-at-Risk variable, until the 14-year bond and after the 20-year bond maturities, exhibit statistical significance with the sample Value-at-Risk variable.

It is difficult to conclude from these results that there were any direct effects from those sectors to influence the market price returns of UK banks after 2005. On the contrary, it may be contended that there is evidence to infer the banking sectors may have influenced these observed significances after 2005. This evidence being that the reported significance is in 2008, the year that banks showed significantly lower stock

market prices, with lower levels of significance reported for 2007. Examining the Control group results, there is strong evidence that shows the market price returns and Value-at-Risk variables experienced similar changes when compared to firms that changed accounting standards in 2005.

The deduction from the results in Table P.8 to Table P.18 is that the UK banks and the Control group reacted similarly to GDP and selected market indicators after 2005. From this evidence, it is difficult to assert that any isolated bank volatility after 2005 would have originated from reactions to the GDP or the selected market indicators. Such a conclusion adds strength to the reasoning that the UK banking sector volatility levels and Value-at-Risk breaches exhibited in Figure 1.1 and Value-at-Risk breaches analysed in Section 4.2.3, Appendix P.2 and P.3, especially during 2005 and 2007, would be as a consequence to factors external to the selected economic and market indicators tested.

4.2.5 Selected Variable Totals Before and After 2005 – Examination of Material Significance

Table 4.3 presents percentage changes for the average samples and control group variable totals: book-to-market ratio (BM), market price, total equity, total assets, total liabilities and net income. In addition, the Historical Value-at-Risk Actual variable measured from 1994 is presented. The average percentage changes presented in Table 4.3 are produced from the results in Table R.4.

Table 4.3Summary Results: Relative Percentage Change of Averages forDescriptive Statistics for the Totals and Historical Value-at-Risk Actual Variables
for Before And After 2005

The Table 4.3 columns present the relative percentage changes of averaged results for *before* 2005 and after 2005, calculated using equation (4.2). The Table 4.3 columns and panels present the following:

The Total Variable column presents the following panels for the samples and the control group respectively: the BM panel presents the book-to-market percentage changes; the Market Price panel presents the market price percentage changes; the Total Equity panel presents the total equity percentage changes; the Total Assets panel presents the total assets percentage changes; the Total Liabilities panel presents the total liabilities percentage changes; the Net Income panel presents the net income percentage changes; and, the VaR panel presents the Historical Value-at-Risk Actual percentage changes.

The Sample and Control Group column presents the samples and the control group result panels for the total variables presented in the Total Variable column.

The Percentage Change of Averages (%) column presents the following columns: the Mean column presents the percentage changes for the mean of the variables in the Total Variable column; the Standard Deviation of Sample Means column presents the percentage changes of the standard deviation of means for the variables presented in the Total Variable column; the 1992 to 2004 column represents the year range applied to calculate the lower bound average in equation (4.2); the 2005 to 2005, 2005 to 2006, 2005 to 2007 and 2005 to 2008 columns represent the year ranges applied to calculate the upper bound averages in equation (4.2).

Total	Sample and Control		М	lean		Standar	Standard Deviation of Sample Means							
Variable	Group	1992 to 2004												
		2005 to 2005	2005 to 2006	2005 to 2007	2005 to 2008	2005 to 2005	2005 to 2006	2005 to 2007	2005 to 2008					
ВМ	Primary and Secondary Control Primary Secondary	52.80 -3.83 4.62 68.26	51.63 -3.83 4.62 67.11	63.38 0.46 26.88 72.87	117.45 32.23 110.36 110.90	-74.23 8.33 -11.56 -75.52	-73.45 -11.98 -7.14 -74.16	-73.45 67.01 56.24 -74.62	-67.40 394.27 205.10 -71.44					
Market Price	Primary and Secondary Control Primary Secondary	64.08 62.03 62.80 64.94	77.08 72.11 72.75 79.34	74.06 70.85 66.80 77.65	42.47 41.81 23.32 51.41	53.70 71.91 102.15 46.41	66.70 96.51 113.83 59.83	86.59 93.75 159.73 73.38	85.34 77.27 155.08 70.69					
Total Equity	Primary and Secondary Control Primary Secondary	113.43 28.06 121.37 42.76	123.60 29.81 131.50 53.22	146.50 26.68 156.68 55.74	159.11 5.31 171.81 45.92	126.25 23.54 130.09 26.58	133.24 25.39 132.91 43.33	153.00 24.87 151.35 47.58	167.74 14.77 160.64 40.78					
Total Assets	Primary and Secondary Control Primary Secondary	176.34 23.34 178.77 38.85	189.66 24.96 192.08 52.76	259.39 22.92 262.79 66.50	355.56 4.23 360.71 64.14	176.99 18.46 173.59 -27.37	189.87 21.71 185.62 -10.14	251.76 22.86 269.08 8.88	331.20 14.02 358.48 18.57					
Total Liabilities	Primary and Secondary Control Primary Secondary	180.33 0.74 182.13 36.92	193.73 1.67 195.50 52.62	264.95 5.69 267.38 71.13	367.36 1.33 371.06 72.43	181.19 -8.18 180.34 -17.74	194.20 -3.00 192.69 -0.16	257.01 11.32 277.91 18.85	340.16 11.17 372.48 26.97					
Net Income	Primary and Secondary Control Primary Secondary	50.12 -32.68 54.45 -46.82	53.30 -28.92 56.32 -14.23	60.92 -28.11 63.02 13.88	-162.86 -28.92 -168.08 -45.80	50.17 -31.46 44.48 -18.40	45.23 -30.27 28.23 -17.81	48.95 -27.38 30.80 -10.36	183.82 -27.44 255.36 -20.89					
VaR	Primary and Secondary Control Primary Secondary	106.11 133.85 93.64 112.15	121.39 131.03 103.18 130.39	69.44 102.82 23.70 87.85	-120.00 -12.82 -246.53 -67.13	-65.49 -51.57 -81.87 -57.69	-54.71 -20.44 -78.85 -47.12	-42.48 -10.06 -51.65 -39.36	13.24 -1.42 34.48 10.00					

Table 4.3 (Continued)

Percentage Change of Averages (%)

Table notes:

Percentage changes calculated using equation (4.2).

Summary percentage change results are produced from the detailed results presented in Table R.4.

Analysis

Examination of Table 4.3 shows that the Primary and Secondary sample exhibits materially significant increases after 2005 for the variables presented. It is worth noting that significant increases are registered for the mean and standard deviation of sample means of the accounting total variables: total equity, total assets, total liabilities and net income, and the market price and Value-at-Risk variables. Closer examination of the

market price variable reveals that it increased from 2005 to 2006 and then decreased in 2007 and 2008. Analysis of the standard deviation of market price sample means reveals that the level of *sample market price volatility* increased from 2005 to 2008. It is noted that the net income increases until 2007 and then exhibits a significant decrease in 2008. This table shows that these significant Primary and Secondary sample changes are mainly due to the Primary sample.

Examination of the book-to-market ratio reveals an interesting observation. It is only during 2008 that the Primary sample banks register a significant increase. The Secondary sample, however, reports increases from 2005 continuing to 2008. This finding suggests that for the banks the book-to-market measurement approach showed a delay in reacting to signs of financial distress risk.

Examination of the Control group results shows less materially significant changes and decreasing trends after 2005.

Generally, the Primary and Secondary sample changes exhibited in Table 4.3 provide strong evidence to suggest that after 2005 the accounting total variables increased while the market price variable decreased.

4.2.6 Null Hypotheses Test using Relative Delta Before to/and After 2005 – Examination of Material Significance

Table 4.4 presents percentage changes for the relative delta variable. Table 4.4 Panel A presents the summary samples and control group percentage change results. The results presented in Panel A are produced from the results in Table 4.13. Table 4.4 Panel B presents summary percentage change results at the bank level. The results presented in Panel B are produced from the results in Table 4.24.

Table 4.4Summary Results: Relative Percentage Change of Averages for the
Relative Delta Variable for Before To/And After 2005

The Table 4.4 Panel A columns present the relative percentage changes of averaged results for *before 2005 to after 2005* calculated using equation (4.1). The Table 4.4 Panel B columns represent the relative percentage changes of averaged results for *before 2005 and after 2005*, calculated using equation (4.2). The Table 4.4 Panel A columns and sub-panels present the following:

The Relative Delta column presents the following sub-panels for the samples and the control group respectively: the E-M sub-panel presents the total equity to market price relative delta percentage changes; the A-M sub-panel presents the total assets to market price relative delta percentage changes; the L-M sub-panel presents the total liabilities to market price relative delta percentage changes; and, the I-M sub-panel presents the net income to market price relative delta delta percentage changes.

The Sample and Control Group column presents the samples and the control group result subpanels for the relative deltas presented in the Relative Delta column.

The Percentage Change of Averages (%) column presents the following columns: the 1994 to 2004 column represents the year range applied to calculate the lower bound average in equation (4.1); the 1994 to 2005, 1994 to 2006, 1994 to 2007 and 1994 to 2008 columns represent the year ranges applied to calculate the upper bound averages in equation (4.1).

The Table 4.4 Panel B columns present, for each of the banks in the Primary sample, the same information as the Table 4.4 Panel A columns, other than in Panel B the Sample and Control Group column is omitted and the 2005 to 2005, 2005 to 2006, 2005 to 2007 and 2005 to 2008 columns represent the year ranges applied to calculate the upper bound averages in equation (4.2).

Percentage Changes for Before To After 2005											
		Percentage Change of Averages (%) 1994 to 2004									
Relative Delta	Sample and Control Group										
		1994 to 2005	1994 to 2006	1994 to 2007	1994 to 2008						
	Primary and Secondary	-1.90	-11.88	-3.48	43.24						
БΜ	Control	-17.86	-33.33	-46.94	-37.78						
E-IVI	Primary	-1.47	-8.58	7.39	70.90						
	Secondary	-1.94	-14.27	-11.09	25.15						
	Primary and Secondary	70.45	28.67	159.09	567.27						
A M	Control	-181.64	-263.94	-350.45	-301.67						
A-IVI	Primary	67.91	47.62	148.69	466.11						
	Secondary	144.44	-90.60	367.06	1,786.30						
	Primary and Secondary	77.67	11.25	133.29	456.79						
тм	Control	-50.90	-53.26	-86.88	-50.15						
L-IVI	Primary	73.84	52.25	157.84	490.94						
	Secondary	92.93	-53.76	108.31	451.71						
	Primary and Secondary	243.42	278.56	295.99	272.06						
T M	Control	-19.58	-0.59	8.58	169.34						
I-IVI	Primary	4.12	-4.85	-0.63	-30.12						
	Secondary	182.37	211.01	223.12	212.25						

 Table 4.4 (Continued)

PANEL A

Panel notes:

Percentage changes calculated using equation (4.1). Summary percentage change results are produced from the detailed results presented in Table 4.13.

Table 4.4 (Continued)

PANEL B Percentage Changes for the UK Banks for Before And After 2005

Percentage Change of Averages (%)

Primary Sample Banks Registered with the LSE's UK Banking Sector																										
HSBC Holdings PLC					Barclays PLC			The Royal Bank of Scotland PLC				Lloyds Banking Group PLC				Standard Chartered PLC										
1993 to 2004			1993 to 2004				1993 to 2004			1999 to 2004				1993 to 2004												
2005 to 2005	2005 to 2006	2005 to 2007	2005 to 2008	2005 to 2005	2005 to 2006	2005 to 2007	2005 to 2008	2005 to 2005	2005 to 2006	2005 to 2007	2005 to 2008	2005 to 2005	2005 to 2006	2005 to 2007	2005 to 2008	2005 to 2005	2005 to 2006	2005 to 2007	2005 to 2008							
2.6	-3.1	8.3	16.0	-122.2	-228.2	1,083.3	4,511.1	-5.8	-12.9	-4.9	27.8	-15.2	-30.7	-19.1	40.7	106.8	129.2	114.6	449.8							
104.9	130.5	257.6	501.8	311.4	240.7	514.5	1,255.5	17.1	7.1	90.3	239.1	-8.9	-23.9	-19.1	56.0	80.8	65.4	64.0	362.9							
135.4	168.9	323.5	624.8	322.5	249.3	528.4	1,285.9	18.6	8.2	92.4	243.6	-8.7	-23.6	-19.0	56.2	77.9	62.0	61.1	339.7							
278.2	146.7	244.5	-170.6	-8.6	-7.2	-8.5	37.8	-0.4	-8.0	-2.8	-110.6	-17.1	-24.8	-22.5	-15.2	264.6	128.6	113.4	770.7							
	2005 to 2005 2.6 104.9 135.4 278.2	HSBC Ho 1993 to 2005 2005 to 2005 2006 2.6 -3.1 104.9 130.5 135.4 168.9 278.2 146.7	HSBC Holdings PL 1993 to 2004 2005 2005 2005 to to to 2007 2.6 -3.1 8.3 104.9 130.5 257.6 135.4 168.9 323.5 278.2 146.7 244.5	HSBC Holdings PLC 1993 to 2004 2005 2005 2005 2005 2005 2006 2007 2008 2.6 -3.1 8.3 16.0 104.9 130.5 257.6 501.8 135.4 168.9 323.5 624.8 278.2 146.7 244.5 -170.6	HSBC Holdings PLC 1993 to 2004 2005 2005 2005 2005 2005 2005 2006 2007 2008 2005 2.6 -3.1 8.3 16.0 -122.2 104.9 130.5 257.6 501.8 311.4 135.4 168.9 323.5 624.8 322.5 278.2 146.7 244.5 -170.6 -8.6	HSBC Holdings PLC Barclay 1993 to 2004 1993 to 2005 2005 2005 2005 to to to to to 2005 2006 2007 2008 2005 2006 2.6 -3.1 8.3 16.0 -122.2 -228.2 104.9 130.5 257.6 501.8 311.4 240.7 135.4 168.9 323.5 624.8 322.5 249.3 278.2 146.7 244.5 -170.6 -8.6 -7.2	Primary 5 HSBC Holdings PLC Barclays PLC 1993 to 2004 1993 to 2004 1993 to 2004 2005 2005 2005 2005 2005 2005 2006 2007 2008 2005 2006 2007 2.6 -3.1 8.3 16.0 -122.2 -228.2 1,083.3 104.9 130.5 257.6 501.8 311.4 240.7 514.5 135.4 168.9 323.5 624.8 322.5 249.3 528.4 278.2 146.7 244.5 -170.6 -8.6 -7.2 -8.5	Primary Sample Ba HSBC Holdings PLC Barclays PLC 1993 to 2004 1993 to 2004 1993 to 2004 2005 2006 2007 2008 2005 2006 2007 2008 2005 2006 2007 2008 2005 2006 2007 2008 2005 2006 2007 2008 2005 2006 2007 2008 2005 1,083.3 4,511.1 104.9 130.5 257.6 501.8 311.4 240.7 514.5 1,255.5 135.4 168.9 323.5 624.8 322.5 249.3 528.4 1,285.9 278.2 146.7 244.5 -170.6 -8.6 -7.2 -8.5	Primary Sample Banks Regist HSBC Holdings PLC Barclays PLC The Ro 1993 to 2004 1993 to 2004 1993 to 2004 2005 <	Primary Sample Banks Registered with HSBC Holdings PLC Barclays PLC The Royal Bank 1993 to 2004 1993 to 2004 1993 to 2004 1993 to 2005 2005	Primary Sample Banks Registered with the LS HSBC Holdings PLC Barclays PLC The Royal Bank of Scotla 1993 to 2004 2005 2006 2007 2006 2007 2006 2007 2006 2007 2006 2007 2006 2007 2006 2007 2006 2007 2006 2007 2006 2007 2006 2007 2006 2007 2006 2007	Primary Sample Banks Registered with the LSE's UK Bank HSBC Holdings PLC Barclays PLC The Royal Bank of Scotland PLC 1993 to 2004 2005 2006 2007 2008 2005 2006 2007 2008 2005 2006 2007 2008 2005 2005 2005 2006 2007 2008 2005 2006 2007 2008 2017 208 2017 208 2017 208 2015 2017 2018 2014	Primary Sample Banks Registered with the LSE's UK Banking Sec HSBC Holdings PLC The Royal Bank of Scotland PLC Lloyds 1993 to 2004 1993 to 2004 Lloyds 2005 2006 2007 2008 2005 2005 2005 2006 2007 2008 2005 2005 2005 2005 2005 <th 2"2"2"2"2"2"2"2"2"2"2"2"2"2"2"2"2"2"<="" colspan="6" td=""><td>Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC Barclays PLC The Royal Bank of Scotland PLC Lloyds Banking 1993 to 2004 1993 to 2004 1993 to 2004 1993 to 2004 1999 to 2005<!--</td--><td>Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC Barclays PLC The Royal Bank of Scotland PLC Lloyds Banking Group 1993 to 2004 1993 to 2004 1993 to 2004 1993 to 2005 20</td><td>Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC The Royal Bank of Scotland PLC Lloyds Banking Group PLC 1993 to 2004 1993 to 2004 1993 to 2004 1993 to 2005 <t< td=""><td>Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC Barclays PLC The Royal Bank of Scotland PLC Lloyds Banking Group PLC St 1993 to 2004 1993 to 2004 Iloyds Banking Group PLC St 2005 <th colspan<="" td=""><td>Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC Barclays PLC The Royal Bank of Scotland PLC Lloyds Banking Group PLC Standard Cl 1993 to 2004 1993 to 2004 1993 to 2004 1993 to 2004 1999 to 2004 1999 to 2005 200</td><td>Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC Barclays PLC The Royal Bank of Scotland PLC Lloyds Banking Group PLC Standard Chartered P 1993 to 2004 1993 to 2004 1993 to 2004 Lloyds Banking Group PLC Standard Chartered P 2005</td></th></td></t<></td></td></th>	<td>Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC Barclays PLC The Royal Bank of Scotland PLC Lloyds Banking 1993 to 2004 1993 to 2004 1993 to 2004 1993 to 2004 1999 to 2005<!--</td--><td>Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC Barclays PLC The Royal Bank of Scotland PLC Lloyds Banking Group 1993 to 2004 1993 to 2004 1993 to 2004 1993 to 2005 20</td><td>Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC The Royal Bank of Scotland PLC Lloyds Banking Group PLC 1993 to 2004 1993 to 2004 1993 to 2004 1993 to 2005 <t< td=""><td>Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC Barclays PLC The Royal Bank of Scotland PLC Lloyds Banking Group PLC St 1993 to 2004 1993 to 2004 Iloyds Banking Group PLC St 2005 <th colspan<="" td=""><td>Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC Barclays PLC The Royal Bank of Scotland PLC Lloyds Banking Group PLC Standard Cl 1993 to 2004 1993 to 2004 1993 to 2004 1993 to 2004 1999 to 2004 1999 to 2005 200</td><td>Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC Barclays PLC The Royal Bank of Scotland PLC Lloyds Banking Group PLC Standard Chartered P 1993 to 2004 1993 to 2004 1993 to 2004 Lloyds Banking Group PLC Standard Chartered P 2005</td></th></td></t<></td></td>						Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC Barclays PLC The Royal Bank of Scotland PLC Lloyds Banking 1993 to 2004 1993 to 2004 1993 to 2004 1993 to 2004 1999 to 2005 </td <td>Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC Barclays PLC The Royal Bank of Scotland PLC Lloyds Banking Group 1993 to 2004 1993 to 2004 1993 to 2004 1993 to 2005 20</td> <td>Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC The Royal Bank of Scotland PLC Lloyds Banking Group PLC 1993 to 2004 1993 to 2004 1993 to 2004 1993 to 2005 <t< td=""><td>Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC Barclays PLC The Royal Bank of Scotland PLC Lloyds Banking Group PLC St 1993 to 2004 1993 to 2004 Iloyds Banking Group PLC St 2005 <th colspan<="" td=""><td>Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC Barclays PLC The Royal Bank of Scotland PLC Lloyds Banking Group PLC Standard Cl 1993 to 2004 1993 to 2004 1993 to 2004 1993 to 2004 1999 to 2004 1999 to 2005 200</td><td>Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC Barclays PLC The Royal Bank of Scotland PLC Lloyds Banking Group PLC Standard Chartered P 1993 to 2004 1993 to 2004 1993 to 2004 Lloyds Banking Group PLC Standard Chartered P 2005</td></th></td></t<></td>	Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC Barclays PLC The Royal Bank of Scotland PLC Lloyds Banking Group 1993 to 2004 1993 to 2004 1993 to 2004 1993 to 2005 20	Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC The Royal Bank of Scotland PLC Lloyds Banking Group PLC 1993 to 2004 1993 to 2004 1993 to 2004 1993 to 2005 2005 <t< td=""><td>Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC Barclays PLC The Royal Bank of Scotland PLC Lloyds Banking Group PLC St 1993 to 2004 1993 to 2004 Iloyds Banking Group PLC St 2005 <th colspan<="" td=""><td>Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC Barclays PLC The Royal Bank of Scotland PLC Lloyds Banking Group PLC Standard Cl 1993 to 2004 1993 to 2004 1993 to 2004 1993 to 2004 1999 to 2004 1999 to 2005 200</td><td>Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC Barclays PLC The Royal Bank of Scotland PLC Lloyds Banking Group PLC Standard Chartered P 1993 to 2004 1993 to 2004 1993 to 2004 Lloyds Banking Group PLC Standard Chartered P 2005</td></th></td></t<>	Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC Barclays PLC The Royal Bank of Scotland PLC Lloyds Banking Group PLC St 1993 to 2004 1993 to 2004 Iloyds Banking Group PLC St 2005 <th colspan<="" td=""><td>Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC Barclays PLC The Royal Bank of Scotland PLC Lloyds Banking Group PLC Standard Cl 1993 to 2004 1993 to 2004 1993 to 2004 1993 to 2004 1999 to 2004 1999 to 2005 200</td><td>Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC Barclays PLC The Royal Bank of Scotland PLC Lloyds Banking Group PLC Standard Chartered P 1993 to 2004 1993 to 2004 1993 to 2004 Lloyds Banking Group PLC Standard Chartered P 2005</td></th>	<td>Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC Barclays PLC The Royal Bank of Scotland PLC Lloyds Banking Group PLC Standard Cl 1993 to 2004 1993 to 2004 1993 to 2004 1993 to 2004 1999 to 2004 1999 to 2005 200</td> <td>Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC Barclays PLC The Royal Bank of Scotland PLC Lloyds Banking Group PLC Standard Chartered P 1993 to 2004 1993 to 2004 1993 to 2004 Lloyds Banking Group PLC Standard Chartered P 2005</td>	Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC Barclays PLC The Royal Bank of Scotland PLC Lloyds Banking Group PLC Standard Cl 1993 to 2004 1993 to 2004 1993 to 2004 1993 to 2004 1999 to 2004 1999 to 2005 200	Primary Sample Banks Registered with the LSE's UK Banking Sector HSBC Holdings PLC Barclays PLC The Royal Bank of Scotland PLC Lloyds Banking Group PLC Standard Chartered P 1993 to 2004 1993 to 2004 1993 to 2004 Lloyds Banking Group PLC Standard Chartered P 2005

Panel notes:

Percentage changes calculated using equation (4.2). Summary percentage change results are produced from the detailed results presented in Table 4.24.
Analysis

The Primary sample results in Table 4.4 Panel A and the bank level results in Panel B presents direct evidence for the levels of *material significance* that this study examines to strengthen or weaken the detailed null hypotheses specified in Section 3.3.2.

In Table 4.4 Panel A the results for the following tests are presented. Sub panel E-M tests the detailed null hypothesis in equation (3.23) by testing equation (E.1) that represents the difference between the change in total equity and the change in market price. Sub-panel A-M tests the detailed null hypothesis in equation (3.24) by testing equation (E.3) that represents the difference between the change in total assets and the change in market price. Sub-panel L-M tests the detailed null hypothesis in equation (3.25) by testing equation (E.5) that represents the difference between the change in total liabilities and the change in market price. Sub-panel I-M tests the detailed null hypothesis the detailed null hypothesis in equation (3.26) by testing equation (E.7) that represents the difference between the change in market price.

Table 4.4 Panel A shows that for the Primary sample and the Secondary sample the average accounting to market price relative deltas exhibit materially significant increases after 2005. With the exception of the Primary sample net income to market price relative delta (I-M), registering a marginal 4.12% increase in 2005 before exhibiting a decrease. Furthermore, it is noted that the Primary sample exhibits a materially significant increase in the total equity to market price relative delta (E-M) only in 2008. Examining the Control group results, besides the net income to market price relative delta that shows a significant increase in 2008, the relative deltas exhibit significant decreases after 2005.

Analysing Table 4.4 Panel B reveals that, in general, each bank exhibited increases and some decreases in relative delta levels to varying degrees after 2005.

In this study, the measure of the relative delta is also referred to as the *level of difference*, or more accurately, the measure of the *difference component* (see Sections 3.2.3, 3.2.5 and Appendix D). From the examination of the relative delta levels exhibited in Table 4.4 Panel A and Panel B, it is evident that the banks and the other

Primary and Secondary sample firms experienced materially significant *levels of difference* after 2005. Given the relationship between the *difference component*, relative delta and accounting quality detailed in Sections 2.9.2 and 2.9.3, this finding indicates that the banks experienced a materially significant deterioration in accounting quality after 2005.

4.2.7 Null Hypotheses Test using Regression Analysis for Relative Delta and Value-at-Risk Before to/and After 2005 – Examination of Statistical Significance

The results presented in Sections 4.2.2 to 4.2.6, other than Section 4.2.4, tests the null hypothesis specified in Section 3.3.1 by examining levels of *material significance*. Section 4.2.4 examines the null hypothesis by analysing levels of *statistical significance*. To further and more directly test the null hypothesis using *statistical significance*, the regression specified in Section 3.9.3, equation (3.40), that relates to the set of regressions specified in Appendix G, is applied. The set of regressions specified in equations (G.1), (G.2), (G.3) and (G.4) are aimed at testing the respective detailed null hypotheses specified in the relationships (3.23), (3.24), (3.25) and (3.26) (see Section 3.9.3 and Appendix O.13. In these respective sections, equation (3.40) specifies the time series regression and equation (O.28) specifies the cross-sectional regression.

The regression tests specified in Appendix G are applied to determine if the presence of statistical significance reported before 2005 remain after 2005. These regressions test for significance from the relationship between the *difference component* (see Section 3.2.5), quantified using the relative delta measure, and Value-at-Risk quantified using the Historical Value-at-Risk Actual measure.

Table 4.5 presents the regression results for the accounting total to market price relative delta variable and the Historical Value-at-Risk Actual variable. Table 4.5 Panel A presents the results for the samples and the control group. These results are reproduced from Table 4.14 to Table 4.17. Table 4.5 Panel B presents the results for the same regressions applied at the bank level. These results are reproduced from Table V.5 to Table V.9.

Table 4.5Summary Results: Regression Results for the Relative DeltaVariable and Historical Value-at-Risk Actual Variable for Before To/And
After 2005

The Table 4.5 Panel A columns and sub-panels present the following:

The Regression column presents the following sub-panels for the samples and the control group: the E-M and V sub-panel presents the total equity to market price relative delta and the Historical Value-at-Risk Actual regression results; the A-M and V sub-panel presents the total assets to market price relative delta and the Historical Value-at-Risk Actual regression results; the L-M and V sub-panel presents the total liabilities to market price relative delta and the Historical Value-at-Risk Actual regression results; and, the I-M and V sub-panel presents the net income to market price relative delta and the Historical Value-at-Risk Actual regression results; and, the I-M and V sub-panel presents the net income to market price relative delta and the Historical Value-at-Risk Actual regression results.

The Sample and Control Group column represents the samples and the control group result sub-panels for the regressions presented in the Regression column.

The Regression Statistics column presents the following columns: the *Slope* column presents the regression slope; the Intercept column presents the regression intercept; the R^2 column presents the Coefficient of Determination for the regression; the *1994 to* column represents the time series regression start year; and, the 2008, 2004, and 2007 columns represent the time series regression end year.

The Table 4.5 Panel B columns present, for each of the banks in the Primary sample, the same information as the Panel A columns, other than in Panel B the Sample and Control Group column is omitted and the *Slope*, *Intercept* and R^2 regression parameters in the Statistics column are presented in sub-panels.

	PANEL A									
			Results 1	for Before T	o After 2005					
		Regression Statistics								
Regression	Sample and Control		Slope			Intercept			R^2	
Regression	Group	1994 to:		1994 to:			1994 to:			
		2008	2004	2007	2008	2004	2007	2008	2004	2007
E-M and V	Primary and Secondary Control Primary Secondary	-0.703* -1.208 -0.519** -0.59	-0.252 -0.009 -0.204 -0.314	-0.32 -0.91 -0.224 -0.372	-0.084 -0.125* -0.103 -0.104	-0.141 -0.176* -0.138* -0.142	-0.104 -0.114* -0.115* -0.103	0.405 0.166 0.548 0.548	0.047 8.2E-06 0.165 0.165	0.068 0.115 0.179 0.179
A-M and V	Primary and Secondary Control Primary Secondary	-0.627* -0.907 -0.575** -0.577	-0.342 0.219 -0.375 -0.372	-0.266 -0.698 -0.256 -0.356	-0.124* -0.167** -0.11* -0.137*	-0.159* -0.178** -0.151* -0.164*	-0.123* -0.145** -0.124* -0.123*	0.396 0.106 0.662 0.662	0.095 0.01 0.165 0.165	0.056 0.079 0.104 0.104
L-M and V	Primary and Secondary Control Primary Secondary	-0.536* -0.275 -0.569** -0.463	-0.298 0.042 -0.371 -0.319	-0.239 -0.15 -0.248 -0.297	-0.126* -0.148** -0.111* -0.135*	-0.158* -0.179** -0.151* -0.162*	-0.122* -0.133* -0.124* -0.121*	0.381 0.119 0.659 0.659	0.11 0.161 0.161	0.071 0.04 0.099 0.099
I-M and V	Primary and Secondary Control Primary Secondary	0.076 -0.111 0.237 0.053	0.119* 0.012 0.009 0.089*	0.066* 0.015 -0.003 0.05*	-0.191** -0.151** -0.233** -0.174**	-0.143* -0.176** -0.163* -0.136*	-0.151** -0.133* -0.142* -0.145**	0.245 0.122 0.129 0.129	0.41 0.001 0.001	0.341 0.001

Table 4.5 (Continued)

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels. Results are re-produced from the detailed results presented in Table 4.14 to Table 4.17.

Table 4.5 ((Continued)
1 4010 100	Commutation,

PANEL B Results for UK Banks for Before To After 2005

		Primary Sample Banks Registered with the LSE's UK Banking Sector														
Regress		HSB	C Holdings	s PLC	Ba	rclays PL	С	The Roya	l Bank of S	cotland PLC	Lloyds B	anking Gro	oup PLC	Standa	ard Charter	ed PLC
ics	sion		1994 to			1994 to			1994 to			1999 to			1994 to	
		2008	2004	2007	2008	2004	2007	2008	2004	2007	2008	2004	2007	2008	2004	2007
	E-M & V	-0.35	-0.355	-0.345	-0.594**	-0.396	-0.415*	-0.265*	-0.118*	-0.125*	-1.042**	-1.181	-1.174	-0.383	-0.145	-0.035
<u>C1</u>	A-M & V	-0.29	-0.491	-0.354	-0.545**	-0.494	-0.293	-0.694**	-0.365*	-0.345**	-0.675**	-0.213	-0.473	-0.348*	-0.086	-0.024
Slope	L-M & V	-0.282	-0.5	-0.349	-0.541**	-0.487	-0.287	-0.691**	-0.363*	-0.343**	-0.665**	-0.2	-0.449	-0.343	-0.081	-0.019
	I-M & V	-0.011	-0.15	-0.074	-0.159	-0.023	-0.024	0.266**	0.033	0.021	-0.285	-0.216	-0.244*	-0.133	0.11	0.112
	E-M & V	-0.11	-0.137	-0.108	-0.118*	-0.133	-0.115*	-0.101	-0.035	-0.054	-0.118	-0.167	-0.104	-0.141*	-0.185*	-0.141*
Intercent	A-M & V	-0.124*	-0.166*	-0.126*	-0.108	-0.142	-0.112	-0.005	-0.044	-0.036	-0.162	-0.318	-0.186	-0.159*	-0.19*	-0.142*
тистеері	L-M & V	-0.125*	-0.167*	-0.127*	-0.108	-0.141	-0.112	-0.005	-0.044	-0.037	-0.163	-0.321	-0.19	-0.16*	-0.19*	-0.142*
	I-M & V	-0.152*	-0.171*	-0.142*	-0.154	-0.13	-0.125	-0.214*	-0.089	-0.105	-0.281	-0.292*	-0.199*	-0.17*	-0.189*	-0.145*
			1 1 1			÷									:	
	E-M & V	0.222	0.241	0.216	0.72	0.255	0.321	0.353	0.481	0.437	0.755	0.27	0.378	0.25	0.025	0.001
\mathbf{p}^2	A-M & V	0.154	0.286	0.162	0.633	0.227	0.151	0.797	0.383	0.495	0.648	0.033	0.125	0.265	0.01	0.001
К	L-M & V	0.148	0.29	0.157	0.629	0.221	0.146	0.796	0.371	0.484	0.644	0.031	0.119	0.263	0.009	-
	I-M & V	0.001	0.046	0.015	0.16	0.009	0.008	0.673	0.02	0.007	0.261	0.589	0.486	0.064	0.049	0.045

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels. Results are re-produced from the detailed results presented in Table V.5 to Table V.9.

Analysis

The Primary sample results in Table 4.5 Panel A and the bank level results in Panel B presents evidence for the levels of *statistical significance* that this study examines to strengthen or weaken the detailed null hypotheses specified in Section 3.3.2.

In Table 4.5, the results for the following tests are presented. Sub-panel E-M and V tests the detailed null hypothesis specified in equation (3.23) by testing equation (G.1) that represents the relationship between the total equity to market price relative delta and Value-at-Risk. Sub-panel A-M and V tests the detailed null hypothesis specified in equation (3.24) by testing equation (G.2) that represents the relationship between the total assets to market price relative delta and Value-at-Risk. Sub-panel L-M and V tests the detailed null hypothesis specified in equation (3.25) by testing equation (G.3) that represents the relationship between the total liabilities to market price relative delta and Value-at-Risk. Sub-panel I-M and V tests the detailed null hypothesis specified in equation (3.26) by testing equation (G.4) that represents the relationship between the net income to market price relative delta and Value-at-Risk.

Examination of the Table 4.5 Panel A results shows statistically significant relationships in 2008 between the Primary sample accounting to market price relative deltas: E-M, A-M and L-M, and the Historical Value-at-Risk Actual variable. The net income to market price relative delta, I-M, and the Historical Value-at-Risk Actual variable relationship is the only exception and does not indicate any significance for before or after 2005.

The results from Table 4.3 (and Table 4.19, V_M column) reveal that the Primary sample Value-at-Risk levels exhibited materially significant increases after 2005. Combining this result with the Table 4.5 Panel A results indicate that for the identified relative deltas the *difference component* increased with statistical significance after 2005. Examining the Table 4.5 Panel A Secondary sample results, only the net income to market price relative delta, I-M, exhibits a statistically significant relationship to Value-at-Risk in 2007 and before 2005. Contrary to the Primary sample and Secondary sample results, the Control group does not exhibit any statistical significance either before or after 2005.

To analyse the bank level results in Table 4.5 Panel B, by applying the sample level analysis approach, first it is necessary to examine the level of Value-at-Risk experienced by the banks. Examining the Primary sample Value-at-Risk levels in Table 4.3 (and Table 4.24 Panel A to Panel E, V_M column) shows that the banks experienced an increase in Value-at-Risk after 2005. Given this increase, the analysis of Table 4.5 Panel B shows that after 2005 60% of the banks (3 banks from 5) exhibit statistically significant increases in the total equity and market price relative delta difference component. These increases translate to 60% of banks experiencing financial distress risk. Further analysis reveals that 80% of the banks (4 banks from 5) exhibit statistically significant increases in the total assets and market price difference component. In addition, 60% of the banks (3 banks from 5) exhibit statistically significant increases in the total liabilities and market price *difference component*. This table also reveals that 20% (1 bank from 5) exhibits a statistically significant increase in the net income and market price *difference component*. Generally, Table 4.5 Panel B exhibits statistically significant increases in the magnitude of the difference component after 2005 than before.

The analysis of the results from Table 4.5 show that the banks experienced statistically significant levels of difference between the changes in accounting variables and the change in the market price variable after 2005. Given the relationship between the *difference component*, relative delta and accounting quality detailed in Sections 2.9.2 and 2.9.3, these results indicate that the banks experienced a *statistically* significant deterioration in accounting quality after 2005. The results from this section support the findings from Section 4.2.6 that also indicate a *materially* significant deterioration in accounting quality after 2005.

The following sections present analysis and detailed results for tests applied to examine this study's null hypothesis. The tests applied and the results presented and analysed are based on the accounting totals and market price change variables, and the Historical Value-at-Risk Actual variable.

4.3 Regressions for Change in Accounting Totals and Market Price Return

The analysis and detailed results for the change in accounting totals and the market price return time series regressions are presented in this section. Appendix U presents

analysis and detailed results for these same variables using cross-sectional regressions. The regressions are in the time range 1994 to 2008. The accounting total variables applied in these regressions are total equity, total equity per share, total assets, total liabilities and net income.

4.3.1 Regressions for Change in Total Equity and Market Price Return

The regression analysis for the change in total equity and market price return is presented for the time range 1994 to 2008 in Table 4.6 (and Table U.1). In the regressions, the market price return variable is tested as the dependent variable, and the change in the total equity variable is tested as the independent variable.

Table 4.6Time Series Regression Analysis for Change in Total Equity and
Market Price Return

Time Series Regression

Table 4.6 Panel A to Panel D show the results for the time series regressions that test the change in total equity and market price return by applying the regression specified in equation (3.32). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.

Time Series Regression Table Description

The Table 4.6 columns represent the following:

The Total Period column represents the time series regression results for the year range 1994 to 2008, with a single year represented from 1st January to 31st December, with the 31st December selected as the variable record date. The Sub-Periods column represents the time series regression results for the year range 1994 to 2004, and for 1994 to 2007, with a single year represented from 1st January to 31st December, and a 31st December variable record date.

The Measure column represents the time series regression results for the total and sub-time periods and is represented by the following data panels:

The *Regression* panel presents the coefficients and related statistics for the regression specified in equation (3.32), and follows the model:

$$dM_{N,t} = a + b \ dE_{N,t} + e_{N,t}$$

Where: $dM_{N,t}$ is the log Market Price return $(M_{t-1,t})$ for N firms at time t, $dE_{N,t}$ is the change in Total Equity $(E_{t-1,t})$ for N firms at time t, and $e_{N,t}$ is the regression error term. The *Lower* and *Upper* are the corresponding coefficient levels predicted at a 99% confidence level.

The *Correlation* panel presents the correlation statistics for the relationship between the dependent and independent variables. The *Descriptive Statistics* panel presents for the variables tested, the average sample means calculated using equation (M.6); the standard deviation, *SD*, of sample means for the time series using equation (M.7); and, the standard error of means, *SE*, for the time series. The *Observations* panel presents: *Firms (n)* (term *N* in the regression) that represents the average number of sample firms in the time series regression (for the Primary and Secondary sample, for the time period 1994 to 2004, the number of firms are rounded down); *Years (T)* represents the number of time series years; *Total (n × T)* represents the number of averaged pooled time series regression observations.

M		Total Period	Sub-F	Periods
Measure		1994-2008	1994-2004	1994-2007
Slope	b	1.192*	0.859*	0.888*
	t (b)	2.656	2.344	2.787
	p (b)	2.0E-02	4.4E-02	1.6E-02
	SE(b)	0.449	0.366	0.319
	Lower	-0.160	-0.332	-0.085
	Upper	2.544	2.050	1.861
Intercept Regression	a t (a) p (a) SE(a)	-0.153 -1.708 1.1E-01 0.089	-0.062 -0.801 4.4E-01 0.078	-0.064 -0.976 3.5E-01 0.066
	Lower	-0.422	-0.314	-0.264
	Upper	0.116	0.190	0.136
Model	R^{2} $s(e)$ $df(e)$	0.352 0.212 13	0.379 0.161 9	0.393 0.146 12
Correlation	r	0.593**	0.616*	0.627**
	p(r)	9.9E-03	2.2E-02	8.2E-03
Descriptive Statistics	E	0.157	0.165	0.165
	SD (E)	0.127	0.139	0.127
	SE(E)	0.033	0.042	0.034
	M	0.035	0.080	0.083
	SD (M)	0.254	0.194	0.180
	SE(M)	0.066	0.059	0.048
Observations	Firms (n) ^a	16	15	16
	Years (T)	15	11	14
	Total ($n \times T$)	233	169	217

PANEL A Time Series Regression Analysis for the Primary and Secondary Sample Change in Total Equity $(E_{t-1, t})$ and Market Price Return $(M_{t-1, t})$

Table 4.6 (Continued)

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

^aFirms(n) is the average number of firms surveyed for each year, rounded to the nearest whole number. In addition, Firms(n) reflects missing data items presented in Table Z.1. These same criteria applies to the presentation of the variable Firms(n) for all the time series results presented in this study.

The sample consists of a maximum of 233 observations from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008.

See main table notes for the regression model applied.

	M		Total Period	Sub-F	Periods
	Measure		1994-2008	1994-2004	1994-2007
	Slope	h	0.973**	0.892**	0.918**
	love p c	<i>t</i> (<i>b</i>)	12.632	11.758	10.561
		p(b)	1.1E-08	9.2E-07	2.0E-07
		SE(b)	0.077	0.076	0.087
		Lower	0.741	0.645	0.652
		Upper	1.205	1.138	1.183
	Intercept	а	-0.025	-0.035	-0.016
Regression	1	t (a)	-1.529	-2.214	-0.941
		p (a)	1.5E-01	5.4E-02	3.7E-01
		SE(a)	0.016	0.016	0.017
		Lower	-0.074	-0.086	-0.069
		Upper	0.024	0.016	0.037
	Model	R^2	0.925	0.939	0.903
		s(e)	0.062	0.049	0.060
		df(e)	13	9	12
		r	0.962**	0 969**	0.95**
Correla	ation	p(r)	5.6E-09	4.6E-07	9.9E-08
		E	0.046	0.067	0.073
		$\overline{SD}(E)$	0.214	0.206	0.192
Deservinting	Statistics	SE(E)	0.055	0.062	0.051
Descriptive	siunsnes	М	0.020	0.025	0.051
		SD(M)	0.217	0.190	0.186
		SE(M)	0.056	0.057	0.050
		Firms (n)	12	12	12
Observa	tions	Years (T)	15	11	14
Observations					

Table 4.6 (Continued)PANEL B

Time Series Regression Analysis for the Control Group Change in Total Equity $(E_{t-I, t})$ and Market Price Return $(M_{t-I, t})$

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The control consists of a maximum of 180 observations from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008.

See main table notes for the regression model applied.

	Martin		Total Period	Sub-F	Periods
	Measure		1994-2008	1994-2004	1994-2007
	Slope	b	0.169	0.123	0.123
		l(0)	0.322 6.1E-01	0.703 5.0E-01	0.745 4 7E-01
		SE(b)	0.324	0.175	0.166
		Lower	-0.808	-0.445	-0.384
		Upper	1.146	0.692	0.631
	Intercept	а	-0.032	0.070	0.053
Regression	Ĩ	t (a)	-0.289	1.032	0.906
		p (a)	7.8E-01	3.3E-01	3.8E-01
		SE(a)	0.111	0.068	0.059
		Lower	-0.367	-0.151	-0.126
		Upper	0.303	0.291	0.232
	Model	R^2	0.021	0.052	0.044
		s(e)	0.351	0.189	0.180
		df(e)	13	9	12
Contract		r	0.143	0.229	0.210
Correl	ation	<i>p(r)</i>	3.1E-01	2.5E-01	2.4E-01
		E	0.199	0.211	0.202
		SD (E)	0.289	0.341	0.300
Descriptive	Statistics	SE(E)	0.075	0.103	0.080
Descriptive	σιαιιστίεσ	М	0.001	0.096	0.078
		SD(M)	0.342	0.184	0.177
		SE(M)	0.088	0.055	0.047
		Firms (n)	5	5	5
Observe	ations	Years (T)	15	11	14
Observations					-

PANEL C Time Series Regression Analysis for the Primary Sample Change in Total Equity $(E_{t-1,t})$

Table 4.6 (Continued)

Panel notes:

_

The sample consists of a maximum of 75 observations from the 5 banks in the LSE that adopted the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008. See main table notes for the regression model applied.

	Magnet		Total Period	Sub-F	Periods
	Measure		1994-2008	1994-2004	1994-2007
	Slope	Ь	1.156*	0.860	0.899*
	1	t (b)	2.753	1.737	2.282
		p (b)	1.6E-02	1.2E-01	4.2E-02
		SE(b)	0.420	0.495	0.394
		Lower	-0.109	-0.749	-0.305
		Upper	2.421	2.469	2.102
	Intercept	а	-0.108	-0.049	-0.047
Regression	1	t (a)	-1.421	-0.540	-0.634
		p (a)	1.8E-01	6.0E-01	5.4E-01
		SE(a)	0.076	0.091	0.074
		Lower	-0.336	-0.346	-0.272
		Upper	0.121	0.247	0.178
	Model	R^2	0.368	0.251	0.303
		s(e)	0.192	0.194	0.172
		df(e)	13	9	12
		r	0.607**	0.501	0.55*
Correl	ation	p(r)	8.2E-03	5.8E-02	2.1E-02
		E	0.136	0.141	0.146
		SD (E)	0.123	0.124	0.121
Descriptive	Statistics	SE(E)	0.032	0.037	0.032
Descriptive	σιαιιστίες	M	0.050	0.072	0.085
		SD(M)	0.233	0.213	0.197
		SE(M)	0.060	0.064	0.053
		Firms (n)	11	11	11
Observa	ations	Years (T)	15	11	14
Observations		T (1 (T))	165	101	154

Table 4.6 (Continued)

PANEL D

Time Series Regression Analysis for the Secondary Sample Change in Total Equity $(E_{t-1, t})$ and Market Price Return $(M_{t-1, t})$

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The sample consists of a maximum of 165 observations from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008.

See main table notes for the regression model applied.

Analysis

The Table 4.6 time series regression slopes show that the change in the total equity variable exhibits statistically significant explanatory power to the market price return variable for the Primary and Secondary sample and the Control group for the three time series periods: 1994 to 2004, 1994 to 2007 and 1994 to 2008.

The Table 4.6 Panel A time series regression slope for the Primary and Secondary sample shows that before 2005, a 1% variation in the change in the total equity variable related to less than a 1% change in the market price return variable. After 2005, the slope shows that for a 1% variation in the change in total equity, the market price return changed more than 1%. However, the slope for the Control group, Panel B, for before and after 2005 shows that for a 1% variation in the change in total equity, the market price return variable changed less than 1%.

The substantial difference between the *coefficients of determination* measures, R^2 , and the absence of significance for the *slope coefficient* measures, *b*, when examining the Primary sample results (Panel C) compared to the Primary and Secondary sample results (Panel A), suggest that the Primary sample banks experienced an effect that differed from the Secondary sample. Examining the standard deviations, the high level exhibited in Panel C from 1994 to 2004 for the change in the total equity variable, strongly suggests that it is this effect that has contributed to the Primary sample's absence of significance (see Panel C, *Descriptive Statistics* sub-panel *SD(E)*)⁴⁴.

The time series slopes for the Primary and Secondary sample suggests that the total equity change variable reacted to a lesser magnitude than the market price return variable after the 2005 accounting change compared to before. For the Control group the total equity change variable reacted with a similar magnitude to the market price return variable before and after 2005.

⁴⁴ For further information, at the yearly level, for the change in the total equity variable that exhibits the suggested increase in standard deviation, see the year 1993 and 2000 standard deviation panel records in APPENDIX S, Table S.3, column dE, and also the year 2000 record in APPENDIX U, Table U.1, column *SD*.

4.3.2 Regressions for Change in Total Equity per Share and Market Price Return

The regression analysis for the change in total equity per share and market price return is presented for the time range 1994 to 2008 in Table 4.7 (and Table U.2). In the regressions, the market price return variable is tested as the dependent variable, and the change in the total equity per share variable is tested as the independent variable.

Table 4.7Time Series Regression Analysis for Change in Total Equity per
Share and Market Price Return

Time Series Regression

Table 4.7 Panel A to Panel D show the results for the time series regressions that test the change in total equity per share and market price return by applying the regression specified in equation (3.32). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.

Time Series Regression Table Description

The Table 4.7 columns represent the following:

The Total Period column represents the time series regression results for the year range 1994 to 2008, with a single year represented from 1st January to 31st December, with the 31st December selected as the variable record date. The Sub-Periods column represents the time series regression results for the year range 1994 to 2004, and for 1994 to 2007, with a single year represented from 1st January to 31st December, and a 31st December variable record date.

The Measure column represents the time series regression results for the total and sub-time periods and is represented by the following data panels:

The *Regression* panel presents the coefficients and related statistics for the regression specified in equation (3.32), and follows the model:

$$dM_{N,t} = a + b d(E/S)_{N,t} + e_{N,t}$$

Where: $dM_{N,t}$ is the log Market Price return $(M_{t-1,t})$ for N firms at time t, $d(E/S)_{N,t}$ is the change in Total Equity per Share $(E_{t-1,t}/S_{t-1,t})$ for N firms at time t, and $e_{N,t}$ is the regression error term. The *Lower* and *Upper* are the corresponding coefficient levels predicted at a 99% confidence level.

The *Correlation* panel presents the correlation statistics for the relationship between the dependent and independent variables. The *Descriptive Statistics* panel presents for the variables tested, the average sample means calculated using equation (M.6); the standard deviation, *SD*, of sample means for the time series using equation (M.7); and, the standard error of means, *SE*, for the time series. The *Observations* panel presents: *Firms (n)* (term *N* in the regression) that represents the average number of sample firms in the time series regression (for the Primary and Secondary sample, for the time period 1994 to 2004, the number of firms are rounded down); *Years (T)* represents the number of time series years; *Total (n × T)* represents the number of averaged pooled time series regression observations.

M		Total Period	Sub-F	Periods
Measu	re	1994-2008	1994-2004	1994-2007
Slope	b	2.254**	1.967**	1.959**
	t (b)	7.611	4.118	4.575
	p (b)	3.8E-06	2.6E-03	6.4E-04
	SE(b)	0.296	0.478	0.428
	Lower	1.362	0.415	0.651
	Upper	3.146	3.520	3.267
Intercep Regression	ot a t (a) p (a) SE(a)	-0.195** -4.645 4.6E-04 0.042	-0.148 -2.240 5.2E-02 0.066	-0.154* -2.570 2.5E-02 0.060
	Lower	-0.321	-0.364	-0.337
	Upper	-0.069	0.067	0.029
Model	R ²	0.817	0.653	0.636
	s(e)	0.113	0.121	0.113
	df(e)	13	9	12
Correlation	r	0.904**	0.808**	0.797**
	p(r)	1.9E-06	1.3E-03	3.2E-04
Descriptive Statistic	E/S	0.102	0.116	0.121
	SD (E/S)	0.102	0.080	0.073
	SE(E/S)	0.026	0.024	0.020
	M	0.035	0.080	0.083
	SD (M)	0.254	0.194	0.180
	SE(M)	0.066	0.059	0.048
Observations	Firms (n)	16	15	16
	Years (T)	15	11	14
	Total ($n \times T$)	233	169	217

Time Series Regression Analysis for the Primary and Secondary Sample Change in Total Equity per Share $(E_{t-1, t}/S_{t-1, t})$ and Market Price Return $(M_{t-1, t})$

Table 4.7 (Continued)
PANEL A

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The sample consists of a maximum of 233 observations from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008.

See main table notes for the regression model applied.

N	*2	Total Period	Sub-F	Periods
Measu	re	1994-2008	1994-2004	1994-2007
Slope	b	0.972**	0.89**	0.926**
	t (b)	20.977	17.903	18.973
	p (b)	2.1E-11	2.4E-08	2.6E-10
	SE(b)	0.046	0.050	0.049
	Lower	0.833	0.729	0.777
	Upper	1.112	1.052	1.075
Interce Regression	$ \begin{array}{l} bt & a \\ t & (a) \\ p & (a) \\ SE(a) \end{array} $	-0.039** -3.840 2.0E-03	-0.036** -3.398 7.9E-03 0.011	-0.031* -3.044 1.0E-02
	Lower	-0.070	-0.070	-0.062
	Upper	-0.008	-0.002	1.1E-04
Model	R ²	0.971	0.973	0.968
	s(e)	0.038	0.033	0.035
	df(e)	13	9	12
Correlation	r	0.986**	0.986**	0.984**
	p(r)	1.0E-11	1.2E-08	1.3E-10
Descriptive Statistic	E/S	0.061	0.068	0.089
	SD (E/S)	0.220	0.210	0.197
	SE(E/S)	0.057	0.063	0.053
Descriptive Statistic	,	0.020	0.025	0.051
	SD (M)	0.217	0.190	0.186
	SE(M)	0.056	0.057	0.050
Observations	Firms (n)	12	12	12
	Years (T)	15	11	14
	Total ($n \times T$)	180	132	168

Table 4.7 (Continued)
PANEL B

Time Series Regression Analysis for the Control Group Change in Total Equity per Share $(E_{t-l,t}/S_{t-l,t})$ and Market Price Return $(M_{t-l,t})$

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The control consists of a maximum of 180 observations from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008.

See main table notes for the regression model applied.

	Маадияа		Total Period	Sub-F	Periods
	Measure		1994-2008	1994-2004	1994-2007
	Slope	b	1.945**	0.885	0.768
	I I I	t (b)	3.130	1.941	1.676
		p (b)	8.0E-03	8.4E-02	1.2E-01
		SE(b)	0.622	0.456	0.458
		Lower	0.073	-0.597	-0.632
		Upper	3.818	2.366	2.168
	Intercept	а	-0.203	-0.007	-0.015
Regression	1	t (a)	-2.134	-0.094	-0.214
		p (a)	5.2E-02	9.3E-01	8.3E-01
		SE(a)	0.095	0.072	0.071
		Lower	-0.490	-0.242	-0.232
		Upper	0.084	0.228	0.202
	Model	R^2	0.430	0.295	0.190
		s(e)	0.268	0.163	0.166
		df(e)	13	9	12
		Ľ	0.656**	0 543*	0.435
Correld	ation	p(r)	4.0E-03	4.2E-02	6.0E-02
		E/S	0 105	0 116	0 121
		SD(E/S)	0.115	0.113	0.100
Degeninting	Statistics	SE(E/S)	0.030	0.034	0.027
Descriptive	sialistics	М	0.001	0.096	0.078
		SD(M)	0.342	0.184	0.177
		SE(M)	0.088	0.055	0.047
		Firms (n)	5	5	5
Observa	tions	Years (T)	15	11	14
Observations		Total $(n \times T)$	75	55	70

Table 4.7 (Continued) PANEL C

Time Series Regression Analysis for the Primary Sample Change in Total Equity per

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The sample consists of a maximum of 75 observations from the 5 banks in the LSE that adopted the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008. See main table notes for the regression model applied.

λſ		Total Period	Sub-F	eriods
Measure	;	1994-2008	1994-2004	1994-2007
Slope	b	1.79**	1.85*	1.852**
1	t (b)	5.289	2.927	3.631
	$p(\dot{b})$	1.5E-04	1.7E-02	3.4E-03
	SE(b)	0.338	0.632	0.510
	Lower	0.771	-0.204	0.294
	Upper	2.810	3.905	3.410
Intercept	а	-0.128*	-0.140	-0.137
Regression	t(a)	-2.633	-1.604	-1.905
C	p (a)	2.1E-02	1.4E-01	8.1E-02
	SE(a)	0.049	0.087	0.072
	Lower	-0.275	-0.423	-0.356
	Upper	0.018	0.143	0.083
Model	R^2	0.683	0.488	0.524
	s(e)	0.136	0.161	0.142
	df(e)	13	9	12
Completion	r	0.826**	0.698**	0.724**
Correlation	p(r)	7.3E-05	8.4E-03	1.7E-03
	F/S	0.100	0.115	0.120
	SD(E/S)	0.108	0.080	0.077
D	SE(E/S)	0.028	0.024	0.021
Descriptive Statistics	М	0.050	0.072	0.085
	SD (M)	0.233	0.213	0.197
	SE(M)	0.060	0.064	0.053
	Firms (n)	11	11	11
Observations	Years (T)	15	11	14
	Total ($n \times T$)	165	121	154

Table 4.7 (Continued) PANEL D

Time Series Regression Analysis for the Secondary Sample Change in Total Equity per

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The sample consists of a maximum of 165 observations from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008.

See main table notes for the regression model applied.

Analysis

Table 4.7 exhibits similar results to Table 4.6.

The Table 4.7 time series regression slopes show that the change in the total equity per share variable exhibits statistically significant explanatory power to the market price return variable for the Primary and Secondary sample and the Control group for the three time series periods: 1994 to 2004, 1994 to 2007 and 1994 to 2008.

The Table 4.7 Panel A time series regression slope for the Primary and Secondary sample shows that before 2005, a 1% variation in the change in total equity per share variable related to less than a 2% change in the market price return variable. After 2005, the slope shows that for a 1% variation in the change in total equity per share, the market price return variable changed more than 2%. However, the slope for the Control group, Panel B, for before and after 2005 shows that for a 1% variation in the change less than 1%.

The time series slopes suggest that for the Primary and Secondary sample the total equity per share change variable reacted to a lesser magnitude than the market price return variable after the 2005 accounting change compared to before. For the Control group the total equity per share change variable reacted with a similar magnitude to the market price return variable before and after 2005.

4.3.3 Regressions for Change in Total Assets and Market Price Return

The regression analysis for the change in total assets and market price return is presented for the time range 1994 to 2008 in Table 4.8 (and Table U.3). In the regressions, the market price return variable is tested as the dependent variable, and the change in the total assets variable is tested as the independent variable.

Table 4.8Time Series Regression Analysis for Change in Total Assets and
Market Price Return

Time Series Regression

Table 4.8 Panel A to Panel D show the results for the time series regressions that test the change in total assets and market price return by applying the regression specified in equation (3.32). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.

Time Series Regression Table Description

The Table 4.8 columns represent the following:

The Total Period column represents the time series regression results for the year range 1994 to 2008, with a single year represented from 1st January to 31st December, with the 31st December selected as the variable record date. The Sub-Periods column represents the time series regression results for the year range 1994 to 2004, and for 1994 to 2007, with a single year represented from 1st January to 31st December, and a 31st December variable record date.

The Measure column represents the time series regression results for the total and sub-time periods and is represented by the following data panels:

The *Regression* panel presents the coefficients and related statistics for the regression specified in equation (3.32), and follows the model:

$$dM_{N,t} = a + b \, dA_{N,t} + e_{N,t}$$

Where: $dM_{N,t}$ is the log Market Price return $(M_{t-1,t})$ for N firms at time t, $dA_{N,t}$ is the log change in Total Assets $(A_{t-1,t})$ for N firms at time t, and $e_{N,t}$ is the regression error term. The *Lower* and *Upper* are the corresponding coefficient levels predicted at a 99% confidence level.

The *Correlation* panel presents the correlation statistics for the relationship between the dependent and independent variables. The *Descriptive Statistics* panel presents for the variables tested, the average sample means calculated using equation (M.6); the standard deviation, *SD*, of sample means for the time series using equation (M.7); and, the standard error of means, *SE*, for the time series. The *Observations* panel presents: *Firms (n)* (term *N* in the regression) that represents the average number of sample firms in the time series regression (for the Primary and Secondary sample, for the time period 1994 to 2004, the number of firms are rounded down); *Years (T)* represents the number of time series years; *Total (n × T)* represents the number of averaged pooled time series regression observations.

N/		Total Period	Sub-I	Periods
Meas	ire	1994-2008	1994-2004	1994-2007
Slope	b	1.565	2.151*	1.220
	t (b)	1.918	2.643	2.160
	p (b)	7.7E-02	2.7E-02	5.2E-02
	SE(b)	0.816	0.814	0.565
	Lower	-0.893	-0.494	-0.505
	Upper	4.024	4.797	2.945
Interce Regression	pt a t (a) p (a) SE(a)	-0.135 -1.260 2.3E-01 0.107	-0.115 -1.324 2.2E-01 0.087	-0.053 -0.699 5.0E-01 0.076
	Lower	-0.456	-0.398	-0.285
	Upper	0.187	0.168	0.179
Mode	l R ²	0.221	0.437	0.280
	s(e)	0.233	0.154	0.159
	df(e)	13	9	12
Correlation	r	0.47*	0.661*	0.529*
	p(r)	3.9E-02	1.3E-02	2.6E-02
Descriptive Statistic	$ \begin{array}{c} A \\ SD(A) \\ SE(A) \end{array} $	0.108 0.076 0.020	0.091 0.060 0.018	0.111 0.078 0.021
	M	0.035	0.080	0.083
	SD (M)	0.254	0.194	0.180
	SE(M)	0.066	0.059	0.048
Observations	Firms (n)	16	15	16
	Years (T)	15	11	14
	Total ($n \times T$)	233	169	217

Table 4.8 (Continued)	
PANEL A	

Time Series Regression Analysis for the Primary and Secondary Sample Change in Total Assets $(A_{t-I, t})$ and Market Price Return $(M_{t-I, t})$

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The sample consists of a maximum of 233 observations from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008.

See main table notes for the regression model applied.

	M		Total Period	Sub-F	Periods
	Measure		1994-2008	1994-2004	1994-2007
	Slope	b	1.128**	1.118**	1.145**
	I I I	t (b)	12.765	10.295	9.976
		p (b)	9.9E-09	2.8E-06	3.7E-07
		SE(b)	0.088	0.109	0.115
		Lower	0.862	0.765	0.795
		Upper	1.394	1.471	1.496
	Intercept	а	0.011	-0.009	0.009
Regression	I I I I I I I I I I I I I I I I I I I	t (a)	0.681	-0.545	0.530
-		p (a)	5.1E-01	6.0E-01	6.1E-01
		SE(a)	0.016	0.017	0.017
		Lower	-0.037	-0.065	-0.044
		Upper	0.058	0.046	0.063
Мос	Model	R^2	0.926	0.922	0.892
		s(e)	0.061	0.056	0.063
		df(e)	13	9	12
		14	0.062**	0.06**	0.045**
Correlat	tion	p(r)	5.0E-09	1.4E-06	1.8E-07
		A	0.008	0.030	0.037
		$\frac{1}{SD}(A)$	0.185	0.163	0.153
Descriptions	tatiation	SE(A)	0.048	0.049	0.041
Descriptive S	manstics	М	0.020	0.025	0.051
		SD(M)	0.217	0.190	0.186
		SE(M)	0.056	0.057	0.050
		Firms (n)	12	12	12
Observat	ions	Years (T)	15	11	14
		Total ($n \times T$)	180	132	168

Table 4.8 (Continued) PANEL B

Time Series Regression Analysis for the Control Group Change in Total Assets $(A_{t-1, t})$ and Market Price Return $(M_{t-1, t})$

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The control consists of a maximum of 180 observations from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008.

See main table notes for the regression model applied.

	Maaguma		Total Period	Sub-F	Periods
	weasure		1994-2008	1994-2004	1994-2007
	Slope	b t (b)	-0.907 -1.405	0.426 0.852	0.094 0.235
		p (b) SE(b)	1.8E-01 0.645	4.2E-01 0.500	8.2E-01 0.401
		Lower	-2.850	-1.198	-1.132
	I	Opper	0.152	2.049	0.064
Regression	Intercept	a t (a) p (a)	0.135 1.111 2.9E-01	0.509 6.2E-01	0.821 4.3E-01
		SE(a)	0.137	0.084	0.078
		Lower Upper	-0.261 0.566	-0.231 0.316	-0.174 0.302
	Model	R ² s(e) df(e)	0.132 0.331 13	0.075 0.187 9	0.005 0.183 12
Correl	ation	r p(r)	-0.363 9.2E-02	0.273 2.1E-01	0.068 4.1E-01
Descriptive	Statistics	A SD (A) SE(A)	0.167 0.137 0.035	0.125 0.118 0.036	0.151 0.127 0.034
Descriptive	5141151105	M SD (M) SE(M)	0.001 0.342 0.088	0.096 0.184 0.055	0.078 0.177 0.047
Observe	ations	Firms (n) Years (T)	5 15	5 11	5 14
20001		Total ($n \times T$)	75	55	70

Table 4.8 (Continued) PANEL C

Time Series Regression Analysis for the Primary Sample Change in Total Assets $(A_{t-1, t})$ and Market Price Return $(M_{t-1, t})$

Panel notes:

The sample consists of a maximum of 75 observations from the 5 banks in the LSE that adopted the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008. See main table notes for the regression model applied.

	Maaguma		Total Period	Sub-F	Periods
	Measure		1994-2008	1994-2004	1994-2007
	Slope	Ь	1.745**	1.707	1.342*
	1	t (b)	3.376	1.921	2.346
		p (b)	5.0E-03	8.7E-02	3.7E-02
		SE(b)	0.517	0.889	0.572
		Lower	0.188	-1.181	-0.405
		Upper	3.303	4.594	3.090
	Intercept	а	-0.091	-0.054	-0.039
Regression	1	t (a)	-1.473	-0.620	-0.563
		p (a)	1.6E-01	5.5E-01	5.8E-01
		SE(a)	0.062	0.087	0.070
		Lower	-0.277	-0.337	-0.252
		Upper	0.095	0.229	0.174
	Model	R^2	0.467	0.291	0.314
		s(e)	0.176	0.189	0.170
		df(e)	13	9	12
		14	0.683**	0.520*	0.561*
Correla	tion	p(r)	2.5E-03	4.3E-02	1.8E-02
		A	0.081	0.074	0.092
		SD(A)	0.091	0.067	0.082
Deservinting	Statistics	SE(A)	0.024	0.020	0.022
Descriptive	sialistics	М	0.050	0.072	0.085
		SD(M)	0.233	0.213	0.197
		SE(M)	0.060	0.064	0.053
		Firms (n)	11	11	11
Observa	tions	Years (T)	15	11	14
		Total ($n \times T$)	165	121	154

Table 4.8 (Continued) PANEL D

Time Series Regression Analysis for the Secondary Sample Change in Total Assets $(A_{t-1, t})$ and Market Price Return $(M_{t-1, t})$

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The sample consists of a maximum of 165 observations from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008.

See main table notes for the regression model applied.

Analysis

The Table 4.8 time series regression slopes show that the change in the total assets variable exhibits statistically significant explanatory power to the market price return variable for the Primary and Secondary sample for the 1994 to 2004 time period. The Control group exhibits statistical significance for the three time series periods: 1994 to 2004, 1994 to 2007 and 1994 to 2008 for the same change variables.

The Table 4.8 Panel A time series regression slope for the Primary and Secondary sample shows that before 2005, a 1% variation in the change in the total assets variable related to more than a 2% change in the market price return variable. After 2005, the slope does not show significance. However, the slope for the Control group, Panel B, for before and after 2005 shows that for a 1% variation in the change in the total assets variable the market price return variable changes more than 1%.

The time series slopes suggest that for the Primary and Secondary sample, the total assets change variable reacted significantly to the market price return variable before the 2005 accounting change and does not show a statistically significant reaction after the change. For the Control group the total assets change variable reacted significantly with a similar magnitude to the market price return variable before and after 2005.

4.3.4 Regressions for Change in Total Liabilities and Market Price Return

The regression analysis for the change in total liabilities and market price return is presented for the time range 1994 to 2008 in Table 4.9 (and Table U.4). In the regressions, the market price return variable is tested as the dependent variable, and the change in the total liabilities variable is tested as the independent variable.

Table 4.9Time Series Regression Analysis for Change in Total Liabilities and
Market Price Return

Time Series Regression

Table 4.9 Panel A to Panel D show the results for the time series regressions that test the change in total liabilities and market price return by applying the regression specified in equation (3.32). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.

Time Series Regression Table Description

The Table 4.9 columns represent the following:

The Total Period column represents the time series regression results for the year range 1994 to 2008, with a single year represented from 1st January to 31st December, with the 31st December selected as the variable record date. The Sub-Periods column represents the time series regression results for the year range 1994 to 2004, and for 1994 to 2007, with a single year represented from 1st January to 31st December, and a 31st December variable record date.

The Measure column represents the time series regression results for the total and sub-time periods and is represented by the following data panels:

The *Regression* panel presents the coefficients and related statistics for the regression specified in equation (3.32), and follows the model:

$dM_{N,t} = a + b \ dL_{N,t} + e_{N,t}$

Where: $dM_{N,t}$ is the log Market Price return $(M_{t-1,t})$ for N firms at time t, $dL_{N,t}$ is the log change in Total Liabilities $(L_{t-1,t})$ for N firms at time t, and $e_{N,t}$ is the regression error term. The *Lower* and *Upper* are the corresponding coefficient levels predicted at a 99% confidence level.

The *Correlation* panel presents the correlation statistics for the relationship between the dependent and independent variables. The *Descriptive Statistics* panel presents for the variables tested, the average sample means calculated using equation (M.6); the standard deviation, *SD*, of sample means for the time series using equation (M.7); and, the standard error of means, *SE*, for the time series. The *Observations* panel presents: *Firms (n)* (term *N* in the regression) that represents the average number of sample firms in the time series regression (for the Primary and Secondary sample, for the time period 1994 to 2004, the number of firms are rounded down); *Years (T)* represents the number of time series years; *Total (n × T)* represents the number of averaged pooled time series regression observations.

	M		Total Period	Sub-F	Periods
	Measure		1994-2008	1994-2004	1994-2007
	Slope	b	0.252	0.302	0.228
	1	t (b)	0.326	0.310	0.416
		p (b)	7.5E-01	7.6E-01	6.8E-01
		SE(b)	0.772	0.975	0.548
		Lower	-2.074	-2.868	-1.445
		Upper	2.578	3.472	1.901
	Intercept	а	0.005	0.051	0.056
Regression	Ĩ	t (a)	0.047	0.462	0.690
-		p (a)	9.6E-01	6.6E-01	5.0E-01
		SE(a)	0.113	0.111	0.081
		Lower	-0.334	-0.308	-0.192
		Upper	0.345	0.410	0.304
Model	Model	R^2	0.008	0.011	0.014
		s(e)	0.263	0.204	0.186
		df(e)	13	9	12
		r	0.090	0 103	0 1 1 9
Correla	tion	p(r)	3.7E-01	3.8E-01	3.4E-01
		L	0.117	0.094	0.117
		$\tilde{SD}(L)$	0.091	0.066	0.094
Descriptive	Statistics	SE(L)	0.023	0.020	0.025
Descriptive	siunsnes	М	0.035	0.080	0.083
		SD(M)	0.254	0.194	0.180
		SE(M)	0.066	0.059	0.048
		Firms (n)	16	15	16
Observa	tions	Years (T)	15	11	14
		Total $(n \times T)$	233	169	217

Table 4.9 (Continued)
PANEL A

Time Series Regression Analysis for the Primary and Secondary Sample Change in Total

Panel notes:

The sample consists of a maximum of 233 observations from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008.

See main table notes for the regression model applied.

	Mangura		Total Period	Sub-P	eriods
	Measure		1994-2008	1994-2004	1994-2007
	Slope	b	0.451	0.431	0.295
		t(b)	1.289	1.157	0.929
		p(b)	2.2E-01	2.8E-01	3./E-01
		SE(b)	0.350	0.3/3	0.317
		Lower	-0.603	-0.780	-0.675
		Upper	1.506	1.643	1.264
-	Intercept	а	-0.003	-0.012	0.034
Regression		t (a)	-0.048	-0.185	0.632
		<i>p</i> (<i>a</i>)	9.6E-01	8.6E-01	5.4E-01
		SE(a)	0.057	0.064	0.053
		Lower	-0.175	-0.221	-0.129
		Upper	0.170	0.198	0.197
	Model	\mathbf{D}^2	0.112	0 1 2 0	0.067
	Mouei	\mathbf{x}	0.113	0.130	0.187
		df(e)	13	9	12
		r	0.337	0 360	0 2 5 9
Correla	tion	p(r)	1.1E-01	1.4E-01	1.9E-01
		L	0.049	0.084	0.059
		SD (L)	0.162	0.158	0.163
Deservit	Ct and in the second	SE(L)	0.042	0.048	0.044
Descriptive	SIGUISTICS	М	0.020	0.025	0.051
		SD (M)	0.217	0.190	0.186
		SE(M)	0.056	0.057	0.050
		Firms (n)	12	12	12
Ohserva	tions	Years (T)	15	11	14
		Total ($n \times T$)	180	132	168

Table 4.9 (Continued) PANEL B

Time Series Regression Analysis for the Control Group Change in Total Liabilities $(L_{t-1, t})$ and Market Price Return $(M_{t-1, t})$

Panel notes:

See main table notes for the regression model applied.

The control consists of a maximum of 180 observations from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008.

	Maagura		Total Period	Sub-Periods	
	Measure		1994-2008	1994-2004	1994-2007
	Slope	b	-0.968	0.421	0.072
		t (b)	-1.528	0.814	0.178
		p (b)	1.5E-01	4.4E-01	8.6E-01
		SE(b)	0.634	0.517	0.405
		Lower	-2.877	-1.259	-1.164
		Upper	0.940	2.101	1.309
	Intercept	а	0.164	0.044	0.067
Regression	1	t (a)	1.207	0.513	0.859
		p (a)	2.5E-01	6.2E-01	4.1E-01
		SE(a)	0.136	0.086	0.078
		Lower	-0.245	-0.234	-0.172
		Upper	0.572	0.322	0.306
λ	Model	R^2	0.152	0.069	0.003
		s(e)	0.327	0.187	0.184
		df(e)	13	9	12
		r	-0 390	0.262	0.051
Correla	ition	p(r)	7.5E-02	2.2E-01	4.3E-01
		L	0 167	0 124	0 150
		$\tilde{SD}(L)$	0.138	0.114	0.126
Description	Statistics	$\overline{SE(L)}$	0.036	0.035	0.034
Descriptive	SIGUISTICS	М	0.001	0.096	0.078
		SD(M)	0.342	0.184	0.177
		SE(M)	0.088	0.055	0.047
		Firms (n)	5	5	5
Observa	tions	Years (T)	15	11	14
Observations					

Table 4.9 (Continued)
PANEL C

Time Series Regression Analysis for the Primary Sample Change in Total Liabilities $(L_{t-1, t})$ and Market Price Return $(M_{t-1, t})$

Panel notes:

The sample consists of a maximum of 75 observations from the 5 banks in the LSE that adopted the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008. See main table notes for the regression model applied.

Maaaaa		Total Period	Sub-Periods	
Measure	;	1994-2008	1994-2004	1994-2007
Slope	b	0.610	-0.149	0.213
	t (b)	0.982	-0.154	0.373
	$p(\dot{b})$	3.4E-01	8.8E-01	7.2E-01
	SE(b)	0.621	0.969	0.571
	Lower	-1.260	-3.299	-1.531
	Upper	2.480	3.000	1.957
Intercept	а	-0.007	0.084	0.063
Regression	t (a)	-0.082	0.818	0.796
	p (a)	9.4E-01	4.3E-01	4.4E-01
	SE(a)	0.084	0.103	0.079
	Lower	-0.258	-0.250	-0.179
	Upper	0.245	0.418	0.306
Model	R^2	0.069	0.003	0.011
	s(e)	0.234	0.224	0.204
	df(e)	13	9	12
	14	0.263	-0.051	0.107
Correlation	p(r)	1.7E-01	4.4E-01	3.6E-01
	L	0.093	0.080	0.101
	$\overline{SD}(L)$	0.101	0.073	0.099
Descriptive Statistics	SE(L)	0.026	0.022	0.027
Descriptive statistics	М	0.050	0.072	0.085
	SD(M)	0.233	0.213	0.197
	SE(M)	0.060	0.064	0.053
	Firms (n)	11	11	11
Observations	Years (T)	15	11	14
	Total ($n \times T$)	165	121	154

Table 4.9 (Continued)
PANEL D

Time Series Regression Analysis for the Secondary Sample Change in Total Liabilities

Panel notes:

The sample consists of a maximum of 165 observations from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008.

See main table notes for the regression model applied.

Analysis

The Table 4.9 time series regression slopes show that for the samples and the control group the change in the total liabilities variable does not exhibit statistically significant explanatory power to the market price return variable. These slopes suggest that the total liabilities and market price change variables of the samples and the control group does not show statistically significant reactions after the 2005 accounting change.

4.3.5 Regressions for Change in Net Income and Market Price Return

The regression analysis for the change in net income and market price return is presented for the time range 1994 to 2008 in Table 4.10 (and Table U.5). In the regressions, the market price return variable is tested as the dependent variable, and the change in the net income variable is tested as the independent variable.

Table 4.10Time Series Regression Analysis for Change in Net Income and
Market Price Return

Time Series Regression

Table 4.10 Panel A to Panel D show the results for the time series regressions that test the change in net income and market price return by applying the regression specified in equation (3.32). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.

Time Series Regression Table Description

The Table 4.10 columns represent the following:

The Total Period column represents the time series regression results for the year range 1994 to 2008, with a single year represented from 1st January to 31st December, with the 31st December selected as the variable record date. The Sub-Periods column represents the time series regression results for the year range 1994 to 2004, and for 1994 to 2007, with a single year represented from 1st January to 31st December, and a 31st December variable record date.

The Measure column represents the time series regression results for the total and sub-time periods and is represented by the following data panels:

The *Regression* panel presents the coefficients and related statistics for the regression specified in equation (3.32), and follows the model:

$dM_{N,t} = a + b \ dI_{N,t} + e_{N,t}$

Where: $dM_{N,t}$ is the log Market Price return $(M_{t-1,t})$ for N firms at time t, $dI_{N,t}$ is the change in Net Income $(I_{t-1,t})$ for N firms at time t, and $e_{N,t}$ is the regression error term. The *Lower* and *Upper* are the corresponding coefficient levels predicted at a 99% confidence level.

The *Correlation* panel presents the correlation statistics for the relationship between the dependent and independent variables. The *Descriptive Statistics* panel presents for the variables tested, the average sample means calculated using equation (M.6); the standard deviation, *SD*, of sample means for the time series using equation (M.7); and, the standard error of means, *SE*, for the time series. The *Observations* panel presents: *Firms (n)* (term *N* in the regression) that represents the average number of sample firms in the time series regression (for the Primary and Secondary sample, for the time period 1994 to 2004, the number of firms are rounded down); *Years (T)* represents the number of time series years; *Total (n × T)* represents the number of averaged pooled time series regression observations.

N		Total Period	Sub-F	Periods
Measure	; 	1994-2008	1994-2004	1994-2007
Slope	b	0.038	-0.025	0.012
1	t (b)	0.840	-0.356	0.343
	p (b)	4.2E-01	7.3E-01	7.4E-01
	SE(b)	0.045	0.070	0.034
	Lower	-0.098	-0.253	-0.091
	Upper	0.173	0.203	0.114
Intercept	a	0.023	0.077	0.078
Regression	t (a)	0.334	1.257	1.506
0	p (a)	7.4E-01	2.4E-01	1.6E-01
	SE(a)	0.068	0.062	0.052
	Lower	-0.182	-0.123	-0.080
	Upper	0.227	0.278	0.236
Model	R^2	0.051	0.014	0.010
	s(e)	0.257	0.203	0.187
	df(e)	13	9	12
	r	0 227	-0 118	0.099
Correlation	p(r)	2.1E-01	3.6E-01	3.7E-01
	I	0.315	-0.084	0.402
		1.526	0.918	1.545
Descriptive Statistics	SE(I)	0.394	0.277	0.413
Descriptive statistics	М	0.035	0.080	0.083
	SD (M)	0.254	0.194	0.180
	SE(M)	0.066	0.059	0.048
	Firms (n)	16	15	16
Observations	Years (T)	15	11	14

PANEL A Time Series Regression Analysis for the Primary and Secondary Sample Change in Net

Table 4.10 (Continued)

Panel notes:

See main table notes for the regression model applied.

The sample consists of a maximum of 233 observations from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008.

	Maaaaaa		Total Period	Sub-F	Periods
	Measure		1994-2008	1994-2004	1994-2007
	Slope	b	-0.263	-0.154	-0.102
		t (b)	-2.096	-0.814	-0.560
		p (b)	5.6E-02	4.4E-01	5.9E-01
		SE(b)	0.125	0.189	0.181
		Lower	-0.641	-0.768	-0.656
		Upper	0.115	0.460	0.452
	Intercept	а	0.037	0.018	0.050
Regression	1	t (a)	0.731	0.302	0.977
		p (a)	4.8E-01	7.7E-01	3.5E-01
		SE(a)	0.051	0.059	0.051
		Lower	-0.116	-0.173	-0.106
		Upper	0.190	0.209	0.206
	Model	R^2	0.253	0.069	0.025
		s(e)	0.194	0.193	0.191
		df(e)	13	9	12
		r	-0.503*	-0 262	-0 160
Correla	ition	p(r)	2.8E-02	2.2E-01	2.9E-01
		Ι	0.067	-0.044	-0.011
		SD (I)	0.414	0.323	0.292
Descripting	Statistics	SE(I)	0.107	0.097	0.078
Descriptive	SIGUISTICS	M	0.020	0.025	0.051
		SD(M)	0.217	0.190	0.186
		SE(M)	0.056	0.057	0.050
		Firms (n)	12	12	12
Observe	itions	Years (T)	15	11	14
Observations					

Table 4.10 (Continued) PANEL B

Time Series Regression Analysis for the Control Group Change in Net Income (I_{t-1, t}) and

Panel notes:

See main table notes for the regression model applied.

^{**, *} Correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The control consists of a maximum of 180 observations from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008.

		Market P	rice Return (<i>M_{t-1}</i> ,	t)	
	N		Total Period	Sub-F	eriods
	Measure		1994-2008	1994-2004	1994-2007
	Slope	b	0.404**	0.093	0.107
	1	t (b)	4.074	0.740	0.911
		p (b)	1.3E-03	4.8E-01	3.8E-01
		SE(b)	0.099	0.126	0.118
		Lower	0.105	-0.316	-0.252
		Upper	0.703	0.503	0.466
	Intercent	а	-0.047	0.072	0.052
Regression		t (a)	-0.752	1.088	0.933
5		p (a)	4.7E-01	3.0E-01	3.7E-01
		SE(a)	0.062	0.066	0.056
		Lower	-0.233	-0.142	-0.118
		Upper	0.140	0.285	0.222
/	Model	R^2	0.561	0.057	0.065
		s(e)	0.235	0.188	0.178
		df(e)	13	9	12
		74	0.740**	0.230	0.254
Correl	ation	p(r)	6.6E-04	2.4E-01	1.9E-01
		I	0 1 1 9	0 264	0.245
		$\frac{1}{SD(I)}$	0.634	0.472	0.420
Deservinting	Ctatistics.	SE(I)	0.164	0.142	0.112
Descriptive	SIGUSTICS	М	0.001	0.096	0.078
		SD (M)	0.342	0.184	0.177
		SE(M)	0.088	0.055	0.047
		Firms (n)	5	5	5
Ohserv	ations	Years (T)	15	11	14
		Total ($n \times T$)	75	55	70

Table 4.10 (Continued)
PANEL C

Time Series Regression Analysis for the Primary Sample Change in Net Income (I_{t-1, t}) and

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The sample consists of a maximum of 75 observations from the 5 banks in the LSE that adopted the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008. See main table notes for the regression model applied.

	Массина		Total Period	Sub-Periods	
Measure		1994-2008	1994-2004	1994-2007	
	Slope	b	0.016	-0.035	0.009
	-	t (b)	0.559	-0.694	0.355
		p (b)	5.9E-01	5.1E-01	7.3E-01
		SE(b)	0.029	0.051	0.025
		Lower	-0.072	-0.199	-0.068
		Upper	0.104	0.129	0.086
	Intercept	а	0.043	0.064	0.081
Regression	1	t (a)	0.688	0.949	1.441
		p (a)	5.0E-01	3.7E-01	1.8E-01
		SE(a)	0.063	0.067	0.056
		Lower	-0.146	-0.154	-0.090
		Upper	0.233	0.282	0.251
	Model	R^2	0.023	0.051	0.010
		s(e)	0.239	0.219	0.204
		df(e)	13	9	12
Correlation		14	0.153	.0.225	0.102
		p(r)	2.9E-01	2.5E-01	3.6E-01
		I	0 404	-0 243	0 473
		SD (I)	2.193	1.368	2.259
Descriptive Statistics		$\overline{SE(I)}$	0.566	0.412	0.604
		М	0.050	0.072	0.085
		SD (M)	0.233	0.213	0.197
		SE(M)	0.060	0.064	0.053
		Firms (n)	11	11	11
Observations		Years (T)	15	11	14
		Total $(n \times T)$	165	121	154

PANEL D Time Series Regression Analysis for the Secondary Sample Change in Net Income $(I_{t-I, t})$ and Market Price Return $(M_{t-1, t})$

Table 4.10 (Continued)

Panel notes:

The sample consists of a maximum of 165 observations from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008.

See main table notes for the regression model applied.
Analysis

The Table 4.10 time series regression slopes generally show that for the samples and the control group the change in the net income variable does not exhibit statistically significant explanatory power to the market price return variable. The only exception is the Primary sample that shows after 2005, a 1% variation in the change in the net income variable related close to a 0.5% (0.404) change in the market price return variable. Other than this, these time series slopes suggest that the net income change variable of the samples and the control group does not show statistically significant reaction to the market price return variable before and after the 2005 accounting change.

4.4 Significance of Differences for Change in Accounting Totals and Market Price Return Regression Slope Coefficients

The analysis of the significance of differences between the change in accounting totals and the market price return time series regressions slope coefficients from Section 4.3 is presented in Table 4.11.

Table 4.11t-Test for Significance of Differences Within and Between Samplesand Control Group for Change in Accounting Totals and Market Price Return
Regression Slopes from Table 4.6 to Table 4.10

Table 4.11 Panel A to Panel E presents the *t-statistics* that provides a comparison of slope coefficients *within the sample*, and for slope coefficients *between the samples*, for the change in accounting totals to market price return time series regressions presented in Table 4.6 to Table 4.10. The *within sample* comparison is between the regression slope coefficients for the time periods 1994 to 2004, 1994 to 2007 and 1994 to 2008. From each of the *within sample* slope coefficients, the *between samples* comparison tests between the Primary and Secondary sample, Control group, Primary sample and Secondary sample. Table 4.11 comprises of four matrices (Panel A to Panel E) and each matrix is designed to be read first vertically and then horizontally. The intersection between the *slope coefficient corresponding to the vertically labelled sample or control group* to the *slope coefficient corresponding to the horizontally labelled sample or control group*.

Panel A presents the *t-statistic* results that compare the slope coefficients for the change in total equity and market price return from Table 4.6 results. Panel B presents the *t-statistic* results that compare the slope coefficients for the change in total equity per share and market price return from Table 4.7 results. Panel C presents the *t-statistic* results that compare the slope coefficients for the change in total exist price return from Table 4.8 results. Panel D presents the *t-statistic* results that compare the slope coefficients for the change in total Assets and market price return from Table 4.8 results. Panel D presents the *t-statistic* results that compare the slope coefficients for the change in total liabilities and market price return from Table 4.9 results. Panel E presents the *t-statistic* results that compare the slope coefficients for the change in net income and market price return from Table 4.10 results.

The *t*-test applied to compare the difference of slope coefficients takes the form (Cohen 1983, Paternoster, Brame, Mazerolle and Piquero1998):

$$t - statistic = \frac{b_1 - b_2}{\sqrt{SE(b_1)^2 + SE(b_2)^2}}$$
(4.3)

Where: t - statistic is the measure of difference between the regression slope coefficients; b_1 and b_2 are the regression slope coefficients compared; and, $SE(b_1)$ and $SE(b_2)$ are the standard errors for the slopes b_1 and b_2 respectively.

The Table 4.11 columns represent the following:

The Sample and Control Group column is the horizontal matrix label. This column presents for the samples and control group the *t-statistic* for the slope coefficient comparisons for the time periods presented in the Time Period column. The PS record presents the Primary and Secondary sample *t-statistics*. The C record presents the Control group *t-statistics*. The P record presents the Primary sample *t-statistics*. The S record presents the Secondary sample *t-statistics*. The Time Periods column presents the regression time period records: 1994-2008, 1994-2007 and 1994-2004. These identify the horizontal *t-statistic* record. The second Samples and Control Group column is the vertical matrix label. This column presents the *t-statistic* corresponding to the sample and control group regression slope coefficients for the specified time period. This column, generally, has the same definitions as the horizontal labels.

<i>t</i> -1	Fest for Sign	ifical	ace of Dif	ferences	Within and	Between Sa	PAN amples an Slope Coo	NEL A nd Control (Group for Ch	ange in To	tal Equity a	nd Market	Price Ret	urn
- - 2 1.		_				xegi ession .	Stope Coe	Sample an	nd Control Gro	oup				
Sample and			Primary	y and Secon	idary (PS)		Control (C	C)		Primary (P)		5	Secondary ((S)
Group	Time Periods		1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008
							t-stat	tistic for diff	ferences betwe	een slopes		-		
PS -	1994-2004 1994-2007 1994-2008	t-statisti	-0.060 -0.575	0.060 - -0.552	0.575 0.552 -	0.088 0.012 -0.659	0.157 0.091 -0.599	0.305 0.259 -0.481	-1.814 -2.103 -2.218	-1.831 -2.127 -2.233	-1.412 -1.581 -1.848	0.002 -0.048 -0.497	0.074 0.022 -0.490	0.533 0.508 -0.059
с	1994-2004 1994-2007 1994-2008	c for differ	-0.088 -0.157 -0.305	-0.012 -0.091 -0.259	0.659 0.599 0.481	-0.225 -0.749	0.225	0.749 0.473	-4.031** -4.068** -4.446**	-4.212** -4.242** -4.645**	-2.173 -2.233 -2.414*	-0.064 -0.115 -0.226	0.017 -0.047 -0.184	0.619 0.555 0.429
Р	1994-2004 1994-2007 1994-2008	ences betw	1.814 1.831 1.412	2.103 2.127 1.581	2.218 2.233 1.848	4.031** 4.212** 2.173	4.068** 4.242** 2.233	4.446** 4.645** 2.414*	0.000 -0.125	0.000	0.125 0.126	1.404 1.412 1.168	1.800 1.815 1.431	2.270 2.287 1.861
s	1994-2004 1994-2007 1994-2008	een slopes	-0.002 -0.074 -0.533	0.048 -0.022 -0.508	0.497 0.490 0.059	0.064 -0.017 -0.619	0.115 0.047 -0.555	0.226 0.184 -0.429	-1.404 -1.800 -2.270	-1.412 -1.815 -2.287	-1.168 -1.431 -1.861	-0.062 -0.456	0.062	0.456 0.446

Table 4.11 (Continued)

Panel notes:

**, * *t-statistic* for the significance of slope coefficients difference is significant at the 0.01 and 0.05 two-tailed levels. A highlighted column identifies a regression slope coefficient with a *p-value* at the 0.05 significance level from Table 4.6.

Table 4.11 (Continued) PANEL B

t-Test for Significance of Differences Within and Between Samples and Control Group for Change in Total Equity per Share and Market Price Return **Regression Slope Coefficients from Table 4.7**

c 1								Sample and	Control Grou	np				
and			Primary	y and Secon	dary (PS)	()	Control (C)			Primary (P))	S	Secondary ((S)
Group	Time Periods		1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008
						-	t-statis	stic for diffe	rences betwee	en slopes				
PS	1994-2004 1994-2007 1994-2008	t-statisti	0.012 -0.510	-0.012 - -0.567	0.510 0.567 -	-2.241 -2.481* -4.544**	-2.166 -2.398* -4.426**	-2.072 -2.293 -4.28**	-1.638 -1.717 -2.518*	-1.811 -1.900 -2.725*	-0.028 -0.019 -0.449	-0.148 -0.143 -0.579	-0.165 -0.161 -0.682	-0.302 -0.310 -1.033
с	1994-2004 1994-2007 1994-2008	c for different	2.241 2.166 2.072	2.481* 2.398* 2.293	4.544** 4.426** 4.28**	-0.514 -1.207	0.514 - -0.684	1.207 0.684 -	-0.011 -0.089 -0.190	-0.265 -0.343 -0.443	1.691 1.633 1.560	1.514 1.458 1.386	1.877 1.807 1.719	2.634* 2.53* 2.398*
Р	1994-2004 1994-2007 1994-2008	ences betwe	1.638 1.811 0.028	1.717 1.900 0.019	2.518* 2.725* 0.449	0.011 0.265 -1.691	0.089 0.343 -1.633	0.190 0.443 -1.560	0.181 -1.374	-0.181	1.374 1.524	1.238 1.386 -0.107	1.413 1.581 -0.116	1.594 1.795 -0.219
S	1994-2004 1994-2007 1994-2008	en slopes	0.148 0.165 0.302	0.143 0.161 0.310	0.579 0.682 1.033	-1.514 -1.877 -2.634*	-1.458 -1.807 -2.53*	-1.386 -1.719 -2.398*	-1.238 -1.413 -1.594	-1.386 -1.581 -1.795	0.107 0.116 0.219	-0.002 0.084	0.002	-0.084 -0.101 -

Panel notes:

**, * *t-statistic* for the significance of slope coefficients difference is significant at the 0.01 and 0.05 two-tailed levels. A highlighted column identifies a regression slope coefficient with a *p-value* at the 0.05 significance level from Table 4.7.

Table 4.11 (Continued)

PANEL C t-Test for Significance of Differences Within and Between Samples and Control Group for Change in Total Assets and Market Price Return Regression Slope Coefficients from Table 4.8

c 1								Sample an	d Control Gi	oup				
and			Primary	and Secon	dary (PS)		Control (C)		Primary (P))	5	Secondary (S)
Group	Time Periods		1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008
							t-stat	<i>istic</i> for diff	ferences betw	veen slopes		1		
	1994-2004	t-5	4	-0.940	-0.508	-1.258	-1.224	-1.249	-1.806	-2.267	-2.944*	-0.368	-0.813	-0.421
PS	1994-2007	ta	0.940	-	0.348	-0.177	-0.130	-0.161	-1.052	-1.625	-2.481*	0.462	0.152	0.686
	1994-2008	tistic	0.508	-0.348		-0.543	-0.510	-0.532	-1.190	-1.618	-2.377	0.118	-0.224	0.186
	1994-2004	; for	1.258	0.177	0.543		0.170	0.071	-1.352	-2.464*	-3.096*	0.658	0.385	1.187
С	1994-2007	d:	1.224	0.130	0.510	-0.170	-	-0.117	-1.401	-2.519*	-3.132**	0.627	0.338	1.133
_	1994-2008	ffere	1.249	0.161	0.532	-0.071	0.117		-1.383	-2.519*	-3.126**	0.648	0.370	1.177
	1994-2004	nce	1.806	1.052	1.190	1.352	1.401	1.383		-0.518	-1.633	1.256	1.206	1.834
P	1994-2007	d s	2.267	1.625	1.618	2.464*	2.519*	2.519*	0.518	-	-1.318	1.654	1.787	2.523*
-	1994-2008	etwe	2.944*	2.481*	2.377	3.096*	3.132**	3.126**	1.633	1.318	<u> </u>	2.380	2.609*	3.208**
	1994-2004	en s	0.368	-0.462	-0.118	-0.658	-0.627	-0.648	-1.256	-1.654	-2.380	1	-0.345	0.037
S	1994-2007	lo	0.813	-0.152	0.224	-0.385	-0.338	-0.370	-1.206	-1.787	-2.609*	0.345	-	0.523
	1994-2008	bes	0.421	-0.686	-0.186	-1.187	-1.133	-1.177	-1.834	-2.523*	-3.208**	-0.037	-0.523	

Panel notes:

**, * *t-statistic* for the significance of slope coefficients difference is significant at the 0.01 and 0.05 two-tailed levels. A highlighted column identifies a regression slope coefficient with a *p-value* at the 0.05 significance level from Table 4.8.

PANEL D *t*-Test for Significance of Differences Within and Between Samples and Control Group for Change in Total Liabilities and Market Price Return Regression Slope Coefficients from Table 4.9

Samula								Sample an	nd Control Gr	oup				
and			Primary	and Secon	dary (PS)		Control (C)		Primary (P)	S	Secondary (S)
Group	Time Periods		1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008
							t-sta	<i>tistic</i> for dif	ferences betw	een slopes				
PS	1994-2004 1994-2007 1994-2008	t-statistic	0.066 0.040	-0.066 -0.025	-0.040 0.025 -	0.124 0.306 0.209	-0.007 0.106 0.052	0.144 0.343 0.235	0.108 0.256 0.182	-0.218 -0.229 -0.206	-1.092 -1.427 -1.221	-0.328 -0.339 -0.324	-0.079 -0.019 -0.041	0.266 0.461 0.361
C	1994-2004 1994-2007 1994-2008	for differe	-0.124 0.007 -0.144	-0.306 -0.106 -0.343	-0.209 -0.052 -0.235	- 0.278 -0.039	-0.278 -0.330	0.039 0.330	-0.016 0.208 -0.048	-0.652 -0.434 -0.708	-1.902 -1.782 -1.959	-0.559 -0.435 -0.582	-0.320 -0.126 -0.355	0.247 0.452 0.223
P	1994-2004 1994-2007 1994-2008	nces betwe	-0.108 0.218 1.092	-0.256 0.229 1.427	-0.182 0.206 1.221	0.016 0.652 1.902	-0.208 0.434 1.782	0.048 0.708 1.959	0.531 1.698	-0.531 - 1.382	-1.698 -1.382	-0.519 -0.210 0.707	-0.270 0.201 1.384	0.234 0.726 1.778
S	1994-2004 1994-2007 1994-2008	en slopes	0.328 0.079 -0.266	0.339 0.019 -0.461	0.324 0.041 -0.361	0.559 0.320 -0.247	0.435 0.126 -0.452	0.582 0.355 -0.223	0.519 0.270 -0.234	0.210 -0.201 -0.726	-0.707 -1.384 -1.778	-0.322 -0.659	0.322 -0.471	0.659 0.471

Panel notes:

There is an absence of statistically significant slope coefficients difference with significances at the 0.01 and 0.05 two-tailed levels. There are no regression slope coefficients with a *p*-value at the 0.05 significance level from Table 4.9.

						1	Table 4.11	(Continued)						
t-Test	for Significa	nce o	of Differenc	ces Within	and Betwee	en Samples a Coe	PAN and Contro efficients fi	EL E ol Group for com Table 4	Change in .10	Net Incom	e and Marke	t Price Retu	rn Regress	ion Slope
								Sample and	Control Gr	oup				
Sample and			Primary	and Second	lary (PS)		Control (C))		Primary (F)		Secondary (S)
Group	Time Periods		1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008
							t-stati	stic for diffe	rences betwe	een slopes				
PS	1994-2004 1994-2007 1994-2008	t-statisti	-0.475 -0.757	0.475	0.757 0.461	-0.640 -0.864 -0.988	-0.397 -0.619 -0.751	-1.661 -2.123 -2.266	0.819 0.621 0.411	0.962 0.774 0.546	3.538** 3.745** 3.366**	-0.115 -0.767 -1.073	0.457 -0.071 -0.563	0.541 0.090 -0.411
с	1994-2004 1994-2007 1994-2008	c for different	0.640 0.397 1.661	0.864 0.619 2.123	0.988 0.751 2.266	-0.199 0.481	0.199	-0.481 -0.732 -	1.087 0.884 2.006	1.171 0.967 2.152	2.615* 2.453* 4.183**	0.608 0.356 1.689	0.855 0.607 2.134	0.889 0.644 2.174
Р	1994-2004 1994-2007 1994-2008	ences betwe	-0.819 -0.962 -3.538**	-0.621 -0.774 -3.745**	-0.411 -0.546 -3.366**	-1.087 -1.171 -2.615*	-0.884 -0.967 -2.453*	-2.006 -2.152 -4.183**	-0.081 -1.941	0.081	1.941 1.928 -	-0.942 -1.105 -3.942**	-0.654 -0.812 -3.868**	-0.596 -0.749 -3.761**
S	1994-2004 1994-2007 1994-2008	en slopes	0.115 -0.457 -0.541	0.767 0.071 -0.090	1.073 0.563 0.411	-0.608 -0.855 -0.889	-0.356 -0.607 -0.644	-1.689 -2.134 -2.174	0.942 0.654 0.596	1.105 0.812 0.749	3.942** 3.868** 3.761**	-0.775 -0.869	0.775 - -0.183	0.869 0.183

Panel notes:

**, * *t-statistic* for the significance of slope coefficients difference is significant at the 0.01 and 0.05 two-tailed levels. A highlighted column identifies a regression slope coefficient with a *p-value* at the 0.05 significance level from Table 4.10.

199

Analysis

The regression tests between the change in accounting totals and market price return variables, presented in Section 4.3, Table 4.6 to Table 4.10, were conducted with the objective to determine first if a statistically significant relationship between the accounting totals and the market price change variables exist. The second objective was to determine if the differences between the statistically significant slope coefficients exhibit an improvement in accounting quality after 2005 when compared to before. To determine if these differences in slopes are also statistically significant within and between the samples and control group for before and after 2005, the *t*-test in equation (4.3) is performed. The results from this test are presented in Table 4.11 Panel A to Panel E.

Examining the Table 4.11 results for the Primary and Secondary sample and the Control group regression slope coefficients, the change in total equity per share to market price return slope difference, presented in Panel B, increased from a 2.241 level before 2005 to a statistically significant 4.28 level after 2005. For the Secondary sample compared to the Control group slope differences, the change in total equity per share to market price return difference increased from a 1.514 level before 2005 to a statistically significant 2.398 level after 2005.

Examining the results for the Primary sample and the Control group slope differences, the total equity to market price return slope difference, presented in Panel A, shows an increase from a 4.031 statistically significant level before 2005, to a 4.242 statistically significant level after 2005 for the 1994 to 2007 time period. However, examining the 1994 to 2008 slope difference, the Primary sample and Control group exhibit a decrease, suggesting a deterioration in accounting quality for the Primary sample firms that was greatest between 2005 and 2007. Examining the Primary sample and Control group slope difference for the change in total assets to market price return, presented in Panel C, shows an increase from a 1.352 level before 2005 to a statistically significant 3.126 level after 2005.

Although the *t*-test applied produces some statistically significant results, it is difficult to determine if the results provide evidence to support this study's null hypothesis that

accounting quality remained unchanged or ideally improved after 2005. It may be contended that accounting quality did not improve after 2005 when examining the levels of differences in slope coefficients exhibited by the total equity per share and the market price return variables. Such evidence supports the Platikanova and Nobes (2006), Paananen and Lin (2009), and Morais and Curto (2008) findings that suggest a deterioration in accounting quality after 2005. Further, from Table 4.11, when examining the before to after 2005 material differences for within the samples and control group slopes, the results suggest that there was a decline in accounting quality. In addition, when comparing sample slope coefficients to the Control group, the after 2005 differences also suggest a decline in accounting quality when compared to the before 2005 slope differences.

From this examination of the accounting totals and market price change variable relationships, this study proceeds to test its null hypothesis by examining the key accounting total variables: total equity, total assets, total liabilities and net income with the market price variable. The tests conducted apply the relative delta and Value-at-Risk measures at the samples and control group levels and also at the individual bank level.

4.5 Relative Delta Measures: Changes in Accounting Totals to Market Price Returns

The analysis and detailed results for the change in key accounting variables, market price returns and the relative delta measure are presented on a yearly basis for the time range 1994 to 2008 in Table 4.13. In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample. Table 4.12 presents relative percentage change results for Table 4.13.

Table 4.12Percentage Changes for Relative Delta Measure Results from Table4.13

Percentage Change Analysis Table Description

The samples and the control group relative delta levels in Table 4.13 for the time period: 1994 to 2004 - before the 2005 accounting change, and the time periods: 1994 to 2007; and 1994 to 2008 - after the accounting change, analysed using relative percentage changes are presented in Table 4.12.

In Table 4.12, the Relative Delta column represents the Description and Symbol columns that describes and *codes* the relative delta measures. The Year column represents the beginning (From) and ending (To) year applied to calculate the percentage changes presented in the Percentage Change column. The Percentage Change column represents the relative delta percentage changes rounded to the nearest whole number. The PS column presents percentage changes for the Primary and Secondary sample. The C column presents the percentage changes for the Primary sample. The S column presents the percentage changes for the Primary sample.

Relative De	elta	Year	Pe	ercentage	Change (%)
Description	Symbol	From - To	PS	C	Р	S
Total Equity to Market Price	(E-M)	1994 - 2004 1994 - 2007 1994 - 2008	-159 203 1,076	-107 -171 10	-99 96 561	-829 1,186 6,143
Total Assets to Market Price	(A-M)	1994 - 2004 1994 - 2007 1994 - 2008	164 1,636 5,114	-133 -180 -72	-28 286 1,139	63 257 631
Total Liabilities to Market Price	(L-M)	1994 - 2004 1994 - 2007 1994 - 2008	-124 294 1,001	-126 -171 -6	-24 291 1,162	-231 277 773
Net Income to Market Price	(I-M)	1994 - 2004 1994 - 2007 1994 - 2008	-52 92 -174	-52 -71 2,340	-59 -79 -147	62 6,708 -1,069

Table 4.13Relative Delta Measures from Change in Accounting Totals and
Market Price Return

Table Description

The Table 4.13 Panel A to Panel D columns represent the following:

The Year column represents the fiscal accounting year. Record Date is the date the variable values are recorded. *Obs.* represents the number of sample firms observed for the year.

For the Change in Variable (Variable Deltas) column:

E and $dE_{t-l,t}$ column represents the relative change in the total equity variable for the *year* time period t - 1 to t calculated by applying equation (3.34); *A* and $dA_{t-l,t}$ column represents the log change in total assets for the *year* time period t - 1 to t calculated by applying equation (3.33); *L* and $dL_{t-l,t}$ column represents the log change in total liabilities for the *year* time period t - 1 to t calculated by applying equation (3.33); *L* and $dL_{t-l,t}$ column represents the log change in total liabilities for the *year* time period t - 1 to t calculated by applying equation (3.33); *I* and $dI_{t-l,t}$ column represents the relative change in the net income variable for the *year* time period t - 1 to t calculated by applying equation (3.34); and M and $dM_{t-l,t}$ column represents the market price return specified by the log change in market price for the *year* time period t - 1 to t and calculated by applying equation (3.35).

For the Relative Delta Measures column:

E-M column represents the Total Equity to Market Price Relative Delta for the *year* time period t - 1 to t calculated by applying equation (3.38), and takes the general form:

$$\delta EM_{i,t} = dE_{i,t} - dM_{i,t}$$

Where: $\delta EM_{i,t}$ is the Total Equity to Market Price Relative Delta for the i^{th} firm at time t, $dE_{i,t}$ is the change in Total Equity $(dE_{t-1,t})$ for the i^{th} firm at time t, and $dM_{i,t}$ is the log change in market price $(dM_{t-1,t})$ for the i^{th} firm at time t.

A-M column represents the Total Assets to Market Price Relative Delta for the *year* time period t - 1 to t calculated by applying equation (3.38), and takes the general form:

$$\delta AM_{i,t} = dA_{i,t} - dM_{i,t}$$

Where: $\delta AM_{i,t}$ is the Total Assets to Market Price Relative Delta for the *i*th firm at time t, $dA_{i,t}$ is the log change in Total Assets $(dA_{t-1,t})$ for the *i*th firm at time t, and $dM_{i,t}$ is the log change in market price $(dM_{t-1,t})$ for the *i*th firm at time t.

L-M column represents the Total Liabilities to Market Price Relative Delta for the *year* time period t - 1 to t calculated by applying equation (3.38), and takes the general form:

$$\delta LM_{i,t} = dL_{i,t} - dM_{i,t}$$

Where: $\delta LM_{i,t}$ is the Total Liabilities to Market Price Relative Delta for the *i*th firm at time t, $dL_{i,t}$ is the log change in Total Liabilities $(dL_{t-1,t})$ for the *i*th firm at time t, and $dM_{i,t}$ is the log change in market price $(dM_{t-1,t})$ for the *i*th firm at time t.

I-M column represents the Net Income to Market Price Relative Delta for the *year* time period t - 1 to t calculated by applying equation (3.38), and takes the general form:

$$\delta IM_{i,t} = dI_{i,t} - dM_{i,t}$$

Where: $\delta IM_{i,t}$ is the Net Income to Market Price Relative Delta for the i^{th} firm at time t, $dI_{i,t}$ is the change in Net Income $(dI_{t-1,t})$ for the i^{th} firm at time t, and $dM_{i,t}$ is the log change in market price $(dM_{t-1,t})$ for the i^{th} firm at time t.

	Table 4.13 (Continued)													
PANEL A Relative Delta Measures for the Primary and Secondary Sample from Change in Accounting Totals and Market Price Return														
	Change in Variables (Variable Delta) Relative Delta Measures													
Year	Record Date	Obs.	E	A	L	Ι	М		4.34					
			$dE_{t-1,t}$	$dA_{t-1,t}$	$dL_{t-1,t}$	$dI_{t-1,t}$	$dM_{t-1,t}$	E-M	A-M	L-M	I-M			
1994	30-Dec-1994	15	0.092	0.020	0.101	0.394	0.034	0.058	-0.014	0.068	0.360			
1995	29-Dec-1995	16	0.409	0.206	0.216	0.508	0.213	0.196	-0.007	0.003	0.295			
1996	31-Dec-1996	16	0.174	0.103	0.067	0.337	0.091	0.083	0.012	-0.024	0.246			
1997	31-Dec-1997	16	0.223	0.119	0.062	-0.034	0.256	-0.033	-0.137	-0.194	-0.290			
1998	31-Dec-1998	16	0.095	0.095	0.093	0.369	0.058	0.037	0.036	0.035	0.311			
1999	31-Dec-1999	16	0.174	0.094	-0.004	0.207	0.259	-0.085	-0.165	-0.263	-0.052			
2000	29-Dec-2000	16	0.420	0.157	0.201	0.172	0.177	0.244	-0.020	0.025	-0.004			
2001	31-Dec-2001	16	0.084	0.063	0.130	-0.239	-0.160	0.244	0.222	0.289	-0.079			
2002	31-Dec-2002	16	-0.018	-0.009	0.040	-0.133	-0.372	0.354	0.363	0.412	0.239			
2003	31-Dec-2003	16	0.110	0.055	0.060	-2.762	0.233	-0.123	-0.178	-0.173	-2.994			
2004	31-Dec-2004	16	0.052	0.095	0.070	0.258	0.086	-0.034	0.009	-0.016	0.172			
2005	30-Dec-2005	16	0.270	0.307	0.355	4.809	0.203	0.066	0.104	0.152	4.606			
2006	29-Dec-2006	16	0.127	0.114	0.053	1.134	0.154	-0.027	-0.041	-0.101	0.980			
2007	31-Dec-2007	16	0.099	0.138	0.191	0.613	-0.077	0.176	0.215	0.268	0.690			
2008	31-Dec-2008	16	0.046	0.066	0.113	-0.903	-0.636	0.682	0.702	0.749	-0.266			

Panel notes: The Change in Variables are sample averages. The sample consists of a maximum of 16 observations per year from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.

PANEL B Relative Delta Measures for the Control Group from Change in Accounting Totals and Market Price Return											
				Change in V	/ariables (Va	ariable Delta	l)		Relative Del	ta Measures	
Year	Record Date	Obs.	E	A	L	Ι	М		4.34		
			$dE_{t-l,t}$	$dA_{t-1,t}$	$dL_{t-1,t}$	$dI_{t-1,t}$	$dM_{t-1,t}$	<i>E-M</i>	A-M	L-M	<i>I-M</i>
1994	30-Dec-1994	12	-0.020	0.008	0.268	-0.028	-0.092	0.073	0.100	0.360	0.065
1995	29-Dec-1995	12	0.164	0.127	0.313	0.081	0.103	0.062	0.024	0.210	-0.022
1996	31-Dec-1996	12	0.158	0.089	0.026	0.095	0.020	0.138	0.069	0.006	0.074
1997	31-Dec-1997	12	0.128	0.097	0.027	0.062	0.103	0.025	-0.006	-0.076	-0.040
1998	31-Dec-1998	12	-0.040	-0.040	0.214	0.248	-0.106	0.066	0.065	0.320	0.354
1999	31-Dec-1999	12	0.446	0.285	0.243	-0.903	0.355	0.091	-0.069	-0.112	-1.257
2000	29-Dec-2000	12	-0.032	-0.040	0.092	0.063	0.015	-0.048	-0.055	0.076	0.048
2001	31-Dec-2001	12	-0.196	-0.196	-0.197	0.001	-0.183	-0.013	-0.013	-0.014	0.184
2002	31-Dec-2002	12	-0.273	-0.275	-0.066	-0.360	-0.307	0.034	0.032	0.241	-0.054
2003	31-Dec-2003	12	0.277	0.188	-0.020	0.103	0.238	0.039	-0.050	-0.258	-0.135
2004	31-Dec-2004	12	0.119	0.091	0.030	0.154	0.124	-0.005	-0.033	-0.094	0.031
2005	30-Dec-2005	12	0.283	0.210	0.025	0.102	0.331	-0.048	-0.121	-0.306	-0.229
2006	29-Dec-2006	12	0.044	0.027	0.105	0.182	0.094	-0.050	-0.067	0.011	0.087
2007	31-Dec-2007	12	-0.032	-0.060	-0.234	0.039	0.020	-0.052	-0.080	-0.254	0.019
2008	31-Dec-2008	12	-0.341	-0.394	-0.084	1.165	-0.421	0.080	0.028	0.338	1.586

Panel notes: The Change in Variables are yearly averages from the Control group firms. The Control group consists of a maximum of 12 observations per year from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.

PANEL C Relative Delta Measures for the Primary Sample from Change in Accounting Totals and Market Price Return											
				Change in V	/ariables (V	ariable Delta	ı)		Relative Del	ta Measures	
Year	Record Date	Obs.	E	A	L	Ι	М				
			$dE_{t-1,t}$	$dA_{t-1,t}$	$dL_{t-1,t}$	$dI_{t-1,t}$	$dM_{t-1,t}$	<i>E-M</i>	A-M	L-M	<i>I-M</i>
1994	30-Dec-1994	4	0.129	0.062	0.060	1.173	-0.056	0.185	0.118	0.117	1.229
1995	29-Dec-1995	5	0.368	0.211	0.207	0.374	0.324	0.044	-0.112	-0.117	0.050
1996	31-Dec-1996	5	0.158	0.124	0.122	0.251	0.230	-0.072	-0.106	-0.108	0.021
1997	31-Dec-1997	5	-0.005	0.043	0.047	-0.150	0.226	-0.231	-0.183	-0.179	-0.375
1998	31-Dec-1998	5	0.029	0.014	0.013	0.034	-0.035	0.064	0.048	0.048	0.069
1999	31-Dec-1999	5	0.222	0.125	0.121	0.150	0.334	-0.111	-0.209	-0.213	-0.184
2000	29-Dec-2000	5	1.179	0.438	0.427	0.546	0.094	1.086	0.345	0.334	0.453
2001	31-Dec-2001	5	0.125	0.095	0.092	-0.066	-0.032	0.158	0.127	0.125	-0.033
2002	31-Dec-2002	5	-0.061	0.047	0.055	-0.600	-0.266	0.205	0.313	0.321	-0.334
2003	31-Dec-2003	5	0.113	0.070	0.068	0.623	0.175	-0.062	-0.105	-0.108	0.447
2004	31-Dec-2004	5	0.068	0.150	0.155	0.569	0.066	0.002	0.085	0.089	0.504
2005	30-Dec-2005	5	0.181	0.352	0.362	0.336	0.085	0.095	0.267	0.277	0.251
2006	29-Dec-2006	5	0.128	0.092	0.090	0.100	0.120	0.007	-0.028	-0.030	-0.021
2007	31-Dec-2007	5	0.193	0.286	0.287	0.089	-0.171	0.363	0.456	0.458	0.259
2008	31-Dec-2008	5	0.151	0.391	0.405	-1.647	-1.071	1.222	1.462	1.476	-0.576

Panel notes: The Change in Variables are sample averages. The sample consists of a maximum of 5 observations per year from the 5 banks in the LSE that adopted the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.

	Table 4.13 (Continued)												
PANEL D													
	Relative Delta Measures for the Secondary Sample from Change in Accounting Totals and Market Price Return												
				Change in V	Variables (V	ariable Delta	ı)	L	Relative Del	ta Measures			
Year	Record Date	Obs.	E	A	L	Ι	М	EM	A M	I M	IM		
			$dE_{t-1,t}$	$dA_{t-1,t}$	$dL_{t-l,t}$	$dI_{t-1,t}$	$dM_{t-1,t}$	<i>E-1VI</i>	A-M	L-IVI	1-111		
1994	30-Dec-1994	11	0.077	0.003	0.118	0.083	0.070	0.007	-0.067	0.048	0.013		
1995	29-Dec-1995	11	0.425	0.204	0.219	0.561	0.169	0.256	0.035	0.051	0.392		
1996	31-Dec-1996	11	0.179	0.096	0.046	0.369	0.040	0.139	0.055	0.006	0.328		
1997	31-Dec-1997	11	0.307	0.147	0.068	0.008	0.267	0.039	-0.121	-0.200	-0.259		
1998	31-Dec-1998	11	0.119	0.124	0.122	0.491	0.092	0.027	0.032	0.030	0.399		
1999	31-Dec-1999	11	0.152	0.079	-0.061	0.233	0.225	-0.073	-0.146	-0.286	0.008		
2000	29-Dec-2000	11	0.075	0.029	0.098	0.003	0.214	-0.139	-0.185	-0.116	-0.212		
2001	31-Dec-2001	11	0.065	0.048	0.147	-0.318	-0.217	0.283	0.265	0.364	-0.100		
2002	31-Dec-2002	11	0.002	-0.034	0.034	0.080	-0.420	0.422	0.386	0.454	0.500		
2003	31-Dec-2003	11	0.108	0.048	0.057	-4.300	0.259	-0.150	-0.211	-0.202	-4.559		
2004	31-Dec-2004	11	0.044	0.070	0.032	0.116	0.095	-0.051	-0.025	-0.063	0.021		
2005	30-Dec-2005	11	0.310	0.287	0.352	6.843	0.257	0.053	0.030	0.095	6.586		
2006	29-Dec-2006	11	0.127	0.124	0.036	1.604	0.170	-0.043	-0.046	-0.134	1.434		
2007	31-Dec-2007	11	0.056	0.071	0.147	0.851	-0.034	0.090	0.105	0.181	0.885		
2008	31-Dec-2008	11	-0.002	-0.082	-0.020	-0.564	-0.439	0.437	0.356	0.419	-0.126		

Panel notes: The Change in Variables are sample averages. The sample consists of a maximum of 11 observations per year from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. The highlighted 2005 record identified the IFRS implementation event year for the adopting firms.

Analysis

Table 4.13 presents results for the relative delta measure. In this table, for the Primary and Secondary sample there is a greater average difference between the change in accounting total variables and the change in the market price variable after the 2005 accounting change than before. In these results, some *difference component* levels exhibit material significance. However, for the Control group this average difference decreased after 2005 compared to before, other than for the total equity and net income to market price relative delta variables that registered an increase after 2005. The evidence from the relative percentage changes for before and after 2005, presented in Table 4.12, suggest support for these findings.

4.6 Regressions for Accounting Total to Market Price Relative Delta and Historical Value-at-Risk Actual

The analysis and detailed results for the *accounting total to market price relative delta* and the *Historical Value-at-Risk Actual* time series regressions are presented in this section. Appendix V.1 to Appendix V.4 presents analysis and detailed results for cross-sectional regressions using the same variable relationship. The regression analysis is conducted within the time range 1994 to 2008.

4.6.1 Regressions for Total Equity to Market Price Relative Delta and Historical Value-at-Risk Actual

The regression analysis for the total equity to market price relative delta and Historical Value-at-Risk Actual is presented for the time range 1994 to 2008 in Table 4.14 (and Table V.1). In the regressions, the Historical Value-at-Risk Actual variable is tested as the dependent variable, and the total equity to market price relative delta variable is tested as the independent variable.

Table 4.14Time Series Regression Analysis for Total Equity to Market Price
Relative Delta and Historical Value-at-Risk Actual

Time Series Regression

Table 4.14 Panel A to Panel D show the results for the time series regressions that test the total equity to market price relative delta and Historical Value-at-Risk Actual by applying the regression specified in equation (3.40). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.

Time Series Regression Table Description

The Table 4.14 columns represent the following:

The Total Period column presents the time series regression results for the year range 1994 to 2008, with a single year represented from 1st January to 31st December, with the 31st December selected as the variable record date. The Sub-Periods column presents the time series regression results for the year range 1994 to 2004, and for 1994 to 2007, with a single year represented from 1st January to 31st December, and a 31st December variable record date.

The Measure column represents the time series regression results for the total and sub-time periods and is represented by the following data panels:

The *Regression* panel presents the coefficients and related statistics for the regression specified in equation (3.40), and follows the model:

 $V_{N,t} = a + b \left(dE - dM \right)_{N,t} + e_{N,t}$

Where: $V_{N,t}$ is the Value-at-Risk (V_t) for N firms at time t, $(dE - dM)_{N,t}$ is the Total Equity to Market Price Relative Delta $(dE_{t-1,t} - dM_{t-1,t})$ for N firms at time t, and $e_{N,t}$ is the regression error term. The *Lower* and *Upper* are the corresponding coefficient levels predicted at a 99% confidence level.

The *Correlation* panel presents the correlation statistics for the relationship between the dependent and independent variables. The *Descriptive Statistics* panel presents for the variables tested, the average sample means calculated using equation (M.6); the standard deviation, *SD*, of sample means for the time series using equation (M.7); and, the standard error of means, *SE*, for the time series. The *Observations* panel presents: *Firms (n)* (term *N* in the regression) that represents the average number of sample firms in the time series regression (for the Primary and Secondary sample, for the time period 1994 to 2004, the number of firms are rounded down); *Years (T)* represents the number of time series years; *Total (n × T)* represents the number of averaged pooled time series regression observations.

М		Total Period	Sub-F	Periods	
Measure		1994-2008	1994-2004	1994-2007	
Slope	b	-0.703*	-0.252	-0.320	
	t (b)	-2.976	-0.667	-0.935	
	p (b)	1.1E-02	5.2E-01	3.7E-01	
	SE(b)	0.236	0.378	0.342	
	Lower	-1.415	-1.479	-1.365	
	Upper	0.009	0.975	0.725	
Intercept Regression	a t (a) p (a) SE(a)	-0.084 -1.520 1.5E-01 0.055	-0.141 -2.192 5.6E-02 0.064	-0.104 -1.905 8.1E-02 0.054	
	Lower	-0.250	-0.350	-0.270	
	Upper	0.082	0.068	0.063	
Model	R ²	0.405	0.047	0.068	
	s(e)	0.182	0.184	0.174	
	df(e)	13	9	12	
Correlation	r	-0.637**	-0.217	-0.261	
	p(r)	5.4E-03	2.6E-01	1.8E-01	
Descriptive Statistics	dE-dM	0.122	0.085	0.082	
	SD (dE-dM)	0.206	0.154	0.141	
	SE(dE-dM)	0.053	0.047	0.038	
	V 95	-0.170	-0.162	-0.130	
	SD (V 95)	0.228	0.179	0.173	
	SE(V 95)	0.059	0.054	0.046	
Observations	Firms (n)	16	15	16	
	Years (T)	15	11	14	
	Years (T) Total $(n \times T)$	233	169	217	

Table 4.14 (Continued) PANEL A

Time Series Regression Analysis for the Primary and Secondary Sample Total Equity to Market Price Relative Delta ($dE_{t-1, t} - dM_{t-1, t}$) and Historical Value-at-Risk Actual (V_t)

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The sample consists of a maximum of 233 observations from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008.

See main table notes for the regression model applied.

	Maaguma		Total Period	Sub-F	Periods
	Measure		1994-2008	1994-2004	1994-2007
	Slope	b	-1.208	-0.009	-0.910 -0.910
		n(b)	1 3E-01	9 9E-01	2 4E-01
		SE(b)	0.751	0.992	0.729
		Lower	-3.470	-3.232	-3.136
		Upper	1.053	3.215	1.316
	Intercept	а	-0.125*	-0.176*	-0.114*
Regression	1	t (a)	-2.621	-2.735	-2.522
		p (a)	2.1E-02	2.3E-02	2.7E-02
		SE(a)	0.048	0.064	0.045
		Lower	-0.268	-0.386	-0.252
		Upper	0.019	0.033	0.024
Model	Model	R^2	0.166	8.2E-06	0.115
		s(e)	0.168	0.163	0.158
		df(e)	13	9	12
	, .	r	-0.408	-0.003	-0.339
Correla	non	<i>p(r)</i>	6.6E-02	5.0E-01	1.2E-01
		dE-dM	0.026	0.042	0.022
		SD (dE-dM)	0.060	0.052	0.060
Descriptive	Statistics	SE(dE-dM)	0.015	0.016	0.016
Descriptive	suusuos	V 95	-0.156	-0.177	-0.134
		SD (V 95)	0.177	0.155	0.161
		SE(V 95)	0.046	0.047	0.043
		Firms (n)	12	12	12
Observat	tions	Years (T)	15	11	14

Table 4.14 (Continued) PANEL B

Time Series Regression Analysis for the Control Group Total Equity to Market Price

Panel notes:

See main table notes for the regression model applied.

^{**, *} Regression coefficients significant at the 0.01 and 0.05 two-tailed levels.

The control consists of a maximum of 180 observations from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008.

Maaaaa		Total Period	Sub-F	Periods
Measure	;	1994-2008	1994-2004	1994-2007
Slope	b	-0.519**	-0.204	-0.224
1	t (b)	-3.969	-1.334	-1.617
	p (b)	1.6E-03	2.1E-01	1.3E-01
	SE(b)	0.131	0.153	0.139
	Lower	-0.913	-0.701	-0.647
	Upper	-0.125	0.293	0.199
Intercept	a	-0.103	-0.138*	-0.115*
Regression	t(a)	-1.757	-2.567	-2.524
-	p(a)	1.0E-01	3.0E-02	2.7E-02
	SE(a)	0.058	0.054	0.045
	Lower	-0.279	-0.313	-0.253
	Upper	0.073	0.037	0.024
Model	R^2	0.548	0.165	0.179
	s(e)	0.203	0.169	0.157
	df(e)	13	9	12
	r	-0 74**	-0 406	-0.423
Correlation	p(r)	8.0E-04	1.1E-01	6.6E-02
	dE-dM	0 197	0.115	0 124
	SD (dE-dM)	0.415	0.349	0.315
Demoniation Continuitor	SE(dE-dM)	0.107	0.105	0.084
Descriptive Statistics	V 95	-0.205	-0.162	-0.142
	SD (V 95)	0.291	0.175	0.167
	SE(V 95)	0.075	0.053	0.045
	Firms (n)	5	5	5
Observations	Years (T)	15	11	14
	$Total (n \times T)$	75	55	70

Table 4.14 (Continued) PANEL C

Time Series Regression Analysis for the Primary Sample Total Equity to Market Price

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The sample consists of a maximum of 75 observations from the 5 banks in the LSE that adopted the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008. See main table notes for the regression model applied.

	Маадияа		Total Period	Sub-F	Periods
	Measure		1994-2008	1994-2004	1994-2007
	Slope	Ь	-0.590	-0.314	-0.372
	I I I	t (b)	-2.033	-0.872	-1.119
		p(b)	6.3E-02	4.1E-01	2.8E-01
		SE(b)	0.290	0.360	0.332
		Lower	-1.465	-1.483	-1.387
		Upper	0.284	0.855	0.643
	Intercept	а	-0.104	-0.142	-0.103
Regression	1	t (a)	-1.796	-2.089	-1.813
		p (a)	9.6E-02	6.6E-02	9.5E-02
		SE(a)	0.058	0.068	0.057
		Lower	-0.279	-0.364	-0.276
		Upper	0.071	0.079	0.070
	Model	R^2	0.241	0.078	0.095
		s(e)	0.203	0.211	0.198
		df(e)	13	9	12
		r	-0 491*	-0 279	-0.307
Correld	ntion	p(r)	3.1E-02	2.0E-01	1.4E-01
		dE-dM	0.086	0.069	0.061
		SD (dE-dM)	0.186	0.185	0.165
Descriptive	Statistics	SE(dE-dM)	0.048	0.056	0.044
Descriptive	SIGUISTICS	V 95	-0.155	-0.164	-0.126
		SD (V 95)	0.224	0.208	0.200
		SE(V 95)	0.058	0.063	0.053
		Firms (n)		11	11
Observa	tions	Years (T)	15	11	14

Table 4.14 (Continued) PANEL D

Time Series Regression Analysis for the Secondary Sample Total Equity to Market Price

Panel notes:

**, * Correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The sample consists of a maximum of 165 observations from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008.

See main table notes for the regression model applied.

Analysis

The Table 4.14 time series regression slopes show that the total equity to market price relative delta variable exhibits statistically significant explanatory power to the Historical Value-at-Risk Actual variable for the Primary and Secondary sample for the 1994 to 2008 time period. The Control group does not exhibit statistical significance for the three time series periods.

The Table 4.14 Panel A time series regression slope for the Primary and Secondary sample shows that before 2005 there is no statistical significance for the total equity to market price relative delta variable and the Historical Value-at-Risk Actual. After 2005, the regression slope shows that a 1% change in the total equity to market price relative delta variable translates to nearly a 1% (-0.703) increase⁴⁵ in the Historical Value-at-Risk Actual Value-at-Risk Actual Value-at-Risk Actual variable.

The time series slopes suggest that for the Primary and Secondary sample, the total equity to market price relative delta variable reacted significantly to the Historical Value-at-Risk Actual variable after the 2005 accounting change, and does not show a statistically significant reaction before the change. For the Control group the total equity to market price relative delta variable does not show a statistically significant reaction to the Historical Value-at-Risk Actual variable for before and after 2005.

⁴⁵ The slope coefficient, although negative, is indicated as an increase due to the application of a decreasing scale to measure Value-at-Risk (Berkowitz and O'Brien 2002). Therefore, when applying this measurement approach, an *increasing* Value-at-Risk level is signified by a measure that is *negatively increasing*.

4.6.2 Regressions for Total Assets to Market Price Relative Delta and Historical Value-at-Risk Actual

The regression analysis for the total assets to market price relative delta and Historical Value-at-Risk Actual is presented for the time range 1994 to 2008 in Table 4.15 (and Table V.2). In the regressions, the Historical Value-at-Risk Actual variable is tested as the dependent variable, and the total assets to market price relative delta variable is tested as the independent variable.

Table 4.15Time Series Regression Analysis for Total Assets to Market Price
Relative Delta and Historical Value-at-Risk Actual

Time Series Regression

Table 4.15 Panel A to Panel D show the results for the time series regressions that test the total assets and market price relative delta and Historical Value-at-Risk Actual by applying the regression specified in equation (3.40). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.

Time Series Regression Table Description

The Table 4.15 columns represent the following:

The Total Period column presents the time series regression results for the year range 1994 to 2008, with a single year represented from 1st January to 31st December, with the 31st December selected as the variable record date. The Sub-Periods column presents the time series regression results for the year range 1994 to 2004, and for 1994 to 2007, with a single year represented from 1st January to 31st December, and a 31st December variable record date.

The Measure column represents the time series regression results for the total and sub-time periods and is represented by the following data panels:

The *Regression* panel presents the coefficients and related statistics for the regression specified in equation (3.40), and follows the model:

$V_{N,t} = a + b (dA - dM)_{N,t} + e_{N,t}$

Where: $V_{N,t}$ is the Value-at-Risk (V_t) for N firms at time t, $(dA - dM)_{N,t}$ is the Total Assets to Market Price Relative Delta $(dA_{t-1,t} - dM_{t-1,t})$ for N firms at time t, and $e_{N,t}$ is the regression error term. The *Lower* and *Upper* are the corresponding coefficient levels predicted at a 99% confidence level.

The *Correlation* panel presents the correlation statistics for the relationship between the dependent and independent variables. The *Descriptive Statistics* panel presents for the variables tested, the average sample means calculated using equation (M.6); the standard deviation, *SD*, of sample means for the time series using equation (M.7); and, the standard error of means, *SE*, for the time series. The *Observations* panel presents: *Firms (n)* (term *N* in the regression) that represents the average number of sample firms in the time series regression (for the Primary and Secondary sample, for the time period 1994 to 2004, the number of firms are rounded down); *Years (T)* represents the number of time series years; *Total (n × T)* represents the number of averaged pooled time series regression observations.

	N f		Total Period	Sub-F	Periods
-	Measure		1994-2008	1994-2004	1994-2007
	Slope	Ь	-0.627*	-0.342	-0.266
	1	t (b)	-2.918	-0.969	-0.841
		p (b)	1.2E-02	3.6E-01	4.2E-01
		SE(b)	0.215	0.353	0.316
		Lower	-1.274	-1.487	-1.230
		Upper	0.020	0.804	0.699
Iı	ntercept	а	-0.124*	-0.159*	-0.123*
Regression	1	t (a)	-2.482	-2.919	-2.567
0		p (a)	2.8E-02	1.7E-02	2.5E-02
		SE(a)	0.050	0.054	0.048
		Lower	-0.275	-0.335	-0.268
		Upper	0.027	0.018	0.023
	Model	R^2	0.396	0.095	0.056
		s(e)	0.184	0.180	0.175
		df(e)	13	9	12
			0.620**	0.307	0.236
Correlati	on	p(r)	6.0E-03	1.8E-01	2.1E-01
		d A-dM	0.073	0.011	0.029
		SD (dA-dM)	0.229	0.161	0.154
Deservinting St	atistics	SE(dA-dM)	0.059	0.049	0.041
Descriptive St	unsiles	V 95	-0.170	-0.162	-0.130
		SD (V 95)	0.228	0.179	0.173
		SE(V 95)	0.059	0.054	0.046
		Firms (n)	16	15	16
Observatio	ons	Years (T)	15	11	14
Obser valle	0113	Total $(n \times T)$	233	169	217

Table 4.15 (Continued)
PANEL A

Time Series Regression Analysis for the Primary and Secondary Sample Total Assets to

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The sample consists of a maximum of 233 observations from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008.

See main table notes for the regression model applied.

Maa		Total Period	Sub-F	Periods
Mea	sure	1994-2008	1994-2004	1994-2007
Slop	e b	-0.907	0.219	-0.698
*	t (b)	-0.907	0.219	-0.698
	p (b)	2.4E-01	8.1E-01	3.3E-01
	SE(b)	0.732	0.910	0.688
	Lower	-3.112	-2.737	-2.799
	Upper	1.297	3.176	1.404
Interd	ept a	-0.167**	-0.178**	-0.145**
Regression	t (a)	-3.655	-3.612	-3.272
	p(a)	2.9E-03	5.6E-03	6.7E-03
	SE(a)	0.046	0.049	0.044
	Lower	-0.305	-0.338	-0.279
	Upper	-0.029	-0.018	-0.010
Моа	$lel R^2$	0.106	0.006	0.079
	s(e)	0.174	0.162	0.161
	df(e)	13	9	12
	r	-0 325	0.080	-0.281
Correlation	p(r)	1.2E-01	4.1E-01	1.7E-01
	dA-dM	-0.012	0.006	-0.015
	SD (dA-dM)	0.063	0.056	0.065
Damainsting Constant	SE(dA-dM)	0.016	0.017	0.017
Descriptive Statist	V 95	-0.156	-0.177	-0.134
	SD (V 95)	0.177	0.155	0.161
	SE(V 95)	0.046	0.047	0.043
	Firms (n)	12	12	12
Observations	Years (T)	15	11	14
		100	122	1(0

Table 4.15 (Continued) PANEL B

Time Series Regression Analysis for the Control Group Total Assets to Market Price

Panel notes:

See main table notes for the regression model applied.

^{**, *} Regression coefficients significant at the 0.01 and 0.05 two-tailed levels.

The control consists of a maximum of 180 observations from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008.

Maagura		Total Period	Sub-F	Periods
Measure		1994-2008	1994-2004	1994-2007
Slope	b	-0.575**	-0.375	-0.256
	t (b)	-5.047	-1.333	-1.180
	p (b)	2.2E-04	2.2E-01	2.6E-01
	SE(b)	0.114	0.281	0.217
	Lower	-0.918	-1.290	-0.917
	Upper	-0.232	0.539	0.406
Intercept Regression	a t (a) p (a) SE(a)	-0.11* -2.242 4.3E-02 0.049	-0.151* -2.927 1.7E-02 0.051	-0.124* -2.655 2.1E-02 0.047
	Lower Upper	-0.258 0.038	-0.318 0.017	-0.266 0.019
Model	R ²	0.662	0.165	0.104
	s(e)	0.176	0.169	0.164
	df(e)	13	9	12
Correlation	r	-0.814**	-0.406	-0.322
	p(r)	1.1E-04	1.1E-01	1.3E-01
Descriptions Statistics	dA-dM	0.165	0.029	0.073
	SD (dA-dM)	0.412	0.190	0.210
	SE(dA-dM)	0.106	0.057	0.056
Descriptive Statistics	V 95	-0.205	-0.162	-0.142
	SD (V 95)	0.291	0.175	0.167
	SE(V 95)	0.075	0.053	0.045
Observations	Firms (n)	5	5	5
	Years (T)	15	11	14
	Total ($n \times T$)	75	55	70

Table 4.15 (Continued) PANEL C

Time Series Regression Analysis for the Primary Sample Total Assets to Market Price

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The sample consists of a maximum of 75 observations from the 5 banks in the LSE that adopted the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008. See main table notes for the regression model applied.

	N		Total Period	Sub-F	eriods
	Measure		1994-2008	1994-2004	1994-2007
	Slope	b	-0.577	-0.372	-0.356
	-	t (b)	-1.931	-1.056	-1.072
		p (b)	7.6E-02	3.2E-01	3.0E-01
		SE(b)	0.299	0.353	0.332
		Lower	-1.476	-1.518	-1.372
		Upper	0.323	0.774	0.659
	Intercept	а	-0.137*	-0.164*	-0.123*
Regression	1	t (a)	-2.558	-2.621	-2.312
		p (a)	2.4E-02	2.8E-02	3.9E-02
		SE(a)	0.054	0.062	0.053
		Lower	-0.299	-0.366	-0.285
		Upper	0.024	0.039	0.039
	Model	R^2	0.223	0.110	0.087
		s(e)	0.205	0.207	0.199
		df(e)	13	9	12
		r	-0.472*	-0.332	-0.296
Correla	ition	p(r)	3.8E-02	1.6E-01	1.5E-01
		dA-dM	0.031	0.002	0.008
		SD (dA-dM)	0.183	0.186	0.166
Deservinting	Statistics	SE(dA-dM)	0.047	0.056	0.044
Descriptive	SIGUISTICS	V 95	-0.155	-0.164	-0.126
		SD (V 95)	0.224	0.208	0.200
		SE(V 95)	0.058	0.063	0.053
		Firms (n)	11	11	11
Observa	tions	Years (T)	15	11	14
		$T = \langle x 1 \langle x T \rangle$	165	121	154

Table 4.15 (Continued) PANEL D

Time Series Regression Analysis for the Secondary Sample Total Assets to Market Price

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The sample consists of a maximum of 165 observations from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008.

See main table notes for the regression model applied.

Analysis

The Table 4.15 time series regression slopes show that the total assets to market price relative delta variable exhibits statistically significant explanatory power to the Historical Value-at-Risk Actual variable for the Primary and Secondary sample for the 1994 to 2008 time period. The Control group does not exhibit statistical significance for the three time series periods.

The Table 4.15 Panel A time series regression slope for the Primary and Secondary sample shows that before 2005 there is no statistical significance for the total assets to market price relative delta variable and the Historical Value-at-Risk Actual variable. After 2005, the regression slope shows that a 1% change in the total assets to market price relative delta variable translates to nearly a 1% (-0.627) increase in the Historical Value-at-Risk Actual variable.

The time series slopes suggest that for the Primary and Secondary sample the total assets to market price relative delta variable reacted significantly to the Historical Value-at-Risk Actual variable after the 2005 accounting change, and does not show a statistically significant reaction before the change. For the Control group the total assets to market price relative delta variable does not show a statistically significant reaction to the Historical Value-at-Risk Actual variable for before and after 2005.

4.6.3 Regressions for Total Liabilities to Market Price Relative Delta and Historical Value-at-Risk Actual

The regression analysis for the total liabilities to market price relative delta and Historical Value-at-Risk Actual is presented for the time range 1994 to 2008 in Table 4.16 (and Table V.3). In the regressions, the Historical Value-at-Risk Actual variable is tested as the dependent variable, and the total liabilities to market price relative delta variable is tested as the independent variable.

Table 4.16Time Series Regression Analysis for Total Liabilities to Market Price
Relative Delta and Historical Value-at-Risk Actual

Time Series Regression

Table 4.16 Panel A to Panel D show the results for the time series regressions that test the total liabilities to market price relative delta and Historical Value-at-Risk Actual by applying the regression specified in equation (3.40). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.

Time Series Regression Table Description

The Table 4.16 columns represent the following:

The Total Period column presents the time series regression results for the year range 1994 to 2008, with a single year represented from 1st January to 31st December, with the 31st December selected as the variable record date. The Sub-Periods column presents the time series regression results for the year range 1994 to 2004, and for 1994 to 2007, with a single year represented from 1st January to 31st December, and a 31st December variable record date.

The Measure column represents the time series regression results for the total and sub-time periods and is represented by the following data panels:

The *Regression* panel presents the coefficients and related statistics for the regression specified in equation (3.40), and follows the model:

 $V_{N,t} = a + b \left(dL - dM \right)_{N,t} + e_{N,t}$

Where: $V_{N,t}$ is the Value-at-Risk (V_t) for N firms at time t, $(dL - dM)_{N,t}$ is the Total Liabilities to Market Price Relative Delta $(dL_{t-1,t} - dM_{t-1,t})$ for N firms at time t, and $e_{N,t}$ is the regression error term. The *Lower* and *Upper* are the corresponding coefficient levels predicted at a 99% confidence level.

The *Correlation* panel presents the correlation statistics for the relationship between the dependent and independent variables. The *Descriptive Statistics* panel presents for the variables tested, the average sample means calculated using equation (M.6); the standard deviation, *SD*, of sample means for the time series using equation (M.7); and, the standard error of means, *SE*, for the time series. The *Observations* panel presents: *Firms (n)* (term *N* in the regression) that represents the average number of sample firms in the time series regression (for the Primary and Secondary sample, for the time period 1994 to 2004, the number of firms are rounded down); *Years (T)* represents the number of time series years; *Total (n × T)* represents the number of averaged pooled time series regression observations.

to Market Price Relative Delta $(dL_{t-1, t} - dM_{t-1, t})$ and Historical Value-at-Risk Actual (V_t)				
М		Total Period	Sub-F	Periods
Measure			1994-2004	1994-2007
Slope	b	-0.536*	-0.298	-0.239
	t (b)	-2.827	-1.052	-0.958
	p (b)	1.4E-02	3.2E-01	3.6E-01
	SE(b)	0.190	0.284	0.250
	Lower	-1.107	-1.220	-1.001
	Upper	0.035	0.623	0.523
Intercept Regression	a t (a) p (a) SE(a)	-0.126* -2.500 2.7E-02 0.050	-0.158* -2.931 1.7E-02 0.054	-0.122* -2.579 2.4E-02 0.047
	Lower	-0.278	-0.333	-0.266
	Upper	0.026	0.017	0.022
Model	R^{2} $s(e)$ $df(e)$	0.381 0.186 13	0.110 0.178 9	0.071 0.174 12
Correlation	r	-0.617**	-0.331	-0.266
	p(r)	7.1E-03	1.6E-01	1.8E-01
Descriptive Statistics	dL-dM	0.082	0.015	0.034
	SD (dL-dM)	0.262	0.199	0.193
	SE(dL-dM)	0.068	0.060	0.052
	V 95	-0.170	-0.162	-0.130
	SD (V 95)	0.228	0.179	0.173
	SE(V 95)	0.059	0.054	0.046
Observations	Firms (n)	16	15	16
	Years (T)	15	11	14
	Total ($n \times T$)	233	169	217

Table 4.16 (Continued)	ļ
PANEL A	

Time Series Regression Analysis for the Primary and Secondary Sample Total Liabilities

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The sample consists of a maximum of 233 observations from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008.

See main table notes for the regression model applied.

	N		Total Period	Sub-F	Periods
	Measure		1994-2008	1994-2004	1994-2007
	Slope	Ь	-0.275	0.042	-0.150
	1	t (b)	-0.275	0.042	-0.150
		p (b)	2.1E-01	8.7E-01	5.0E-01
		SE(b)	0.207	0.259	0.214
		Lower	-0.899	-0.799	-0.803
		Upper	0.349	0.884	0.503
I	ntercept	а	-0.148**	-0.179**	-0.133*
Regression	I I I	t (a)	-3.295	-3.482	-3.030
		p(a)	5.8E-03	6.9E-03	1.0E-02
		SE(a)	0.045	0.051	0.044
		Lower	-0.284	-0.346	-0.267
		Upper	-0.013	-0.012	0.001
	Model	R^2	0.119	0.003	0.040
		s(e)	0.173	0.163	0.164
		df(e)	13	9	12
~		r	-0.345	0.055	-0.199
Correlati	on	p(r)	1.0E-01	4.4E-01	2.5E-01
		dL-dM	0.030	0.060	0.008
		SD (dL-dM)	0.222	0.199	0.213
Degewinting St	tatistics	SE(dL-dM)	0.057	0.060	0.057
Descriptive St	ausucs	V 95	-0.156	-0.177	-0.134
		SD (V 95)	0.177	0.155	0.161
		SE(V 95)	0.046	0.047	0.043
		Firms (n)	12	12	12
Observati	ons	Years (T)	15	11	14
			100	122	1(0

Table 4.16 (Continued) PANEL B

Time Series Regression Analysis for the Control Group Total Liabilities to Market Price

Panel notes:

See main table notes for the regression model applied.

^{**, *} Regression coefficients significant at the 0.01 and 0.05 two-tailed levels.

The control consists of a maximum of 180 observations from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008.

	Maaguma		Total Period	Sub-F	Periods
	Measure		1994-2008	1994-2004	1994-2007
	Slope	Ь	-0.569**	-0.371	-0.248
	I I	t (b)	-5.017	-1.315	-1.149
		p (b)	2.4E-04	2.2E-01	2.7E-01
		SE(b)	0.113	0.282	0.216
		Lower	-0.910	-1.287	-0.908
		Upper	-0.227	0.546	0.412
	Intercept	а	-0.111*	-0.151*	-0.124*
Regression	1	t (a)	-2.246	-2.933	-2.663
		p (a)	4.3E-02	1.7E-02	2.1E-02
		SE(a)	0.049	0.052	0.047
		Lower	-0.259	-0.319	-0.267
		Upper	0.038	0.016	0.018
	Model	R^2	0.659	0.161	0.099
		s(e)	0.176	0.169	0.165
		df(e)	13	9	12
		r	-0.812**	-0 401	-0.315
Correla	tion	<i>p(r)</i>	1.2E-04	1.1E-01	1.4E-01
		dL-dM	0.166	0.028	0.072
		SD (dL-dM)	0.416	0.190	0.212
Descripting	Statistics	SE(dL-dM)	0.107	0.057	0.057
Descriptive S	naustics	V 95	-0.205	-0.162	-0.142
		SD (V 95)	0.291	0.175	0.167
		SE(V 95)	0.075	0.053	0.045
		Firms (n)	5	5	5
Observat	tions	Years (T)	15	11	14

Table 4.16 (Continued) PANEL C

Time Series Regression Analysis for the Primary Sample Total Liabilities to Market Price

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The sample consists of a maximum of 75 observations from the 5 banks in the LSE that adopted the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008. See main table notes for the regression model applied.

М		Total Period	Sub-F	Periods
IVIE	casure	1994-2008	1994-2004	1994-2007
Sle	ope b	-0.463	-0.319	-0.297
	t (b)	-1.934	-1.123	-1.146
	p(b)	7.5E-02	2.9E-01	2.7E-01
	SE(b)	0.240	0.284	0.259
	Lower	-1.185	-1.242	-1.090
	Upper	0.258	0.604	0.495
Inte	rcept a	-0.135*	-0.162*	-0.121*
Regression	t (a)	-2.509	-2.608	-2.282
	p(a)	2.6E-02	2.8E-02	4.2E-02
	SE(a)	0.054	0.062	0.053
	Lower	-0.298	-0.363	-0.283
	Upper	0.027	0.040	0.041
Ма	odel R^2	0.223	0.123	0.099
	s(e)	0.205	0.205	0.198
	df(e)	13	9	12
	r	-0 473*	-0.350	-0 314
Correlation	p(r)	3.8E-02	1.5E-01	1.4E-01
	dL-dM	0.043	0.008	0.016
	SD (dL-dM)	0.229	0.229	0.211
Descriptive Stati	SE(dL-dM)	0.059	0.069	0.056
	V 95	-0.155	-0.164	-0.126
	SD (V 95)	0.224	0.208	0.200
	SE(V 95)	0.058	0.063	0.053
	Firms (n)	11	11	11
Observations	Years (T)	15	11	14

Table 4.16 (Continued) PANEL D

Time Series Regression Analysis for the Secondary Sample Total Liabilities to Market

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The sample consists of a maximum of 165 observations from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008.

See main table notes for the regression model applied.

Analysis

The Table 4.16 Panel A time series regression slope for the Primary and Secondary sample shows that before 2005 there is no statistical significance for the total liabilities to market price relative delta variable and the Historical Value-at-Risk Actual variable. After 2005, the regression slope shows that a 1% change in the total liabilities to market price relative delta variable translates to just above a 0.5% (-0.536) increase in the Historical Value-at-Risk Actual variable.

The time series slopes suggest that for the Primary and Secondary sample the total liabilities to market price relative delta variable reacted significantly to the Historical Value-at-Risk Actual variable after the 2005 accounting change, and does not show a statistically significant reaction before the change. For the Control group the total liabilities to market price relative delta variable does not show a statistically significant reaction to the Historical Value-at-Risk Actual variable Actual variable does not show a statistically significant reaction to the Historical Value-at-Risk Actual variable before and after 2005.

4.6.4 Regressions for Net income to Market Price Relative Delta and Historical Value-at-Risk Actual

The regression analysis for the net income to market price relative delta and Historical Value-at-Risk Actual is presented for the time range 1994 to 2008 in Table 4.17 (and Table V.4). In the regressions, the Historical Value-at-Risk Actual variable is tested as the dependent variable, and the net income to market price relative delta variable is tested as the independent variable.

Table 4.17Time Series Regression Analysis for Net Income to Market Price
Relative Delta and Historical Value-at-Risk Actual

Time Series Regression

Table 4.17 Panel A to Panel D show the results for the time series regressions that test the net income to market price relative delta and Historical Value-at-Risk Actual by applying the regression specified in equation (3.40). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.

Time Series Regression Table Description

The Table 4.17 columns represent the following:

The Total Period column presents the time series regression results for the year range 1994 to 2008, with a single year represented from 1st January to 31st December, with the 31st December selected as the variable record date. The Sub-Periods column presents the time series regression results for the year range 1994 to 2004, and for 1994 to 2007, with a single year represented from 1st January to 31st December, and a 31st December variable record date.

The Measure column represents the time series regression results for the total and sub-time periods and is represented by the following data panels:

The *Regression* panel presents the coefficients and related statistics for the regression specified in equation (3.40), and follows the model:

 $V_{N,t} = a + b (dI - dM)_{N,t} + e_{N,t}$

Where: $V_{N,t}$ is the Value-at-Risk (V_t) for N firms at time t, $(dI - dM)_{N,t}$ is the Total Net Income to Market Price Relative Delta, $(dI_{t-1,t} - dM_{t-1,t})$, for N firms at time t, and $e_{N,t}$ is the regression error term. The *Lower* and *Upper* are the corresponding coefficient levels predicted at a 99% confidence level.

The *Correlation* panel presents the correlation statistics for the relationship between the dependent and independent variables. The *Descriptive Statistics* panel presents for the variables tested, the average sample means calculated using equation (M.6); the standard deviation, *SD*, of sample means for the time series using equation (M.7); and, the standard error of means, *SE*, for the time series. The *Observations* panel presents: *Firms (n)* (term *N* in the regression) that represents the average number of sample firms in the time series regression (for the Primary and Secondary sample, for the time period 1994 to 2004, the number of firms are rounded down); *Years (T)* represents the number of time series years; *Total (n × T)* represents the number of averaged pooled time series regression observations.

Measure		Total Period	Sub-Periods	
		1994-2008	1994-2004	1994-2007
Slope	b	0.076	0.119*	0.066*
	<i>t</i> (<i>b</i>)	2.054	2.500	2.490
	$p(\dot{b})$	6.1E-02	3.4E-02	2.8E-02
	SE(b)	0.037	0.048	0.026
	Lower	-0.035	-0.036	-0.015
	Upper	0.187	0.275	0.147
Intercept Regression	а	-0.191**	-0.143*	-0.151**
	t (a)	-3.542	-3.215	-3.773
	p (a)	3.6E-03	1.1E-02	2.7E-03
	SE(a)	0.054	0.044	0.040
	Lower	-0.354	-0.287	-0.274
	Upper	-0.029	0.002	-0.029
Model	R^2	0.245	0.410	0.341
	s(e)	0.205	0.145	0.147
	df(e)	13	9	12
Correlation	14	0.405*	0.64*	0.58/1*
	p(r)	3.0E-02	1.7E-02	1.4E-02
Descriptive Statistics	dI dM	0.281	-0.163	0.320
	SD (dI-dM)	1 489	0.961	1 538
	SE(dI-dM)	0.385	0.290	0.411
	V 95	-0.170	-0.162	-0.130
	SD (V 95)	0.228	0.179	0.173
	SE(V 95)	0.059	0.054	0.046
Observations	Firms (n)	16	15	16
	Years (T)	15	11	14
		222	1(0	017

Table 4.17 (Continued) PANEL A

Time Series Regression Analysis for the Primary and Secondary Sample Net Income to

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The sample consists of a maximum of 233 observations from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008.

See main table notes for the regression model applied.
Maagur		Total Period	Sub-F	Periods
Measure		1994-2008	1994-2004	1994-2007
Slope	b	-0.111	0.012	0.015
1	t (b)	-0.111	0.012	0.015
	p (b)	2.0E-01	9.2E-01	9.0E-01
	SE(b)	0.083	0.124	0.126
	Lower	-0.361	-0.390	-0.368
	Upper	0.139	0.415	0.399
Intercept	a	-0.151**	-0.176**	-0.133*
Regression	t (a)	-3.384	-3.528	-2.933
-	p(a)	4.9E-03	6.4E-03	1.3E-02
	SE(a)	0.045	0.050	0.045
	Lower	-0.286	-0.338	-0.272
	Upper	-0.017	-0.014	0.006
Model	R^2	0.122	0.001	0.001
	s(e)	0.172	0.163	0.168
	df(e)	13	9	12
		0.240	0.023	0.035
Correlation	p(r)	1.0E-01	4.6E-01	4.5E-01
	dI-dM	0.047	-0.068	-0.063
	SD (dI-dM)	0.555	0.415	0.370
Degewinting Statistics	SE(dI-dM)	0.143	0.125	0.099
Descriptive statistics	V 95	-0.156	-0.177	-0.134
	SD (V 95)	0.177	0.155	0.161
	SE(V 95)	0.046	0.047	0.043
	Firms (n)	12	12	12
Observations	Years (T)	15	11	14
	Total ($n \times T$)	180	132	168

Table 4.17 (Continued)
PANEL B

Time Series Regression Analysis for the Control Group Net Income to Market Price

Panel notes:

See main table notes for the regression model applied.

Regression coefficients and the related statistics are calculated using OLS regression.

^{**, *} Regression coefficients significant at the 0.01 and 0.05 two-tailed levels.

The control consists of a maximum of 180 observations from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008.

	M		Total Period	Sub-F	eriods
	Measure		1994-2008	1994-2004	1994-2007
	Slope	b	0.237	0.009	-0.003
		t (b)	1.386	0.070	-0.029
		p (b)	1.9E-01	9.5E-01	9.8E-01
		SE(b)	0.171	0.126	0.117
		Lower	-0.278	-0.400	-0.360
		Upper	0.752	0.417	0.354
	Intercept	а	-0.233**	-0.163*	-0.142*
Regression	1	t (a)	-3.082	-2.742	-2.818
		p (a)	8.7E-03	2.3E-02	1.6E-02
		SE(a)	0.076	0.059	0.050
		Lower	-0.460	-0.356	-0.295
		Upper	-0.005	0.030	0.012
	Model	R^2	0.129	0.001	7.1E-05
		s(e)	0.282	0.184	0.174
		df(e)	13	9	12
		r	0 359	0.023	-0.008
Correl	ation	p(r)	9.5E-02	4.7E-01	4.9E-01
		dI-dM	0.117	0.168	0.167
		SD (dI-dM)	0.441	0.464	0.412
Deservinting	Continution	SE(dI-dM)	0.114	0.140	0.110
Descriptive	SIGUSTICS	V 95	-0.205	-0.162	-0.142
		SD (V 95)	0.291	0.175	0.167
		SE(V 95)	0.075	0.053	0.045
		Firms (n)	5	5	5
Observ	ations	Years (T)	15	11	14
		$T_{\rm rest} = 1 \left(c_{\rm res} - T \right)$	75	55	70

Table 4.17 (Continued) PANEL C

Time Series Regression Analysis for the Primary Net Income to Market Price Relative

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels.

The sample consists of a maximum of 75 observations from the 5 banks in the LSE that adopted the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008. See main table notes for the regression model applied.

Regression coefficients and the related statistics are calculated using OLS regression.

M		Total Period	Sub-F	Periods
Measure		1994-2008	1994-2004	1994-2007
Slope	b	0.053	0.089*	0.05*
1	t (b)	2.156	2.302	2.352
	p (b)	5.0E-02	4.7E-02	3.7E-02
	SE(b)	0.025	0.038	0.021
	Lower	-0.021	-0.036	-0.015
	Upper	0.127	0.214	0.115
Intercept	a	-0.174**	-0.136*	-0.145**
Regression	t (a)	-3.330	-2.530	-3.105
	p (a)	5.4E-03	3.2E-02	9.1E-03
	SE(a)	0.052	0.054	0.047
	Lower	-0.331	-0.311	-0.288
	Upper	-0.017	0.039	-0.002
Model	R^2	0.263	0.371	0.316
	s(e)	0.2	0.174	0.172
	df(e)	13	9	12
	r	0 513*	0 609*	0.562*
Correlation	p(r)	2.5E-02	2.3E-02	1.8E-02
	dI-dM	0.354	-0.315	0.388
	SD (dI-dM)	2.170	1.431	2.248
Descriptive Statistics	SE(dI-dM)	0.560	0.431	0.601
Descriptive statistics	V 95	-0.155	-0.164	-0.126
	SD (V 95)	0.224	0.208	0.200
	SE(V 95)	0.058	0.063	0.053
	Firms (n)	11	11	11
	Years (T)	15	11	14
Observations				

Table 4.17 (Continued) PANEL D

Time Series Regression Analysis for the Secondary Sample Net Income to Market Price

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The sample consists of a maximum of 165 observations from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. Observations are taken on a yearly basis from 1994 to 2008.

See main table notes for the regression model applied.

Regression coefficients and the related statistics are calculated using OLS regression.

Analysis

The Table 4.17 time series regression slopes show that the net income to market price relative delta variable exhibits statistically significant explanatory power to the Historical Value-at-Risk Actual variable for the Primary and Secondary sample for the 1994 to 2004 and 1994 to 2007 time periods. The Control group does not exhibit statistical significance for the three time series periods.

The Table 4.17 panel A time series regression slope for the Primary and Secondary sample shows that before 2005, a 1% change in the net income to market price relative delta variable translates to just above a 0.1% (0.119) decrease in the Historical Value-at-Risk Actual variable. After 2005, the regression slope shows that for the 1994 to 2007 time period, a 1% change in the net income to market price relative delta variable relates to nearly a 0% (0.066) level in the Historical Value-at-Risk Actual variable.

The time series slopes suggest that for the Primary and Secondary sample the net income to market price relative delta variable reacted significantly to the Historical Value-at-Risk Actual variable before and just after the 2005 accounting change. For the Control group the net income to market price relative delta variable does not show a statistically significant reaction to the Historical Value-at-Risk Actual variable before and after 2005.

4.7 Accounting Value-at-Risk Estimates and Relationships

The analysis and detailed results for the *accounting Value-at-Risk*⁴⁶ estimates are presented in this section. In addition, an analysis for the relationship between accounting Value-at-Risk, market price return Value-at-Risk measured using the Historical Value-at-Risk Actual, and the relative delta is also presented.

4.7.1 Estimates for Accounting Value-at-Risk

The analysis and results for the accounting Value-at-Risk estimates are presented on a yearly basis for the time range 1994 to 2008 in Table 4.19. In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the

⁴⁶ This study refers to the *accounting Value-at-Risk* measure also as the *change in accounting total Value-at-Risk* or the *change in accounting totals Value-at-Risk*.

Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample. Table 4.18 presents relative percentage change results for Table 4.19.

Table 4.18Percentage Changes for Accounting Value-at-Risk Results from
Table 4.19

Percentage Change Analysis Table Description

Accounting Value-at-Risk estimated levels in Table 4.19 for the samples and the control group, for the time period: 1994 to 2004 - before the 2005 accounting change, and the time periods: 1994 to 2007; and 1994 to 2008 - after the accounting change, analysed using relative percentage changes are presented in Table 4.18.

In Table 4.18, the Accounting Value-at-Risk column represents the Description and Symbol columns that describes and *codes* the accounting Value-at-Risk measures. The Year column represents the beginning (From) and ending (To) year applied to calculate the percentage changes presented in the Percentage Change column. The Percentage Change column represents the accounting Value-at-Risk percentage changes rounded to the nearest whole number. The PS column presents the percentage changes for the Primary and Secondary sample. The C column presents the percentage changes for the Control group. The P column presents the percentage changes for the S column presents the per

Accounting Value	e-at-Risk	Year	Ре	Percentage Change (%)					
Description	Symbol	From - To	PS	С	Р	S			
Total Equity Value-at-Risk	V_E	1994 - 2004 1994 - 2007 1994 - 2008	-228 -36 -148	52 -308 -1,436	-127 -17 -16	-328 -59 -286			
Total Assets Value-at-Risk	V_A	1994 - 2004 1994 - 2007 1994 - 2008	-412 292 -208	-2,100 -6,600 -21,950	-61 133 283	-4,433 1,533 -6,967			
Total Liabilities Value-at-Risk	V_L	1994 - 2004 1994 - 2007 1994 - 2008	-198 44 -81	-139 -216 -148	-56 140 305	-255 7 -234			
Net Income Value-at-Risk	V _I	1994 - 2004 1994 - 2007 1994 - 2008	-80 44 -348	170 6 3,500	-62 -97 -237	-217 970 -1,001			

Table Description

The Table 4.19 columns represent the following:

The Year column represents the fiscal accounting year. Record Date is the date the variable values are recorded. *Obs.* represents the number of sample firms observed for the year.

For the Relative Delta Measures column:

E-M column represents the total equity to market price relative delta for the *year* time period t - 1 to t calculated by applying equation (3.38); *A-M* column represents the total assets to market price relative delta for the *year* time period t - 1 to t calculated by applying equation (3.38); *L-M* column represents the total liabilities to market price relative delta for the *year* time period t - 1 to t calculated by applying equation (3.38); *L-M* column represents the total liabilities to market price relative delta for the *year* time period t - 1 to t calculated by applying equation (3.38); and, *I-M* column represents the net income to market price relative delta for the *year* time period t - 1 to t calculated by applying equation (3.38).

The V 95 column represents the Historical Value-at-Risk Actual, V_M , calculated at the 95% confidence level by applying equation (N.34).

For the Accounting V Estimates column:

 V_E and $(E-M)+V_M$ column represents the estimates for the Total Equity Value-at-Risk Actual at the 95% confidence level for the *year* time period t - 1 to t calculated by applying equation (3.44), and takes the general form:

$$V_{E:i,t} = \delta E M_{i,t} + V_{M:i,t}$$

Where: $V_{E:i,t}$ is the Total Equity Value-at-Risk $(V_{E:t})$ for the i^{th} firms at time t, $\delta EM_{i,t}$ is the Total Equity to Market Price Relative Delta $(dE_{t-1,t} - dM_{t-1,t})$ for the i^{th} firm at time t, and $V_{M:i,t}$ is the Market Price Value-at-Risk (V_t) for the i^{th} firm at time t.

 V_A and $(A-M)+V_M$ column represents the estimates for the Total Assets Value-at-Risk Actual at the 95% confidence level for the *year* time period t - 1 to t calculated by applying equation (3.44), and takes the general form:

$$V_{A:i,t} = \delta A M_{i,t} + V_{M:i,t}$$

Where: $V_{A:i,t}$ is the Total Assets Value-at-Risk $(V_{A:t})$ for the i^{th} firms at time t, $\delta AM_{i,t}$ is the Total Assets to Market Price Relative Delta $(dA_{t-1,t} - dM_{t-1,t})$ for the i^{th} firm at time t, and $V_{M:i,t}$ is the Market Price Value-at-Risk (V_t) for the i^{th} firm at time t.

 V_L and $(L-M)+V_M$ column represents the estimates for the Total Liabilities Value-at-Risk Actual at the 95% confidence level for the *year* time period t - 1 to t calculated by applying equation (3.44), and takes the general form:

$$V_{L:i,t} = \delta L M_{i,t} + V_{M:i,t}$$

Where: $V_{L:i,t}$ is the Total Liabilities Value-at-Risk $(V_{L:t})$ for the i^{th} firms at time t, $\delta LM_{i,t}$ is the Total Liabilities to Market Price Relative Delta $(dL_{t-1,t} - dM_{t-1,t})$ for the i^{th} firm at time t, and $V_{M:i,t}$ is the Market Price Value-at-Risk (V_t) for the i^{th} firm at time t.

 V_I and $(I-M)+V_M$ column represents the estimates for the Net Income Value-at-Risk Actual at the 95% confidence level for the *year* time period t - 1 to t calculated by applying equation (3.44), and takes the general form:

$$V_{I:i,t} = \delta I M_{i,t} + V_{M:i,t}$$

Where: $V_{I:i,t}$ is the Net Income Value-at-Risk $(V_{I:t})$ for the i^{th} firms at time t, $\delta IM_{i,t}$ is the Net Income to Market Price Relative Delta $(dI_{t-1,t} - dM_{t-1,t})$ for the i^{th} firm at time t, and $V_{M:i,t}$ is the Market Price Value-at-Risk (V_t) for the i^{th} firm at time t.

	Table 4.19 (Continued)												
					PA	ANEL A							
	Estimates for Acc	ounting Va	lue-at-Risk f	or the Prima	ary and Seco	ondary Samp	le from Relat	ive Delta and F	listorical Valu	ie-at-Risk Act	tual		
				Relative Del	lta Measures		V 95	Accounting V Estimates					
Year	Record Date	Obs.	E M	л М	I M	I M	 V	V_E	V_A	V_L	V _I		
			<i>E-1v</i> 1	<i>A-M</i>	L-M	1-1/1	* M	$(E-M)+V_M$	$(A-M)+V_M$	$(L-M)+V_M$	$(I-M)+V_M$		
1994	30-Dec-1994	15	0.058	-0.014	0.068	0.360	0.040	0.098	0.026	0.108	0.401		
1995	29-Dec-1995	16	0.196	-0.007	0.003	0.295	-0.183	0.012	-0.190	-0.181	0.111		
1996	31-Dec-1996	16	0.083	0.012	-0.024	0.246	0.014	0.096	0.026	-0.011	0.260		
1997	31-Dec-1997	16	-0.033	-0.137	-0.194	-0.290	0.007	-0.026	-0.130	-0.187	-0.283		
1998	31-Dec-1998	16	0.037	0.036	0.035	0.311	-0.104	-0.068	-0.068	-0.070	0.207		
1999	31-Dec-1999	16	-0.085	-0.165	-0.263	-0.052	-0.107	-0.192	-0.272	-0.370	-0.159		
2000	29-Dec-2000	16	0.244	-0.020	0.025	-0.004	-0.135	0.109	-0.155	-0.110	-0.139		
2001	31-Dec-2001	16	0.244	0.222	0.289	-0.079	-0.296	-0.053	-0.074	-0.007	-0.376		
2002	31-Dec-2002	16	0.354	0.363	0.412	0.239	-0.414	-0.060	-0.051	-0.002	-0.175		
2003	31-Dec-2003	16	-0.123	-0.178	-0.173	-2.994	-0.517	-0.640	-0.695	-0.690	-3.512		
2004	31-Dec-2004	16	-0.034	0.009	-0.016	0.172	-0.090	-0.125	-0.081	-0.106	0.081		
2005	30-Dec-2005	16	0.066	0.104	0.152	4.606	0.014	0.080	0.118	0.166	4.619		
2006	29-Dec-2006	16	-0.027	-0.041	-0.101	0.980	0.063	0.036	0.023	-0.038	1.043		
2007	31-Dec-2007	16	0.176	0.215	0.268	0.690	-0.113	0.063	0.102	0.155	0.577		
2008	31-Dec-2008	16	0.682	0.702	0.749	-0.266	-0.730	-0.047	-0.028	0.020	-0.996		

Panel notes: The *Relative Delta Measures* and the *V 95* measure are based on sample averages. The sample consists of a maximum of 16 observations per year from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.

						Table 4.	19 (Continue	d)					
		Estimates	s for Accou	nting Value-	at-Risk for 1	P. the Control	ANEL B Group from 1	Relative Delta	and Historica	l Value-at-Ris	k Actual		
					Relative Delta Measures				Accounting V Estimates				
	Year	Record Date	Obs.	 	4-M	I_M	I_M		V_E	V_A	V_L	V_I	
				<i>L</i> /-1 <i>v</i> 1	21-101	L-1 VI	1-111	• M	$(E-M)+V_M$	$(A-M)+V_M$	$(L-M)+V_M$	$(I-M)+V_M$	
	1994	30-Dec-1994	12	0.073	0.100	0.360	0.065	-0.098	-0.025	0.002	0.262	-0.033	
	1995	29-Dec-1995	12	0.062	0.024	0.210	-0.022	-0.188	-0.126	-0.164	0.022	-0.210	
	1996	31-Dec-1996	12	0.138	0.069	0.006	0.074	-0.031	0.107	0.038	-0.025	0.044	
	1997	31-Dec-1997	12	0.025	-0.006	-0.076	-0.040	-0.049	-0.023	-0.054	-0.125	-0.089	
)	1998	31-Dec-1998	12	0.066	0.065	0.320	0.354	-0.240	-0.174	-0.174	0.080	0.114	
	1999	31-Dec-1999	12	0.091	-0.069	-0.112	-1.257	-0.172	-0.081	-0.242	-0.284	-1.430	
	2000	29-Dec-2000	12	-0.048	-0.055	0.076	0.048	-0.016	-0.064	-0.072	0.060	0.031	
	2001	31-Dec-2001	12	-0.013	-0.013	-0.014	0.184	-0.303	-0.316	-0.316	-0.317	-0.119	
	2002	31-Dec-2002	12	0.034	0.032	0.241	-0.054	-0.383	-0.349	-0.351	-0.142	-0.436	
	2003	31-Dec-2003	12	0.039	-0.050	-0.258	-0.135	-0.456	-0.416	-0.505	-0.713	-0.591	
	2004	31-Dec-2004	12	-0.005	-0.033	-0.094	0.031	-0.007	-0.012	-0.040	-0.101	0.023	
	2005	30-Dec-2005	12	-0.048	-0.121	-0.306	-0.229	0.057	0.009	-0.064	-0.249	-0.172	
	2006	29-Dec-2006	12	-0.050	-0.067	0.011	0.087	0.054	0.005	-0.012	0.065	0.142	
	2007	31-Dec-2007	12	-0.052	-0.080	-0.254	0.019	-0.050	-0.102	-0.130	-0.304	-0.031	
	2008	31-Dec-2008	12	0.080	0.028	0.338	1.586	-0.464	-0.384	-0.437	-0.127	1.122	

Panel notes: The *Relative Delta Measures* and the *V* 95 measure are based on the Control group averages. The Control group consists of a maximum of 12 observations per year from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.

					Table 4.	19 (Continue	d)					
	Estimates	for Accoun	nting Value-a	nt-Risk for t	P. he Primary S	ANEL C Sample from	Relative Delt	a and Historica	l Value-at-Ri	sk Actual		
				Relative Delta Measures				Accounting V Estimates				
Year	Record Date	Obs.	E-M	A-M	I-M	I-M	 	V_E	V_A	V_L	VI	
						1 1/1	, M	$(E-M)+V_M$	$(A-M)+V_M$	$(L-M)+V_M$	$(I-M)+V_M$	
1994	30-Dec-1994	4	0.185	0.118	0.117	1.229	-0.019	0.166	0.099	0.097	1.210	
1995	29-Dec-1995	5	0.044	-0.112	-0.117	0.050	-0.222	-0.178	-0.335	-0.339	-0.172	
1996	31-Dec-1996	5	-0.072	-0.106	-0.108	0.021	0.118	0.046	0.012	0.010	0.139	
1997	31-Dec-1997	5	-0.231	-0.183	-0.179	-0.375	0.056	-0.175	-0.127	-0.123	-0.319	
1998	31-Dec-1998	5	0.064	0.048	0.048	0.069	-0.371	-0.307	-0.323	-0.323	-0.303	
1999	31-Dec-1999	5	-0.111	-0.209	-0.213	-0.184	-0.116	-0.227	-0.325	-0.328	-0.300	
2000	29-Dec-2000	5	1.086	0.345	0.334	0.453	-0.312	0.773	0.032	0.021	0.140	
2001	31-Dec-2001	5	0.158	0.127	0.125	-0.033	-0.162	-0.004	-0.035	-0.037	-0.195	
2002	31-Dec-2002	5	0.205	0.313	0.321	-0.334	-0.323	-0.117	-0.010	-0.001	-0.656	
2003	31-Dec-2003	5	-0.062	-0.105	-0.108	0.447	-0.382	-0.444	-0.487	-0.490	0.065	
2004	31-Dec-2004	5	0.002	0.085	0.089	0.504	-0.045	-0.044	0.039	0.043	0.458	
2005	30-Dec-2005	5	0.095	0.267	0.277	0.251	-0.007	0.089	0.261	0.271	0.244	
2006	29-Dec-2006	5	0.007	-0.028	-0.030	-0.021	0.017	0.025	-0.011	-0.013	-0.003	
2007	31-Dec-2007	5	0.363	0.456	0.458	0.259	-0.225	0.138	0.231	0.233	0.034	
2008	31-Dec-2008	5	1.222	1.462	1.476	-0.576	-1.083	0.139	0.379	0.393	-1.659	
	Year 1994 1995 1996 1997 1998 1999 2000 2001 2002 2003 2004 2005 2006 2007 2008	Estimates Year Record Date 1994 30-Dec-1994 1995 29-Dec-1995 1996 31-Dec-1996 1997 31-Dec-1997 1998 31-Dec-1998 1999 31-Dec-1998 1999 31-Dec-2001 2000 29-Dec-2000 2001 31-Dec-2001 2002 31-Dec-2002 2003 31-Dec-2003 2004 31-Dec-2005 2006 29-Dec-2006 2007 31-Dec-2007 2008 31-Dec-2008	Estimates for Accourt Year Record Date Obs. 1994 30-Dec-1994 4 1995 29-Dec-1995 5 1996 31-Dec-1996 5 1997 31-Dec-1997 5 1998 31-Dec-1998 5 1999 31-Dec-1999 5 2000 29-Dec-2000 5 2001 31-Dec-2001 5 2002 31-Dec-2002 5 2003 31-Dec-2003 5 2004 31-Dec-2005 5 2005 30-Dec-2006 5 2006 29-Dec-2007 5 2007 31-Dec-2007 5 2008 31-Dec-2008 5	Estimates for Accounting Value-aYearRecord DateObs. $E-M$ 199430-Dec-199440.185199529-Dec-1995199631-Dec-199631-Dec-199650.044199631-Dec-19975-0.072199731-Dec-199850.064199931-Dec-19995-0.111200029-Dec-200051.086200131-Dec-200150.158200231-Dec-200250.205200331-Dec-20035-0.062200431-Dec-200450.002200530-Dec-200550.095200629-Dec-200650.007200731-Dec-200750.363200831-Dec-200851.222	Estimates for Accounting Value-at-Risk for the Relative Determination of the Relative Determinati	Table 4.P.Estimates for Accounting Value-at-Risk for the Primary 9Relative Delta MeasuresYearRecord DateObs. $E-M$ A-ML-M199430-Dec-199440.1850.1180.117199529-Dec-199550.044-0.112-0.117199631-Dec-19965-0.072-0.106-0.108199731-Dec-19975-0.231-0.183-0.179199831-Dec-199850.0640.0480.048199931-Dec-19995-0.111-0.209-0.213200029-Dec-200051.0860.3450.334200131-Dec-200150.1580.1270.125200231-Dec-20035-0.062-0.105-0.108200431-Dec-200450.0020.0850.089200530-Dec-200550.007-0.028-0.030200731-Dec-200750.3630.4560.458200831-Dec-200851.2221.4621.476	Table 4.19 (ContinuePANEL CEstimates for Accounting Value-at-Risk for the Primary Sample fromYearRecord DateObs.Relative Delta Measures1994 30 -Dec-19944 0.185 0.118 0.117 1.229 199529-Dec-19955 0.044 -0.112 -0.117 0.050 1996 31 -Dec-19965 -0.072 -0.106 -0.108 0.211 1997 31 -Dec-19975 -0.231 -0.183 -0.179 -0.375 1998 31 -Dec-19985 0.064 0.048 0.048 0.069 1999 31 -Dec-19995 -0.111 -0.209 -0.213 -0.184 200029-Dec-20005 1.086 0.345 0.334 0.453 2001 31 -Dec-20015 0.205 0.313 0.321 -0.334 2003 31 -Dec-20045 0.002 0.085 0.089 0.504 2004 30 -Dec-20055 0.095 0.267 0.277 0.251 2006 29 -Dec-20065 0.007 -0.028 -0.030 -0.021 2007 31 -Dec-20075 0.363 0.456 0.458 0.259 2008 31 -Dec-20085 1.222 1.462 1.476 -0.576	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{tabular}{ c c c c c } \hline Table 4.19 (Continued) \\ \hline PANEL C \\ \hline Estimates for Accounting Value-at-Risk for the Primary Sample from Relative Delta and Historica \\ \hline Panel C \\ \hline Estimates for Accounting Value-at-Risk for the Primary Sample from Relative Delta and Historica \\ \hline Panel C \\ $	Table 4.19 (Continued) PANEL C Estimates for Accounting Value-at-Risk for the Primary Sample from Relative Delta and Historical Value-at-Risk Year Record Date Obs. Relative Delta Measures V 95 Accounting 1994 30-Dec-1994 4 0.185 0.118 0.117 1.229 -0.019 0.166 0.099 1995 29-Dec-1995 5 0.044 -0.112 -0.117 0.050 -0.222 -0.178 -0.335 1996 31-Dec-1996 5 -0.072 -0.106 -0.108 0.021 0.118 0.046 0.012 1997 31-Dec-1998 5 0.064 0.048 0.048 0.069 -0.375 0.056 -0.175 -0.127 1998 31-Dec-1999 5 -0.111 -0.209 -0.213 -0.184 -0.116 -0.227 -0.323 1999 31-Dec-1999 5 -0.0111 -0.209 -0.213 -0.184 -0.116 -0.227 -0.323 101	Table 4.19 (Continued) FANEL C Setimates for Accounting Value-at-Risk for the Primary Sample from Relative Delta and Historical Value-at-Risk Actual Year Record Date Obs. $E-M$ A-M L-M V 95 Accounting V Estimates 1994 30-Dec-1994 4 0.118 0.117 1.229 -0.019 0.166 0.099 0.097 1994 30-Dec-1994 4 0.118 0.117 1.229 -0.019 0.166 0.099 0.097 1995 29-Dec-1995 5 0.044 -0.112 -0.117 0.050 -0.222 -0.178 -0.335 -0.339 -0.198 0.127 -0.123 1996 31-Dec-1997 5 -0.231 -0.183 -0.179 -0.375 0.056 -0.175 -0.127 -0.123 -0.338 -0.332 -0.323 -0.323	

Panel notes: The *Relative Delta Measures* and the *V*95 measure are based on sample averages. The sample consists of a maximum of 5 observations per year from the 5 banks in the LSE that adopted the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.

					Table 4.	19 (Continue	d)					
			• • • •		P.	ANEL D						
	Estimates 1	or Account	ing value-at	-RISK for th	e Secondary	Sample from	h Relative Del	ta and Historic	al value-at-k	ISK ACTUAI		
				Relative De	lta Measures		V 95		Accounting V Estimates			
Year	Record Date	Obs.	E M	4 M	I M	I M		V_E	V_A	V_L	V_I	
			<i>E-W</i>	A-M	L-111	1-1/1	<i>V</i> M	$(E-M)+V_M$	$(A-M)+V_M$	$(L-M)+V_M$	$(I-M)+V_M$	
1994	30-Dec-1994	11	0.007	-0.067	0.048	0.013	0.064	0.071	-0.003	0.112	0.077	
1995	29-Dec-1995	11	0.256	0.035	0.051	0.392	-0.168	0.088	-0.133	-0.117	0.225	
1996	31-Dec-1996	11	0.139	0.055	0.006	0.328	-0.024	0.115	0.031	-0.018	0.304	
1997	31-Dec-1997	11	0.039	-0.121	-0.200	-0.259	-0.011	0.029	-0.131	-0.210	-0.270	
1998	31-Dec-1998	11	0.027	0.032	0.030	0.399	-0.007	0.020	0.025	0.023	0.392	
1999	31-Dec-1999	11	-0.073	-0.146	-0.286	0.008	-0.103	-0.176	-0.249	-0.389	-0.095	
2000	29-Dec-2000	11	-0.139	-0.185	-0.116	-0.212	-0.054	-0.193	-0.240	-0.170	-0.266	
2001	31-Dec-2001	11	0.283	0.265	0.364	-0.100	-0.357	-0.075	-0.092	0.007	-0.458	
2002	31-Dec-2002	11	0.422	0.386	0.454	0.500	-0.456	-0.034	-0.070	-0.002	0.044	
2003	31-Dec-2003	11	-0.150	-0.211	-0.202	-4.559	-0.579	-0.729	-0.790	-0.781	-5.138	
2004	31-Dec-2004	11	-0.051	-0.025	-0.063	0.021	-0.111	-0.162	-0.136	-0.174	-0.090	
2005	30-Dec-2005	11	0.053	0.030	0.095	6.586	0.023	0.076	0.053	0.118	6.608	
2006	29-Dec-2006	11	-0.043	-0.046	-0.134	1.434	0.084	0.042	0.038	-0.049	1.519	
2007	31-Dec-2007	11	0.090	0.105	0.181	0.885	-0.061	0.029	0.043	0.120	0.824	
2008	31-Dec-2008	11	0.437	0.356	0.419	-0.126	-0.569	-0.132	-0.212	-0.150	-0.694	

Panel notes: The *Relative Delta Measures* and the *V* 95 measure are based on sample averages. The sample consists of a maximum of 11 observations per year from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. The highlighted 2005 record identified the IFRS implementation event year for the adopting firms.

Analysis

Table 4.19 presents estimates for the accounting Historical Value-at-Risk Actual measure, referred to as the accounting Value-at-Risk. The accounting Value-at-Risk measure estimated in this study by definition is an equivalent measurement to the market price return Historical Value-at-Risk Actual specified in Appendix N.8.

The Table 4.19 Primary and Secondary sample accounting Value-at-Risk estimates suggest that, after the 2005 accounting change compared to before, there was a marginally significant decrease in average accounting Value-at-Risk estimates. The only exception is the net income Value-at-Risk estimates that shows a significant material increase. The market price return Historical Value-at-Risk Actual generally exhibits varying levels after 2005. However, materially significant increases are reported after 2005.

The Control group estimates of accounting Value-at-Risk exhibit varying levels after 2005, with a significant material decrease reported for the net income Value-at-Risk in 2008. The market price return Historical Value-at-Risk Actual also exhibits varying levels. However, a materially significant increase after 2005, in 2008, is reported.

Further examination of these results suggest that generally for the Primary and Secondary sample there is a wider and materially significant difference between accounting Value-at-Risk and market price Value-at-Risk after the 2005 accounting change than before. The Control group, however, shows that the accounting Value-at-Risk better reflects the market price Value-at-Risk after 2005 compared to before. The exception for both the samples and the control group is the difference between net income Value-at-Risk and market price Value-at-Risk that exhibits a *widening* after 2005. The evidence from the relative percentage changes for before and after 2005, presented in Table 4.18, suggest support for these findings.

4.7.2 Relationship Between Accounting Value-at-Risk, Historical Value-at-Risk Actual and Relative Delta

The arithmetic difference between accounting Value-at-Risk and market price Value-at-Risk and the relationship of this *arithmetic difference* to the relative delta measure is presented, on a yearly basis for the time range 1994 to 2008, in Table 4.20. In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.

Table 4.20Relationship Between the Difference in Accounting Value-at-Risk
and Historical Value-at-Risk Actual to Relative Delta

Table Description

The Table 4.20 Panel A to Panel D columns represent the following:

The Year column represents the fiscal accounting year. Record Date is the date the variable values are recorded. *Obs.* represents the number of sample firms observed for the year.

For the *Accounting V* column:

 V_E column represents the estimates for the Total Equity Value-at-Risk Actual at the 95% confidence level for the *year* time period t - 1 to t calculated by applying equation (3.44); V_A column represents the estimates for the Total Assets Value-at-Risk Actual at the 95% confidence level for the *year* time period t - 1 to t calculated by applying equation (3.44); V_L column represents the estimates for the Total Liabilities Value-at-Risk Actual at the 95% confidence level for the year time period t - 1 to t calculated by applying equation (3.44); V_L column represents the estimates for the Total Liabilities Value-at-Risk Actual at the 95% confidence level for the year time period t - 1 to t calculated by applying equation (3.44); and, V_I column represents the estimates for the Net Income Value-at-Risk Actual at the 95% confidence level for the *year* time period t - 1 to t calculated by applying equation (3.44).

The V 95 column represents the Historical Value-at-Risk Actual, V_M , calculated at the 95% confidence level by applying equation (N.34).

 V_E - V_M and E-M column represents the Total Equity to Market Price Relative Delta for the *year* time period t - 1 to t calculated by applying equation (3.45), and takes the general form:

$$V_{E:i,t} - V_{M:i,t} = \delta E M_{i,t}$$

Where: $V_{E:i,t}$ is the Total Equity Value-at-Risk $(V_{E:t})$ for the i^{th} firms at time t, $V_{M:i,t}$ is the Market Price Value-at-Risk (V_t) for the i^{th} firms at time t, and $\delta EM_{i,t}$ is the Total Equity to Market Price Relative Delta $(dE_{t-1,t} - dM_{t-1,t})$ for the i^{th} firm at time t.

 V_A - V_M and A-M column represents the Total Assets to Market Price Relative Delta for the *year* time period t - 1 to t calculated by applying equation (3.45), and takes the general form:

$$V_{A:i,t} - V_{M:i,t} = \delta A M_{i,t}$$

Where: $V_{A:i,t}$ is the Total Assets Value-at-Risk $(V_{A:t})$ for the *i*th firms at time *t*, $V_{M:i,t}$ is the Market Price Value-at-Risk (V_t) for the *i*th firms at time *t*, and $\delta AM_{i,t}$ is the Total Assets to Market Price Relative Delta $(dA_{t-1,t} - dM_{t-1,t})$ for the *i*th firm at time *t*.

 V_L - V_M and *L*-*M* column represents the Total Liabilities to Market Price Relative Delta for the *year* time period t - 1 to t calculated by applying equation (3.45), and takes the general form: $V_{L,i,t} - V_{M,i,t} = \delta L M_{i,t}$

 $V_{L:i,t} - V_{M:i,t} = \delta LM_{i,t}$ Where: $V_{L:i,t}$ is the Total Liabilities Value-at-Risk ($V_{L:t}$) for the i^{th} firms at time t, $V_{M:i,t}$ is the Market Price Value-at-Risk (V_t) for the i^{th} firms at time t, and $\delta LM_{i,t}$ is the Total Liabilities to Market Price Relative Delta ($dL_{t-1,t} - dM_{t-1,t}$) for the i^{th} firm at time t.

 V_I and *I-M* column represents the Net Income to Market Price Relative Delta for the *year* time period t - 1 to t calculated by applying equation (3.45), and takes the general form:

$$V_{I:i,t} - V_{M:i,t} = \delta I M_{i,t}$$

Where: $V_{I:i,t}$ is the Net Income Value-at-Risk $(V_{I:t})$ for the i^{th} firms at time t, $V_{M:i,t}$ is the Market Price Value-at-Risk (V_t) for the i^{th} firms at time t, and $\delta IM_{i,t}$ is the Net Income to Market Price Relative Delta $(dI_{t-1,1} - dM_{t-1,t})$ for the i^{th} firm at time t.

				r -	Fable 4.20 (C	Continued)						
Relatio	PANEL A Relationship for the Primary and Secondary Sample Between the Difference in Accounting Value-at-Risk and Historical Value-at-Risk Actual to Relative Delta											
				Accou	nting V		V 95		Accounting V	V - Market V		
Year	Record Date	Obs.		IZ.	IZ.	IZ.		$V_E - V_M$	V_A - V_M	$V_L - V_M$	$V_I - V_M$	
			<i>V</i> _E	VA	V _L	VI	V _M	E-M	A-M	L-M	I-M	
1994	30-Dec-1994	15	0.098	0.026	0.108	0.401	0.040	0.058	-0.014	0.068	0.360	
1995	29-Dec-1995	16	0.012	-0.190	-0.181	0.111	-0.183	0.196	-0.007	0.003	0.295	
1996	31-Dec-1996	16	0.096	0.026	-0.011	0.260	0.014	0.083	0.012	-0.024	0.246	
1997	31-Dec-1997	16	-0.026	-0.130	-0.187	-0.283	0.007	-0.033	-0.137	-0.194	-0.290	
1998	31-Dec-1998	16	-0.068	-0.068	-0.070	0.207	-0.104	0.037	0.036	0.035	0.311	
1999	31-Dec-1999	16	-0.192	-0.272	-0.370	-0.159	-0.107	-0.085	-0.165	-0.263	-0.052	
2000	29-Dec-2000	16	0.109	-0.155	-0.110	-0.139	-0.135	0.244	-0.020	0.025	-0.004	
2001	31-Dec-2001	16	-0.053	-0.074	-0.007	-0.376	-0.296	0.244	0.222	0.289	-0.079	
2002	31-Dec-2002	16	-0.060	-0.051	-0.002	-0.175	-0.414	0.354	0.363	0.412	0.239	
2003	31-Dec-2003	16	-0.640	-0.695	-0.690	-3.512	-0.517	-0.123	-0.178	-0.173	-2.994	
2004	31-Dec-2004	16	-0.125	-0.081	-0.106	0.081	-0.090	-0.034	0.009	-0.016	0.172	
2005	30-Dec-2005	16	0.080	0.118	0.166	4.619	0.014	0.066	0.104	0.152	4.606	
2006	29-Dec-2006	16	0.036	0.023	-0.038	1.043	0.063	-0.027	-0.041	-0.101	0.980	
2007	31-Dec-2007	16	0.063	0.102	0.155	0.577	-0.113	0.176	0.215	0.268	0.690	
2008	31-Dec-2008	16	-0.047	-0.028	0.020	-0.996	-0.730	0.682	0.702	0.749	-0.266	

Panel notes: The *Accounting V* and *V95* measures are based on sample averages. The sample consists of a maximum of 16 observations per year from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.

				Т	able 4.20 (C	continued)					
Rolati	anshin for the Ca	ntrol Groun	Rotwoon the	Difference i	PANE n Accountin	L B g Value_at_B	Pick and Histo	rical Value_a	t-Risk Actu	al to Rolativ	va Dalta
Relati			between the	Accou	nting V		V 95		Accounting	V - Market V	
Year	Record Date	Obs.	V	V	V	V		$V_E - V_M$	V_A - V_M	V_L - V_M	$V_I - V_M$
			VE	V _A	V _L	V _I	V _M	E-M	A-M	L-M	I-M
1994	30-Dec-1994	12	-0.025	0.002	0.262	-0.033	-0.098	0.073	0.100	0.360	0.065
1995	29-Dec-1995	12	-0.126	-0.164	0.022	-0.210	-0.188	0.062	0.024	0.210	-0.022
1996	31-Dec-1996	12	0.107	0.038	-0.025	0.044	-0.031	0.138	0.069	0.006	0.074
1997	31-Dec-1997	12	-0.023	-0.054	-0.125	-0.089	-0.049	0.025	-0.006	-0.076	-0.040
1998	31-Dec-1998	12	-0.174	-0.174	0.080	0.114	-0.240	0.066	0.065	0.320	0.354
1999	31-Dec-1999	12	-0.081	-0.242	-0.284	-1.430	-0.172	0.091	-0.069	-0.112	-1.257
2000	29-Dec-2000	12	-0.064	-0.072	0.060	0.031	-0.016	-0.048	-0.055	0.076	0.048
2001	31-Dec-2001	12	-0.316	-0.316	-0.317	-0.119	-0.303	-0.013	-0.013	-0.014	0.184
2002	31-Dec-2002	12	-0.349	-0.351	-0.142	-0.436	-0.383	0.034	0.032	0.241	-0.054
2003	31-Dec-2003	12	-0.416	-0.505	-0.713	-0.591	-0.456	0.039	-0.050	-0.258	-0.135
2004	31-Dec-2004	12	-0.012	-0.040	-0.101	0.023	-0.007	-0.005	-0.033	-0.094	0.031
2005	30-Dec-2005	12	0.009	-0.064	-0.249	-0.172	0.057	-0.048	-0.121	-0.306	-0.229
2006	29-Dec-2006	12	0.005	-0.012	0.065	0.142	0.054	-0.050	-0.067	0.011	0.087
2007	31-Dec-2007	12	-0.102	-0.130	-0.304	-0.031	-0.050	-0.052	-0.080	-0.254	0.019
2008	31-Dec-2008	12	-0.384	-0.437	-0.127	1.122	-0.464	0.080	0.028	0.338	1.586

Panel notes: The *Accounting V* and *V95* measures are based on the Control group averages. The Control group consists of a maximum of 12 observations per year from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.

				Т	able 4.20 (C	continued)					
Relatio	PANEL C Relationship for the Primary Sample Between the Difference in Accounting Value-at-Risk and Historical Value-at-Risk Actual to Relative Delta										
				Accou	nting V		V 95	2	Accounting	V - Market V	7
Year	Record Date	Obs.	V	V	V	V		$V_E - V_M$	V_A - V_M	$V_L - V_M$	$V_I - V_M$
			VE	V _A	V L	V I	V _M	E-M	A-M	L-M	I-M
1994	30-Dec-1994	4	0.166	0.099	0.097	1.210	-0.019	0.185	0.118	0.117	1.229
1995	29-Dec-1995	5	-0.178	-0.335	-0.339	-0.172	-0.222	0.044	-0.112	-0.117	0.050
1996	31-Dec-1996	5	0.046	0.012	0.010	0.139	0.118	-0.072	-0.106	-0.108	0.021
1997	31-Dec-1997	5	-0.175	-0.127	-0.123	-0.319	0.056	-0.231	-0.183	-0.179	-0.375
1998	31-Dec-1998	5	-0.307	-0.323	-0.323	-0.303	-0.371	0.064	0.048	0.048	0.069
1999	31-Dec-1999	5	-0.227	-0.325	-0.328	-0.300	-0.116	-0.111	-0.209	-0.213	-0.184
2000	29-Dec-2000	5	0.773	0.032	0.021	0.140	-0.312	1.086	0.345	0.334	0.453
2001	31-Dec-2001	5	-0.004	-0.035	-0.037	-0.195	-0.162	0.158	0.127	0.125	-0.033
2002	31-Dec-2002	5	-0.117	-0.010	-0.001	-0.656	-0.323	0.205	0.313	0.321	-0.334
2003	31-Dec-2003	5	-0.444	-0.487	-0.490	0.065	-0.382	-0.062	-0.105	-0.108	0.447
2004	31-Dec-2004	5	-0.044	0.039	0.043	0.458	-0.045	0.002	0.085	0.089	0.504
2005	30-Dec-2005	5	0.089	0.261	0.271	0.244	-0.007	0.095	0.267	0.277	0.251
2006	29-Dec-2006	5	0.025	-0.011	-0.013	-0.003	0.017	0.007	-0.028	-0.030	-0.021
2007	31-Dec-2007	5	0.138	0.231	0.233	0.034	-0.225	0.363	0.456	0.458	0.259
2008	31-Dec-2008	5	0.139	0.379	0.393	-1.659	-1.083	1.222	1.462	1.476	-0.576

Panel notes: The *Accounting V* and *V95* measures are based on sample averages. The sample consists of a maximum of 5 observations per year from the 5 banks in the LSE that adopted the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.

				Т	able 4.20 (C	ontinued)						
PANEL D Relationship for the Secondary Sample Between the Difference in Accounting Value-at-Risk and Historical Value-at-Risk Actual to Relative Delta												
				Accou	nting V		V 95	2	Accounting	V - Market V	7	
Year	Record Date	Obs.	V	V	V	V		$V_E - V_M$	V_A - V_M	$V_L - V_M$	$V_I - V_M$	
			<i>V</i> _E	VA	V _L	VI	V _M	E-M	A-M	L-M	I-M	
1994	30-Dec-1994	11	0.071	-0.003	0.112	0.077	0.064	0.007	-0.067	0.048	0.013	
1995	29-Dec-1995	11	0.088	-0.133	-0.117	0.225	-0.168	0.256	0.035	0.051	0.392	
1996	31-Dec-1996	11	0.115	0.031	-0.018	0.304	-0.024	0.139	0.055	0.006	0.328	
1997	31-Dec-1997	11	0.029	-0.131	-0.210	-0.270	-0.011	0.039	-0.121	-0.200	-0.259	
1998	31-Dec-1998	11	0.020	0.025	0.023	0.392	-0.007	0.027	0.032	0.030	0.399	
1999	31-Dec-1999	11	-0.176	-0.249	-0.389	-0.095	-0.103	-0.073	-0.146	-0.286	0.008	
2000	29-Dec-2000	11	-0.193	-0.240	-0.170	-0.266	-0.054	-0.139	-0.185	-0.116	-0.212	
2001	31-Dec-2001	11	-0.075	-0.092	0.007	-0.458	-0.357	0.283	0.265	0.364	-0.100	
2002	31-Dec-2002	11	-0.034	-0.070	-0.002	0.044	-0.456	0.422	0.386	0.454	0.500	
2003	31-Dec-2003	11	-0.729	-0.790	-0.781	-5.138	-0.579	-0.150	-0.211	-0.202	-4.559	
2004	31-Dec-2004	11	-0.162	-0.136	-0.174	-0.090	-0.111	-0.051	-0.025	-0.063	0.021	
2005	30-Dec-2005	11	0.076	0.053	0.118	6.608	0.023	0.053	0.030	0.095	6.586	
2006	29-Dec-2006	11	0.042	0.038	-0.049	1.519	0.084	-0.043	-0.046	-0.134	1.434	
2007	31-Dec-2007	11	0.029	0.043	0.120	0.824	-0.061	0.090	0.105	0.181	0.885	
2008	31-Dec-2008	11	-0.132	-0.212	-0.150	-0.694	-0.569	0.437	0.356	0.419	-0.126	

Panel notes: The *Accounting V* and *V95* are based on sample averages. The sample consists of a maximum of 11 observations per year from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. The highlighted 2005 record identified the IFRS implementation event year for the adopting firms.

Analysis

Table 4.20 presents results for the difference between the accounting totals Historical Value-at-Risk Actual estimates and the market price return Historical Value-at-Risk Actual and its relation to the accounting total to market price relative delta. Examining Table 4.20, it is evident that a measure for the difference between the accounting Value-at-Risk and market price return Value-at-Risk is shown to be equivalent to the relative delta measure.

The results presented in Table 4.20 are produced from the accounting Historical Valueat-Risk Actual estimates presented in Table 4.19 and the accounting to market price relative delta results presented in Table 4.13.

4.8 Relative Delta Measures for UK Banks

The analysis and results for the change in key accounting variables, market price returns and the relative delta measure for the UK banks are presented on a yearly basis for the time range 1994 to 2008 in this section. In this section, Table 4.22 Panel A presents results for HSBC Holdings PLC; Panel B presents results for Barclays PLC; Panel C presents results for The Royal Bank of Scotland Group PLC; Panel D presents results for Lloyds Banking Group PLC; and, Panel E presents results for Standard Chartered PLC. Table 4.21 presents relative percentage change results for Table 4.22.

Table 4.21Percentage Changes for Relative Delta Measure Results for UK
Banks from Table 4.22

Percentage Change Analysis Table Description

The banks' relative delta levels in Table 4.22 for time period: 1994 to 2004 - before the 2005 accounting change, and the time periods: 1994 to 2007; and 1994 to 2008 - after the accounting change, analysed using relative percentage changes are presented in Table 4.21.

In Table 4.21, the Relative Delta column represents the Description and Symbol columns that describes and *codes* the relative delta measures. The Year column represents the beginning (From) and ending (To) year applied to calculate the percentage changes presented in the Percentage Change column. The Percentage Change column represents the relative delta percentage changes rounded to the nearest whole number. The H column presents the percentage changes for HSBC Holdings PLC. The B column presents the percentage changes for Barclays PLC. The R column presents the percentage changes for The Royal Bank of Scotland Group PLC. The L column presents the percentage changes for Lloyds Banking Group PLC. The S column presents the percentage changes for Standard Chartered PLC.

Relative De	elta	Year		Percer	tage Chang	ge (%)	
Description	Symbol	From - To	Н	В	R	L	S
Total Equity to Market Price	(E-M)	1994 - 2004 1994 - 2007 1994 - 2008	-83 -45 -52	-149 172 685	133 431 1,077	-100 8 321	-110 -98 285
Total Assets to Market Price	(A-M)	1994 - 2004 1994 - 2007 1994 - 2008	-53 6 113	-95 2,942 8,726	1,263 7,481 14,169	-100 50 947	-74 -109 608
Total Liabilities to Market Price	(L-M)	1994 - 2004 1994 - 2007 1994 - 2008	-50 12 127	-73 3,760 11,120	3,317 19,867 37,867	-100 55 994	-71 -109 633
Net Income to Market Price	(I-M)	1994 - 2004 1994 - 2007 1994 - 2008	-88 -39 -322	-101 -94 -41	137 -51 -556	-100 2 61	-62 -100 142

Table Description

The Table 4.22 Panel A to Panel E columns represent the following:

The Year column represents the fiscal accounting year. Record Date is the date the variable values are recorded. *Obs.* represents the number of sample firms observed for the year.

For the Change in Variables (Variable Delta) column:

E and $dE_{t-l,t}$ column represents the relative change in the total equity variable for the *year* time period t - 1 to t calculated by applying equation (3.34); A and $dA_{t-l,t}$ column represents the log change in total assets for the *year* time period t - 1 to t calculated by applying equation (3.33); L and $dL_{t-l,t}$ column represents the log change in total liabilities for the *year* time period t - 1 to t calculated by applying equation (3.33); I and $dL_{t-l,t}$ column represents the log change in total liabilities for the *year* time period t - 1 to t calculated by applying equation (3.33); I and $dI_{t-l,t}$ column represents the relative change in the net income variable for the *year* time period t - 1 to t calculated by applying equation (3.34); and M and $dM_{t-l,t}$ column represents the market price return specified by the log change in market price for the *year* time period t - 1 to t and calculated by applying equation (3.35).

For the Relative Delta Measures column:

E-M column represents the total Equity to Market Price Relative Delta for the *year* time period t - 1 to t calculated by applying equation (3.38) for one firm, and takes the general form:

$$\delta EM_t = dE_{i,t} - dM_t$$

Where: δEM_t is the Total Equity to Market Price Relative Delta at time t, dE_t is the change in Total Equity $(dE_{t-1,t})$ at time t, and dM_t is the log change in market price $(dM_{t-1,t})$ at time t.

A-M column represents the Total Assets to Market Price Relative Delta for the *year* time period t - 1 to t calculated by applying equation (3.38) for one firm, and takes the general form:

$$\delta AM_t = dA_t - dM_t$$

Where: δAM_t is the Total Assets to Market Price Relative Delta at time t, dA_t is the log change in Total Assets $(dA_{t-1,t})$ at time t, and dM_t is the log change in market price $(dM_{t-1,t})$ at time t.

L-M column represents the Total Liabilities to Market Price Relative Delta for the *year* time period t - 1 to t calculated by applying equation (3.38) for one firm, and takes the general form:

$\delta LM_t = dL_t - dM_t$

Where: δLM_t is the Total Liabilities to Market Price Relative Delta at time t, dL_t is the log change in Total Liabilities $(dL_{t-1,t})$ at time t, and dM_t is the log change in market price $(dM_{t-1,t})$ at time t.

I-M column represents the net income to market price relative delta for the *year* time period t - 1 to t calculated by applying equation (3.38) for one firm, and takes the general form:

$\delta IM_t = dI_t - dM_t$

Where: δIM_t is the Net Income to Market Price Relative Delta at time *t*, dI_t is the change in Net Income $(dI_{t-1,t})$ at time *t*, and dM_t is the log change in market price $(dM_{t-1,t})$ at time *t*.

			1101001101		100 101 1102	e norango r					
			Change in V	Variables (V	ariable Delta	l)	Relative Delta Measures				
Year	Record Date	E	A	L	Ι	М		() (
		$dE_{t-l,t}$	$dA_{t-l,t}$	$dL_{t-l,t}$	$dI_{t-1,t}$	$dM_{t-1,t}$	<i>E-M</i>	A-M	L-M	I-M	
1994	30-Dec-1994	0.156	-0.022	-0.032	0.137	-0.285	0.477	0.299	0.289	0.458	
1995	29-Dec-1995	0.927	0.558	0.552	0.896	0.363	0.573	0.205	0.198	0.542	
1996	31-Dec-1996	0.251	0.140	0.134	0.258	0.272	-0.010	-0.121	-0.127	-0.003	
1997	31-Dec-1997	-0.368	-0.347	-0.338	-0.315	0.183	-0.545	-0.524	-0.516	-0.493	
1998	31-Dec-1998	0.002	0.014	0.015	-0.227	0.054	-0.042	-0.030	-0.028	-0.270	
1999	31-Dec-1999	0.258	0.196	0.194	0.293	1.624	-0.204	-0.267	-0.269	-0.170	
2000	29-Dec-2000	0.454	0.244	0.230	0.323	0.222	0.322	0.112	0.098	0.190	
2001	31-Dec-2001	0.036	0.058	0.061	-0.163	-0.190	0.236	0.258	0.262	0.038	
2002	31-Dec-2002	0.033	-0.014	-0.017	-0.528	-0.148	0.193	0.146	0.144	-0.368	
2003	31-Dec-2003	0.291	0.202	0.196	0.459	0.392	0.045	-0.044	-0.050	0.213	
2004	31-Dec-2004	0.085	0.142	0.146	0.054	0.020	0.083	0.140	0.145	0.053	
2005	30-Dec-2005	0.193	0.275	0.288	0.654	0.074	0.134	0.215	0.229	0.594	
2006	29-Dec-2006	0.028	0.083	0.087	-0.232	0.019	0.030	0.085	0.089	-0.230	
2007	31-Dec-2007	0.163	0.218	0.222	0.179	-0.079	0.263	0.318	0.323	0.279	
2008	31-Dec-2008	-0.012	0.396	0.415	-1.256	-0.217	0.229	0.636	0.656	-1.016	

PANEL A

Table 4.22 (Continued)

Relative Delta Measures for HSBC Holdings PLC

			Relativ	ve Delta Me	easures for l	Barclays PLC				
			Change in V	/ariables (V	ariable Delta	Relative Delta Measures				
Year	Record Date	E	A	L	Ι	М		4.34		IM
		$dE_{t-1,t}$	$dA_{t-1,t}$	$dL_{t-1,t}$	$dI_{t-1,t}$	$dM_{t-1,t}$	<i>E-M</i>	A-M	L-IVI	<i>I-M</i>
1994	30-Dec-1994	0.160	-0.022	-0.026	2.767	-0.033	0.201	0.019	0.015	2.808
1995	29-Dec-1995	0.141	0.039	0.035	0.157	0.183	-0.050	-0.152	-0.156	-0.034
1996	31-Dec-1996	0.034	0.097	0.100	0.202	0.252	-0.269	-0.206	-0.203	-0.101
1997	31-Dec-1997	0.043	0.232	0.240	-0.311	0.469	-0.438	-0.249	-0.241	-0.791
1998	31-Dec-1998	0.040	-0.067	-0.071	0.181	-0.230	0.262	0.155	0.151	0.403
1999	31-Dec-1999	0.076	0.149	0.152	0.318	0.308	-0.243	-0.169	-0.166	-0.001
2000	29-Dec-2000	0.555	0.216	0.203	0.406	0.258	0.404	0.065	0.053	0.255
2001	31-Dec-2001	0.100	0.120	0.121	-0.003	0.098	0.007	0.027	0.027	-0.097
2002	31-Dec-2002	0.045	0.122	0.131	-0.596	-0.391	0.435	0.512	0.521	-0.205
2003	31-Dec-2003	0.081	0.095	0.096	0.365	0.240	-0.177	-0.163	-0.162	0.107
2004	31-Dec-2004	0.064	0.164	0.166	0.140	0.145	-0.099	0.001	0.004	-0.022
2005	30-Dec-2005	0.001	0.571	0.579	0.035	0.047	-0.041	0.529	0.538	-0.007
2006	29-Dec-2006	0.136	0.075	0.074	0.410	0.185	-0.042	-0.103	-0.104	0.232
2007	31-Dec-2007	0.176	0.208	0.209	-0.212	-0.360	0.547	0.578	0.579	0.158
2008	31-Dec-2008	0.415	0.514	0.520	0.494	-0.952	1.578	1.677	1.683	1.657

PANEL B

Table 4.22 (Continued)

		Relative	e Delta Meas	sures for Th	e Royal Ba	nk of Scotland	Group PLC				
			Change in V	variables (V	ariable Delta	.)	Relative Delta Measures				
Year	Record Date	E	A	L	Ι	М		4.14			
		$dE_{t-1,t}$	$dA_{t-1,t}$	$dL_{t-l,t}$	$dI_{t-1,t}$	$dM_{t-1,t}$	<i>E-M</i>	A-M	L-M	<i>I-M</i>	
1994	30-Sep-1994	0.021	0.222	0.232	1.381	0.247	-0.218	-0.016	-0.006	1.143	
1995	29-Sep-1995	0.118	0.119	0.119	0.063	0.087	0.037	0.038	0.038	-0.018	
1996	30-Sep-1996	0.159	0.180	0.179	0.247	0.095	0.076	0.097	0.096	0.164	
1997	30-Sep-1997	0.212	0.172	0.171	0.041	0.403	-0.133	-0.173	-0.174	-0.304	
1998	30-Sep-1998	-0.029	0.093	0.100	0.394	-0.018	0.001	0.123	0.130	0.424	
1999	30-Sep-1999	0.423	0.109	0.098	0.218	0.684	-0.242	-0.556	-0.567	-0.447	
2000	29-Dec-2000	4.501	1.281	1.255	0.905	1.294	4.229	1.010	0.983	0.633	
2001	31-Dec-2001	0.195	0.142	0.139	0.264	0.118	0.140	0.087	0.084	0.209	
2002	31-Dec-2002	-0.021	0.111	0.118	-0.660	-0.102	0.096	0.227	0.234	-0.543	
2003	31-Dec-2003	0.039	0.100	0.103	-0.102	0.125	-0.062	-0.001	0.002	-0.203	
2004	31-Dec-2004	0.134	0.248	0.255	2.772	0.131	0.072	0.186	0.193	2.710	
2005	30-Dec-2005	0.112	0.286	0.300	0.329	0.006	0.110	0.284	0.298	0.328	
2006	29-Dec-2006	0.135	0.115	0.111	0.128	0.115	0.008	-0.012	-0.016	0.001	
2007	31-Dec-2007	0.318	0.778	0.783	0.163	-0.345	0.721	1.181	1.186	0.565	
2008	31-Dec-2008	0.110	0.233	0.248	-7.226	-0.824	2.129	2.251	2.266	-5.208	

PANEL C Relative Delta Measures for The Royal Bank of Scotland Group PL

		R	elative Delta	a Measures	for Lloyds	Banking Grou	p PLC			
			Change in V	/ariables (V	ariable Delta	Relative Delta Measures				
Year	Record Date	E	A	L	Ι	М		4.34		1.14
		$dE_{t-1,t}$	$dA_{t-1,t}$	$dL_{t-1,t}$	$dI_{t-1,t}$	$dM_{t-1,t}$	<i>E-M</i>	A-M	L-M	<i>I-M</i>
1994	30-Sep-1994	-	-	-	-	-	-	-	-	-
1995	29-Sep-1995	-	-	-	-	-	-	-	-	-
1996	30-Sep-1996	-	-	-	-	-	-	-	-	-
1997	30-Sep-1997	-	-	-	-	-	-	-	-	-
1998	31-Dec-1998	-	-	-	-	-	-	-	-	-
1999	31-Dec-1999	0.162	0.047	0.042	0.186	-0.092	0.260	0.146	0.141	0.285
2000	29-Dec-2000	0.121	0.213	0.216	0.084	-0.085	0.211	0.303	0.306	0.173
2001	31-Dec-2001	0.106	0.082	0.079	-0.082	0.066	0.054	0.029	0.027	-0.135
2002	31-Dec-2002	-0.259	0.066	0.082	-0.733	-0.513	0.255	0.581	0.597	-0.218
2003	31-Dec-2003	0.210	-0.003	-0.010	2.142	0.006	0.205	-0.007	-0.014	2.138
2004	31-Dec-2004	0.037	0.105	0.108	-0.363	0.055	-0.018	0.050	0.054	-0.417
2005	30-Dec-2005	0.022	0.102	0.105	-0.028	0.033	-0.010	0.069	0.073	-0.060
2006	29-Dec-2006	0.094	0.104	0.105	0.221	0.163	-0.063	-0.053	-0.052	0.064
2007	31-Dec-2007	0.088	0.028	0.026	0.101	-0.189	0.280	0.219	0.218	0.292
2008	31-Dec-2008	-0.226	0.208	0.222	-0.861	-0.904	1.094	1.529	1.542	0.460

Table 4.22 (Continued)
PANEL D

			Change in V	variables (Va	ariable Delta)	i i i i i i i i i i i i i i i i i i i	Relative Deli	ta Measures	
Year	Record Date	E	A	L	Ι	М		4.14	T M	
		$dE_{t-1,t}$	$dA_{t-1,t}$	$dL_{t-1,t}$	$dI_{t-1,t}$	$dM_{t-1,t}$	E-M	A-M	L-M	<i>I-M</i>
1994	30-Dec-1994	0.180	0.071	0.067	0.406	-0.095	0.281	0.172	0.168	0.508
1995	29-Dec-1995	0.288	0.129	0.123	0.379	0.676	-0.382	-0.540	-0.547	-0.291
1996	31-Dec-1996	0.187	0.079	0.074	0.297	0.284	-0.084	-0.193	-0.198	0.026
1997	31-Dec-1997	0.091	0.113	0.115	-0.014	-0.091	0.192	0.214	0.215	0.087
1998	31-Dec-1998	0.105	0.014	0.009	-0.213	0.077	0.035	-0.055	-0.060	-0.282
1999	31-Dec-1999	0.194	0.123	0.120	-0.266	0.380	-0.128	-0.199	-0.202	-0.588
2000	29-Dec-2000	0.266	0.237	0.232	1.015	0.065	0.262	0.233	0.228	1.012
2001	31-Dec-2001	0.190	0.073	0.062	-0.343	-0.159	0.353	0.235	0.224	-0.181
2002	31-Dec-2002	-0.103	-0.052	-0.040	-0.484	-0.116	0.047	0.098	0.110	-0.334
2003	31-Dec-2003	-0.053	-0.044	-0.046	0.250	0.272	-0.320	-0.311	-0.314	-0.017
2004	31-Dec-2004	0.019	0.094	0.097	0.243	0.052	-0.029	0.045	0.049	0.194
2005	30-Dec-2005	0.575	0.529	0.538	0.690	0.389	0.285	0.238	0.248	0.399
2006	29-Dec-2006	0.244	0.082	0.073	-0.029	0.204	0.103	-0.060	-0.068	-0.171
2007	31-Dec-2007	0.216	0.196	0.197	0.214	0.228	0.005	-0.015	-0.015	0.002
2008	31-Dec-2008	0.470	0.604	0.618	0.615	-0.448	1.082	1.217	1.231	1.227

Table 4.22 (Continued)PANEL E

Analysis

Table 4.22 present accounting totals to market price relative delta results for the UK banks.

These tables bank level relative delta measures suggest that there is a greater average difference between the change in accounting total variables and the change in the market price variable after the 2005 accounting change than before. In these results, some *difference component* levels exhibit material significance. For after 2005, the only decrease in this *difference component* is reported from the net income to market price relative delta measure for HSBC Holdings PLC, The Royal Bank of Scotland Group PLC and Lloyds Banking Group PLC.

Examining the *absolute* average bank level relative deltas in Table 4.22 Panel A to Panel E for before and after 2005 suggest that the banks experienced, in general, greater materially significant levels of the *difference component* after the 2005 accounting change than before. The evidence from the relative percentage changes for before and after 2005, presented in Table 4.21, suggest support for this finding.

4.9 Estimates of Accounting Value-at-Risk for UK Banks

The accounting Value-at-Risk estimates are presented for the UK banks on a yearly basis for the time range 1994 to 2008 in Table 4.24. Table 4.24 Panel A presents results for HSBC Holdings PLC; Panel B presents results for Barclays PLC; Panel C presents results for The Royal Bank of Scotland Group PLC; Panel D presents results for Lloyds Banking Group PLC; and, Panel E presents results for Standard Chartered PLC. Table 4.23 presents relative percentage change results for Table 4.24.

Table 4.23Percentage Changes for Accounting Value-at-Risk Results for UK
Banks from Table 4.24

Percentage Change Analysis Table Description

The banks accounting Value-at-Risk estimates in Table 4.24 for the time period: 1994 to 2004 - before the 2005 accounting change, and the time periods: 1994 to 2007; and 1994 to 2008 - after the accounting change, analysed using relative percentage changes are presented in Table 4.23.

In Table 4.23, the Accounting Value-at-Risk column represents the Description and Symbol columns that describes and *codes* the relative delta measures. The Year column represents the beginning (From) and ending (To) year applied to calculate the percentage changes presented in the Percentage Change column. The Percentage Change column represents the accounting Value-at-Risk percentage changes rounded to the nearest whole number. The H column presents the percentage changes for HSBC Holdings PLC. The B column presents the percentage changes for The Royal Bank of Scotland Group PLC. The L column presents the percentage changes for Lloyds Banking Group PLC. The S column presents the percentage changes for Standard Chartered PLC.

Accounting Value	e-at-Risk	Year		Percen	tage Char	nge (%)	
Description	Symbol	From - To	Н	В	R	L	S
Total Equity		1994 - 2004	-70	-171	-128	-249	-100
Value at Rick	V_E	1994 - 2007	-39	19	524	-19	-113
v alue-at-ixisk		1994 - 2008	-100	142	204	-233	45
Total Assats		1994 - 2004	98	21	-59	-1,229	-30
Total Assets	V_A	1994 - 2007	212	1,284	202	486	-145
v alue-at-RISK		1994 - 2008	526	2,695	6	4,257	326
Tatal Lighilitian		1994 - 2004	144	48	-58	-650	-23
Volue et Diele	V_L	1994 - 2007	278	1,083	192	317	-147
v alue-at-RISK		1994 - 2008	675	2,265	8	2,633	357
		1994 - 2004	-82	-101	87	-524	-49
Net Income	V_I	1994 - 2007	-27	-107	-91	-24	-107
Value-at-Risk	-	1994 - 2008	-656	-83	-612	-690	3
		1994 - 2004	95	58	-132	6	141
Market Price	V_M	1994 - 2007	51	-829	-265	-25	53
Value-at-Risk	- 141	1994 - 2008	2	-3,018	-856	-709	-1,037

Table Description

The Table 4.24 Panel A to Panel E columns represent the following:

The Year column represents the fiscal accounting year. Record Date is the date the variable values are recorded. Obs. represents the number of sample firms observed for the year.

For the Relative Delta Measures column:

E-M column represents the total equity to market price relative delta for the *year* time period t-1 to t calculated by applying equation (3.38) for one firm; A-M column represents the total assets to market price relative delta for the *year* time period t - 1 to t calculated by applying equation (3.38) for one firm; L-M column represents the total liabilities to market price relative delta for the year time period t - 1 to t calculated by applying equation (3.38) for one firm; and, I-M column represents the net income to market price relative delta for the year time period t - 1 to t calculated by applying equation (3.38) for one firm.

The V 95 column represents the Historical Value-at-Risk Actual, V_M , calculated at the 95% confidence level by applying equation (N.34).

For the Accounting V Estimates column:

 V_E column represents the estimates for the Total Equity Value-at-Risk Actual at the 95% confidence level for the year time period t - 1 to t calculated by applying equation (3.44), and takes the general form:

$$V_{E:t} = \delta E M_t + V_{M:t}$$

Where: $V_{E:t}$ is the Total Equity Value-at-Risk at time t, δEM_t is the Total Equity to Market Price Relative Delta $(dE_{t-1,t} - dM_{t-1,t})$ at time t, and $V_{M:i,t}$ is the Market Price Value-at-Risk (V_t) at time t.

 V_A column represents the estimates for the Total Assets Value-at-Risk Actual at the 95% confidence level for the year time period t - 1 to t calculated by applying equation (3.44), and takes the general form:

 $V_{A:t} = \delta AM_t + V_{M:t}$ Where: $V_{A:t}$ is the Total Assets Value-at-Risk at time t, δAM_t is the Total Assets to Market Price Relative Delta $(dA_{t-1,t} - dM_{t-1,t})$ at time t, and $V_{M:i,t}$ is the Market Price Value-at-Risk (V_t) at time t.

 V_L column represents the estimates for the Total Liabilities Value-at-Risk Actual at the 95% confidence level for the year time period t - 1 to t calculated by applying equation (3.44), and takes the general form:

$$V_{L:t} = \delta L M_t + V_{M:t}$$

Where: $V_{L:t}$ is the Total Liabilities Value-at-Risk at time t, δLM_t is the Total Liabilities to Market Price Relative Delta $(dL_{t-1,t} - dM_{t-1,t})$ at time t, and $V_{M:t}$ is the Market Price Value-at-Risk (V_t) at time t.

V₁ column represents the estimates for the Net Income Value-at-Risk Actual at the 95% confidence level for the *year* time period t-1 to t calculated by applying equation (3.44), and takes the general form:

$$V_{I:t} = \delta I M_t + V_{M:t}$$

Where: $V_{I:t}$ is the Net Income Value-at-Risk at time t, δIM_t is the Net Income to Market Price Relative Delta $(dI_{t-1,t} - dM_{t-1,t})$ at time t, and $V_{M:t}$ is the Market Price Value-at-Risk (V_t) at time t.

Vear	Record Date		Relativ Accounting	e Delta V - Market V		V 95	Accounting V				
I Cui	Record Date	E-M	A-M	L-M	I-M	V _M	V_E	V_A	V_L	V _I	
		$V_E - V_M$	V_A - V_M	$V_L - V_M$	$V_I - V_M$						
1994	30-Dec-1994	0.477	0.299	0.289	0.458	-0.234	0.243	0.065	0.055	0.224	
1995	29-Dec-1995	0.573	0.205	0.198	0.542	-0.404	0.169	-0.200	-0.206	0.138	
1996	31-Dec-1996	-0.010	-0.121	-0.127	-0.003	0.158	0.148	0.037	0.032	0.156	
1997	31-Dec-1997	-0.545	-0.524	-0.516	-0.493	0.140	-0.405	-0.384	-0.375	-0.35	
1998	31-Dec-1998	-0.042	-0.030	-0.028	-0.270	-0.588	-0.630	-0.619	-0.617	-0.85	
1999	31-Dec-1999	-0.204	-0.267	-0.269	-0.170	-0.074	-0.279	-0.342	-0.343	-0.24	
2000	29-Dec-2000	0.322	0.112	0.098	0.190	-0.077	0.244	0.035	0.021	0.11.	
2001	31-Dec-2001	0.236	0.258	0.262	0.038	-0.314	-0.077	-0.055	-0.052	-0.27	
2002	31-Dec-2002	0.193	0.146	0.144	-0.368	-0.285	-0.092	-0.139	-0.141	-0.65	
2003	31-Dec-2003	0.045	-0.044	-0.050	0.213	-0.217	-0.173	-0.261	-0.267	-0.00	
2004	31-Dec-2004	0.083	0.140	0.145	0.053	-0.011	0.072	0.129	0.134	0.04	
2005	30-Dec-2005	0.134	0.215	0.229	0.594	-0.018	0.115	0.197	0.210	0.57	
2006	29-Dec-2006	0.030	0.085	0.089	-0.230	-0.008	0.023	0.078	0.081	-0.23	
2007	31-Dec-2007	0.263	0.318	0.323	0.279	-0.115	0.148	0.203	0.208	0.16	
2008	31-Dec-2008	0.229	0.636	0.656	-1.016	-0.230	-0.001	0.407	0.426	-1.24	

PANEL A Estimates for HSBC Holdings PLC Accounting Value-at-Risk from Relative Delta and Historical Value-at-Risk Actual

	Estimates for H	Barclays PL	C Accountir	ng Value-at-	Risk from Ro	elative Delta a	nd Historica	l Value-at-R	isk Actual		
Voor	Record Date		Relativ Accounting	e Delta V - Market V		V 95	Accounting V				
i cai	Record Date	E-M	A-M	L-M	I-M						
		$V_E - V_M$	V_A - V_M	V_L - V_M	$V_I - V_M$	V _M	V_E	V_A	V_L	V_I	
1994	30-Dec-1994	0.201	0.019	0.015	2.808	-0.038	0.163	-0.019	-0.023	2.770	
1995	29-Dec-1995	-0.050	-0.152	-0.156	-0.034	-0.054	-0.104	-0.206	-0.210	-0.088	
1996	31-Dec-1996	-0.269	-0.206	-0.203	-0.101	0.094	-0.175	-0.113	-0.110	-0.008	
1997	31-Dec-1997	-0.438	-0.249	-0.241	-0.791	0.249	-0.189	0.001	0.008	-0.542	
1998	31-Dec-1998	0.262	0.155	0.151	0.403	-0.413	-0.151	-0.258	-0.262	-0.010	
1999	31-Dec-1999	-0.243	-0.169	-0.166	-0.001	-0.296	-0.539	-0.465	-0.463	-0.297	
2000	29-Dec-2000	0.404	0.065	0.053	0.255	-0.201	0.203	-0.136	-0.149	0.054	
2001	31-Dec-2001	0.007	0.027	0.027	-0.097	0.006	0.013	0.033	0.034	-0.090	
2002	31-Dec-2002	0.435	0.512	0.521	-0.205	-0.360	0.075	0.153	0.161	-0.565	
2003	31-Dec-2003	-0.177	-0.163	-0.162	0.107	-0.450	-0.627	-0.613	-0.612	-0.343	
2004	31-Dec-2004	-0.099	0.001	0.004	-0.022	-0.016	-0.115	-0.015	-0.012	-0.038	
2005	30-Dec-2005	-0.041	0.529	0.538	-0.007	0.002	-0.039	0.531	0.539	-0.005	
2006	29-Dec-2006	-0.042	-0.103	-0.104	0.232	0.018	-0.024	-0.085	-0.086	0.250	
2007	31-Dec-2007	0.547	0.578	0.579	0.158	-0.353	0.194	0.225	0.226	-0.195	
2008	31-Dec-2008	1.578	1.677	1.683	1.657	-1.185	0.394	0.493	0.498	0.472	

PANEL B Estimates for Barclays PLC Accounting Value-at-Risk from Relative Delta and Historical Value-at-Risk Actual

Vear	Record Date		Relativ Accounting	e Delta V - Market V		V 95	Accounting V				
i cai	Record Date	E-M	A-M	L-M	I-M					V	
		V_E - V_M	V_A - V_M	V_L - V_M	$V_I - V_M$	V _M	V_E	V_A	V_L	V_I	
1994	30-Sep-1994	-0.218	-0.016	-0.006	1.143	0.263	0.046	0.247	0.257	1.406	
1995	29-Sep-1995	0.037	0.038	0.038	-0.018	-0.208	-0.171	-0.170	-0.170	-0.22	
1996	30-Sep-1996	0.076	0.097	0.096	0.164	0.048	0.125	0.145	0.145	0.212	
1997	30-Sep-1997	-0.133	-0.173	-0.174	-0.304	-0.104	-0.237	-0.277	-0.278	-0.40	
1998	30-Sep-1998	0.001	0.123	0.130	0.424	0.162	0.163	0.285	0.292	0.58	
1999	30-Sep-1999	-0.242	-0.556	-0.567	-0.447	0.052	-0.190	-0.504	-0.515	-0.39	
2000	29-Dec-2000	4.229	1.010	0.983	0.633	-0.521	3.709	0.489	0.462	0.112	
2001	31-Dec-2001	0.140	0.087	0.084	0.209	0.007	0.147	0.094	0.091	0.21	
2002	31-Dec-2002	0.096	0.227	0.234	-0.543	-0.197	-0.101	0.031	0.037	-0.74	
2003	31-Dec-2003	-0.062	-0.001	0.002	-0.203	-0.275	-0.337	-0.276	-0.273	-0.47	
2004	31-Dec-2004	0.072	0.186	0.193	2.710	-0.084	-0.013	0.101	0.108	2.62	
2005	30-Dec-2005	0.110	0.284	0.298	0.328	-0.048	0.063	0.237	0.250	0.28	
2006	29-Dec-2006	0.008	-0.012	-0.016	0.001	-0.027	-0.019	-0.039	-0.043	-0.02	
2007	31-Dec-2007	0.721	1.181	1.186	0.565	-0.434	0.287	0.747	0.751	0.13	
2008	31-Dec-2008	2.129	2.251	2.266	-5.208	-1.989	0.140	0.262	0.277	-7.19	

Esti	mates for Lloyds l	Banking Gro	oup PLC Ac	counting Va	PANEL D llue-at-Risk f	from Relative	Delta and H	istorical Val	ue-at-Risk A	Actual
Voor	Pasard Data		Relativ Accounting	e Delta V - Market V		V 95		Accoun	ting V	
real	Record Date	E-M	A-M	L-M	I-M					
		$V_E - V_M$	V_A - V_M	V_L - V_M	$V_I - V_M$	V_M	V_E	V_A	V_L	V_I
1994	30-Sep-1994	_	_	_	_	_	_	_	_	_
1995	29-Sep-1995	-	-	-	-	-	-	-	-	-
1996	30-Sep-1996	-	-	-	-	-	-	-	-	-
1997	30-Sep-1997	-	-	-	-	-	-	-	-	-
1998	31-Dec-1998	-	-	-	-	-	-	-	-	-
1999	31-Dec-1999	0.260	0.146	0.141	0.285	-0.153	0.108	-0.007	-0.012	0.132
2000	29-Dec-2000	0.211	0.303	0.306	0.173	-0.500	-0.289	-0.197	-0.195	-0.327
2001	31-Dec-2001	0.054	0.029	0.027	-0.135	-0.119	-0.065	-0.089	-0.092	-0.253
2002	31-Dec-2002	0.255	0.581	0.597	-0.218	-0.462	-0.207	0.118	0.134	-0.681
2003	31-Dec-2003	0.205	-0.007	-0.014	2.138	-0.767	-0.562	-0.775	-0.782	1.370
2004	31-Dec-2004	-0.018	0.050	0.054	-0.417	-0.144	-0.161	-0.093	-0.090	-0.560
2005	30-Dec-2005	-0.010	0.069	0.073	-0.060	0.004	-0.007	0.073	0.077	-0.056
2006	29-Dec-2006	-0.063	-0.053	-0.052	0.064	0.022	-0.041	-0.031	-0.030	0.086
2007	31-Dec-2007	0.280	0.219	0.218	0.292	-0.192	0.088	0.027	0.026	0.100
2008	31-Dec-2008	1.094	1.529	1.542	0.460	-1.238	-0.144	0.291	0.304	-0.779

PANEL E Estimates for Standard Chartered PLC Accounting Value-at-Risk from Relative Delta and Historical Value-at-Risk Actual										
Year	Record Date	<i>Relative Delta</i> <i>Accounting V - Market V</i>			V 95	Accounting V				
		E-M	A-M	L-M	I-M	V_M	V_E	V_A	V _L	V _I
		$V_E - V_M$	V_A - V_M	$V_L - V_M$	$V_I - V_M$					
1994	30-Dec-1994	0.281	0.172	0.168	0.508	-0.068	0.213	0.104	0.100	0.439
1995	29-Dec-1995	-0.382	-0.540	-0.547	-0.291	-0.223	-0.605	-0.764	-0.770	-0.514
1996	31-Dec-1996	-0.084	-0.193	-0.198	0.026	0.172	0.088	-0.021	-0.026	0.198
1997	31-Dec-1997	0.192	0.214	0.215	0.087	-0.060	0.132	0.154	0.155	0.027
1998	31-Dec-1998	0.035	-0.055	-0.060	-0.282	-0.646	-0.610	-0.701	-0.706	-0.928
1999	31-Dec-1999	-0.128	-0.199	-0.202	-0.588	-0.106	-0.235	-0.305	-0.309	-0.69
2000	29-Dec-2000	0.262	0.233	0.228	1.012	-0.261	0.001	-0.029	-0.033	0.750
2001	31-Dec-2001	0.353	0.235	0.224	-0.181	-0.391	-0.039	-0.156	-0.167	-0.572
2002	31-Dec-2002	0.047	0.098	0.110	-0.334	-0.308	-0.261	-0.210	-0.199	-0.643
2003	31-Dec-2003	-0.320	-0.311	-0.314	-0.017	-0.201	-0.521	-0.512	-0.515	-0.218
2004	31-Dec-2004	-0.029	0.045	0.049	0.194	0.028	-0.001	0.073	0.077	0.222
2005	30-Dec-2005	0.285	0.238	0.248	0.399	0.028	0.313	0.266	0.276	0.427
2006	29-Dec-2006	0.103	-0.060	-0.068	-0.171	0.081	0.184	0.021	0.013	-0.089
2007	31-Dec-2007	0.005	-0.015	-0.015	0.002	-0.032	-0.027	-0.047	-0.047	-0.030
2008	31-Dec-2008	1.082	1.217	1.231	1.227	-0.773	0.309	0.443	0.457	0.454

Analysis

Table 4.24 Panel A to Panel E present *accounting Historical Value-at-Risk Actual*, referred to as the *accounting Value-at-Risk*, estimates for the UK banks. As stated earlier in Section 4.7.1, Analysis, the banks accounting Value-at-Risk measure estimated in this study by definition, is an equivalent measurement to the market price return Historical Value-at-Risk Actual specified in Appendix N.8.

The Table 4.24 bank accounting Value-at-Risk estimates suggest varying accounting Value-at-Risk levels for before and after the 2005 accounting change. The results suggest that the banks experienced significant material decreases in total assets Value-at-Risk and total liabilities Value-at-Risk after 2005 compared to before. The total equity Value-at-Risk and net-income Value-at-Risk levels, however, exhibited both increases and decreases.

Generally, all the banks exhibited increases in market price return Historical Value-at-Risk Actual and mainly decreases in accounting Value-at-Risk after 2005 compared to before. The evidence from the relative percentage changes for before and after 2005, presented in Table 4.23, suggest support for this finding

4.10 Regulatory Relative Delta Framework: Estimates for Minimum Regulatory Capital for UK Banks

The adjusted Basel (2011) minimum regulatory capital requirement, estimated by the proposed regulatory relative delta framework (see Sections 3.9.6, Appendix K and Appendix L), is presented for the UK banks on a yearly basis for the time range 1994 to 2008 in Table 4.27. In this table, Panel A presents results for HSBC Holdings PLC; Panel B presents results for Barclays PLC; Panel C presents results for The Royal Bank of Scotland PLC; Panel D presents results for Lloyds Banking Group PLC; and, Panel E presents results for Standard Chartered PLC.

Table 4.25 presents, for the banks, the adjusted Basel minimum regulatory capital level estimated on a yearly basis from 2004 to 2008 using the average total equity to market price regulatory relative delta component from Table 4.27. Table 4.26 presents relative percentage change results for Table 4.27.

Table 4.25Adjusted Minimum Basel Regulatory Capital from Total Equity for
UK Banks

The table shows the estimated average change to the minimum Basel regulatory capital requirement, adjusted by the level of difference between the change in total equity and change in market price for banks registered with the LSE's UK banking sector. The Time column presents the Year and Period columns. The Year and Period columns present the corresponding year and time period, respectively, for the regulatory relative delta measure presented in the ΔR column. The *Change in 8% Capital Requirement (%)* column presents the adjustment applied to the Basel minimum regulatory capital, at the 8% level, using the formula: $8\% \times (1 + \Delta R)$ formula. The Increase (%) column presents the percentage increase from the 8% minimum capital requirement level to the adjusted level presented in the *Change in 8% Capital Requirement (%)* column.

	Time	ΔR^{a}	Change in 8% Capital Requirement (%)	Increase (%)	
Year	Period		$8\% \times (1 + \Delta R)$		
2004	01-Jan-04 to 31 Dec-04	0.002	8.02	0.2	
2005	01-Jan-05 to 31-Dec-05	0.095	8.76	9.5	
2006	01-Jan-06 to 31-Dec-06	0.007	8.06	0.7	
2007	01-Jan-07 to 31-Dec-07	0.363	10.90	36.3	
2008	01-Jan-08 to 31-Dec-08	1.222	17.78	122.2	

Table note: ^a ΔR is the total equity to market price regulatory relative delta calculated by applying equation (3.51).

The increases presented in Table 4.25 also signify the level of exposure to financial distress risk experienced by the UK banks during 2004 to 2008. Appendix L presents the approach this study applied to develop the regulatory relative delta estimates in Table 4.25.

Examining the results for the system of accounting totals, the Table 4.27 bank regulatory relative delta measures, suggest that 80% (4 banks from 5) exhibit materially significant levels for the absolute maximum accounting totals to market price relative delta after the 2005 accounting change when compared to before.

That is, the primary UK banks experienced a materially significant increase in the level of the absolute maximum difference between the change in accounting total variables and the change in the market price variable after the 2005 accounting standards change compared to before. The evidence from the relative percentage changes for before and after 2005, presented in Table 4.26, suggest support for this finding.

Table 4.26Percentage Changes for Regulatory Relative Delta Results for UK
Banks from Table 4.27

Percentage Change Analysis Table Description

The banks regulatory relative delta, ΔR , levels in Table 4.27 for time period: 1994 to 2004 - before the 2005 accounting change, and the time periods: 1994 to 2007; and 1994 to 2008 - after the accounting change, analysed using relative percentage changes are presented in Table 4.26.^a

In Table 4.26, the Description and Symbol columns describes and *codes* the regulatory relative delta measure. The Year column represents the beginning (From) and ending (To) year applied to calculate the percentage changes presented in the Percentage Change column. The Percentage Change column represents the regulatory relative delta percentage changes rounded to the nearest whole number. The H column presents the percentage changes for HSBC Holdings PLC. The B column presents the percentage changes for Barclays PLC. The R column presents the percentage changes for The Royal Bank of Scotland Group PLC. The L column presents the percentage changes for Lloyds Banking Group PLC. The S column presents the percentage changes for Standard Chartered PLC.

Description	Symbol	Year	Percentage Change (%)					
2 •••••••p •••••		From - To	Н	В	R	L	S	
Regulatory Relative Delta	ΔR	1994 - 2004 1994 - 2007 1994 - 2008	-70 -32 113	-96 -79 -40	137 4 356	46 2 441	-62 -97 142	

Table note: ^a Although Table 4.26 attempts to present for the banks summary information, it is recommended that Table 4.27 is examined for a more *accurate* view of the regulatory relative delta results.
Table 4.27Regulatory Relative Delta Framework: Estimates for MinimumRegulatory Capital for UK Banks from the Maximum Absolute Relative Delta
Adjusted to Specified Threshold Levels

Table Description

The Table 4.27 Panel A to Panel E columns represent the following:

The Year column represents the fiscal accounting year. Record Date is the date the variable values are recorded. *Obs.* represents the number of sample firms observed for the year.

For the Relative Delta Measures (δ) column:

E-M column represents the total equity to market price relative delta for the *year* time period t - 1 to t calculated by applying equation (3.38); *A-M* column represents the total assets to market price relative delta for the *year* time period t - 1 to t calculated by applying equation (3.38); *L-M* column represents the total liabilities to market price relative delta for the *year* time period t - 1 to t calculated by applying equation (3.38); *L-M* column represents the total liabilities to market price relative delta for the *year* time period t - 1 to t calculated by applying equation (3.38); and, *I-M* column represents the net income to market price relative delta for the *year* time period t - 1 to t calculated by applying equation (3.38).

The *Regulatory Relative Delta* $(\Delta R_{t-1,t})$ column represents the regulatory relative delta measure calculated by applying equation (3.51), and takes the general form:

$$\Delta R_t = |dAC - dM|_{t:model}$$

Where: ΔR_t is the Regulatory Relative Delta at time t, $|dAC - dM|_{t:max}$ is the absolute maximum of the *E-M*, *A-M*, *L-M*, and *I-M* relative delta levels at time t.

The *Regulatory Relative Delta Thresholds Levels* (λ) column presents illustrative threshold levels: 0%; 10%; 20%; 30%; 40%; 50% and 100% that may be applied to determine the activation level for the proposed regulatory relative delta adjustment. The *Adjusted Minimum Capital Requirement:* 8% × (1 + $\Delta R_{t-1,t}$) column represents the adjusted minimum capital requirement estimates for corresponding threshold levels calculated by applying expression (3.55), and takes the general form:

 $if \ \Delta R_t \leq \lambda \ then$ Minimum (Min.)Regulatory Capital = 8%

 $if \ \Delta R_t > \lambda \ then$ Adjusted Min. Regulatory Capital = 8% × (1 + $\Delta R_{t-1,t}$)

					Tal	ole 4.27 (Continued)							
Regula	ntory Relative Del	ta Framewo	rk for HSBC Maximu	' Holdings Pl n Absolute F	LC to Estima Relative Delt	PANEL A ate Changes to the F a Measure Adjusted	Regulatory 1 I to Specifie	Minimum d Thresho	Capital Re ld Levels	quirement	Levels Ca	lculated fi	om the
								Regulat	ory Relativ	e Delta Thr	esholds Le	vels (λ)	
Year	Record Date		Relative Delta	a Measure (δ)	Regulatory Relative Delta	0%	10%	20%	30%	40%	50%	100%
		E-M	A-M	L-M	I-M	$(\Delta R_{t-1,t})$	Ad	justed Mini	mum Capit	al Requirer	nent: 8% ×	$(1 + \Delta R_{t-})$	1.t)
1994	30-Dec-1994	0.477	0.299	0.289	0.458	0.477	11.82	11.82	11.82	11.82	11.82	8	8
1995	29-Dec-1995	0.573	0.205	0.198	0.542	0.573	12.58	12.58	12.58	12.58	12.58	12.58	8
1996	31-Dec-1996	-0.010	-0.121	-0.127	-0.003	0.127	9.01	9.01	8	8	8	8	8
1997	31-Dec-1997	-0.545	-0.524	-0.516	-0.493	0.545	12.36	12.36	12.36	12.36	12.36	12.36	8
1998	31-Dec-1998	-0.042	-0.030	-0.028	-0.270	0.270	10.16	10.16	10.16	8	8	8	8
1999	31-Dec-1999	-0.204	-0.267	-0.269	-0.170	0.269	10.15	10.15	10.15	8	8	8	8
2000	29-Dec-2000	0.322	0.112	0.098	0.190	0.322	10.57	10.57	10.57	10.57	8	8	8
2001	31-Dec-2001	0.236	0.258	0.262	0.038	0.262	10.09	10.09	10.09	8	8	8	8
2002	31-Dec-2002	0.193	0.146	0.144	-0.368	0.368	10.94	10.94	10.94	10.94	8	8	8
2003	31-Dec-2003	0.045	-0.044	-0.050	0.213	0.213	9.71	9.71	9.71	8	8	8	8
2004	31-Dec-2004	0.083	0.140	0.145	0.053	0.145	9.16	9.16	8	8	8	8	8
2005	30-Dec-2005	0.134	0.215	0.229	0.594	0.594	12.75	12.75	12.75	12.75	12.75	12.75	8
2006	29-Dec-2006	0.030	0.085	0.089	-0.230	0.230	9.84	9.84	9.84	8	8	8	8
2007	31-Dec-2007	0.263	0.318	0.323	0.279	0.323	10.58	10.58	10.58	10.58	8	8	8
2008	31-Dec-2008	0.229	0.636	0.656	-1.016	1.016	16.13	16.13	16.13	16.13	16.13	16.13	16.13

					Tal	ble 4.27 (Continued))						
				I DIG		PANEL B							
Re	gulatory Relative	Della Frame	Maximu	n Absolute H	Relative Delt	a Measure Adjusted	l to Specifie	d Thresho	ld Levels	rement Lev	vers Carcu	ated from	the
			Relative Delt	a Measure (δ)	Regulatory		Reg	ulatory Rel	ative Delta	Thresholds	(λ)	
Year	Record Date				-	Relative Delta	0%	10%	20%	30%	40%	50%	100%
		E-M	A-M	L-M	I-M	$(\Delta R_{t-1,t})$	Adjusted Minimum Capital Re	tal Requirer	rement: $8\% \times (1 + \Delta R_{t-1,t})$				
1994	30-Dec-1994	0.201	0.019	0.015	2.808	2.808	30.46	30.46	30.46	30.46	30.46	30.46	30.46
1995	29-Dec-1995	-0.050	-0.152	-0.156	-0.034	0.156	9.25	9.25	8	8	8	8	8
1996	31-Dec-1996	-0.269	-0.206	-0.203	-0.101	0.269	10.15	10.15	10.15	8	8	8	8
1997	31-Dec-1997	-0.438	-0.249	-0.241	-0.791	0.791	14.33	14.33	14.33	14.33	14.33	14.33	8
1998	31-Dec-1998	0.262	0.155	0.151	0.403	0.403	11.23	11.23	11.23	11.23	11.23	8	8
1999	31-Dec-1999	-0.243	-0.169	-0.166	-0.001	0.243	9.94	9.94	9.94	8	8	8	8
2000	29-Dec-2000	0.404	0.065	0.053	0.255	0.404	11.23	11.23	11.23	11.23	11.23	8	8
2001	31-Dec-2001	0.007	0.027	0.027	-0.097	0.097	8.77	8	8	8	8	8	8
2002	31-Dec-2002	0.435	0.512	0.521	-0.205	0.521	12.17	12.17	12.17	12.17	12.17	12.17	8
2003	31-Dec-2003	-0.177	-0.163	-0.162	0.107	0.177	9.42	9.42	8	8	8	8	8
2004	31-Dec-2004	-0.099	0.001	0.004	-0.022	0.099	8.79	8	8	8	8	8	8
2005	30-Dec-2005	-0.041	0.529	0.538	-0.007	0.538	12.30	12.30	12.30	12.30	12.30	12.30	8
2006	29-Dec-2006	-0.042	-0.103	-0.104	0.232	0.232	9.85	9.85	9.85	8	8	8	8
2007	31-Dec-2007	0.547	0.578	0.579	0.158	0.579	12.63	12.63	12.63	12.63	12.63	12.63	8
2008	31-Dec-2008	1.578	1.677	1.683	1.657	1.683	21.46	21.46	21.46	21.46	21.46	21.46	21.46

Table 4.27 (Continued)

PANEL C Regulatory Relative Delta Framework for The Royal Bank of Scotland PLC to Estimate Changes to the Regulatory Minimum Capital Requirement Levels Calculated from the Maximum Absolute Relative Delta Measure Adjusted to Specified Threshold Levels

		j	Relative Delta	Measure (δ))	Regulatory	Regulatory Relative Delta Thresholds (λ)						
Year	Record Date					Relative Delta	0%	10%	20%	30%	40%	50%	100%
		E-M	A-M	L-M	I-M	$(\Delta R_{t-1,t})$	Adj	justed Mini	mum Capit	al Requiren	nent: 8% ×	$(1 + \Delta R_{t-})$	1 <i>,t</i>)
1994	30-Sep-1994	-0.218	-0.016	-0.006	1.143	1.143	17.14	17.14	17.14	17.14	17.14	17.14	17.14
1995	29-Sep-1995	0.037	0.0379	0.0383	-0.018	0.038	8.31	8	8	8	8	8	8
1996	30-Sep-1996	0.076	0.097	0.096	0.164	0.164	9.31	9.31	8	8	8	8	8
1997	30-Sep-1997	-0.133	-0.173	-0.174	-0.304	0.304	10.43	10.43	10.43	10.43	8	8	8
1998	30-Sep-1998	0.001	0.123	0.130	0.424	0.424	11.39	11.39	11.39	11.39	11.39	8	8
1999	30-Sep-1999	-0.242	-0.556	-0.567	-0.447	0.567	12.54	12.54	12.54	12.54	12.54	12.54	8
2000	29-Dec-2000	4.229	1.010	0.983	0.633	4.229	41.83	41.83	41.83	41.83	41.83	41.83	41.83
2001	31-Dec-2001	0.140	0.087	0.084	0.209	0.209	9.67	9.67	9.67	8	8	8	8
2002	31-Dec-2002	0.096	0.227	0.234	-0.543	0.543	12.35	12.35	12.35	12.35	12.35	12.35	8
2003	31-Dec-2003	-0.062	-0.001	0.002	-0.203	0.203	9.63	9.63	9.63	8	8	8	8
2004	31-Dec-2004	0.072	0.186	0.193	2.710	2.710	29.68	29.68	29.68	29.68	29.68	29.68	29.68
2005	30-Dec-2005	0.110	0.284	0.298	0.328	0.328	10.62	10.62	10.62	10.62	8	8	8
2006	29-Dec-2006	0.008	-0.012	-0.016	0.001	0.016	8.13	8	8	8	8	8	8
2007	31-Dec-2007	0.721	1.181	1.186	0.565	1.186	17.48	17.48	17.48	17.48	17.48	17.48	17.48
2008	31-Dec-2008	2.129	2.251	2.266	-5.208	5.208	49.66	49.66	49.66	49.66	49.66	49.66	49.66

Table 4.27 (Continued)	ł
PANEL D	

Regulatory Relative Delta Framework for Lloyds Banking Group PLC to Estimate Changes to the Regulatory Minimum Capital Requirement Levels Calculated from the Maximum Absolute Relative Delta Measure Adjusted to Specified Threshold Levels

		Ì	Relative Delta	α Measure (δ`)	$\begin{array}{c c c c c c c c c c c c c c c c c c c $							
Year	Record Date				, 		50%	100%					
		E-M	A-M	L-M	I-M	$\left(\Delta R_{t-1,t}\right)$	Adj	justed Mini	mum Capit	al Requiren	nent: 8% ×	$\times (1 + \Delta R_{t-})$	_{1,t})
1994	30-Sep-1994	-	-	-	-	-	-	-	-	-	-	-	-
1995	29-Sep-1995	-	-	-	-	-	-	-	-	-	-	-	-
1996	30-Sep-1996	-	-	-	-	-	-	-	-	-	-	-	-
1997	30-Sep-1997	-	-	-	-	-	-	-	-	-	-	-	-
1998	31-Dec-1998	-	-	-	-	-	-	-	-	-	-	-	-
1999	31-Dec-1999	0.260	0.146	0.141	0.285	0.285	10.28	10.28	10.28	8	8	8	8
2000	29-Dec-2000	0.211	0.303	0.306	0.173	0.306	10.45	10.45	10.45	10.45	8	8	8
2001	31-Dec-2001	0.054	0.029	0.027	-0.135	0.135	9.08	9.08	8	8	8	8	8
2002	31-Dec-2002	0.255	0.581	0.597	-0.218	0.597	12.77	12.77	12.77	12.77	12.77	12.77	8
2003	31-Dec-2003	0.205	-0.007	-0.014	2.138	2.138	25.10	25.10	25.10	25.10	25.10	25.10	25.10
2004	31-Dec-2004	-0.018	0.050	0.054	-0.417	0.417	11.33	11.33	11.33	11.33	11.33	8	8
2005	30-Dec-2005	-0.010	0.069	0.073	-0.060	0.073	8.58	8	8	8	8	8	8
2006	29-Dec-2006	-0.063	-0.053	-0.052	0.064	0.064	8.52	8	8	8	8	8	8
2007	31-Dec-2007	0.280	0.219	0.218	0.292	0.292	10.34	10.34	10.34	8	8	8	8
2008	31-Dec-2008	1.094	1.529	1.542	0.460	1.542	20.34	20.34	20.34	20.34	20.34	20.34	20.34

PANEL E Regulatory Relative Delta Framework for Standard Chartered PLC to Estimate Changes to the Regulatory Minimum Capital Requirement Levels Calculated from the Maximum Absolute Relative Delta Measure Adjusted to Specified Threshold Levels

 Table 4.27 (Continued)

			Relative Delta	$Measure (\delta)$	$(\delta) \qquad \qquad$								
Year	Record Date				, 	Relative Delta	0%	10%	20%	30%	40%	50%	100%
		E-M	A-M	L-M	I-M	$(\Delta R_{t-1,t})$	Adj	iusted Mini	mum Capit	al Requiren	nent: 8% ×	$\lesssim (1 + \Delta R_{t-1})$	_{1,t})
1994	30-Dec-1994	0.281	0.172	0.168	0.508	0.508	12.06	12.06	12.06	12.06	12.06	12.06	8
1995	29-Dec-1995	-0.382	-0.540	-0.547	-0.291	0.547	12.37	12.37	12.37	12.37	12.37	12.37	8
1996	31-Dec-1996	-0.084	-0.193	-0.198	0.026	0.198	9.58	9.58	8	8	8	8	8
1997	31-Dec-1997	0.192	0.214	0.215	0.087	0.215	9.72	9.72	9.72	8	8	8	8
1998	31-Dec-1998	0.035	-0.055	-0.060	-0.282	0.282	10.26	10.26	10.26	8	8	8	8
1999	31-Dec-1999	-0.128	-0.199	-0.202	-0.588	0.588	12.70	12.70	12.70	12.70	12.70	12.70	8
2000	29-Dec-2000	0.262	0.233	0.228	1.012	1.012	16.09	16.09	16.09	16.09	16.09	16.09	16.09
2001	31-Dec-2001	0.353	0.235	0.224	-0.181	0.353	10.82	10.82	10.82	10.82	8	8	8
2002	31-Dec-2002	0.047	0.098	0.110	-0.334	0.334	10.67	10.67	10.67	10.67	8	8	8
2003	31-Dec-2003	-0.320	-0.311	-0.314	-0.017	0.320	10.56	10.56	10.56	10.56	8	8	8
2004	31-Dec-2004	-0.029	0.045	0.049	0.194	0.194	9.55	9.55	8	8	8	8	8
2005	30-Dec-2005	0.285	0.238	0.248	0.399	0.399	11.19	11.19	11.19	11.19	8	8	8
2006	29-Dec-2006	0.103	-0.060	-0.068	-0.171	0.171	9.36	9.36	8	8	8	8	8
2007	31-Dec-2007	0.005	-0.015	-0.015	0.002	0.015	8.12	8	8	8	8	8	8
2008	31-Dec-2008	1.082	1.217	1.231	1.227	1.231	17.84	17.84	17.84	17.84	17.84	17.84	17.84

Analysis

Table 4.27 presents the regulatory relative delta measurement framework's results for the UK banks. This framework operates by adjusting the minimum regulatory capital levels using the absolute maximum relative delta measure from the key accounting totals. This regulatory relative delta framework is proposed in this study to inform the regulatory risk monitoring and management framework at firmwide and regional levels.

Examining Table 4.27 for the fiscal years 2004 to 2008, the regulatory relative delta component, ΔR , for the banks *E-M* relative delta measure reveals the changes presented in Table 4.25. Effectively, Table 4.25 shows the adjustments, based on the levels exhibited from the *total equity to market price relative delta* component, applied to the 8% Basel minimum capital requirement.

Table 4.25 shows that this capital adjustment alters the 8% capital requirement for the 2004 year by 0.2%. For years 2005, 2007 and 2008 the adjustment effectively increases the minimum capital reserve requirement by 9.5%, 36.3% and 122.2% respectively.

4.11 Regulatory Capital Measurement and the Technology Framework

As part of this research's discovery, this study proposes the relative delta measure to inform the regulatory framework of a firm's or a portfolio of firms' exposure to financial distress and other risks arising from accounting and market price information. It is proposed that these risks have the ability to be aggregated and reported at national, regional and the global regulatory levels.

Table 4.13 and Table 4.22 present results for the relative delta measure that are calculated by extending the Fama and French (2008) time dependent treatment for the book-to-market ratio measure. As referred to earlier, such an approach, by just measuring the equity to market component alone, substantiates the promise to measure levels of exposure to financial distress risk of a single firm or a portfolio of firms. This study proposes that monitoring the information exhibited in Table 4.13, at an intrayearly or even an intra-daily frequency, would provide an effective model to help measure, at minimum, exposures to financial distress risk at the firmwide and regulatory levels. Effectively this study proposes extending the Fama and French (2008) book-tomarket ratio to a model that is able to monitor the system of relative deltas based on: total equity to market price, total assets to market price, total liabilities to market price and net income to market price. In addition, the proposed measurement model is specified to measure differences in accounting Value-at-Risk and market price return Value-at-Risk. This proposed Value-at-Risk measurement model is similar to that exhibited in Table 4.19 and Table 4.24. As detailed in Section 4.7.2, this model reports the same levels quantified by the system of relative deltas.

This system of relative deltas is applied in this study to develop the regulatory relative delta framework specified in Section 3.9.6. The regulatory relative delta framework is proposed to measure the potential *add-on* effects to the minimum regulatory capital requirement as a consequence of the levels recorded from the accounting to market price relative delta measures. In effect, the regulatory relative delta framework proposes to report information similar to that presented in Table 4.27.

To effectively implement the proposed relative delta, Value-at-Risk and regulatory relative delta frameworks, this study recommends monitoring accounting totals and market price information on a less than a yearly frequency. This study proposes that the

technology framework applied to monitor accounting and market price information should ideally report information at a *real-time* or *near real-time* basis.

Adding to this research's findings, a schematic overview of an ideal bank technology framework presented in Figure 4.1 is proposed. The principal feature of this framework is the application of a real-time pricing process to the contemporary technology framework presented in Section 2.13, Figure 2.1.

The proposed framework presented in Figure 4.1 shows that at time t = T, denoted by t_T (Transaction/Trade time), transacted items from the trade process are posed to both the settlement process and the accounting process simultaneously. The *pre-settlement* items posted directly on to the accounting system at time t_T are updated with current or *real-time* price information at time t = C, denoted by t_C (Current time, effectively the time stamp $t_T = t_C$). At time t_{T+S} the settled items are posted to the accounting system, and reconciled with the corresponding items posted to the accounting process prior to settlement at time $t_T = t_C$. The *reconciled* items recorded on the accounting system continue to be updated at time t_C with current price information, until the items are derecognised and archived.

Such a framework is proposed to allow the measurement of individual financial items, accounting totals and the market price information, on a *real-time* or close to real-time basis. This real-time valuation process attached to the accounting process would avoid any potential delays during the settlement process and provide an instantaneous view of a firm's accounting quality and exposure to risks arising from accounting and market price information bias. For settled items, the *valuation* or *pricing* process would remain to permit the instantaneous monitoring of post settlement exposures to these risks.

From this research's findings, the proposed framework presented in Figure 4.1 is expected to be forward looking, secure and highly transparent to both financial products and technology. It is proposed to be an adaptable system to changes in financial instrument products and technology interfaces, platforms and processes. That is, the proposed system possesses qualities that enable an encouragingly manageable integration and replacement of financial instruments and other bank product lifecycles, and the technology product lifecycles.

Figure 4.1Proposed Bank Technology Framework OverviewProposed banking technology framework for financial instrument and financial item: pricing, trade, risk management, and accounting.



4.12 Summary

This chapter has presented the results and analyses from the tests specified in Chapter 3. These tests were applied to determine the level of verifiability, or conversely falsifiability, for this study's null hypothesis, presented in Section 3.3.1, and the detailed set of null hypotheses, presented in Section 3.3.2.

From the basis of tests applied at both the individual bank and Primary sample levels, the results show that the UK banks, after 2005, were exposed to market price volatility, increased Value-at-Risk and an increase to the Basel minimum capital requirement. When testing accounting totals, the levels for total equity, total assets and total liabilities exhibited materially and statistically significant increases after 2005. The exception is the net income total that generally exhibited varying levels after 2005. These results indicate that after 2005 the banks became exposed to increased levels of financial distress risk and experienced deterioration in accounting quality, and thus experienced an increase in the level of exposure to accounting quality risk. The same tests conducted on the Secondary sample largely confirmed these findings. The control group of firms, tested to control for the 2005 accounting change, mostly exhibited variable levels that were significantly different to the Primary and Secondary sample, especially for the time period 2005 to 2007. For this time period, generally the control group exhibited lower levels of financial distress risk and improved accounting quality.

An interesting finding is that the tests conducted for GDP and selected market indices, produced similar results before and after 2005 for both the samples and the control group market price variable. These results indicate an arguable finding that the bank market price volatility observed after 2005, especially between 2005 and 2007, were unlikely to have been due to reactions to these economic and market indicators.

From the basis of these results, there is evidence to suggest that the relative delta, accounting Value-at-Risk and regulatory relative delta measures, that are applied to test this study's null hypothesis, has the proposed potential to report risks that arise from accounting and market price information bias. From this finding, a technology architecture that has the capability to report these measures effectively at the bank and regulatory levels is prescribed in this chapter.

Chapter 5 Conclusion

This research set out to examine whether accounting quality improved for the UK banks that adopted in 2005 the IASB framework's IFRS accounting standards (see Chapter 1). This study has quantified the changes in the level of accounting quality experienced by UK banks after the 2005 accounting change, its relation to levels of Value-at-Risk, and its effect on the Basel minimum regulatory capital levels. The theoretical and empirical literature in this subject area that is reviewed in Chapter 2, generally examines European firms during this adoption event. This study extends this literature by focusing examination on the UK banking sector. In so doing, this study begins by first *formulating* and then asking the question:

Does accounting quality improve for UK banks that adopted the IFRS accounting standards in 2005?

In order to address this research question, in Chapter 3, Section 3.3.1, the following null hypothesis is formally introduced:

Accounting quality in UK banks was not affected by the adoption of the IFRS accounting standards in 2005. For the UK banks that adopted the new accounting standards in 2005, there is no significant difference between the change in accounting totals and the change in market price after 2005 when compared to before.

Chapter 3 then specifies the methodological approach and details the tests applied to determine the level of verifiability or falsifiability of the null hypothesis. The results from these tests are presented at summary and detailed levels in Chapter 4.

Section 4.2.6, Table 4.4 Panel B, and Section 4.8, Table 4.22, present at summary and detailed levels respectively, *key* results for the relative delta measure specified in Section 3.2.4 that is applied to test the set of detailed null hypotheses specified in Section 3.3.2.

The evidence from this test, focusing on the *total equity to market price relative delta* applied to examine the null hypothesis, shows an interesting result. For the time range 1993 to 2008, the evidence favours more, at approximately the 80% level, the research hypothesis (see Section 3.4.1). This result, concluded in the first panel in Table 5.6, weakens the null hypothesis and finds that the adopting UK banks did not show an improvement in accounting quality after 2005.

Critical examination of this result shows that the banks did not exhibit an improvement in accounting quality and suggests that they experienced a significant deterioration in accounting quality between 2005 and 2008. This evidence, from the UK banking sector, supports the findings from the sampled firms in Europe reported by Platikanova and Nobes (2006), Paananen and Lin (2009) and Morais and Curto (2008). This result also suggests that within the risk management framework a strong implicit accounting role became evident after 2005, as opposed to the normal explicit role that financial accounting is expected to play.

However, this 80% null hypothesis falsification level is based entirely on the individual bank's *total equity to market price relative delta measure* results applied to test the corresponding *detailed null hypothesis* in equation (3.23). It does not signify evidence from the other tests conducted on the individual banks singularly or as a primary sample, nor does this falsification level take into consideration the secondary sample results. Importantly, this result does not take into account the control sample's reactions before and after 2005.

To attempt to strengthen the null hypothesis, and to examine the important requirements set by the EC (1992) and IASB (2009), further analysis is performed by testing the statements that after 2005: *market price returns increased*; *market price return volatility levels decreased*; and *market price Value-at-Risk decreased*. The outcomes to these statements are presented in Table 5.1. The conclusion drawn from Table 5.1 is that the null hypothesis is further weakened and the research hypothesis further strengthened.

To build evidence to show that significant effects were experienced by the sample market price variable after 2005 due to normal economic and financial market conditions, the statements presented in Table 5.2 are determined. It may be contended that for these statements to add strength to the null hypothesis, it is expected that the

Control group would show significantly different results when compared to the samples after 2005. Such an outcome, would suggest that any periods of volatility experienced by the UK banks arose as a consequence of normal economic and market conditions, and not as a consequence to other influences such as the IFRS accounting standards adoption.

Conversely, if the same level of reaction were signified by the samples and the control group after 2005 for a tested statement, the study would find that the economic factors (tested using GDP levels) and market factors would not have influenced the samples' market prices to a significant level compared to the Control group's market prices before and after 2005. Such an outcome would falsify the tested statement and thus weaken the null hypothesis.

From Table 5.2 the conclusion drawn is that the statements evaluated to corroborate the null hypothesis based on economic and market conditions are falsified. This conclusion deduces that any significant sample differences between the accounting and market price change variables, and thus significant bank *difference component* levels, after 2005 would be as a consequence of factors external to the tested statements. The conclusion drawn from this test is the same as the previous test, resulting in the null hypothesis being further weakened and the research hypothesis strengthened.

The *statement based conclusive evidence* provided thus far has focused on the *market price variable* and the *market price Value-at-Risk variable*. To build definitive evidence to verify the null hypothesis and to claim that the null hypothesis has been wrongly weakened, that is, being subject to a *Type I statistical error*, conclusions are now drawn from tests performed on the accounting and market price variables.

To build evidence to strengthen the null hypothesis, it is expected for the samples and the control group to report significantly similar levels of differences between the accounting and the market price variables after the 2005 accounting standards change. When systematically examining evidence for this study's *null hypothesis* on the basis of statements evaluated to corroborate the *detailed null hypotheses*, the following outcome is expected:

Given that the market price returns did not increase after 2005, the percentage change in accounting totals was not significant from before to after 2005 for the Primary and the Secondary sample compared to the Control group

Testing this statement, the outcomes presented in Table 5.3 conclude that each of the detailed null hypotheses has become weakened. That is, compared to the Control group, the accounting total levels for the samples exhibit a significant increase after 2005. Proceeding then to examine the evidence drawn from the statements based on the *relative delta* and *regulatory relative delta* measures, presented in Table 5.4 to Table 5.11, an arguable conclusion is revealed. This general conclusion arguably further weakens the null hypothesis and strengthens the alternative research hypothesis from the approximate 80% to an approximate 90% level.

It had been expected that the objectives of the EC (1992) and IASB (2009) to increase market stability and increase value relevance - reflected by an improvement in accounting quality, would have been realised in the UK banking sector after the 2005 accounting change. Instead, this study reveals that accounting quality for the UK banks had declined significantly from 2005 to 2008.

This conclusion suggests that within the risk management framework a strong implicit accounting role became evident for the UK banks after 2005, as opposed to financial accounting's expected explicit role.

However, what requires further investigation is whether this observed decline in accounting quality after 2005 had any part to play in the observed volatility after 2005. In Section 2.9.3, it is stated that a decrease in accounting quality is revealed when an increase in financial distress risk is evident. Based on this reasoning and drawing from the conclusion reported in Table 5.11 and principally the evidence from Table 4.25, it is evident that the level of exposure to financial distress risk experienced by UK banks increased from a 0.2% level in 2004 before the accounting change, to a 9.5% level in 2005. This exposure increased significantly to 36.3% and 122.2% in 2007 and 2008 respectively. This gradually increasing level of financial distress risk from 2005 to 2008, reflected by declining accounting quality, may be contended to have produced investor uncertainty. This uncertainty, in turn, may be argued to have contributed to the observed market price volatility, especially from 2005 to 2007.

5.1 Conclusive Statement Analysis

Table 5.1 to Table 5.11, present a sequence of concluding statements that are aligned to this study's null hypothesis. The statements are constructed in such a way that they are easily testable and are similar to the statements termed 'predictions' by Popper (1992, p 33). These statements present evidence that conclusively examines the null hypothesis specified in Section 3.3.1 and equation (3.22) and the detailed null hypotheses specified in Section 3.3.2 and equations (3.23) to (3.26).

Table 5.1 to Table 5.11 are organised to first present the concluding statements being evaluated, and when presented, followed by corresponding variables and equations. The table then indicates whether the findings drawn in this study result in evidence to verify or falsify these statements. The tables also record the literature supported by this evidence. In addition, the tables present a description of the analysis method applied and the *results tables* used to produce the concluding evidence.

Table 5.1Conclusions for Market Price Returns, Market Price Return
Volatility and Market Price Return Value-at-Risk

The table shows in the Statement Tested column the statement examined to determine if the evidence from this study's results provide strength to verify or falsify the statement. The Verified or Falsified column presents the conclusion met, based on the analysis of the results. The Literature Supported column presents the literature that this study supports based on the evidence of the statement tested. The Analysis column presents the quantitative analytical method applied to test the statement. The Tables column presents the result tables analysed.

Statement Tested	Verified or Falsified	Literature Supported	Analysis Method	Tables
<i>Market price returns</i> increased after 2005	falsified	-	Mean	
		Walton and Aerts (2006)		
Market price return volatility		Walton (2004)	Standard	1 1 1 1 1 1
levels decreased after 2005	after 2005 falsified Cheney (2008)	Cheney (2008)	Deviation	Table 4.1 Table P.1
		IASB1 (2008)		
		Wild and Poole (2008)		
Market price Return Value-at-	falsified	Leone (2008)	VaR models: Var-covar,	Table 4.2 Table P.2
<i>Risk</i> decreased after 2005		Platikanova and Nobes (2006)	Historical, MCS	

Table 5.2Conclusions for GDP and General Market Conditions to the Samples and the Control Group Market Price Returns and
Value-at-Risk

The table shows in the Statement Tested column the statement examined to determine if the evidence from this study's results provide strength to verify or falsify the statement. The Verified or Falsified column presents the conclusion met, based on the analysis of the results. The Literature Supported column presents the literature that this study supports based on the evidence of the statement tested. The Analysis column presents the quantitative analytical method applied to test the statement. The Tables column presents the result tables analysed.

Statement Tested	Verified or Falsified	Literature Supported	Analysis Method	Tables
Selected <i>GDP</i> levels showed significantly different magnitude and direction effects with the <i>Primary and Secondary sample</i> when compared with the <i>Control group</i> returns and VaR	partially falsified	Walton and Aerts (2006)		Table P.8 Table P.9
Selected <i>stock market indices</i> levels showed significantly different magnitude and direction effects with the <i>Primary and Secondary sample</i> when compared with the	falsified	Walton (2004)		Table P.11
Control group returns and VaR		Cheney (2008)		Table P.12
Selected <i>LIBOR rates</i> showed significantly different magnitude and direction effects with the <i>Primary and Secondary sample</i> when compared with the <i>Control group</i> returns	falsified	IASB1 (2008)	Time series	Table P.13
and VaR		Wild and Poole (2008)	regression	Table P.14
Selected GILT Government bond rates showed significantly different magnitude and direction effects with the Primary and Secondary sample when compared with the	partially	Leone (2008)		Table P.15
Control group returns and VaR	laisined	Platikanova and Nobes		Table P.10
Selected <i>foreign exchange rates</i> showed significantly different magnitude and direction effects with the <i>Primary and Secondary sample</i> when compared with the <i>Control group</i> returns and VaR	partially falsified	(2006)		Table P.17 Table P.18

Table 5.3 Conclusions for Null Hypothesis Tests Based on Accounting Totals

Statement Tested	Accounting Total Tested	Detailed Null Hypothesis Tested	Verified or Falsified	Literature Supported	Analysis Method	Tables
Given that the <i>market price returns</i> did not increase after	Total Equity	$H_{0:E}:\delta EM_{t\leq 2004}=\delta EM_{t\geq 2005}$	falsified	Platikanova and Nobes (2006)		
2005, the percentage change in <i>accounting totals</i> was not	Total Assets	$H_{0:A}: \delta AM_{t \le 2004} = \delta AM_{t \ge 2005}$	falsified	Paananen and	Descriptive Statistics	Table 4.3
significant from before to after 2005 for the Primary and	Total Liabilities	$H_{0:L}:\delta LM_{t\leq2004}=\delta LM_{t\geq2005}$	falsified	Lin (2009)		Table R.4
Secondary sample compared to the Control group	Net Income	$H_{0:I}: \delta IM_{t \le 2004} = \delta IM_{t \ge 2005}$	falsified Morais and Curto (2008)			

Table 5.4 Conclusions for Null Hypothesis Tests Based on the Change in Accounting Totals to Change in Market Price

Statement Tested	Accounting Total Tested	Detailed Null Hypothesis Tested	Verified or Falsified	Literature Supported	Analysis Method	Tables
The difference between the change in accounting	Total Equity	$H_{0:E}:\delta EM_{t\leq 2004}=\delta EM_{t\geq 2005}$	falsified	Platikanova and Nobes (2006)		Table 4.6 Table U.1
totals were not significantly different to the change in market price	Total Assets	$H_{0:A}: \delta AM_{t \le 2004} = \delta AM_{t \ge 2005}$	falsified	Nobes (2000)TirPaananen andandLin (2009)seregMorais and	Time series and Cross-	Table 4.8 Table U.3 Table 4.11
after 2005 for the Primary and Secondary sample and this difference was less	Total Liabilities	$H_{0:L}:\delta LM_{t\leq 2004} = \delta LM_{t\geq 2005}$	-		regression	Table 4.9 Table U.4
than the Control group	Net Income	$H_{0:I}: \delta IM_{t \le 2004} = \delta IM_{t \ge 2005}$	-	Curto (2008)		Table 4.10 Table U.5

Table 5.5 Conclusions for Null Hypothesis Tests Based on Accounting to Market Price Relative Delta

Statement Tested	Accounting Total Tested	Detailed Null Hypothesis Tested	Verified or Falsified	Literature Supported	Analysis Method	Tables
Using the Relative Delta Measure	Total Equity	$H_{0:E}:\delta EM_{t\leq 2004}=\delta EM_{t\geq 2005}$	falsified	Platikanova and Nobes (2006)		
change in accounting totals was not significantly different to the	Total Assets	$H_{0:A}: \delta AM_{t \le 2004} = \delta AM_{t \ge 2005}$	falsified	Paananen and	Relative changes	Table 4.4 (Panel A)
for the Primary and Secondary sample and this difference was	Total Liabilities	$H_{0:L}:\delta LM_{t\leq 2004} = \delta LM_{t\geq 2005}$	falsified	Lin (2009) Morais and		Table 4.12 Table 4.13
less than that of the Control group	Net Income	$H_{0:I}:\delta IM_{t\leq 2004}=\delta IM_{t\geq 2005}$	falsified	Curto (2008)		

Table 5.6 Conclusions for Null Hypothesis Tests Based on Accounting to Market Price Relative Delta for UK Banks

Statement Tested	Accounting Total Tested	Detailed Null Hypothesis Tested	Falsification Level	Literature Supported	Analysis Method	Tables
	Total Equity	$H_{0:E}:\delta EM_{t \le 2004} = \delta EM_{t \ge 2005}$	falsified 4 banks from 5	Platikanova and		
For the UK banks, using the Relative Delta Measure - The difference between the change in accounting totals	Total Assets	$H_{0:A}: \delta AM_{t \leq 2004} = \delta AM_{t \geq 2005}$	falsified 5 banks from 5	Nobes (2006) Paananen and	Relative	Table 4.4 (Panel B)
different to the change in market price after 2005	Total Liabilities	$H_{0:L}:\delta LM_{t\leq 2004} = \delta LM_{t\geq 2005}$	falsified 5 banks from 5	d Lin (2009) om 5 Morais and	changes	Table 4.21 Table 4.22
-	Net Income	$H_{0:I}: \delta IM_{t \le 2004} = \delta IM_{t \ge 2005}$	falsified 4 banks from 5			

Table 5.7 Conclusions for Null Hypothesis Tests Based on Relative Delta and Market Price Value-at-Risk

Statement Tested	Accounting Total Tested	Detailed Null Hypothesis Tested	Verified or Falsified	Literature Supported	Analysis Method	Tables
The relative delta measure showed no significance to market price Value-at-Risk after 2005 for the Primary and Secondary sample and for the Control group	Total Equity	$H_{0:E}: \delta EM_{t \le 2004} = \delta EM_{t \ge 2005}$	falsified		Time series and Cross-sectional regression	Table 4.5 (Panel A) Table 4.14
	Total Assets	$H_{0:A}: \delta AM_{t \le 2004} = \delta AM_{t \ge 2005}$	falsified	Platikanova and Nobes (2006) Paananen and		Table 4.5 (Panel A) Table 4.15 Table 4.5 (Panel A) Table 4.16
	Total Liabilities	$H_{0:L}: \delta LM_{t \le 2004} = \delta LM_{t \ge 2005}$	falsified	Lin (2009) Morais and Curto (2008)		
	Net Income	$H_{0:I}: \delta IM_{t \le 2004} = \delta IM_{t \ge 2005}$	falsified			Table 4.5 (Panel A) Table 4.17

Table 5.8 Conclusions for Null Hypothesis Tests Based on Relative Delta and Market Price Value-at-Risk for UK Banks

Statement Tested	Accounting Total Tested	Detailed Null Hypothesis Tested	Falsification Level	Literature Supported	Analysis Method	Tables
For the UK banks, the relative delta measure showed no significance to market price Value-at-Risk after 2005	Total Equity	$H_{0:E}:\delta EM_{t\leq 2004}=\delta EM_{t\geq 2005}$	partially falsified 3 banks from 5	Platikanova and Nobes (2006)	Time series and Cross-sectional regression	Table 4.5 (Panel B) Table V.5 to Table V.9
	Total Assets	$H_{0:A}: \delta AM_{t \le 2004} = \delta AM_{t \ge 2005}$	falsified 5 banks from 5	Paananen and Lin (2009)		
	Total Liabilities	$H_{0:L}:\delta LM_{t \le 2004} = \delta LM_{t \ge 2005}$	partially falsified 3 banks from 5	Morais and Curto (2008)		
	Net Income	$H_{0:I}: \delta IM_{t \le 2004} = \delta IM_{t \ge 2005}$	partially verified 1 bank from 5	EC (2002) IASB (2009)		

Table 5.9 Conclusions for Null Hypothesis Tests Based on Accounting Value-at-Risk and Market Price Value-at-Risk

Statement Tested	Accounting Total Tested	Detailed Null Hypothesis Tested	Verified or Falsified	Literature Supported	Analysis Method	Tables
	Total Equity	$H_{0:E}:\delta EM_{t\leq 2004}=\delta EM_{t\geq 2005}$	falsified	Platikanova and Nobes (2006) Paanapen and	Material	Table 4.18
For the Primary and Secondary sample the accounting VaB	Total Assets	$H_{0:A}:\delta AM_{t\leq 2004}=\delta AM_{t\geq 2005}$	falsified			
showed similar levels to the Market Price VaR after 2005	Total Liabilities	$H_{0:L}:\delta LM_{t\leq 2004}=\delta LM_{t\geq 2005}$	falsified	Lin (2009)	differences	Table 4.19
	Net Income	$H_{0:I}: \delta IM_{t \le 2004} = \delta IM_{t \ge 2005}$	partially falsified	Morais and Curto (2008)		

Table 5.10 Conclusions for Null Hypothesis Tests Based on Accounting Value-at-Risk and Market Price Value-at-Risk for UK Banks

Statement Tested	Accounting Total Tested	Detailed Null Hypothesis Tested	Falsification Level	Literature Supported	Analysis Method	Tables
For the UK banks the accounting VaR showed similar levels to the Market Price VaR after 2005	Total Equity	$H_{0:E}:\delta EM_{t\leq 2004} = \delta EM_{t\geq 2005}$	partially falsified 3 banks from 5	Platikanova and	Material differences	Table 4.23 Table 4.24
	Total Assets	$H_{0:A}:\delta AM_{t \le 2004} = \delta AM_{t \ge 2005}$	falsified 4 banks from 5	Nobes (2006) Paananen and Lin (2009) Morais and		
	Total Liabilities	$H_{0:L}:\delta LM_{t\leq 2004} = \delta LM_{t\geq 2005}$	falsified 4 banks from 5			
	Net Income	$H_{0:I}: \delta IM_{t \le 2004} = \delta IM_{t \ge 2005}$	partially falsified 4 banks from 5	Curto (2008)		

Table 5.11 Conclusions for Null Hypothesis Tests Based on Regulatory Relative Relative Delta for UK Banks

Statement Tested	Null Hypothesis Tested	Verified or Falsified	Literature Supported	Analysis Method	Table
For the UK banks, the level of minimum regulatory capital did not increase after 2005 when applying the regulatory relative delta adjustment	$H_0: \delta_{t \le 2004} = \delta_{t \ge 2005}$	falsified	Platikanova and Nobes (2006) Paananen and Lin (2009) Morais and Curto (2008)	Relative changes	Table 4.25 Table 4.26 Table 4.27

5.2 Extension to the Fama and French Time Dependent Treatment for the Bookto-Market Ratio

This study has extended and applied the Fama and French (2008) book-to-market ratio's time dependent treatment by developing a system of equations that measures the difference between the change in the financial statement accounting totals: total equity, total assets, total liabilities and net income, to the change in market price. This measure of *difference* represented by the *difference component* is termed the *relative delta measure*. The relative delta measure has been developed to provide direct tests for the series of detailed null hypotheses specified in Section 3.3.2. Founded on the results from these direct tests, this study has built evidence to determine the level of acceptance or rejection for its null hypothesis (see Sections 4.2.6, 4.5, 4.8 and Table 5.7 and Table 5.8). This study has also extended the Fama and French (2008) book-to-market ratio treatment, together with the Basel regulatory policy (Basel 2011), to produce the regulatory relative delta framework. As presented in Section 4.10, the regulatory relative delta framework has been applied to test directly the individual banks in the UK banking sector. The conclusion from this test is tabulated in Table 5.11.

5.3 Fama and French Baseline Regression and its Inference to this Study

Table Q.1 presents the Fama and French (2008) baseline regression results for the bookto-market ratio and market price return applied to the Primary and Secondary sample and the Control group. These results exhibit a similar pattern when compared to the Fama and French (2008, p 2979, *Table I*) results. However, this study's predictions for expected market price returns using the market value alone do not exhibit strong explanatory evidence.

The Fama and French (2008) sample consisted of all the stocks in the New York Stock Exchange (NYSE) from 1927 to 2006. From the Fama and French results it is determined that the larger the magnitude of the regression slope the higher the returns, and the higher the book-to-market ratio measure, and thus the greater the difference between its book value and market value. Fama and French exhibits a significant slope at 0.21 for stocks listed on the NYSE from 1927 to 2006 that are above the 20th percentile of capitalisation (termed *high capped* stocks). With a slope of 0.3 for stocks *below* the 20th percentile of capitalisation (termed *low capped* stocks).

The baseline regression in this study includes the application of *financially distressed* 2008 stock prices to predict expected market price returns using the market value and the book-to-market ratio measured during 2007. Together with the market value effects, the slopes exhibit evidence of *skewing the 2005 accounting change effect* to produce a higher baseline book-to-market ratio slope for returns before 2005 than after. Taking this effect into account, this study's results produce a slope with a magnitude of 0.667 after 2005 and 0.728 before 2005 for the Primary and Secondary sample; and, 0.767 before 2005 and 0.710 after 2005 for the Control group. If evaluating the regression slopes to be comparative to the difference between the change in *total equity per share* and the change in market price, then this study exhibits greater levels of *difference* than the slopes from the Fama and French baseline regression. The time range applied in this study for its baseline regression is from 1992 to 2007.

By examining the results in Table P.11, it is observed that the relationship between market price returns for the NYSE Dow Jones Industrials index and the LSE FTSE indices is statistically significant from 1994 to 2008. In general, if it may be implied that this same relationship was exhibited from 1927 to 2006 between the indices, then it may be contended that a relationship for the book-to-market ratio observed for stocks registered on the NYSE exchange would have a similar relative relationship for the book-to-market ratio measured for stocks registered on the LSE exchange. If this reasoning is to hold true then it may be said that the levels observed for the book-to-market ratio slopes for the samples and the control group that comprise firms from the LSE, observed an average level of *difference* after 2005 that is greater than the level observed from 1927 to 2006. Therefore, it may be suggested that the levels of difference between the accounting and market price change variables reported in this study, including the *differences* presented in Table 4.25, may be greater than those observed since 1927. In Chapter 6, it is recommended to test the historical effects implied in this section by testing historical data that extends further than 1994.

5.4 Regulatory and Firmwide Technology Framework Findings

This study proposes to inform the banking regulatory authorities by implementing the regulatory relative delta measurement framework, specified in Section 3.9.6 and detailed in Appendix K. This regulatory relative delta framework is proposed to better monitor regulatory capital levels. In addition, the regulatory relative delta framework

and the relative delta measure are proposed in this study to provide firms with risk management tools to better facilitate regulatory and firmwide risk management aims. These aims aligned to accurately monitor, maintain and report minimum regulatory capital requirements.

In addition, this study proposes the relative delta measure to inform the accounting standards and firmwide supervisory authorities. The relative delta measure is, in this instance, proposed to report accounting quality, by measuring accounting totals reported by firms under accounting guidelines and investor reactions to these totals reported by the market price.

This study has approached the implementation for such a reporting framework by specifying a technology framework that is intended to integrate more closely the entire banking transaction function. This study proposes that such a system has the ability to synchronise financial instrument and item prices at a *real-time* or *near real-time basis* to its accounting function. The overview for this system is introduced in Section 4.11 and its schematic overview presented in Figure 4.1.

It was stated in the introduction to this thesis that accounting has two roles it may play in risk management: the explicit and the implicit accounting roles. Implementing the accounting total to market price relative delta measures, introduces a third role accounting may play in risk management. This role being a hybrid of the explicit and implicit roles where the function of accounting is applied to help monitor potential financial risks that are attributed to accounting quality.

5.5 Policy Implications

The Basel III regulatory policy (Basel 2011) is the principal banking regulatory framework applied by regional regulators. These regulations are aimed at quantifying a bank's ability to handle potential financial loss by measuring a bank's capital adequacy level. Since its formal inception in 1988 (Basel 1998), the Basel policies have evolved by addressing *credit risk, market risk* and aspects of *operational risk*. The measurement framework that Basel regulations apply is based on, in principle, the accounting measures: *Total Assets* whose items are primarily applied to calculate the Basel *risk weighted assets* component; and, *Total Shareholders' Equity* whose items are primarily

applied to calculate a bank's capital adequacy level. Although the Basel framework is heavily reliant on accounting based measurements, the policy does not factor adjustments to address accounting quality issues.

The findings from this thesis show that the *regulatory relative delta measure* has the ability to adjust the Basel minimum regulatory capital levels. In so doing, the effects from *measurement bias* between accounting and market valuations would be factored into the Basel regulatory framework. This study recommends that a measure in keeping with the sophisticated measurement framework proposed by the *regulatory relative delta* be applied within the regulatory policy framework. This study recommends, from its findings, that by incorporating such a measurement into policy would provide both regulators and firmwide risk managers a more accurate perspective to the risks faced by a banking institution.

In addition, it is proposed that *unregulated* internal firmwide risk management policies should also incorporate the measure termed the *relative delta*. This measure is developed in this thesis as a measure of the *difference* between the change in accounting totals and the change in market price. This *difference component* also quantifying a firm's level of exposure to financial distress risk. Incorporating the relative delta measure would inform the risk management processes of bias between accounting totals and market valuations. In addition, the measure would guide risk managers to the financial instrument level, where the *difference component* and thus risk exposures are reported to be the highest.

From this research's findings, it is also evident that informing the accounting framework of accounting total and market price *difference* levels is just as important as informing the regulatory and firmwide risk management frameworks. Therefore, it is also proposed that accounting standard policies and standard setters would find it beneficial to stipulate the measures specified principally by the *relative delta* and also the *regulatory relative delta*.

The findings in this research also have implications for policies created by Governments, the European Community and other global regional communities. Administered by central financial regulatory bodies and central banks, the governing bodies now have a process that contributes to help determine the level of effectiveness economic policy has when aimed at stabalising regional and global financial markets.

In summary, such policy implementations are proposed in order to bring stability to the financial system by identifying and mitigating risks that may arise from information bias between accounting and market price data.

5.6 Managerial Implications

Managerial implications based on the findings in this research focus primarily on the policy implementations. That is, firmwide, regulatory and accounting standards setting management processes would be affected by the day-to-day operational process definitions and implementations. These include the technology process definitions and implementations. The primary focus would be on the implementation and monitoring of the relative delta and the regulatory relative delta measurement framework. This study has proposed, in Section 4.11, a technology framework to monitor these measures. From a firmwide perspective, the implementation of this system encompasses, at the least, coordination between the functions: bank front office trading, bank supervision, risk management and control, accounting and technology. Such an implementation at minimum would have an effect on the management duties within these functions and include human resource planning, procurement and training.

This effect on management duties is considered to begin from the technology framework's planning and implementation phases to its day-to-day operations management. These same effects will be evident when new financial products are introduced. In addition, when new technologies and processes are introduced these enhancements will have a significant effect on the change management of the currently recognised and unrecognised functions.

5.7 Technology Implications

The technology implications from this research centres on platforms and interfaces to compute the *relative delta* and *accounting Value-at-Risk* measures, and the implementation of the *regulatory relative delta framework*. These platforms and interfaces are aligned to the proposed technology framework presented in Section 4.11 and specified in Figure 4.1. The technology implemented is to be a *real-time* data

system, with *real-time* updates of transacted financial instruments and items, and *real-time* updates of prices for these instruments and items. The technology framework implemented is to be highly scalable and secure, with the ability to handle large transaction volumes and be *fast* and precise in analysing and reporting information. The framework is to have the ability to adapt efficiently to changes in policy, financial product and technology lifecycles. In order to fulfill such requirements, resources would need to be allocated to financial product research and integration, technology platforms and interface research, training, and systems development, implementation, maintenance and technology lifecycles would include: budgeting, resource allocation, project management, business analysis, systems analysis, research and development, systems security and recovery, software engineering, hardware engineering as well as computer networks and communications engineering.

Monitoring system performance, change management to new technologies, and implementing new and changes to existing policies and financial products, would also have a significant effect on technology processes and their functions.

5.8 Statistical Implications

The measurement approach specified in Appendix M.5.2 quantifies the level of accuracy by which the coefficient of determination R^2 reports a regression model. This measurement approach would add value to both statistical research and commercial statistical product development. Such a measure can be efficiently implemented within a regression model to add an additional layer to determine either if the R^2 measure is further strengthened or if further investigation is necessary to determine a tested regressive relationship.

Statistical implications are also evident with the proposed algorithm and smoothing formula applied to calculate the standard normal inverse - cumulative distribution function (CDF). The proposed process specified in Appendix N.6.2 and N.6.3 would be useful for systems that primarily apply simulation techniques. In addition, this process would also add value to any statistical process that applies the standard normal inverse CDF.

5.9 Thesis Limitations

This thesis has been limited by the relative delta formula being implemented on a yearly basis. If the system of relative delta equations, specified in Section 3.2.5 were applied on a daily basis, then a similar number of observations to the market price variable (over 3,000 observations per firm for the 1992 to 2009 time range) would have been available for analysis. However, this research approached its analysis from the basis of audited accounting figures that are reported on a yearly basis. By applying the recommendations of this research to present accounting total information on a real-time or near real-time basis, then accounting information would be available for the daily relative delta analysis.

This study was also limited to the number of years observed. This limitation arose as a consequence of market and accounting data availability. Although, this research attaches the significance of its analysis from 1992 to 2009 to the Fama and French (2008) analysis from 1927 to 2006, it is recommended in Chapter 6 that increasing the time range would test the relative delta measure and also test this study's conclusions.

Regarding the number of firms surveyed, this research's population of interest consists of five banking firms. The Secondary sample that was tested alongside the banks consists of 11 banking related firms. The Control group that was also tested consists of 12 banking related firms. This study considers that increasing the number of firms tested in the Secondary sample and the Control group would test further the conclusions reported. The number of firms selected for the Secondary sample and the Control group would test further the control group was limited due to the stringent selection criteria detailed in Section 3.6. In addition, some firms were excluded during the quality assurance phase due to missing data.

This thesis has been challenged by the log change formula's (see Section 3.9.2, equation (3.33)) mathematical limitation when calculating changes that involve negative variable values. The total equity and net income variables both report negative levels. To compute the relative delta measure, the changes in these two variables have been computed by applying the relative change formula (see Section 3.9.2, equation (3.34)), and changes in the market price variable have been computed using the *conventional* log change formula. To provide a comparison, Table S.3 presents *log changes* (where negative values were set to zero) and *relative changes* for both the total equity and net

income variables. To test this study's conclusions, it is recommended in Chapter 6 that the relative delta applies variable changes computed using a consistent formula (or *process*) that is able to handle negative values. This would mean that the formula (or *process*) applied to calculate changes in the total equity and the net income variables, is also applied to calculate the changes in the market price variable.

This study considers that the results and analysis presented in this research effectively and conclusively examines the null hypothesis specified in Section 3.3.1. The limitations stated in this section, although important, are considered unlikely to falsify the findings reported. Further areas of analysis and work presented in Chapter 6 may be also considered as addressing this research's limitations. However, unless explicitly aligned to the above limitations this study considers them to be outside its direct research scope.

5.10 Summary

It is deduced, from examination of the test results, that this study's null hypothesis is falsified at approximately the 90% level. Analysis of the results show that instead of the expected improvement in accounting quality after the 2005 accounting change, the UK banks experienced a deterioration in this quality. This examination finds that the role accounting played in risk management after 2005 was influenced by its *implicit* role. In addition, this analysis finds that the UK banks became exposed to financial distress risk and an increase in the minimum regulatory capital requirement after 2005. The general conclusion from these findings identifies evidence that the UK banks became exposed to increased levels of accounting quality risk, and information bias between accounting and market data, after the 2005 accounting change.

It is also concluded that there are *far-reaching* regulatory policy, managerial and technological implications based on the implementation of the relative delta and the regulatory relative delta framework developed and applied in this study. In addition, statistical implications arising from analytical approaches applied in this research are also presented. Limitations to this thesis are also presented that centre on the time range analysed and the number of firms selected for the Secondary sample and the Control group. In addition, the limitation with the log change formula is recognised.

Based on the evidence from this research, it may be possible to summarise the theoretical findings in a proposition. Specifying the accounting totals: *total shareholders' equity, total assets, total liabilities* and *net income* collectively as the *set of accounting total information*, and individually as *accounting total information*, the axiom would state:

Prior to any risk mitigation action by a bank, for a given time period if there is a significant level of bias between the *change in a bank's <u>set</u> of accounting total information* and the *change in its market price*, then the bank will become exposed to market price volatility as a direct consequence of this bias.

Prior to any risk mitigation action by a bank, for a given time period if there is no bias between the *change in a bank's accounting total information* and the *change in its market price*, then the bank will not become exposed to market price volatility as a direct consequence of this non-bias.
Chapter 6 Future Work

6.1 Chapter Introduction

This chapter presents some future research directions aimed at extending this study. Some recommended work is aimed at addressing the thesis limitations reported in Chapter 5. In addition, some recommendations are suggested to validate the findings presented in this thesis. It is noted that this chapter makes extensive reference to the relative delta and the regulatory relative delta framework. However, in a general sense, it is not the intention of this chapter to follow any sequential pattern in its recommendations. The following paragraphs are organised for each to present one or more recommendations. These may be examined singularly or together with similar recommended work.

6.2 Recommendations for Future Work

This study would be furthered by researching areas that test the performance and accuracy of the relative delta measure. An important examination would be to determine how the measurement model performs with a longer time series range for firms within the UK banking sector. In addition, this measure could be tested to determine how well it performs with banking sectors outside the UK. Such results would enable the relative delta to be tested in differing environments to determine whether the results add strength to its relationship to financial distress risk, value relevance and accounting quality. This measure may be tested comparatively to the banking sector with firms in sectors and markets that are outside the banking sector. In general, such work would also provide a test of the concluding axiom presented in Section 5.10.

The other area that would require further research is the regulatory relative delta framework. This framework has only been tested within the LSE's UK banking sector. However, this framework's intention is to provide aggregated measures not only for domestic regulators and banking supervisors, but also for international and regional regulators and supervisors. Tests that determine how well this framework performs in other national sectors and international markets would be an important aspect of this measure. Such tests are important not only to determine how well the proposed

measurement framework operates but also to provide valuable information as to how best to calibrate its threshold levels.

This study applies consolidated accounting figures to represent its accounting totals. Testing the relative delta measure by applying also unconsolidated accounting totals would provide comparable results for this measure's performance on a consolidated and unconsolidated basis. Such results may provide an insight to the holding firm's levels of exposure to financial distress risk compared to its subsidiaries. On an operational level, monitoring such data would provide important information to help direct action to mitigate exposures to accounting quality risk and financial distress risk.

An important aspect of the relative delta measurement system is its *promised* ability to trace financial distress risk to the item level. This would mean that exposures to financial distress risk at the financial statement level can be traced to their origin at the financial item or financial instrument level. Further tests could be performed to determine this system's effectiveness to move from monitoring aggregated or summary information, such as industry concentrations at regional and economic area levels, to investigating this information at the detailed level. Such tests would provide important metrics to evaluate the systems forensic performance.

To further the strength of evidence from the general market indices, the number of market indices and foreign exchange currency pairs could be increased. The commodities markets that include oil and gold indices could also be tested. In addition, increasing the number of GDP sectors analysed and the inclusion of the Consumer Price Index (CPI) would add to the strength of evidence for market price reactions originating from economic indicators.

Investor and general bank sentiments for the 2005 accounting change is reported in this study by surveying secondary information from published sources. It may also be recommended to determine views to the 2005 change by directly surveying investors and key banking personnel. Such an approach would provide further insights that relate to the accounting change. In addition, this approach would provide a method to assess potential benefits and issues for the proposed relative delta measure and the regulatory relative delta framework. It is also recommended that face-to-face surveys may also be

302

augmented by investor and bank personnel responses to a paper or electronic questionnaire.

This study would also be furthered by examining evidence from detailed case studies. Information may be gathered from sources that include bank financial statements. By way of motivation, analysis of such information may uncover the precise reasons for the post 2005 increases in volatility and deterioration in accounting quality that is reported in this study. This examination would also include a detailed study of the accounting standard pronouncements that may be linked to these increases. This examination could focus on areas that include the fair value pricing approach, hedge accounting, netting and loan collateralisation policies. In addition, effects from the balance sheet assets and liabilities categorisations: fair value recognition, held-to-maturity, loans and receivables, and available-for-sale, may exhibit further evidence⁴⁷. Such examinations would assist in determining how those standards if applied unchanged may react in the future. Furthermore, from such an analysis, it would be possible to inform the accounting standards authorities as to possible actions that may be applied to reduce levels of observed differences between the accounting and market price change variables.

This study has found that the market price returns and the Value-at-Risk levels for banks in the UK banking sector observed significant changes after 2005. Highly significant changes were exhibited during 2008, the year that witnessed several large banking organisation and financial service firm failures. Future work could be considered that directs attention to the banking firms that failed with the view to determining additional factors that may provide evidence for the observed market price and Value-at-Risk levels. In addition, attention could also be directed to external *stressors* such as national and international political issues, and national and international security issues, as well as natural and unnatural disasters. Investigation with regard to *stressors* aligned to aspects such as reputational risk, moral hazard and tax law misinterpretations may also provide further evidence to the observed market price and Value-at-Risk levels. Such evidence would provide useful information in

⁴⁷ These four measurement categories were replaced in November 2009 by two measurement categories: recognition at fair value and recognition at amortised cost (IASB 2011, 2012). The two measurement categories are specified under the IFRS 9 accounting standard and are to be mandatorily applied from 1st January 2015 (IASB 2011, 2012).

identifying and therefore assisting in resolving such issues.

The Fama and French (2008) baseline regression results have been compared to the results in this study. This comparison implies that the UK banks' accounting to market price *difference* levels after 2005 are greater than those reported for NYSE firms from 1927 to 2006. This examination may be furthered by directly analysing historical data that is further than the 1992 time range investigated in this study. Such an examination would determine if these post 2005 *differences* were, as reported, significant. This study may also be furthered by taking the time range closer to the current date. Such an analysis would provide an examination of the *difference* effects beyond the 2008/2009 upper time range investigated.

For variables that have the potential to exhibit negative levels, such as the total equity and the net income variables, this study has computed variable changes using the relative change formula presented in Section 3.9.2, equation (3.34). Otherwise, changes are computed using the log change formula presented in Section 3.9.2, equation (3.33). This log formula is limited, in that, it is unable to compute changes involving negative numbers. This study has applied these change variables to compute the relative delta measure. However, it would be useful to measure the relative delta using change variables that are computed using a consistent formula that has the ability to handle negative values. It is considered that such a study would provide further analysis of the results reported using the total equity to market price and net income to market price relative deltas.

As its convention, this study examines the *second half-year* net income figures after 2002 (see Appendix R.1.2). Although, Appendix R.1.2 presents a comparison for the Primary and Secondary sample *full year*, *first half-year* and *second half-year* net income figures, it may be useful to provide a further comparison of the full year results for after 2002. This analysis may be performed by computing the net income to market price relative delta using the full year net income figures for before and after 2002. In addition, it may be useful to compare the relative delta levels when using individual yearly net income figures that are applied in this study, with net income levels calculated using running totals. Such analyses, although not expected to affect the findings in this research, would provide additional tests for the null hypothesis, and also further test the performance of the relative delta measure.

This study applied the Value-at-Risk models: variance-covariance, historical and Monte Carlo simulation (MCS) to analyse the change in Value-at-Risk levels before and after 2005. The Value-at-Risk testing approach consisted of backtesting market price returns calculated on the same day count basis as the Value-at-Risk time horizons. To extend this analysis it would be beneficial to determine how each model applied to different time horizons perform to the 1-day market price return profile. In addition, the Value-at-Risk measures have only been tested for the 95%, 99% and 99.9% confidence levels. It would also be beneficial to determine the *Value-at-Risk models* performance at the *extreme* 100% and 0% confidence levels. Specifically at the 0% confidence level, such an analysis is similar to the approach applied to evaluate *conditional VaR*.

The mathematical model in the MCS Value-at-Risk approach applied in this study utilises future market price movements based on historical market price returns. Further work is recommended in this area to test if future market price movements can be better modelled by applying other variables. These other variables include benchmark interest rate information such as interest swap rates, LIBOR rates, GILT bond yields, implied volatilities and historical implied volatilities and, stock market index averages. The aim of the proposed tests is to determine the variables that help enhance MCS Value-at-Risk measurements.

To better understand the relationship between the types of risk that this study references, it may be beneficial to research the relationship between financial distress risk, Value-at-Risk and the levels of *difference* quantified by the relative delta measure. In addition, prices applied to financial instruments that are based on pricing models could be investigated to determine how well the relative delta quantifies *pricing model risk*.

This study has introduced credit risk from the perspective of the Basel regulatory framework (Basel 2011). Although this framework advocates credit risk to be measured using the Basel framework or an audited internal based system, this framework understands some firms may apply credit risk levels assessed by external credit assessment institutions. Three such credit assessment institutions are Standard and Poor's, Moody's and Fitch Ratings. These institutions assign credit ratings to reflect credit risk for firms that are considered as being *investment grade companies* (Fabozzi 2010). It may be beneficial to investigate if a relationship exists between the relative

delta measure and the risk levels reported by these three institutions. Such an investigation would further the scope of tests performed using the relative delta measure. In addition, such an investigation would provide an assessment of the relative delta measure with respect to the approaches applied by the credit assessment institutions to quantify credit risk.

This study has reported estimates for the change in accounting total Value-at-Risk. These estimates have been made by applying the market price return Value-at-Risk and the relative delta measure. To further the study of accounting Value-at-Risk, it may be possible to determine, from the regressions for the market price Value-at-Risk and the relative delta measure, the true level of market price Value-at-Risk from the level reported by the regression's *intercept parameter*. The difference between the *intercept parameter* and the observed market price return Value-at-Risk, theoretically, should equate to the accounting Value-at-Risk. In addition to this regression process, accounting Value-at-Risk may also be determined from the process applied in this study. In addition, it may be determined by a direct evaluation from actual accounting data on a yearly and/or *daily basis*. Evaluation of accounting Value-at-Risk using these methods would allow the measure of its component that may be embedded in the market price return Value-at-Risk.

This study has introduced, in Appendix M.5.2, the R^2 assessment measure with the objective of testing how well the coefficient of determination R^2 quantifies a regression model. Although some tests for the R^2 assessment has been performed in this research, further study for this measure would provide evidence to how well it performs to its theoretical expectations.

A major aspect of investigation arising from this study is the implementation approaches for the proposed relative delta system and the regulatory relative delta framework. This implementation is proposed at the firmwide, the regional and the global regulatory and supervisory levels. Such a *framework* requires its implementation characteristics to be highly scalable and adaptable to react efficiently and effectively to regulatory policy changes. It must also react in the same way to operational process and financial product changes. The implementation approach that this study emphasises is to build a technology solution around the relative delta and regulatory relative delta framework. Future work regarding this approach that detail and test technology interfaces and platforms would provide important performance information. Building on this work, an implementation approach may be applied to develop a system that is highly scalable and secure with the ability to adapt efficiently to changes in policy, product and technology lifecycles. Such an implementation holds the promise to be extremely beneficial to the financial system.

6.3 Summary

This chapter has directed future work in several research areas. It has recommended that the relative delta system and the regulatory relative delta framework be further tested. These tests are aimed at evaluating the system's and framework's regulatory, risk management and implementation performance. In addition, these tests are aimed at examining the conclusions presented in this thesis. To examine further the findings presented in this thesis, it is recommended that additional variable and longitudinal data be gathered and tested. With this same objective, it is recommended that additional data gathering approaches and measurement techniques be applied. To support this research's findings, the use of case studies and investigation of external *stressors* are recommended. This chapter also recommends work based on the market price and accounting Value-at-Risk measures. In addition, future work that examines the *assessment measure* proposed to test the coefficient of determination, R^2 , is recommended.

Chapter 7 References

- Abbink, John B. (2011), 'Constructing stress tests', Journal of Risk Finance, Vol. 12, No. 5, pp. 421-434.
- Alfredson, Keith, Ken J. Leo, Ruth Picker, Paul Pacter, Jennie Radford, and Victoria Wise (2007), 'Applying International Financial Reporting Standards', Enhanced Edition, John Wiley and Sons.
- Amir, Eli, Trevor S. Harris, and Elizabeth K. Venuti (1993), 'A comparison of the value-relevance of U.S. versus non-U.S. GAAP accounting measures using Form 20-F reconciliations', Journal of Accounting Research, Vol. 31, Studies on International Accounting, pp. 230-264.
- Ang, Andrew, Monika Piazzesi, and Min Wei (2006), 'What does the yield curve tell us about GDP growth?', Journal of Econometrics, Vol. 131, No. 1-2, pp. 359-403.
- Armstrong, Christopher S., Mary E. Barth, Alan D. Jagolinzer, and Edward J. Riedl (2010), 'Market reaction to the adoption of IFRS in Europe', Accounting Review, Vol. 85, No. 1 (Jan., 2010), pp. 31-61.
- Banks, Eric (2002), 'The simple rules of risk: Revisiting the art of financial risk management', Wiley Finance.
- Banz, Rolf W. (1981), 'The relationship between return and market value of common stocks', Journal of Financial Economics, Vol. 9, No. 1 (Mar., 1981), pp. 3-18.

Barclays (2006), 'Barclays PLC Annual Report 2005', Barclays plc.

Barlas, Stephen, Roland Madison, Bob Randall, and Curt Verschoor (2003), 'SEC proposed new Regulation G on non-GAAP financials', Strategic Finance, Vol. 84, No. 7 (Jan., 2003), pp. 19-22.

- Barth, Mary E., Wayne R. Landsman, and Mark H. Lang (2008), 'International accounting standards and accounting quality', Journal of Accounting Research, Vol. 46, No. 3 (Jun., 2008), pp. 467-498.
- Barth, Mary E., William H. Beaver, and Wayne R. Landsman (2001), 'The relevance of the value-relevance literature for financial accounting standard setting: another view', Journal of Accounting & Economics, Vol. 31, No. 1-3 (Sep., 2001), pp. 77-104.
- Basel (1998), 'International convergence of capital measurement and capital standards (July 1988, Updated To April 1998)', Bank of International Settlements.
- Basel (2004), 'International convergence of capital measurement and capital standards: A revised framework', Basel II, Basel Committee on Banking Supervision (November 2004), Bank of International Settlements.
- Basel (2005), 'Amendment to the Capital Accord to incorporate market risks', Basel II, Basel Committee on Banking Supervision (November 2005 Revision), Bank of International Settlements.
- Basel (2006), 'International convergence of capital measurement and capital standards:A revised framework: Comprehensive version', Basel II, Basel Committee on Banking Supervision (June 2006), Bank of International Settlements.
- Basel (2009), 'Revisions to the Basel II market risk framework', Basel Committee on Banking Supervision, (July 2010), Bank of International Settlements.
- Basel (2010), 'Basel III: A global regulatory framework for more resilient banks and banking systems', Basel III, Basel Committee on Banking Supervision, (December 2010), Bank of International Settlements.
- Basel1 (2010), 'Basel III: International framework for liquidity risk measurement, standards and monitoring', Basel III, Basel Committee on Banking Supervision, (December 2010), Bank of International Settlements.

- Basel (2011), 'Basel III: A global regulatory framework for more resilient banks and banking systems', Basel III, Basel Committee on Banking Supervision, (June 2011 Revision), Bank of International Settlements.
- Beaver, William H., and George Parker (1995), 'Risk management: Problems and solutions', McGraw-Hill, New York.
- Beaver, William H., Paul Kettler, and Myron S. Scholes (1970), 'The association between market determined and accounting determined risk measures', Accounting Review, Vol. 45, No. 4 (Oct., 1970), pp. 654-682.
- Berkowitz, Jeremy, and James O'Brien (2002), 'How accurate are Value-at-Risk models at commercial banks?', The Journal of Finance, Vol. 57, No. 3 (Jun., 2002), pp. 1093-1111.
- Bernoulli, Daniel (1738), 'Exposition of a new theory on the measurement of risk', Econometrica: The Econometric Society, Vol. 22, No. 1 (Jan., 1994), pp. 23-36.
- Beuselinck, Christof, Philip Joos, Inder K. Khurana, and Sofie Van der Meulen (2010),
 'Mandatory adoption of IFRS and analysts' forecasts information properties',
 Working paper, Tilburg University, available:
 http://arno.uvt.nl/show.cgi?fid=112996 [accessed 14th July 2013].
- Bhandari, Laxmi Chand (1988), 'Debt/equity ratio and expected common stock returns: Empirical evidence', The Journal of Finance, Vol. 43, No. 2 (Jun., 1988), pp. 507-528.
- Bodie, Zvi, Alex Kane, and Alan J. Marcus (2002), 'Investments', Fifth Edition, McGraw-Hill Irwin.
- Brown, Robert (1828), 'A brief account of microscopical observations made in the months of June, July and August, 1827, on the particles contained in the pollen of plants; and on the general existence of active molecules in organic and inorganic bodies', The miscellaneous botanical works of Robert Brown, Vol. 1.,

available: http://sciweb.nybg.org/science2/pdfs/dws/Brownian.pdf [accessed 14th February 2012].

BvDEP (2011), 'Orbis', Bureau van Dijk Electronic Publishing, available: http://www.bvdinfo.com/About-BvD/Brochure-Library/Brochures/ORBISbrochure [accessed 11th February 2012].

Byrne, David (2002), 'Interpreting Quantitative Data', SAGE Publications.

- CFA (1997), 'New requirements on derivatives disclosure would help investors and improve efficient use of capital', Testimony to the U.S. Senate Banking Committee's Subcommittee on Securities, available: http://www.cfainstitute.org/about/press/release/Documents/1997_press_releases. pdf [accessed 30th December 2011].
- Chen, Yi-Wen (2006), 'An empirical study of using derivatives on multinational corporation strategies in Taiwan', Journal of American Academy of Business, Vol. 10, No. 1 (Sep., 2006), pp. 110-116.
- Cheney, Glenn (2008), 'FASB, IASB call for comments on financial instruments', Accounting Today, Vol. 22, No. 9 (May, 2008), pp. 14-17.
- Coffee, Jr, John C. (2005), 'Can lawyers wear blinders? Gatekeepers and third-party opinions', Texas Law Review, Vol. 84, No. 1 (Nov., 2005), pp. 59-74.
- Cohen, Ayala (1983) 'Comparing regression coefficients across subsamples: A study of the statistical test', Sociological Methods and Research, Vol. 12 No. 1 (Aug., 1983), pp.77-94.
- Corner, David (2006), 'The United Kingdom private finance initiative: The challenge of allocating risk', OECD Journal on Budgeting, Vol. 5, No. 3, pp. 37-55.
- Crane, Dwight B., Zvi Bodie, Kenneth A. Froot, Scott P. Mason, Robert C. Merton, Andre F. Perold, Erik Sirri, and Peter Tufano, (1995), 'The global financial system: A functional perspective', Boston, MA: Harvard Business School Press.

- Culp, Christopher L. (2002), 'The art of risk management: Alternative risk transfer, capital structure and convergence of insurance and capital markets', Wiley Finance.
- Cunningham, Coleen (2005), 'Effective simplification needed to overcome complexity', Financial Executive, Vol. 21, No. 9 (Nov., 2005), p. 6.
- Daniel, Kent, and Sheridan Titman (2006), 'Market reactions to tangible and intangible information', The Journal of Finance, Vol. 61, No. 4 (Aug., 2006), pp. 1605–1643.
- Dichev, Ilia D. (1998) 'Is the risk of bankruptcy a systematic risk?', The Journal of Finance, Vol. 53, No. 3 (Jun., 1998), pp. 1131-1147.
- Dosi, Cesare, and Michele Moretto (2003), 'Global warming and financial umbrellas', Journal of Risk Finance, Vol. 4, No. 4 (Summer 2003), pp. 18-25.
- Dowd, Kevin (2003), 'Beyond value at risk, the new science of risk management', John Wiley & Sons.
- EC (2002), 'Regulation (EC) No. 1606/2002 of The European Parliament and of The Council of 19 July 2002 on the application of international accounting standards', Official Journal of the European Communities, L 243/1, (11 Nov., 2002), English Version.
- Elliot, Barry, and Jamie Elliot (2008), 'Financial accounting and reporting', 12 Edition, Financial Times Prentice Hall.
- Fabozzi, Frank J. (2010), 'Bond markets analysis and strategies', Seventh Edition, Pearson Prentice Hall.
- Fama, Eugene F. (1965), 'The behavior of stock-market prices', The Journal of Business, Vol. 38, No. 1 (Jan., 1965), pp. 34-105.

- Fama, Eugene F. (1970), 'Efficient capital markets: A review of theory and empirical work', The Journal of Finance, Vol. 25, No. 2 (May, 1970), pp. 383-417.
- Fama, Eugene F., and James D. MacBeth (1973), 'Risk, return, and equilibrium: Empirical tests', The Journal of Political Economy, Vol. 81, No. 3 (May - Jun., 1973), pp. 607-636.
- Fama, Eugene F., and Kenneth R. French (1992), 'The cross-section of expected stock returns', The Journal of Finance, Vol. 47 (XLVII), No. 2 (Jun., 1992), pp. 427-465.
- Fama, Eugene F., and Kenneth R. French (1993), 'Common risk factors in the returns on stocks and bonds', The Journal of Financial Economics, Vol. 33, No. 1 (Feb., 1993), pp. 3-56.
- Fama, Eugene F., and Kenneth R. French (2008), 'Average returns, B/M, and share issues', The Journal of Finance, Vol. 63 (LXIII), No. 6 (Dec., 2008), pp. 2971-2995.
- Freire, Paula Gomes, André Figueiredo, and Pedro Cassiano Santos (2005), 'Portugal ready to embrace new asset classes', International Financial Law Review, Vol. 24, No. 9, pp. 83-86.
- French, Craig W. (2003), 'The Treynor capital asset pricing model', Journal of Investment Management, Vol. 1, No. 2. (2003), pp. 60-72.
- Gietzmann Miles, Ivana Raonic (2013), 'Thinly Traded Growth Stocks: A Joint Examination of Transparency in Communication and the Trading Platform', European Accounting Review (Published online: 20 Mar 2013).
- Guan, Lim Kian, and Wong Soo Chen (1998), 'Trends in financial markets: Possible Scenarios', Singapore Management Review, Vol. 20, No. 2 (Jul., 1998), pp. 11-24.

- Gup, Benton E. (2004), 'The new Basel Capital Accord', South-Western, Division of Thomson Learning.
- Horton, Joanne, George Serafeim, and Ioanna Serafiem (2013), 'Does mandatory IFRS adoption improve the information environment', Contemporary Accounting Research, Vol. 30, No. 1 (Spring 2013), pp.388-423.

HSBC (2006), 'Annual Report and Accounts 2005', HSBC Holdings plc.

- Hull, John C. (2009), 'Options, futures, and other derivatives', Seventh Edition, Pearson Prentice Hall.
- Humphreys, Peter (2004), 'Structured finance challenges for new issuers and new assets: An overview', Journal of Structured Finance, Vol. 10, No. 3 (Fall 2004), pp. 56-61.
- IASB (2008), 'International Financial Reporting Standards', International Accounting Standards Board and International Accounting Standards Committee Foundation.
- IASB (2009), 'International Financial Reporting Standards', International Accounting Standards Committee Foundation.
- IASB (2011), 'International Financial Reporting Standards; Part A: The conceptual framework and requirement; Part B: The accompanying documents', International Financial Reporting Standards Foundation.
- IASB (2012), 'International Financial Reporting Standards; Part A: The conceptual framework and requirement; Part B: The accompanying documents', International Financial Reporting Standards Foundation.
- IASB1 (2008), 'Reducing complexity in reporting financial instruments', Discussion Paper, International Accounting Standards Committee Foundation Publications, (March 2008).

- Ito, Kiyoshi (1951), 'On stochastic differential equations', Memoirs of The American Mathematical Society; No. 4, American Mathematical Society, pp. 1-51.
- Jensen, Michael C., Fischer S. Black, and Myron S. Scholes (1972), 'The capital asset pricing model: Some empirical tests', in Studies in the Theory of Capital Markets, Michael C. Jensen, ed., New York, Praeger Publishers, pp. 79-121.
- Jorion, Philippe (2009), 'Risk management lessons from the credit crisis', European Financial Management, Vol. 15, No. 5 (Nov., 2009), pp. 923-933.
- JPMorgan Chase and Reuters (1996), 'RiskMetrics Technical document', Fourth Edition, New York: Morgan Guaranty Trust Company, (Dec., 1996).
- Korajczyk, Robert A. (1999), 'Asset pricing and portfolio performance: Models, strategy and performance metrics', London, Risk Books.
- Lakonishok, Josef, Andrei Shleifer, and Robert W. Vishny (1994), 'Contrarian investment, extrapolation, and risk', The Journal of Finance, Vol. 49, No. 5 (Dec., 1994), pp. 1541–1578.
- Lander, Gerald, and Kathleen Auger (2008), 'The economic impact of the lack of transparency in financial reporting', Atlantic Economic Journal, Vol. 36, No. 1 (Mar., 2008), pp. 105-116.
- Langley, Paul (2006), 'Securitising suburbia: The transformation of Anglo-American mortgage finance', Competition & Change, Vol. 10, No. 3 (Sep., 2006), pp. 283-299.
- Leone, Marie (2008), 'Bankers: Fair value is like throwing gasoline on a fire', CFO.com, US, April 14, 2008, available: http://www.cfo.com/article.cfm/11039958?f=search [accessed 11th February 2012].

- Lintner, John V. (1965), 'The valuation of risk assets and the selection of risky investments in stock portfolios and capital budgets', The Review of Economics and Statistics, Vol. 47, No. 1 (Feb., 1965), pp. 13-37.
- Liu, Chunhui, Lee J. Yao, Nan Hu, and Ling Liu (2011), 'The impact of IFRS on accounting quality in a regulated market: An empirical study of China', Journal of Accounting, Auditing and Finance, Vol. 26, No. 4 (Sep., 2011), pp. 659-676.

Lloyds (2006), 'Annual Report and Accounts 2005', Lloyds TSB Group plc.

- Markowitz, Harry (1952), 'Portfolio selection', The Journal of Finance, Vol. 7, No. 1 (Mar., 1952), pp. 77-91.
- Mateus, Cesario, Jan Bartholdy (2011), 'Debt and taxes for Private Firms', International Review of Financial Analysis, Vol. 20, No. 3 (Jun., 2011), pp. 177–189.
- Merton, Robert C. (1995), 'A functional perspective of financial intermediation', Financial Management, Vol. 24, No. 2 (Summer 1995), pp. 23-41.
- Merton, Robert C., and Zvi Bodie (1996), 'A conceptual framework for analyzing the financial system', Paper based on chapters 1 and 8 of 'The Global Financial System: A Functional Perspective', by Crane et al., Harvard Business School Press, 1995. Harvard Business School Press, available: http://www.nek.lu.se/NEKeno/Finance B/A Framework for Analyzing the Financial System.pdf [accessed 17th March 2012].
- Metropolis, Nicholas, and Stanislaw Marcin Ulam (1949), 'The Monte Carlo method', Journal of American Statistics Association, Vol. 44, No. 247 (Sep., 1949), pp. 335-341.
- Monson, Dennis W. (2001), 'The conceptual framework and accounting for leases', Accounting Horizons, Vol. 15, No. 3 (Sep., 2001), pp. 275-287.
- Morais, Ana Isabel, and José Dias Curto (2008), 'Accounting quality and the adoption of IASB standards – Portuguese evidence', Revista Contabilidade & Finanças

(Accounting & Finance Magazine), University of São Paulo, São Paulo, Vol. 19, No. 48 (Sep. – Dec., 2008), pp. 103-111.

- Mossin, Jan (1966), 'Equilibrium in a capital asset market', Econometrica, Vol. 34, No. 4 (Oct., 1966), pp. 768-783.
- Mountain, James R. (2008), 'Securitization accounting: Are you ready for the credit crunch challenge?', Bank Accounting & Finance, Vol. 21, No. 3 (Apr., 2008), pp. 3-10.
- MSCI (1999), 'Risk management: A practical guide', First Edition, RiskMetrics Group.
- NACE (2008) 'NACE Rev. 2, Statistical classification of economic activities in the European Community', Eurostat, European Commission, available: http://epp.eurostat.ec.europa.eu/cache/ITY_OFFPUB/KS-RA-07-015/EN/KS-RA-07-015-EN.pdf [accessed 5th July 2012].
- Newbold, Paul (1995), 'Statistics for business and economics', Fourth Edition, Prentice-Hall, Inc.
- O'Haver, Russ R. (2003), 'Management intangibles: Capitalizing on your IP assets', Journal of Internet Law, Vol. 7, No. 6 (Dec., 2003), pp. 16-21.
- Ongkrutaraksa, Worapot (1999), 'Risk and fair value accounting: A critical analysis of FASB 119', University of Kent State, Working Paper, (Fall 1999), available: http://independent.academia.edu/DrWorapot/Papers/1153971/Risk_and_Fair_Va lue_Accounting_A_Critical_Analysis_of_FASB_No._119 [accessed 23rd April 2012].
- Paananen, Mari, and Henghsiu Lin (2009), 'The development of accounting quality of IAS and IFRS over time: The case of Germany', Journal of International Accounting Research, Vol. 8, No. 1, pp. 31–55.

- Paternoster, Raymond, Robert Brame, Paul Mazerolle, and Alex Piquero (1998), 'Using the correct statistical test for the equality of regression coefficients', Journal of Quantitative Criminology, Vol. 14, No. 3 (Sep., 1998), pp. 245-261
- Pearson, Karl, G. B. Jeffery, and Ethel M. Elderton (1929), 'On the distribution of the first product moment-coefficient, in samples drawn from an indefinitely large normal population', Biometrika, Vol. 21, No. 1-4 (Dec., 1929), pp. 164-201.
- Peterkort, Robert F., and James F. Nielsen (2005), 'Is the book-to-market ratio a measure of risk?', The Journal of Financial Research, Vol. 28, No. 4 (Winter 2005), pp. 487-502.
- Platikanova, Petya, and Christopher Nobes (2006), 'Was the introduction of IFRS in Europe value-relevant?', University of Pompeu Fabra and University of Reading; Working paper Series, Social Science Research Network, Social Science Electronic Publishing Inc.
- Popper, Karl R. (1992), 'The logic of scientific discovery', originally published in 1934 and the first English translation published in 1959 by Hutchinson Education, and reprinted and simultaneously published in USA and Canada by Routeledge, London and New York.
- Ramanna, Karthik, and Ewa Sletten (2009), 'Why do countries adopt International Financial Reporting Standards?', Harvard Business School Division of Research, Working Paper, No. 09-102, pp. 1-46.
- Raonic, Ivana, Stuart J. McLeay, Ioannis Asimakopolous (2004), 'The Timeliness of Income Recognition by European Companies: An Analysis of Institutional and Market Complexity', Journal of Business Finance and Accounting, Vol. 31, No, 1-2 (Jan. - Mar., 2004), pp.115-148.
- RBS (2006), 'Annual Report and Accounts 2005', The Royal Bank of Scotland Group plc.

- Šević, Željko, Chandrasekhar Krishnamurti, Aleksandar Šević (2005), 'Voluntary disclosure, transparency, and market quality: Evidence from emerging market ADRs', Journal of Multinational Financial Management, Vol. 15, No. 4-5 (Oct., 2005), pp. 435-454.
- Sharpe, William F. (1964), 'Capital Asset Prices: A theory of market equilibrium under conditions of risk', The Journal of Finance, Vol. 19 (XIX), No. 3 (Sep., 1964), pp. 425-442.
- SI (2008), 'Information and Incentives: The heart of risk management', Securities Industry News, Vol. 20, No. 11 (Mar., 2008), pp. 4-7.
- Standard Chartered (2006), 'Standard Chartered Annual Report and Accounts 2005', Standard Chartered plc.
- Stojanovic, Alex, Alec Chrystal, and Peter Crossan (2002), 'The impact on the City of UK Eurozone membership: The banking industry', Corporation of London.
- Treynor, Jack L. (1962), 'Toward a theory of market value of risky assets', Unpublished manuscript, referenced version from 'Asset Pricing and Portfolio Performance: Models, Strategy and Performance Metrics', by Robert A. Korajczyk (editor) London: Risk Books, (published in 1999), pp. 15–22.
- Vuolteenaho, Tuomo (2002), 'What drives firm-level stock returns?', The Journal of Finance, Vol. 57, No. 1 (Dec., 2002), pp. 233–264.
- Walton, Peter (2004), 'IAS 39: Where different accounting models collide', Accounting in Europe, Vol. 1, pp. 5-16.
- Walton, Peter, and Walter Aerts (2006), 'Global financial accounting and reporting, principles and analysis', Thomson Learning.
- Wang, Mulong, Min-Ming Wen, and Charles C. Yang (2010), 'Weather derivatives, price forwards, and corporate risk management', Journal of Risk Finance, Vol. 11, No. 4, pp. 358-376.

Wild, Ken, and Veronica Poole (2008), 'Financial reporting does the credit crisis herald the end of that period of stability for standard-setters', Accountancy Ireland, Vol. 40, No. 3 (Jun., 2008), pp. 10-12.

APPENDIX A THE FAMA AND FRENCH TIME DEPENDENT BOOK-TO-MARKET RATIO TREATMENT AND THE RELATIVE DELTA

This appendix introduces the extension to the Fama and French (2008) treatment of the book-to-market ratio that is applied in this study. This appendix also introduces the rationale in applying this extension to examine the null hypothesis specified in Section 3.3.1. For reasons of convenience, this null hypothesis is reproduced in this appendix.

This study extends the Fama and French (2008) treatment of the book-to-market ratio, from the following:

$$BM_t = BM_{t-1} + [dB_{t-1,t} - dM_{t-1,t}]$$
(A.1)

to an implementation of the following *delta* component from equation (A.1):

$$\delta_{t-1,t} = [dB_{t-1,t} - dM_{t-1,t}] \tag{A.2}$$

Where in equation (A.1), the BM_t is the natural log of the book-to-market ratio at time t, and is a measure of, BM_{t-1} , the book-to-market ratio measured at the preceding time period t - 1, plus the relative change or difference between, $dB_{t-1,t}$, the change in the natural log of book equity, and, $dM_{t-1,t}$, the change in the natural log of market price from time t - 1 to t.

Equation (A.2), and in its generalised form specified in equation (B.2), is the component that forms the basis of the extension to the treatment specified in equation (A.1). Where in equation (A.2), the delta component, $\delta_{t-1,t}$, in its extended form, measures at time t from t - 1, the difference between the natural log change in the accounting total, *total equity*, (and also the natural log change in accounting totals - *total assets, total liabilities* and *net income*) and the natural log change in the market price. For convenience, this study refers to the general approach of this delta measure, δ , with the generalised term - *relative delta*.

The term *relative delta* is applied to equation (A.2) because the change or the *delta* values dB and dM are first calculated from t - 1 to t, then the difference between two delta variables evaluated to determine their *vector* relationship. The relative delta measure could as well be named the *delta relative*, *delta difference* or *difference delta* as each conveys the same idea. The important mechanism that the name for this formula should translate is the process where, first, the delta for each variable is computed, and second, their relative difference is computed.

This relative delta measure is developed in this research to test the following null hypothesis:

Accounting quality in UK banks was not affected by the adoption of the IFRS accounting standards in 2005. For the UK banks that adopted the new accounting standards in 2005, there is no significant difference between the change in accounting totals and the change in market price after 2005 when compared to the level of difference before.

Fama and French (2008) and Daniel and Titman (2006) provide evidence to show that the book-to-market ratio's past changes contain predictive information. This predictive nature translates to these past changes having the ability to predict the level of a firm's future exposure to financial distress risk. Based on this evidence it could be contended that the motivation to verify the null hypothesis would be strong. In that, doing so would provide evidence for both the changes in accounting quality, and also the changes in the predictive levels of exposure to financial distress risk during the 2005 accounting change. A stronger motivation may become apparent when examining evidence from Fama and French (1992) and Peterkort and Nielsen (2005). This evidence shows that the measure of financial distress risk from the classical book-to-market approach relates to a measure of *stock market price risk*.

APPENDIX B BOOK DELTA AND ACCOUNTING DELTA TO MARKET DELTA

This appendix details the change in the accounting totals applied to extend the Fama and French (2008) treatment of the book-to-market ratio. This appendix also specifies the *relative delta measure* and its alignment to the concept of the *difference component*. In addition, the generalised optimal computational form of the *accounting to market price relative delta* measure is presented.

B.1 Book Delta to Market Delta

The Fama and French (2008) development of the book-to-market ratio (BM) and *book delta* to *market delta* measure takes the following decomposition:

$$BM_t = BM_{t-k} + [dB_{t-k,t} - dM_{t-k,t}]$$
(B.1)

Where BM_t - the \log^{48} of BM at time t is expressed as BM_{t-k} - the log of the BM at time t - k plus the difference between $dB_{t-k,t}$ - the change or *delta* in the log of book equity from time t - k to t, and $dM_{t-k,t}$ - the change or *delta* in the log market price for the same time period t - k to t.

B.2 Accounting Delta to Market Delta

The expression specified within the brackets in equation (B.1) is extended to the measure termed as the *relative delta*. Maintaining the subscript notation from Fama and French (2008) in equation (B.1) and replacing the measure of book equity B, with the more general form AC, to represent the financial statement accounting totals, the generic form of the *relative delta* measure is expressed by the following:

$$\delta_{t-k,t} = dAC_{t-k,t} - dM_{t-k,t} \tag{B.2}$$

⁴⁸ As stated in Appendix A, the term log is used to specify the natural logarithm to the base e and is also denoted as ln(X), where (X) is the termed being logged.

Where $\delta_{t-k,t}$ is specified as a firm's *relative delta*⁴⁹, and with reference to equation (B.1), equation (B.2) is expressed as the difference between $dAC_{t-k,t}$ - the log change in the financial statement accounting total from time t - k to t, and $dM_{t-k,t}$ - the log change in market price for the same time period t - k to t.

To define the relative delta measure used in this study more precisely as a single time unit measure, while maintaining the Fama and French (2008) notation, equation (B.2) takes the following form:

$$\delta_{t-1,t} = dAC_{t-1,t} - dM_{t-1,t}$$
(B.3)

Where $\delta_{t-1,t}$ is the measure of the relative delta for a firm from time period t-1 to a single incremental forward time period t, and is evaluated as the difference between $dAC_{t-1,t}$ - the log change in the financial statement accounting total for the unit time period t-1 to t, and $dM_{t-1,t}$ - the log change in the market price for the same unit time period t-1 to t.

Specifying equation (B.3) using a slightly different notation gives:

$$\delta_{t,t+1} = dAC_{t,t+1} - dM_{t,t+1}$$
(B.4)

Where $\delta_{t,t+1}$ is the measure of the relative delta for a firm from time period t to a single incremental forward time period t+1, and is evaluated as the difference between $dAC_{t,t+1}$ - the log change in the financial statement accounting total for the unit time period t to t+1, and $dM_{t,t+1}$ - the log change in market price for the same unit time period t to t+1.

This study applies the notation presented in equation (B.3), however, the same principles would apply if using the approach specified in equation (B.4).

⁴⁹ The Relative Delta measure $\delta_{t-k,t} = dAC_{t-k,t} - dM_{t-k,t}$ may also be represented by $\Delta_{t-k,t} = \Delta AC_{t-k,t} - \Delta M_{t-k,t}$.

Restating equation (B.3) and then expressing its components in the natural *log* form provides the following definitions:

$$\delta_{t-1,t} = dAC_{t-1,t} - dM_{t-1,t}$$

$$\delta_{t-1,t} = ln\left(\frac{AC_t}{AC_{t-1}}\right) - ln\left(\frac{P_t}{P_{t-1}}\right)$$
(B.5)

Applying the same definitions as in equation (B.3), the expression $ln(AC_t/AC_{t-1})$ is the natural log⁵⁰ change in the accounting total from time period t - 1 to t. This *change component* is termed in this study as the *accounting delta*. The expression $ln(P_t/P_{t-1})$ is the log change in market price for the same time period t - 1 to t. This *change component* is termed in this study as the *market price delta* or the *market price return*.

The *difference component* $\delta_{t-1,t}$ - the *relative delta* measure in equation (B.5), may be expressed in the more familiar *relative* form. This form, specified in equation (B.6), combines the *delta log* components to provide a more computationally optimal single log form. This *relative* log form is:

$$\delta_{t-1,t} = ln \left[\left(\frac{AC_t}{AC_{t-1}} \right) / \left(\frac{P_t}{P_{t-1}} \right) \right]$$
(B.6)

The definitions for the variables in equation (B.6) are the same as in equation (B.5).

⁵⁰ Unless otherwise stated, this study applies the term ln to specify the natural log, that is, the logarithm to the base e.

APPENDIX C THE RELATIVE DELTA MEASURE

To specify the characteristics of the *relative delta measure*, this appendix presents comparisons of its behaviour with the conventional book-to-market ratio. These comparisons are made with reference to the measure of *financial distress risk*, and the *difference component*. The latter measure applied in this study to examine accounting quality (see also Section 2.9.3).

C.1 Book Delta to Market Delta and The Relative Delta Measure

The *relative delta measure* is the primary measure built and applied to evaluate the levels of *difference* between the change in accounting totals and the change in market price.⁵¹ The *relative delta* forms the basis of the measurement system applied to test this study's null hypothesis. These tests effectively measure, for UK banks, the level of change in financial statement accounting totals to the level of change in market price before and after the 2005 accounting change. In effect, this study proposes this relative delta to quantify changes in accounting quality by measuring the changes in the deltas for the accounting variables and the market price variable.

The development of the relative delta measure specified in equation (B.3) as part of its extension to the Fama and French (2008) book-to-market ratio, applies a different measurement of the book equity component to that specified in equation (A.2). In equation (A.2), Fama and French (2008) specify the book-to-market ratio as the ratio between the variables *book equity per share* and *market price*, or as the ratio between *book equity* and *market value*. Where, *market value* is specified as the number of a firm's outstanding shares multiplied by its market price. Section 3.2.3 provides a detailed analysis of the way each of these *explicitly share adjusted variables* relate to the *relative delta variable*. Examining the change component specified in equation (A.2) to the *relative delta* measure specified in its generic form in equation (B.3), the former applies the *book equity per share* variable while the latter, in comparison,

⁵¹ The component *change in market price* specified in the *relative delta measure* is in effect a measure of the market price return, and may be referred to as either *change in market price* or *market price return*. In addition, this *change in market price* may also be termed the *market price delta*.

applies the *total equity* variable. As stated earlier in this section, the former equation (A.2) adjusts its equity or market price component using the number of shares outstanding. The *relative delta measure* in equation (B.3) does not make this adjustment. This study specifies the *total equity to market price relative delta* to be the measure of the difference between the change in *total shareholders' equity* and the change in *market price*. It is possible to use the *relative delta* measure without the adjustment for outstanding shares due to its specification. Since the *relative delta* measure is formulated based on the change in the numerator variable to the change in the denominator variable (see equation (B.6)), in effect, this measure scales itself. This specification of the relative delta measure is used in this study to determine the level of difference between changes in one variable compared to changes in another variable. This specification is also applied to determine the relative delta's behavior without the explicit adjustment for outstanding shares.

C.2 Relative Delta Measure Characteristics

To maintain the identity of the variables used in the Fama and French (2008) work, the relative delta characteristics will be detailed by referencing the *book equity to market price relative delta* (BM Relative Delta measure). This is equivalent to the earlier stated *total equity to market price relative delta* that is developed later in its general form as the *accounting to market price relative delta*.

The difference between the *change* in *book equity* and *market price* variables signify the *change* for a firm's value from the *book equity* perspective to its value from the *market price* perspective. This *book equity valuation* measures a firm's end of year value from the perspective of prepared financial statements, and this *market price valuation* measures a firm's almost instantaneous value from the perspective of its market investors.

To determine the characteristics of the Fama and French (2008) motivated *BM relative delta measure*, presented in equation (A.2), in relation to the conventional book-tomarket ratio (BM or BM Measure), the primary characteristics of this conventional measure are now considered. Fama and French (1992, 1993) found evidence that the *conventional* book-to-market ratio has the ability to measure both financial distress risk and market price return for a firm's stock. Based on a firm's BM level its stock is classified as either a value stock or a growth stock. For a firm with a high BM measure, its shares issued in a stock market are termed value stocks, and for a firm with a low BM its shares are termed growth stocks. Fama and French (1992, 1993) evaluated that a value stock with a high BM exhibited higher returns to compensate for higher levels of *financial distress risk* exposure to the stock's *underlying firm*. In comparison, a growth stock with a low BM, exhibited lower returns and signified a lower level of financial distress to the stock's *underlying firm*.

These book-to-market ratio characteristics may be summarised by: value stocks with high BM exhibit higher market price returns but also have a higher level of financial distress risk; growth stocks with low BM exhibit lower market price returns and have a lower level of financial distress risk. The Fama and French (2008) BM delta characteristics may be summarised by the finding that the *BM delta measure*, in general, enhance estimates of expected returns. From this and the Fama and French (1992) and Peterkort and Nielsen (2005) findings, it may be contended that this *BM delta measure* enhances its characteristic to predict a firm's level of exposure to financial distress risk.

Table C.1 presents the behavior of the BM measure and the BM delta for increasing returns and risk and conversely decreasing returns and risk.

Table C.1Book-to-Market Ratio and Book Equity to Market Price Relative
Delta

The table shows the relationship between the book-to-market ratio and the book equity to market price relative delta measure. The Return column presents the increase or decrease of market price return for the corresponding system of inequalities presented in the BM Measure column, and the BM Relative Delta Measure column. The BM Measure column presents the book-to-market ratio, and the BM Relative Delta Measure column presents the book equity to market price relative delta measure. The Financial Distress Risk column shows the increase or decrease of financial distress risk for the corresponding system of inequality equations in the BM Measure column, and the BM Relative Delta Measure column. The BM Measure column presents the system of inequalities where for a firm: B is the book equity measure; S is the number of shares outstanding; and, M is the market price. The BM Relative Delta Measure column presents the system of inequality equations where for a firm: dB is the natural log of the change in book equity for a unit time period; and, dM is the natural log of the change in market price for the same unit time period.

Table C.1 (Continued)				
Return	Financial Distress Risk	BM Measure	BM Relative Delta Measure	
increase	increase	$\frac{B}{S \times M} > 1$	dB - dM > 0	
decrease	decrease	$\frac{B}{S \times M} < 1$	dB - dM < 0	

From Table C.1 it is observed that the *increase* and *decrease* relationship specified for return and financial distress risk, represented by the Return and Financial Distress Risk columns respectively, are reflected by the inequality levels for the book-to-market ratio, represented by the BM Measure column, and the relative delta measure, represented by the BM Relative Delta Measure⁵² column. Examining Table C.1 shows that a firm's stock return and its financial distress risk both increase if the BM measure is greater than 1, (that is, if the ratio of book equity to number of shares outstanding multiplied by the BM Relative Delta Measure when it is greater than 0, (that is, if the difference in the change in book equity to market price return is greater than 0). Table C.1 also shows that a firm's stock return and its exposure to financial distress risk both decrease if the BM measure is less than 1. The same decrease in return and *risk* is specified for the BM measure when it is less than 0.

This is to say that in the normal course of events, from Table C.1 the expressions for the BM measure and the BM delta measure both signify for a stock, its gradually increasing and decreasing expected return and risk measurement. From the system of inequalities presented in Table C.1, if a stock's BM measure is increasing above 1, this would translate to a corresponding increase in expected return and also the related increase in financial distress risk. If however, the BM measure reverses and is decreasing below 1, this would signify the corresponding decrease in the stock's expected return and also a decrease in its related financial distress risk. Following the same rationale, if the BM delta measure is increasing above 0, this would translate to an increase in expected return and also the increase in related financial distress risk. If however the BM delta measure reverses and is decreasing above 0, this would translate to an increase in expected return and also the increase in related financial distress risk. If however the BM delta measure reverses and is decreasing below 0, i.e. negatively increasing, this would

⁵² This study refers to the BM Relative Delta Measure also as the BM delta or BM delta measure.

signify a decrease in expected return and a decrease in financial distress risk. In addition, Fama and French (2008) find evidence that shows the *BM delta* reversal of returns is predictive of the *BM* reversal of returns. They find evidence that both the change in market price, dM, and the change in book equity, dB, are equally related to the BM delta's predictive power.

From Section 2.6, in its general form the BM delta measure is specified as the accounting to market price relative delta measure or referred to simply as the *relative delta*. Table C.2 presents this measure to reflect the return and risk relationships presented in Table C.1.

Table C.2Market Price Return, Financial Distress Risk and Accounting Totals
to Market Price Relative Delta

The table shows the relationship between market price return, financial distress risk and accounting totals to the market price relative delta measure. The Return column presents the increase, unchanged or decrease of the market price return for the corresponding system of inequality and equality equations in the Relative Delta Measure column. The Financial Distress Risk column shows the increase or decrease of financial distress risk for the corresponding system of equations in the Relative Delta Measure column. The Relative Delta Measure column presents the system of equations where for a firm: dAC is the natural log of the change in accounting totals for a unit time period; and dM is the natural log of the change in market price for the same unit time period.

Return	Financial Distress Risk	Relative Delta Measure
increase	increase	dAC - dM > 0
unchanged	unchanged	dAC - dM = 0
decrease	decrease	dAC - dM < 0

Table C.2 shows that a firm's stock return and its financial distress risk both increase if the Relative Delta measure is greater than 0, that is, if the difference in the change in accounting totals to market price is greater than 0. Table C.2 also shows that a firm's stock return and its financial distress risk both decrease if the Relative Delta measure is less than 0, that is, if the difference in the change in accounting totals to market price is less than 0. When compared to the system of equations presented in Table C.1, Table C.2 presents in the Relative Delta Measure column an additional equality. This equality, represented by dAC - dM = 0, specifies that a firm's stock return and its financial distress risk both remain unchanged if the Relative Delta measure equates to 0, that is,

the change in accounting totals are the same, in both magnitude and direction, as the change in market price. If this equality were introduced in Table C.1, within the BM Measure context, this equality would be: $B/(S \times M) = 1$.

However, this equality is introduced in Table C.2 because the accounting total to market price relative delta measure is applied to test this research's null hypothesis by providing a measure of *difference* between a firm's change in key accounting totals and its change in market price. In order to provide a coherent measure of *difference*, also termed the *difference component*, from the system of equations presented in Table C.2, the accounting total to market price relative delta measure takes the definitions specified in Table C.3.

Table C.3Measure of the Difference Component using the Accounting Totals
to Market Price Relative Delta

The table shows the relationship between levels of the difference component and the accounting totals to market price relative delta measure. The Difference Component column presents the increase or unchanged level of difference for the corresponding system of inequality and equality equations presented in the Relative Delta Measure column. In this system of inequality and equality equations, dAC is the natural log of the change in accounting totals in a unit time period, and dM is the natural log of the change in market price for the same unit time period.

Difference Component	Relative Delta Measure
increase	dAC - dM > 0
unchanged	dAC - dM = 0
increase	dAC - dM < 0

Aligned with the system of equations presented in Table C.2, Table C.3 shows that a firm's *difference component* increases if the Relative Delta measure is greater than 0. That is, if the difference in the change in accounting totals to market price is greater than 0. The equality represented by dAC - dM = 0, specifies that there is no difference between a firm's change in accounting totals and the change in its market price. Unlike Table C.2, Table C.3 shows that for a firm, its *difference component* level increases if the Relative Delta measure is less than 0. That is, if the difference in the change in accounting totals to market price.

The expressions specified in Table C.3 are central to testing this study's null hypothesis. In that they quantify the level of material difference between the firm's change in key financial statement accounting totals and its change in market price. This study approaches the measure of the *difference component* from the perspective that the accounting standard applied would maintain a level of *difference* year-on-year between its change in key financial statement totals and its change in market price. This level of difference expected to be the same or less than that measured for a previous accounting standard. That is to say, the accounting standards change in 2005 would be expected to maintain or ideally, exhibit a decrease in the difference between the measures of a firm's value based on key accounting totals from its balance sheet and income statement transactions to that of the firm's value based on its market price transactions. This ideal decrease in difference would translate to an increase in the level of accounting quality, (see Section 2.9.3), after the 2005 adoption. In Table C.3, the key accounting total components used for the relative delta in this study are the following balance sheet amounts: total shareholders' equity, total assets and total liabilities, and the income statement amount net income.

APPENDIX D RELATIONSHIP BETWEEN FINANCIAL DISTRESS RISK, DIFFERENCE COMPONENT AND RELATIVE DELTA

This appendix focuses on the relative delta measure's behaviour with respect to financial distress risk and the measure of the *difference component*.

D.1 Financial Distress Risk and the Relative Delta Measure

From the book-to-market ratio measure in Fama and French (1992) and Peterkort and Nielsen (2005), it is considered that the level of book equity per share compared to the level of market price is a measure of a firm's expected returns. This level is also considered in these studies as a measure of a firm's exposure to financial distress risk. This relationship is developed in Appendix C and presented in Table C.1 in the BM Measure column.

This classical treatment of the BM measure is extended by Fama and French (2008). They interpret that the level of a firm's change in the book equity per share to the level of change to market price enhances the measure of its expected returns. This study develops this measure in Appendix C using the term the *relative delta* measure. The *relative delta* measure provides for a firm a measure of the difference between its change in the key financial statement accounting totals to the change in its market price. Additionally, Appendix C and Table C.2, develops for a firm the relationship between its *relative delta* and its expected return and its financial distress risk. Appendix C Table C.3 presents the relational characteristics of the *relative delta* measure applied in this study. If Table C.3 is adapted to present the relationship between the *difference component* and financial distress risk presented in Table C.2, their resulting relationship to the *relative delta* would be that presented in Table D.1.

Table D.1The Difference Component, Financial Distress Risk and the
Accounting Totals to Market Price Relative Delta

The table shows the relationship between levels of the difference component to financial distress risk and the accounting totals to market price relative delta measure. The Difference Component column presents the increase or unchanged level of difference for the corresponding system of inequality and equality equations that are presented in the Relative Delta Measure column. The Financial Distress Risk column shows the increase of financial distress risk for the corresponding system of inequality and equality and equality equations in the Relative Delta Measure column. The Relative Delta Measure column presents the system of inequality and equality equations. Where, for a firm, dAC is the natural log of the change in accounting totals and dM is the natural log of the change in market price.

Difference Component	Financial Distress Risk	Relative Delta Measure
increase	increase	dAC - dM > 0
unchanged	unchanged	dAC - dM = 0
increase	increase	dAC - dM < 0

The obvious difference in the interpretation of financial distress risk between Table C.2 and Table D.1 is that in the latter it increases for the relative delta inequality, dAC - dM < 0. This Table D.1 specification interprets an increase to a firm's level of financial distress risk, irrespective of whether an increase in return is evident or not. This compares with Table C.2 where the same inequality represents a firm's decrease in financial distress risk and a resulting decrease for the firm's expected return.

The relationships specified in Table D.1 between the *difference component*, financial distress risk, and the *relative delta*, are applied in this study to test its null hypothesis specified in Section 3.3.1. With, principally the *difference component*, applied to test for accounting quality (see Section 2.9).

APPENDIX E ACCOUNTING TO MARKET PRICE RELATIVE DELTA DEFINITIONS

This appendix details the *accounting to market price relative delta* measures applied in this study. In addition, the specific implementations of these measures are specified.

E.1 Accounting Totals to Market Price Relative Deltas

This study, as specified in Appendix B, terms the generic form of the *relative delta measure* as the *accounting to market price relative delta*. From this general specification, this section details the four relative delta measure applied in this study.

E.1.1 Total Equity to Market Price Relative Delta

From the basis of the Fama and French (2008) measure for the change in book equity to the change in market price, this study specifies the total equity to market price relative delta. As discussed in Appendix C, there is a subtle difference from the way that Fama and French (2008) specifies the change in book equity to the change in market price to that specified in this study. In a strict sense, the Fama and French (2008) change in book equity to the change in market price is in fact the change in book equity per share, to the change in market price. This study evaluates the formula - change in book equity per share) to the change in market price; and also the formula - change in *total equity* (by applying the change in Total Shareholders' Equity) to the change in market price. The latter formulation, in effect, does not explicitly specify effects that relate to the number of shares outstanding.

This study specifies the formula to calculate the change in total shareholders' equity to the market price return relative delta by the following equation:

$$\delta E M_{t-1,t} = d E_{t-1,t} - d M_{t-1,t} \tag{E.1}$$

Where $\delta EM_{t-1,t}$ is a firm's change in total shareholders' equity to its change in market price from time t - 1 to a single incremental forward time period, t. This study terms

the $\delta EM_{t-1,t}$ as the total shareholders' equity to market price relative delta. The $\delta EM_{t-1,t}$ is evaluated as the difference between $dE_{t-1,t}$: the change in the log financial statement total shareholders' equity accounting total for the unit time period t - 1 to t, termed in this study as the total shareholders' equity delta or simply as the equity delta; and, $dM_{t-1,t}$: the change in market price for the same unit time period t - 1 to t, termed in this study as the market price return, market delta or price delta.

Presenting equation (E.1) in log form gives:

$$\delta EM_{t-1} = ln\left(\frac{E_t}{E_{t-1}}\right) - ln\left(\frac{P_t}{P_{t-1}}\right)$$
(E.2)

The expression $ln(E_t/E_{t-1})$ is the log change in the *total shareholders' equity* accounting total from time t - 1 to t. The variable E_t is the *total shareholders' equity* accounting total at time t, and E_{t-1} is the *total shareholders' equity* accounting total at time t - 1. The expression $ln(P_t/P_{t-1})$ is the log change in *market price* for the same time period t - 1 to t. The variable P_t is the market price at time t - 1.

E.1.2 Total Assets to Market Price Relative Delta

From the basis of the accounting equation presented in Section 2.8.1, equation (2.3), a measure may be specified to determine the difference between the change in total assets and the change in market price. This measure, would in principle, allow this study to determine the level of the change in balance sheet asset totals to the change in market price before and after the 2005 accounting change.

Such a measure would allow examination of: 1) the *standalone change* in assets; and, 2) the *relative change* in assets on an explanatory basis from equation (2.3). The latter examination would provide evidence to explain changes in equity. That is, the ability to explain if observed changes in equity are primarily based on changes in assets or changes in liabilities, or both.

To provide this measure of the difference between the change in total assets and the change in market price, the *total assets to market price relative delta* measure is
developed for the total assets variable. This study specifies the formula for this measure using the following equation:

$$\delta AM_{t-1,t} = dA_{t-1,t} - dM_{t-1,t}$$
(E.3)

Where $\delta AM_{t-1,t}$ is a firm's change in total assets to its change in market price from time t - 1 to a single incremental forward time period, t. This study terms the $\delta AM_{t-1,t}$ as the *total assets to market price relative delta*. The $\delta AM_{t-1,t}$ is evaluated as the difference between $dA_{t-1,t}$: the change in the log financial statement *total assets* accounting total for the unit time period t - 1 to t, termed in this study as the *total assets delta* or simply as the *assets delta*; and, $dM_{t-1,t}$: the change in *market price* for the same unit time period t - 1 to t.

Presenting equation (E.3) in log form gives:

$$\delta AM_{t-1,t} = \ln\left(\frac{A_t}{A_{t-1}}\right) - \ln\left(\frac{P_t}{P_{t-1}}\right) \tag{E.4}$$

Where the expression $ln(A_t/A_{t-1})$ is the log change in the *total assets* accounting total from time t - 1 to t. The variable A_t is the *total assets* accounting total at time t, and A_{t-1} is the *total assets* accounting total at time t - 1. The expression $ln(P_t/P_{t-1})$ is the log change in *market price* for the same time period t - 1 to t. The variable P_t is the market price at time period t, and P_{t-1} is the *market price* at time t - 1.

E.1.3 Total Liabilities to Market Price Relative Delta

From the accounting equation presented in equation (2.3), a measure may be specified to determine the difference between the change in total liabilities and the change in market price. This measure, would in principle, allow this study to determine the level of the change in balance sheet assets and liabilities totals to the change in market price before and after the 2005 accounting change.

Same as with the total assets variable (see Appendix E.1.2), such a measure would allow the examination of: 1) the *standalone change* in liabilities; and, 2) the *relative change* in liabilities on an explanatory basis from equation (2.3). The latter examination

would provide evidence to explain changes in equity and also assets. That is, the ability to explain if observed changes in equity are primarily based on changes in liabilities or changes in assets, or both.

To provide this measure of difference between the change in total liabilities and the change in market price, the *total liabilities to market price relative delta* measure is developed for the total liabilities variable. This study specifies the formula for this measure using the following equation:

$$\delta LM_{t-1,t} = dL_{t-1,t} - dM_{t-1,t}$$
(E.5)

Where $\delta LM_{t-1,t}$ is a firm's change in total liabilities to its change in market price from time t - 1 to a single incremental forward time period, t. This study terms the $\delta LM_{t-1,t}$ as the *total liabilities to market price relative delta*. The $\delta LM_{t-1,t}$ is evaluated as the difference between $dL_{t-1,t}$: the change in the log financial statement *total liabilities* accounting total for the unit time period t - 1 to t, termed in this study as the *total liabilities delta*, or simply as the *liabilities delta*; and, $dM_{t-1,t}$: the change in *market price* for the same unit time period t - 1 to t.

Presenting equation (E.5) in log form gives:

$$\delta LM_{t-1,t} = ln\left(\frac{L_t}{L_{t-1}}\right) - ln\left(\frac{P_t}{P_{t-1}}\right)$$
(E.6)

Where the expression $ln(L_t/L_{t-1})$ is the log change in the *total liabilities* accounting total at time t - 1 to t, the variable L_t is the *total liabilities* accounting total at time t, and L_{t-1} is the *total liabilities* accounting total at time t - 1. The expression $ln(P_t/P_{t-1})$ is the log change in *market price* for the same time period t - 1 to t. The variable P_t is the market price at time period t, and P_{t-1} is the *market price* at time t - 1.

E.1.4 Net Income to Market Price Relative Delta

The development of a measure to determine the difference between the change in net income and the change in market price, in principle, would allow this study to determine the level of the change in net income to the change in market price before and after the 2005 accounting change.

Such a measure would allow the examination of: 1) the *standalone change* in net income; and, 2) the *relative change* in net income as a comparison and on an explanatory basis to changes in equity, assets and liabilities. The latter examination would provide evidence to determine if changes in equity, assets and liabilities were reflected by changes in net income. This study specifies the formula to calculate the change in net income to the market price relative delta measure by the following equation:

$$\delta IM_{t-1,t} = dI_{t-1,t} - dM_{t-1,t} \tag{E.7}$$

Where $\delta IM_{t-1,t}$ is a firm's change in net income to its change in market price from time t - 1 to a single incremental forward time period, t, this study terms the $\delta IM_{t-1,t}$ as the *net income to market price relative delta*. The $\delta IM_{t-1,t}$ is evaluated as the difference between $dI_{t-1,t}$: the change in the log financial statement *net income* accounting total for the unit time period t - 1 to t, termed in this study as the *net income delta*; and, $dM_{t-1,t}$: the change in *market price* for the same unit time period t - 1 to t.

Presenting equation (E.7) in log form gives:

$$\delta IM_{t-1,t} = ln\left(\frac{I_t}{I_{t-1}}\right) - ln\left(\frac{P_t}{P_{t-1}}\right)$$
(E.8)

Where the expression $ln(I_t/I_{t-1})$ is the log change in the *net income* accounting total at time t - 1 to t, the variable I_t is the *net income* accounting total at time t, and I_{t-1} is the *net income* accounting total at time t - 1. The expression $ln(P_t/P_{t-1})$ is the log change in *market price* for the same time period t - 1 to t. The variable P_t is the market price at time period t, and P_{t-1} is the *market price* at time t - 1.

APPENDIX F APPROACH FOR CHANGE IN ACCOUNTING TOTALS AND MARKET PRICE RETURN REGRESSIONS

From the Fama and French (2008) specification for the *book delta* to *market delta* presented in Appendix B and the relative delta specifications presented in Appendix E, it is possible to test the regressive relationship between the change in key financial statement accounting totals (specified in Section 2.8) and the change in market price⁵³. This appendix details the regressions applied to examine this relationship.

The results from these regressions are expected to provide a guide to the statistical relationship between the accounting and market price change variables that make up the relative delta. It is also expected that these results may provide evidence to evaluate this study's null hypothesis. This study uses the Fama and French (2008) cross-sectional regression approach and the Jensen, Black and Scholes (1972) time series regression approach.

F.1 Cross-Sectional Regression

This section presents the cross-sectional regressions applied to test the relationships between the accounting and market price change variables. These regressions are guided by the approach specified by Fama and French (2008).

F.1.1 Change in Accounting Totals and Market Price Return

The cross-sectional regression approach is applied to evaluate the relationship between a chosen independent variable and a dependent variable. The cross-sectional regression approach applied in this study takes the following general form:

$$R_{d:t} = b_t dR_{i:t} + a_{d:t} + e_t$$
 (F.1)

⁵³ From Fama and French (2008) it would be acceptable to refer to the change in market price as the market price return or simply as the return.

Where for a time period t, generally one year, R_d represents the dependent or explained variable; R_j represents the independent or explanatory variable; b is the slope representing the relationship between the independent variable to the dependent variable; a_d is the intercept representing the level of the dependent variable if the independent variable were zero; and, e the error term representing the difference between the observed measure for the dependent variable, R_d , and its predicted measure using the regression equation (F.1).

By applying the cross-sectional regression approach to test this research's null hypothesis, it would be expected that to strengthen the null hypothesis the relationships observed would be closer to unity after the 2005 accounting change when compared to before. That is, the slope parameter, b_t in equation (F.1), that measures the relationship between the accounting and the market price change variables, is expected to be closer to 1 after the 2005 accounting change when compared to before.

F.1.2 Change in Total Equity and Market Price Return

The cross-sectional regression applied to evaluate the relationship between the change in total shareholders' equity and the change in market price takes the following general form:

$$dM_t = b_t dE_t + a_t + e_t \tag{F.2}$$

In this regression, for the banks surveyed: dM_t is the sample firms' average change in market price for the year t; dE_t is the sample firms' average change in total shareholders' equity for the year t; b_t is the slope representing the relationship between the average change in total shareholders' equity and the average market price return; a_t is the intercept and represents the level of market price return if the average change in total shareholders' equity were zero; and, e_t the error representing the difference between the observed average return and the regression model return.

F.1.3 Change in Total Assets and Market Price Return

The cross-sectional regression applied to evaluate the relationship between the change in total assets and the change in market price takes the following general form:

$$dM_t = b_t dA_t + a_t + e_t \tag{F.3}$$

In this regression, for the banks surveyed: dM_t is the sample firms' average change in market price for the year t; dA_t is the sample firms average change in total assets for year t; b_t is the slope representing the relationship between the average change in total assets and the average market price return; a_t is the intercept, and represents the level of market price return if the average change in total assets were zero; and, e_t the error representing the difference between the observed average return and the regression model return.

F.1.4 Change in Total Liabilities and Market Price Return

The cross-sectional regression applied to evaluate the relationship between the change in total liabilities and the change in market price takes the following general form:

$$dM_t = b_t dL_t + a_t + e_t \tag{F.4}$$

In this regression, for the banks surveyed: dM_t is the sample firms average change in market price for the year t; dL_t is the sample firms average change in total liabilities for year t; b_t is the slope representing the relationship between the average change in total liabilities and the average market price return; a_t is the intercept and represents the level of market price return if the average change in total liabilities were zero; and, e_t the error representing the difference between the observed average return and the regression model return.

F.1.5 Change in Net Income and Market Price Return

The cross-sectional regression applied to evaluate the relationship between the change in net income and the change in market price takes the following general form:

$$dM_t = b_t dI_t + a_t + e_t \tag{F.5}$$

In this regression, for the banks surveyed: dM_t is the sample firms' average change in market price for the year t; dI_t is the sample firms average change in net income for year ; b_t is the slope representing the relationship between the average change in net income and the average market price return; a_t is the intercept and represents the level of market price return if the average change in net income were zero; and, e_t is the error representing the difference between the observed average return and the regression model return.

F.2 Time Series Regression

This study also tests its null hypothesis by applying the time series regression approach specified by Jensen, Black and Scholes (1972).

F.2.1 Change in Accounting Totals and Market Price Return

This study applies the time series regression approach to evaluate the relationship between a chosen independent variable and a dependent variable. The time series regression approach applied in this study takes the following general form:

$$R_{d:t,T} = b_{t,T} dR_{i:t,T} + a_{d:t,T} + e_{t,T}$$
(F.6)

Where for a time period t = 1, 2, ..., T: R_d represents the dependent or explained variable; R_i represents the independent or explanatory variable; b is the slope representing the relationship between the independent variable to the dependent variable; a_d is the intercept representing the level of the dependent variable if the independent variable were zero; and, e the error term representing the difference between the observed measure for the dependent variable, R_d , and its predicted measure using regression (F.6).

This research applies the same rationale to test its null hypothesis using the time series regression approach as it does with the cross-sectional regression approach. By applying the time series regression approach, it would be expected that, to strengthen the null hypothesis, any relationship observed would be closer to unity after the 2005 accounting change when compared to before. That is, the slope parameter, $b_{t,T}$ in equation (F.6), that measures the relationship between the accounting and the market price change

variables, is expected to be closer to 1 after the 2005 accounting change when compared to before.

The time series regressions are applied over three time periods specified by a total time period from 1994 to 2008, and sub-time periods 1994 to 2004 and 1994 to 2007.

F.2.2 Change in Total Equity and Market Price Return

The time series regression applied to evaluate the relationship between the change in total shareholders' equity and the change in market price takes the following general form:

$$dM_{t,T} = b_{t,T} dE_{t,T} + a_{t,T} + e_{t,T}$$
(F.7)

In this regression, for the firms surveyed: t represents the year count from the beginning year 1994 to the end year T; T represents the final year for the time series regressions, that is 2004, 2007 and 2008; $dM_{t,T}$ is the average change in market price for the sample firms at year t, with averages calculated on a yearly basis for the time period t to T; $dE_{t,T}$ is the average change in total shareholders' equity for the sample firms at year t, with averages calculated on a yearly basis for the time period t to T; $b_{t,T}$ is the slope representing the relationship between the yearly average change in total shareholders' equity and the average change in market price for the time period t to T; $a_{t,T}$ is the intercept and represents the level of average market price change if the yearly average change in total shareholders' equity were zero; and, $e_{t,T}$ the error representing the difference between the actual yearly average change in market price to the regression model's yearly average change in market price.

F.2.3 Change in Total Assets and Market Price Return

The time series regression applied to evaluate the relationship between the change in total assets and the change in market price takes the following general form:

$$dM_{t,T} = b_{t,T} dA_{t,T} + a_{t,T} + e_{t,T}$$
(F.8)

In this regression, for the firms surveyed: t represents the year count from the beginning year 1994 to the end year T; T represents the final year for the time series regressions, that is 2004, 2007 and 2008; $dM_{t,T}$ is the average change in market price for the sample firms at year t, with averages calculated on a yearly basis for the time period t to T; $dA_{t,T}$ is the average change in total assets for the sample firms at year t, with averages calculated on a yearly basis for the time period t to T; $b_{t,T}$ is the slope representing the relationship between the yearly average change in total assets and the average change in market price for the time period t to T; $a_{t,T}$ is the intercept and represents the level of average market price change if the yearly average change in total assets were zero; and, $e_{t,T}$ the error representing the difference between the actual yearly average change in market price to the regression model's yearly average change in market price.

F.2.4 Change in Total Liabilities and Market Price Return

The time series regression applied to evaluate the relationship between the change in total liabilities and the change in market price takes the following general form:

$$dM_{t,T} = b_{t,T} dL_{t,T} + a_{t,T} + e_{t,T}$$
(F.9)

In this regression, for the firms surveyed: t represents the year count from the beginning year 1994 to the end year T; T represents the final year for the time series regressions, that is 2004, 2007 and 2008; $dM_{t,T}$ is the average change in market price for the sample firms at year t, with averages calculated on a yearly basis for the time period t to T; $dL_{t,T}$ is the average change in total liabilities for the sample firms at year t, with averages calculated on a yearly basis for the time period t to T; $b_{t,T}$ is the slope representing the relationship between the yearly average change in total liabilities and the average change in market price for the time period t to T; $a_{t,T}$ is the intercept and represents the level of average market price change if the yearly average change in total liabilities were zero; and, $e_{t,T}$ the error representing the difference between the actual yearly average change in market price to the regression model's yearly average change in market price.

F.2.5 Change in Net Income and Market Price Return

The time series regression applied to evaluate the relationship between the change in net income and the change in market price takes the following general form:

$$dM_{t,T} = b_{t,T}dI_{t,T} + a_{t,T} + e_{t,T}$$
(F.10)

In this regression, for the firms surveyed: t represents the year count from the beginning year 1994 to the end year T; T represents the final year for the time series regressions, that is 2004, 2007 and 2008; $dM_{t,T}$ is the average change in market price for the sample firms at year t, with averages calculated on a yearly basis for the time period t to T; $dI_{t,T}$ is the average change in net income for the sample firms at year t, with averages calculated on a yearly basis for the time period t to T; $b_{t,T}$ is the slope representing the relationship between the yearly average change in net income and the average change in market price for the time period t to T; $a_{t,T}$ is the intercept and represents the level of average market price change if the yearly average change in net income were zero; and, $e_{t,T}$ the error representing the difference between the actual yearly average change in market price to the regression model's yearly average change in market price.

APPENDIX G APPROACH FOR ACCOUNTING TO MARKET PRICE RELATIVE DELTA AND VALUE-AT-RISK REGRESSIONS

The regressions specified in this appendix are applied to test the statistical significance of this study's null hypothesis. These regressions specify the relationship between the accounting to market price relative delta and the market price return Value-at-Risk.

G.1 Total Equity Regression

For a time period t the regression to test the relationship between the total equity to market price relative delta and the market price Value-at-Risk is given by:

$$V_{dM_t} = b_t (dE_t - dM_t) + a_t + e_t$$
 (G.1)

Where, for a time period t, $dE_t - dM_t$ represents the total equity to market price relative delta measure, V_{dM_t} represents the Value-at-Risk for the market price variable, b_t represents the slope of the regression equation, a_t represents the intercept for the regression, and e_t is the error term that represents the difference between the actual value for V_{dM_t} and its modelled value using the regression equation (G.1).

G.2 Total Assets Regression

For a time period t the regression to test the relationship between the total assets to market price relative delta and the market price return Value-at-Risk is given by:

$$V_{dM_t} = b_t (dA_t - dM_t) + a_t + e_t$$
 (G.2)

Where, for a time period t, $dA_t - dM_t$ represents the total assets to market price relative delta measure. The definitions for V_{dM_t} and a_t are the same as in equation (G.1). The parameter e_t is the error term that represents the difference between the actual value for V_{dM_t} and its modelled value using the regression equation (G.2).

G.3 Total Liabilities Regression

For a time period t the regression to test the relationship between the total liabilities to market price relative delta and the market price return Value-at-Risk is given by:

$$V_{dM_t} = b_t (dL_t - dM_t) + a_t + e_t$$
(G.3)

Where, for a time period t, $dL_t - dM_t$ represents the total liabilities to market price relative delta measure. The definitions for V_{dM_t} and a_t are the same as in equation (G.1). The parameter e_t is the error term that represents the difference between the actual value for V_{dM_t} and its modelled value using the regression equation (G.3).

G.4 Net Income Regression

For a time period t the regression to test the relationship between the net income to market price relative delta and the market price Value-at-Risk is given by:

$$V_{dM_t} = b_t (dI_t - dM_t) + a_t + e_t$$
 (G.4)

Where, for a time period t, $dI_t - dM_t$ represents the net Income to market Price relative delta measure. The definitions for V_{dM_t} and a_t are the same as in equation (G.1). The parameter e_t is the error term that represents the difference between the actual value for V_{dM_t} and its modelled value using the regression equation (G.4).

APPENDIX H DERIVATION AND PROOF OF ACCOUNTING VALUE-AT-RISK

This appendix details the derivation of the accounting Value-at-Risk measurement approach applied in this study.

In Table D.1 the *relative delta* measure equality dAC - dM = 0 represents the effect exhibited when changes in the accounting variable during a specified time interval relate directly to changes in the market price variable for the same time interval. This observation is simply how the book-to-market measure is interpreted by Fama and French (1992, 2008) during the shift between growth stocks and value stocks.

It would then be reasonable to state that the change in the accounting measure is simply reporting the change in the market price. In interpreting this effect further, it would be reasonable to state that the equality dAC - dM = 0 represents a state of equilibrium where the level of risk exhibited by the market price variable would be the same as, or matched by, the accounting variable.

In such a state of equilibrium, it would be reasonable to declare a firm's market price return Value-at-Risk would be matched by its accounting Value-at-Risk. Where the accounting Value-at-Risk is measured using the accounting totals. This latter measure attributed to the accounting variable has been reported by HSBC (2005), Barclays (2005) and Dowd (2003). This study develops the *relative delta* equilibrium relationship dAC - dM = 0 and market price return Value-at-Risk to derive a measure of accounting Value-at-Risk.

H.1 Derivation and Proof of Accounting Value-at-Risk from Market Price Value-at-Risk and the Relative Delta

For a time period t consider the relative delta measure where the change in the market price variable is reported by the change in the accounting variable, thus:

$$dAC_t - dM_t = 0 \tag{H.1}$$

Where dAC_t is the change in the accounting variable during the time interval specified by t and dM_t is the change in the market price variable during the same time interval.

It follows that the Value-at-Risk measured for the change in the accounting variable, represented by dAC_t , and the Value-at-Risk measured for the change in the market price variable, represented by dM_t , can be specified by:

$$V_{dAC_t} - V_{dM_t} = 0 \tag{H.2}$$

Where V_{dAC_t} represents the Value-at-Risk measure for the accounting variable during the time period *t*, and V_{dM_t} represents the Value-at-Risk measure for the market price variable for the same time period.

Rearranging equation (H.2) gives:

$$V_{dM_t} = V_{dAC_t} \tag{H.3}$$

Introducing to RHS of equation (H.3) the relative delta component from equation (H.1) gives:

$$V_{dM_t} = (dAC_t - dM_t) + V_{dAC_t}$$
(H.4)

Where V_{dAC_t} represents the measure of accounting Value-at-Risk.

It is possible to test the relationship specified in equation (H.4) using the following regression:

$$V_{dM_t} = b_t (dAC_t - dM_t) + a_t + e_t \tag{H.5}$$

Where V_{dM_t} represents the Value-at-Risk for the market price variable for time interval t (in this regression, this Value-at-Risk variable contains the accounting Value-at-Risk component), b_t represents the slope of the regression equation for time interval t, $dAC_t - dM_t$ represents the relative delta measure for time interval t, a_t represents the intercept for the regression at time interval t (in this regression, this variable represents the market price return Value-at-Risk that does not contain the accounting Value-at-Risk component), and e_t is the error term that represents the difference between the actual value for V_{dM_t} and its modelled value using the regression equation (H.5) for the time interval t.

APPENDIX I ACCOUNTING VALUE-AT-RISK

A measure of accounting Value-at-Risk is presented in HSBC (2005), Barclays (2005) and Dowd (2003). To measure *accounting Value-at-Risk* by directly applying the models specified in Section 2.12 involves several historical time periods. JPMorgan Chase and Reuters (1996) specifies two years or 500 days of historical data points, and the Basel (2011) framework specifies a minimum of three years or 800 days. This study computes the market price return Value-at-Risk based on both the 500 and 800 days of historical price information. Applying the same approach to measure Value-at-Risk for the accounting total variables would mean having at most historical accounting total data for the past 800 days.

For banking firms, this study primarily tests audited accounting totals reported in annual financial statements. The total time period principally examined is from 1992 to 2008. This time range provides 17 data points. It is mathematically possible to apply the Value-at-Risk models specified in Section 2.12 to evaluate this *accounting Value-at-Risk* for an accounting total using the 17 historical data points.

The historical simulation Value-at-Risk model specified in Section 2.12.2 could be applied to evaluate and backtest a one-year time horizon accounting Value-at-Risk. Using this approach, to compute the 2008 year accounting Value-at-Risk, 16 historical data points (from 1992 to 2007) may be used. This measure will provide a confidence level (that is less than 100%) at approximately the 93.75% (computed using $[15/16] \times 100\%$). To compute the one year time horizon accounting Value-at-Risk for the remaining years (2007 year to 1992), there would be a proportional decrease in the number of data points available for the Value-at-Risk calculation, with a resulting decrease in confidence level. That is, a maximum of 15 data points would be available to evaluate Value-at-Risk for the year 2007 and 2008, at a confidence level of 93.33%; 14 data points for the year 2006, 2007 and 2008, at a confidence level of 92.86%; continuing to 2 data points required for the year 1994 to 2008, at a confidence level of 50%. To illustrate, 1 data point would be required for the year 1993 to 2008, technically at an uncertain confidence level of 0%.

I.1 Generalised Accounting Value-at-Risk

Although it is mathematically possible to compute accounting Value-at-Risk by directly applying the historical simulation Value-at-Risk model using only 17 data points, this research develops an alternative approach. This study estimates accounting Value-at-Risk by utilising the relationship between the market price Value-at-Risk and the relative delta measure. This approach is developed by applying the following relationship.

At a state of *equilibrium*, equating equation (H.1): $dAC_t - dM_t = 0$, and equation (H.2): $V_{dAC_t} - V_{dM_t} = 0$ gives:

$$V_{dAC_t} - V_{dM_t} = dAC_t - dM_t \tag{I.1}$$

Where V_{dAC_t} for time period *t* represents the Value-at-Risk measure for the accounting variable (termed the accounting Value-at-Risk), and V_{dM_t} represents the Value-at-Risk measure for the market price return variable (termed the market price return Value-at-Risk) for the same time period. dAC_t is the change in the accounting variable during the time interval specified by *t*, and dM_t is the change in the market price variable during the same time interval.

Rearranging equation (I.1) to make accounting Value-at-Risk the subject provides the following general form:

$$V_{dAC_t} = (dAC_t - dM_t) + V_{dM_t}$$
(I.2)

Where the definitions for the variables in equation (I.2) are the same as in equation (I.1).

I.2 Total Equity Value-at-Risk

For a time period *t* the formula for the *change in total equity* Value-at-Risk estimated by applying the total equity to market price relative delta and the market price Value-at-Risk is given by:

$$V_{dE_t} = (dE_t - dM_t) + V_{dM_t}$$
(I.3)

Where for the time period t, V_{dE_t} represents the Value-at-Risk measure for the change in the total equity variable, and V_{dM_t} represents the Value-at-Risk measure for the market price return variable for the same time period. dE_t is the change in the total equity variable during the time interval specified by t, and dM_t is the change in the market price variable during the same time interval.

I.3 Total Assets Value-at-Risk

For a time period *t* the formula for the *change in total assets* Value-at-Risk estimated by applying the total assets to market price relative delta and the market price Value-at-Risk is given by:

$$V_{dA_t} = (dA_t - dM_t) + V_{dM_t}$$
(I.4)

Where for the time period t, V_{dA_t} represents the Value-at-Risk measure for the change in the total assets variable, and V_{dM_t} represents the Value-at-Risk measure for the market price return variable for the same time period. dA_t is the change in the total assets variable during the time interval specified by t, and dM_t is the change in the market price variable during the same time interval.

I.4 Total Liabilities Value-at-Risk

For a time period t the formula for the *change in total liabilities* Value-at-Risk estimated by applying the total liabilities to market price relative delta and the market price Value-at-Risk is given by:

$$V_{dL_t} = (dL_t - dM_t) + V_{dM_t}$$
(I.5)

Where for the time period t, V_{dL_t} represents the Value-at-Risk measure for the change in the total liabilities variable, and V_{dM_t} represents the Value-at-Risk measure for the market price return variable for the same time period. dL_t is the change in the total liabilities variable during the time interval specified by t, and dM_t is the change in the market price variable during the same time interval.

I.5 Net Income Value-at-Risk

For a time period *t* the formula for the *change in net income* Value-at-Risk estimated by applying the net income to market price relative delta and the market price Value-at-Risk is given by:

$$V_{dl_t} = (dI_t - dM_t) + V_{dM_t}$$
(I.6)

Where for the time period t, V_{dI_t} represents the Value-at-Risk measure for the change in the net income variable, and V_{dM_t} represents the Value-at-Risk measure for the market price return variable for the same time period. dI_t is the change in the net income variable during the time interval specified by t, and dM_t is the change in the market price variable during the same time interval.

APPENDIX J ACCOUNTING VALUE-AT-RISK, MARKET PRICE VALUE-AT-RISK AND RELATIVE DELTA

This appendix presents the implied relationship between the relative delta measure and the level of difference between the accounting Value-at-Risk and the market price Value-at-Risk. This relationship's implication to this study is also highlighted.

The general accounting Value-at-Risk formula developed in this research is specified in equation (J.1):

$$V_{dAC_t} = (dAC_t - dM_t) + V_{dM_t}$$
(J.1)

Where, for time period t, V_{dAC_t} represents the Value-at-Risk measure for the change in the accounting variable, and V_{dM_t} represents the Value-at-Risk measure for the market price return variable for the same time period. dAC_t is the change in the accounting variable during the time interval specified by t, and dM_t is the change in the market price return variable during the same time interval.

Equation (J.1) proposes a measure to assign a level of Value-at-Risk to the accounting variable. In this study this measure is termed the *accounting Value-at-Risk*. Although the accounting Value-at-Risk alone may hold significance as a measure for risk attached to accounting totals, this study evaluates its significance comparatively to the market price Value-at-Risk. The formula applied to compare the accounting and market price Value-at-Risk measures is simply the difference between these two Value-at-Risk variables. This relationship may be symbolised by:

$$V_{dAC_t} - V_{dM_t} \tag{J.2}$$

Where the definitions for the variables in equation (J.2) are the same as in equation (J.1).

Examining the expression presented in (J.2), it becomes apparent that it is simply the right-hand side of the Value-at-Risk and relative delta relationship presented in equation (I.1). For convenience, reproducing equation (I.1) presents, $V_{dAC_t} - V_{dM_t} = dAC_t - dM_t$.

The left-hand side (LHS) of equation (I.1) specifies the expression:

$$dAC_t - dM_t \tag{J.3}$$

The expression presented in (J.3) is the accounting totals to market price relative delta. Thus this relative delta may be applied to estimate the difference between the accounting Value-at-Risk and the market price return Value-at-Risk.

In addition to the market price Value-at-Risk, it would be ideal to compute the accounting Value-at-Risk by applying the sophisticated models presented in Section 2.12 and referenced in Appendix I. However, the proposition that the relative delta has the ability to measure the difference in accounting and market price return Value-at-Risks is the approach applied in this study. The implication that such an approach has is that it avoids the implementation of complex models, such as those presented in Section 2.12.

APPENDIX K REGULATORY CAPITAL AND REGULATORY VALUE-AT-RISK

This appendix details the adjustment proposed to the Basel regulatory capital formula. The approach applied to specify this adjustment is by building on the measure of difference between the accounting Value-at-Risk and the market price Value-at-Risk. Effectively the proposition recommends that the adjustment applied to the minimum capital requirement be calculated using the *regulatory relative delta framework*. This appendix also details how this *regulatory relative delta framework* is developed from the system of relative delta measures.

The Basel regulatory capital requirement formula presented in Section 2.10, and reproduced here specifies the following:

$$Capital Requirement = \frac{Total Regulatory Capital}{RWA}$$
(K.1)

Where the capital requirement formula in equation (K.1) represents the capital adequacy for a firm. This capital adequacy calculated using the ratio of the firm's total regulatory capital to its risk weighted assets (RWA).

It was suggested in Section 2.10 that the Basel Regulatory Framework (Basel 2011) does not specify an adjustment to take account of differences between accounting risk and the market price risk. This study proposes such an adjustment to be applied to the capital requirement *risk weighted assets (RWA)* component in formula (K.1).

The proposed adjustment to the risk weighted assets component is based on the difference between the accounting Value-at-Risk and the market price Value-at-Risk. This difference, in effect, is quantified by the relative delta. This associative relationship between the relative delta measure and the difference between the accounting Value-at-Risk and the market price Value-at-Risk is detailed in Appendix J. The relative delta measure has been developed in this study from the framework of the conventional book-

to-market measure prescribed by Fama and French (1992) and Peterkort and Nielsen (2005), and the delta measure specified by Fama and French (2008).

Developing a system of relative delta equations based on the change in accounting totals, and conforming to the Value-at-Risk to relative delta relationship represented in equations (J.2): $V_{dAC_t} - V_{dM_t}$; and (J.3): $dAC_t - dM_t$, would give the following series of relationships:

$$\delta EM_t = dE_t - dM_t \tag{K.2}$$

$$\delta AM_t = dA_t - dM_t \tag{K.3}$$

$$\delta LM_t = dL_t - dM_t \tag{K.4}$$

$$\delta IM_t = dI_t - dM_t \tag{K.5}$$

Where for the series of equations (K.2), (K.3), (K.4) and (K.5) for time period t, δEM_t is a firm's total equity to market price relative delta, $dE_t - dM_t$ is the explicit difference between the change in total shareholders' equity and the change in market price; δAM_t is a firm's total assets to market price relative delta, $dA_t - dM_t$ is the explicit difference between the change in total assets and the change in market price; δLM_t is a firm's total liabilities to market price relative delta, $dL_t - dM_t$ is the explicit difference between the change in total liabilities and the change in market price; and, δIM_t is a firm's net income to market price relative delta, $dI_t - dM_t$ is the explicit difference between the change in net income and the change in market price.

The conventional accounting equation that measures the key total accounting variables used in this study is specified by: Equity = Assets - Liabilities. The other accounting total variable, *Net Income* is specified as the measures for the change in *Equity*, that is, the change in the level of *Assets* and *Liabilities* (IASB 2011).

By monitoring the relative delta equation for Equity, (K.2), for amounts measured on the balance sheet or income statement, any change in the level of the total equity to market price relative delta can be traced to changes in the levels of the total assets and total liabilities to market price relative deltas represented by equations (K.3) and (K.4) respectively. In addition, the net income relative delta given the net income variable's relationship to the equity, assets and liabilities variables should also reflect the change in the total equity to market price relative delta levels. In effect, the monitoring of the total equity to market price relative delta specified in equation (K.2) would have a direct relation to a firm's level of exposure to financial distress risk and to the other accounting totals, and also provide a measure for the difference between a firm's total equity Value-at-Risk and market price Value-at-Risk.

This study proposes that the risk weighted assets component (RWA) in the Basel capital requirement formula, specified in formula (K.1), is adjusted to reflect the absolute measures of the total equity to market price relative delta. The modified Basel capital requirement formula incorporating the proposed adjustment is given by:

$$Capital Requirement = \frac{Total Regulatory Capital}{RWA \times Abs(\delta EM_t)}$$
(K.6)

Where $Abs(\delta EM_t)$ represents the positive or absolute measure of the total equity to market price relative delta for a given time period t, and is guided by the relative delta measure specification presented in Table D.1. The definitions for the other variables in equation (K.6) are the same as in equation (K.1).

It may be contended that by monitoring the series of four relative delta equations (K.2), (K.3), (K.4) and (K.5), any change in one of the relative deltas, including the total equity to market price relative delta, would reflect the level of a firm's exposure to financial distress risk.

This study also proposes that the risk weighted assets component (RWA) in the Basel capital requirement formula is adjusted to reflect the highest absolute measure of the difference component from the relative delta equations. That is, if the highest level recorded for a given time period is from the total equity to market price relative delta, then this relative delta is used to adjust the capital requirement formula. If, however, the total assets to market price relative delta registers a higher level, then this level would be used to adjust the capital requirement formula. The same criterion is applied to the total liabilities and net income relative delta levels. Such an adjustment to the Basel regulatory capital formula is presented by the following formula:

$$Capital Requirement = \frac{Total Regulatory Capital}{RWA \times Abs(\delta_t)}$$
(K.7)

Where $Abs(\delta_t)$ represents the positive or absolute measure of the *highest accounting* total to market price return relative delta for a given time period t. The same as with equation (K.6), equation (K.7) is guided by the relative delta measure specification presented in Table D.1. The definitions for the other variables in equation (K.7) are the same as in equation (K.1).

This study applies the approach presented in equation (K.7) to the banks in the UK banking sector on a yearly basis from 1994 to 2008. Specifically this study applies the adjustment to the minimum 8% level of total regulatory capital specified in the Basel III framework (Basel 2011).

This study proposes that the implementation of this measure uses a threshold based application of the adjustment applied to the risk weighted assets (*RWA*). Such that, a 0% threshold would apply the maximum level of the relative delta adjustment to the RWA, and for example, a 50% threshold applies the relative delta adjustment if the relative delta level is greater than 50%. That is, the maximum change for an accounting total has seen an increase or decrease of more than 50% compared to the change in the market price variable. A 100% threshold would apply the relative delta adjustment to the RWA if the relative delta measure were greater than a 100%.

The threshold levels presented in this study may be monitored and set internally by a banking firm as part of its risk management process. Alternatively, the threshold levels may be applied by regulatory authorities that have the ability to monitor banking firms at national and international levels. In accordance with the relative delta specifications presented in Table D.1, this study proposes that high thresholds could be set for banks that continually exhibit low financial distress risk. That is, banks that exhibit low relative delta levels at about the 40% level. With medium to low thresholds applied to banks that continually exhibit medium to high levels of exposure to financial distress risk. That is, medium to high relative delta levels at about the 100% and 200% levels respectively.

Applying this framework would allow the Basel regulatory equation (see equation (K.1)) to factor in any persistent and significant accounting to market price differences. It is envisaged that such an application would encourage banks at the firmwide level to monitor this effect closer. In addition, the regulatory framework, the accounting framework, and the risk management framework now have a measure that monitors the absolute level of a firm's financial distress risk based on the effect between accounting totals and the market price, an effect that is measured by the relative delta measure.

It could be contended that firms may smooth levels of accounting totals or take other actions, such as issue shares, to adjust levels of relative delta, with the aim to lessen the level of regulatory capital requirement. However, the proposed relative delta measure can be applied primarily as a risk management tool at the firm level to determine the accounting total variables that may be contributing to large relative delta levels when compared to the market price variables. Theoretically, it is then possible to investigate the accounting totals at an item or instrument level to determine the cause of significant increases to relative delta levels. Such an approach would provide the opportunity for a firm to take appropriate steps to mitigate any recognised financial distress risk exposures at the item or instrument level. This action carried out prior to any of these exposures recorded as *persistent observations* at the regulatory level.

The approach detailed in this appendix, that applies a system of relative delta equations to provide an adjustment for the Basel minimum regulatory capital, is termed in this study as the *regulatory relative delta framework*.

APPENDIX L ADJUSTMENT TO THE MINIMUM BASEL REGULATORY CAPITAL FOR UK BANKS USING THE REGULATORY RELATIVE DELTA

The Basel regulatory framework (Basel 2011) specifies that a banking firm at minimum should provision an 8% capital reserve for 100% of assets that are considered to be at risk. For the banks registered with LSE's UK banking sector an adjustment applied to the 8% capital reserve, effectively alters the 8% capital reserve for the 2004 year by 0.2%. This adjustment is computed using the *regulatory relative delta framework* developed by extending the Fama and French (2008) book-to-market ratio treatment. This adjustment is discussed in this appendix. For years 2005, 2007 and 2008 the adjustment effectively increases the minimum capital reserve requirement by 9.5%, 36.3% and 122.2% respectively. Table L.1 presents these yearly-adjusted changes to the 8% Basel minimum capital requirement from 2004 to 2008.

Table L.1 Adjusted Minimum Basel Regulatory Capital for UK Banks

The table shows the estimated average change to the minimum Basel regulatory capital requirement, adjusted by the level of difference between the change in total equity and change in market price for banks registered with the LSE's UK banking sector. The Time column presents the Year and Period columns. The Year and Period columns present the corresponding year and time period, respectively, for the regulatory relative delta measure presented in the ΔR column. The *Change in 8% Capital Requirement (%)* column presents the adjustment applied to the Basel minimum regulatory capital, at the 8% level, using the formula: 8% × $(1 + \Delta R)$ formula. The Increase (%) column presents the percentage increase from the 8% minimum capital requirement level to the adjusted level presented in the *Change in 8% Capital Requirement (%)* column.

Time		ΔR^{a}	Change in 8% Capital Requirement (%)	Increase (%)
Year	Period		$8\% \times (1 + \Delta R)$	
2004	01-Jan-04 to 31 Dec-04	0.002	8.02	0.2
2005	01-Jan-05 to 31-Dec-05	0.095	8.76	9.5
2006	01-Jan-06 to 31-Dec-06	0.007	8.06	0.7
2007	01-Jan-07 to 31-Dec-07	0.363	10.90	36.3
2008	01-Jan-08 to 31-Dec-08	1.222	17.78	122.2

Table note: ^a ΔR is the Total Equity to Market Price Regulatory Relative Delta calculated by applying equation (3.51).

Table L.1⁵⁴ shows the modified delta component, ΔR , based on the Fama and French (2008) time dependent treatment of the book-to-market ratio, and exhibits the effective adjustment applied to the 8% minimum capital reserve for the fiscal years 2004 to 2008. The ΔR component is developed in this study as the *regulatory relative delta* measure and is specified in equations (L.1) and (L.2). This ΔR measure is evaluated from the log change in total equity, ΔE , and the log change in market price, ΔM . Specifying 1st January by t - 1 and 31st December by t, then ΔR is computed using:

$$\Delta R_{t-1,t} = \Delta E_{t-1,t} - \Delta M_{t-1,t} \tag{L.1}$$

Applying the notation from Fama and French (2008), the log change in total equity (book value) becomes dE, and the log change in market price or the market price return becomes dM. Then maintaining the ΔR symbol to allow for its development in this study, equation (L.1) may be described by:

$$\Delta R_{t-1,t} = dE_{t-1,t} - dM_{t-1,t}$$
(L.2)

The adjustment applied to the minimum regulatory capital, presented in Table L.1 in the column *Change in 8% Capital Requirement* is specified by:

Adjusted Regulatory Capital_{t-1,t} =
$$8\% \times (1 + \Delta R_{t-1,t})$$
 (L.3)

As detailed in Section 3.9.6, the *regulatory relative delta framework* specified in (L.3) is developed to a sophisticated level in this study. As indicated earlier, the building blocks for this framework are provided by extending the Fama and French (2008) book-to-market ratio treatment. The first part of this extension is termed the *relative delta measure*. The specification for the relative delta is detailed in Appendix C. It can be seen by examining Section 3.2.4 and 3.9.6 that the regulatory relative delta framework is a special implementation of the relative delta measurement model.

⁵⁴ For ease of referencing, Table L.1 reproduces Table 4.25 presented in Section 4.10.

APPENDIX M QUANTITATIVE ANALYTICAL APPROACHES APPLIED IN THIS STUDY

This appendix presents the principle quantitative analytical approaches applied in this study. These are descriptive statistics and descriptive distribution statistics, and correlation and regression analysis. In addition, the formulas applied to calculate the change in variables and the market price return are detailed.

M.1 Descriptive Statistics

This study applies a number of statistical measures based on the *statistical mean* and *statistical standard* deviation. The foundations for these measures are detailed in the following sections.

M.1.1 Time Series Data Mean and Standard Deviation

For the time series periods: t = 1, 2, ..., T, the random variable Ω is applied to represent a population of interest. Populations that would be examined as a basis of this variable that represents an observable and quantifiable attribute, are the Gross Domestic Product (GDP), stock market, interest rate, and foreign exchange indices. The *statistical mean* (also termed the average) of this random variable, Ω , is given by:

$$\mu_T = \frac{1}{T} \sum_{t=1}^T \Omega_t \tag{M.1}$$

Where:

 μ_T = population statistical mean at time T

- Ω_t = random variable at time t
- T =maximum time index

t = time index

From the *population statistical mean* calculated using equation (M.1), the population's *statistical standard deviation* is given by:

$$\sigma_{T} = \sqrt{\frac{1}{T} \sum_{t=1}^{T} (\Omega_{t} - \mu_{T})^{2}}$$
(M.2)

Where:

- σ_T = population statistical standard deviation at time *T*
- μ_T = population statistical mean at time *T*
- Ω_t = random variable at time *t*
- T =maximum time index
- t = time index

In addition to the population standard deviation specified in equation (M.2), this study calculates the standard deviation for samples. The sample standard deviation is calculated in a similar way to the population standard deviation. However, a degrees of freedom adjustment is applied to the observation count, *T*. To calculate the sample standard deviation, a (T - 1) adjustment is applied to equation (M.2). This adjustment effectively substitutes the $\frac{1}{T}$ population parameter with a sample parameter, $\frac{1}{(T-1)}$.

M.1.2 Cross-Sectional Data Mean and Standard Deviation

At time *t*, for a population of *N* firms, the *statistical mean* for a random variable Γ that represents for all *N* firms an observable and quantifiable attribute, is given by:

$$\mu_{N,t} = \frac{1}{N} \sum_{i=1}^{N} \Gamma_{i,t} \tag{M.3}$$

Where:

 $\mu_{N,t}$ = population statistical mean for N firms at time t

 $\Gamma_{i,t}$ = random variable for the i^{th} firm at time t

N = number of firms that represent the population at time t

i =firm index

t = time index

From the population's statistical mean calculated using equation (M.3), the population's *statistical standard deviation* of *N* firms is given by:

$$\sigma_{N,t} = \sqrt{\frac{1}{N} \sum_{i=1}^{N} (\Gamma_{i,t} - \mu_{N,t})^2}$$
(M.4)

Where:

 $\sigma_{N,t}$ = population standard deviation for N firms at time t

 $\mu_{N,t}$ = population statistical mean for N firms at time t

 $\Gamma_{i,t}$ = random variable for the i^{th} firm at time t

N = number of firms that represent the population at time t

i =firm index

t = time index

M.1.3 Time Series Mean and Standard Deviation using Cross-Sectional Data

To determine the time series mean for a population of N firms comprises of two calculations. First the cross-sectional *statistical mean* for the N firms for each time period t = 1, 2, ..., T is calculated. Second, the *statistical mean* of the series of these cross-sectional *means* is calculated.

Using the same formula as (M.3), the formula to first calculate the cross-sectional *statistical mean* for N firms at time t is given by:

$$\mu_{N,t} = \frac{1}{N} \sum_{i=1}^{N} \Gamma_{i,t} \tag{M.5}$$

Where the parameter definitions are the same as equation (M.3).

Second, using equation (M.5) the mean for the population of N firms over the time period: t = 1, 2, ..., T, is calculated:

$$\mu_{N,T} = \frac{1}{T} \sum_{t=1}^{T} \mu_{N,t} \tag{M.6}$$

Where:

 $\mu_{N,T}$ = population statistical mean for *N* firms at time *T* $\mu_{N,t}$ = population statistical mean for *N* firms at time *t*

- N = number of firms that represent the population at time t
- T =maximum time index

t = time index

From the population's *statistical mean* calculated using equation (M.6), the population's *statistical standard deviation* for *N* firms is given by:

$$\sigma_{N,T} = \sqrt{\frac{1}{T} \sum_{t=1}^{T} (\mu_{N,t} - \mu_{N,T})^2}$$
(M.7)

Where:

 $\sigma_{N,T}$ = population statistical standard deviation for N firms at time T

 $\mu_{N,t}$ = population statistical mean for N firms at time t

 $\mu_{N,T}$ = population statistical mean for N firms at time T

N = number of firms that represent the population at time t

T =maximum time index

t = time index

M.1.4 Panel Data Mean and Standard Deviation

To determine the mean for a population of N firms, where a single panel of data consists of the time series periods: $\tau = 1, 2, ..., \Psi$, such that the single data panel consists of $\Psi \times N$ data elements, comprises of two calculations. First the *statistical mean* for each firm over the time period $\tau = 1, 2, ..., \Psi$ is calculated. Second, the statistical mean of the series of these N individual *firm means* is calculated.

The formula to first calculate the statistical mean for one firm, over the time period: $\tau = 1, 2, ..., \Psi$ is given by:

$$\mu_{\Psi,i} = \frac{1}{\Psi} \sum_{\tau=1}^{\Psi} \Gamma_{\tau,i} \tag{M.8}$$

Where:

 $\mu_{\Psi,i}$ = population statistical mean for i^{th} firm at time Ψ $\Gamma_{\tau,i}$ = random variable for the i^{th} firm at time τ N = number of firms that represent the population at time Ψ

- i =firm index
- Ψ = maximum time index for one data panel
- τ = time index within one data panel

Second, using equation (M.8) the mean for the population of N firms is calculated using the following formula:

$$\mu_{\Psi,N} = \frac{1}{N} \sum_{i=1}^{N} \mu_{\Psi,i}$$
(M.9)

Where:

 $\mu_{\Psi,N}$ = population statistical mean for *N* firms at time Ψ $\mu_{\Psi,i}$ = population statistical mean for the *i*th firm at time Ψ *N* = number of firms that represent the population at time Ψ *i* = firm index Ψ = maximum time index for one data panel τ = time index within one data panel

Equations (M.8) and (M.9) can be abbreviated to produce the following formula:

$$\mu_{\Psi,N} = \frac{1}{N} \sum_{i=1}^{N} \left(\frac{1}{\Psi} \sum_{\tau=1}^{\Psi} \Gamma_{\tau,i} \right) \tag{M.10}$$

Where the parameter definitions are the same as equations (M.8) and (M.9).

From the population's *statistical mean* calculated using equation (M.10), the population's *statistical standard deviation* for N firms is given by:

$$\sigma_{\Psi,N} = \sqrt{\frac{1}{N} \sum_{i=1}^{N} (\mu_{\Psi,i} - \mu_{\Psi,N})^2}$$
(M.11)

Where:

 $\sigma_{\Psi,N}$ = population standard deviation for *N* firms at time Ψ

 $\mu_{\Psi,i}$ = population statistical mean for i^{th} firm at time Ψ $\mu_{\Psi,N}$ = population statistical mean for *N* firms at time Ψ *N* = number of firms that represent the population at time Ψ *i* = firm index Ψ = maximum time index for one data panel

 τ = time index within one data panel

M.2 Correlation Analysis

Correlation analysis provides a measure of how closely two random variables vary together. The Pearson correlation coefficient r specified by Pearson, Jeffery, Elderton (1929), Hull (2009) and Newbold (1995) is applied in this study to quantify this relationship between selected variables. If applying the Pearson correlation approach to determine the relationship between the accounting variable, denoted by AC, and the market price variable, denoted by M, then:

M.2.1 Correlation Equations

For a given time series specified by the time values:

$$t = 1, 2, \dots, T$$
 (M.12)

The Pearson correlation coefficient is calculated by:

$$r(AC, M) = \frac{\sum_{t=1}^{T} (AC_t - \overline{AC})(M_t - \overline{M})}{\sqrt{\sum_{t=1}^{T} (AC_t - \overline{A})^2} \sqrt{\sum_{t=1}^{T} (M_t - \overline{M})^2}}$$
(M.13)

Where:

r = Pearson correlation coefficient

 AC_t = accounting variables value at time t

 M_t = market price variables value at time t

 \overline{AC} = accounting variable mean for the time series period specified between t and T

 \overline{M} = market variable mean for the time series period specified between t and T

T = maximum time index

t = time index

The Pearson correlation coefficient r calculated using equation (M.13) evaluates to a number between 1 and -1. If the accounting variable, AC, varies exactly in the same way as the market price variable, M, then r equates to 1 signifying perfect correlation between the variables. If the accounting variable, AC, varies exactly the opposite way as the market price variable, M, then r equates to -1 signifying negative but perfect correlation. If the accounting variable, AC, does not vary the same way as the market price variable, M, then r equates to 0 signifying an absence of correlation between the variables. For a measure of r between 1 to 0 and 0 to -1 signifies the relative strength or magnitude of the variation and its direction. This measure quantifying the relationship between the two variables (Newbold 1995).

An equivalent form of equation (M.13) presented by Newbold (1995) that provides a simpler computation of the Pearson correlation coefficient is:

$$r(AC, M) = \frac{\sum_{t=1}^{T} AC_t M_t - t\overline{AC}\overline{M}}{\sqrt{(\sum_{t=1}^{T} AC_t^2 - t\overline{AC}^2)(\sum_{t=1}^{T} M_t^2 - t\overline{M}^2)}}$$
(M.14)

Where parameter definitions are the same as equation (M.13).

M.2.2 Correlation Significance Level

Specified by Newbold (1995), the level of significance for the Pearson correlation coefficient in equations (M.13) and (M.14) is calculated by:

$$p\text{-value} = \frac{r}{\sqrt{(1-r^2)/(T-2)}}$$
 (M.15)

Where:

p-value = probability value or the level of significance based on the students *t*-test, sometime referred to as the *t-score*

r = Pearson correlation coefficient

T - 2 = degrees of freedom

T =maximum time index

M.3 Degrees of Freedom

Degrees of freedom, as explained by Newbold (1995), specifies the number of observations minus the number of parameters estimated to evaluate a statistical measure. These measures, as specified in equation (M.15), include the *p*-value.

The degrees of freedom are generally specified by the following formula:

degrees of freedom =
$$I$$
 – number of estimated parameters (M.16)

Where:

I = maximum index count

Examining the Pearson correlation coefficient in equation (M.13), there are two estimated parameters. These are the accounting variable mean and the market price variable mean. To account for these 2 estimations when determining the level of significance in equation (M.13), the correlation coefficient's *p*-value applies T - 2 degrees of freedom.

M.4 Regression Analysis

This study applies regression analysis as one of its principal measurement methods to determine statistical significance when testing its null hypothesis.

Regression analysis, similar to correlation analysis presented in Appendix M.2, provides a measure for the relationship between variables. Specified by Fama and French (1992, 1996, 2008) and Jensen, Black and Scholes (1972), the regression relationship differs in that it is specified using a predictive model. This predictive model is based on the variables categorised either as a dependent variable or as an independent variable. This model then establishes a *straight-line mathematical form* that estimates the dependent variable given values for the independent variable.

To test the null hypothesis in Section 3.3.1 this research applies the cross-sectional regression approach specified by Fama and French (2008), and the time series regression approach specified by Jensen, Black and Scholes (1972). Appendix O presents the cross-sectional and time series regression tests applied in this study. In
general, the *mathematical form* for the *population* regression line is specified by (Newbold 1995):

$$Y_t = \alpha + \beta X_t + \epsilon_t \tag{M.17}$$

Where:

 β = regression slope

 Y_t = dependent variable value at time t

 X_t = independent variable value at time t

 α = intercept

 ϵ_t = error term, representing the difference between the observed and modelled values for Y_t

t = time index

From equation (M.17), the regression analysis approach applied to determine the relationship between the accounting variable, AC, and the market price variable, M, is specified by:

$$M_t = a + bAC_t + e_t \tag{M.18}$$

Where:

b = sample regression slope

 M_t = dependent market price variable at time t

 AC_t = independent accounting variable at time t

a = intercept

 e_t = error term, representing the difference between the observed and modelled values for M_t

t = time index

M.5 Coefficient of Determination, R^2

As informed by Newbold (1995), to indicate how well the regression line presented in equation (M.18) *fits* observed data points, the coefficient of determination R^2 is used.

M.5.1 Coefficient of Determination, R^2 , Equation

The coefficient of determination is the mathematical square of the correlation coefficient presented in Appendix M.2. The formal specification for the coefficient of determination is given by (Newbold 1995):

$$R^2 = 1 - \frac{SSE}{SST} \tag{M.19}$$

Where:

 $R^2 = \text{coefficient of determination}$

SSE = Error Sum of Squares

SST = Total Sum of Squares

The Error Sum of Squares is specified as:

$$SSE = \sum_{t=1}^{T} e_t^2 \tag{M.20}$$

Where:

SSE = Error Sum of Squares

e = error term, representing the difference between the regression model's dependent variable's observed values and modelled values

t = time index

The Total Sum of Squares is specified as:

$$SST = \sum_{t=1}^{T} (M_t - \overline{M})^2$$
 (M.21)

Where:

SST = Total Sum of Squares

 M_t = dependent market price variable at time t

 \overline{M} = mean of the dependent market price variable M for the full time series

T = maximum time index

M.5.2 R^2 Assessment

As stated, the coefficient of determination R^2 specified in equation (M.19), quantifies for a regression model, the level an independent variable explains a dependent variable. It is monitored to determine the goodness of fit for a regression model, with a value close to one indicating a strong regression model, and a value close to zero indicating a weaker model. This research develops and tests a further analytical measure to assist in determining the accuracy of the R^2 measure.

From equation (M.17) this research determines that for a regression model to be accurate, the independent variable X (denoted also by x), and the *residue* variable ϵ (denoted also by e), must be randomly distributed. Utilising the Pearson product-moment correlation coefficient, specified in Appendix M.2, this research proposes and measures the correlation between the independent variable, x, and residue variable, e, to evaluate the relationship's randomness. The resulting correlation level is tested to determine if it is able to provide an assessment, or a check, to augment the R^2 measure. Thus providing a secondary measure to determine how well the regression model predicts observed data. If the proposed correlation coefficient test for x and e is approximately between 0.5 and -0.5 then it is concluded the measure of R^2 is a valid measure for the model's *strength*. If the x and e correlation is significantly above or below zero (greater than 0.5 or less than -0.5 respectively), then further examination of the independent variables regression relationship may be considered.

This correlation, proposed as an assessment for the regression coefficient of determination R^2 , is implemented in this research using the following:

$$r(x,e) = \frac{\sum_{t=1}^{T} (x_t - \overline{MP})(e_t - \bar{e})}{\sqrt{\sum_{t=1}^{T} (x_t - \bar{x})^2} \sqrt{\sum_{t=1}^{T} (e_t - \bar{e})^2}}$$
(M.22).

Where:

r = Pearson correlation coefficient

- x_t = independent variable at time t
- \bar{x} = independent variable mean for the full time series
- e_t = error variable at time t
- \bar{e} = error variable mean for the full time series
- T = maximum time index

This R^2 assessment, in equation (M.22), is tested by applying it to the individual bank regressions specified in Section 3.9.7.

M.6 Market Price Change

Specified by JPMorgan Chase and Reuters (1996) and Hull (2009) the price change for a financial security can be measured using a number of methods. Three such methods are termed the absolute price change, relative price change and the log price change.

M.6.1 Absolute Price Change

For a financial instrument with price P_t , at a time instance t, the formula specified by JPMorgan Chase and Reuters (1996) to calculate the absolute price change from time t - 1 to t, is given by:

$$D_{t-1,t} = P_t - P_{t-1} \tag{M.23}$$

Where:

M.6.2

 $D_{t-1,t}$ = absolute price change from time t - 1 to t P_t = price at time t P_{t-1} = price at time t - 1t = time index

Relative Price Change

For the same time period, t - 1 to t, the formula for calculating the relative price change⁵⁵ for a financial instrument is given by:

$$R_{t-1,t} = \frac{P_t - P_{t-1}}{P_{t-1}} \tag{M.24}$$

Where:

 $R_{t-1,t}$ = relative price change from time t - 1 to t

 P_t = price at time t

 P_{t-1} = price at time t-1

⁵⁵ JPMorgan Chase and Reuters (1996) states that the relative price change is also termed the *percent return*. However, the relative price change is normally stated as a decimal number.

M.6.3 Log Price Change

The relative price change formula, specified in equation (M.24), takes the same structure as the standard arithmetic percentage change formula.

Simplifying equation (M.24) gives:

$$R_{t-1,t} = \frac{P_t}{P_{t-1}} - \frac{P_{t-1}}{P_{t-1}}$$

$$= \frac{P_t}{P_{t-1}} - 1$$
(M.25)

Rearranging equation (M.25) gives:

$$1 + R_{t-1,t} = \frac{P_t}{P_{t-1}} \tag{M.26}$$

Where the fractional, or decimal form, of the gross return for a financial instrument can be calculated by using the right-hand side expression presented in equation (M.26), that is P_t/P_{t-1} . Thus, the left-hand side of equation (M.26) represents the more familiar gross return formula: $1 + R_{t-1,t}$. From the definitions presented by JPMorgan Chase and Reuters (1996), taking the natural logarithm of the gross return presented by the left-hand side of equation (M.26) gives:

$$r_{t-1,t} = ln(1 + R_{t-1,t})$$
 (M.27)

Where, for a financial instrument, the term $ln(1 + R_{t-1,t})$ represents the instrument's continuously compounded return.

Substituting the right-hand side from equation (M.26) into equation (M.27) gives the formula for calculating the log price change. This log price change formula for a financial instrument from time t - 1 to t, is given by:

$$r_{t-1,t} = ln\left(\frac{P_t}{P_{t-1}}\right) \tag{M.28}$$

Where:

 $r_{t-1,t} = \log \text{ price change from time } t - 1 \text{ to } t$ ln = natural logarithm to the base e $P_t = \text{price at time } t$ $P_{t-1} = \text{price at time } t - 1$ t = time index

The *relative* expression presented in equation (M.28) can also be presented by using the logarithm identity: $\ln (x/y) = \ln (x) - \ln (y)$ to give:

$$r_{t-1,t} = ln(P_t) - ln(P_{t-1})$$
(M.29)

Where the parameter definitions are the same as equation (M.28).

M.7 Market Price Return

JPMorgan Chase and Reuters (1996) specify a *return* to be a measure of the change in price relative to an initial price. Both the relative price change, specified in equation (M.24), and the log price change measure, specified in equation (M.28), maintain this relative quality. That is, both quantify a price change relative to an initial price level. Thus both these measures, the relative and log price changes, are classified as *returns*. The absolute price change, specified in equation (M.23), also measures the price from an initial level, however, it does not proportion the change based on the initial value.

This study applies the log price change formula specified in equation (M.28) to measure the market price variable's relative price changes. In order to emphasise - *the change in the market price variable*, this change is also referred to as the *market price return*, or simply as the *return*.

This study also applies the structure of equations (M.24) and (M.29) to measure the relative change and log change respectively for economic, market and accounting variables.

M.8 General Variable Change

The formulas specified in equations (M.24) for the relative price change for the market price variable, and equations (M.28) and (M.29) for the log price change, can also be applied to measure relative and log changes for any time series variable.

Equation (M.24) applied to the measure of a general variable's relative change is given by:

$$\Gamma_{t-1,t} = \frac{\Gamma_t - \Gamma_{t-1}}{\Gamma_{t-1}} \tag{M.30}$$

Where:

 $\Gamma_{t-1,t}$ = relative variable change from time t - 1 to t Γ_t = variable value at time t Γ_{t-1} = variable value at time t - 1t = time index

Equation (M.28) applied to the measure a general variable's log change is given by:

$$\Gamma_{t-1,t} = ln\left(\frac{\Gamma_t}{\Gamma_{t-1}}\right)$$
 (M.31)

Where:

 $\Gamma_{t-1,t} = \log \text{ variable change from time } t - 1 \text{ to } t$ ln = natural logarithm to the base e $\Gamma_t = \text{variable value at time } t$ $\Gamma_{t-1} = \text{variable value at time } t - 1$ t = time index

Applying the *relative* expression specified in equation (M.29) to equation (M.31), to measure a variable's log change, gives:

$$\Gamma_{t-1,t} = ln(\Gamma_t) - ln(\Gamma_{t-1}) \tag{M.32}$$

Where the parameter definitions are the same as equation (M.31).

M.9 Notation Applied for Referencing Variable Change

This section presents the association with *variable change* notations applied in this study to the notations presented and referred to in Appendix M.6, M.7 and M.8.

This study refers to *variable changes* by applying the delta notation. Thus the change in market price for both the relative and log changes, given by the following -

$$R_{t-1,t}$$
 or $r_{t-1,t}$ is represented by:
 $dM_{t-1,t}$
(M.33)

Where:

 $R_{t-1,t}$ = relative price change variable from time t - 1 to t $r_{t-1,t}$ = log price change variable from time t - 1 to t $dM_{t-1,t}$ = market price change or market price return from time t - 1 to tt = time index

The change in variables for both the relative and log changes, given by the following -

$$\Gamma_{t-1,t}$$
 is represented by:
 $d\Gamma_{t-1,t}$
(M.34)

Where:

 $\Gamma_{t-1,t}$ = relative or log price change variable from time t - 1 to t $d\Gamma_{t-1,t}$ = general variable change from time t - 1 to tt = time index

M.10 Market Price Return Descriptive Statistics

From the basis of the measures detailed in Appendix M.1, the specific sample mean and standard deviation descriptive statistics applied in this study are presented in this section.

M.10.1 Mean

From the basis of equation (M.10), the formula for calculating the average market price return, that is the mean market price return for a sample of N firms for the time period t = 1, 2, ..., T, is given by:

$$dM_{T,N} = \frac{1}{N} \sum_{i=1}^{N} \left(\frac{1}{T} \sum_{t=1}^{T} dM_{t,i} \right)$$
(M.35)

Where:

 $dM_{T,N}$ = mean sample return for *N* firms for time or year *T*

 $dM_{t,i}$ = return for the i^{th} firm at time t

N = number of sample firms

i = firm index

T = maximum time index representing the maximum for the number of observation days in 1 year

t = time index

M.10.2 Standard Deviation

From the basis of equation (M.11), the formula for calculating the standard deviation of the average returns, or mean returns, for a sample of N firms at time T is given by:

$$SD_{T,N} = \sqrt{\frac{1}{N-1} \sum_{i=1}^{N} (dM_{T,i} - dM_{T,N})^2}$$
 (M.36)

Where:

 $SD_{T,N}$ = standard deviation for *N* firms at time *T*

 $dM_{T,N}$ = mean sample return for *N* firms for year *T*

 $dM_{T,i}$ = mean return for the i^{th} firm at time T

N = number of sample firms

i =firm index

T = maximum time index representing the maximum for the number of observation days in 1 year

M.11 Distribution Descriptive Statistics

To describe the distribution for 1-day market price returns, this study applies a series of distribution related descriptive statistics. These statistics applied in this study are the distribution: mean, standard deviation, skew, kurtosis and excess kurtosis. These distribution statistics are built on the mean and standard deviation formulas presented by equations (M.35) and (M.36) respectively, and are applied to 1-day market price returns calculated using 300 days of historical 1-day returns.

M.11.1 Mean and Standard Deviation

Specifying equations (M.35) and (M.36) based on a 300 day basis gives the following formulas:

The formula for the 300-day sample distribution mean is given by:

$$dM_{300,N} = \frac{1}{N} \sum_{i=1}^{N} \left(\frac{1}{300} \sum_{t=1}^{300} dM_{t,i} \right)$$
(M.37)

Where:

 $dM_{300,N}$ = mean sample 1-day returns for *N* firms for time *T* calculated using 300 days of historical 1-day returns

 $dM_{t,i}$ = return for the i^{th} firm at time t

N = number of sample firms

i =firm index

T = maximum time index representing the maximum number of distribution observation days (300 days)

t = time index representing 1 day

The formula to calculate the standard deviation for the mean sample distribution is given by:

$$SD_{300,N} = \sqrt{\frac{1}{N-1} \sum_{i=1}^{N} (dM_{300,i} - dM_{300,N})^2}$$
 (M.38)

Where:

 $SD_{300,N}$ = standard deviation of distribution means for N firms at time T = 300 days calculated using the mean sample $dM_{300,N}$ $dM_{300,i}$ = mean return for the i^{th} firm at time T $dM_{t,i}$ = return for the i^{th} firm at time t N = number of sample firms i = firm index T = maximum time index representing the maximum number of distribution observation

days (300 days)

M.11.2 Skew

The formula to calculate the distribution skew at the firm level, building from the mean and standard deviation equations (M.35) and (M.36), is given by:

$$Skew_{T,i} = \frac{T}{(T-1)(T-2)} \left[\sum_{t=1}^{T} \left(\frac{dM_{t,i} - dM_{T,i}}{SD_{T,i}} \right) \right]^3$$
(M.39)

Where:

 $Skew_{T,i}$ = distribution skew for the i^{th} firm at time T

 $dM_{t,i}$ = return for the i^{th} firm at time t

 $dM_{T,i}$ = mean return for the i^{th} firm at time T

 $SD_{T.i}$ = standard deviation for the i^{th} firm at time T

i =firm index

T = maximum time index representing the maximum number of distribution observation days (300 days)

t = time index

The formula to calculate the average distribution skew for a sample is given by:

$$Skew_{T,N} = \frac{1}{N} \sum_{i=1}^{N} Skew_{T,i}$$
(M.40)

Where:

 $Skew_{T,N}$ = mean distribution skew for N firms at time T

 $Skew_{T,i}$ = distribution skew for the i^{th} firm at time T

N = number of sample firms

i =firm index

T = maximum time index representing the maximum number of distribution observation days (300 days)

The formula to calculate the standard deviation of mean sample distribution skews is given by:

$$SD \; Skew_{T,N} = \sqrt{\frac{1}{N-1} \sum_{i=1}^{N} (Skew_{T,i} - Skew_{T,N})^2} \tag{M.41}$$

Where:

*SD Skew*_{*T*,*N*} = skew standard deviation of distribution skew means for *N* firms at time *T Skew*_{*T*,*i*} = distribution skew for the *i*th firm at time *T*

 $Skew_{T,N}$ = mean distribution skew for N firms at time T

N = number of sample firms

i =firm index

T = maximum time index representing the maximum number of distribution observation days (300 days)

M.11.3 Kurtosis

The formula to calculate the distribution kurtosis at the firm level, building from the mean and standard deviation equations (M.35) and (M.36), is given by:

$$Kurtosis_{T,i} = \frac{T(T+1)}{(T-1)(T-2)(T-3)} \times \left[\sum_{t=1}^{T} \left(\frac{dM_{t,i} - dM_{T,i}}{SD_{T,i}^{4}}\right)\right]^{4}$$
(M.42)

Where:

 $Kurtosis_{T,i}$ = distribution kurtosis for the i^{th} firm at time T

 $dM_{t,i}$ = return for the i^{th} firm at time t

 $dM_{T,i}$ = mean return for the i^{th} firm at time T

 $SD_{T,N}$ = standard deviation of distribution means for *N* firms at time *T* calculated using the mean sample $dM_{T,N}$

i =firm index

T = maximum time index representing the maximum number of distribution observation days (300 days)

t = time index

The formula to calculate the average distribution kurtosis for a sample is given by:

$$Kurtosis_{T,N} = \frac{1}{N} \sum_{i=1}^{N} Kurtosis_{T,i}$$
(M.43)

Where:

 $Kurtosis_{T,N}$ = mean distribution kurtosis for N firms at time T

 $Kurtosis_{T,i}$ = distribution kurtosis for the i^{th} firm at time T

N = number of sample firms

i =firm index

T = maximum time index representing the maximum number of distribution observation days (300 days)

The formula to calculate the standard deviation of mean sample distribution kurtosis is given by:

$$SD \ Kurtosis_{T,N} = \sqrt{\frac{1}{N-1} \sum_{i=1}^{N} (Kurtosis_{T,i} - Kurtosis_{T,N})^2} \qquad (M.44)$$

Where:

SD $Kurtosis_{T,N}$ = kurtosis standard deviation of distribution kurtosis means for N firms at time T

 $Kurtosis_{T,N}$ = mean distribution kurtosis for N firms at time T

 $Kurtosis_{T,i}$ = distribution kurtosis for the i^{th} firm at time T

N = number of sample firms

i =firm index

T = maximum time index representing the maximum number of distribution observation days (300 days)

M.11.4 Excess Kurtosis

The formula to calculate the distribution excess kurtosis at the firm level is given by:

$$EKurtosis_{T,i} = Kurtosis_{T,i} - 3\frac{(T-1)^2}{(T-2)(T-3)}$$
(M.45)

Where:

 $EKurtosis_{T,i}$ = excess kurtosis for the *i*th firm at time *T*

 $Kurtosis_{T,i}$ = distribution kurtosis for the i^{th} firm at time T

i =firm index

T = maximum time index representing the maximum number of distribution observation days (300 days)

The formula to calculate the average distribution kurtosis for a sample is given by:

$$EKurtosi_{T,N} = \frac{1}{N} \sum_{i=1}^{N} EKurtosis_{T,i}$$
(M.46)

Where:

 $EKurtosis_{T,N}$ = mean distribution excess kurtosis for N firms at time T

 $EKurtosis_{T,i}$ = excess kurtosis for the i^{th} firm at time T

i =firm index

N = number of sample firms

T = maximum time index representing the maximum number of distribution observation days (300 days)

The formula to calculate the standard deviation of mean sample excess kurtosis is given by:

$$SD \ EKurtosis_{T,N} = \sqrt{\frac{1}{N-1} \sum_{i=1}^{N} (EKurtosis_{T,i} - EKurtosis_{T,N})^2} \qquad (M.47)$$

Where:

SD EKurtosis_{T,N} = excess kurtosis standard deviation of distribution excess kurtosis for N firms at time T

 $EKurtosis_{T,N}$ = mean distribution excess kurtosis for N firms at time T

 $EKurtosis_{T,i}$ = distribution excess kurtosis for the i^{th} firm at time T

N = number of sample firms

i =firm index

T = maximum time index representing the maximum number of distribution observation days (300 days)

APPENDIX N VALUE-AT-RISK IMPLEMENTATION DETAILS

This appendix details the variance-covariance, historical and Monte Carlo simulation Value-at-Risk implementation approaches applied in this study. The Value-at-Risk backtesting implementation details are also presented. In addition, the *Historical Value-at-Risk Actual* measure applied throughout this study is specified.

N.1 Value-at-Risk Modelling Approach

Market price return Value-at-Risk analysis is conducted by applying the Value-at-Risk models specified in Table N.1.

Table N.1	Value-at-Risk Model Summary
-----------	-----------------------------

The table shows a summary of the specifications for the Value-at-Risk (VaR) model approaches applied for tests. The VaR Model column presents the names of the three Value-at-Risk models applied. The Confidence Level (%) column presents the three probability confidence levels that each VaR model applies. The Sig α column presents for the confidence levels the three corresponding probability significance levels. The Time Horizon Θ (days) column presents the number of time horizon days that each VaR model applies for testing. The Historical Observations Z (days) column presents the number of days of historical observations applied to calculate the respective VaR models. The table shows that for each VaR model time horizon, the 1 and 250 day VaR is calculated using both 300 and 800 days of historical observations.

VaR Model	Confidence Level (%)	Sig. a ^G	Time Horizon O (days)	Historical Observations Z (days)	
Variance- Covariance	95%	0.05	1	300 500	800
			250		
			500		
			1		
Historical	99%	0.01	250	300	800
MCS			500	500	-
	99.9%	0.001	1	300	800
			250		
			500	500	

Table N.1 (Continued): Table notes: ^G For brevity this study also refers to the Value-at-Risk measure in formulas with the subscript significance alpha (α) symbol: V_{α} . However, in complete form this subscript is written as a confidence level, for example as V_{95} (for the 95% confidence level that refers to the significance level $\alpha = 0.05$), that is as $V_{(100\times[1-\alpha])}$. Tables in the Analysis and Results chapter, Chapter 4, refer to the 0.05 significance level or the 95% confidence level Value-at-Risk as VaR05.

The Value-at-Risk models in Table N.1 are applied for the probabilities specified by the corresponding percentage confidence levels (also translated to the significance levels in the Sig. α column) over the specified time horizon, presented in the Time Horizon Θ column. The Historical Observations column specifies the number of historical observations applied to compute the Value-at-Risk levels. The 300 and 500-days historical observations are under the RiskMetrics specification (JPMorgan Chase and Reuters 1996), and the 800-days historical observations are from the Basel III specification (Basel 2011).

N.2 Market Price Return Value-at-Risk Calculation

The calculation for the market price returns applied for each of the Value-at-Risk models is based on the number of time horizon days presented in Table N.1.

For a time horizon of Θ days, to *determine* the return for the future Θ days, the historical returns for the past Θ days are used. The formula for calculating the return for a single Θ -day historical time interval is given by:

$$dM_{t-\Theta,t} = ln\left(\frac{P_t}{P_{t-\Theta}}\right) \tag{N.1}$$

Where:

 $dM_{t-\Theta,t} = \log$ price return from time t to $t - \Theta$ ln = natural logarithm to the base e $P_t =$ price at time t $P_{t-\Theta} =$ price at time $t - \Theta$ t = time at time 0 (the Value-at-Risk *calculation date*) $\Theta =$ number of time horizon days Equation (N.1) specifies the return for one historical time period of $t - \Theta$ days. To provide a more robust estimate of the expected return for a Θ -day time horizon in the future, a series of historical returns are taken maintaining the Θ -day time interval. The formula for calculating a total of *Z* historical returns is given by:

$$dM_{t-\zeta-\Theta,t-\zeta} = ln\left(\frac{P_{t-\zeta}}{P_{t-\zeta-\Theta}}\right) \tag{N.2}$$

Where:

 $dM_{t-\zeta-\Theta,t-\zeta} = \log \text{ price return from time } t - \zeta \text{ to } t - \zeta - \Theta$ ln = natural logarithm to the base e $P_{t-\zeta} = \text{ price at time } t - \zeta$ $P_{t-\zeta-\Theta} = \text{ price at time } t - \zeta - \Theta$ $t = \text{ time at time } 0 \text{ (the Value-at-Risk$ *calculation date* $)}$ Z = maximum number of historical observations $\zeta = \text{ time index from } 0, 1, 2, ..., Z$ $\Theta = \text{ number of time horizon days}$

Presented in Table N.1, and stated earlier, the number of observations, *Z*, applied in this study, range from 300 to 800 days.

N.3 Variance-Covariance Value-at-Risk Model

The variance-covariance Value-at-Risk, referred to also as var-covar Value-at-Risk, is evaluated by first calculating, for a given time horizon Θ , and a specified number of historical returns, *Z*, a series of returns using equation (N.2). The series of historical returns are then used to calculate the mean and standard deviation for the time horizon Θ . The formula used to calculate the mean for the variance-covariance Value-at-Risk model for a time horizon Θ is given by:

$$\mu_{\Theta} = \frac{1}{Z} \sum_{\zeta=1}^{Z} dM_{t-\zeta-\Theta,t-\zeta}$$
(N.3)

Where:

 μ_{Θ} = population mean return for *Z* observations

 $dM_{t-\zeta-\Theta,t-\zeta} = \log$ price return from time $t - \zeta - \Theta$ to $t - \zeta$

- *t* = time at time 0 (the Value-at-Risk *calculation date*)
- Z = maximum number of historical observations
- ζ = time index from 0, 1, 2, ..., Z
- Θ = number of time horizon days

The population standard deviation⁵⁶ based on the historical returns is given by:

$$\sigma_{\Theta} = \sqrt{\frac{1}{Z} \sum_{\zeta=1}^{Z} \left(dM_{t-\zeta-\Theta,t-\zeta} - \mu_{\Theta} \right)^2}$$
(N.4)

Where:

 σ_{Θ} = population standard deviation

 μ_{Θ} = population mean return for *Z* observations

 $dM_{t-\zeta-\Theta,t-\zeta} = \log$ price return from time $t - \zeta - \Theta$ to $t - \zeta$

t = time at time 0 (the Value-at-Risk *calculation date*)

Z = maximum number of historical observations

 ζ = time index from 0, 1, 2, ..., Z

 Θ = number of time horizon days

Applying the one-tailed properties of the normal distribution from the evaluated mean and standard deviation, the variance-covariance Value-at-Risk for the 95% and 99% confidence levels (JPMorgan Chase and Reuters 1996, Dowd 2003, Hull 2009) and the 99.9% confidence level are calculated using the following parameters:

$$V_{95\%} = -1.65 \times \sigma_{\Theta} \tag{N.5}$$

$$V_{99\%} = -2.33 \times \sigma_{\Theta} \tag{N.6}$$

$$V_{99.9\%} = -3.09 \times \sigma_{\Theta}$$
 (N.7)

Where in equations (N.5) to (N.7):

 $V_{95\%}$ = Value-at-Risk at the 95% confidence level

 $^{^{56}\}sigma_{\Theta}$ is the population standard deviation specified by the 1/Z devisor and would be applied for the Gross Domestic Product (GDP) and market indicator variance-covariance Value-at-Risk calculations. Generally however, for this study, standard deviation results are produced by applying the sample based 1/(Z-1) devisor.

 $V_{99\%}$ = Value-at-Risk at the 99% confidence level $V_{99.9\%}$ = Value-at-Risk at the 99.9% confidence level σ_{Θ} = population standard deviation from equation (N.4) in Appendix N.3

N.4 Historical Value-at-Risk Model

The historical Value-at-Risk model is evaluated by calculating, for a given time horizon Θ , and a specified number of historical returns, *Z*, a series of returns using equation (N.3). The historical returns are then arranged into a histogram by arranging the returns from the lowest to the highest values.

The historical Value-at-Risk measure is specified such that the lowest return level corresponds to the 0th percentile. Therefore, for 1,001 historical returns, the 51st lowest return read corresponds to the historical Value-at-Risk at the 95% confidence level; the 11th lowest return corresponds to the historical Value-at-Risk at the 99% confidence level (Dowd 2003); and, the 2nd lowest return corresponds to the historical Value-at-Risk at the 99.9% confidence level. However, for the historical days used in this study that are based on the 300 (301); 500 (501); and 800 (801) days, the historical Value-at-Risk that correspond to the 99.9% confidence level would be the same as the smallest return corresponding to the 0th percentile.

The specific approach this study applies to compute historical Value-at-Risk is as follows. For a series of Z market price returns, dM, the *returns* are arranged from the smallest to the largest return such that:

$$dM_{0\%}, dM_{1\%}, \dots, dM_{100\%}$$
 (N.8)

The historical Value-at-Risk is evaluated for the 95%, 99%, and 99.9% confidence levels by applying the following expressions:

$$V_{95\%} = dM_{5\%} \tag{N.9}$$

$$V_{99\%} = dM_{1\%} \tag{N.10}$$

$$V_{99.9\%} = dM_{0.1\%} \tag{N.11}$$

Where:

 $V_{95\%}$ = Value-at-Risk at the 95% confidence level

 $V_{99\%}$ = Value-at-Risk at the 99% confidence level $V_{99.9\%}$ = Value-at-Risk at the 99.9% confidence level $dM_{5\%}$ = log price return at the fifth percentile (5th percentile) $dM_{1\%}$ = log price return at the first percentile (1st percentile) $dM_{0.1\%}$ = log price return at the one tenth percentile (0.1st percentile)

N.5 Monte Carlo Simulation Value-at-Risk Model

The Monte Carlo simulation (MCS) Value-at-Risk and the historical Value-at-Risk models are very similar. They both read the Value-at-Risk levels from a histogram constructed from an ordered set of returns.

The difference between these two models is the method used to generate returns. The MCS Value-at-Risk approach used in this study applies a mathematical model to simulate returns, while the historical Value-at-Risk approach applies actual historical returns.

The simulation approach applied in this study for the Monte Carlo simulation Value-at-Risk model is based on the Monte Carlo simulation approach detailed by Metropolis and Ulam (1949), Dowd (2003) and Hull (2009). In this study, the approach simulates returns dM from market prices denoted by the random variable *P*. To simulate the market price returns the following processes are applied (Dowd 2003, Hull 2009):

$$dM_{t,t+\Delta t} = \ln P(t+\Delta t) - \ln P(t)$$
(N.12)

$$ln P(t + \Delta t) - ln P(t) = \hat{\mu} \Delta t + \sigma \epsilon \sqrt{\Delta t}$$
(N.13)

Where in equations (N.12) and (N.13):

 $dM_{t,t+\Delta t}$ = market price return from time *t* to Δt

 $P(t + \Delta t)$ = future price at a time $t + \Delta t$

P(t) = current price at time t

 $\hat{\mu}$ = mean market price return

 σ = standard deviation or volatility of market price returns

 ϵ = stochastic variable that is represented by a random variable generated from a standard normal cumulative distribution with mean zero and a standard deviation of 1.0

 Δt = time increment t = time index

Equation (N.13) represents the geometric Brownian motion specified by Hull (2009) and named after Robert Brown (1828).

The following process presented in expression (N.14) is attributed to the *Ito Lemma process* named after Kiyoshi Ito (Ito 1951), and is applied to equation (N.13) to produce equation (N.15).

$$\left(\hat{\mu} - \frac{\sigma^2}{2}\right)\Delta t \tag{N.14}$$

Where:

 $\hat{\mu}$ = mean market price return

 σ = standard deviation or volatility of market price returns

 $\Delta t = \text{time increment}$

The process specified in expression (N.14) is applied to a stochastic process, which adjusts the mean market price variable $\hat{\mu}$ for time period Δt by one-half its variance. Application of this *Ito Lemma process*, in equation (N.14), to the process in equation (N.13), evaluates to equations (N.15) and (N.16) (Hull 2009):

$$ln P(t + \Delta t) - ln P(t) = \left(\hat{\mu} - \frac{\sigma^2}{2}\right) \Delta t + \sigma \epsilon \sqrt{\Delta t}$$
(N.15)

$$P(t + \Delta t) = P(t) \times exp^{\left[\left(\hat{\mu} - \frac{\sigma^2}{2}\right)\Delta t + \sigma\epsilon\sqrt{\Delta t}\right]}$$
(N.16)

Where in equations (N.15) and (N.16):

 $P(t + \Delta t)$ = future price at a time $t + \Delta t$

P(t) = current price at time t

exp = exponent of: the mathematical constant approximately equal to 2.718281828

 $\hat{\mu}$ = mean market price return

 σ = standard deviation or volatility of market price returns

 ϵ = stochastic variable that is represented by a random variable generated from a standard normal cumulative distribution with mean zero and a standard deviation of 1.0 Δt = time increment

N.6 Monte Carlo Simulation

The Monte Carlo simulation Value-at-Risk method implements, in this study, the process specified in equation (N.16). There are several methods for determining the standard deviation for the specified process in equation (N.16), including the use of the ARMA plus GARCH models⁵⁷ detailed by Berkowitz and O'Brien (2002).

N.6.1 Monte Carlo Simulation Approach

In this study, the process specified by equation (N.16) is implemented by calculating the mean and standard deviation of the daily returns by observing actual historical time series returns. These returns are calculated based on Θ -day time horizons (where, Θ = number of days for a specified time horizon). The same as with the variance-covariance Value-at-Risk, the mean based on Θ -day returns, denoted by μ_{Θ} , is calculated using the statistical mean specified in equation (N.3) and the standard deviation, denoted by σ_{Θ} , is calculated using the statistical standard deviation specified in equation (N.4).

Substituting the evaluated mean μ_{Θ} and standard deviation σ_{Θ} into equation (N.16) produces:

$$P(t + \Delta t) = P(t)exp^{\left[\left(\mu_{\Theta} - \frac{\sigma_{\Theta}^{2}}{2}\right)\Delta t + \sigma_{\Theta}\epsilon\sqrt{\Delta t}\right]}$$
(N.17)

$$ln\left[\frac{P(t+\Delta t)}{P(t)}\right] = \left(\mu_{\Theta} - \frac{{\sigma_{\Theta}}^2}{2}\right)\Delta t + \sigma_{\Theta}\epsilon\sqrt{\Delta t}$$
(N.18)

Substituting to the right-hand side of equation (N.18) the time increment $\Delta t = 1$, gives:

⁵⁷ The ARMA plus GARCH model is a combination of the Autoregressive Moving Average (ARMA) model and the Generalised Autoregressive Conditional Heteroskedasticity (GARCH) model.

$$ln\left[\frac{P(t+\Delta t)}{P(t)}\right] = \left(\mu_{\Theta} - \frac{{\sigma_{\Theta}}^2}{2}\right) + \sigma_{\Theta}\epsilon \qquad (N.19)$$

$$dM_{t,t+\Delta t} = \left(\mu_{\Theta} - \frac{{\sigma_{\Theta}}^2}{2}\right) + \sigma_{\Theta}\epsilon$$
 (N.20)

Where in equations (N.17) to (N.20):

 $P(t + \Delta t)$ = future price at a time $t + \Delta t$

P(t) = current price at time t

 $dM_{t,t+\Delta t}$ = market price return from t to Δt

 μ_{Θ} = mean of Z returns with returns calculated for Θ -day time horizon

 σ_{Θ} = standard deviation of Z returns with returns calculated for Θ -day time horizon

Z = maximum number of historical observations

 Θ = number of time horizon days

 $exp = exponent \ of$: the mathematical constant approximately equal to 2.718281828 $\epsilon =$ stochastic variable that is represented by a random sample distribution from a standard normal cumulative distribution with mean zero and a standard deviation of 1.0 $\Delta t =$ time increment set to 1 time interval

The formula presented in expression (N.20) is simulated several times, with each simulation applying a differing level of *shock* generated by the stochastic process ϵ . This study applies for each MCS Value-at-Risk calculation 1 million and one simulations. Thus producing 1 million and one simulated returns for the given Θ -day time horizon. Applying the same process as the historical Value-at-Risk model, the simulated returns are arranged into a histogram from the lowest to the highest values.

Referring to the historical Value-at-Risk model description presented in Appendix N.4, the Monte Carlo simulation Value-at-Risk measure is specified such that the lowest return corresponds to the 0th percentile. Therefore, for 1,000,001 historical returns the 50,001st lowest return read, that is the 5th percentile, corresponds to the historical Value-at-Risk at the 95% confidence level; the 10,001st lowest return read, that is the 1st percentile, corresponds to the historical Value-at-Risk at the 95% confidence level; the 10,001st lowest return read, that is the 1st percentile, corresponds to the historical Value-at-Risk at the 99% confidence level; and, the 1,001st lowest return read, that is the one 0.1st percentile, corresponds to the historical Value-at-Risk at the 99.9% confidence level.

The specific approach this study applies to compute MCS Value-at-Risk is as follows. The same as with the historical Value-at-Risk, for the MCS Value-at-Risk, a series of Z = 1,000,001 (one million and one) market price returns, dM, are arranged from the smallest to the largest return such that:

$$dM_{0\%}, dM_{1\%}, \dots, dM_{100\%}$$
 (N.21)

The MCS Value-at-Risk is evaluated for the 95%, 99%, and 99.9% confidence levels by applying the following expressions:

$$V_{95\%} = dM_{5\%} \tag{N.22}$$

$$V_{99\%} = dM_{1\%} \tag{N.23}$$

$$V_{99.9\%} = dM_{0.1\%} \tag{N.24}$$

Where:

 $V_{95\%}$ = Value-at-Risk at the 95% confidence level $V_{99\%}$ = Value-at-Risk at the 99% confidence level $V_{99.9\%}$ = Value-at-Risk at the 99.9% confidence level $dM_{5\%}$ = log price return at the fifth percentile (5th percentile) $dM_{1\%}$ = log price return at the first percentile (1st percentile) $dM_{0.1\%}$ = log price return at the one tenth percentile (0.1st percentile)

N.6.2 Evaluation of the Stochastic Process ϵ

The value of the stochastic process variable ϵ is evaluated by first assigning a probability level for ϵ , represented by $p(\epsilon)$. From this probability level, using a standard normal distribution, its inverse is computed to reveal the value of ϵ that corresponds to this probability level. That is, for a given level of probability, $p(\epsilon)$, ϵ represents the area under the standard normal distribution. The area under the standard normal distribution is evaluated using the inverse standard normal cumulative distribution function, termed the inverse standard normal CDF (Hull 2009).

This study evaluates the inverse standard normal CDF using the following process:

For:

$$p(\epsilon) = 0.5; \ \epsilon = 0$$
 (N.25)

$$p(\epsilon) < 0.5; \ \epsilon = -\sqrt{2 \ln\left[\frac{1}{p(\epsilon)\sqrt{2\pi}}\right]}$$
 (N.26)

$$p(\epsilon) > 0.5$$
; $\epsilon = \sqrt{2 \ln \left[\frac{1}{(1 - p(\epsilon))\sqrt{2\pi}}\right]}$ (N.27)

Where:

 $p(\epsilon)$ = probability of the stochastic variable ϵ , with values between 1 and 0

 ϵ = random stochastic variable with values between + ∞ and - ∞ ⁵⁸

ln = natural logarithm to the base e

 π = mathematical constant approximately equal to 3.141592654

N.6.3 ϵ Adjustment

To adjust the stochastic process variable, ϵ , to better fit a normal distribution, equations (N.26) and (N.27) are adjusted using the following cubic polynomials:

For:
$$p(\epsilon) < 0.5$$
;
 $\epsilon_{error} = Ap(\epsilon)^3 - Bp(\epsilon)^2 + Cp(\epsilon) + D$
(N.28)

For:
$$p(\epsilon) > 0.5$$
;
 $\epsilon_{error} = -Ap(1-\epsilon)^3 - Bp(1-\epsilon)^2 + Cp(1-\epsilon) + D$
(N.29)

Where:

 $p(\epsilon)$ = probability of the stochastic variable ϵ , with values between 1 and 0 ϵ = random stochastic variable with values between + ∞ and $-\infty$ ³³ ϵ_{error} = stochastic process error from the standard normal distribution A, B, C, D = constant parameters

⁵⁸ The $\pm \infty$ (infinity levels) are bounded in this study to an appropriate level of \pm standard deviations (less than ± 20 standard deviations).

The constant parameter coefficients for A, B, C, D are evaluated (rounded up to 4 decimal places) to be:

$$A = 2.6243$$

 $B = 3.5068$
 $C = 0.3295$
 $D = 0.3841$

The adjusted stochastic variable from the inverse standard normal CDF is evaluated as follows:

$$\epsilon_{adjusted} = \epsilon + \epsilon_{error} \tag{N.30}$$

Where:

 $\epsilon_{adjusted}$ = stochastic variable adjusted

 ϵ = stochastic variable

 ϵ_{error} = stochastic variable error

N.7 Backtesting

The variance-covariance, the historical, and the Monte Carlo simulation Value-at-Risk measures are each tested using the actual price return movements observed from the LSE market.

N.7.1 Backtesting Approach

The backtest process specified by Hull (2009), and observed by Berkowitz and O'Brien (2002), evaluates the following process:

$$V_{1,2,\dots,T} > dM_{1,2,\dots,T}$$
 (N.31)

Where:

 $V_{1,2,...,T}$ = Value-at-Risk measure for time periods t = 1, 2, ..., T $dM_{1,2,...,T}$ = market price return value for time periods t = 1, 2, ..., T

T =maximum time index

Expression (N.31) represents the number of times that the actual market price return variable, dM, is less than the predicted Value-at-Risk measure V, for the specified time horizon (Hull 2009). The count of the event where the actual price, based on market price returns, is less than the Value-at-Risk level specifies the number of times the tested Value-at-Risk model is breached⁵⁹.

In this study, the number of times a Value-at-Risk model is breached provides a measure of the level of market price risk for the period of measurement.

N.7.2 In-Time Horizon Backtesting

In-time horizon backtesting is a test performed to evaluate the number of Value-at-Risk breaches from a start date to the Θ -day time horizon date. The return used in this study is a rolling return that maintains the time horizon days. The number of breaches recorded when applying *in-time horizon backtesting* is termed in this study as the *In-Time Horizon Breach Count*.

From equation (N.31) the in-time horizon breach count is represented by:

$$V_{1,2,...,\Theta} > dM_{1,2,...,\Theta}$$
 (N.32)

Where:

 $V_{1,2,...,\Theta}$ = Value-at-Risk measure for time periods $t = 1, 2, ..., \Theta$ $dM_{1,2,...,\Theta}$ = market price return for time periods $t = 1, 2, ..., \Theta$ Θ = time horizon t = time index

N.7.3 End-of-Time Horizon Backtesting

End-of-time horizon backtesting is a test performed to evaluate if the actual return breached the Value-at-Risk level on the Θ -day time horizon date. The return used in this study is a rolling return that maintains the time horizon days. The number of breaches recorded when applying *end-of-time horizon backtesting* is termed in this study as the *End-of-Time Horizon Breach Count*.

⁵⁹ The term VaR model *breach* is also termed VaR model *violation*, (Berkowitz and O'Brien 2002).

From equation (N.32) the end-of-time horizon breach count is represented by:

$$V_{\Theta} > dM_{\Theta} \tag{N.33}$$

Where:

 V_{Θ} = Value-at-Risk measure at time period Θ dM_{Θ} = market price return value for time periods Θ Θ = time horizon

N.8 Historical Value-at-Risk Actual

Applying the Value-at-Risk approaches presented in Appendix N.1, provides a measure of Value-at-Risk for a time interval from day t to a number of days in the future. The number of days in the future is specified by the Θ -day time horizon. This study develops the historical Value-at-Risk approach specified in Appendix N.4 to provide a measure of the actual Value-at-Risk for a specific day, t, for a given Θ -day time horizon. This actual measure of Value-at-Risk is termed in this study as the *Historical Value-at-Risk Actual*.

The expression applied to calculate the *Historical Value-at-Risk Actual* measure at time *t* is given by:

$$V_{\alpha:t:\Theta} = dM_{\alpha:\Theta:Z} \tag{N.34}$$

Where:

 $V_{\alpha:t:\Theta}$ = Historical Value-at-Risk Actual

 $dM_{\alpha:\Theta:Z} = \log$ price return $M_{t-\zeta-\Theta,t-\zeta}$ from time $t-\zeta$ to $t-\zeta-\Theta$ represented by equation (N.2)

 α = significance level: confidence level = $[100 \times (1 - \alpha)]\%$

t = time at time 0 (the Value-at-Risk *calculation date*)

 Θ = number of time horizon days

Z = maximum number of historical observations

 ζ = time index from 0, 1, 2, ..., Z

APPENDIX O QUANTITATIVE ANALYTICAL METHODS IMPLEMENTATION DETAILS

This appendix details the mean and standard deviation calculations and the crosssectional and time series regression implementation approaches applied in this study.

These approaches include examination of *Historical Value-at-Risk Actual* variable characteristics. The regression approach applied to examine the market price returns and *Historical Value-at-Risk Actual* relationship is also specified.

The implementation approach for GDP and selected market indices time series regressions are specified. In addition, the implementation details for market value and book-to-market ratio regressions that apply the Fama and McBeth t-Statistic are specified.

The descriptive statistics and correlation analysis implementation details for accounting total variables, market price variable, selected market variables and the Value-at-Risk variable are specified. In addition, descriptive statistics and correlation analysis implementation details for the *change in* accounting totals and *change in* selected market variables are also detailed.

Implementation details for the change in market value and market price return regression, and the change in key accounting totals and market price return cross-sectional regressions are specified. In addition, the implementation details for the relative delta and Historical Value-at-Risk Actual cross-sectional regressions that are applied to directly test this study's null hypothesis is also presented.

O.1 Historical Value-at-Risk Actual for the 1-Day, 250-Day and 500-Day Time Horizon at the 95% Confidence Level

This study proposes the application of the *Historical Value-At-Risk Actual* measure at the 95% confidence level, and at the 250-day time horizon. For the sample firms used in this study, and to provide a comparison of the *Historical Value-At-Risk Actual* measure

at the 95% confidence level for the 1-day, 250-day, and 500-day time horizons, the mean and the standard deviation on a yearly basis is calculated. The sample mean for time t is calculated by:

$$\bar{V}_{\alpha:T:\Theta,N} = \frac{1}{N} \sum_{i=1}^{N} \left(\frac{1}{T} \sum_{t=1}^{T} V_{\alpha:\Theta:Z,t,i} \right)$$
(O.1)

Where:

 $\overline{V}_{\alpha:T:\Theta,N}$ = mean sample Historical Value-at-Risk Actual for N firms at time T

N = number of sample firms

i =firm index

 $V_{\alpha:\Theta:\mathbf{Z},t,i}$ = Historical Value-at-Risk Actual for the i^{th} firm at time t

 α = significance level: confidence level = $[100 \times (1 - \alpha)]\%$

T =maximum time index

t = time at time 0 (the Value-at-Risk *calculation date*)

 Θ = number of time horizon days

Z = maximum number of historical observations

The standard deviation of the Value-at-Risk sample means is calculated by:

$$SD_{\alpha:T:\Theta,N} = \sqrt{\frac{1}{N-1} \sum_{i=1}^{N} (\bar{V}_{\alpha:T:\Theta,i} - \bar{V}_{\alpha:T:\Theta,N})^2}$$
(O.2)

Where:

 $SD_{\alpha:T:\Theta,N}$ = standard deviation for N firms at year T

 $\overline{V}_{\alpha:T:\Theta,i}$ = mean sample Historical Value-at-Risk Actual for i^{th} firms at time T

 $\overline{V}_{\alpha:T:\Theta,N}$ = mean sample Historical Value-at-Risk Actual for N firms at time T

N = number of sample firms

i =firm index

 α = significance level: confidence level = $[100 \times (1 - \alpha)]\%$

T =maximum time index

 Θ = number of time horizon days

Z = maximum number of historical observations

O.2 Regressions for Historical Value-at-Risk Actual and the 99% and 99.9% Confidence Level to the 95% Confidence Level

The regression analysis method is applied to provide an analysis of the different significance levels for the 250-day *Historical Value-At-Risk Actual*. Cross-sectional and time series regressions are applied that examine the 95% confidence level 250-day time horizon *Historical Value-At-Risk Actual* with the 99% and 99.9% 250-day time horizon *Historical Value-At-Risk Actual*.

O.2.1 Cross-Sectional Regressions for Historical Value-at-Risk Actual at the 99% and 99.9% Confidence Level and the 95% Confidence Level

The cross-sectional regression is applied to the sample on a yearly basis by first using equation (M.8) from Appendix M.1.4 to calculate the yearly Value-at-Risk levels from the daily levels. For each firm, averages are calculated for each confidence level (95%, 99% and 99.9%). Regression is then applied to compare these yearly averages (99% to 95%, and 99.9% to 95%) for firms i = 1, 2, ..., N that represents the regression for that year. This regression process is repeated for year *panels* from 1994 to 2008.

The cross-sectional regression equation applied in this study for the 99% and 95% confidence levels is given by:

$$\bar{V}_{95:\Psi:\Theta,i,t} = a_{95:\Psi:\Theta,i,t} + b\bar{V}_{99:\Psi:\Theta,i,t} + e_{95:\Psi:\Theta,i,t}$$
(O.3)

Where:

 $\overline{V}_{95:\Psi:\Theta,i,t}$ = average Value-at-Risk at the 95% confidence level for the i^{th} firm for year t $\overline{V}_{99:\Psi:\Theta,i,t}$ = average Value-at-Risk at the 99% confidence level for the i^{th} firm for year tb = regression slope

a = regression intercept

e = error term representing the difference between the actual dependent variable values to the calculated dependent variable values using the regression equation

$$i = \text{firm index: } i = 1, 2, ..., N$$

N = number of sample firms

 Ψ = maximum time index based on a single year

t = time index at the year level

 Θ = number of time horizon days

The cross-sectional regression equation applied in this study for the 99.9% to 95% confidence levels is given by:

$$V_{95;\Psi:\Theta,i,t} = a_{95;\Psi:\Theta,i,t} + bV_{99,9;\Psi:\Theta,i,t} + e_{95;\Psi:\Theta,i,t}$$
(O.4)

Where:

 $\overline{V}_{95:\Psi:\Theta,i,t}$ = average Value-at-Risk at the 95% confidence level for the *i*th firm for year t $\overline{V}_{99:\Psi:\Theta,i,t}$ = average Value-at-Risk at the 99% confidence level for the *i*th firm for year t b = regression slope

a = regression intercept

e = error term representing the difference between the actual dependent variable values to the calculated dependent variable values using the regression equation

i = firm index: i = 1, 2, ..., N

N = number of sample firms

 Ψ = maximum time index based on a single year

t = time index at the year level

 Θ = number of time horizon days

O.2.2 Time Series Regressions for Historical Value-at-Risk Actual at the 99% and 99.9% Confidence Level and the 95% Confidence Level

The time series regression is applied to the sample on a yearly basis by first applying equation (M.10) from Appendix M.1.4 to calculate the yearly Value-at-Risk levels from the daily levels. For each year, averages are calculated for each confidence level (95%, 99% and 99.9%). Regression is then applied to compare these yearly averages (99% to 95%, and 99.9% to 95%) for the time series years t = 1, 2, ..., T. This regression process is applied for the time period 1994 to 2008, and the sub-time periods 1994 to 2004 and 1994 to 2007.

The time series regression equation applied in this study for the 99% to 95% confidence level is given by:

$$\bar{V}_{95:\Theta,N,t} = a_{95:\Theta,N,t} + b\bar{V}_{99:\Theta,N,t} + e_{95:\Theta,N,t}$$
(O.5)

Where:

 $\overline{V}_{95:\Theta,N,t}$ = panel average Value-at-Risk at the 95% confidence level at the sample level *N* for year *t*

 $\overline{V}_{99:\Theta,N,t}$ = panel average Value-at-Risk at the 99% confidence level at the sample level N for year t

b = regression slope

a = regression intercept

e = error term representing the difference between the actual dependent variable values to the calculated dependent variable values using the regression equation

N = average number of sample firms

T = maximum time index based on time series interval

t = time index at the year level: t = 1, 2, ..., T

 Θ = number of time horizon days

The time series regression equation applied in this study for the 99.9% to 95% confidence level is given by:

$$\bar{V}_{95:\Theta,N,t} = a_{95:\Theta,N,t} + b\bar{V}_{99.9:\Theta,N,t} + e_{95:\Theta,N,t}$$
(O.6)

Where:

 $\overline{V}_{95:\Theta,N,t}$ = panel average Value-at-Risk at the 95% confidence level at the sample level *N* for year *t*

 $\overline{V}_{99.9:\Theta,N,t}$ = panel average Value-at-Risk at the 99.9% confidence level at the sample level N for year t

b = regression slope

a = regression intercept

e = error term representing the difference between the actual dependent variable values to the calculated dependent variable values using the regression equation

N = average number of sample firms

T = maximum time index based on time series interval

t = time index at the year level: t = 1, 2, ..., T

 Θ = number of time horizon days

O.3 Regressions for Market Price Returns and Historical Value-at-Risk Actual

The regression analysis method is applied to the yearly market price returns as the independent variable and the *Historical Value-At-Risk Actual* at the 95% confidence level for the 250-day time horizon as the dependent variable. The regression for the market price returns and the *Historical Value-At-Risk Actual* is principally applied to identify differences between the samples and the control group.

O.3.1 Cross-Sectional Regressions for Returns and Historical Value-at-Risk Actual

The cross-sectional and time series regressions are applied to the average market price returns to the average *Historical Value-At-Risk Actual*.

To allow a comparative measure of market price returns to that of the 250-day *Historical Value-At-Risk Actual*, the market price returns are also measured on a 250-day basis. The market price returns are calculated using the following:

$$dM_{i:\Psi-1,\Psi} = ln\left(\frac{P_{i:\Psi}}{P_{i:\Psi-1}}\right) \tag{O.7}$$

Where:

 $dM_{i:\Psi-1,\Psi}$ = market price return for the *i*th firm: that is, the log price change specified in equations (M.28) and (M.29) representing the change in market price

ln = natural logarithm to the base e

 $P_{i:\Psi}$ = market price for the i^{th} firm at year Ψ

 $P_{i:\Psi-1}$ = market price for the *i*th firm at year $\Psi - 1$

 Ψ = year count at the firm level

N = average number of sample firms

$$i = \text{firm index: } i = 1, 2, ..., N$$

The *Historical Value-At-Risk Actual* at the 95% confidence level for the 250-day time horizon at the firm level is specified by the following formula:

$$V_{i:\Psi} = V_{i:95:\Psi:\Theta} \tag{O.8}$$

Where:

 $V_{i:\Psi}$ = *Historical Value-At-Risk Actual* for the *i*th firm at the 95% confidence level for the 250-day time horizon, measured at the year-end of the year represented by the year count Ψ

 $V_{95:\Psi:\Theta}$ = *Historical Value-At-Risk Actual* for the *i*th firm at the 95% confidence level for the 250-day time horizon, measured at the year-end of year for the year represented by the year count Ψ

95 = Value-at-Risk confidence level

i = firm index: i = 1, 2, ..., N

 Ψ = year count at the firm level

 Θ = number of time horizon days

The cross-sectional regression equation applied in this study for the market price return to the Value-at-Risk is given by:

$$V_{i:\Psi} = a_{\Psi} + b_{\Psi} dM_{i:\Psi-1,\Psi} + e_{\Psi} \tag{O.9}$$

Where:

 $V_{i:\Psi}$ = *Historical Value-At-Risk Actual* for the *i*th firm at the 95% confidence level for the 250-day time horizon, measured at the year-end of the year represented by the year count Ψ

 $dM_{i:\Psi-1,\Psi}$ = market price return for the i^{th} firm: that is, the log price change

 b_{Ψ} = regression slope

 a_{Ψ} = regression intercept

 e_{ψ} = error term representing the difference between the actual dependent variable values to the calculated dependent variable values using the regression equation

N = average number of sample firms

i = firm index: i = 1, 2, ..., N

 Ψ = maximum time index based on a single year

O.3.2 Time Series Regressions for Market Price Returns and Historical Valueat-Risk Actual

The time series regression applies sample level variable averages on a yearly basis. These averages are calculated by first applying equations (O.7) and (O.8) to calculate
respective market price returns and Historical Value-at-Risk Actuals at the company level. Then equation (M.3) (specified for the population mean) from Appendix M.1.2, is applied to these company level figures to calculate the average at the sample level on a yearly basis.

To calculate the market price return average at the sample level on a yearly (250-day) basis, applying equation (M.3) to equation (0.7) gives:

$$d\bar{M}_{N:t-1,t} = \frac{1}{N} \sum_{i=1}^{N} dM_{i:\Psi-1,\Psi}$$
(O.10)

Where:

 $d\overline{M}_{N:t-1,t}$ = average market price return at time *t* for *N* sample firms $dM_{i:\Psi-1,\Psi}$ = market price return for the *i*th firm: that is, the log price change specified in equations (M.28) and (M.29) representing the change in market price ln = natural logarithm to the base *e* $P_{i:\Psi}$ = market price for the *i*th firm at year Ψ $P_{i:\Psi-1}$ = market price for the *i*th firm at year $\Psi - 1$ N = average number of sample firms i = firm index: i = 1, 2, ..., N Ψ = year count at the firm level T = maximum year count at the sample level t = time series index: t = 1, 2, ..., T

To calculate the Historical Value-at-Risk Actual average at the sample level on a yearly (250-day) basis, applying equation (M.3) to equation (O.8) gives:

$$\bar{V}_{N:t} = \frac{1}{N} \sum_{i=1}^{N} V_{i:\Psi}$$
(O.11)

Where:

 $\overline{V}_{N:t}$ = average *Historical Value-At-Risk Actual* at time *t* for *N* sample firms at the 95% confidence level for the 250-day time horizon

 $V_{i:\Psi}$ = *Historical Value-At-Risk Actual* for the *i*th firm at the 95% confidence level for the 250-day time horizon, measured at the year-end of the year represented by the year count Ψ

N = average number of sample firms

i = firm index: i = 1, 2, ..., N

 Ψ = year count at the firm level

T = maximum year count at the sample level

t =time series index: t = 1, 2, ..., T

The time series regression equation applied in this study for the market price return to the Value-at-Risk is given by:

$$\bar{V}_{N:t} = a_T + b_T d\bar{M}_{N:t-1,t} + e_{N:t}$$
 (O.12)

Where:

 $\overline{V}_{N:t}$ = average *Historical Value-At-Risk Actual* at time *t* for *N* sample firms at the 95% confidence level for the 250-day time horizon

 $d\overline{M}_{N:t-1,t}$ = average market price return at time t for N sample firms

 b_T = regression slope

 a_T = regression intercept

 $e_{N:t}$ = error term representing the difference between the actual dependent variable values to the calculated dependent variable values using the regression equation

N = average number of sample firms

T = maximum year count at the sample level

t =time series index: t = 1, 2, ..., T

O.4 Regressions for GDP and Market Indices and Market Price Return and Value-at-Risk

The GDP and market indices variables selected in this study are treated as representing the respective populations of interest.

The regression analysis method is applied to the yearly change in the Gross Domestic Product (GDP) and market indices variables as the independent variable and market price return as the dependent variable. This regression analysis is applied to identify differences in the market price returns to the change in GDP and indices values.

Regression analysis is also applied to the *GDP and market indices variable Historical Value-At-Risk Actual* and the *market price return Historical Value-At-Risk Actual*. These Historical Value-at-Risk Actual variables are calculated at the 95% confidence level for the 250-day time horizon. In this regression, the GDP and market indices Historical Value-At-Risk Actual is treated as the independent variable and the market price return Historical Value-At-Risk Actual is treated as the dependent variable. This regression analysis is applied to identify differences in the market price return Value-at-Risk to the change in GDP and indices Value-at-Risk levels.

O.4.1 Time Series Regressions for Change in GDP and Market Indices, and Market Price Return and Value-at-Risk

This study applies time series regression to analyse the GDP and market indices variables and the market price return and *Historical Value-At-Risk Actual* variables.

The GDP and market indices variables are calculated on a 250-day basis by applying the form specified in equation (M.31). The GDP and market indices Historical Value-at-Risk Actual is computed on a 250-day basis using the approach specified in Appendix N.8. The principles specified in Appendix O.3.2 are applied to calculate the yearly sample averages for the market price returns, and the yearly sample averages for its Historical Value-at-Risk Actual.

The time series regression equation applied for the GDP and market indices variables to the market price return variable is given by:

$$d\bar{M}_{N:t-1,t} = a_T + b_T d\Gamma_{t-1,t} + e_{N:t-1,t}$$
(O.13)

Where:

 $d\overline{M}_{N:t-1,t}$ = average market price return at time t for N sample firms

 $d\Gamma_{t-1,t}$ = GDP and market indices change

 b_T = regression slope

 a_T = regression intercept

 $e_{N:t-1,t}$ = error term representing the difference between the actual dependent variable values to the calculated dependent variable values using the regression equation

N = average number of sample firms

T = maximum year count at the sample level

t =time series index: t = 1, 2, ..., T

The time series regression equation applied for the *GDP and market indices Historical Value-At-Risk Actual* variables and the *market price return Historical Value-at-Risk Actual* variable is given by:

$$\overline{V}_{N:t} = a_T + b_T V_{\Gamma:t} + e_{N:t} \tag{O.14}$$

Where:

 $\overline{V}_{N:t}$ = average *Historical Value-At-Risk Actual* at time *t* for *N* sample firms at the 95% confidence level for the 250-day time horizon

 $V_{\Gamma:t}$ = GDP and market indices *Historical Value-At-Risk Actual* at time *t* at the 95% confidence level for the 250-day time horizon

 b_T = regression slope

 a_T = regression intercept

 $e_{N:t}$ = error term representing the difference between the actual dependent variable values to the calculated dependent variable values using the regression equation

N = average number of sample firms

T = maximum year count at the sample level

t =time series index: t = 1, 2, ..., T

O.5 Regressions for Market Value and Book-to-Market Ratio to Predict Market Price Returns

The market value variable and book-to-market ratio variable applied to predict market price returns takes the following cross-sectional regression form:

$$dM_{i:t,t+1} = a_{t+1} + b_{1:t+1}MV_{i:t} + b_{2:t+1}BM_{i:t} + e_{i:t,t+1}$$
(O.15)

Where:

 $dM_{i:t,t+1}^{60}$ = market price return for the i^{th} firm from year t to year t + 1

 $MV_{i:t} = \log of market value for the ith firm for year t$

 $BM_{i:t} = \log of the book-to-market ratio for the$ *i*th firm for year t

 $b_{1:t+1}$ = regression slope corresponding to the market value variable MV

 $b_{2:t+1}$ = regression slope corresponding to the book-to-market ratio variable BM

⁶⁰ Fama and French (2008) refer to the future return variable $dM_{i:t,t+1}$ as R_{t+1} . This study refers to the one period future return variable as $dM_{i:t,t+1}$ to maintain continuity with the market price return variable dM used throughout this study.

a_{t+1} = regression intercept

 $e_{i:t,t+1}$ = error term representing the difference between the actual dependent variable values to the calculated dependent variable values using the regression equation

$$N =$$
 average number of sample firms

i = firm index: i = 1, 2, ..., N

T = maximum year count at the sample level

t = time index: t = 1, 2, ..., T

O.6 The Fama and McBeth t-Statistic for Regressions for Market Value and Book-to-Market Ratio to Predict Market Price Returns

The slope and intercept parameters from the regression equation (0.15) is averaged for the time series period t = 1, 2, ..., T. Denoting the averaged regression parameters from equation (0.15) for the time series maximum year count T as $\bar{b}_{1:T}$; $\bar{b}_{2:T}$; and, \bar{a}_{T} . Representing for each regression parameter the general reference $\bar{\gamma}_{T}$, the Fama and McBeth (1973) *t-statistic* is applied using the following formula:

$$t(\bar{\gamma}_T) = \frac{\bar{\gamma}_T}{SD(\bar{\gamma}_T)/\sqrt{T}} \tag{O.16}$$

Where:

 $t(\bar{\gamma}_T)$ = Fama and McBeth t-statistic

 $\bar{\gamma}_T$ = generic representation of the parameters: $\bar{b}_{1:T}$; $\bar{b}_{2:T}$; and, \bar{a}_T from equation (O.15)

 $SD(\bar{\gamma}_T)$ = sample standard deviation at time T for the $\bar{\gamma}_T$ parameters

T = maximum year count at the sample level representing the number of years in the time series

The Fama and McBeth (1973) *t-statistic* is proposed to measure the level of significance aligned to the *p-value* measure of significance. In that the *t-statistic* represents a measure of the standard deviation, while the *p-value* represents the probability level calculated from the *t-statistic* standard deviation.

When applying the Fama and McBeth (1973) *t-statistic*, the R^2 measure is measured by taking the average of the cross-sectional regressions R^2 measure for years t = 1, 2, ..., T.

O.7 Descriptive Statistics for Market and Accounting Totals and Market Price and Value-at-Risk

To determine the descriptive statistics for the mean and standard deviation relationships between the variables book-to-market ratio, market value, market price, and the accounting totals, and Value-at-Risk represented by the Historical Value-At-Risk Actual, descriptive statistics are measured using the formulas presented in equations (0.17) and (0.18).

The statistical mean for the selected variables is measured by developing the population cross-sectional mean equation (M.3) from Appendix M.1.2. The formula to measure the statistical mean for the selected variables is given by:

$$\overline{\Lambda}_{N,t} = \frac{1}{N} \sum_{i=1}^{N} \Lambda_{i,t} \tag{O.17}$$

Where:

 $\overline{\Lambda}_{N,t}$ = mean market and accounting variables total amounts, market price, and Value-at-Risk (substituted to $\overline{V}_{N,t}$ for the Value-at-Risk variable) at time *t* for *N* sample firms $\Lambda_{i,t}$ = market and accounting variables total amounts, market price, and Value-at-Risk (substituted to $V_{N,t}$ for the Value-at-Risk variable) for the *i*th firm at time *t* N = number of firms that represent the sample at time *t* i = firm index t = time index

Developed from equation (M.4) presented to measure the population cross-sectional standard deviation in Appendix M.1.2, the formula to measure the standard deviation of means for market and accounting variable total amounts, market price, and Value-at-Risk is given by:

$$SD(\overline{\Lambda}_{N,t}) = \sqrt{\frac{1}{N-1} \sum_{i=1}^{N} (\Lambda_{i,t} - \overline{\Lambda}_{N,t})^2}$$
(O.18)

Where:

 $SD(\overline{\Lambda}_{N,t}) =$ standard deviation of sample means for market and accounting variables total amounts, market price, and Value-at-Risk at time t for N sample firms $\overline{\Lambda}_{N,t} =$ mean market and accounting variables total amounts, market price, and Value-at-Risk (substituted to $\overline{V}_{N,t}$ for the Value-at-Risk variable) at time t for N sample firms $\Lambda_{i,t} =$ market and accounting variables total amounts, market price, and Value-at-Risk (substituted to $V_{N,t}$ for the Value-at-Risk variable) for the t for N sample firms $\Lambda_{i,t} =$ market and accounting variables total amounts, market price, and Value-at-Risk (substituted to $V_{N,t}$ for the Value-at-Risk variable) for the i^{th} firm at time t N = number of firms that represent the sample at time t i = firm index

t = time index

O.8 Correlations for Market and Accounting Totals and Market Price to Valueat-Risk

To determine the relationship between the Historical Value-At-Risk Actual measure and the *total* variables specified in Appendix O.7, correlation analysis is applied. Pearson correlation, presented in equation (M.13) from Appendix M.2, is the correlation analysis method used and is specified as follows:

For a given time series specified by the time values:

$$t = 1, 2, \dots, T$$
 (0.19)

The Pearson correlation coefficient is calculated by:

$$r(\overline{\Lambda}_{N,T}, \overline{\nabla}_{N,T}) = \frac{\sum_{t=1}^{T} (\overline{\Lambda}_{N,t} - \overline{\Lambda}_{N,T}) (\overline{\nabla}_{N,t} - \overline{\nabla}_{N,T})}{\sqrt{\sum_{t=1}^{T} (\overline{\Lambda}_{N,t} - \overline{\Lambda}_{N,T})^2} \sqrt{\sum_{t=1}^{T} (\overline{\nabla}_{N,t} - \overline{\nabla}_{N,T})^2}$$
(0.20)

Where:

 $r(\overline{\Lambda}_{N,T}, \overline{V}_{N,T}) =$ Pearson correlation coefficient

 $\overline{\Lambda}_{N,T}$ = mean market and accounting variables total amounts and market price variable at time *T* for *N* sample firms

 $\overline{V}_{N,T}$ = mean Historical Value-At-Risk Actual measure at the 95% confidence level, and at the 250-day time horizon at time *T* for *N* sample firms

 $\overline{\Lambda}_{N,t}$ = mean market and accounting variables total amounts and market price variable at time *t* for *N* sample firms

 $\overline{V}_{N,t}$ = mean Historical Value-At-Risk Actual measure at the 95% confidence level, and at the 250-day time horizon at time *t* for *N* sample firms

T = maximum time series index

t = time series index at the year level: t = 1, 2, ..., T

O.9 Descriptive Statistics for Changes in Market and Accounting Totals and Market Price Returns

To calculate the change in the variables in Appendix O.7, the log change equation (M.31) from Appendix M.8 is applied. For variables that contain negative values these changes are omitted from the sample average log change calculation. Variable log changes that are calculated using this approach are labelled with the prefix - *dln*. For such variables, the changes are also calculated, including the negative values, using the relative variable change formula specified in equation (M.30) in Appendix M.8. If the same variable is not presented twice in a result set, that is once with the *dln* prefix label and then without this label, then its change has been calculated using the log change formula specified in equation (M.31).

Equations (M.3) and (M.4) in Appendix M.1.2 are applied to determine the mean and standard deviation respectively for the market and accounting total variable changes and market price change (market price return). From equations (M.3), the formula to measure the statistical mean for the variable changes is given by:

$$\overline{\Lambda}_{N,t-1,t} = \frac{1}{N} \sum_{i=1}^{N} \Lambda_{i,t-1,t}$$
(O.21)

Where:

 $\overline{\Lambda}_{N,t-1,t}$ = average change in the market and accounting variables total amounts and market price (market price return) at time *t* for *N* sample firms

 $\Lambda_{i,t}$ = change in market and accounting variables total amounts and market price (market price return) for the *i*th firm in *N* sample firms at time *t*

N = number of firms that represent the sample at time t

i =firm index

t = time index

Developed from formula (M.4) in Appendix M.1.2, the formula to measure the standard deviation for the variable changes is given by:

$$SD(\overline{\Lambda}_{N,t-1,t}) = \sqrt{\frac{1}{N-1} \sum_{i=1}^{N} (\Lambda_{i,t-1,t} - \overline{\Lambda}_{N,t-1,t})^2}$$
(O.22)

Where:

 $SD(\overline{\Lambda}_{N,t-1,t})$ = standard deviation for the change in market and accounting variables total amounts and market price at time *t* for *N* sample firms

 $\overline{\Lambda}_{N,t-1}$ = average change in market and accounting variables total amounts and market price (market price return) at time *t* for *N* sample firms

 $\Lambda_{i,t}$ = change in the market and accounting variables total amounts and market price (market price return) for the *i*th firm at time *t*

N = number of firms that represent the sample at time t

i =firm index

t = time index

O.10 Correlations for Changes in Market and Accounting Totals and Market Price Returns to Value-at-Risk

To determine the relationship between the Historical Value-At-Risk Actual measure and the change variables specified in Appendix O.7, correlation analysis is applied. Pearson correlation, presented in equation (M.13) from Appendix M.2, is the correlation analysis method used and is specified as follows:

For a given time series specified by the time values:

$$t = 1, 2, \dots, T$$
 (0.23)

The Pearson correlation coefficient is calculated by:

$$r(\overline{\Lambda}_{N,T-1,T}, \overline{\nabla}_{N,T}) = \frac{\sum_{t=1}^{T} (\overline{\Lambda}_{N,t-1,t} - \overline{\Lambda}_{N,T-1,T}) (\overline{\nabla}_{N,t} - \overline{\nabla}_{N,T})}{\sqrt{\sum_{t=1}^{T} (\overline{\Lambda}_{N,t-1,t} - \overline{\Lambda}_{N,T-1,T})^2} \sqrt{\sum_{t=1}^{T} (\overline{\nabla}_{N,t} - \overline{\nabla}_{N,T})^2}$$
(O.24)

Where:

 $r(\overline{\Lambda}_{N,T-1,T}, \overline{V}_{N,T-1,T}) =$ Pearson correlation coefficient

 $\overline{\Lambda}_{N,T-1}$ = average change in the market and accounting variables total amounts and market price variable from time T - 1 to T for N sample firms

 $\overline{V}_{N,T}$ = mean Historical Value-At-Risk Actual measure at the 95% confidence level, and at the 250-day time horizon at time *T* for *N* sample firms

 $\overline{\Lambda}_{N,t-1,t}$ = average change market and accounting variables total amounts and market price variable from time *t*-1 to *t* for *N* sample firms

 $\overline{V}_{N,t}$ = mean Historical Value-At-Risk Actual measure at the 95% confidence level, and at the 250-day time horizon at time *t* for *N* sample firms

T = maximum time series index

t =time series index at the year level: t = 1, 2, ..., T

0.11 Regressions for Change in Market Value and Market Price Return

In Section 3.2.3 it is detailed that the book-to-market ratio measure uses *book equity* to *market value*. This study develops and applies the relative delta measure specified in Section 3.2.4 based on the change in *total equity* to the change in *market price*. The main difference between these two measures is that, the book-to-market ratio applies the market value as its denominator, and the relative delta measure applies the market price as its denominator.

To analyse the relationship between *market value* and *market price*, cross-sectional and time series regressions are applied. The specific variables examined are the *change in the market value* and the *change in market price*.

O.11.1 Cross-Sectional Regressions for Change in Market Value and Market Price Return

The cross-sectional regression applied to determine the relationship between the change in the market value variable and the change in the market price (market price return) variable is specified by the following:

$$dM_{i:\Psi-1,\Psi} = a_{\Psi} + b_{\Psi} dM V_{i:\Psi-1,\Psi} + e_{i:\Psi-1,\Psi}$$
(O.25)

Where:

 $dM_{i:\Psi-1,\Psi}$ = market price return variable for the i^{th} firm: that is, the log price change $dMV_{i:\Psi-1,\Psi}$ = natural log change in the market value variable for the i^{th} firm b_{Ψ} = regression slope

 a_{Ψ} = regression intercept

 $e_{i:\Psi-1,\Psi}$ = error term representing the difference between the actual dependent variable values to the calculated dependent variable values using the regression equation

N = average number of sample firms

i =firm index: i = 1, 2, ..., N

 Ψ = maximum time index based on a single year

O.11.2 Time Series Regressions for Change in Market Value and Market Price Return

The time series regression applied to determine the relationship between the change in the market value variable and the market price return variable is specified by the following:

$$d\bar{M}_{N:t-1,t} = a_T + b_T d\overline{MV}_{N:t-1,t} + e_{N:t-1,t}$$
(O.26)

Where:

 $d\overline{M}_{N:t-1,t}$ = mean market price return at time t for N sample firms

 $d\overline{MV}_{N:t-1,t}$ = average natural log change in the market value at time *t* for *N* sample firms

 b_T = regression slope

 a_T = regression intercept

 $e_{N:t-1,t}$ = error term representing the difference between the actual dependent variable values to the calculated dependent variable values using the regression equation

N = average number of sample firms

T = maximum year count at the sample level

t =time series index: t = 1, 2, ..., T

O.12 Cross-Sectional Regressions for Change in Key Accounting Totals and Market Price Return

The cross-sectional regression applied to determine the relationship between the change in key accounting variables and the market price return variable is specified by the following:

$$dM_{i:\Psi-1,\Psi} = a_{\Psi} + b_{\Psi} dAC_{i:\Psi-1,\Psi} + e_{i:\Psi-1,\Psi}$$
(O.27)

Where:

 $dM_{i:\Psi-1,\Psi}$ = market price return for the i^{th} firm: that is, the log price change

 $dAC_{i:\Psi-1,\Psi}$ = change in the key accounting total variable for the *i*th firm

 b_{Ψ} = regression slope

 a_{Ψ} = regression intercept

 $e_{i:\Psi-1,\Psi}$ = error term representing the difference between the actual dependent variable values to the calculated dependent variable values using the regression equation

N = average number of sample firms

i = firm index: i = 1, 2, ..., N

 Ψ = maximum time index based on a single year

This regression approach is presented in a generalised form for each key accounting total and market price return in Appendix F.1.

O.13 Cross-Sectional Regressions for Relative Delta and Historical Value-at-Risk Actual – Null Hypothesis Evaluation

The cross-sectional regression applied to determine the relationship between the relative delta variable and the Historical Value-at-Risk Actual variable is specified by the following:

$$V_{i:\Psi} = a_{\Psi} + b_{\Psi} (dAC - dM)_{i:\Psi-1,\Psi} + e_{i:\Psi}$$
(O.28)

Where:

 $V_{i:\Psi}$ = *Historical Value-At-Risk Actual* for the *i*th firm measured at the year-end of the year represented by the year count Ψ

 $(dAC - dM)_{i:\Psi-1,\Psi}$ = relative delta measure: change in the key accounting total variable to the market price return variable for the *i*th firm

 b_{ψ} = regression slope

 a_{Ψ} = regression intercept

 $e_{i:\Psi}$ = error term representing the difference between the actual dependent variable values to the calculated dependent variable values using the regression equation

N = average number of sample firms

i = firm index: i = 1, 2, ..., N

 Ψ = maximum time index based on a single year

O.14 Computer System, Data Analysis and Modelling Products

Data analysis and simulations are performed on a computer system with the specifications presented in Table O.1.

Table O.1 Computer System Description and Specification

The table shows information for the computer system used for data analysis. The System Description column presents the systems main characteristics. The Specification column presents the system details that correspond to the system descriptions.

System Description	Specification					
Operating system	Windows 7, 64-bit version					
Processors	Intel Pentium Dual-Core (2 core) Processor					
Processor Speed	2.10 GHz per processor					
Random Access Memory (RAM)	4 Giga bytes					

The data analysis methods and the products, used to perform data analysis in this study, are presented in Table O.2.

Table O.2Data Analysis Description and Products

The table shows the quantitative analysis approaches applied and the computer system products and programming languages used to perform data analysis. The Analysis Description column presents the analysis methods applied. The Products column presents the computer system products and programming languages used to perform the corresponding analysis presented in the Analysis Description column.

Analysis Description	Products
Descriptive statistics (mean and standard deviation) and significance tests	PASW Statistics version 17 ^c SPSS version 17 ^c Excel 2007 SQL Server 2008
Value-at-Risk modelling	SQL Server 2008 Excel 2007 C and C++
Correlation analysis: (inc. Pearson correlation coefficient)	PASW Statistics version 17 ^c SPSS version 17 ^c Excel 2007
Regression analysis: Ordinary Least Squares (OLS) Method	PASW Statistics version 17 ^c SPSS version 17 ^c
Other tests and quality assurance	Excel 2007 PASW Statistics version 17 ^c SPSS version 17 ^c

Table notes: ^C The product SPSS (Statistical Package for the Social Sciences) version 17 was used for the first part of 2009. The statistical product PAWS (Predictive Analytics SoftWare) version17 was used during the latter part of 2009 and thereafter. The PAWS product is the International Business Machines (IBM) Corporation's version of the previously named SPSS product. In 2010, this product was renamed to IBM SPSS Statistics.

APPENDIX P ANALYSIS AND RESULTS FOR MARKET PRICE RETURNS, VALUE-AT-RISK, GDP AND GENERAL MARKET CONDITIONS

The analysis and results for descriptive statistics and distribution descriptive statistics applied to the change in market price (market price return) variable are presented in this appendix. Descriptive statistics are also presented for the variance-covariance Value-at-Risk measure.

This appendix also presents the variance-covariance, historical and Monte Carlo simulation Value-at-Risk model analysis based on average yearly backtested results. The backtested results are presented for selected Value-at-Risk time horizons at the 95% confidence level and calculated for a selected number of historical and simulated return observations.

In addition, analysis and results for the Historical Value-at-Risk Actual measure for selected time horizons and confidence levels are presented. Analysis and results are also presented for the regressions that test the Historical Value-at-Risk Actual variable and the market price return variable.

This appendix also presents analysis and results for reactions to economic and market conditions for the samples and the control group market price return and Historical Value-at-Risk Actual. This analysis is conducted using regressions. Reactions to economic condition are assessed by analysing results from tests with selected GDP sector levels. Reactions to market conditions are assessed by analysing results from tests with selected stock market indices, short and long-term benchmark interest rates, and foreign exchange currency pair rates.

P.1 Market Price Return Distribution and Variance-Covariance Value-at-Risk Analysis

The descriptive statistics results for the natural log market price returns, the market price return distribution, and the variance-covariance Value-at-Risk for a 1-day time

horizon at the 95% confidence level, are presented on a yearly basis for the time range 1993 to 2009 in Table P.1. In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.

Table P.1Market Price Return Distribution and Value-at-Risk Descriptive
Statistics

Table Description

The Table P.1 columns represent the following:

Year is the *panel data* year represented by the number of days from 1st January to 31st December, and also represents the 31st December variable record date for each year. *Obs.* represents the number of sample firms observed for the year.

For the Mean result column: Return is the average 1-day returns for the sample firms observed for year (Year), calculated using equation (M.35). Dist. is the average 1-day return distribution over 300 days for the sample firms, calculated using equation (M.37). Skew is the average skewness based on the distribution, calculated using equation (M.40). Kurtosis is the average kurtosis based on the distribution, calculated using equation (M.43). Excess Kurtosis is the average excess kurtosis based on the distribution, calculated using equation (M.43). Excess Kurtosis is the average 1-day time horizon variance-covariance Value-at-Risk at the 95% confidence level calculated using equation (M.5) for the sample firms observed for year (Year), with the averages calculated by applying equation (M.35).

For the Standard Deviation column: Return represents the standard deviation of 1 day sample return yearly averages, calculated by equation (M.36). Dist. represents the standard deviation of 1 day sample return averages over 300 days, calculated using equation (M.38). Skew is the standard deviation of the sample average skewness, calculated using equation (M.41). Kurtosis is the standard deviation of the sample average kurtosis, calculated using equation (M.44). Excess Kurtosis is the standard deviation of the sample average kurtosis, calculated using equation (M.47). V is the standard deviation of the sample average 1-day time horizon variance-covariance Value-at-Risk at the 95% confidence level, calculated by applying equation (M.36).

The Primary and Secondary Sample Market Price Return Distribution Descriptive Statistics and Variance-Covariance Value-at-Risk using 300 historical observations based on 1-day returns

Table P.1 (Continued)
PANEL A

			Mean							Standard Deviation						
Year	Obs.	Return	Dist.	Skew	Kurtosis	Excess Kurtosis	V	R	eturn	Dist.	Skew	Kurtosis	Excess Kurtosis	V		
1993 1994	15 15	0.0019	0.0022	$0.876 \\ 0.148$	15.98 14 67	12.95 11 64	-0.023 -0.021	0	.0020 0008	$0.0010 \\ 0.0008$	1.962 1.839	14.61 15 82	14.61 15.82	0.020		
1995	16	0.0009	0.0003	-0.314	14.08	11.05	-0.018	0	.0009	0.0006	1.469	15.73	15.73	0.010		
1996	16	0.0004	0.0007	0.279	11.83	8.80	-0.016	0	.0005	0.0006	1.006	9.25	9.25	0.006		
1997	16	0.0010	0.0008	0.225	12.13	9.10	-0.018	0	.0008	0.0006	1.082	10.73	10.73	0.007		
1998	16	0.0002	0.0008	0.161	10.52	7.49	-0.027	0	.0006	0.0007	0.715	7.10	7.10	0.012		
1999	16	0.0012	0.0006	0.687	14.52	11.49	-0.034	0	.0012	0.0005	2.293	30.96	30.96	0.013		
2000	16	0.0006	0.0006	0.420	9.82	6.79	-0.032	0	.0009	0.0013	0.736	10.58	10.58	0.014		
2001	16	-0.0006	0.0001	-0.202	7.76	4.73	-0.032	0	.0007	0.0006	0.518	5.08	5.08	0.013		
2002	16	-0.0015	-0.0008	-0.312	8.07	5.04	-0.034	0	.0015	0.0009	0.513	4.06	4.06	0.014		
2003	16	0.0009	-0.0006	-0.044	6.13	3.10	-0.036	0	.0006	0.0010	0.407	2.49	2.49	0.020		
2004	16	0.0004	0.0006	0.112	6.18	3.15	-0.023	0	.0004	0.0004	0.427	3.26	3.26	0.010		
2005	16	0.0007	0.0006	-0.020	6.91	3.88	-0.019	0	.0007	0.0004	0.571	3.68	3.68	0.006		
2006	16	0.0006	0.0008	0.026	6.55	3.52	-0.021	0	.0003	0.0006	0.518	2.03	2.03	0.006		
2007	16	-0.0006	0.0002	-0.072	6.45	3.42	-0.024	0	.0016	0.0006	0.329	2.31	2.31	0.007		
2008	16	-0.0024	-0.0015	-0.049	8.79	5.76	-0.043	0	.0022	0.0019	0.808	7.12	7.12	0.022		
2009	16	0.0012	-0.0014	-0.140	11.30	8.27	-0.071	0	.0014	0.0018	1.327	12.50	12.50	0.035		

Panel notes: The variable means are sample averages. The sample consists of a maximum of 16 observations per year from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.

PANEL B The Control Group Market Price Return Distribution Descriptive Statistics and Variance-Covariance Value-at-Risk using 300 historical observations based on 1-day returns

Table P.1 (Continued)

Vear Obs				М	ean			Standard Deviation						
Year	Obs.	Return	Dist.	Skew	Kurtosis	Excess Kurtosis	V	Return	Dist.	Skew	Kurtosis	Excess Kurtosis	V	
1993	12	0.0014	0.0015	0.560	7.89	4.86	-0.016	0.0009	0.0005	0.461	3.14	3.14	0.005	
1994	12	-0.0005	0.0006	-0.037	7.50	4.47	-0.017	0.0003	0.0003	0.520	4.39	4.39	0.006	
1995	12	0.0003	-0.0002	-0.407	6.30	3.27	-0.014	0.0004	0.0003	0.196	2.95	2.95	0.004	
1996	12	0.0001	0.0004	-0.440	7.23	4.19	-0.011	0.0005	0.0002	0.439	2.25	2.25	0.003	
1997	12	0.0001	0.0001	-0.718	8.72	5.69	-0.012	0.0010	0.0007	0.319	2.51	2.51	0.004	
1998	12	-0.0001	-0.0001	-0.666	10.45	7.42	-0.020	0.0006	0.0009	0.422	4.78	4.78	0.009	
1999	12	0.0017	0.0005	-0.358	5.62	2.59	-0.021	0.0016	0.0008	0.282	1.74	1.74	0.008	
2000	12	-0.0002	0.0010	-0.474	6.98	3.95	-0.017	0.0009	0.0006	0.430	3.01	3.01	0.006	
2001	12	-0.0006	-0.0004	-0.544	7.67	4.64	-0.018	0.0003	0.0004	0.327	3.55	3.55	0.005	
2002	12	-0.0012	-0.0007	-0.521	8.18	5.15	-0.021	0.0005	0.0004	0.657	5.29	5.29	0.003	
2003	12	0.0010	-0.0005	-0.119	5.74	2.71	-0.023	0.0005	0.0003	0.256	1.59	1.59	0.005	
2004	12	0.0005	0.0007	-0.140	6.17	3.14	-0.016	0.0004	0.0005	0.404	2.87	2.87	0.003	
2005	12	0.0011	0.0008	-0.285	7.88	4.85	-0.013	0.0005	0.0002	0.535	4.58	4.58	0.003	
2006	12	0.0004	0.0009	-0.575	8.28	5.24	-0.017	0.0007	0.0002	0.492	2.37	2.37	0.003	
2007	12	0.0001	0.0005	-0.528	7.24	4.21	-0.018	0.0008	0.0005	0.402	1.66	1.66	0.003	
2008	12	-0.0018	-0.0007	-0.112	5.81	2.78	-0.026	0.0007	0.0006	0.334	1.00	1.00	0.002	
2009	12	0.0012	-0.0009	-0.146	6.39	3.36	-0.039	0.0006	0.0003	0.427	1.65	1.65	0.005	

Panel notes: The variable means are yearly averages from the Control group firms. The Control group consists of a maximum of 12 observations per year from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.

PANEL C The Primary Sample Market Price Return Distribution Descriptive Statistics and Variance-Covariance Value-at-Risk using 300 historical observations based on 1-day returns

Table P.1 (Continued)

			Mean							Standard Deviation						
Year	Obs.	Return	Dist.	Skew	Kurtosis	Excess Kurtosis	V	_	Return	Dist.	Skew	Kurtosis	Excess Kurtosis	V		
1993	4	0.0031	0.0020	-0.852	17.65	14.62	-0.023		0.0021	0.0012	2.514	24.27	24.27	0.013		
1994	4	-0.0005	0.0009	-1.078	16.43	13.40	-0.022		0.0005	0.0006	2.422	24.86	24.86	0.012		
1995	5	0.0012	0.0004	-1.112	15.32	12.29	-0.021		0.0010	0.0004	2.351	25.49	25.49	0.012		
1996	5	0.0008	0.0010	-0.069	6.40	3.37	-0.022		0.0005	0.0008	0.557	5.28	5.28	0.004		
1997	5	0.0011	0.0013	-0.021	4.78	1.75	-0.025		0.0011	0.0006	0.130	0.75	0.75	0.002		
1998	5	0.0001	0.0005	0.134	4.82	1.79	-0.039		0.0006	0.0010	0.201	0.99	0.99	0.006		
1999	5	0.0009	0.0008	0.179	4.20	1.17	-0.047		0.0008	0.0005	0.157	0.43	0.43	0.005		
2000	5	0.0005	0.0000	0.185	4.44	1.41	-0.043		0.0008	0.0006	0.188	1.15	1.15	0.006		
2001	5	-0.0001	0.0005	-0.078	5.21	2.18	-0.039		0.0005	0.0006	0.239	1.02	1.02	0.003		
2002	5	-0.0010	-0.0003	-0.319	6.47	3.44	-0.038		0.0007	0.0003	0.257	2.47	2.47	0.004		
2003	5	0.0007	-0.0005	0.320	4.35	1.32	-0.039		0.0004	0.0007	0.224	0.52	0.52	0.009		
2004	5	0.0003	0.0004	0.249	4.98	1.95	-0.023		0.0002	0.0003	0.214	1.64	1.64	0.003		
2005	5	0.0003	0.0003	-0.207	6.15	3.12	-0.017		0.0004	0.0002	0.464	1.92	1.92	0.003		
2006	5	0.0005	0.0005	0.004	5.41	2.38	-0.018		0.0003	0.0003	0.333	1.52	1.52	0.004		
2007	5	-0.0007	0.0001	0.026	5.08	2.05	-0.020		0.0009	0.0003	0.081	0.60	0.60	0.004		
2008	5	-0.0041	-0.0018	0.209	6.83	3.80	-0.045		0.0026	0.0014	0.234	1.81	1.81	0.011		
2009	5	0.0011	-0.0030	-0.833	19.64	16.61	-0.108		0.0018	0.0026	2.358	20.80	20.80	0.038		

Panel notes: The variable means are sample averages. The sample consists of a maximum of 5 observations per year from the 5 banks in the LSE that adopted the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.

PANEL D The Secondary Sample Market Price Return Distribution Descriptive Statistics and Variance-Covariance Value-at-Risk using 300 historical observations based on 1-day returns

Table P.1 (Continued)

Year Obs.			М	ean			Standard Deviation						
Year	Obs.	Return	Dist.	Skew	Kurtosis	Excess Kurtosis	V	Return	Dist.	Skew	Kurtosis	Excess Kurtosis	V
1993	11	0.0012	0 0021	1 662	15 22	12 19	-0.023	0.0016	0.0010	1 037	9 09	9 09	0.023
1994	11	0.0001	0.0012	0.705	13.88	10.85	-0.021	0.0009	0.0008	1.277	11.23	11.23	0.014
1995	11	0.0007	0.0002	0.049	13.51	10.48	-0.017	0.0008	0.0006	0.749	10.50	10.50	0.008
1996	11	0.0002	0.0006	0.438	14.30	11.27	-0.014	0.0004	0.0004	1.142	9.79	9.79	0.005
1997	11	0.0009	0.0006	0.336	15.48	12.45	-0.014	0.0007	0.0005	1.307	11.54	11.54	0.005
1998	11	0.0002	0.0009	0.173	13.11	10.08	-0.022	0.0006	0.0005	0.866	7.18	7.18	0.009
1999	11	0.0014	0.0005	0.918	19.21	16.18	-0.028	0.0013	0.0005	2.772	36.89	36.89	0.012
2000	11	0.0007	0.0009	0.528	12.27	9.24	-0.027	0.0010	0.0014	0.871	12.09	12.09	0.013
2001	11	-0.0009	-0.0001	-0.259	8.92	5.89	-0.028	0.0007	0.0006	0.608	5.80	5.80	0.015
2002	11	-0.0017	-0.0010	-0.308	8.80	5.77	-0.032	0.0017	0.0010	0.606	4.52	4.52	0.017
2003	11	0.0010	-0.0006	-0.210	6.93	3.90	-0.034	0.0007	0.0012	0.363	2.62	2.62	0.023
2004	11	0.0005	0.0007	0.050	6.73	3.70	-0.023	0.0004	0.0005	0.491	3.71	3.71	0.012
2005	11	0.0009	0.0007	0.065	7.25	4.22	-0.019	0.0007	0.0005	0.614	4.30	4.30	0.007
2006	11	0.0006	0.0010	0.036	7.07	4.04	-0.023	0.0004	0.0006	0.598	2.08	2.08	0.006
2007	11	-0.0006	0.0003	-0.116	7.08	4.05	-0.026	0.0019	0.0007	0.391	2.55	2.55	0.007
2008	11	-0.0017	-0.0013	-0.166	9.69	6.66	-0.042	0.0015	0.0021	0.954	8.48	8.48	0.025
2009	11	0.0012	-0.0006	0.175	7.52	4.49	-0.054	0.0012	0.0005	0.264	3.31	3.31	0.017

Panel notes: The variable means are sample averages. The sample consists of a maximum of 11 observations per year from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.

Analysis

The general inference, for the Primary and Secondary sample, the Control group, the Primary sample and the Secondary sample (collectively referred to as the samples and the control group), from analysing the Table P.1 results is that there are materially significant changes to both daily average returns and market price variance-covariance Value-at-Risk immediately after the 2005 accounting change. Table P.1 results, however, infer that the Primary and Secondary sample firms exhibited lower returns and higher Value-at-Risk levels after 2005 compared to the Control group.

Further analysis of these results reveal that the samples and the control group yearly average market price returns (calculated using daily returns), presented in the Return column, infer materially significant changes after 2005 for the time periods 2006 to 2008, and 2008 to 2009, when compared to similar time periods before 2005. Materially significant changes in returns for before 2005 are also inferred for the time period 1999 to 2002. When compared to before 2005, the results show that the sample market price return volatility level, measured using the standard deviation of sample means, increased for the Primary and Secondary sample after 2005. This compares with a decrease in the level of volatility for the Control group after 2005.

The average distribution characteristics for the 300 day average returns are presented in Table P.1 by the distribution descriptive statistics Dist. (Distribution), Skew, Kurtosis and Excess Kurtosis. It is expected that the accuracy for the variance-covariance Valueat-Risk model increase when the property of the distribution tends to the normal distribution. The normal distribution is specified by a mean of 0 and a standard deviation of 1. This distribution property is also characterised by a Skew measure of 0, and an Excess Kurtosis measure of 0. Examining Table P.1 the distribution mean presented in column Dist. does show near 0 levels for the samples and the Control group distributions. However, the Skew and Excess Kurtosis properties deviate for the optimal 0 level. This suggests that the variance-covariance Value-at-Risk results presented in Table P.1 would require further examination.

In addition to the variance-covariance Value-at-Risk model, in Appendix N.3, this study presents results for the historical and Monte Carlo simulation Value-at-Risk models.

P.2 Value-at-Risk Analysis

The Value-at-Risk and *Value-at-Risk backtest breach* results for the variancecovariance, historical and Monte Carlo Simulation Value-at-Risk models for 1-day, 250-day and 500-day time horizons at the 95% confidence level are presented on a yearly basis for the time range 1993 to 2009 in Table P.2. This table presents Value-at-Risk levels for the 1-day time horizon calculated using 300 1-day historical returns, the 250-day time horizon calculated using 300 historical 250-day returns and the 500-day time horizon calculated using 500 historical 500-day returns. Table P.2 presents descriptive statistics for these backtests on a yearly basis for the time range 1993 to 2009.

Table P.2 Panel A presents Value-at-Risk results for the Primary and Secondary Sample for the 1-day time horizon; Panel B presents Value-at-Risk results for the Control group for the 1-day time horizon; Panel C presents Value-at-Risk results for the Primary and Secondary Sample for the 250-day time horizon; Panel D presents Value-at-Risk results for the Control group for the 250-day time horizon; E presents Value-at-Risk results for the Primary and Secondary Sample for the 250-day time horizon; E presents Value-at-Risk results for the Primary and Secondary Sample for the 500-day time horizon; and, Panel F presents Value-at-Risk results for the Control group for the Control group for the 500-day time horizon.

Table P.2 Value-at-Risk Analysis for Market Price Returns

Table Description

The Table P.2 columns represent the following:

Year is the *panel data* year, represented by the number of days from 1st January to 31st December, and also represents the 31st December variable record date for each year. *Obs.* represents the number of sample firms observed for the year.

The Average Return column represents the average returns for the sample firms observed for year (Year). The average returns are based on the stated time horizon days and calculated using 300 days historical returns for Panel A to Panel D and 500 days historical returns for Panels E and F. The average returns are calculated using equation (M.35).

For the Variance-Covariance Average, Historical Average and Monte Carlo Simulation Average columns:

V 95% is the average Value-at-Risk for the stated time horizons at the 95% confidence level. The Value-at-Risk parameters are calculated for the stated time horizon using 300 days historical returns for Panel A to Panel D and 500 days historical returns for Panel E and Panel F. The variance-covariance Value-at-Risk is calculated using equation (N.5); the historical Value-at-Risk is calculated using equation Value-at-Risk is calculated using equation Value-at-Risk is calculated using equation (N.22). For the Value-at-Risk models the average Value-at-Risk for the sample firms observed (*Obs.*) for year (Year) is calculated by applying equation (M.35).

In Hori. Breach Count (*In-time Horizon Breach* Count) is the backtested count for the number of times the actual market price return variable breached the Value-at-Risk level within the number of time horizon days. The In Hori. Breach Count is measured using equation (N.32).

End Hori. Breach Count (*End-time Horizon Breach* Count)^a is the backtested binary code that is set to 1 if the actual market price return variable breached the Value-at-Risk level at the time horizon date. End Hori. Breach Count is calculated using equation (N.33). In Table P.2, the End Hori. Breach Count is calculated by determining if the average return from the Average Return column is greater than the average Value-at-Risk from the V95% column.

Table description notes:

^aFor a given year, the Value-at-Risk levels (reported in the V95% column) and the in-time horizon breach counts (reported in the In Hori. Breach Count column) may differ from expectation when comparing levels between the three models. This difference is due primarily to sample averaging and rounding effects.

Table	e P.2 ((Cont	inued)
	PAN	EL A	

The Primary and Secondary sample Value-at-Risk results for the 1-day time horizon Variance-Covariance, Historical and Monte Carlo Simulation models

			Variance	-Covariance	Average	His	torical Avera	ige	Monte Car	lo Simulation	n Average
Year	Obs.	Average Return	V 95%	In Hori. Breach Count ^a	End Hori. Breach Count ^b	V 95%	In Hori. Breach Count ^a	End Hori. Breach Count ^b	V 95%	In Hori. Breach Count ^a	End Hor. Breach Count ^b
1993	15	0.0019	-0.023	0		-0.0170	0		-0.0249	0	
1994	15	-0.0001	-0.021	0		-0.0183	0		-0.0232	0	
1995	16	0.0009	-0.0180	0		-0.0171	0		-0.0184	0	
1996	16	0.0004	-0.0162	0		-0.0134	0		-0.0172	0	
1997	16	0.0010	-0.0178	0		-0.0152	0		-0.0179	0	
1998	16	0.0002	-0.0270	0		-0.0240	0		-0.0277	0	
1999	16	0.0012	-0.0340	0		-0.0309	0		-0.0356	0	
2000	16	0.0006	-0.0319	0		-0.0289	0		-0.0339	0	
2001	16	-0.0006	-0.0317	0		-0.0305	0		-0.0338	0	
2002	16	-0.0015	-0.0336	0		-0.0333	0		-0.0364	0	
2003	16	0.0009	-0.0355	0		-0.0343	0		-0.0379	0	
2004	16	0.0004	-0.0228	0		-0.0206	0		-0.0234	0	
2005	16	0.0007	-0.0185	0		-0.0166	0		-0.0189	0	
2006	16	0.0006	-0.0212	0		-0.0191	0		-0.0214	0	
2007	16	-0.0006	-0.0239	0		-0.0226	0		-0.0250	0	
2008	16	-0.0024	-0.0432	0		-0.0381	0		-0.0481	0	
2009	16	0.0012	-0.0705	0		-0.0636	0		-0.0778	0	

Panel notes: ^{a,b} In Table P.2 yearly breaches are rounded to the nearest whole number.

The figures presented are sample averages. The sample consists of a maximum of 16 observations per year from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.

	Table P.2 (Continued)													
The C	PANEL B The Control group Value-at-Risk results for the 1-day time horizon Variance-Covariance, Historical and Monte Carlo Simulation models													
			Variance	-Covariance	Average	His	torical Avera	ige	Monte Carlo Simulation Average					
Year	Obs.	Average Return	V 95%	In Hori. Breach Count ^a	End Hori. Breach Count ^b	V 95%	In Hori. Breach Count ^a	End Hori. Breach Count ^b	V 95%	In Hori. Breach Count ^a	End Hori. Breach Count ^b			
1993	12	0.0014	-0.0162	0		-0.0138	0		-0.0139	0				
1994	12	-0.0005	-0.0166	0		-0.0158	0		-0.0152	0				
1995	12	0.0003	-0.0137	0		-0.0141	0		-0.0136	0				
1996	12	0.0001	-0.0108	0		-0.0101	0		-0.0101	0				
1997	12	0.0001	-0.0119	0		-0.0114	0		-0.0111	0				
1998	12	-0.0001	-0.0196	0		-0.0197	0		-0.0173	0				
1999	12	0.0017	-0.0212	0		-0.0218	0		-0.0190	0				
2000	12	-0.0002	-0.0172	0		-0.0156	0		-0.0153	0				
2001	12	-0.0006	-0.0180	0		-0.0179	0		-0.0173	0				
2002	12	-0.0012	-0.0211	0		-0.0207	0		-0.0213	0				
2003	12	0.0010	-0.0231	0		-0.0237	0		-0.0238	0				
2004	12	0.0005	-0.0164	0		-0.0153	0		-0.0156	0				
2005	12	0.0011	-0.0134	0		-0.0121	0		-0.0125	0				
2006	12	0.0004	-0.0167	0		-0.0151	0		-0.0162	0				
2007	12	0.0001	-0.0182	0		-0.0174	0		-0.0183	0				
2008	12	-0.0018	-0.0258	0		-0.0266	0		-0.0264	0				
2009	12	0.0012	-0.0391	0		-0.0394	0		-0.0399	0				

Panel notes: ^{a,b} In Table P.2 yearly breaches are rounded to the nearest whole number.

The figures presented are yearly averages from the Control group firms. The Control group consists of a maximum of 12 observations per year from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.

Table I . (Continueu)

PANEL C The Primary and Secondary sample Value-at-Risk results for the 250-day time horizon Variance-Covariance, Historical and Monte Carlo Simulation models

		Variance	-Covariance	Average	His	torical Avera	ige	Monte Carlo Simulation Average			
Year	Obs.	Average Return	V 95%	In Hori. Breach Count	End Hori. Breach Count	V 95%	In Hori. Breach Count	End Hori. Breach Count	V 95%	In Hori. Breach Count	End Hori. Breach Count
1993	-	-	-	-	-	-	-	-	-	-	-
1994	-	-	-	-	-	-	-	-	-	-	-
1995	15	0.169	-0.291	14		0.062	117		0.021	68	
1996	16	0.156	-0.271	1		-0.181	4		-0.225	5	
1997	16	0.228	-0.184	3		-0.030	45		-0.028	40	
1998	16	0.176	-0.185	7		0.019	27		0.001	30	
1999	16	0.157	-0.234	17		-0.003	88		-0.008	59	
2000	16	0.127	-0.304	20		-0.162	38		-0.224	20	
2001	16	-0.010	-0.317	21		-0.141	65		-0.208	31	
2002	16	-0.210	-0.350	57		-0.236	100		-0.295	67	
2003	16	-0.102	-0.249	119		-0.362	82		-0.405	56	
2004	16	0.144	-0.324	15		-0.517	3		-0.608	10	
2005	16	0.161	-0.342	0		-0.250	8		-0.283	11	
2006	16	0.203	-0.161	3		-0.025	9		-0.053	5	
2007	16	0.035	-0.147	8		0.064	87	1	0.043	55	1
2008	16	-0.414	-0.176	100	1	-0.046	159	1	-0.073	105	1
2009	16	-0.283	-0.313	151		-0.453	136		-0.535	99	

Panel notes: The figures presented are sample averages. The sample consists of a maximum of 16 observations per year from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.

	PANEL D												
The Co	ontrol grouj	p Value-at-Risk	results for th	te 250-day t	time horizon V	ariance-Cov	ariance, Hi	storical and N	Ionte Carlo S	Simulation	models		
			Variance	-Covariance	Average	His	torical Avera	ige	Monte Carlo Simulation Average				
Year	Obs.	Average Return	V 95%	In Hori. Breach Count	End Hori. Breach Count	V 95%	In Hori. Breach Count	End Hori. Breach Count	V 95%	In Hori. Breach Count	End Hori. Breach Count		
1993	-	-	-	-	-	-	-	-	-	-	-		
1994	-	-	-	-	-	-	-	-	-	-	-		
1995	12	0.047	-0.210	25		-0.004	162		-0.014	42			
1996	12	0.099	-0.196	2		-0.200	4		-0.208	12			
1997	12	0.030	-0.142	14		-0.077	38		-0.080	6			
1998	12	-0.016	-0.119	37		-0.059	47		-0.059	10			
1999	12	0.154	-0.156	48		-0.139	93		-0.077	18			
2000	12	0.207	-0.281	0		-0.232	2		-0.239	6			
2001	12	-0.119	-0.256	28		-0.007	109	1	-0.036	10	1		
2002	12	-0.182	-0.244	88		-0.186	127		-0.202	27			
2003	12	-0.095	-0.171	163		-0.323	100		-0.330	19			
2004	12	0.172	-0.276	19		-0.445	7		-0.526	9			
2005	12	0.202	-0.318	0		-0.213	4		-0.230	4			
2006	12	0.220	-0.158	0		0.003	7		-0.012	5			
2007	12	0.109	-0.150	3		0.098	90		0.083	12			
2008	12	-0.229	-0.151	76	1	0.016	129	1	-0.017	17	1		
2009	12	-0.156	-0.225	173		-0.258	151		-0.240	28			

Panel notes: The figures presented are yearly averages from the Control group firms. The Control group consists of a maximum of 12 observations per year from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.

Table P.2 (Continued)

Table I . (Continueu)

PANEL E The Primary and Secondary sample Value-at-Risk results for the 500-day time horizon Variance-Covariance, Historical and Monte Carlo Simulation models

		bs. Average Return	Variance-Covariance Average			Historical Average			Monte Carlo Simulation Average		
Year	Obs.		V 95%	In Hori. Breach Count	End Hori. Breach Count	V 95%	In Hori. Breach Count	End Hori. Breach Count	V 95%	In Hori. Breach Count	End Hori. Breach Count
1993	-	-	-	-	-	-	-	-	-	-	-
1994	-	-	-	-	-	-	-	-	-	-	-
1995	-	-	-	-	-	-	-	-	-	-	-
1996	-	-	-	-	-	-	-	-	-	-	-
1997	-	-	-	-	-	-	-	-	-	-	-
1998	15	0.378	-0.254	1		0.077	56		-0.001	133	
1999	15	0.331	-0.210	3		0.069	75		0.055	139	
2000	16	0.295	-0.293	13		0.105	128		0.100	140	
2001	16	0.117	-0.311	16		0.062	144		0.034	189	
2002	16	-0.238	-0.375	46		-0.077	151	1	-0.140	210	1
2003	16	-0.321	-0.374	160		-0.119	276	1	-0.164	134	1
2004	16	0.073	-0.481	126		-0.371	211		-0.478	131	
2005	16	0.296	-0.474	19		-0.657	5		-0.810	8	
2006	16	0.369	-0.418	3		-0.635	0		-0.726	41	
2007	16	0.234	-0.384	4		-0.267	10		-0.331	123	
2008	16	-0.395	-0.218	57	1	0.072	177	1	0.071	165	1
2009	16	-0.694	-0.168	250	1	0.140	405	1	0.144	144	1

Panel notes: The figures presented are sample averages. The sample consists of a maximum of 16 observations per year from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.

PANEL F											
The Control group Value-at-Risk results for the 500-day time horizon Variance-Covariance, Historical and Monte Carlo Simulation models											
			Variance-Covariance Average			Historical Average			Monte Carlo Simulation Average		
Year	Obs.	Average Return	V 95%	In Hori. Breach Count	End Hori. Breach Count	V 95%	In Hori. Breach Count	End Hori. Breach Count	V95%	In Hori. Breach Count	End Hori. Breach Count
1993	-	-	-	-	-	-	-	-	-	-	-
1994	-	-	-	-	-	-	-	-	-	-	-
1995	-	-	-	-	-	-	-	-	-	-	-
1996	-	-	-	-	-	-	-	-	-	-	-
1997	-	-	-	-	-	-	-	-	-	-	-
1998	12	0.002	-0.185	46		-0.070	64		-0.101	65	
1999	12	0.138	-0.154	87		-0.080	106		-0.065	72	
2000	12	0.379	-0.198	42		-0.111	97		-0.044	78	
2001	12	0.069	-0.250	5		-0.174	62		-0.139	171	
2002	12	-0.308	-0.340	84		-0.145	186	1	-0.126	356	1
2003	12	-0.284	-0.321	231		-0.048	397	1	-0.055	405	1
2004	12	0.106	-0.443	169		-0.368	196		-0.462	72	
2005	12	0.367	-0.345	21		-0.572	2		-0.687	0	
2006	12	0.430	-0.381	2		-0.554	0		-0.657	0	
2007	12	0.322	-0.373	1		-0.208	4		-0.285	31	
2008	12	-0.126	-0.206	36		0.150	161	1	0.143	322	1
2009	12	-0.383	-0.159	225	1	0.196	399	1	0.226	440	1

Table P.2 (Continued)

Panel notes: The figures presented are yearly averages from the Control group firms. The Control group consists of a maximum of 12 observations per year from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.

Analysis

From the analysis of Table P.2, it is evident that the three Value-at-Risk models variance-covariance, historical and Monte Carlo simulation at the 95% confidence level, produce similar results.

Panels A and B that present results for the 1-day time horizon do not record any breaches for the Primary and Secondary sample and the Control group.

Panels C and D that present results for the 250-day time horizon show that all three models register end-of-time horizon breaches for 2008. The historical and Monte Carlo simulation models also register end-of-time horizon breaches for the Primary and Secondary sample 2007 year, and for the Control group for the 2001 year.

Panels E and F that present results for the 500-day time horizon show that all three models register end-of-time horizon breaches for 2009. All three models also breach the Primary and Secondary sample 2008 year. The historical and Monte Carlo simulation models also register end-of-time horizon breaches for the Primary and Secondary sample and for the Control group for the years 2002, 2003, 2008 and 2009.

In summary, from the results presented in Table P.2, it is evident that all three models for the 250-day and 500-day time horizons report increasing frequency for end-of-time horizon Value-at-Risk breaches after the 2005 accounting change. The in-time horizon breach counts also show evidence of increase of Value-at-Risk breaches after 2005, with breach counts that are comparable to, or exceed those recorded before 2005.

P.3 Historical Value-at-Risk Actual for the 1-Day, 250-Day and 500-Day Time Horizon at the 95% Confidence Level

The *Historical Value-at-Risk Actual* descriptive statistics for the 1-day, 250-day and 500-day time horizons at the 95% confidence level are presented on a yearly basis for the time range 1993 to 2009 in Table P.3. This table presents the Historical Value-at-Risk Actual results for the 1-day time horizon calculated using 300 and 800 historical 1-day returns, the 250-day time horizon calculated using 300 and 800 historical 250-day returns and the 500-day time horizon calculated using 500 and 800 historical 500-day returns.

Table P.3 Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.

For the charts presented in Panels A, B, C and D respectively:

Figures P.1, P.7, P.13 and P.19 plot the Historical Value-at-Risk Actual for the 1-day time horizon calculated using 300 historical 1-day market price returns averaged for the respective sample and control group. The corresponding daily sample market price returns and standard deviation of sample mean returns are also plotted.

Figures P.2, P.8, P.14 and P.20 plot the Historical Value-at-Risk Actual for the 1-day time horizon calculated using 800 historical 1-day market price returns averaged for the respective sample and control group. The corresponding daily sample market price returns and standard deviation of sample mean returns are also plotted.

Figures P.3, P.9, P.15 and P.21 plot the Historical Value-at-Risk Actual for the 250-day time horizon calculated using 300 historical 250-day market price returns averaged for the respective sample and control group. The corresponding 250-day sample market price returns and standard deviation of sample mean returns are also plotted.

Figures P.4, P.10, P.16 and P.22 plot the Historical Value-at-Risk Actual for the 250-day time horizon calculated using 800 historical 250-day market price returns averaged for the respective sample and control group. The corresponding 250-day

sample market price returns and standard deviation of sample mean returns are also plotted.

Figures P.5, P.11, P.17 and P.23 plot the Historical Value-at-Risk Actual for the 500-day time horizon calculated using 500 historical 500-day market price returns averaged for the respective sample and control group. The corresponding 500-day market price returns and standard deviation of sample mean returns are also plotted.

Figures P.6, P.12, P.18 and P.24 plot the Historical Value-at-Risk Actual for the 500-day time horizon calculated using 800 historical 500-day market price returns averaged for the respective sample and control group. The corresponding 500-day sample market price returns and standard deviation of sample mean returns are also plotted.

Table P.3 Descriptive Statistics for Historical Value-at-Risk Actual

Table Description

The Table P.3 Panel A to Panel D columns represent the following:

The Statistic column represents the panels: Mean calculated using equation (O.1); and, Standard Deviation of sample means calculated using equation (O.2), for the sample firms observed (*Obs.*) for year (Year). Year is the *panel data* year represented by the number of days from 1st January to 31st December, and also represents the 31st December variable record date for each year.

For the Time Horizon (Days) column: 1, 250 and 500 represent the time horizon days columns; Historical Data (Days) represents the number of historical returns corresponding to the time horizon days; 300, 800 and 500 columns represent the number of historical returns applied to calculate Historical Value-at-Risk Actual. *Obs.* represents the number of sample firms observed for the year.

			Time Horizon (Days)								
	Year		1	2	50	50	500				
Statistic		Historical Data (Days)									
		300	800	300	800	500	800				
	1993	-0.017	-	-	-	-	-	15			
	1994	-0.018	-	0.073	-	-	-	15			
	1995	-0.017	-0.017	-0.176	-	-	-	16			
	1996	-0.013	-0.016	-0.038	-0.147	0.198	-	16			
	1997	-0.015	-0.015	0.005	-0.138	0.222	0.066	16			
	1998	-0.024	-0.020	0.005	-0.080	0.142	0.023	16			
	1999	-0.031	-0.025	-0.164	-0.145	-0.027	0.038	16			
	2000	-0.029	-0.029	-0.140	-0.211	-0.197	-0.080	16			
Mean	2001	-0.030	-0.030	-0.228	-0.279	-0.194	-0.114	16			
	2002	-0.033	-0.031	-0.354	-0.388	-0.172	-0.348	16			
	2003	-0.034	-0.033	-0.516	-0.510	-0.408	-0.623	16			
	2004	-0.021	-0.030	-0.270	-0.506	-0.418	-0.627	16			
	2005	-0.017	-0.024	-0.027	-0.466	-0.221	-0.619	16			
	2006	-0.019	-0.019	0.063	-0.177	-0.014	-0.330	16			
	2007	-0.023	-0.020	-0.034	-0.050	0.069	-0.004	16			
	2008	-0.038	-0.029	-0.458	-0.378	-0.426	-0.318	16			
	2009	-0.064	-0.043	-0.916	-0.822	-1.678	-0.970	16			
	1993	0.002	-	-	-	-	-				
	1994	0.002	-	0.051	-	-	-				
	1995	0.002	0.001	0.032	-	-	-				
	1996	0.002	4.9E-04	0.101	0.004	0.003	-				
	1997	0.002	0.001	0.026	0.002	0.049	0.006				
	1998	0.005	0.003	0.120	0.055	0.113	0.022				
	1999	0.003	0.001	0.041	0.019	0.023	0.035				
Standard	2000	0.002	0.001	0.093	0.041	0.096	0.058				
Deviations	2001	0.004	0.001	0.150	0.064	0.021	0.047				
	2002	0.004	0.002	0.072	0.037	0.078	0.140				
	2003	0.004	0.001	0.032	0.031	0.029	0.023				
	2004	0.003	0.002	0.196	0.007	0.015	0.002				
	2005	0.001	0.002	0.024	0.013	0.068	0.015				
	2006	0.002	0.001	0.032	0.165	0.086	0.144				
	2007	0.003	0.001	0.077	0.044	0.036	0.138				
	2008	0.007	0.004	0.157	0.133	0.292	0.177				
	2009	0.004	0.002	0.070	0.058	0.172	0.079				

The Primary and Secondary sample Descriptive Statistics for Historical Value-at-Risk

 Table P.3 (Continued)
 PANEL A

Panel notes: The variable means are sample averages. The sample consists of a maximum of 16 observations per year from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.

Figures P.1 to P.24 Plots of Historical Value-at-Risk Actual

Figure Description

The chart plots in Figures P.1 to P.24 represent the following:

The Market Price Return Average plot is the sample average return calculated on a daily basis for the specified time horizon using equation (M.35), where the returns are calculated on a daily basis for the specified time horizon using equation (N.1); the Market Price Return Standard Deviation plot is the standard deviation of the sample mean returns for the specified time horizon, and calculated on a daily basis by applying equation (M.36) with the maximum time increment set to 1 day.

The Value-at-Risk Average plot is the plot of the Historical Value-at-Risk Actual and the corresponding time horizon at the specified confidence level using the specified number of historical returns, and is calculated using equation (O.1) with the maximum time increment set to 1 day.



Figure P.1 The Primary and Secondary Sample 1-Day Historical Value-at-Risk Actual at the 95% Confidence Level with 300 Days Historical Data from September 1993 to November 2009



Figure P.2 The Primary and Secondary Sample 1-Day Historical Value-at-Risk Actual at the 95% Confidence Level with 800 Days Historical Data from August 1995 to November 2009


Figure P.3 The Primary and Secondary Sample 250-Day Historical Value-at-Risk Actual at the 95% Confidence Level with 300 Days Historical Data from August 1994 to November 2009



Figure P.4 The Primary and Secondary Sample 250-Day Historical Value-at-Risk Actual at the 95% Confidence Level with 800 Days Historical Data from July 1996 to November 2009



Figure P.5 The Primary and Secondary Sample 500-Day Historical Value-at-Risk Actual at the 95% Confidence Level with 500 Days Historical Data from May 1996 to November 2009



Figure P.6 The Primary and Secondary Sample 500-Day Historical Value-at-Risk Actual at the 95% Confidence Level with 800 Days Historical Data from July 1997 to November 2009

				Time Hori	zon (Days))		
			1	2:	50	50)0	
Statistic	Year			Historical	Data (Days	5)		Obs
		300	800	300	800	500	800	_
	1993	-0.014	-	-	-	-	-	12
	1994	-0.016	-	0.007	-	-	-	12
	1995	-0.014	-0.015	-0.196	-	-	-	12
	1996	-0.010	-0.014	-0.085	-0.167	-0.070	-	12
	1997	-0.011	-0.012	-0.057	-0.176	-0.078	-0.069	12
	1998	-0.020	-0.014	-0.132	-0.147	-0.105	-0.164	12
	1999	-0.022	-0.018	-0.237	-0.204	-0.174	-0.171	12
	2000	-0.016	-0.019	-0.010	-0.203	-0.157	-0.168	12
Mean	2001	-0.018	-0.019	-0.174	-0.241	-0.033	-0.192	12
	2002	-0.021	-0.018	-0.322	-0.295	-0.339	-0.324	12
	2003	-0.024	-0.021	-0.440	-0.429	-0.570	-0.554	12
	2004	-0.015	-0.020	-0.232	-0.427	-0.557	-0.560	12
	2005	-0.012	-0.017	0.001	-0.420	0.123	-0.273	12
	2000	-0.017	-0.014	0.022	-0.004	0.125	0.064	12
	2007	-0.027	-0.021	-0.241	-0 179	-0.109	-0.072	12
	2009	-0.039	-0.029	-0.549	-0.489	-0.554	-0.531	12
	1993	0.001	-	-	-	-	-	
	1994	0.001	-	0.052	-	-	-	
	1995	0.002	4.4E-04	0.020	-	-	-	
	1996	0.001	3.8E-04	0.080	0.004	0.001	-	
	1997	0.002	0.001	0.028	0.002	0.036	0.015	
	1998	0.004	0.002	0.078	0.055	0.057	0.028	
	1999	0.003	3.7E-04	0.045	0.019	0.004	0.016	
Standard	2000	0.001	0.001	0.115	0.041	0.053	0.008	
Deviations	2001	0.002	0.001	0.102	0.064	0.108	0.055	
	2002	0.003	0.002	0.040	0.037	0.125	0.132	
	2003	0.003	0.001	0.037	0.031	0.013	0.011	
	2004	0.002	0.001	0.205	0.00/	0.150	0.001	
	2005	0 001		111140	0013	0150		
	2005	0.001	0.001	0.030	0.165	0.130	0.165	
	2005 2006 2007	0.001 0.003 0.002	0.001	0.030	0.165	0.132	0.165	
	2005 2006 2007 2008	0.001 0.003 0.002 0.005	0.001 0.001 0.001 0.003	0.038	0.165 0.044 0.133	0.132 0.056 0.158	0.165 0.138 0.131	

PANEL B The Control Group Descriptive Statistics for Historical Value-at-Risk Actual for the 1-Day 250-Day and 500-Day Time Horizons at the 95% Confidence Level

Table P.3 (Continued)

Panel notes: The variable means are yearly averages from the Control group firms. The Control group consists of a maximum of 12 observations per year from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.



Figure P.7 The Control Group 1-Day Historical Value-at-Risk Actual at the 95% Confidence Level with 300 Days Historical Data from September 1993 to November 2009



Figure P.8 The Control Group 1-Day Historical Value-at-Risk Actual at the 95% Confidence Level with 800 Days Historical Data from August 1995 to November 2009



Figure P.9 The Control Group 250-Day Historical Value-at-Risk Actual at the 95% Confidence Level with 300 Days Historical Data from August 1994 to November 2009



Figure P.10 The Control Group 250-Day Historical Value-at-Risk Actual at the 95% Confidence Level with 800 Days Historical Data from July 1996 to November 2009



Figure P.11 The Control Group 500-Day Historical Value-at-Risk Actual at the 95% Confidence Level with 500 Days Historical Data from May 1996 to November 2009



Figure P.12 The Control Group 500-Day Historical Value-at-Risk Actual at the 95% Confidence Level with 800 Days Historical Data from July 1997 to November 2009

1-D	ay, 250-	Day and 5	00-Day T	ime Horizo	ns at the	95% Confid	ence Leve	el
				Time Hori	izon (Day	rs)		
			1	2:	50	50	0	
Statistic	Year			Historical	Data (Day	ys)		Obs.
		300	800	300	800	500	800	
	1993	-0.024	-		_		_	4
	1994	-0.025	-	0.047	-	-	-	4
	1995	-0.025	-0.025	-0.217	-	-	-	5
	1996	-0.021	-0.024	0.030	-0.131	0.198	-	5
	1997	-0.023	-0.023	0.077	-0.133	0.222	0.205	5
	1998	-0.037	-0.034	-0.057	-0.122	0.142	0.104	5
	1999	-0.044	-0.036	-0.319	-0.277	-0.027	0.023	5
	2000	-0.042	-0.042	-0.276	-0.358	-0.197	-0.171	5
Mean	2001	-0.038	-0.042	-0.236	-0.364	-0.194	-0.178	5
	2002	-0.037	-0.040	-0.203	-0.319	-0.172	-0.219	5
	2003	-0.037	-0.037	-0.385	-0.387	-0.408	-0.375	5
	2004	-0.021	-0.032	-0.214	-0.388	-0.418	-0.397	5
	2005	-0.015	-0.026	-0.029	-0.348	-0.221	-0.396	5
	2006	-0.016	-0.017	0.005	-0.140	-0.014	-0.228	5
	2007	-0.019	-0.017	-0.038	-0.039	0.069	-0.036	5
	2008	-0.042	-0.029	-0.512	-0.400	-0.426	-0.352	5
	2009	-0.096	-0.054	-1.507	-1.308	-1.678	-1.551	5
	1993	0.001	-	-	-	-	-	
	1994	0.002	-	0.073	-	-	-	
	1995	0.002	0.001	0.021	-	-	-	
	1996	0.001	3.2E-04	0.145	0.004	0.003	-	
	1997	0.004	0.001	0.031	0.002	0.049	0.014	
	1998	0.005	0.005	0.213	0.055	0.113	0.045	
	1999	0.003	0.001	0.082	0.019	0.023	0.010	
Ctore do ed	2000	0.002	0.001	0.081	0.041	0.096	0.052	
Standard	2001	0.004	0.001	0.115	0.064	0.021	0.009	
Deviations	2002	0.004	0.002	0.069	0.037	0.078	0.048	
	2003	0.004	0.001	0.039	0.031	0.029	0.021	
	2004	0.003	0.002	0.144	0.007	0.015	0.006	
	2005	0.001	0.002	0.017	0.013	0.068	0.007	
	2006	0.001	0.001	0.009	0.165	0.086	0.081	
	2007	0.004	0.002	0.070	0.044	0.036	0.082	
	2008	0.012	0.006	0.241	0.133	0.292	0.245	
	2009	0.008	0.005	0.151	0.058	0.172	0.171	

PANEL C The Primary sample Descriptive Statistics for Historical Value-at-Risk Actual for the 1-Day, 250-Day and 500-Day Time Horizons at the 95% Confidence Level

Table P.3 (Continued)

Panel notes: The variable means are sample averages. The sample consists of a maximum of 5 observations per year from the 5 banks in the LSE that adopted the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.



Figure P.13 The Primary Sample 1-Day Historical Value-at-Risk Actual at the 95% Confidence Level with 300 Days Historical Data from September 1993 to November 2009



Figure P.14 The Primary Sample 1-Day Historical Value-at-Risk Actual at the 95% Confidence Level with 800 Days Historical Data from August 1995 to November 2009



Figure P.15 The Primary Sample 250-Day Historical Value-at-Risk Actual at the 95% Confidence Level with 300 Days Historical Data from August 1994 to November 2009



Figure P.16 The Primary Sample 250-Day Historical Value-at-Risk Actual at the 95% Confidence Level with 800 Days Historical Data from July 1996 to November 2009



Figure P.17 The Primary Sample 500-Day Historical Value-at-Risk Actual at the 95% Confidence Level with 500 Days Historical Data from May 1996 to November 2009



Figure P.18 The Primary Sample 500-Day Historical Value-at-Risk Actual at the 95% Confidence Level with 800 Days Historical Data from July 1997 to November 2009

Time Horizon (Days) Statistic Year 1 250 500 Obs. 300 800 300 800 500 800 1993 -0.014 - - - - 11 11 1994 -0.016 - 0.084 - - 11 1995 -0.014 -0.012 - 1 11 1995 -0.014 -0.012 - 1 11 1997 -0.011 -0.012 -0.021 -0.140 0.007 0.011 1998 -0.018 -0.014 0.065 0.099 - 11 1997 -0.013 -0.0224 -0.240 -0.065 0.043 11 2000 -0.023 -0.024 -0.240 -0.065 -0.044 11 2001 -0.021 -0.024 -0.240 -0.065 -0.044 11 2002 -0.031 -0.024 -0.240 -0.045<	the 1-	-Day, 25	0-Day and	500-Day	Time Horiz	zons at the	e 95% Con	fidence Le	vel
Statistic Year 1 250 500 300 800 300 800 500 800 1993 -0.014 - - - - 11 1994 -0.014 - - - - 11 1995 -0.014 -0.062 - - 11 1996 -0.014 -0.012 -0.021 -0.140 0.007 0.011 1998 -0.014 -0.012 -0.021 -0.140 0.007 0.011 11 1999 -0.025 -0.019 -0.024 -0.085 0.105 0.003 11 2000 -0.025 -0.024 -0.240 -0.065 -0.084 11 2001 -0.025 -0.224 -0.240 -0.065 -0.084 11 2002 -0.031 -0.25 -0.224 -0.206 -0.731 11 2004 -0.020 -0.029 -0.560 -0.747 -0.731 11					Time Hor	izon (Days	5)		
Statistic Year Historical Data (Days) Obs. 300 800 300 800 500 800 1993 -0.014 - - - - 11 1995 -0.014 -0.062 - - 11 1996 -0.011 -0.012 -0.021 -0.140 0.007 - 11 1997 -0.011 -0.012 -0.021 -0.140 0.007 0.009 11 1998 -0.018 -0.019 -0.094 -0.065 0.095 -0.009 11 1999 -0.023 -0.022 -0.78 -0.143 -0.010 -0.033 11 2000 -0.023 -0.024 -0.240 -0.065 -0.084 11 2001 -0.027 -0.029 -0.295 -0.560 -0.747 -0.731 11 2004 -0.020 -0.029 -0.295 -0.560 -1.74 -0.302 11 2005 -0.024				1	2	50	5	00	
$ \frac{300 \ 800 \ 300 \ 800 \ 500 \ 800 \ }{500 \ 800 \ }} $	Statistic	Year			Historical	Data (Day	s)		Obs.
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $			300	800	300	800	500	800	
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		1993	-0.014	-	-	-	-	-	11
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		1994	-0.016	-	0.084	-	-	-	11
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		1995	-0.014	-0.014	-0.162	-	-	-	11
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		1996	-0.011	-0.014	-0.062	-0.153	0.029	-	11
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		1997	-0.011	-0.012	-0.021	-0.140	0.007	0.011	11
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		1998	-0.018	-0.014	0.033	-0.065	0.095	-0.009	11
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		1999	-0.025	-0.019	-0.094	-0.085	0.105	0.043	11
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		2000	-0.023	-0.022	-0.078	-0.143	-0.010	-0.039	11
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Mean	2001	-0.027	-0.025	-0.224	-0.240	-0.065	-0.084	11
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		2002	-0.031	-0.028	-0.423	-0.420	-0.411	-0.407	11
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		2003	-0.033	-0.031	-0.575	-0.566	-0.763	-0.735	11
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		2004	-0.020	-0.029	-0.295	-0.560	-0.747	-0.731	11
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		2005	-0.017	-0.024	-0.027	-0.519	-0.335	-0.720	11
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		2006	-0.020	-0.020	0.090	-0.193	0.074	-0.3//	11
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		2007	-0.024	-0.021	-0.032	-0.056	0.1/3	0.011	11
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		2008	-0.037	-0.029	-0.434	-0.369	-0.337	-0.302	11
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		2009	-0.049	-0.039	-0.048	-0.602	-0.740	-0.706	11
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		1993	0.002	-	-	-	-	-	
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		1994	0.002	-	0.042	-	-	-	
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		1995	0.003	0.001 5 5E 04	0.030	0.004	0.007	-	
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		1990	0.002	0.001	0.084	0.004	0.007	0.003	
$ \begin{array}{c} \mbox{Standard} \\ \mbox{Deviations} \end{array} \begin{array}{c} 1999 & 0.003 & 0.001 & 0.023 & 0.019 & 0.035 & 0.044 \\ 2000 & 0.002 & 0.001 & 0.099 & 0.041 & 0.068 & 0.060 \\ 2001 & 0.003 & 0.002 & 0.166 & 0.064 & 0.084 & 0.064 \\ 2002 & 0.004 & 0.002 & 0.073 & 0.037 & 0.223 & 0.182 \\ 2003 & 0.004 & 0.001 & 0.029 & 0.031 & 0.016 & 0.024 \\ 2004 & 0.002 & 0.002 & 0.219 & 0.007 & 0.036 & 4.1E-08 \\ \hline 2005 & 0.001 & 0.002 & 0.027 & 0.013 & 0.163 & 0.019 \\ \hline 2006 & 0.003 & 0.001 & 0.042 & 0.165 & 0.131 & 0.174 \\ 2007 & 0.002 & 0.001 & 0.080 & 0.044 & 0.084 & 0.164 \\ 2008 & 0.005 & 0.003 & 0.119 & 0.133 & 0.152 & 0.146 \\ 2009 & 0.002 & 0.001 & 0.032 & 0.058 & 0.033 & 0.037 \end{array}$		1997	0.001	0.001	0.023	0.002	0.020	0.003	
$ \begin{array}{c} \mbox{Standard} \\ \mbox{Deviations} \end{array} \begin{array}{c} 1777 & 0.003 & 0.001 & 0.023 & 0.017 & 0.033 & 0.044 \\ 2000 & 0.002 & 0.001 & 0.099 & 0.041 & 0.068 & 0.060 \\ 2001 & 0.003 & 0.002 & 0.166 & 0.064 & 0.084 & 0.064 \\ 2002 & 0.004 & 0.002 & 0.073 & 0.037 & 0.223 & 0.182 \\ 2003 & 0.004 & 0.001 & 0.029 & 0.031 & 0.016 & 0.024 \\ 2004 & 0.002 & 0.002 & 0.219 & 0.007 & 0.036 & 4.1E-08 \\ 2005 & 0.001 & 0.002 & 0.027 & 0.013 & 0.163 & 0.019 \\ 2006 & 0.003 & 0.001 & 0.042 & 0.165 & 0.131 & 0.174 \\ 2007 & 0.002 & 0.001 & 0.080 & 0.044 & 0.084 & 0.164 \\ 2008 & 0.005 & 0.003 & 0.119 & 0.133 & 0.152 & 0.146 \\ 2009 & 0.002 & 0.001 & 0.032 & 0.058 & 0.033 & 0.037 \end{array}$		1990	0.003	0.002	0.077	0.035	0.045	0.012	
Standard Deviations 2000 0.002 0.001 0.003 0.004 0.003 0.004 0.003 0.004 0.003 0.004 0.003 0.004 0.003 0.004 0.002 0.166 0.064 0.084 0.064 2002 0.004 0.002 0.073 0.037 0.223 0.182 2003 0.004 0.001 0.029 0.031 0.016 0.024 2004 0.002 0.002 0.219 0.007 0.036 4.1E-08 2005 0.001 0.002 0.027 0.013 0.163 0.019 2006 0.003 0.001 0.042 0.165 0.131 0.174 2007 0.002 0.001 0.080 0.044 0.084 0.164 2008 0.005 0.003 0.119 0.133 0.152 0.146 2009 0.002 0.001 0.032 0.058 0.033 0.037		2000	0.003	0.001	0.025	0.017	0.055	0.044	
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Standard	2000	0.002	0.001	0.166	0.041	0.000	0.000	
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Deviations	2001	0.005	0.002	0.073	0.004	0.223	0.182	
2005 0.004 0.001 0.025 0.051 0.016 0.024 2004 0.002 0.002 0.219 0.007 0.036 4.1E-08 2005 0.001 0.002 0.027 0.013 0.163 0.019 2006 0.003 0.001 0.042 0.165 0.131 0.174 2007 0.002 0.001 0.080 0.044 0.084 0.164 2008 0.005 0.003 0.119 0.133 0.152 0.146 2009 0.002 0.001 0.032 0.058 0.033 0.037		2002	0.004	0.002	0.079	0.037	0.225	0.182	
2001 0.002 0.027 0.013 0.163 0.019 2005 0.001 0.002 0.027 0.013 0.163 0.019 2006 0.003 0.001 0.042 0.165 0.131 0.174 2007 0.002 0.001 0.080 0.044 0.084 0.164 2008 0.005 0.003 0.119 0.133 0.152 0.146 2009 0.002 0.001 0.032 0.058 0.033 0.037		2003	0.007	0.001	0.219	0.007	0.036	4 1E-08	
2006 0.003 0.001 0.042 0.165 0.131 0.174 2007 0.002 0.001 0.080 0.044 0.084 0.164 2008 0.005 0.003 0.119 0.133 0.152 0.146 2009 0.002 0.001 0.032 0.058 0.033 0.037		2004	0.002	0.002	0.027	0.007	0.050	0.019	
2007 0.002 0.001 0.080 0.044 0.084 0.164 2008 0.005 0.003 0.119 0.133 0.152 0.146 2009 0.002 0.001 0.032 0.058 0.033 0.037		2006	0.003	0.001	0.042	0.165	0.131	0 174	
2008 0.005 0.003 0.119 0.133 0.152 0.146 2009 0.002 0.001 0.032 0.058 0.033 0.037		2007	0.002	0.001	0.080	0.044	0.084	0 164	
2009 0.002 0.001 0.032 0.058 0.033 0.037		2008	0.005	0.003	0.119	0.133	0.152	0.146	
		2009	0.002	0.001	0.032	0.058	0.033	0.037	

PANEL D The Secondary sample Descriptive Statistics for Historical Value-at-Risk Actual for the 1-Day 250-Day and 500-Day Time Horizons at the 95% Confidence Level

Table P.3 (Continued)

Panel notes: The variable means are sample averages. The sample consists of a maximum of 11 observations per year from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.



Figure P.19 The Secondary Sample 1-Day Historical Value-at-Risk Actual at the 95% Confidence Level with 300 Days Historical Data from September 1993 to November 2009



Figure P.20 The Secondary Sample 1-Day Historical Value-at-Risk Actual at the 95% Confidence Level with 800 Days Historical Data from August 1995 to November 2009



Figure P.21 The Secondary Sample 250-Day Historical Value-at-Risk Actual at the 95% Confidence Level with 300 Days Historical Data from August 1994 to November 2009



Figure P.22 The Secondary Sample 250-Day Historical Value-at-Risk Actual at the 95% Confidence Level with 800 Days Historical Data from July 1996 to November 2009



Figure P.23 The Secondary Sample 500-Day Historical Value-at-Risk Actual at the 95% Confidence Level with 500 Days Historical Data from May 1996 to November 2009



Figure P.24 The Secondary Sample 500-Day Historical Value-at-Risk Actual at the 95% Confidence Level with 800 Days Historical Data from July 1997 to November 2009

Analysis

Table P.3 and the corresponding Figures P.1 to P.24 present descriptive statistics and plots for the Historical Value-at-Risk Actual measure at the 95% confidence level. The specification for this historical Value-at-Risk model is presented in Appendix N.8. The time horizons 1-day, 250-day and 500-day tested for the samples and the control group with the 300 and 500 days historical returns, generally exhibit a lower level of Historical Value-at-Risk Actual for a shorter time period when compared to the 800 days historical returns. This signifies the expected result that for the 95% confidence level, the Historical Value-at-Risk Actual level at the 5% distribution tail changes to a new level more frequently the less the number of historical returns applied to construct its distribution. Conversely, the Value-at-Risk level changes to a new level less frequently the more the number of historical returns applied to construct its distribution.

Examination of Figures P.1 to P.24, it is evident that Value-at-Risk levels were breached for the Primary and Secondary sample and the Control group after 2005.

P.4 Historical Value-at-Risk Actual Confidence Level Analysis with the 95% Confidence Level

The Historical Value-at-Risk Actual regression analysis for the 250-day time horizon calculated using 300 historical 250-day returns at the 99% and 99.9%, and the 95% confidence levels are presented in Table P.4 and Table P.5. This analysis is presented for the time range 1994 to 2008. In the regressions, the Historical Value-at-Risk Actual at the 95% confidence level is tested as the dependent variable, and the Historical Value-at-Risk Actual at the 99% and 99.9% confidence levels are individually tested as the independent variable.

Table P.4Cross-Sectional Regression Analysis for Historical Value-at-Risk
Confidence Levels

Cross-Sectional Regression

Table P.4 Panel A to Panel D show the results for the cross-sectional regressions that test the Historical Value-at-Risk Actual at the 99% and the 95% confidence levels by applying the regression specified in equation (O.3). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.

Table P.4 Panels E to H show the results for the cross-sectional regressions that test the Historical Value-at-Risk Actual at the 99.9% and the 95% confidence levels by applying the regression specified in equation (O.4). In this table, Panel E presents results for the Primary and Secondary sample; Panel F presents results for the Control group; Panel G presents results for the Primary sample; and, Panel H presents results for the Secondary sample.

Cross-Sectional Regression Table Description

The Table P.4 columns represent the following:

Year is the *panel data* year represented by the number of days from 1st January to 31st December, and also represents the 31st December variable record date for each year.

The Regression column presents the coefficients and related statistics for the regressions specified in equations (O.3) and (O.4), and follows the respective models:

$$V_{95:i,t} = a + bV_{99:i,i} + e_{95:i,t}$$

$$V_{95:i,t} = a + bV_{99:9:i,i} + e_{95:i,t}$$

Where: $V_{95:i,t}$ is the Value-at-Risk at the 95% confidence level for the i^{th} firm at time t; $V_{99:i,t}$ and $V_{99.9:i,t}$ is the Value-at-Risk for the 99% and 99.9% confidence levels respectively for the i^{th} firm at time t, and $e_{95:i,t}$ is the regression error term. The *Lower* and *Upper* are the corresponding coefficient levels predicted at a 99% confidence level.

The Correlation column presents the correlation statistics for the relationship between the dependent and independent variables. The Descriptive Statistics column presents for the independent variable, the sample mean calculated using equation (O.1) and the sample standard deviation, SD, using equation (O.2). The Obs. column presents the number of sample firms observed for the year.

								Table P.4	(Contin	ued)								
	The Prime	ry and 9	Seconders	v Somnla	Cross S	actional Pa	grassia	PA n Analysi	NEL A s for His	torical V	alua at 1	Dielz A	etual at	the 00% o	nd 05% Co	nfidanca l	avals	
		i y anu k	secondary	y Sampro		Regre	ssion	11 7 Mary 51.	5 101 1115		aiuc-at-1			. ene <i>99</i> 70 a	iiu)570 C0	Deser	intivo	
Year			Slope					Intercept				Model		Corre	elation	Statis	stics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	1.094**	21.36	6.4E-11	0.94	1.25	0.053**	5.55	1.3E-04	0.02	0.08	0.97	0.04	12	0.987**	3.2E-11	-0.01	0.19	14
1995	0.911**	35.85	2.2E-14	0.83	0.99	0.019*	2.39	3.3E-02	-4.8E-03	0.04	0.99	0.02	13	0.995**	1.1E-14	-0.22	0.21	15
1996	0.992**	16.01	6.2E-10	0.80	1.18	0.043**	4.02	1.5E-03	0.01	0.08	0.95	0.04	13	0.976**	3.1E-10	-0.03	0.18	15
1997	0.99**	22.52	8.4E-12	0.86	1.12	0.028**	5.33	1.4E-04	0.01	0.04	0.98	0.02	13	0.987**	4.2E-12	-0.02	0.12	15
1998	0.945**	25.50	1.7E-12	0.83	1.06	0.066**	5.53	9.7E-05	0.03	0.10	0.98	0.04	13	0.99**	8.7E-13	-0.18	0.28	15
1999	0.987**	13.04	3.2E-09	0.76	1.21	0.048**	3.16	6.9E-03	2.9E-03	0.09	0.92	0.04	14	0.961**	1.6E-09	-0.16	0.13	16
2000	1.017**	30.21	3.8E-14	0.92	1.12	0.05**	4.29	7.5E-04	0.02	0.08	0.98	0.04	14	0.992**	1.9E-14	-0.18	0.30	16
2001	0.808**	13.90	1.4E-09	0.63	0.98	0.010	0.41	6.9E-01	-0.07	0.09	0.93	0.05	14	0.966**	7.0E-10	-0.38	0.22	16
2002	0.916**	14.98	5.2E-10	0.73	1.10	0.024	0.73	4.8E-01	-0.07	0.12	0.94	0.06	14	0.97**	2.6E-10	-0.48	0.26	16
2003	0.885**	149.8	7.7E-24	0.87	0.90	-0.003	-0.70	5.0E-01	-0.02	0.01	1.00	0.01	14	1**	3.8E-24	-0.58	0.53	16
2004	0.904**	30.59	3.2E-14	0.82	0.99	0.037**	4.43	5.7E-04	0.01	0.06	0.99	0.03	14	0.993**	1.6E-14	-0.14	0.25	16
2005	0.821**	14.68	6.8E-10	0.65	0.99	0.03**	5.51	7.7E-05	0.01	0.05	0.94	0.02	14	0.969**	3.4E-10	-0.02	0.10	16
2006	0.991**	26.60	2.2E-13	0.88	1.10	0.029**	6.01	3.2E-05	0.01	0.04	0.98	0.02	14	0.99**	1.1E-13	0.03	0.13	16
2007	0.791**	18.67	2.7E-11	0.66	0.92	0.021	1.73	1.1E-01	-0.01	0.06	0.96	0.04	14	0.981**	1.4E-11	-0.17	0.23	16
2008	0.949**	77.13	8.3E-20	0.91	0.99	0.065**	4.94	2.2E-04	0.03	0.10	1.00	0.03	14	0.999**	4.1E-20	-0.84	0.68	16

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 16 observations per yearly regression from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

								Table P.4	l (Contir	nued)								
	Tł	ne Conti	rol Group	Cross-S	Sectional	Regression	Analys	PA sis for His	NEL B torical V	′alue-at-F	Risk Act	ual at	the 99%	6 and 95%	Confidence	e Levels		
						Regre	ession									Descri	intive	
Year			Slope					Intercept				Model		Corre	elation	Statis	stics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	0.805**	7.91	2.4E-05	0.47	1.14	0.008	0.51	6.2E-01	-0.04	0.06	0.87	0.03	9	0.935**	1.2E-05	-0.13	0.09	11
1995	0.934**	31.35	2.6E-11	0.84	1.03	0.017*	2.36	4.0E-02	-0.01	0.04	0.99	0.01	10	0.995**	1.3E-11	-0.22	0.12	12
1996	0.954**	38.88	3.0E-12	0.88	1.03	0.017**	4.50	1.1E-03	4.9E-03	0.03	0.99	0.01	10	0.997**	1.5E-12	-0.05	0.15	12
1997	0.991**	40.74	1.9E-12	0.91	1.07	0.023**	4.95	5.8E-04	0.01	0.04	0.99	0.02	10	0.997**	9.5E-13	-0.07	0.19	12
1998	0.943**	41.75	1.5E-12	0.87	1.01	0.032**	3.83	3.3E-03	0.01	0.06	0.99	0.02	10	0.997**	7.5E-13	-0.29	0.23	12
1999	0.963**	66.66	1.4E-14	0.92	1.01	0.018**	3.81	3.4E-03	2.9E-03	0.03	1.00	0.01	10	0.999**	7.0E-15	-0.20	0.26	12
2000	0.987**	23.20	5.0E-10	0.85	1.12	0.037**	3.82	3.4E-03	0.01	0.07	0.98	0.03	10	0.991**	2.5E-10	-0.05	0.23	12
2001	1.016**	15.16	3.2E-08	0.80	1.23	0.076*	2.88	1.6E-02	-0.01	0.16	0.96	0.03	10	0.979**	1.6E-08	-0.37	0.13	12
2002	0.928**	11.65	3.9E-07	0.68	1.18	0.011	0.32	7.6E-01	-0.10	0.12	0.93	0.02	10	0.965**	1.9E-07	-0.42	0.08	12
2003	0.949**	12.81	1.6E-07	0.71	1.18	0.019	0.50	6.3E-01	-0.10	0.14	0.94	0.02	10	0.971**	7.9E-08	-0.50	0.09	12
2004	1.045**	18.40	4.8E-09	0.87	1.23	0.037**	8.31	8.4E-06	0.02	0.05	0.97	0.01	10	0.986**	2.4E-09	-0.04	0.07	12
2005	0.904**	27.55	9.2E-11	0.80	1.01	0.028**	10.1	1.4E-06	0.02	0.04	0.99	0.01	10	0.993**	4.6E-11	0.03	0.08	12
2006	0.988**	67.44	1.3E-14	0.94	1.03	0.02**	8.60	6.2E-06	0.01	0.03	1.00	0.01	10	0.999**	6.3E-15	0.03	0.16	12
2007	0.888**	17.41	8.3E-09	0.73	1.05	0.031**	3.19	9.7E-03	1.9E-04	0.06	0.97	0.03	10	0.984**	4.1E-09	-0.09	0.17	12
2008	0.927**	21.71	9.6E-10	0.79	1.06	0.042	1.71	1.2E-01	-0.04	0.12	0.98	0.03	10	0.99**	4.8E-10	-0.55	0.19	12

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The Control group consists of a maximum of 12 observations per yearly regression from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

								Table P.4	l (Conti	nued)								
	TL	. D		C	G	D	1.	PA	NEL C	17 - 1 4 ⁻	Diala A a	4	41 - 000)/	C C	. T		
	1 1	e Prima	iry Sampi	e Cross-	Sectional	Regression	ssion	sis for His	storical	value-at-	KISK AC	tual a	the 99%	70 anu 9570	Confidenc	e Leveis		
Year			Slope					Intercept				Model		Corre	elation	Statis	stics	Obs.
	b	t(b)	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	0.995**	11.18	7.9E-03	0.11	1.88	0.067	3.79	6.3E-02	-0.11	0.24	0.98	0.03	2	0.992**	4.0E-03	-0.09	0.21	4
1995	0.812**	19.78	2.5E-03	0.40	1.22	0.012	0.91	4.6E-01	-0.12	0.15	0.99	0.01	2	0.997**	1.3E-03	-0.29	0.18	4
1996	0.692	2.56	1.2E-01	-1.99	3.37	0.077	3.28	8.2E-02	-0.16	0.31	0.77	0.03	2	0.875	6.2E-02	0.06	0.07	4
1997	1.085**	50.03	4.0E-04	0.87	1.30	0.047**	16.1	3.8E-03	0.02	0.08	1.00	0.01	2	1**	2.0E-04	0.01	0.15	4
1998	1.055**	30.98	1.0E-03	0.72	1.39	0.142*	7.29	1.8E-02	-0.05	0.34	1.00	0.02	2	0.999**	5.2E-04	-0.49	0.35	4
1999	1.274*	3.77	3.3E-02	-0.70	3.25	0.124	1.79	1.7E-01	-0.28	0.53	0.83	0.06	3	0.909*	1.6E-02	-0.19	0.09	5
2000	0.919**	23.06	1.8E-04	0.69	1.15	0.017	1.05	3.7E-01	-0.08	0.11	0.99	0.02	3	0.997**	8.9E-05	-0.36	0.21	5
2001	0.971**	26.69	1.2E-04	0.76	1.18	0.056*	5.52	1.2E-02	-3.3E-03	0.12	1.00	0.01	3	0.998**	5.8E-05	-0.23	0.19	5
2002	0.969**	5.88	9.8E-03	0.01	1.93	0.039	0.61	5.8E-01	-0.33	0.41	0.92	0.03	3	0.959**	4.9E-03	-0.37	0.10	5
2003	0.926**	263.5	1.2E-07	0.91	0.95	0.009*	5.34	1.3E-02	-8.5E-04	0.02	1.00	2E-03	3	1**	6.0E-08	-0.42	0.26	5
2004	0.586**	6.83	6.4E-03	0.08	1.09	0.013	1.10	3.5E-01	-0.06	0.08	0.94	0.02	3	0.969**	3.2E-03	-0.10	0.11	5
2005	0.917*	4.53	2.0E-02	-0.26	2.10	0.024	2.82	6.7E-02	-0.03	0.07	0.87	0.01	3	0.934*	1.0E-02	-0.03	0.03	5
2006	0.963**	16.05	5.3E-04	0.61	1.31	0.021**	9.29	2.6E-03	0.01	0.03	0.99	0.01	3	0.994**	2.6E-04	-4.1E-03	0.04	5
2007	0.961**	20.71	2.5E-04	0.69	1.23	0.031	2.19	1.2E-01	-0.05	0.11	0.99	0.02	3	0.997**	1.2E-04	-0.27	0.17	5
2008	0.962**	30.38	7.8E-05	0.78	1.15	0.083	1.93	1.5E-01	-0.17	0.33	1.00	0.04	3	0.998**	3.9E-05	-1.21	0.67	5

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 5 observations per yearly regression from the 5 banks in the LSE that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

_								Table P.4	(Contin	nued)								
		~ · ·	~		a			PA	NEL D	.								
	The	Second	ary Samp	ole Cross	-Sectiona	I Regressio	on Anal	ysis for H	istorical	Value-at	-Risk A	ctual a	it the 99	% and 95%	6 Confiden	ce Levels		
Year			Slope			Regic		Intercept				Model		Corre	elation	Descri Statis	ptive stics	Obs.
	Ь	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	1.163**	21.71	2.1E-08	0.98	1.34	0.042**	4.38	2.3E-03	0.01	0.07	0.98	0.03	8	0.992**	1.1E-08	0.02	0.19	10
1995	0.944**	69.03	1.4E-13	0.90	0.99	0.017**	4.19	2.3E-03	0.00	0.03	1.00	0.01	9	0.999**	7.1E-14	-0.20	0.23	11
1996	0.983**	14.10	1.9E-07	0.76	1.21	0.036*	2.69	2.5E-02	-0.01	0.08	0.96	0.04	9	0.978**	9.6E-08	-0.06	0.19	11
1997	0.911**	25.96	9.0E-10	0.80	1.03	0.018**	4.57	1.3E-03	0.01	0.03	0.99	0.01	9	0.993**	4.5E-10	-0.03	0.11	11
1998	0.925**	11.99	7.8E-07	0.67	1.18	0.057**	4.76	1.0E-03	0.02	0.10	0.94	0.04	9	0.97**	3.9E-07	-0.07	0.15	11
1999	0.965**	22.70	3.0E-09	0.83	1.10	0.036**	4.17	2.4E-03	0.01	0.06	0.98	0.02	9	0.991**	1.5E-09	-0.14	0.15	11
2000	1.038**	22.38	3.4E-09	0.89	1.19	0.051**	3.49	6.8E-03	3.5E-03	0.10	0.98	0.05	9	0.991**	1.7E-09	-0.10	0.31	11
2001	0.729**	8.62	1.2E-05	0.45	1.00	-0.029	-0.70	5.0E-01	-0.16	0.11	0.89	0.06	9	0.944**	6.1E-06	-0.45	0.21	11
2002	0.918**	11.69	9.6E-07	0.66	1.17	0.028	0.58	5.7E-01	-0.13	0.18	0.94	0.07	9	0.969**	4.8E-07	-0.53	0.30	11
2003	0.883**	138.0	2.8E-16	0.86	0.90	-0.002	-0.28	7.8E-01	-0.02	0.02	1.00	0.01	9	1**	1.4E-16	-0.65	0.61	11
2004	0.919**	37.49	3.4E-11	0.84	1.00	0.036**	4.45	1.6E-03	0.01	0.06	0.99	0.02	9	0.997**	1.7E-11	-0.16	0.30	11
2005	0.813**	12.53	5.3E-07	0.60	1.02	0.034**	4.60	1.3E-03	0.01	0.06	0.95	0.02	9	0.972**	2.7E-07	-0.01	0.12	11
2006	0.982**	21.65	4.5E-09	0.83	1.13	0.033**	4.72	1.1E-03	0.01	0.06	0.98	0.02	9	0.991**	2.3E-09	0.05	0.15	11
2007	0.717**	26.76	6.9E-10	0.63	0.80	0.027**	3.81	4.2E-03	4.0E-03	0.05	0.99	0.02	9	0.994**	3.4E-10	-0.12	0.25	11
2008	0.945**	61.05	4.3E-13	0.89	0.99	0.061**	4.35	1.9E-03	0.02	0.11	1.00	0.03	9	0.999**	2.1E-13	-0.67	0.64	11

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 11 observations per yearly regression from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

475

							,	Table P.4	(Contir	nued)								
]	The Primar	y and S	econdary	Sample	Cross-Se	ectional Reg	ression .	PA Analysis	NEL E for Hist	orical Val	lue-at-R	isk Ac	ctual at 1	the 99.9% a	nd 95% Co	onfidence	Levels	
						Regre	ssion									Descri	ntive	
Year			Slope					Intercept				Model		Corre	lation	Statis	tics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	1.098**	19.41	2.0E-10	0.93	1.27	0.072*	6.78	2.1E-02	0.04	0.10	0.97	0.04	12	0.984**	9.9E-11	-0.03	0.19	14
1995	0.88**	24.69	2.6E-12	0.77	0.99	0.031**	2.68	2.0E-03	0.00	0.07	0.98	0.03	13	0.99**	1.3E-12	-0.25	0.22	15
1996	0.957**	12.65	1.1E-08	0.73	1.19	0.056	4.12	5.4E-01	0.02	0.10	0.92	0.05	13	0.962**	5.5E-09	-0.04	0.18	15
1997	0.961**	14.88	1.5E-09	0.77	1.15	0.039	5.00	7.8E-01	0.02	0.06	0.94	0.03	13	0.972**	7.6E-10	-0.03	0.12	15
1998	0.891**	18.03	1.4E-10	0.74	1.04	0.095	5.34	4.5E-01	0.04	0.15	0.96	0.05	13	0.981**	7.0E-11	-0.22	0.29	15
1999	0.893**	8.68	5.2E-07	0.59	1.20	0.06*	2.54	3.0E-02	-0.01	0.13	0.84	0.06	14	0.918**	2.6E-07	-0.19	0.14	16
2000	1.017**	19.23	1.8E-11	0.86	1.17	0.074	3.96	4.5E-01	0.02	0.13	0.96	0.06	14	0.982**	9.1E-12	-0.21	0.30	16
2001	0.786**	9.23	2.5E-07	0.53	1.04	0.047	1.14	6.3E-01	-0.08	0.17	0.86	0.07	14	0.927**	1.3E-07	-0.44	0.22	16
2002	0.731**	13.19	2.7E-09	0.57	0.90	-0.028	-0.82	2.2E-01	-0.13	0.07	0.93	0.07	14	0.962**	1.4E-09	-0.53	0.32	16
2003	0.807**	61.87	1.8E-18	0.77	0.85	-0.010	-0.92	7.8E-01	-0.04	0.02	1.00	0.03	14	0.998**	8.9E-19	-0.63	0.58	16
2004	0.862**	30.00	4.2E-14	0.78	0.95	0.048*	5.52	2.4E-02	0.02	0.07	0.98	0.03	14	0.992**	2.1E-14	-0.16	0.27	16
2005	0.799**	15.47	3.4E-10	0.65	0.95	0.044*	8.18	3.3E-02	0.03	0.06	0.94	0.02	14	0.972**	1.7E-10	-0.04	0.10	16
2006	1.03**	14.92	5.5E-10	0.82	1.24	0.049	5.92	7.7E-02	0.02	0.07	0.94	0.03	14	0.97**	2.7E-10	0.01	0.12	16
2007	0.764**	14.59	7.4E-10	0.61	0.92	0.033	2.10	5.2E-01	-0.01	0.08	0.94	0.05	14	0.969**	3.7E-10	-0.19	0.24	16
2008	0.92**	68.44	4.4E-19	0.88	0.96	0.092	6.10	3.7E-01	0.05	0.14	1.00	0.04	14	0.999**	2.2E-19	-0.89	0.70	16

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 16 observations per yearly regression from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

	Th	e Contr	ol Group	Cross-S	ectional F	Regression .	Analysi	PA is for Histo	NEL F orical Va	alue-at-R	isk Actu	al at t	he 99.9'	% and 95%	Confidenc	e Levels		
			1			Regre	ession									Descr	intive	
Year			Slope					Intercept				Model	l	Corre	elation	Stati	stics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	0.757**	6.74	8.4E-05	0.39	1.12	0.013	0.68	5.1E-01	-0.05	0.08	0.83	0.03	9	0.914**	4.2E-05	-0.15	0.10	11
1995	0.919**	20.33	1.8E-09	0.78	1.06	0.028*	2.35	4.1E-02	-0.01	0.07	0.98	0.02	10	0.988**	9.2E-10	-0.24	0.12	12
1996	0.949**	32.23	1.9E-11	0.86	1.04	0.026**	5.82	1.7E-04	0.01	0.04	0.99	0.01	10	0.995**	9.7E-12	-0.06	0.15	12
1997	0.971**	33.24	1.4E-11	0.88	1.06	0.03**	5.08	4.8E-04	0.01	0.05	0.99	0.02	10	0.996**	7.2E-12	-0.08	0.19	12
1998	0.937**	38.45	3.4E-12	0.86	1.01	0.055**	5.82	1.7E-04	0.02	0.08	0.99	0.02	10	0.997**	1.7E-12	-0.31	0.24	12
1999	0.951**	39.09	2.9E-12	0.87	1.03	0.031**	3.86	3.2E-03	0.01	0.06	0.99	0.02	10	0.997**	1.4E-12	-0.21	0.26	12
2000	0.952**	20.27	1.9E-09	0.80	1.10	0.054**	4.77	7.6E-04	0.02	0.09	0.98	0.04	10	0.988**	9.4E-10	-0.07	0.24	12
2001	0.956**	9.14	3.6E-06	0.62	1.29	0.102	2.21	5.1E-02	-0.04	0.25	0.89	0.05	10	0.945**	1.8E-06	-0.42	0.13	12
2002	0.735**	7.09	3.3E-05	0.41	1.06	-0.042	-0.86	4.1E-01	-0.20	0.11	0.83	0.03	10	0.913**	1.7E-05	-0.46	0.10	12
2003	0.849**	7.70	1.6E-05	0.50	1.20	-0.010	-0.17	8.7E-01	-0.20	0.18	0.86	0.03	10	0.925**	8.2E-06	-0.53	0.10	12
2004	1.058**	18.41	4.8E-09	0.88	1.24	0.047**	9.86	1.8E-06	0.03	0.06	0.97	0.01	10	0.986**	2.4E-09	-0.05	0.07	12
2005	0.878**	19.18	3.2E-09	0.73	1.02	0.04**	10.47	1.0E-06	0.03	0.05	0.97	0.01	10	0.987**	1.6E-09	0.02	0.08	12
2006	0.988**	43.62	9.6E-13	0.92	1.06	0.036**	10.00	1.6E-06	0.02	0.05	0.99	0.01	10	0.997**	4.8E-13	0.02	0.16	12
2007	0.831**	17.66	7.2E-09	0.68	0.98	0.046**	4.64	9.3E-04	0.01	0.08	0.97	0.03	10	0.984**	3.6E-09	-0.12	0.18	12
2008	0.818**	18.74	4.1E-09	0.68	0.96	0.021	0.75	4.7E-01	-0.07	0.11	0.97	0.03	10	0.986**	2.0E-09	-0.59	0.22	12

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The Control group consists of a maximum of 12 observations per yearly regression from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

Table P.4 (Continued)

								Table P.4	(Contir	nued)								
	ТІ	D		Cross	• • • • • • • • • • •	D	A	PA For History	NEL G				4h o 00 0	0/ and 050/	Confident	T arral-		
	1 ne	Primar	y sample	Cross-2	bectional.	Regression	ession	SIS IOF HIS	lorical v	alue-al-R	ASK ACU	ual at	the 99.9	% and 95%	o Connaeno	e Levels		
Year			Slope					Intercept				Model		Corre	elation	Descri Statis	ptive stics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	1.014**	12.62	6.2E-03	0.22	1.81	0.093*	5.56	3.1E-02	-0.07	0.26	0.99	0.03	2	0.994**	3.1E-03	-0.11	0.20	4
1995	0.733**	44.66	5.0E-04	0.57	0.90	0.012	2.02	1.8E-01	-0.05	0.07	1.00	0.01	2	0.999**	2.5E-04	-0.32	0.20	4
1996	0.545	1.98	1.9E-01	-2.18	3.27	0.099*	4.35	4.9E-02	-0.13	0.32	0.66	0.04	2	0.814	9.3E-02	0.04	0.09	4
1997	1.053**	12.07	6.8E-03	0.19	1.92	0.075*	6.24	2.5E-02	-0.04	0.19	0.99	0.02	2	0.993**	3.4E-03	-0.02	0.16	4
1998	1.029**	10.36	9.2E-03	0.04	2.01	0.198	3.14	8.8E-02	-0.43	0.82	0.98	0.06	2	0.991**	4.6E-03	-0.55	0.36	4
1999	1.188	1.95	1.5E-01	-2.37	4.74	0.163	1.09	3.5E-01	-0.71	1.03	0.56	0.10	3	0.748	7.3E-02	-0.23	0.08	5
2000	0.861**	15.63	5.7E-04	0.54	1.18	0.018	0.76	5.0E-01	-0.12	0.16	0.99	0.02	3	0.994**	2.8E-04	-0.38	0.22	5
2001	0.947**	7.61	4.7E-03	0.22	1.67	0.134	3.03	5.6E-02	-0.12	0.39	0.95	0.05	3	0.975**	2.4E-03	-0.31	0.19	5
2002	0.856*	4.31	2.3E-02	-0.30	2.02	0.020	0.24	8.3E-01	-0.46	0.50	0.86	0.04	3	0.928*	1.2E-02	-0.40	0.11	5
2003	0.922**	35.01	5.1E-05	0.77	1.08	0.059*	4.20	2.5E-02	-0.02	0.14	1.00	0.01	3	0.999**	2.6E-05	-0.48	0.26	5
2004	0.583**	6.46	7.5E-03	0.06	1.11	0.022	1.61	2.1E-01	-0.06	0.10	0.93	0.02	3	0.966**	3.8E-03	-0.12	0.11	5
2005	0.718*	3.50	3.9E-02	-0.48	1.91	0.031	2.48	9.0E-02	-0.04	0.10	0.80	0.01	3	0.896*	2.0E-02	-0.05	0.04	5
2006	0.902**	16.81	4.6E-04	0.59	1.22	0.031**	13.4	8.9E-04	0.02	0.04	0.99	5.E-03	3	0.995**	2.3E-04	-0.02	0.05	5
2007	0.965**	13.81	8.2E-04	0.56	1.37	0.044	1.96	1.4E-01	-0.09	0.17	0.98	0.02	3	0.992**	4.1E-04	-0.28	0.17	5
2008	0.91**	26.22	1.2E-04	0.71	1.11	0.083	1.67	1.9E-01	-0.21	0.37	1.00	0.05	3	0.998**	6.1E-05	-1.28	0.71	5

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 5 observations per yearly regression from the 5 banks in the LSE that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

								Table P.4	l (Contin	nued)								
								РА	NEL H			_				_		
	The	Seconda	ry Sampl	e Cross-	Sectional	Regression	n Analy	sis for Hi	storical	Value-at-	Risk Ac	tual a	t the 99.	9% and 95	% Confide	nce Levels		
						Regre	ession							G	.	Descri	iptive	
Year			Slope					Intercept				Model	l	Corre	elation	Statis	stics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	1.175**	19.87	4.3E-08	0.98	1.37	0.059**	5.63	4.9E-04	0.02	0.09	0.98	0.03	8	0.99**	2.1E-08	0.00	0.19	10
1995	0.932**	37.57	3.3E-11	0.85	1.01	0.032**	4.20	2.3E-03	0.01	0.06	0.99	0.02	9	0.997**	1.7E-11	-0.22	0.23	11
1996	0.95**	12.06	7.4E-07	0.69	1.21	0.045*	2.81	2.0E-02	-0.01	0.10	0.94	0.05	9	0.97**	3.7E-07	-0.07	0.20	11
1997	0.886**	19.23	1.3E-08	0.74	1.04	0.024**	4.54	1.4E-03	0.01	0.04	0.98	0.02	9	0.988**	6.4E-09	-0.04	0.11	11
1998	0.868**	8.95	8.9E-06	0.55	1.18	0.083**	4.81	9.6E-04	0.03	0.14	0.90	0.05	9	0.948**	4.5E-06	-0.10	0.15	11
1999	0.909**	20.83	6.4E-09	0.77	1.05	0.048**	4.89	8.6E-04	0.02	0.08	0.98	0.02	9	0.99**	3.2E-09	-0.17	0.16	11
2000	1.056**	14.37	1.6E-07	0.82	1.29	0.078**	3.36	8.4E-03	0.00	0.15	0.96	0.07	9	0.979**	8.2E-08	-0.13	0.30	11
2001	0.667**	6.32	1.4E-04	0.32	1.01	-0.028	-0.50	6.3E-01	-0.21	0.16	0.82	0.07	9	0.903**	6.9E-05	-0.49	0.22	11
2002	0.729**	10.46	2.5E-06	0.50	0.96	-0.028	-0.59	5.7E-01	-0.18	0.13	0.92	0.08	9	0.961**	1.2E-06	-0.59	0.37	11
2003	0.797**	65.47	2.3E-13	0.76	0.84	-0.023	-2.01	7.5E-02	-0.06	0.01	1.00	0.03	9	0.999**	1.1E-13	-0.70	0.68	11
2004	0.874**	32.26	1.3E-10	0.79	0.96	0.047**	4.96	7.8E-04	0.02	0.08	0.99	0.03	9	0.996**	6.5E-11	-0.18	0.32	11
2005	0.795**	13.68	2.5E-07	0.61	0.98	0.048**	6.94	6.7E-05	0.03	0.07	0.95	0.02	9	0.977**	1.3E-07	-0.03	0.12	11
2006	1.019**	12.30	6.2E-07	0.75	1.29	0.056**	4.81	9.5E-04	0.02	0.09	0.94	0.04	9	0.972**	3.1E-07	0.03	0.15	11
2007	0.685**	20.17	8.4E-09	0.57	0.80	0.041**	4.23	2.2E-03	0.01	0.07	0.98	0.03	9	0.989**	4.2E-09	-0.15	0.26	11
2008	0.928**	55.48	1.0E-12	0.87	0.98	0.096**	6.02	2.0E-04	0.04	0.15	1.00	0.03	9	0.999**	5.0E-13	-0.72	0.65	11

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 11 observations per yearly regression from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

Table P.5Time Series Regression Analysis for Historical Value-at-Risk
Confidence Levels

Time Series Regression

Table P.5 Panel A to Panel D show the results for the time series regressions that test the Historical Value-at-Risk Actual at the 99% and the 95% confidence levels by applying the regression specified in equation (O.5). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.

Table P.5 Panels E to H show the results for the time series regressions that test the Historical Value-at-Risk Actual at the 99.9% and the 95% confidence levels by applying the regression specified in equation (O.6). Table P.5 Panel E presents results for the Primary and Secondary sample; Panel F presents results for the Control group; Panel G presents results for the Primary sample; and, Panel H presents results for the Secondary sample.

Time Series Regression Table Description

The Table P.5 columns represent the following:

The Total Period column represents the time series regression results for the year range 1994 to 2008, with a single year represented by the number of days from 1st January to 31st December. The Sub-Periods column represents the time series regression results for the year range 1994 to 2004, and for 1994 to 2007, with a single year represented by the number of days from 1st January to 31st December.

The *Regression* panel presents the coefficients and related statistics for the regressions specified in equations (O.5) and (O.6), and follows the respective models:

 $V_{95:N,t} = a + bV_{99:N,i} + e_{95:N,t}$ $V_{95:N,t} = a + bV_{99:N,i} + e_{95:N,t}$

Where: $V_{95:N,t}$ is the Value-at-Risk at the 95% confidence level for N firms at time t, $V_{99:N,i}$ and $bV_{99:9,N,i}$ is the Value-at-Risk at the 99% and 99.9% confidence levels respectively for N firms at time t, and $e_{N,t}$ is the regression error term. The *Lower* and *Upper* are the corresponding coefficient levels predicted at a 99% confidence level.

The *Correlation* panel presents the correlation statistics for the relationship between the dependent and independent variables. The *Descriptive Statistics* panel presents for the variables tested, the average sample means calculated using equation (M.6), that applies the sample average for each time series panel year, calculated using equation (M.10); the standard deviation, *SD*, of sample means for the time series using equation (M.7); and, the standard error of means, *SE*, for the time series. The *Observations* panel presents: *Firms (n)* (term *N* in the regression) that represents the average number of sample firms in the time series regression (for the Primary and Secondary sample, for the time period 1994 to 2004, the number of firms are rounded down); *Years (T)* represents the number of time series years; *Total (n × T)* represents the number of averaged pooled time series regression observations.
Maaaaa		Total Period	Sub-Periods					
Measure		1994-2008	1994-2004	1994-2007				
Slope	b	0.926**	0.948**	0.938**				
	t (b)	67.24	40.64	48.85				
	p (b)	6.5E-18	1.6E-11	3.5E-15				
	SE(b)	0.014	0.023	0.019				
	Lower	0.885	0.872	0.879				
	Upper	0.968	1.024	0.997				
Intercept Regression	a t (a) p (a) SE(a)	0.038** 8.50 1.1E-06 0.005	0.043** 6.54 1.1E-04 0.007	0.04** 8.17 3.0E-06 0.005				
	Lower	0.025	0.022	0.025				
	Upper	0.052	0.064	0.055				
Model	R ²	0.997	0.995	0.995				
	s(e)	0.013	0.014	0.013				
	df(e)	13	9	12				
Correlation	r	0.999**	0.997**	0.997**				
	p(r)	3.2E-18	8.2E-12	1.8E-15				
Descriptive Statistics	V	-0.225	-0.217	-0.181				
	SD (V)	0.246	0.189	0.184				
	SE(V)	0.063	0.057	0.049				
	V 95	-0.170	-0.163	-0.130				
	SD (V 95)	0.228	0.179	0.173				
	SE (V 95) SE(V 95)	0.059	0.054	0.046				
Observations	Firms (n) ^a	16	15	16				
	Years (T)	15	11	14				
	Total ($n \times T$)	234	170	218				

PANEL A The Primary and Secondary Sample Time Series Regression Analysis for Historical

Value-at-Risk Actual at the 99% and 95% Confidence Levels

Table P.5 (Continued)

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

^aFirms(n) is the average number of firms surveyed for each year, rounded to the nearest whole number. In addition, Firms(n) reflects missing data items presented in Table Z.1. These same criteria applies to the presentation of the variable Firms(n) for all the time series results presented in this study.

The sample consists of a maximum of 234 observations from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. Observations are based on yearly averages calculated using daily levels from 1994 to 2008.

See main table notes for the regression model applied.

М		Total Period	Sub-F	Periods
Mea	asure	1994-2008	1994-2004	1994-2007
Slo	ne h	0.928**	0.945**	0.946**
	<i>t</i> (b)	56.65	42.70	55.88
	$p(\dot{b})$	6.0E-17	1.1E-11	7.1E-16
	SE(b)	0.016	0.022	0.017
	Lower	0.879	0.874	0.894
	Upper	0.977	1.017	0.998
Inter	cept a	0.024**	0.026**	0.026**
Regression	t (a)	5.57	4.37	6.57
-	p(a)	9.1E-05	1.8E-03	2.7E-05
	SE(a)	0.004	0.006	0.004
	Lower	0.011	0.007	0.014
	Upper	0.038	0.045	0.038
Mo	del R^2	0.996	0.995	0.996
	s(e)	0.012	0.011	0.010
	df(e)	13	9	12
	V	0 998**	0 998**	0 998**
Correlation	p(r)	3.0E-17	5.3E-12	3.6E-16
	V	-0.195	-0.214	-0.170
	SD(V)	0.190	0.163	0.170
Dogovinting Statis	SE(V)	0.049	0.049	0.045
Descriptive statis	V 95	-0.156	-0.177	-0.134
	SD (V 95)	0.177	0.155	0.161
	SE(V 95)	0.046	0.047	0.043
	Firms (n)	12	12	12
Observations	Years (T)	15	11	14

Table P.5 (Continued) PANEL B

The Control Group Time Series Regression Analysis for Historical Value-at-Risk Actual

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The control consists of a maximum of 179 observations from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. Observations are based on yearly averages calculated using daily levels from 1994 to 2008.

See main table notes for the regression model applied.

М		Total Period	Sub-F	Periods
Measu	ire	1994-2008	1994-2004	1994-2007
Slope	h	0.932**	0.969**	0.944**
stope	$\tilde{t}(b)$	52.35	26.72	27.80
	p (b)	1.7E-16	7.0E-10	2.9E-12
	SE(b)	0.018	0.036	0.034
	Lower	0.878	0.851	0.840
	Upper	0.986	1.086	1.048
Interce	pt a	0.042**	0.055**	0.044**
Regression	t (a)	5.88	5.38	4.99
	p (a)	5.4E-05	4.4E-04	3.1E-04
	SE(a)	0.007	0.010	0.009
	Lower	0.021	0.022	0.017
	Upper	0.064	0.088	0.071
Mode	R^2	0.995	0.988	0.985
	s(e)	0.021	0.021	0.021
	df(e)	13	9	12
	v	0 998**	0 994**	0 992**
Correlation	p(r)	8.3E-17	3.5E-10	1.5E-12
	V	-0.265	-0 224	-0 197
	, SD (V)	0.312	0.180	0.175
Description Statistic	$\widetilde{SE(V)}$	0.080	0.054	0.047
Descriptive statistic	s V 95	-0.205	-0.162	-0.142
	SD (V 95)	0.291	0.175	0.167
	SE(V 95)	0.075	0.053	0.045
	Firms (n)	5	5	5
Observations	Years (T)	15	11	14
		70	71	()

Table P.5 (Continued) PANEL C

The Primary Sample Time Series Regression Analysis for Historical Value-at-Risk Actual

Panel notes:

See main table notes for the regression model applied.

^{**, *} Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The sample consists of a maximum of 70 observations from the 5 banks in the LSE that adopted the IFRS standards in 2005. Observations are based on yearly averages calculated using daily levels from 1994 to 2008.

Ac	tual at the 99%	and 95% Confide	ence Levels	
Maaaaa		Total Period	Sub-F	Periods
Measure		1994-2008	1994-2004	1994-2007
Slope	b	0.923**	0.929**	0.93**
	t (b)	60.29	42.93	51.14
	p (b)	2.7E-17	1.0E-11	2.1E-15
	SE(b)	0.015	0.022	0.018
	Lower	0.877	0.859	0.875
	Upper	0.969	1.000	0.986
Intercept	а	0.037**	0.036**	0.038**
Regression	t (a)	7.71	5.55	7.63
-	p(a)	3.3E-06	3.6E-04	6.1E-06
	SE(a)	0.005	0.007	0.005
	Lower	0.023	0.015	0.023
	Upper	0.051	0.058	0.053
Model	R^2	0.996	0.995	0.995
	s(e)	0.014	0.015	0.014
	df(e)	13	9	12
	14	0.008**	0.008**	0.008**
Correlation	p(r)	1.3E-17	5.0E-12	1.0E-15
	V	-0 209	-0.216	-0.176
	, SD (V)	0.242	0.223	0.214
	SE(V)	0.063	0.067	0.057
Descriptive Statistics	V 05	0 156	0 165	0.126
	V 95 SD (V 05)	0 224	0.208	0 200
	SE(V 95)	0.058	0.063	0.053
	Firms (n)	11	11	11
Observations	Years (T)	15	11	14
	$T_{otal}(m \times T)$	164	120	153

Table P.5 (Continued) PANEL D

The Secondary Sample Time Series Regression Analysis for Historical Value-at-Risk

Panel notes:

^{**, *} Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The sample consists of a maximum of 164 observations from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. Observations are based on yearly averages calculated using daily levels from 1994 to 2008.

See main table notes for the regression model applied.

Regression coefficients and the related statistics are calculated using OLS regression.

Measure 1994-2008 1994-2004 1994-2004 Slope b 0.877** 0.881** 0.876** $f(b)$ 5.3E-16 7.3E-10 3.0E-13 SE(b) 0.018 0.033 0.026 Lower 0.932 0.989 0.956 Regression Intercept a 0.054** 0.056** 0.054** $p(a)$ 1.7E-06 4.7E-04 9.9E-06 0.007 0.010 0.007 $Model$ R^2 0.994 0.021 0.031 0.022 0.031 $Model$ R^2 0.994 0.022 0.031 0.007 $Model$ R^2 0.994 0.997** 0.990 0.995** $Descriptive Statistics$ Y -0.255 -0.248 -0.210 0.053 $P(r)$ 0.259 0.202 0.197 0.197 0.197 Descriptive Statistics Y 9.525 -0.170 -0.163 -0.130 $Descriptive Statistics$ Y	Maarree		Total Period	Sub-F	Periods
Slope b t (b) 47.836 26.591 33.649 3.0E-13 0.026 b.53E-16 7.3E-10 0.033 0.026 0.0018 0.033 0.026 0.026 0.0018 0.033 0.026 0.026 0.0018 0.033 0.026 0.026 0.0018 0.033 0.026 0.026 0.0018 0.033 0.026 0.026 0.0018 0.033 0.026 0.026 0.026 0.0018 0.033 0.026 0.026 0.0018 0.033 0.026 0.001 0.007 0.010 0.007 0.001 0.007 0.001 0.007 0.001 0.007 0.010 0.007 0.010 0.007 0.010 0.007 0.010 0.007 0.010 0.007 0.010 0.007 0.0118 0.021 0.0053 0.025 0.020 0.051 0.055 0.020 0.051 0.055 0.020 0.051 0.055 0.020 0.054 0.046	Measure		1994-2008	1994-2004	1994-2007
t(b) 47.836 26.591 33.649 $p(b)$ $5.3E-16$ $7.3E-10$ $3.0E-13$ $SE(b)$ 0.018 0.033 0.026 Lower 0.822 0.773 0.797 0.932 0.989 0.956 Regression Intercept $t(a)$ a 0.054^{**} 0.056^{**} 0.054^{**} $Intercept$ a 0.054^{**} 0.056^{**} 0.054^{**} 0.054^{**} $Intercept$ a 0.007 0.010 0.054^{**} 0.054^{**} $Intercept$ a 0.034 0.022 0.031 0.007^{**} $Intercept$ a 0.034 0.022 0.031 0.095^{**}	Slope	b	0.877**	0.881**	0.876**
$Regression \begin{tabular}{ c c c c c c c c c c c c c c c c c c c$		t (b)	47.836	26.591	33.649
SE(b) 0.018 0.033 0.026 Lower 0.932 0.773 0.797 0.989 0.956 Regression Intercept a 0.054** 0.056** 0.054** 0.056** 0.054** Regression Intercept a 0.054** 0.056** 0.054** 0.054** Lower 0.007 0.007 0.010 0.007 0.010 0.007 Model R^2 0.994 0.997 0.990 0.021 0.018 0.990 Model R^2 0.997** 0.021 0.994** 0.995** 0.112 Descriptive Statistics V 0.0259 0.202 0.197 0.153 0.197 0.153		p(b)	5.3E-16	7.3E-10	3.0E-13
Lower Upper 0.822 $0.9320.7730.9890.7970.956RegressionInterceptt (a)p (a)at (a)p (a)SE(a)0.054^{**}0.0070.056^{**}0.0100.054^{**}0.007LowerUpper0.0340.0070.0220.0100.0310.007ModelR^2s(e)df(e)0.9941.30.9870.0210.0180.9900.0210.018ModelR^2f(e)0.997^{**}2.7E-160.994^{**}3.6E-100.995^{**}1.5E-13Descriptive StatisticsVSD (V)SE(V)0.2550.2280.0590.2020.0540.995^{**}0.054ObservationsFirms (n)Years (T)161515111614$		SE(b)	0.018	0.033	0.026
Upper 0.932 0.989 0.956 Regression Intercept t(a) p(a) SE(a) 0.054** $1.7E-06SE(a) 0.056**1.7E-064.7E-040.010 0.054**9.9E-060.007 LowerUpper 0.0340.073 0.0220.0310.089 0.0310.076 Model R^2s(e)df(e) 0.99413 0.9879 0.9900.018$ Correlation $rp(r)$ 0.997** 2.7E-16 0.994** 3.6E-10 0.995** 1.5E-13 Descriptive Statistics $VSD(V)SE(V)$ -0.255 0.228 0.170 -0.248 0.019 -0.210 0.053 $VSE(V 95)$ -0.170 0.059 -0.163 0.054 -0.130 0.046 Observations Firms (n) Years (T) 16 15 16 11 14		Lower	0.822	0.773	0.797
Intercept r (a) a r (a) 0.054** r (a) 0.056** r (a) 0.056** r (a) 0.054** r (a) 0.054** r (a) 0.054** r (a) 0.054** r (a) 0.054** r (a) 0.056** r (a) 0.056** r (a) 0.056** r (a) 0.056** r (a) 0.056** r (a) 0.056** r (a) 0.054** r (a) 0.056** r (a) 0.056** $r (a)$ 0.057 0.010 0.007 Model R^2 $s(e)$ 0.091 0.022 0.031 0.007 Model R^2 s(e) 0.994 0.987 0.990 0.018 0.021 0.995** Correlation r p(r) $2.7E-16$ $0.994**$ $0.995**$ $0.5E-10$ $1.5E-13$ Descriptive Statistics V $SD(V)$ 0.0259 0.202 0.197 0.173 Descriptive Statistics Firms (n) Years (T) 16 15 16 Observations Firms (n) Years (T) 16 15		Upper	0.932	0.989	0.956
Regression t (a) 8.186 5.341 7.268 $p(a)$ 1.7E-06 4.7E-04 9.9E-06 $SE(a)$ 0.007 0.010 0.007 Lower 0.034 0.022 0.031 $Upper$ 0.073 0.089 0.076 Model R^2 0.994 0.987 0.990 $s(e)$ 0.018 0.021 0.018 $df(e)$ 13 9 12 Correlation r 0.997** 0.994** 0.995** $p(r)$ 2.7E-16 3.6E-10 1.5E-13 Descriptive Statistics V -0.255 -0.248 -0.210 $V = 5$ 0.067 0.061 0.053 $V = 5$ -0.170 -0.163 -0.130 $V = 5$ 0.228 0.179 0.173 $Observations$ Firms (n) 16 15 16 $Pears (T)$ 15 11 14	Intercept	а	0.054**	0.056**	0.054**
$\frac{p'(a)}{SE(a)} = \frac{1.7E-06}{0.007} = \frac{4.7E-04}{0.010} = \frac{9.9E-06}{0.007}$ $\frac{Lower}{Upper} = \frac{0.034}{0.073} = \frac{0.022}{0.089} = \frac{0.031}{0.076}$ $\frac{Model}{R^2} = \frac{R^2}{0.994} = \frac{0.994}{0.018} = \frac{0.987}{0.021} = \frac{0.990}{0.018}$ $\frac{r}{p(r)} = \frac{0.997^{**}}{2.7E-16} = \frac{0.994^{**}}{3.6E-10} = \frac{0.995^{**}}{1.5E-13}$ $\frac{V}{SD(V)} = \frac{0.255}{0.229} = \frac{0.248}{0.202} = \frac{0.210}{0.197}$ $\frac{V}{SE(V)} = \frac{0.255}{0.067} = \frac{0.248}{0.061} = \frac{0.210}{0.053}$ $\frac{V}{SD(V)} = \frac{0.067}{0.067} = \frac{0.061}{0.053} = \frac{0.130}{0.054}$ $\frac{V}{Descriptions} = \frac{Firms(n)}{Years(T)} = \frac{16}{15} = \frac{16}{14}$	Regression	t (a)	8.186	5.341	7.268
$\frac{SE(a)}{Upper} = 0.007 = 0.010 = 0.007$ $Lower Upper = 0.034 = 0.022 = 0.031$ $Model = \begin{array}{c} R^2 & 0.994 & 0.987 & 0.990 \\ s(e) & 0.018 & 0.021 & 0.018 \\ df(e) & 13 & 9 & 12 \end{array}$ $\frac{Correlation}{Descriptive Statistics} = \begin{array}{c} r & 0.997^{**} & 0.997^{**} \\ SD(V) & 0.255 & -0.248 & -0.210 \\ SE(V) & 0.067 & 0.061 & 0.053 \\ V & 95 & -0.170 & -0.163 & -0.130 \\ SD(V & 95) & 0.228 & 0.179 & 0.173 \\ SE(V & 95) & 0.228 & 0.179 & 0.173 \\ SE(V & 95) & 0.259 & 0.054 & 0.046 \\ \hline Observations & Firms(n) \\ Years(T) & 16 & 15 & 16 \\ 11 & 14 & 0 \\ \hline \end{array}$		p (a)	1.7E-06	4.7E-04	9.9E-06
$\frac{Lower}{Upper}$ $\frac{0.034}{0.073}$ $\frac{0.022}{0.089}$ $\frac{0.031}{0.076}$ $\frac{0.076}{0.089}$ $\frac{0.091}{0.076}$ $\frac{0.990}{0.021}$ $\frac{0.990}{0.018}$ $\frac{0.021}{0.018}$ $\frac{0.995}{12}$ $\frac{0.994^{**}}{1.2}$ $\frac{0.995^{**}}{2.7E-16}$ $\frac{0.994^{**}}{3.6E-10}$ $\frac{0.995^{**}}{1.5E-13}$ $\frac{V}{SD(V)}$ $\frac{-0.255}{0.259}$ $\frac{-0.248}{0.202}$ $\frac{-0.210}{0.197}$ $\frac{0.995^{**}}{0.061}$ $\frac{-0.210}{0.053}$ $\frac{V}{SD(V)}$ $\frac{V}{SD(V)}$ $\frac{-0.170}{0.067}$ $\frac{-0.163}{0.059}$ $\frac{-0.130}{0.173}$ $\frac{-0.130}{0.046}$ $\frac{-0.130}{0.046}$ $\frac{-0.130}{0.046}$ $\frac{-0.146}{14}$		SE(a)	0.007	0.010	0.007
Upper 0.073 0.089 0.076 Model R^2 $s(e)$ $df(e)$ 0.994 0.018 13 0.987 0.021 9 0.990 0.018 12 Correlation r $p(r)$ 0.997^{**} $2.7E-16$ 0.994^{**} $3.6E-10$ 0.995^{**} $1.5E-13$ Descriptive Statistics V $SD(V)$ $SE(V)$ -0.255 0.259 0.202 0.061 -0.210 0.053 Descriptive Statistics V $SD(V)$ $SE(V)$ -0.170 0.067 -0.163 0.051 -0.130 0.053 Descriptive Statistics $Firms(n)$ $Fears(T)$ 16 15 15 11 16 14		Lower	0.034	0.022	0.031
Model R^2 $s(e)$ $df(e)$ 0.994 0.018 0.987 0.021 9 0.990 0.018 12 Correlationr $p(r)$ 0.997** $2.7E-16$ 0.994** $3.6E-10$ 0.995** $1.5E-13$ Descriptive StatisticsV $SD(V)$ $SE(V)$ -0.255 0.259 0.067 -0.248 0.061 -0.210 0.053 Descriptive StatisticsV $SD(V)$ $SE(V)$ -0.170 0.067 -0.163 0.054 -0.130 0.173 0.054 Descriptive StatisticsV 95 $SD(V 95)$ $SE(V 95)$ -0.170 0.228 0.059 -0.163 0.054 -0.130 0.173 0.046 DiscriptionsFirms (n) Years (T)16 15 15 11 16 14		Upper	0.073	0.089	0.076
$s(e)$ $df(e)$ 0.018 13 0.021 9 0.018 12 Correlation r $p(r)$ 0.997^{**} $2.7E-16$ 0.994^{**} $3.6E-10$ 0.995^{**} $1.5E-13$ Descriptive Statistics V $SE(V)$ -0.255 0.259 0.202 -0.248 0.061 -0.210 0.053 Descriptive Statistics V $SD(V,95)$ $SE(V,95)$ -0.170 0.228 -0.163 0.179 0.054 -0.130 0.173 0.046 ObservationsFirms (n) Years (T) 16 15 15 11 16 14	Model	R^2	0.994	0.987	0.990
$df(e)$ 13912Correlationr p(r) 0.997^{**} $2.7E-160.994^{**}3.6E-100.995^{**}1.5E-13Descriptive StatisticsVSD (V)SE(V)-0.2550.2590.067-0.2480.061-0.2100.053Descriptive StatisticsVV 95SD (V 95)SE(V 95)-0.1700.2280.059-0.1630.054-0.1300.1730.046Descriptive StatisticsFirms (n)Years (T)1615151614$		s(e)	0.018	0.021	0.018
Correlationr p(r) 0.997^{**} $2.7E-160.994^{**}3.6E-100.995^{**}1.5E-13Descriptive StatisticsVSE(V)-0.2550.2590.067-0.2480.061-0.2100.053V 95SE(V)0.0670.0670.0610.0530.061V 95SE(V 95)-0.1700.2280.059-0.1300.054ObservationsFirms (n)Years (T)16151511$		df(e)	13	9	12
Correlation $p(r)$ 2.7E-16 3.6E-10 1.5E-13 $P(r)$ $P($		r	0.997**	0.994**	0.995**
V SD (V) SE(V) -0.255 0.259 -0.248 0.202 -0.210 0.197 Descriptive Statistics V $SE(V)$ 0.067 0.061 0.053 V SD (V SD (V SD (V $SE(V)$) -0.170 0.228 -0.163 0.179 -0.130 0.173 0.054 V $SE(V$ $SE(V)$ -0.163 0.059 -0.130 0.054 V $SE(V$ $SE(V$ $SE(V$ $SE(V)$ 16 15 15 11 16 14 V $SE(V$ $SE(V)$ 16 15 15 11 16 14	Correlation	p(r)	2.7E-16	3.6E-10	1.5E-13
SD (V) SE(V) 0.259 0.067 $0.2020.061$ $0.1970.053$ Descriptive Statistics $V 95SD (V 95) -0.1700.228$ $-0.1630.179$ $-0.1300.173$ $SD (V 95)SE(V 95) 0.059 0.054 -0.1300.046$ Observations Firms (n) Years (T) 16 15 1511 1614		V	-0.255	-0.248	-0.210
Descriptive Statistics $SE(V)$ 0.067 0.061 0.053 $V 95$ -0.170 -0.163 -0.130 $SD(V 95)$ 0.228 0.179 0.173 $SE(V 95)$ 0.059 0.054 0.046 Firms (n) $Vears(T)$ 15 11 14		SD (V)	0.259	0.202	0.197
Descriptive statistics $V 95$ -0.170 -0.163 -0.130 $SD (V 95)$ 0.228 0.179 0.173 $SE(V 95)$ 0.059 0.054 0.046 Firms (n) 16 15 16 $Observations$ $Firms (T)$ 15 11 14	Descripting Statistics	SE(V)	0.067	0.061	0.053
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Descriptive statistics	V 95	-0.170	-0.163	-0.130
SE(V 95) 0.059 0.054 0.046 Firms (n) 16 15 16 Observations Firms (T) 15 11 14		SD (V 95)	0.228	0.179	0.173
$\begin{array}{c c} Firms(n) & 16 & 15 & 16 \\ Years(T) & 15 & 11 & 14 \end{array}$		SE(V 95)	0.059	0.054	0.046
$\begin{array}{c} \text{Observations} \\ \text{Vears}(T) \\ \text{Vears}(T) \\ \text{15} \\ \text{11} \\ 14 \\ \text{16} \\ 14 \\ \text{16} \\ 14 \\ \text{16} \\ 14 \\ \text{16} \\ 14 \\ 14 \\ 14 \\ 14 \\ 14 \\ 14 \\ 14 \\ 1$		Firms (n)	16	15	16
Observations	01	Years (T)	15	11	14
$T_{a4al}(x, T) = 224 = 170 = 210$	Observations	$T_{adv} 1 (T)$	224	170	219

Table P.5 (Continued) PANEL E

Panel notes:

See main table notes for the regression model applied.

^{**, *} Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The sample consists of a maximum of 234 observations from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. Observations are based on yearly averages calculated using daily levels from 1994 to 2008.

24		Total Period	Sub-F	Periods
Measure		1994-2008	1994-2004	1994-2007
Slope	b	0.876**	0.885**	0.896**
	t (b)	36.98	26.19	33.92
	p (b)	1.5E-14	8.3E-10	2.7E-13
	SE(b)	0.024	0.034	0.026
	Lower	0.804	0.776	0.815
	Upper	0.947	0.995	0.977
Intercept	a	0.034**	0.032**	0.036**
Regression	t (a)	4.90	3.26	5.33
	p (a)	2.9E-04	9.8E-03	1.8E-04
	SE(a)	0.007	0.010	0.007
	Lower	0.013	1.3E-04	0.015
	Upper	0.055	0.063	0.057
Model	R^2	0.991	0.987	0.990
	s(e)	0.018	0.019	0.017
	df(e)	13	9	12
	14	0 005**	0.00/**	0 005**
Correlation	p(r)	7.4E-15	4.2E-10	1.4E-13
	V	-0.217	-0.235	-0 190
	, SD (V)	0.201	0.173	0.179
Description Statistics	$\widetilde{SE}(V)$	0.052	0.052	0.048
Descriptive statistics	V 95	-0.156	-0.177	-0.134
	SD (V 95)	0.177	0.155	0.161
	SE(V 95)	0.046	0.047	0.043
	Firms (n)	12	12	12
Observations	Years (T)	15	11	14
	\mathbf{T}_{a} (\mathbf{n}^{1} (\mathbf{n} \mathbf{T})	170	121	167

Table P.5 (Continued) PANEL F

The Control Group Time Series Regression Analysis for Historical Value-at-Risk Actual

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The control consists of a maximum of 179 observations from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. Observations are based on yearly averages calculated using daily levels from 1994 to 2008.

See main table notes for the regression model applied.

λ		Total Period	Sub-F	Periods
Me	asure	1994-2008	1994-2004	1994-2007
Slo	ppe b	0.889**	0.903**	0.874**
	t (b)	33.50	16.86	17.62
	p(b)	5.3E-14 0.027	4.1E-08 0.054	6.1E-10 0.050
	Lower	0.809	0.729	0.722
	Upper	0.969	1.077	1.025
Inter Regression	ccept a t (a) p (a) SE(a)	0.063** 5.42 1.2E-04 0.012	0.076** 4.42 0.002 0.017	0.060** 4.10 0.001 0.015
	Lower	0.028	0.020	0.015
	Upper	0.097	0.131	0.104
Мо	del R ²	0.989	0.969	0.963
	s(e)	0.032	0.032	0.033
	df(e)	13	9	12
Correlation	r	0.994**	0.985**	0.981**
	p(r)	2.6E-14	2.0E-08	3.0E-10
Descriptive Statis	V	-0.301	-0.263	-0.231
	SD (V)	0.326	0.191	0.187
	SE(V)	0.084	0.058	0.050
Descriptive Statis	V 95	-0.205	-0.162	-0.142
	SD (V 95)	0.291	0.175	0.167
	SE(V 95)	0.075	0.053	0.045
Observations	Firms (n)	5	5	5
	Years (T)	15	11	14
	Total ($n \times T$)	70	51	64

Table P.5 (Continued)	
PANEL G	

The Primary Sample Time Series Regression Analysis for Historical Value-at-Risk Actual

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The sample consists of a maximum of 70 observations from the 5 banks in the LSE that adopted the IFRS standards in 2005. Observations are based on yearly averages calculated using daily levels from 1994 to 2008.

See main table notes for the regression model applied.

	Act	ual at the 99.9%	and 95% Confid	ence Levels	
	Maaaaaa		Total Period	Sub-P	Periods
	Measure		1994-2008	1994-2004	1994-2007
	Slope	Ь	0.875**	0.874**	0.88**
	1	t (b)	44.45	30.85	36.90
		p(b)	1.4E-15	1.9E-10	1.0E-13
		SE(b)	0.020	0.028	0.024
		Lower	0.816	0.782	0.807
		Upper	0.935	0.966	0.953
	Intercent	a	0.052**	0.048**	0.052**
Regression	inter e epr	t(a)	7.67	5.13	7.35
0		n(a)	3.5E-06	6.2E-04	8.9E-06
		SE(a)	0.007	0.009	0.007
		Lower	0.018	0.030	
		Upper	0.072	0.079	0.074
	Model	R^2	0.993	0.991	0.991
		s(e)	0.019	0.021	0.019
		df(e)	13	9	12
		74	0.007**	0.005**	0.006**
Correle	ation	p(r)	6.9E-16	9.7E-11	5.0E-14
		V	-0 237	-0 244	-0.202
		, $SD(V)$	0.255	0.237	0.226
	G	SE(V)	0.066	0.071	0.060
Descriptive	Statistics	V 95	-0.156	-0.165	-0.126
		SD (V 95)	0.224	0.208	0.200
		SE(V 95)	0.058	0.063	0.053
		Einmag (m)	11		11
Ohservi	ations	Firms (n) Years (T)	15	11	14
		Total ($n \times T$)	164	120	153

Table P.5 (Continued) PANEL H

The Secondary Sample Time Series Regression Analysis for Historical Value-at-Risk

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The sample consists of a maximum of 164 observations from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. Observations are based on yearly averages calculated using daily levels from 1994 to 2008.

See main table notes for the regression model applied.

Analysis

For the samples and the control group Table P.4 and Table P.5 show that the cross-sectional and time series regressions for the Historical Value-at-Risk Actual (specified for the 250-day time horizon) at the 95% confidence level exhibit similar magnitude slopes with the 99% and the 99.9% confidence levels. Table P.4 and Table P.5 also exhibit the expected result that the Historical Value-at-Risk Actual at the 95% confidence level, vary in magnitude with greater statistical significance to the 99% than the 99.9% confidence level.

P.5 Market Price Returns and Historical Value-at-Risk Actual

The regression analysis for the market price returns and Historical Value-at-Risk Actual is presented for the time range 1994 to 2008 in Table P.6 and Table P.7. In the regressions, the Historical Value-at-Risk Actual is tested as the dependent variable, and the market price return variable is tested as the independent variable.

Table P.6 **Cross-Sectional Regression Analysis for Market Price Returns and Historical Value-at-Risk Actual**

Cross-Sectional Regression

Table P.6 Panel A to Panel D show the results for the cross-sectional regressions that test the market price returns and Historical Value-at-Risk Actual by applying the regression specified in equation (0.9). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.

Cross-Sectional Regression Table Description

The Table P.6 columns represent the following:

Year is the *panel data* year represented by the number of days from 1st January to 31st December, and also represents the 31st December variable record date for each year.

The Regression column presents the coefficients and related statistics for the regression specified in equation (0.9) and follows the model:

 $V_{i,t} = a + b \ dM_{i,t} + e_{i,t}$ Where: $V_{i,t}$ is the Value-at-Risk (V_t) for the i^{th} firm at time t, $dM_{i,t}$ is the log Market Price return $(M_{t-1,t})$ for the i^{th} firm at time t, and $e_{N,t}$ is the regression error term. The Lower and Upper are the corresponding coefficient levels predicted at a 99% confidence level.

The Correlation column presents the correlation statistics for the relationship between the dependent and independent variables. The Descriptive Statistics column presents for the independent variable, the sample mean calculated using equation (0.17) and the sample standard deviation, SD, using equation (0.18). The Obs. column presents the number of sample firms observed for the year.

								Table P.6	6 (Contin	nued)								
	The	Duiman	wand Saa	andawy (Samula (waa Saatia	nal Dag	PA	NEL A	for Morty	ot Duigo	Dotum	ng and l	Historical V	alua at Dia	lr A atual		
	The	riillar	y and sec	onuary s	Sample C	Regree	ssion	ression A				Ketur	iis anu i	nistorical v	alue-at-Kis	Desori	ntivo	
Year			Slope				Intercept					Model		Corre	Statistics		Obs.	
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	0.904**	14.67	5.0E-09	0.72	1.09	0.010	0.01	4.9E-01	-0.03	0.05	0.95	0.05	12	0.973**	2.5E-09	0.03	0.23	14
1995	0.472*	2.53	2.7E-02	-0.10	1.04	-0.284**	-0.28	5.2E-04	-0.47	-0.10	0.35	0.17	12	0.589*	1.3E-02	0.21	0.25	14
1996	0.841*	2.82	1.5E-02	-0.06	1.74	-0.063	-0.06	2.0E-01	-0.20	0.08	0.38	0.15	13	0.616**	7.3E-03	0.09	0.13	15
1997	0.192	1.06	3.1E-01	-0.35	0.74	-0.042	-0.04	4.7E-01	-0.21	0.13	0.08	0.12	13	0.282	1.5E-01	0.26	0.18	15
1998	0.520	1.73	1.1E-01	-0.38	1.42	-0.135	-0.13	6.5E-02	-0.34	0.07	0.19	0.25	13	0.433	5.3E-02	0.06	0.22	15
1999	0.293*	2.40	3.1E-02	-0.07	0.66	-0.183**	-0.18	8.3E-04	-0.31	-0.05	0.29	0.12	14	0.54*	1.5E-02	0.26	0.25	16
2000	0.485*	2.41	3.0E-02	-0.11	1.08	-0.221*	-0.22	1.2E-02	-0.45	0.01	0.29	0.27	14	0.541*	1.5E-02	0.18	0.35	16
2001	0.425*	2.59	2.1E-02	-0.06	0.91	-0.229**	-0.23	2.8E-04	-0.37	-0.09	0.32	0.16	14	0.569*	1.1E-02	-0.16	0.25	16
2002	0.721**	9.73	1.3E-07	0.50	0.94	-0.146**	-0.15	1.1E-03	-0.25	-0.04	0.87	0.09	14	0.933**	6.5E-08	-0.37	0.32	16
2003	1.230	2.05	6.0E-02	-0.56	3.02	-0.804**	-0.80	4.4E-04	-1.33	-0.28	0.23	0.43	14	0.48*	3.0E-02	0.23	0.18	16
2004	-0.581	-0.83	4.2E-01	-2.66	1.50	-0.040	-0.04	6.4E-01	-0.29	0.21	0.05	0.23	14	-0.217	2.1E-01	0.09	0.09	16
2005	0.194	2.14	5.0E-02	-0.08	0.46	-0.026	-0.03	3.4E-01	-0.10	0.05	0.25	0.07	14	0.497*	2.5E-02	0.20	0.21	16
2006	0.635	1.99	6.7E-02	-0.32	1.59	-0.035	-0.03	5.6E-01	-0.21	0.14	0.22	0.12	14	0.469*	3.3E-02	0.15	0.10	16
2007	0.651**	10.81	3.5E-08	0.47	0.83	-0.063**	-0.06	2.0E-03	-0.11	-0.01	0.89	0.06	14	0.945**	1.8E-08	-0.08	0.27	16
2008	0.993**	16.16	1.9E-10	0.81	1.18	-0.098	-0.10	9.3E-02	-0.26	0.06	0.95	0.15	14	0.974**	9.4E-11	-0.64	0.64	16

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 16 observations per yearly regression from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

							I	Table P.6	(Contin	nued)								
				a	2 2			PA	NEL B				.					
		The	e Control	Group (cross-Sec	Regres	ession A	Analysis fo	or Mark	tet Price I	Keturns	and F	listorica	l Value-at-l	Risk Actual	l		
Year			Slope				Intercept					Model		Corre	lation	Descriptive Statistics		Obs.
	b	t(b)	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	0.353	1.89	9.2E-02	-0.25	0.96	-0.065*	-0.07	4.3E-02	-0.16	0.02	0.28	0.07	9	0.532*	4.6E-02	-0.09	0.12	11
1995	1.223*	3.19	1.1E-02	-0.02	2.47	-0.313**	-0.31	9.0E-05	-0.47	-0.16	0.53	0.08	9	0.728**	5.5E-03	0.10	0.07	11
1996	0.758**	8.64	6.0E-06	0.48	1.04	-0.046*	-0.05	1.1E-02	-0.09	0.00	0.88	0.05	10	0.939**	3.0E-06	0.02	0.17	12
1997	1.002**	5.10	4.7E-04	0.38	1.62	-0.152**	-0.15	1.8E-03	-0.27	-0.04	0.72	0.10	10	0.85**	2.3E-04	0.10	0.16	12
1998	0.749**	4.22	1.8E-03	0.19	1.31	-0.161**	-0.16	4.7E-03	-0.30	-0.02	0.64	0.14	10	0.801**	8.8E-04	-0.11	0.24	12
1999	-0.078	-0.35	7.3E-01	-0.78	0.62	-0.144	-0.14	2.2E-01	-0.49	0.20	0.01	0.26	10	-0.112	3.7E-01	0.35	0.36	12
2000	0.873**	11.09	6.1E-07	0.62	1.12	-0.030	-0.03	1.5E-01	-0.09	0.03	0.92	0.07	10	0.962**	3.1E-07	0.02	0.26	12
2001	0.268	0.52	6.1E-01	-1.36	1.90	-0.254*	-0.25	3.2E-02	-0.58	0.07	0.03	0.14	10	0.162	3.1E-01	-0.18	0.08	12
2002	0.223	1.84	9.6E-02	-0.16	0.61	-0.314**	-0.31	2.4E-05	-0.45	-0.18	0.25	0.07	10	0.502*	4.8E-02	-0.31	0.18	12
2003	-0.092	-0.44	6.7E-01	-0.75	0.57	-0.434**	-0.43	1.6E-05	-0.61	-0.26	0.02	0.09	10	-0.138	3.3E-01	0.24	0.13	12
2004	0.801**	6.89	4.2E-05	0.43	1.17	-0.106**	-0.11	9.7E-05	-0.16	-0.05	0.83	0.03	10	0.909**	2.1E-05	0.12	0.08	12
2005	-0.272	-1.54	1.5E-01	-0.83	0.29	0.147*	0.15	3.9E-02	-0.05	0.34	0.19	0.07	10	-0.438	7.7E-02	0.33	0.12	12
2006	0.878**	14.46	5.0E-08	0.69	1.07	-0.028*	-0.03	3.8E-02	-0.07	0.01	0.95	0.04	10	0.977**	2.5E-08	0.09	0.18	12
2007	0.578**	6.51	6.8E-05	0.30	0.86	-0.062*	-0.06	1.4E-02	-0.13	0.00	0.81	0.07	10	0.9**	3.4E-05	0.02	0.24	12
2008	0.695**	3.93	2.8E-03	0.13	1.26	-0.172	-0.17	6.3E-02	-0.43	0.09	0.61	0.12	10	0.779**	1.4E-03	-0.42	0.20	12

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The Control group consists of a maximum of 12 observations per yearly regression from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

								Table P.6	(Contir	nued)								
				a 1	c c			PA	NEL C		D (
		The	Primary	Sample	Cross-See	Ctional Regi	ression	Analysis	tor Mar	ket Price	Return	s and l	Historic	al Value-at	-Risk Actua	al		
														Corre	elation	Descri	iptive	
Year			Slope				Intercept									Statis	Statistics	
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	0.893**	11.38	7.6E-03	0.11	1.67	0.031	1.91	2.0E-01	-0.13	0.19	0.98	0.03	2	0.992**	3.8E-03	-0.06	0.23	4
1995	-0.160	-0.42	7.1E-01	-3.92	3.60	-0.170	-1.15	3.7E-01	-1.65	1.31	0.08	0.17	2	-0.286	3.6E-01	0.32	0.26	4
1996	0.403	1.37	3.1E-01	-2.53	3.34	0.025	0.35	7.6E-01	-0.69	0.74	0.48	0.05	2	0.695	1.5E-01	0.23	0.10	4
1997	0.341	0.84	4.9E-01	-3.67	4.36	-0.021	-0.16	8.9E-01	-1.28	1.24	0.26	0.18	2	0.512	2.4E-01	0.23	0.25	4
1998	-0.679	-0.35	7.6E-01	-19.76	18.40	-0.395	-1.72	2.3E-01	-2.67	1.88	0.06	0.44	2	-0.242	3.8E-01	-0.03	0.13	4
1999	0.254	1.18	3.2E-01	-1.01	1.51	-0.200	-2.22	1.1E-01	-0.73	0.33	0.32	0.12	3	0.562	1.6E-01	0.33	0.28	5
2000	0.104	0.13	9.0E-01	-4.54	4.74	-0.322	-2.59	8.1E-02	-1.05	0.40	0.01	0.22	3	0.076	4.5E-01	0.09	0.14	5
2001	1.252*	4.79	1.7E-02	-0.27	2.78	-0.122*	-3.65	3.5E-02	-0.32	0.07	0.88	0.07	3	0.94**	8.6E-03	-0.03	0.14	5
2002	0.514*	4.18	2.5E-02	-0.20	1.23	-0.186*	-4.89	1.6E-02	-0.41	0.04	0.85	0.04	3	0.924*	1.2E-02	-0.27	0.18	5
2003	1.464	1.82	1.7E-01	-3.23	6.16	-0.639*	-3.89	3.0E-02	-1.60	0.32	0.53	0.19	3	0.725	8.3E-02	0.18	0.12	5
2004	0.045	0.07	9.5E-01	-3.84	3.93	-0.048	-0.86	4.5E-01	-0.38	0.28	0.00	0.08	3	0.039	4.7E-01	0.07	0.06	5
2005	0.182	1.96	1.4E-01	-0.36	0.72	-0.022	-1.76	1.8E-01	-0.09	0.05	0.56	0.02	3	0.750	7.2E-02	0.09	0.12	5
2006	0.214	0.70	5.4E-01	-1.59	2.01	-0.009	-0.20	8.5E-01	-0.25	0.24	0.14	0.04	3	0.372	2.7E-01	0.12	0.07	5
2007	0.631*	4.81	1.7E-02	-0.14	1.40	-0.117*	-3.20	4.9E-02	-0.33	0.10	0.89	0.07	3	0.941**	8.6E-03	-0.17	0.25	5
2008	0.94**	13.59	8.6E-04	0.54	1.34	-0.076	-0.89	4.4E-01	-0.57	0.42	0.98	0.09	3	0.992**	4.3E-04	-1.07	0.68	5

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 5 observations per yearly regression from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

								Table P.6	(Contin	ued)								
								PA	NEL D									
		The	Secondar	y Sampl	e Cross-S	ectional Reg	gressio	n Analysis	for Ma	rket Price	e Returr	is and	Histori	cal Value-a	t-Risk Actu	ıal		
						Regre	ssion									Descr	iptive	
Year			Slope			Intercept					Model			Correlation		Statistics		Obs.
	в	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	a	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	0.93**	11.28	3.4E-06	0.65	1.21	-0.001	-0.05	9.6E-01	-0.06	0.06	0.94	0.06	8	0.97**	1.7E-06	0.07	0.23	10
1995	0.792**	5.10	9.3E-04	0.27	1.31	-0.301**	-6.66	1.6E-04	-0.45	-0.15	0.76	0.12	8	0.875**	4.6E-04	0.17	0.25	10
1996	1.052	2.00	7.6E-02	-0.65	2.76	-0.067	-1.20	2.6E-01	-0.25	0.11	0.31	0.17	9	0.556*	3.8E-02	0.04	0.10	11
1997	0.114	0.53	6.1E-01	-0.59	0.82	-0.041	-0.62	5.5E-01	-0.26	0.17	0.03	0.11	9	0.173	3.1E-01	0.27	0.16	11
1998	0.437**	3.53	6.4E-03	0.03	0.84	-0.047	-1.54	1.6E-01	-0.15	0.05	0.58	0.10	9	0.762**	3.2E-03	0.09	0.24	11
1999	0.344	2.08	6.7E-02	-0.19	0.88	-0.18**	-3.41	7.8E-03	-0.35	-0.01	0.32	0.13	9	0.57*	3.4E-02	0.23	0.24	11
2000	0.453	2.07	6.8E-02	-0.26	1.17	-0.151	-1.55	1.5E-01	-0.47	0.17	0.32	0.28	9	0.568*	3.4E-02	0.21	0.41	11
2001	0.240	1.34	2.1E-01	-0.34	0.82	-0.305**	-5.05	6.9E-04	-0.50	-0.11	0.17	0.15	9	0.409	1.1E-01	-0.22	0.27	11
2002	0.732**	7.80	2.7E-05	0.43	1.04	-0.149*	-2.92	1.7E-02	-0.31	0.02	0.87	0.11	9	0.933**	1.3E-05	-0.42	0.36	11
2003	1.401	1.88	9.2E-02	-1.02	3.82	-0.941**	-3.89	3.6E-03	-1.73	-0.16	0.28	0.49	9	0.532*	4.6E-02	0.26	0.21	11
2004	-0.621	-0.67	5.2E-01	-3.63	2.38	-0.052	-0.42	6.9E-01	-0.45	0.35	0.05	0.29	9	-0.219	2.6E-01	0.10	0.10	11
2005	0.200	1.57	1.5E-01	-0.21	0.61	-0.029	-0.67	5.2E-01	-0.17	0.11	0.21	0.09	9	0.463	7.6E-02	0.26	0.23	11
2006	0.657	1.52	1.6E-01	-0.75	2.06	-0.027	-0.32	7.6E-01	-0.30	0.25	0.20	0.14	9	0.452	8.1E-02	0.17	0.10	11
2007	0.613**	10.87	1.8E-06	0.43	0.80	-0.04*	-2.65	2.6E-02	-0.09	0.01	0.93	0.05	9	0.964**	8.9E-07	-0.03	0.28	11
2008	1.115**	12.50	5.4E-07	0.83	1.40	-0.080	-1.33	2.1E-01	-0.27	0.11	0.95	0.15	9	0.972**	2.7E-07	-0.44	0.53	11

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 11 observations per yearly regression from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

Table P.7Analysis for Market Price Returns and Historical Value-at-Risk
Actual

Time Series Regression

Table P.7 Panel A to Panel D show the results for the time series regressions that test the market price returns and Historical Value-at-Risk Actual by applying the regression specified in equation (O.12). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.

Time Series Regression Table Description

The Table P.7 columns represent the following:

The Total Period column represents the time series regression results for the year range 1994 to 2008, with a single year represented by the number of days from 1st January to 31st December. The Sub-Periods column represents the time series regression results for the year range 1994 to 2004, and for 1994 to 2007, with a single year represented by the number of days from 1st January to 31st December.

Measure column represents the time series regression results for the total and sub-time periods and is represented by the data panels:

The *Regression* panel presents the coefficients and related statistics for the regression specified in equation (0.12) and follows the model:

$$V_{N,t} = a + b \ dM_{N,t} + e_{N,t}$$

Where: $V_{N,t}$ is the Value-at-Risk (V_t) for N firms at time t, $dM_{N,t}$ is the log Market Price return (dM_t) for N firms at time t, and $e_{N,t}$ is the regression error term. The *Lower* and *Upper* are the corresponding coefficient levels predicted at a 99% confidence level.

The *Correlation* panel presents the correlation statistics for the relationship between the dependent and independent variables. The *Descriptive Statistics* panel presents for the variables tested, the average sample means calculated using equation (M.6); the standard deviation, *SD*, of sample means for the time series using equation (M.7); standard error of means, *SE*, for the time series. The *Observations* panel presents: *Firms* (*n*) (term *N* in the regression) that represents the average number of sample firms in the time series regression (for the Primary and Secondary sample, for the time period 1994 to 2004, the number of firms are rounded down); *Years* (*T*) represents the number of time series years; *Total* ($n \times T$) represents the number of averaged pooled time series regression observations.

Маалия		Total Period	Sub-F	Periods
Weasure		1994-2008	1994-2004	1994-2007
Slope	b	0.613**	0.326	0.361
	t (b)	3.38	1.13	1.40
	p (b) SE(b)	4.9E-03 0.181	2.9E-01 0.287	0.257
	Lower	0.067	-0.608	-0.425
	Upper	1.159	1.260	1.147
Intercept Regression	a t (a) p (a) SE(a)	-0.191** -4.26 9.4E-04 0.045	-0.188* -3.25 1.0E-02 0.058	-0.16** -3.23 7.2E-03 0.050
	Lower	-0.327	-0.377	-0.311
	Upper	-0.056	8.2E-05	-0.009
Model	R^{2} $s(e)$ $df(e)$	0.468 0.172 13	0.125 0.177 9	0.141 0.167 12
Correlation	r	0.684**	0.353	0.375
	p(r)	2.5E-03	1.4E-01	9.3E-02
Descriptive Statistics	M	0.035	0.080	0.083
	SD (M)	0.254	0.194	0.180
	SE(M)	0.066	0.059	0.048
Descriptive Statistics	V 95	-0.170	-0.162	-0.130
	SD (V 95)	0.228	0.179	0.173
	SE(V 95)	0.059	0.054	0.046
Observations	Firms (n)	16	15	16
	Years (T)	15	11	14
	Total ($n \times T$)	233	169	217

Table P.7 (Continued) PANEL A

The Primary and Secondary Sample Time Series Regression Analysis for Market Price

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The sample consists of a maximum of 233 observations from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. Observations are based on yearly averages calculated using daily levels from 1994 to 2008.

See main table notes for the regression model applied.

Маал		Total Period	Sub-F	Periods	
Ivieas	ure	1994-2008	1994-2004	1994-2007	
Slop	$\begin{array}{c} e & b \\ t & (b) \\ t & (d) \end{array}$	0.42* 2.16	0.165 0.62	0.291 1.24	
	p (b)	5.0E-02	5.5E-01	2.4E-01	
	SE(b)	0.194	0.266	0.236	
	Lower	-0.165	-0.699	-0.429	
	Upper	1.006	1.029	1.012	
Interc Regression	ept a t (a) p (a) SE(a)	-0.165** -4.03 1.4E-03 0.041	-0.181** -3.72 4.8E-03 0.049	-0.149** -3.40 5.3E-03 0.044	
	Lower	-0.288	-0.338	-0.283	
	Upper	-0.041	-0.023	-0.015	
Mod	$ \begin{array}{c} el & R^2 \\ s(e) \\ df(e) \end{array} $	0.264 0.158 13	0.041 0.159 9	0.113 0.158 12	
Correlation	r	0.514*	0.203	0.336	
	p(r)	2.5E-02	2.8E-01	1.2E-01	
Descriptive Statisti	M	0.020	0.025	0.051	
	SD (M)	0.217	0.190	0.186	
	SE(M)	0.056	0.057	0.050	
Deserptive Statisti	V 95	-0.156	-0.177	-0.134	
	SD (V 95)	0.177	0.155	0.161	
	SE(V 95)	0.046	0.047	0.043	
Observations	Firms (n)	12	12	12	
	Years (T)	15	11	14	
	Total ($n \times T$)	178	130	166	

Table P.7 (Continued) PANEL B

The Control Group Time Series Regression Analysis for Market Price Returns and

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The control consists of a maximum of 178 observations from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. Observations are based on yearly averages calculated using daily levels from 1994 to 2008.

See main table notes for the regression model applied.

	Maaguma		Total Period	Sub-F	Periods
	Measure		1994-2008	1994-2004	1994-2007
	Slope	b	0.696**	0.314	0.322
Regression Correl Descriptive	~~r~r~	<i>t</i> (<i>b</i>)	5.11	1.05	1.26
		p(b)	2.0E-04	3.2E-01	2.3E-01
		SE(b)	0.136	0.299	0.256
		Lower	0.286	-0.659	-0.460
		Upper	1.106	1.287	1.104
Regression	Intercent	a	-0.206**	-0.192*	-0.167**
Regression	inter copt	t(a)	-4.58	-3.20	-3.49
		p (a)	5.1E-04	1.1E-02	4.4E-03
		SE(a)	0.045	0.060	0.048
		Lower	-0.342	-0.387	-0.314
		Upper	-0.071	0.003	-0.021
	Model	R^2	0.668	0.109	0.116
		s(e)	0.174	0.174	0.163
		df(e)	13	9	12
			0.917**	0.220	0.241
Correla	ation	p(r)	1.0E-04	1.6E-01	1.2E-01
		М	0.001	0.096	0.078
		SD(M)	0.342	0.184	0.177
Regression Correl Descriptive Observe	Statistics	SE(M)	0.088	0.055	0.047
	SIGUISTICS	V 95	-0.205	-0.162	-0.142
		SD (V 95)	0.291	0.175	0.167
		SE(V 95)	0.075	0.053	0.045
		Firms (n)	5	5	5
Observa	ations	Years (T)	15	11	14
		Total ($n \times T$)	70	51	64

Table P.7 (Continued) PANEL C

The Primary Sample Time Series Regression Analysis for Market Price Returns and

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The sample consists of a maximum of 70 observations from the 5 banks in the LSE that adopted the IFRS standards in 2005. Observations are based on yearly averages calculated using daily levels from 1994 to 2008.

See main table notes for the regression model applied.

Маалия	-	Total Period	Sub-F	Periods	
Measur	8	1994-2008	1994-2004	1994-2007	
Slope	b	0.569*	0.388	0.429	
	t (b)	2.65	1.30	1.62	
	p (b)	2.0E-02	2.3E-01	1.3E-01	
	SE(b)	0.215	0.299	0.265	
	Lower	-0.077	-0.583	-0.380	
	Upper	1.215	1.359	1.238	
Intercept Regression	t (a) p (a) SE(a)	-0.184** -3.71 2.6E-03 0.050	-0.192* -2.98 1.5E-02 0.064	-0.162* -2.94 1.2E-02 0.055	
	Lower	-0.333	-0.402	-0.331	
	Upper	-0.034	0.017	0.006	
Model	R^{2} s(e) df(e)	0.351 0.187 13	0.158 0.201 9	0.179 0.189 12	
Correlation	r	0.593**	0.397	0.423	
	p(r)	1.0E-02	1.1E-01	6.6E-02	
Descriptive Statistics	M	0.050	0.072	0.085	
	SD (M)	0.233	0.213	0.197	
	SE(M)	0.060	0.064	0.053	
Descriptive Statistics	V 95	-0.155	-0.164	-0.126	
	SD (V 95)	0.224	0.208	0.200	
	SE(V 95)	0.058	0.063	0.053	
Observations	Firms (n)	11	11	11	
	Years (T)	15	11	14	
	Total ($n \times T$)	163	119	152	

Table P.7 (Continued) PANEL D

The Secondary Sample Time Series Regression Analysis for Market Price Returns and

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The sample consists of a maximum of 163 observations from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. Observations are based on yearly averages calculated using daily levels from 1994 to 2008.

See main table notes for the regression model applied.

Analysis

The Table P.6 cross-sectional regression slopes show that the samples and the control group market price return variable exhibits some statistically significant explanatory power to the Historical Value-at-Risk Actual variable at varying times both before and after the 2005 accounting change.

The Table P.7 time series regression slopes show that the samples and the control group market price return variable exhibits statistically significant explanatory power to the Historical Value-at-Risk Actual variable for the 1994 to 2008 time period. However, for the sub-time periods 1994 to 2004 and 1994 to 2007 statistical significance is not evident.

This suggests that the relationship between market price returns and Value-at-Risk became significant after 2007. Examining the regression slopes, the Primary and Secondary sample exhibit greater magnitude and significance when compared to the Control group. This reveals that the Primary and Secondary sample market price returns varied closely with Value-at-Risk levels after the 2005 accounting change than before when compared to the Control group.

P.6 Change in GDP Price Indices and Market Price Return

The regression analysis for the change in selected Gross Domestic Product (GDP) indices and the market price return is presented for the time range 1994 to 2008 in Table P.8. In these regressions, the market price return variable is tested as the dependent variable, and the change in the GDP sector indices All Production and Business Services are individually tested as the independent variable.

Table P.8Time Series Regression Analysis for Change in GDP Price Indices
and Market Price Return

Time Series Regression

Table P.8 Panels A to C show the results for the time series regressions that test the change in selected GDP sector indices and the market price return by applying the regression specified in equation (0.13). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; and, Panel C presents results for the Primary sample.

Time Series Regression Table Description

The Table P.8 columns represent the following:

The GDP Index Sector column represents the samples and the control group result panels for the selected indices.

The Returns Time Periods column represents the following:

The Total Period column represents the time series regression results for the year range 1994 to 2008, with a single year represented by the 4 yearly quarters: January 1st to March 31st; April 1st to June 30th; July 1st to September 30th; and, October 1st to December 31st. The Sub-Periods column represents the time series regression results for the year range 1994 to 2004, and for 1994 to 2007, with a single year represented by the 4 yearly quarters: January 1st to March 31st; April 1st to June 30th; July 1st to September 30th; and, October 1st to December 31st.

The total and sub-period columns present the coefficients and related statistics for the regression specified in equation (O.13), and follows the model:

$$dM_{N,t} = a + b \ dGDP_t + e_{N,t}$$

Where: $dM_{N,t}$ is the log Market Price return $(M_{t-1,t})$ for N firms at time t, $dGDP_t$ is the change in the selected GDP index $(GDP_{t-1,t})$ at time t, and $e_{N,t}$ is the regression error term.

				Т	able P.8 (Co	ontinued)							
	Returns Time Periods												
CDD In day Sector	Total Period	Sub-l	Periods	Total Sub-Periods Period		Total Period	Total Sub-Periods		Total Period	Sub-Periods			
GDP Index Sector	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	
		b			t			p			R^2		
		P	ANEL A:	GDP	vs. Primary	and Secon	dary Sample	e Return					
All Production Business Services	6.044** 8.982**	5.007 4.943	3.713 5.194	3.504 3.166	1.655 1.247	1.344 1.504	0.004 0.007	0.132 0.244	0.204 0.158	0.486 0.435	0.233 0.147	0.131 0.159	
			PAN	EL B:	GDP vs.	Control G	roup Return						
All Production Business Services	4.128* 5.889	4.483 1.889	2.521 3.117	2.358 2.067	1.539 0.470	0.877 0.852	0.035 0.059	0.158 0.649	0.398 0.411	0.300 0.247	0.208 0.024	0.060 0.057	
			PAN	EL C:	GDP vs. P	rimary Sa	mple Return	l					
All Production Business Services	8.7** 10.581*	3.629 1.593	3.219 1.692	4.427 2.709	1.290 0.423	1.235 0.486	0.001 0.018	0.229 0.682	0.240 0.636	0.601 0.361	0.156 0.019	0.113 0.019	

Table note: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels.

Analysis

The Table P.8 time series regression slopes show that the change in the GDP *All Production* sector index exhibits statistically significant explanatory power to the samples and the control group market price return variable for the time period 1994 to 2008. The Primary and Secondary sample slopes also show statistical significance for the change in the GDP *Business Services* sector index for the same time period.

Examining the significant regression slopes, the Primary and Secondary sample exhibits greater magnitude, statistical significance and model strength when compared to the Control group for both the GDP sector indices tested.

The results suggest that the GDP All Production and Business Services sectors reacted in relation to the Primary and Secondary sample market price returns after the 2005 accounting change compared to before.

P.7 Change in GDP Price Indices Value-at-Risk Actual and Historical Value-at-Risk Actual

The regression analysis for the selected GDP indices Historical Value-at-Risk Actual and market price return Historical Value-at-Risk Actual is presented for the time range 1994 to 2008 in Table P.9. In these regressions, the Historical Value-at-Risk Actual variable is tested as the dependent variable, and the GDP indices Value-at-Risk Actual for sector indices All Production and Business Services are individually tested as the independent variable.

Table P.9Time Series Regression Analysis for GDP Indices Historical Value-
at-Risk Actual and Historical Value-at-Risk Actual

Time Series Regression

Table P.9 Panels A to C show the results for the time series regressions that test the change in selected GDP indices Value-at-Risk Actual and Historical Value-at-Risk Actual by applying the regression specified in equation (O.14). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; and, Panel C presents results for the Primary sample.

Time Series Regression Table Description

The Table P.9 columns represent the following:

The GDP Index Sector column represents the samples and the control group result panels for the selected indices.

The *V* Time Periods column represents the following:

The Total Period column represents the time series regression results for the year range 1994 to 2008, with a single year represented by the 4 yearly quarters: January 1st to March 31st; April 1st to June 30th; July 1st to September 30th; and, October 1st to December 31st. The Sub-Periods column represents the time series regression results for the year range 1994 to 2004, and for 1994 to 2007, with a single year represented by the 4 yearly quarters: January 1st to March 31st; April 1st to June 30th; July 1st to September 30th; and, October 1st to December 31st.

The total and sub-period columns present the coefficients and related statistics for the regression specified in equation (O.14), and follows the model:

$$V_{N,t} = a + b V G D P_t + e_{N,t}$$

Where: $V_{N,t}$ is the Value-at-Risk (V_t) for N firms at time t, $VGDP_t$ is the Value-at-Risk for the selected GDP index at time t, and $e_{N,t}$ is the regression error term.

				Ta	ble P.9 (Co	ntinued)							
	V Time Periods												
CDD In day, Sector	Total Period	Sub-Po	eriods	Total Sub-Periods Period		Total Period	Sub-Periods		Total Period	Sub-Periods			
GDP Index Sector	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	
		b		t			<i>p</i>			R^2			
		Р	ANEL A:	GDP	V vs. Prima	ary and Se	condary Sam	ple V					
All Production Business Services	6.398** 9.708**	7.054** 6.502	4.846 6.864*	4.207 3.889	3.618 2.206	2.055 2.581	0.001 0.002	0.006 0.055	0.062 0.024	0.577 0.538	0.593 0.351	0.260 0.357	
			PA	NEL B:	GDP V	vs. Contro	l Group V						
All Production Business Services	3.706* 6.628**	5.356* 4.735	3.100 5.572	2.472 3.032	2.764 1.733	1.299 2.120	0.028 0.010	0.022 0.117	0.218 0.056	0.320 0.414	0.459 0.250	0.123 0.272	
			PAN	EL C:	GDP V v	s. Primary	y Sample V						
All Production Business Services	7.728** 8.965*	3.557 1.766	2.085 1.968	0.002 2.268	0.213 0.515	0.420 0.644	0.002 0.041	0.213 0.619	0.420 0.532	0.520 0.283	0.167 0.029	0.055 0.033	

Table note: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels.

Analysis

The Table P.9 time series regression slopes show that the GDP All Production and Business Services sector indices Historical Value-at-Risk Actual exhibit statistically significant explanatory power to the samples and the control group Historical Value-at-Risk Actual for the time periods 1994 to 2008 and 1994 to 2004. The Primary and Secondary sample slopes also show statistical significance for the GDP *Business Services* sector index Historical Value-at-Risk Actual for the time period 1994 to 2007.

Examining the significant regression slopes, the Primary and Secondary sample exhibits greater magnitude, statistical significance, and model strength when compared to the Control group slope for the two GDP sector indices tested.

The results suggest that the All Production and Business Services sector indices Historical Value-at-Risk Actual reacted in relation to the Primary and Secondary sample after the 2005 accounting change compared to before, than the Control group reactions.

The regression slopes also show that the GDP All Production sector index Historical Value-at-Risk Actual reacted in relation to the samples and the control group Historical Value-at-Risk Actual before and after the 2005 accounting change. This reveals that any correlation between the GDP All Production sector index and *the samples and the control group* Historical Value-at-Risk Actuals was also evident before 2005. However, this relationship is not evident with the Primary sample that exhibits a stronger relationship to the selected GDP indices Historical Value-at-Risk Actuals after the 2005 accounting change.

P.8 Change in Stock Indices and Market Price Return

The regression analysis for the change in key LSE stock indices and selected world stock indices, presented in Table P.10, and the market price return is presented for the time range 1994 to 2008 in Table P.11. In these regressions, the market price return variable is tested as the dependent variable, and the change in the key LSE and selected stock indices variables are individually tested as the independent variable.

Index ID ^a	Equity Index Name ^b	Region ^c
FTSE 350	FTSE 350 PRICE INDEX	UK
FTSE 100	FTSE 100 PRICE INDEX	UK
FTSE ALL	FTSE ALL SHARE PRICE INDEX	UK
DJIA	DOW JONES INDUSTRIALS	USA
NIKKEI	NIKKEI 225 STOCK AVERAGE	Japan
S&P 500	S&P 500 COMPOSITE	USA
NASDAQ	NASDAQ COMPOSITE	USA
CAC	FRANCE CAC 40	France
DAX	DAX 30 PERFORMANCE	Germany
RUSSIA	RUSSIA RTS INDEX	Russia
HANG SENG	HANG SENG PRICE INDEX	Hong Kong, China
COLOMBO	COLOMBO SE ALL SHARE	Sri Lanka
AUSTRALIA	AUSTRALIA S&P/ASX 200	Australia
CANADA	CANADA S&P/TSX COMPOSITE	Canada
BRAZIL	FTSE BRAZIL	Brazil
SOUTH AFRICA	MSCI SOUTH AFRICA	South Africa
FTSE WORLD	FTSE ALL WORLD \$	Global

 Table P.10
 Stock Indices Selected for Testing

Table notes:

^a The equity stock index identifier that is applied in Table P.11 and Table P.12 to identify the indices.

^b The general price index name from the Thomson Reuters Datastream financial data database product.

^c The region that most describes the geographical area for the index constituents.

Table P.11Time Series Regression Analysis for Change in Equity Stock Indices
and Market Price Return

Time Series Regression

Table P.11 Panels A to C show the results for the time series regressions that test the change in selected stock indices and the market price return by applying the regression specified in equation (O.13). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents the results for the Control group; and, Panel C presents the results for the Secondary sample.

Time Series Regression Table Description

The Table P.11 columns represent the following:

The Equity Stock Index Description column represents the samples and the control group result panels for the selected indices.

The Returns Time Periods column represents the following:

The Total Period column represents the time series regression results for the year range 1994 to 2008, with a single year represented by the number of days from 1st January to 31st December. The Sub-Periods column represents the time series regression results for the year range 1994 to 2004, and for 1994 to 2007, with a single year represented by the number of days from 1st January to 31st December.

The total and sub-period columns present the coefficients and related statistics for the regression specified in equation (O.13), and follows the model:

 $dM_{N,t} = a + b \ dSTOCK_INDEX_t + e_{N,t}$

Where: $dM_{N,t}$ is the log Market Price return $(M_{t-1,1})$ for N firms at time t, $dSTOCK_INDEX_t$ is the change in the selected stock market index $(STOCK_INDEX_{t-1,t})$ at time t, and $e_{N,t}$ is the regression error term.

PANEL A: Tir	ne Series Re	gression An	alysis of the	Primary ar	nd Seconda	ary Sampl	e for Change i	n Equity Sto	ock Indices a	nd Market l	Price Retu	rn		
				Returns Time Periods										
Equity Stock	Total Period	Sub-F	Periods	Total Period Sub-Periods 1994- 1994- 1994- 2008 2004 2007		Total Sub-Periods			Total Period	Sub-Periods				
Description	1994- 2008	1994- 2004	1994- 2007			1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007		
	b			t				<i>p</i>			R^2			
FTSE 350	1.418**	1.122**	1.123**	7.288	5.522	5.943	6.1E-06	3.7E-04	6.8E-05	0.803	0.772	0.746		
FTSE 100	1.409**	1.091**	1.096**	6.588	5.083	5.396	1.7E-05	6.6E-04	1.6E-04	0.769	0.742	0.708		
FTSE ALL	1.42**	1.13**	1.131**	7.552	5.735	6.180	4.2E-06	2.8E-04	4.7E-05	0.814	0.785	0.761		
DJIA	1.271**	0.956*	0.922**	5.110	3.083	3.083	2.0E-04	0.013	9.5E-03	0.668	0.514	0.442		
NIKKEI	0.725**	0.423	0.426	3.037	1.442	1.916	9.5E-03	0.180	0.080	0.415	0.188	0.234		
S&P 500	1.112**	0.833**	0.845**	5.839	3.680	3.888	5.8E-05	5.1E-03	2.2E-03	0.724	0.601	0.557		
NASDAQ	0.532**	0.357*	0.355*	3.086	2.391	2.502	8.7E-03	0.040	0.028	0.423	0.388	0.343		
CAC	0.929**	0.697**	0.715**	6.248	3.994	4.604	3.0E-05	3.1E-03	6.1E-04	0.750	0.639	0.639		
DAX	0.8**	0.643**	0.608**	5.754	5.650	4.994	6.7E-05	3.1E-04	3.1E-04	0.718	0.780	0.675		
RUSSIA	0.154	0.031	0.041	1.735	0.328	0.505	0.110	0.750	0.620	0.215	0.015	0.025		
HANG SENG	0.57**	0.391	0.309	3.105	1.896	1.636	8.4E-03	0.090	0.130	0.426	0.285	0.182		
COLOMBO	0.152	-0.391	-0.226	0.607	-1.804	-1.153	0.550	0.100	0.270	0.028	0.266	0.100		
AUSTRALIA	1.241**	1.042	0.971*	5.059	2.212	2.364	2.2E-04	0.054	0.036	0.663	0.352	0.318		
CANADA	0.43*	0.253	0.215	7.078	4.247	4.368	8.3E-06	2.2E-03	9.2E-04	0.794	0.667	0.614		
BRAZIL	0.43*	0.253	0.215	2.367	1.379	1.305	0.036	0.210	0.220	0.318	0.192	0.134		
SOUTH AFRICA	0.609	0.250	0.286	1.783	0.693	1.026	0.098	0.510	0.330	0.196	0.051	0.081		
ETSE WORLD	1.119**	0.92**	0.913**	6.270	3.462	3.601	2.9E-05	7.1E-03	3.6E-03	0.751	0.571	0.519		

Table P.11 (Continued) .

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels.

PANE	L B: Time S	eries Regre	ssion Analy	sis of the Co	ontrol Gro	up for Cha	ange in Equity	Stock Indic	es and Marl	ket Price Ret	urn		
						Returns 7	Time Periods						
Equity Stock	Total Period	Total Sub-Periods		Total Sub-Periods		Total Sub-Periods		eriods	Total Period	Sub-Periods			
Description	1994-	1994-	1994-	1994-	1994-	1994-	1994-	1994-	1994-	1994-	1994-	1994-	
	2008	2004	2007	2008	2004	2007	2008	2004	2007	2008	2004	2007	
		b			t			p			R^2		
FTSE 350	1.184**	1.023**	1.102**	6.619	4.314	5.017	1.7E-05	1.9E-03	3.0E-04	0.771	0.674	$0.677 \\ 0.600 \\ 0.709$	
FTSE 100	1.153**	0.954**	1.039**	5.638	3.638	4.245	8.1E-05	5.4E-03	1.1E-03	0.710	0.595		
FTSE ALL	1.195**	1.049**	1.125**	7.101	4.689	5.405	8.0E-06	1.1E-03	1.6E-04	0.795	0.710		
DJIA	0.926**	0.866*	0.761*	3.524	2.677	2.181	3.7E-03	0.025	0.050	0.489	0.443	0.284	
NIKKEI	0.797**	0.679*	0.701**	5.405	3.038	4.217	1.2E-04	0.014	1.2E-03	0.692	0.506	0.597	
S&P 500	0.803**	0.691*	0.671*	3.751	2.628	2.437	2.4E-03	0.027	0.031	0.520	0.434	0.331	
NASDAQ	0.476**	0.386*	0.381*	3.365	2.857	2.671	5.1E-03	0.019	0.020	0.466	0.476	0.373	
CAC	0.759**	0.636**	0.691**	5.370	3.378	3.920	1.3E-04	8.1E-03	2.0E-03	0.689	0.559	0.562	
DAX	0.675**	0.568**	0.597**	5.581	3.982	4.359	8.9E-05	3.2E-03	9.3E-04	0.706	0.638	0.613	
RUSSIA	0.17*	0.099	0.117	2.466	1.166	1.516	0.031	0.280	0.160	0.356	0.163	0.187	
HANG SENG COLOMBO AUSTRALIA	0.548** 0.180 1.027**	0.505* -0.155 1.159*	0.457* -0.044 1.217**	3.924 0.853 4.663	3.011 -0.643 2.754	2.694 -0.208 3.265	1.7E-03 0.410 4.4E-04	$\begin{array}{c} 0.015 \\ 0.540 \\ 0.022 \end{array}$	0.020 0.840 6.8E-03	0.542 0.053 0.626	$0.502 \\ 0.044 \\ 0.457$	$0.377 \\ 0.004 \\ 0.470$	
CANADA	0.504**	0.424**	0.422**	8.382	5.334	6.621	1.3E-06	4.7E-04	2.5E-05	0.844	0.760	0.785	
BRAZIL	0.504**	0.424**	0.422**	4.390	3.377	3.368	8.8E-04	9.7E-03	6.3E-03	0.616	0.588	0.508	
SOUTH AFRICA	0.751**	0.502	0.591*	3.017	1.571	2.402	9.9E-03	0.150	0.033	0.412	0.215	0.325	
FISE WORLD	0.899**	0.901**	0.938**	3.121	5.490	3.381	2.0E-04	6.8E-03	3.8E-03	0.669	0.575	0.51/	

Table P.11 (Continued)

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels.

	PANE	L C: Time Se	eries Regres	sion Analys	is of the Pri	mary Sam	ple for Ch	ange in Equity	y Stock India	ces and Mar	ket Price Re	turn	
							Returns 7	Time Periods					
	Equity Stock	Total Period	Sub-P	ub-Periods Total Period		Sub-P	eriods	Total Period	Sub-Periods		Total Period	Sub-Periods	
	Description	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007
			b			t			р			R^2	
	FTSE 350 FTSE 100 FTSE ALL	1.708** 1.686** 1.715**	1.045** 1.008** 1.058**	0.988** 0.955** 1.001**	4.852 4.499 4.983	5.177 4.655 5.467	4.263 3.918 4.437	3.2E-04 6.0E-04 2.5E-04	5.8E-04 1.2E-03 4.0E-04	1.1E-03 2.0E-03 8.1E-04	0.644 0.609 0.656	0.749 0.707 0.769	0.602 0.561 0.621
511	DJIA NIKKEI	1.772** 0.916*	1.027** 0.438	1.034** 0.363	5.755 2.742	4.200 1.615	4.069 1.609	6.7E-05 0.017	2.3E-03 0.140	1.6E-03 0.130	0.718 0.366	0.662 0.225	0.580 0.177
	S&P 500 NASDAQ	1.439** 0.681*	0.774** 0.367*	0.807** 0.368*	5.134 2.836	3.517 2.761	3.681 2.734	1.9E-04 0.014	6.5E-03 0.022	3.1E-03 0.018	0.670 0.382	0.579 0.459	0.530 0.384
	CAC DAX RUSSIA HANG SENG COLOMBO AUSTRALIA	1.112** 0.898** 0.274* 0.794** 0.358 1.766**	0.595* 0.549** 0.104 0.458* -0.350 1.191*	0.587** 0.472* 0.102 0.322 -0.211 0.959*	4.357 3.610 2.632 3.306 1.095 6.128	3.122 3.960 1.452 2.648 -1.675 3.083	3.120 2.969 1.520 1.769 -1.092 2.392	7.8E-04 3.2E-03 0.023 5.7E-03 0.290 3.6E-05	0.012 3.3E-03 0.190 0.027 1.3E-01 0.013	8.9E-03 0.012 0.160 0.100 0.300 0.034	$\begin{array}{c} 0.593 \\ 0.501 \\ 0.386 \\ 0.457 \\ 0.084 \\ 0.743 \end{array}$	0.520 0.635 0.232 0.438 0.238 0.514	$\begin{array}{c} 0.448 \\ 0.423 \\ 0.188 \\ 0.207 \\ 0.090 \\ 0.323 \end{array}$
	CANADA BRAZIL SOUTH AFRICA FTSE WORLD	0.598* 0.598* 0.791 <u>1.434**</u>	0.309 0.309 0.343 0.778*	0.228 0.228 0.247 0.73*	6.179 2.496 1.707 5.292	5.648 2.046 1.036 2.747	3.786 1.468 0.896 2.520	3.3E-05 0.028 0.110 1.5E-04	3.1E-04 0.075 0.330 0.023	2.6E-03 0.170 0.390 0.027	0.746 0.342 0.183 0.683	0.780 0.344 0.107 0.456	0.544 0.164 0.063 0.346
	ranei notes. ••, * Re	gression coer	ncients sign	incant at the	0.01 and 0.0	is two-talle	cu levels.						

Analysis

The Table P.11 time series regression slopes show that the change in the selected equity stock indices generally exhibit statistically significant explanatory power to the samples and the control group market price return variable for the three time periods tested. Examining the significant regression slopes, the samples and the control group exhibit greater magnitude, and marginally higher significance for the time period 1994 to 2008 than for the time periods 1994 to 2004 and 1994 to 2007. This suggests that the change in the selected indices exhibited a smaller change in relation to the samples and the control group market price return variable after the 2005 accounting change compared to before.

P.9 Stock Indices Value-at-Risk Actual and Historical Value-at-Risk Actual

The regression analysis for key LSE stock indices and selected world stock indices Historical Value-at-Risk Actual and market price return Historical Value-at-Risk Actual is presented for the time range 1994 to 2008 in Table P.12. In the regression analysis, the Historical Value-at-Risk Actual variable is tested as the dependent variable, and the key LSE and selected stock indices Value-at-Risk Actual variables are individually tested as the independent variable.

Table P.12Time Series Regression Analysis for Equity Stock Indices Historical
Value-at-Risk Actual and Historical Value-at-Risk Actual

Time Series Regression

Table P.12 Panels A to C show the results for the time series regressions that test the key LSE and selected stock indices Value-at-Risk Actual and the Historical Value-at-Risk Actual by applying the regression specified in equation (O.14). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; and, Panel C presents results for the Primary sample.

Time Series Regression Table Description

The Table P.12 columns represent the following:

The Equity Stock Index Description column represents the samples and the control group result panels for the selected indices.

The V Time Periods column represents the following:

The Total Period column represents the time series regression results for the year range 1994 to 2008, with a single year represented by the number of days from 1st January to 31st December. The Sub-Periods column represents the time series regression results for the year range 1994 to 2004, and for 1994 to 2007, with a single year represented by the number of days from 1st January to 31st December.

The total and sub-period columns present the coefficients and related statistics for the regression specified in equation (O.14), and follows the model:

 $V_{N,t} = a + b VSTOCK_INDEX_t + e_{N,t}$ Where: $V_{N,t}$ is the Value-at-Risk (V_t) for N firms at time t, $VSTOCK_INDEX_t$ is the Value-at-Risk for the selected stock market index at time t, and $e_{N,t}$ is the regression error term.

Total Period	Sub-Periods	
Total Period	Sub-Periods	
	Sub-Periods	
1994- 2008	1994- 2004 2007	
R^2		
0.880 0.869 0.885	$\begin{array}{ccc} 0.817 & 0.838 \\ 0.813 & 0.829 \\ 0.823 & 0.844 \end{array}$	
0.839 0.739	0.831 0.717 0.518 0.594	
0.841 0.518	$\begin{array}{ccc} 0.782 & 0.733 \\ 0.444 & 0.442 \end{array}$	
0.752 0.788 0.039	0.670 0.698 0.872 0.843 0.059 0.001	
0.370 0.014 0.680	$\begin{array}{cccc} 0.105 & 0.178 \\ 0.151 & 0.027 \\ 0.345 & 0.409 \end{array}$	
0.645 0.276 0.463 0.800	0.400 0.463 0.016 0.098 0.210 0.357 0.802 0.702	
	$\begin{array}{c} 0.880\\ 0.869\\ 0.885\\ 0.839\\ 0.739\\ 0.739\\ 0.841\\ 0.518\\ 0.752\\ 0.788\\ 0.039\\ 0.370\\ 0.014\\ 0.680\\ 0.645\\ 0.276\\ 0.463\\ 0.890\\ \end{array}$	

PANEL A: Time Series Regression Analysis of the Primary and Secondary Sample for Equity Stock Indices Value-at-Risk Actual and Historical Value-at-Risk Actual

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels.

Equity Stock Index Description	V Time Periods												
	Total Period	Sub-Periods		Total Period	Sub-Periods		Total Period	Sub-Periods		Total Period	Sub-Periods		
	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	
	b			t			p			R^2			
FTSE 350	1.184**	1.023**	1.102**	6.619	4.314	5.017	1.7E-05	1.9E-03	3.0E-04	0.771	0.674	$0.677 \\ 0.600 \\ 0.709$	
FTSE 100	1.153**	0.954**	1.039**	5.638	3.638	4.245	8.1E-05	5.4E-03	1.1E-03	0.710	0.595		
FTSE ALL	1.195**	1.049**	1.125**	7.101	4.689	5.405	8.0E-06	1.1E-03	1.6E-04	0.795	0.710		
DJIA	0.926**	0.866*	0.761*	3.524	2.677	2.181	3.7E-03	0.025	0.050	0.489	0.443	0.284	
NIKKEI	0.797**	0.679*	0.701**	5.405	3.038	4.217	1.2E-04	0.014	1.2E-03	0.692	0.506	0.597	
S&P 500	0.803**	0.691*	0.671*	3.751	2.628	2.437	2.4E-03	0.027	0.031	0.520	0.434	0.331	
NASDAQ	0.476**	0.386*	0.381*	3.365	2.857	2.671	5.1E-03	0.019	0.020	0.466	0.476	0.373	
CAC	0.759**	0.636**	0.691**	5.370	3.378	3.920	1.3E-04	8.1E-03	2.0E-03	0.689	0.559	0.562	
DAX	0.675**	0.568**	0.597**	5.581	3.982	4.359	8.9E-05	3.2E-03	9.3E-04	0.706	0.638	0.613	
RUSSIA	0.17*	0.099	0.117	2.466	1.166	1.516	0.031	0.280	0.160	0.356	0.163	0.187	
HANG SENG	0.548**	0.505*	0.457*	3.924	3.011	2.694	1.7E-03	0.015	0.020	0.542	0.502	0.377	
COLOMBO	0.180	-0.155	-0.044	0.853	-0.643	-0.208	0.410	0.540	0.840	0.053	0.044	0.004	
CANADA BRAZIL	0.504** 0.504**	0.424** 0.424**	0.422** 0.422**	4.003 8.382 4.390	2.734 5.334 3.377	5.265 6.621 3.368	4.4E-04 1.3E-06 8.8E-04	4.7E-04 9.7E-03	0.8E-03 2.5E-05 6.3E-03	0.826 0.844 0.616	0.437 0.760 0.588	0.470 0.785 0.508	
OUTH AFRICA	0.751**	0.502	0.591*	3.017	1.571	2.402	9.9E-03	0.150	0.033	0.412	0.215	0.325	
FTSE WORLD	0.899**	0.901**	0.938**	5.121	3.490	3.581	2.0E-04	6.8E-03	3.8E-03	0.669	0.575	0.517	

Table P.12 (Continued)

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels.

Equity Stock Index Description	V Time Periods													
	Total Period	Sub-Periods		Total Period	Sub-Periods		Total Period	Sub-Periods		Total Period	Sub-Periods			
	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007		
	b			t			<i>p</i>			R^2				
FTSE 350 FTSE 100 FTSE ALL	1.708** 1.686** 1.715**	1.045** 1.008** 1.058**	0.988** 0.955** 1.001**	4.852 4.499 4.983	5.177 4.655 5.467	4.263 3.918 4.437	3.2E-04 6.0E-04 2.5E-04	5.8E-04 1.2E-03 4.0E-04	1.1E-03 2.0E-03 8.1E-04	0.644 0.609 0.656	0.749 0.707 0.769	0.602 0.561 0.621		
DJIA NIKKEI	1.772** 0.916*	1.027** 0.438	1.034** 0.363	5.755 2.742	4.200 1.615	4.069 1.609	6.7E-05 0.017	2.3E-03 0.140	1.6E-03 0.130	0.718 0.366	0.662 0.225	0.580 0.177		
S&P 500 NASDAQ	1.439** 0.681*	0.774** 0.367*	0.807** 0.368*	5.134 2.836	3.517 2.761	3.681 2.734	1.9E-04 0.014	6.5E-03 0.022	3.1E-03 0.018	0.670 0.382	0.579 0.459	0.530 0.384		
CAC DAX RUSSIA HANG SENG COLOMBO AUSTRALIA	1.112** 0.898** 0.274* 0.794** 0.358 1.766**	0.595* 0.549** 0.104 0.458* -0.350 1.191*	0.587** 0.472* 0.102 0.322 -0.211 0.959*	4.357 3.610 2.632 3.306 1.095 6.128	3.122 3.960 1.452 2.648 -1.675 3.083	3.120 2.969 1.520 1.769 -1.092 2.392	7.8E-04 3.2E-03 0.023 5.7E-03 0.290 3.6E-05	0.012 3.3E-03 0.190 0.027 0.130 0.013	8.9E-03 0.012 0.160 0.100 0.300 0.034	$\begin{array}{c} 0.593 \\ 0.501 \\ 0.386 \\ 0.457 \\ 0.084 \\ 0.743 \end{array}$	0.520 0.635 0.232 0.438 0.238 0.514	0.448 0.423 0.188 0.207 0.090 0.323		
CANADA BRAZIL SOUTH AFRICA FTSE WORLD	0.598* 0.598* 0.791 1 434**	0.309 0.309 0.343 0.778*	0.228 0.228 0.247 0.73*	6.179 2.496 1.707 5.292	5.648 2.046 1.036 2.747	2.392 3.786 1.468 0.896 2.520	3.3E-05 0.028 0.110 1.5E-04	3.1E-04 0.075 0.330 0.023	2.6E-03 0.170 0.390 0.027	0.746 0.342 0.183 0.683	0.314 0.780 0.344 0.107 0.456	0.525 0.544 0.164 0.063 0.346		

Table P.12 (Continued)
The Table P.12 time series regression slopes, similar to the market price returns results in Table P.11, show that the selected equity stock indices Historical Value-at-Risk Actual variable exhibits statistically significant explanatory power to the samples and the control group Historical Value-at-Risk Actual variable for the three time periods tested.

Examining the significant regression slopes, the samples and the control group exhibit greater magnitude, and marginally higher significance for the time period 1994 to 2008 than for the time periods 1994 to 2004 and 1994 to 2007.

This suggests that the Historical Value-at-Risk Actual for selected equity indices exhibited a smaller change in relation to the samples and the control group Historical Value-at-Risk Actual variable after the 2005 accounting change compared to before.

P.10 Change in LIBOR Rates and Market Price Return

The regression analysis for the change in selected London Interbank Offered Rates (LIBOR) and the market price return is presented for the time range 1994 to 2008 in Table P.13. In these regressions, the market price return variable is tested as the dependent variable, and the change in the selected LIBOR rates, for the range *overnight* (O/N) to 1 year, are individually tested as the independent variable.

Table P.13Time Series Regression Analysis for Change in LIBOR Rates and
Market Price Return

Time Series Regression

Table P.13 Panels A to C show the results for the time series regressions that test the change in selected LIBOR rates and the market price return by applying the regression specified in equation (O.13). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; and, Panel C presents results for the Primary sample.

Time Series Regression Table Description

The Table P.13 columns represent the following:

The LIBOR Rate column represents the samples and the control group result panels for the selected rates.

The Rates and Returns Time Periods column represents the following:

The Total Period column represents the time series regression results for the year range 1994 to 2008, with a single year represented by the number of days from 1st January to 31st December. The Sub-Periods column represents the time series regression results for the year range 1994 to 2004, and for 1994 to 2007, with a single year represented by the number of days from 1st January to 31st December.

The total and sub-period columns present the coefficients and related statistics for the regression specified in equation (O.13), and follows the model:

 $dM_{N,t} = a + b \ dLIBOR_t + e_{N,t}$

Where: $dM_{N,t}$ is the log Market Price return $(M_{t-1,t})$ for N firms at time t, $dLIBOR_t$ is the change in the selected LIBOR rate $(LIBOR_{t-1,t})$ at time t, and $e_{N,t}$ is the regression error term.

_							- (
			Rates and Returns Time Periods													
	LIBOP Pata	Total Period	Sub-]	Periods	Total Period	Sub-	Sub-Periods Tota Perio		Sub-	Periods	Total Period	Sub-	Periods			
		1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007			
			b	b		t			р			R^2				
			PA	ANEL A:	LIBO	R Rates vs.										
	Over Night 1 Month 3 Months 6 Months 1 Year	0.451** 0.645** 0.716* 0.7* 0.612	$\begin{array}{c} 0.240 \\ 0.667 \\ 0.667 \\ 0.663 \\ 0.467 \end{array}$	0.188 0.339 0.370 0.335 0.231	3.167 3.217 2.806 2.419 1.870	0.706 1.727 1.933 1.843 1.225	0.614 1.091 1.257 1.105 0.733	7.4E-03 6.7E-03 0.015 0.031 0.084	0.500 0.120 0.085 0.098 0.250	$\begin{array}{c} 0.550 \\ 0.300 \\ 0.230 \\ 0.290 \\ 0.480 \end{array}$	0.435 0.443 0.377 0.310 0.212	$\begin{array}{c} 0.052 \\ 0.249 \\ 0.293 \\ 0.274 \\ 0.143 \end{array}$	$\begin{array}{c} 0.030 \\ 0.090 \\ 0.116 \\ 0.092 \\ 0.043 \end{array}$			
	Over Night 1 Month 3 Months 6 Months 1 Year	0.290 0.492* 0.551* 0.542 0.480	$\begin{array}{c} 0.049 \\ 0.411 \\ 0.411 \\ 0.409 \\ 0.294 \end{array}$	0.042 0.360 0.351 0.319 0.239	2.012 2.597 2.331 0.060 1.649	0.149 1.029 1.128 0.305 0.779	0.134 1.166 1.181 0.315 0.759	0.065 0.022 0.036 0.060 0.120	0.880 0.330 0.290 0.310 0.460	$\begin{array}{c} 0.900 \\ 0.270 \\ 0.260 \\ 0.320 \\ 0.460 \end{array}$	$\begin{array}{c} 0.237 \\ 0.342 \\ 0.295 \\ 0.246 \\ 0.1729 \end{array}$	$\begin{array}{c} 0.002 \\ 0.105 \\ 0.124 \\ 0.116 \\ 0.063 \end{array}$	$\begin{array}{c} 0.001 \\ 0.102 \\ 0.104 \\ 0.084 \\ 0.046 \end{array}$			
				PAN	EL C:	LIBOR F	Rates vs. P	rimary Samp	ole Return							
	Over Night 1 Month 3 Months 6 Months 1 Year	0.697** 0.871** 0.898* 0.836* 0.716	$\begin{array}{c} 0.247 \\ 0.565 \\ 0.585 \\ 0.540 \\ 0.333 \end{array}$	0.156 0.117 0.181 0.121 0.017	4.619 3.487 2.668 2.163 1.651	0.829 1.631 1.906 1.645 0.957	$\begin{array}{c} 0.543 \\ 0.387 \\ 0.627 \\ 0.409 \\ 0.056 \end{array}$	4.8E-04 4.0E-03 0.019 0.050 0.120	$\begin{array}{c} 0.430 \\ 0.140 \\ 0.089 \\ 0.130 \\ 0.360 \end{array}$	0.600 0.710 0.540 0.690 0.960	0.621 0.483 0.354 0.265 0.173	$\begin{array}{c} 0.071 \\ 0.228 \\ 0.288 \\ 0.231 \\ 0.092 \end{array}$	0.024 0.012 0.032 0.014 2.7E-04			

Table P.13 (Continued)

The Table P.13 time series regression slopes show that some selected LIBOR rate changes exhibit statistically significant explanatory power to the samples and the control group market price returns for the time period 1994 to 2008. The Primary and Secondary sample slopes show statistical significance for all the selected rates other than the 1 year rate. The Control group shows statistical significance for 1 and 3 month rates only.

Examining the significant regression slopes, the Primary and Secondary sample exhibits greater magnitude, statistical significance, and model strength when compared to the Control group for the 1 and 3 month LIBOR rates.

The statistically significant results suggest that the change in LIBOR rates reacted in relation to the Primary and Secondary sample market price return variable after the 2005 accounting change compared to before, than the Control group reactions.

P.11 LIBOR Value-at-Risk Actual and Historical Value-at-Risk Actual

The regression analysis for selected LIBOR Historical Value-at-Risk Actual and market price return Historical Value-at-Risk Actual is presented for the time range 1994 to 2008 in Table P.14. In these regressions, the Historical Value-at-Risk Actual variable is tested as the dependent variable, and the LIBOR Historical Value-at-Risk Actual variables for the range *over-night* (O/N) to 1 year are individually tested as the independent variable.

Table P.14Time Series Regression Analysis for LIBOR Rates Historical Value-
at-Risk Actual and Historical Value-at-Risk Actual

Time Series Regression

Table P.14 Panels A to C show the results for the time series regressions that test the selected LIBOR rate Value-at-Risk Actual and Historical Value-at-Risk Actual by applying the regression specified in equation (O.14). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; and, Panel C presents results for the Primary sample.

Time Series Regression Table Description

The Table P.14 columns represent the following:

The LIBOR Rate column represents the samples and the control group result panels for the selected rates.

The V Time Periods column represents the following:

The Total Period column represents the time series regression results for the year range 1994 to 2008, with a single year represented by the number of days from 1st January to 31st December. The Sub-Periods column represents the time series regression results for the year range 1994 to 2004, and for 1994 to 2007, with a single year represented by the number of days from 1st January to 31st December.

The total and sub-period columns present the coefficients and related statistics for the regression specified in equation (O.14), and follows the model:

 $V_{N,t} = a + b V LIBOR_t + e_{N,t}$

Where: $V_{N,t}$ is the Value-at-Risk (V_t) for N firms at time t, $VLIBOR_t$ is the Value-at-Risk for the selected LIBOR rate at time t, and $e_{N,t}$ is the regression error term.

Table 1.14 (Continued)													
						VTin	ne Periods						
	Total Period	Sub-	Periods	Total Period	Sub-	Periods	Total Period	Sub-	Periods	Total Period	Sub-]	Periods	
LIBOR Rate	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	
		b			t			р		R^2			
]	PANEL A:	LIBOR Rates V vs. Primary and Secondary Sample V									
Over Night 1 Month 3 Months 6 Months 1 Year	0.739** 0.551** 0.7** 0.687** 0.578*	$\begin{array}{c} 0.460 \\ 0.326 \\ 0.436 \\ 0.390 \\ 0.285 \end{array}$	0.459 0.369 0.467 0.422 0.316	3.416 3.616 3.708 3.479 2.900	1.245 0.957 1.268 1.272 1.142	1.716 1.227 1.532 1.545 1.400	4.6E-03 3.1E-03 2.6E-03 4.1E-03 0.012	$\begin{array}{c} 0.240 \\ 0.360 \\ 0.240 \\ 0.240 \\ 0.280 \end{array}$	0.110 0.240 0.150 0.150 0.190	$\begin{array}{c} 0.473 \\ 0.501 \\ 0.514 \\ 0.482 \\ 0.393 \end{array}$	$\begin{array}{c} 0.147 \\ 0.092 \\ 0.152 \\ 0.152 \\ 0.127 \end{array}$	$\begin{array}{c} 0.197 \\ 0.111 \\ 0.164 \\ 0.166 \\ 0.140 \end{array}$	
			PA	NEL B:	LIBOI	R Rates V	vs. Control G	roup V					
Over Night 1 Month 3 Months 6 Months 1 Year	0.545** 0.379* 0.477* 0.454* 0.347	$\begin{array}{c} 0.376 \\ 0.431 \\ 0.450 \\ 0.344 \\ 0.185 \end{array}$	0.491 0.487 0.497 0.387 0.220	3.104 2.900 2.915 0.021 1.996	1.168 1.583 1.580 0.224 0.831	2.057 1.869 1.808 0.155 1.009	8.4E-03 0.012 0.021 0.067	$\begin{array}{c} 0.270 \\ 0.150 \\ 0.150 \\ 0.220 \\ 0.430 \end{array}$	$\begin{array}{c} 0.062 \\ 0.086 \\ 0.096 \\ 0.150 \\ 0.330 \end{array}$	0.426 0.393 0.395 0.348 0.235	0.132 0.218 0.217 0.159 0.071	0.261 0.225 0.214 0.161 0.078	
			PAN	IEL C:	LIBOR	Rates V v	s. Primary S	ample V					
Over Night 1 Month 3 Months 6 Months 1 Year	0.838* 0.786** 0.988** 0.989** 0.863**	0.168 0.296 0.472 0.497 0.425	0.162 0.276 0.437 0.475 0.425*	2.797 4.718 4.734 4.581 3.923	$\begin{array}{c} 0.445 \\ 0.910 \\ 1.482 \\ 1.843 \\ 2.011 \end{array}$	0.587 0.957 1.530 1.942 2.208	0.015 4.0E-04 3.9E-04 5.2E-04 1.7E-03	$\begin{array}{c} 0.670 \\ 0.390 \\ 0.170 \\ 0.098 \\ 0.075 \end{array}$	0.570 0.360 0.150 0.076 0.047	0.376 0.631 0.633 0.617 0.542	$\begin{array}{c} 0.022 \\ 0.084 \\ 0.196 \\ 0.274 \\ 0.310 \end{array}$	$\begin{array}{c} 0.028 \\ 0.071 \\ 0.163 \\ 0.239 \\ 0.289 \end{array}$	

 Table P.14 (Continued)

The Table P.14 time series regression slopes show the selected LIBOR rate Historical Value-at-Risk variable exhibits statistically significant explanatory power to the samples and the control group Historical Value-at-Risk Actual variable for the time period 1994 to 2008. The Primary and Secondary sample slopes show statistical significance for all the selected rates. The Control group also shows statistical significance for all the selected rates other than the 1 year rate.

Examining the significant regression slopes, the Primary and Secondary sample exhibits greater magnitude, statistical significance, and model strength when compared to the Control group.

This suggests that the Historical Value-at-Risk Actual for the selected LIBOR rates reacted with a greater magnitude in relation to the Primary and Secondary sample Historical Value-at-Risk Actual variable after the 2005 accounting change compared to before, than the Control group reactions.

P.12 Change in GILT Rates and Market Price Return

The regression analysis for the change in selected GILT Government bond benchmark rates and the market price return is presented for the time range 1994 to 2008 in Table P.15. In these regressions, the market price return variable is tested as the dependent variable, and the change in the GILT Government bond benchmark rate variables for maturities in the range 2.5 to 25 years are individually tested as the independent variable.

Table P.15Time Series Regression Analysis for Change in GILT Rates and
Market Price Return

Time Series Regression

Table P.15 Panels A1 to C2 show the results for the time series regressions that test the change in selected GILT rates and the market price return by applying the regression specified in equation (O.13).

In Table P.15 the change in the GILT Government bond benchmark rates:- for the Primary and Secondary sample, Panel A1 presents results for the maturity range 2.5 to 14.5 year rates and Panel A2 presents results for the maturity range 15 to 25 year rates; for the Control group, Panel B1 presents results for the maturity range 2.5 to 14.5 year rates and Panel B2 presents results for the maturity range 2.5 to 14.5 year rates and Panel C1 presents results for the maturity range 15 to 25 year rates.

Time Series Regression Table Description

The Table P.15 columns represent the following:

The GILT Rate Benchmark Year column represents bond maturities.

The Rates and Returns Time Periods column represents the following:

The Total Period column represents the time series regression results for the year range 1994 to 2008, with a single year represented by the number of days from 1st January to 31st December. The Sub-Periods column represents the time series regression results for the year range 1994 to 2004, and for 1994 to 2007, with a single year represented by the number of days from 1st January to 31st December.

The total and sub-period columns present the coefficients and related statistics for the regression specified in equation (0.13), and follows the model:

$dM_{N,t} = a + b \ dGILTS_t + e_{N,t}$

Where: $dM_{N,t}$ is the log Market Price return $(M_{t-1,t})$ for N firms at time t, $dGILTS_t$ is the change in the selected GILT rate $(GILT_{t-1,t})$ at time t, and $e_{N,t}$ is the regression error term.

					to 14.5 y	ears						
					Rate	es and Return	ns Time Period	S				
GILT Rate	Total Period	Sub-	Periods	Total Period	Sub-P	eriods	Total Period	Sub-P	eriods	Total Period	Sub-P	eriods
Benchmark Year	1994- 2008	1994- 2004 b	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008		1994- 2007
GILT 2.5 GILT 3 GILT 3.5 GILT 4 GILT 4.5	0.100 -0.153 -0.142 -0.132 -0.129	0.100 0.107 0.105 0.108 0.109	0.100 0.133 0.132 0.135 0.133	0.364 -0.562 -0.530 -0.462 -0.460	0.364 0.350 0.364 0.356 0.382	0.364 0.569 0.587 0.576 0.582	0.728 0.587 0.607 0.653 0.653	P 0.728 0.738 0.727 0.732 0.711	0.728 0.583 0.570 0.577 0.571	0.022 0.031 0.025 0.019 0.016	0.022 0.020 0.019 0.018 0.016	0.022 0.035 0.033 0.032 0.027
GILT 5 GILT 5.5 GILT 6 GILT 6.5 GILT 7.5 GILT 7.5 GILT 8 GILT 8.5 GILT 9 GILT 9.5	-0.125 -0.125 -0.127 -0.133 -0.142 -0.153 -0.166 -0.180 -0.195 -0.210	$\begin{array}{c} 0.102\\ 0.092\\ 0.081\\ 0.069\\ 0.056\\ 0.044\\ 0.031\\ 0.018\\ 0.004\\ -0.009 \end{array}$	$\begin{array}{c} 0.127\\ 0.118\\ 0.108\\ 0.097\\ 0.086\\ 0.075\\ 0.064\\ 0.053\\ 0.042\\ 0.030\\ \end{array}$	$\begin{array}{c} -0.432\\ -0.421\\ -0.423\\ 0.669\\ -0.461\\ -0.492\\ -0.528\\ -0.567\\ -0.608\\ -0.650\end{array}$	$\begin{array}{c} 0.348\\ 0.309\\ 0.267\\ 0.827\\ 0.182\\ 0.139\\ 0.096\\ 0.055\\ 0.013\\ -0.027\\ \end{array}$	$\begin{array}{c} 0.541 \\ 0.494 \\ 0.446 \\ 0.699 \\ 0.348 \\ 0.299 \\ 0.252 \\ 0.206 \\ 0.160 \\ 0.115 \end{array}$	$\begin{array}{c} 0.673 \\ 0.681 \\ 0.679 \\ 0.669 \\ 0.652 \\ 0.631 \\ 0.606 \\ 0.580 \\ 0.554 \\ 0.527 \end{array}$	$\begin{array}{c} 0.736\\ 0.764\\ 0.795\\ 0.827\\ 0.860\\ 0.893\\ 0.925\\ 0.958\\ 0.990\\ 0.979\\ \end{array}$	$\begin{array}{c} 0.599\\ 0.630\\ 0.664\\ 0.699\\ 0.734\\ 0.770\\ 0.805\\ 0.840\\ 0.875\\ 0.910\\ \end{array}$	$\begin{array}{c} 0.014\\ 0.013\\ 0.014\\ 0.015\\ 0.016\\ 0.018\\ 0.021\\ 0.024\\ 0.028\\ 0.031\\ \end{array}$	$\begin{array}{c} 0.013\\ 0.010\\ 0.008\\ 0.006\\ 0.004\\ 0.002\\ 0.001\\ 3.3E-04\\ 2.0E-05\\ 8.2E-05\\ \end{array}$	$\begin{array}{c} 0.024\\ 0.020\\ 0.016\\ 0.013\\ 0.010\\ 0.007\\ 0.005\\ 0.004\\ 0.002\\ 0.001\\ \end{array}$
GILT 10 GILT 10.5 GILT 11 GILT 11.5 GILT 12 GILT 12.5 GILT 13 GILT 13.5 GILT 14 GILT 14.5	$\begin{array}{c} -0.240\\ -0.240\\ -0.255\\ -0.269\\ -0.331\\ -0.344\\ -0.355\\ -0.364\\ -0.373\\ -0.379\end{array}$	-0.037 -0.037 -0.051 -0.065 -0.113 -0.127 -0.142 -0.156 -0.171 -0.185	$\begin{array}{c} 0.007\\ 0.007\\ -0.004\\ -0.016\\ -0.047\\ -0.060\\ -0.072\\ -0.084\\ -0.097\\ -0.109 \end{array}$	$\begin{array}{c} -0.692\\ -0.733\\ -0.772\\ -0.808\\ -0.953\\ -0.982\\ -1.007\\ -1.027\\ -1.043\\ -1.053\end{array}$	$\begin{array}{c} -0.067 \\ -0.107 \\ -0.146 \\ -0.185 \\ -0.329 \\ -0.368 \\ -0.407 \\ -0.445 \\ -0.484 \\ -0.523 \end{array}$	$\begin{array}{c} 0.071 \\ 0.028 \\ -0.016 \\ -0.059 \\ -0.177 \\ -0.220 \\ -0.264 \\ -0.307 \\ -0.351 \\ -0.395 \end{array}$	$\begin{array}{c} 0.501 \\ 0.477 \\ 0.454 \\ 0.434 \\ 0.358 \\ 0.344 \\ 0.332 \\ 0.323 \\ 0.316 \\ 0.311 \end{array}$	$\begin{array}{c} 0.948\\ 0.917\\ 0.887\\ 0.749\\ 0.721\\ 0.694\\ 0.666\\ 0.640\\ 0.614\\ \end{array}$	$\begin{array}{c} 0.944\\ 0.978\\ 0.988\\ 0.954\\ 0.863\\ 0.829\\ 0.796\\ 0.764\\ 0.732\\ 0.700\\ \end{array}$	$\begin{array}{c} 0.036\\ 0.040\\ 0.044\\ 0.048\\ 0.065\\ 0.069\\ 0.072\\ 0.075\\ 0.077\\ 0.079\\ \end{array}$	$\begin{array}{c} 0.001\\ 0.001\\ 0.002\\ 0.004\\ 0.012\\ 0.015\\ 0.018\\ 0.022\\ 0.025\\ 0.029\\ \end{array}$	$\begin{array}{c} 4.2E\text{-}04\\ 6.4E\text{-}05\\ 2.1E\text{-}05\\ 2.9E\text{-}04\\ 0.003\\ 0.004\\ 0.006\\ 0.008\\ 0.010\\ 0.013\\ \end{array}$

Table P.15 (Continued)
PANEL A1: Time Series Regression Analysis of the Primary and Secondary Sample for Change in GILT Rates and Market Price Return from 2.5 years

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels.

525

					J	J ~								
	Rates and Returns Time Periods													
GILT Rate	Total Period	Sub-	Periods	eriods Total Period		Sub-Periods		Sub-P	eriods	Total Period	Sub-Periods			
Benchmark Year	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007		
	b			<i>t</i>			р				R^2	R^2		
GILT 15 GILT 15.5 GILT 16.5 GILT 16.5 GILT 17 GILT 17.5 GILT 18 GILT 18.5 GILT 19 GILT 19.5	-0.384 -0.387 -0.388 -0.387 -0.384 -0.378 -0.371 -0.360 -0.348 -0.333	-0.199 -0.213 -0.227 -0.241 -0.254 -0.267 -0.280 -0.292 -0.304 -0.315	-0.122 -0.134 -0.147 -0.159 -0.171 -0.183 -0.195 -0.206 -0.217 -0.228	-1.059 -1.059 -1.054 -1.044 -1.028 -1.006 -0.980 -0.948 -0.911 -0.869	-0.562 -0.600 -0.639 -0.678 -0.717 -0.756 -0.795 -0.834 -0.873 -0.911	-0.439 -0.483 -0.527 -0.571 -0.614 -0.658 -0.702 -0.745 -0.788 -0.831	$\begin{array}{c} 0.309\\ 0.309\\ 0.311\\ 0.316\\ 0.323\\ 0.333\\ 0.345\\ 0.361\\ 0.379\\ 0.401\\ \end{array}$	$\begin{array}{c} 0.588\\ 0.563\\ 0.539\\ 0.515\\ 0.491\\ 0.469\\ 0.447\\ 0.426\\ 0.405\\ 0.386\end{array}$	$\begin{array}{c} 0.669\\ 0.638\\ 0.608\\ 0.579\\ 0.550\\ 0.523\\ 0.496\\ 0.471\\ 0.446\\ 0.422 \end{array}$	$\begin{array}{c} 0.079\\ 0.079\\ 0.079\\ 0.077\\ 0.075\\ 0.072\\ 0.069\\ 0.065\\ 0.060\\ 0.055\\ \end{array}$	$\begin{array}{c} 0.034\\ 0.039\\ 0.043\\ 0.049\\ 0.054\\ 0.060\\ 0.066\\ 0.072\\ 0.078\\ 0.085\end{array}$	$\begin{array}{c} 0.016\\ 0.019\\ 0.023\\ 0.026\\ 0.031\\ 0.035\\ 0.039\\ 0.044\\ 0.049\\ 0.054\\ \end{array}$		
GILT 20 GILT 20.5 GILT 21 GILT 21.5 GILT 22.5 GILT 23 GILT 23 GILT 23 GILT 24 GILT 24.5 GILT 25	$\begin{array}{c} -0.337\\ -0.320\\ -0.428\\ -0.399\\ -0.346\\ -0.302\\ -0.258\\ -0.216\\ -0.162\\ -0.119\\ -0.079\end{array}$	-0.332 -0.344 -0.485 -0.496 -0.756 -0.758 -0.757 -0.754 -0.626 -0.622 -0.617	-0.248 -0.259 -0.327 -0.339 -0.420 -0.432 -0.442 -0.451 -0.440 -0.443 -0.444	$\begin{array}{c} -0.820\\ -0.778\\ -0.860\\ -0.803\\ -0.550\\ -0.487\\ -0.423\\ -0.360\\ -0.256\\ -0.192\\ -0.130\end{array}$	-0.872 -0.911 -0.998 -1.039 -1.057 -1.096 -1.132 -1.165 -0.785 -0.798 -0.811	-0.842 -0.885 -0.918 -0.964 -0.964 -1.099 -1.054 -1.096 -1.099 -1.034 -1.059	$\begin{array}{c} 0.430\\ 0.453\\ 0.410\\ 0.441\\ 0.597\\ 0.639\\ 0.683\\ 0.728\\ 0.807\\ 0.854\\ 0.901 \end{array}$	$\begin{array}{c} 0.412\\ 0.393\\ 0.357\\ 0.339\\ 0.350\\ 0.325\\ 0.321\\ 0.309\\ 0.515\\ 0.508\\ 0.503\end{array}$	$\begin{array}{c} 0.420\\ 0.397\\ 0.382\\ 0.360\\ 0.367\\ 0.346\\ 0.327\\ 0.309\\ 0.359\\ 0.348\\ 0.338\end{array}$	$\begin{array}{c} 0.058\\ 0.052\\ 0.069\\ 0.061\\ 0.036\\ 0.029\\ 0.022\\ 0.016\\ 0.011\\ 0.006\\ 0.003\\ \end{array}$	$\begin{array}{c} 0.098\\ 0.106\\ 0.142\\ 0.152\\ 0.218\\ 0.231\\ 0.242\\ 0.253\\ 0.236\\ 0.242\\ 0.247\\ \end{array}$	$\begin{array}{c} 0.066\\ 0.073\\ 0.086\\ 0.094\\ 0.117\\ 0.127\\ 0.137\\ 0.147\\ 0.169\\ 0.176\\ 0.183\\ \end{array}$		

PANEL A2: Time Series Regression Analysis of the Primary and Secondary Sample for Change in GILT Rates and Market Price Return from 15 vears to 25 years

Table P.15 (Continued)

	Rates and Returns Time Periods													
GILT Rate	Total Period	Sub-	Periods	Total Period	Sub-P	eriods	Total Period	Sub-P	eriods	Total Period	Sub-P	eriods		
Benchmark Year	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007		
		b			t			р			R^2			
GILT 2.5 GILT 3 GILT 3.5 GILT 4 GILT 4.5	0.130 -0.112 -0.075 -0.082 -0.094	$\begin{array}{c} 0.130 \\ 0.113 \\ 0.136 \\ 0.120 \\ 0.101 \end{array}$	$\begin{array}{c} 0.130 \\ 0.092 \\ 0.121 \\ 0.106 \\ 0.085 \end{array}$	0.500 -0.465 -0.310 -0.320 -0.383	$\begin{array}{c} 0.500 \\ 0.387 \\ 0.485 \\ 0.407 \\ 0.373 \end{array}$	$\begin{array}{c} 0.500 \\ 0.389 \\ 0.519 \\ 0.434 \\ 0.367 \end{array}$	$\begin{array}{c} 0.635\\ 0.652\\ 0.762\\ 0.755\\ 0.708\end{array}$	$\begin{array}{c} 0.635 \\ 0.712 \\ 0.642 \\ 0.696 \\ 0.718 \end{array}$	$\begin{array}{c} 0.635 \\ 0.706 \\ 0.615 \\ 0.673 \\ 0.720 \end{array}$	$\begin{array}{c} 0.040 \\ 0.021 \\ 0.009 \\ 0.009 \\ 0.011 \end{array}$	$\begin{array}{c} 0.040 \\ 0.024 \\ 0.033 \\ 0.023 \\ 0.015 \end{array}$	$\begin{array}{c} 0.040 \\ 0.017 \\ 0.026 \\ 0.019 \\ 0.011 \end{array}$		
GILT 5 GILT 5.5 GILT 6 GILT 6.5 GILT 7.5 GILT 7.5 GILT 8 GILT 8.5 GILT 9 GILT 9.5	-0.107 -0.121 -0.135 -0.150 -0.165 -0.180 -0.196 -0.211 -0.226 -0.241	$\begin{array}{c} 0.079\\ 0.058\\ 0.038\\ 0.021\\ 0.006\\ -0.007\\ -0.019\\ -0.030\\ -0.039\\ -0.049 \end{array}$	$\begin{array}{c} 0.063\\ 0.041\\ 0.021\\ 0.002\\ -0.015\\ -0.031\\ -0.046\\ -0.060\\ -0.073\\ -0.086\\ \end{array}$	$\begin{array}{c} -0.423\\ -0.468\\ -0.517\\ 0.580\\ -0.620\\ -0.671\\ -0.722\\ -0.772\\ -0.820\\ -0.867\end{array}$	$\begin{array}{c} 0.284\\ 0.204\\ 0.134\\ 0.943\\ 0.021\\ -0.024\\ -0.063\\ -0.097\\ -0.127\\ -0.155\end{array}$	0.267 0.172 0.086 0.993 -0.061 -0.124 -0.181 -0.233 -0.281 -0.327	$\begin{array}{c} 0.679\\ 0.647\\ 0.614\\ 0.580\\ 0.546\\ 0.514\\ 0.483\\ 0.454\\ 0.427\\ 0.402\\ \end{array}$	$\begin{array}{c} 0.783 \\ 0.843 \\ 0.997 \\ 0.943 \\ 0.984 \\ 0.982 \\ 0.951 \\ 0.925 \\ 0.902 \\ 0.880 \end{array}$	$\begin{array}{c} 0.794 \\ 0.866 \\ 0.933 \\ 0.993 \\ 0.952 \\ 0.903 \\ 0.859 \\ 0.820 \\ 0.783 \\ 0.749 \end{array}$	$\begin{array}{c} 0.014\\ 0.017\\ 0.020\\ 0.024\\ 0.029\\ 0.034\\ 0.039\\ 0.044\\ 0.049\\ 0.055 \end{array}$	$\begin{array}{c} 0.009\\ 0.005\\ 0.002\\ 0.001\\ 4.9E\text{-}05\\ 6.3E\text{-}05\\ 4.4E\text{-}04\\ 0.001\\ 0.002\\ 0.003\\ \end{array}$	$\begin{array}{c} 0.006\\ 0.002\\ 0.001\\ 6.1E\text{-}06\\ 3.1E\text{-}04\\ 0.001\\ 0.003\\ 0.005\\ 0.007\\ 0.009\\ \end{array}$		
GILT 10 GILT 10.5 GILT 11 GILT 11.5 GILT 12.5 GILT 12.5 GILT 13.5 GILT 14 GILT 14.5	-0.269 -0.269 -0.283 -0.296 -0.346 -0.358 -0.369 -0.379 -0.389 -0.398	-0.067 -0.067 -0.076 -0.085 -0.114 -0.123 -0.132 -0.142 -0.153 -0.164	$\begin{array}{c} -0.110\\ -0.110\\ -0.122\\ -0.134\\ -0.173\\ -0.185\\ -0.197\\ -0.209\\ -0.221\\ -0.233\end{array}$	-0.913 -0.956 -0.998 -1.038 -1.191 -1.224 -1.254 -1.281 -1.305 -1.327	$\begin{array}{c} -0.181\\ -0.206\\ -0.231\\ -0.257\\ -0.340\\ -0.364\\ -0.389\\ -0.416\\ -0.444\\ -0.474\end{array}$	-0.370 -0.412 -0.454 -0.495 -0.636 -0.675 -0.714 -0.755 -0.796 -0.838	$\begin{array}{c} 0.378\\ 0.356\\ 0.337\\ 0.318\\ 0.255\\ 0.243\\ 0.232\\ 0.223\\ 0.214\\ 0.207\\ \end{array}$	$\begin{array}{c} 0.860\\ 0.841\\ 0.822\\ 0.803\\ 0.742\\ 0.724\\ 0.706\\ 0.687\\ 0.667\\ 0.647\end{array}$	$\begin{array}{c} 0.718\\ 0.687\\ 0.658\\ 0.630\\ 0.537\\ 0.513\\ 0.489\\ 0.465\\ 0.442\\ 0.419\\ \end{array}$	$\begin{array}{c} 0.060\\ 0.066\\ 0.071\\ 0.076\\ 0.098\\ 0.103\\ 0.108\\ 0.112\\ 0.116\\ 0.119\\ \end{array}$	$\begin{array}{c} 0.004\\ 0.005\\ 0.006\\ 0.007\\ 0.013\\ 0.015\\ 0.017\\ 0.019\\ 0.021\\ 0.024 \end{array}$	$\begin{array}{c} 0.011\\ 0.014\\ 0.017\\ 0.020\\ 0.033\\ 0.037\\ 0.041\\ 0.045\\ 0.050\\ 0.055\\ \end{array}$		

Table P.15 (Continued)

PANEL B1: Time Series Regression Analysis of the Control Group for Change in GILT Rates and Market Price Return from 2.5 years to 14.5 Years

PANEL B2: Time	Series Regre	ession Ana	lysis of the	Control Gro	up for Cha	ange in GII	LT Rates and	l Market I	Price Retu	rn from 15 y	ears to 25	years
					Rate	es and Return	ns Time Period	ls				
GILT Rate	Total Period	Sub-	Periods	Total Period	Total Period Sub-Periods 1994- 1994- 1994- 2008 2004 2007		Total Period	Sub-Periods 1994- 1994- 2004 2007		Total Period	Sub-P	eriods
Benchmark Year	1994- 2008	1994- 2004	1994- 2007	1994- 2008			1994- 2008			1994- 2008	1994- 2004	1994- 2007
		b		<i>t</i>			р			R^2		
GILT 15 GILT 15.5 GILT 16.5 GILT 16.5 GILT 17 GILT 17.5 GILT 18 GILT 18.5 GILT 19 GILT 19.5	-0.406 -0.413 -0.419 -0.424 -0.427 -0.429 -0.430 -0.430 -0.430 -0.427 -0.423	-0.175 -0.187 -0.199 -0.211 -0.224 -0.237 -0.250 -0.263 -0.263 -0.276 -0.288	$\begin{array}{c} -0.246\\ -0.258\\ -0.271\\ -0.284\\ -0.296\\ -0.309\\ -0.321\\ -0.333\\ -0.345\\ -0.356\end{array}$	-1.345 -1.360 -1.371 -1.379 -1.383 -1.383 -1.380 -1.372 -1.360 -1.344	$\begin{array}{c} -0.505\\ -0.537\\ -0.571\\ -0.607\\ -0.644\\ -0.683\\ -0.723\\ -0.764\\ -0.807\\ -0.851\end{array}$	-0.881 -0.925 -0.970 -1.015 -1.062 -1.110 -1.159 -1.209 -1.259 -1.310	$\begin{array}{c} 0.202\\ 0.197\\ 0.194\\ 0.191\\ 0.190\\ 0.190\\ 0.191\\ 0.193\\ 0.197\\ 0.202\\ \end{array}$	$\begin{array}{c} 0.626\\ 0.604\\ 0.582\\ 0.559\\ 0.535\\ 0.512\\ 0.488\\ 0.464\\ 0.441\\ 0.417\end{array}$	$\begin{array}{c} 0.396\\ 0.373\\ 0.351\\ 0.330\\ 0.309\\ 0.289\\ 0.269\\ 0.250\\ 0.232\\ 0.215\\ \end{array}$	$\begin{array}{c} 0.122\\ 0.125\\ 0.126\\ 0.128\\ 0.128\\ 0.128\\ 0.128\\ 0.128\\ 0.126\\ 0.125\\ 0.122\\ \end{array}$	$\begin{array}{c} 0.028\\ 0.031\\ 0.035\\ 0.039\\ 0.044\\ 0.049\\ 0.055\\ 0.061\\ 0.067\\ 0.074\\ \end{array}$	$\begin{array}{c} 0.061 \\ 0.066 \\ 0.073 \\ 0.079 \\ 0.086 \\ 0.093 \\ 0.101 \\ 0.109 \\ 0.117 \\ 0.125 \end{array}$
GILT 20 GILT 20.5 GILT 21 GILT 21.5 GILT 22.5 GILT 23 GILT 23.5 GILT 24 GILT 24.5 GILT 25	$\begin{array}{c} -0.427\\ -0.421\\ -0.446\\ -0.433\\ -0.698\\ -0.664\\ -0.627\\ -0.589\\ -0.477\\ -0.435\\ -0.394\end{array}$	-0.305 -0.318 -0.322 -0.340 -0.973 -0.974 -0.972 -0.967 -0.739 -0.732 -0.724	$\begin{array}{c} -0.370\\ -0.381\\ -0.378\\ -0.392\\ -0.753\\ -0.761\\ -0.767\\ -0.772\\ -0.696\\ -0.695\\ -0.692\end{array}$	$\begin{array}{c} -1.243\\ -1.225\\ -1.063\\ -1.035\\ -1.385\\ -1.327\\ -1.268\\ -1.207\\ -0.926\\ -0.856\\ -0.788\end{array}$	$\begin{array}{c} -0.798\\ -0.841\\ -0.650\\ -0.699\\ -1.505\\ -1.568\\ -1.630\\ -1.692\\ -1.061\\ -1.077\\ -1.092\end{array}$	-1.241 -1.291 -1.051 -1.105 -1.951 -2.024 -2.097 -2.170 -1.873 -1.906 -1.937	$\begin{array}{c} 0.240\\ 0.246\\ 0.313\\ 0.325\\ 0.203\\ 0.221\\ 0.241\\ 0.262\\ 0.390\\ 0.425\\ 0.461\\ \end{array}$	$\begin{array}{c} 0.451\\ 0.428\\ 0.540\\ 0.510\\ 0.207\\ 0.192\\ 0.178\\ 0.166\\ 0.400\\ 0.394\\ 0.389\end{array}$	$\begin{array}{c} 0.243\\ 0.226\\ 0.321\\ 0.298\\ 0.092\\ 0.083\\ 0.074\\ 0.067\\ 0.120\\ 0.115\\ 0.110\\ \end{array}$	$\begin{array}{c} 0.123\\ 0.120\\ 0.102\\ 0.097\\ 0.193\\ 0.181\\ 0.167\\ 0.154\\ 0.125\\ 0.109\\ 0.094 \end{array}$	$\begin{array}{c} 0.083\\ 0.092\\ 0.066\\ 0.075\\ 0.362\\ 0.381\\ 0.399\\ 0.417\\ 0.360\\ 0.367\\ 0.374 \end{array}$	$\begin{array}{c} 0.133\\ 0.143\\ 0.109\\ 0.119\\ 0.352\\ 0.369\\ 0.386\\ 0.402\\ 0.412\\ 0.421\\ 0.429\\ \end{array}$

Panel notes: **, * Regression	coefficients significant at the	0.01 and 0.05 two-tailed levels.
-------------------------------	---------------------------------	----------------------------------

PANEL C1: Time So	eries Regress	ion Analys	sis of the P	rimary Samp	ole for Cha Rate	nge in GII	T Rates and	l Market P ls	rice Retur	n from 2.5 y	ears to 14.	5 years	
GILT Rate	Total Period	Sub-l	Periods	Total Period	Sub-P	eriods	Total Period	Sub-P	eriods	Total Period	Sub-P	eriods	
Benchmark Year	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 1994- 2004 2007		1994- 2008	1994- 1994- 2004 2007		1994- 2008	1994- 2004	1994- 2007	
		b		t				р			R^2		
GILT 2.5 GILT 3 GILT 3.5 GILT 4 GILT 4.5	-0.032 -0.400 -0.362 -0.343 -0.326	-0.032 -0.036 0.007 0.019 0.037	-0.032 0.029 0.056 0.068 0.082	-0.131 -1.196 -1.083 -0.959 -0.915	-0.131 -0.132 0.027 0.069 0.146	-0.131 0.132 0.267 0.311 0.381	0.900 0.259 0.302 0.358 0.377	0.900 0.899 0.979 0.947 0.887	0.900 0.898 0.795 0.762 0.710	$\begin{array}{c} 0.003 \\ 0.125 \\ 0.096 \\ 0.077 \\ 0.061 \end{array}$	0.003 0.003 1.0E-04 0.001 0.002	$\begin{array}{c} 0.003 \\ 0.002 \\ 0.007 \\ 0.010 \\ 0.012 \end{array}$	
GILT 5 GILT 5.5 GILT 6 GILT 6.5 GILT 7 GILT 7.5 GILT 8 GILT 8.5 GILT 9 GILT 9.5	-0.308 -0.294 -0.284 -0.279 -0.279 -0.282 -0.288 -0.297 -0.307 -0.318	$\begin{array}{c} 0.040\\ 0.040\\ 0.039\\ 0.036\\ 0.031\\ 0.026\\ 0.020\\ 0.014\\ 0.006\\ -0.001 \end{array}$	$\begin{array}{c} 0.086\\ 0.088\\ 0.088\\ 0.088\\ 0.086\\ 0.084\\ 0.082\\ 0.079\\ 0.076\\ 0.072\\ \end{array}$	$\begin{array}{c} -0.837\\ -0.779\\ -0.739\\ 0.487\\ -0.705\\ -0.706\\ -0.714\\ -0.728\\ -0.746\\ -0.766\end{array}$	$\begin{array}{c} 0.154\\ 0.153\\ 0.145\\ 0.898\\ 0.114\\ 0.094\\ 0.071\\ 0.047\\ 0.022\\ -0.004 \end{array}$	$\begin{array}{c} 0.391 \\ 0.393 \\ 0.391 \\ 0.708 \\ 0.374 \\ 0.361 \\ 0.347 \\ 0.331 \\ 0.315 \\ 0.296 \end{array}$	$\begin{array}{c} 0.417\\ 0.450\\ 0.473\\ 0.487\\ 0.493\\ 0.493\\ 0.488\\ 0.479\\ 0.469\\ 0.457\end{array}$	$\begin{array}{c} 0.881 \\ 0.882 \\ 0.888 \\ 0.912 \\ 0.928 \\ 0.945 \\ 0.963 \\ 0.983 \\ 0.997 \end{array}$	$\begin{array}{c} 0.703\\ 0.701\\ 0.703\\ 0.708\\ 0.715\\ 0.724\\ 0.735\\ 0.746\\ 0.759\\ 0.772\\ \end{array}$	$\begin{array}{c} 0.051\\ 0.045\\ 0.040\\ 0.038\\ 0.037\\ 0.037\\ 0.038\\ 0.039\\ 0.041\\ 0.043\\ \end{array}$	$\begin{array}{c} 0.003\\ 0.003\\ 0.002\\ 0.002\\ 0.001\\ 0.001\\ 2.5E\text{-}04\\ 5.4E\text{-}05\\ 1.6E\text{-}06\end{array}$	$\begin{array}{c} 0.013\\ 0.013\\ 0.013\\ 0.012\\ 0.011\\ 0.011\\ 0.010\\ 0.009\\ 0.008\\ 0.007\\ \end{array}$	
GILT 10 GILT 10.5 GILT 11 GILT 11.5 GILT 12.5 GILT 12.5 GILT 13 GILT 13.5 GILT 14 GILT 14.5	-0.341 -0.352 -0.362 -0.405 -0.410 -0.414 -0.415 -0.414 -0.410	-0.017 -0.026 -0.035 -0.048 -0.054 -0.060 -0.067 -0.074 -0.081	$\begin{array}{c} 0.064\\ 0.064\\ 0.060\\ 0.055\\ 0.062\\ 0.058\\ 0.054\\ 0.054\\ 0.049\\ 0.044\\ 0.039 \end{array}$	$\begin{array}{c} -0.787\\ -0.807\\ -0.826\\ -0.842\\ -0.861\\ -0.865\\ -0.865\\ -0.859\\ -0.859\\ -0.850\\ -0.834\end{array}$	-0.030 -0.057 -0.084 -0.112 -0.148 -0.165 -0.182 -0.201 -0.220 -0.241	$\begin{array}{c} 0.277\\ 0.257\\ 0.236\\ 0.214\\ 0.235\\ 0.219\\ 0.202\\ 0.183\\ 0.163\\ 0.142\\ \end{array}$	$\begin{array}{c} 0.446\\ 0.434\\ 0.424\\ 0.415\\ 0.405\\ 0.403\\ 0.403\\ 0.406\\ 0.411\\ 0.419\\ \end{array}$	$\begin{array}{c} 0.976\\ 0.956\\ 0.935\\ 0.913\\ 0.886\\ 0.873\\ 0.859\\ 0.845\\ 0.831\\ 0.815\\ \end{array}$	$\begin{array}{c} 0.786\\ 0.801\\ 0.817\\ 0.834\\ 0.818\\ 0.830\\ 0.843\\ 0.858\\ 0.873\\ 0.873\\ 0.890\\ \end{array}$	$\begin{array}{c} 0.045\\ 0.048\\ 0.050\\ 0.052\\ 0.054\\ 0.054\\ 0.054\\ 0.054\\ 0.054\\ 0.053\\ 0.051\\ \end{array}$	$\begin{array}{c} 1.0E{-}04\\ 3.6E{-}04\\ 0.001\\ 0.002\\ 0.003\\ 0.004\\ 0.004\\ 0.005\\ 0.006\end{array}$	$\begin{array}{c} 0.006\\ 0.005\\ 0.005\\ 0.004\\ 0.005\\ 0.004\\ 0.003\\ 0.003\\ 0.002\\ 0.002\\ \end{array}$	

Table P.15 (Continued)

					Rate	s and Return	ns Time Period	ls				
GILT Rate	Total Period	Sub-I	Periods	iods Total Period		Sub-Periods		Sub-Periods		Total Period	Sub-F	Periods
Benchmark Year	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 1994- 2004 2007		1994- 2008	1994- 1994- 2004 2007		1994- 2008	1994- 2004	1994- 2007
	b			t		р				R^2	R^2	
GILT 15 GILT 15.5 GILT 16.5 GILT 17 GILT 17.5 GILT 18.5 GILT 18.5 GILT 19 GILT 19.5	-0.404 -0.395 -0.382 -0.367 -0.348 -0.326 -0.301 -0.272 -0.241 -0.206	-0.089 -0.097 -0.105 -0.114 -0.122 -0.131 -0.139 -0.148 -0.157 -0.165	$\begin{array}{c} 0.033\\ 0.026\\ 0.019\\ 0.012\\ 0.005\\ -0.003\\ -0.011\\ -0.018\\ -0.026\\ -0.034 \end{array}$	$\begin{array}{c} -0.814\\ -0.788\\ -0.757\\ -0.720\\ -0.678\\ -0.631\\ -0.578\\ -0.520\\ -0.458\\ -0.391\end{array}$	$\begin{array}{c} -0.262 \\ -0.284 \\ -0.308 \\ -0.332 \\ -0.356 \\ -0.382 \\ -0.408 \\ -0.435 \\ -0.462 \\ -0.490 \end{array}$	$\begin{array}{c} 0.119\\ 0.095\\ 0.070\\ 0.044\\ 0.017\\ -0.010\\ -0.038\\ -0.067\\ -0.096\\ -0.125\\ \end{array}$	$\begin{array}{c} 0.430\\ 0.445\\ 0.463\\ 0.484\\ 0.510\\ 0.539\\ 0.573\\ 0.612\\ 0.655\\ 0.702\\ \end{array}$	$\begin{array}{c} 0.799\\ 0.783\\ 0.765\\ 0.748\\ 0.730\\ 0.711\\ 0.693\\ 0.674\\ 0.655\\ 0.636\end{array}$	$\begin{array}{c} 0.907\\ 0.926\\ 0.945\\ 0.966\\ 0.986\\ 0.992\\ 0.970\\ 0.948\\ 0.925\\ 0.903\\ \end{array}$	$\begin{array}{c} 0.048\\ 0.046\\ 0.042\\ 0.038\\ 0.034\\ 0.030\\ 0.025\\ 0.020\\ 0.016\\ 0.012\\ \end{array}$	$\begin{array}{c} 0.008\\ 0.009\\ 0.010\\ 0.012\\ 0.014\\ 0.016\\ 0.018\\ 0.021\\ 0.023\\ 0.026\end{array}$	0.001 0.001 4.1E-04 1.6E-04 2.5E-05 8.4E-00 1.2E-04 3.7E-04 0.001 0.001
GILT 20 GILT 20.5 GILT 21.5 GILT 22.5 GILT 22.5 GILT 23.5 GILT 24 GILT 24.5 GILT 25	-0.229 -0.193 -0.195 -0.143 -4.9E-04 0.068 0.133 0.194 0.333 0.392 0.446	-0.209 -0.219 -0.163 -0.179 -0.482 -0.494 -0.504 -0.511 -0.284 -0.285 -0.285	$\begin{array}{c} -0.085\\ -0.096\\ -0.032\\ -0.046\\ -0.119\\ -0.136\\ -0.152\\ -0.167\\ -0.096\\ -0.101\\ -0.106\end{array}$	$\begin{array}{c} -0.419\\ -0.354\\ -0.293\\ -0.215\\ -0.001\\ 0.082\\ 0.165\\ 0.245\\ 0.401\\ 0.486\\ 0.569\end{array}$	$\begin{array}{c} -0.634 \\ -0.672 \\ -0.384 \\ -0.428 \\ -0.796 \\ -0.844 \\ -0.890 \\ -0.936 \\ -0.466 \\ -0.477 \\ -0.488 \end{array}$	$\begin{array}{c} -0.321\\ -0.362\\ -0.100\\ -0.146\\ -0.307\\ -0.356\\ -0.405\\ -0.453\\ -0.276\\ -0.296\\ -0.315\end{array}$	$\begin{array}{c} 0.683\\ 0.730\\ 0.775\\ 0.834\\ 1.000\\ 0.936\\ 0.873\\ 0.812\\ 0.703\\ 0.644\\ 0.590 \end{array}$	$\begin{array}{c} 0.546\\ 0.523\\ 0.714\\ 0.684\\ 0.471\\ 0.446\\ 0.424\\ 0.402\\ 0.687\\ 0.680\\ 0.674\end{array}$	$\begin{array}{c} 0.755\\ 0.725\\ 0.923\\ 0.887\\ 0.768\\ 0.732\\ 0.698\\ 0.665\\ 0.794\\ 0.779\\ 0.765\end{array}$	$\begin{array}{c} 0.016\\ 0.011\\ 0.009\\ 0.005\\ 4.3E\text{-}08\\ 0.001\\ 0.003\\ 0.007\\ 0.026\\ 0.038\\ 0.051\\ \end{array}$	$\begin{array}{c} 0.054\\ 0.061\\ 0.024\\ 0.030\\ 0.137\\ 0.151\\ 0.165\\ 0.180\\ 0.098\\ 0.102\\ 0.106\end{array}$	$\begin{array}{c} 0.010\\ 0.013\\ 0.001\\ 0.002\\ 0.013\\ 0.018\\ 0.023\\ 0.028\\ 0.015\\ 0.017\\ 0.019\\ \end{array}$

Table P.15 (Continued)

530

The Table P.15 time series regression slopes show that the change in selected GILT Government bond benchmark rates do not exhibit statistically significant explanatory power with the samples and the control group market price return variable.

This suggests that changes in the GILT Government bond benchmark rates reacted with the samples and the control group market price returns similarly before and after the 2005 accounting change.

P.13 GILT Rate Value-at-Risk Actual and Historical Value-at-Risk Actual

The regression analysis for selected GILT Government bond benchmark rate Historical Value-at-Risk Actual and market price return Historical Value-at-Risk Actual is presented for the time range 1994 to 2008 in Table P.16. In these regressions, the Historical Value-at-Risk Actual variable is tested as the dependent variable, and the GILT rate Value-at-Risk Actual variables for the maturities in the range 2.5 to 25 years are individually tested as the independent variable.

Table P.16Time Series Regression Analysis for GILT Rates Historical Value-
at-Risk Actual and Historical Value-at-Risk Actual

Time Series Regression

Table P.16 Panels A to C show the results for the time series regressions that test the selected GILT Government bond benchmark rate Historical Value-at-Risk Actual and Historical Value-at-Risk Actual by applying the regression specified in equation (O.14).

In Table P.16 the GILT Government bond benchmark rate Historical Value-at-Risk Actual:- for the Primary and Secondary sample, Panel A1 presents results for the maturity range 2.5 to 14.5 year rates and Panel A2 presents results for the maturity range 15 to 25 year rates; for the Control group, Panel B1 presents results for the maturity range 2.5 to 14.5 year rates and Panel B2 presents results for the maturity range 15 to 25 year rates and Panel C1 presents results for the maturity range 2.5 to 14.5 year rates for the maturity range 15 to 25 year rates.

Time Series Regression Table Description

The Table P.16 columns represent the following:

The GILT Rate Benchmark Year column represents bond maturities.

The V Time Periods column represents the following:

The Total Period column represents the time series regression results for the year range 1994 to 2008, with a single year represented by the number of days from 1st January to 31st December. The Sub-Periods column represents the time series regression results for the year range 1994 to 2004, and for 1994 to 2007, with a single year represented by the number of days from 1st January to 31st December.

The total and sub-period columns present the coefficients and related statistics for the regression specified in equation (O.14), and follows the model:

$$V_{N,t} = a + b V GILT_t + e_{N,t}$$

Where: $V_{N,t}$ is the Value-at-Risk (V_t) for N firms at time t, $VGILT_t$ is the Value-at-Risk for the selected GILT rate at time t, and $e_{N,t}$ is the regression error term.

Actual from 2.5 years to 14.5 Years													
						V Time	Periods						
GILT Rate	Total Period	Sub-	Periods	Total Period	Sub-P	eriods	Total Period	Sub-P	eriods	Total Period	Sub-P	eriods	
Benchmark Year	1994- 2008	1994- 2004 b	1994- 2007	1994- 2008	1994- 2004 t	1994- 2007	1994- 2008	1994- 2004 <i>p</i>	1994- 2007	1994- 2008	1994- 2004 <i>R</i> ²	1994- 2007	
GILT 2.5 GILT 3 GILT 3.5 GILT 4 GILT 4.5	0.846* 0.564** 0.621** 0.67** 0.694**	$\begin{array}{c} 0.846*\\ 0.441*\\ 0.465\\ 0.411\\ 0.449\end{array}$	0.846* 0.464* 0.473 0.475 0.464	8.052 5.564 4.006 3.582 3.751	8.052 3.013 1.981 1.508 1.699	8.052 3.506 2.230 1.846 1.853	0.015 0.001 0.004 0.005 0.002	0.015 0.039 0.095 0.182 0.124	0.015 0.017 0.061 0.098 0.089	0.970 0.838 0.667 0.562 0.520	$\begin{array}{c} 0.970 \\ 0.694 \\ 0.395 \\ 0.275 \\ 0.243 \end{array}$	$\begin{array}{c} 0.970 \\ 0.711 \\ 0.415 \\ 0.275 \\ 0.222 \end{array}$	
GILT 5 GILT 5.5 GILT 6 GILT 6.5 GILT 7.5 GILT 7.5 GILT 8 GILT 8.5 GILT 9 GILT 9.5	$\begin{array}{c} 0.693^{**}\\ 0.68^{**}\\ 0.662^{*}\\ 0.634^{*}\\ 0.508^{*}\\ 0.587\\ 0.568\\ 0.544\\ 0.525\\ 0.510\\ \end{array}$	$\begin{array}{c} 0.411\\ 0.368\\ 0.325\\ 0.275\\ 0.230\\ 0.198\\ 0.169\\ 0.136\\ 0.106\\ 0.077\\ \end{array}$	$\begin{array}{c} 0.414\\ 0.357\\ 0.299\\ 0.235\\ 0.178\\ 0.131\\ 0.089\\ 0.043\\ -0.001\\ -0.045\\ \end{array}$	$\begin{array}{c} 3.365\\ 3.042\\ 2.763\\ 0.026\\ 2.286\\ 2.116\\ 1.988\\ 1.820\\ 1.664\\ 1.562\end{array}$	$\begin{array}{c} 1.427\\ 1.201\\ 1.016\\ 0.426\\ 0.680\\ 0.572\\ 0.475\\ 0.373\\ 0.282\\ 0.198\end{array}$	$\begin{array}{c} 1.522\\ 1.240\\ 0.994\\ 0.462\\ 0.564\\ 0.407\\ 0.272\\ 0.129\\ -0.004\\ -0.129\end{array}$	$\begin{array}{c} 0.005\\ 0.009\\ 0.016\\ 0.026\\ 0.040\\ 0.054\\ 0.068\\ 0.092\\ 0.120\\ 0.142\\ \end{array}$	$\begin{array}{c} 0.187\\ 0.260\\ 0.336\\ 0.426\\ 0.514\\ 0.581\\ 0.646\\ 0.718\\ 0.784\\ 0.847\\ \end{array}$	$\begin{array}{c} 0.154\\ 0.239\\ 0.340\\ 0.462\\ 0.583\\ 0.691\\ 0.790\\ 0.899\\ 0.997\\ 0.899\end{array}$	$\begin{array}{c} 0.465\\ 0.416\\ 0.370\\ 0.325\\ 0.287\\ 0.256\\ 0.233\\ 0.203\\ 0.176\\ 0.158\end{array}$	$\begin{array}{c} 0.184\\ 0.138\\ 0.103\\ 0.072\\ 0.049\\ 0.035\\ 0.024\\ 0.015\\ 0.009\\ 0.004\\ \end{array}$	$\begin{array}{c} 0.162\\ 0.114\\ 0.076\\ 0.046\\ 0.026\\ 0.014\\ 0.006\\ 0.001\\ 1.3E\text{-}06\\ 0.001\\ \end{array}$	
GILT 10 GILT 10.5 GILT 11 GILT 11.5 GILT 12.5 GILT 13 GILT 13.5 GILT 14 GILT 14.5	$\begin{array}{c} 0.481 \\ 0.481 \\ 0.470 \\ 0.459 \\ 0.442 \\ 0.427 \\ 0.412 \\ 0.399 \\ 0.388 \\ 0.372 \end{array}$	$\begin{array}{c} 0.021\\ 0.021\\ -0.004\\ -0.034\\ -0.065\\ -0.099\\ -0.136\\ -0.165\\ -0.195\\ -0.230\\ \end{array}$	$\begin{array}{c} -0.125\\ -0.125\\ -0.161\\ -0.202\\ -0.241\\ -0.282\\ -0.322\\ -0.356\\ -0.384\\ -0.418\end{array}$	$\begin{array}{c} 1.483\\ 1.417\\ 1.361\\ 1.307\\ 1.244\\ 1.192\\ 1.145\\ 1.108\\ 1.079\\ 1.029\end{array}$	$\begin{array}{c} 0.118\\ 0.051\\ -0.009\\ -0.079\\ -0.149\\ -0.224\\ -0.302\\ -0.362\\ -0.422\\ -0.493\end{array}$	-0.249 -0.353 -0.449 -0.555 -0.658 -0.767 -0.878 -0.972 -1.053 -1.153	$\begin{array}{c} 0.162\\ 0.180\\ 0.197\\ 0.214\\ 0.235\\ 0.254\\ 0.273\\ 0.288\\ 0.300\\ 0.322\\ \end{array}$	$\begin{array}{c} 0.909\\ 0.961\\ 0.993\\ 0.939\\ 0.885\\ 0.828\\ 0.770\\ 0.726\\ 0.683\\ 0.634\end{array}$	$\begin{array}{c} 0.807\\ 0.730\\ 0.662\\ 0.589\\ 0.523\\ 0.458\\ 0.397\\ 0.350\\ 0.313\\ 0.271 \end{array}$	$\begin{array}{c} 0.145\\ 0.134\\ 0.125\\ 0.116\\ 0.106\\ 0.099\\ 0.092\\ 0.086\\ 0.082\\ 0.075\\ \end{array}$	$\begin{array}{c} 0.002\\ 2.9E\text{-}04\\ 8.6E\text{-}06\\ 0.001\\ 0.002\\ 0.006\\ 0.010\\ 0.014\\ 0.019\\ 0.026\end{array}$	$\begin{array}{c} 0.005\\ 0.010\\ 0.017\\ 0.025\\ 0.035\\ 0.047\\ 0.060\\ 0.073\\ 0.085\\ 0.100\\ \end{array}$	

Table P.16	(Continued)
1 abic 1 .10	Commuta	

PANEL A1: Time Series Regression Analysis of the Primary and Secondary Sample for GILT Rate Value-at-Risk Actual and Historical Value-at-Risk Actual from 2.5 years to 14.5 Years

	-	-		Actual f	rom 15 yea	ars to 25 ye	ears					
						V Time I	Periods					
GILT Rate	Total Period	Sub-	Periods	Total Period	Sub-F	Periods	Total Period	Sub-F	Periods	Total Period	Sub-P	eriods
Benchmark Year	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007
		b			t			р			R^2	
GILT 15 GILT 15.5 GILT 16.5 GILT 16.5 GILT 17 GILT 17.5 GILT 18 GILT 18.5 GILT 19 GILT 19.5	$\begin{array}{c} 0.355 \\ 0.341 \\ 0.325 \\ 0.315 \\ 0.306 \\ 0.296 \\ 0.288 \\ 0.279 \\ 0.269 \\ 0.256 \end{array}$	$\begin{array}{c} -0.261 \\ -0.288 \\ -0.313 \\ -0.333 \\ -0.355 \\ -0.379 \\ -0.400 \\ -0.414 \\ -0.434 \\ -0.450 \end{array}$	-0.445 -0.468 -0.488 -0.504 -0.520 -0.535 -0.546 -0.553 -0.560 -0.565	$\begin{array}{c} 0.975\\ 0.940\\ 0.891\\ 0.871\\ 0.854\\ 0.840\\ 0.829\\ 0.810\\ 0.791\\ 0.763\end{array}$	$\begin{array}{c} -0.553 \\ -0.608 \\ -0.655 \\ -0.696 \\ -0.742 \\ -0.795 \\ -0.844 \\ -0.876 \\ -0.926 \\ -0.971 \end{array}$	-1.236 -1.312 -1.384 -1.445 -1.511 -1.580 -1.641 -1.687 -1.742 -1.794	$\begin{array}{c} 0.347\\ 0.365\\ 0.389\\ 0.400\\ 0.408\\ 0.416\\ 0.422\\ 0.433\\ 0.443\\ 0.459\end{array}$	$\begin{array}{c} 0.593\\ 0.558\\ 0.529\\ 0.504\\ 0.477\\ 0.447\\ 0.421\\ 0.404\\ 0.379\\ 0.357\end{array}$	$\begin{array}{c} 0.240\\ 0.214\\ 0.192\\ 0.174\\ 0.157\\ 0.140\\ 0.127\\ 0.117\\ 0.107\\ 0.098 \end{array}$	$\begin{array}{c} 0.068\\ 0.064\\ 0.058\\ 0.055\\ 0.053\\ 0.051\\ 0.050\\ 0.048\\ 0.046\\ 0.043\\ \end{array}$	$\begin{array}{c} 0.033\\ 0.040\\ 0.046\\ 0.051\\ 0.058\\ 0.066\\ 0.073\\ 0.079\\ 0.087\\ 0.095\\ \end{array}$	$\begin{array}{c} 0.113\\ 0.125\\ 0.138\\ 0.148\\ 0.160\\ 0.172\\ 0.183\\ 0.192\\ 0.202\\ 0.211\\ \end{array}$
GILT 20 GILT 20.5 GILT 21 GILT 21.5 GILT 22.5 GILT 23.5 GILT 23.5 GILT 24 GILT 24.5 GILT 25	$\begin{array}{c} 0.151 \\ 0.144 \\ 0.097 \\ 0.084 \\ 0.148 \\ 0.137 \\ 0.128 \\ 0.118 \\ 0.136 \\ 0.125 \\ 0.116 \end{array}$	-0.878 -0.890 -0.879 -0.889 -1.546 -1.526 -1.503 -1.470 -1.910 -1.869 -1.829	-0.857* -0.859* -0.887* -0.887** -1.024* -1.014* -1.002* -0.991* -1.074* -1.059* -1.045*	$\begin{array}{c} 0.377\\ 0.366\\ 0.247\\ 0.216\\ 0.339\\ 0.317\\ 0.300\\ 0.281\\ 0.258\\ 0.242\\ 0.228\\ \end{array}$	-1.877 -1.938 -2.357 -2.462 -1.921 -2.045 -2.271 -2.831 -2.983 -3.135	-2.714 -2.786 -3.275 -3.388 -3.287 -3.389 -3.472 -3.559 -3.157 -3.228 -3.296	$\begin{array}{c} 0.714\\ 0.722\\ 0.810\\ 0.834\\ 0.744\\ 0.760\\ 0.773\\ 0.787\\ 0.807\\ 0.819\\ 0.829\end{array}$	$\begin{array}{c} 0.110\\ 0.101\\ 0.065\\ 0.057\\ 0.150\\ 0.133\\ 0.119\\ 0.108\\ 0.216\\ 0.206\\ 0.197\\ \end{array}$	$\begin{array}{c} 0.024\\ 0.021\\ 0.011\\ 0.010\\ 0.017\\ 0.015\\ 0.013\\ 0.012\\ 0.034\\ 0.032\\ 0.030\\ \end{array}$	$\begin{array}{c} 0.014\\ 0.013\\ 0.007\\ 0.005\\ 0.016\\ 0.014\\ 0.013\\ 0.011\\ 0.013\\ 0.012\\ 0.010\\ \end{array}$	$\begin{array}{c} 0.370\\ 0.385\\ 0.526\\ 0.548\\ 0.552\\ 0.582\\ 0.610\\ 0.632\\ 0.889\\ 0.899\\ 0.908\\ \end{array}$	$\begin{array}{c} 0.450\\ 0.463\\ 0.573\\ 0.589\\ 0.643\\ 0.657\\ 0.668\\ 0.679\\ 0.714\\ 0.723\\ 0.731\\ \end{array}$

PANEL A2: Time Series Regression Analysis of the Primary and Secondary Sample for GILT Rate Value-at-Risk Actual and Historical Value-at-Risk Actual from 15 years to 25 years

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels.

Table P.16 (Continued)

	8	j·		y	ears to 14.	5 Years						
						V Time	Periods					
GILT Rate	Total Period	Sub-	Periods	Total Period	Sub-P	eriods	Total Period	Sub-P	eriods	Total Period	Sub-P	eriods
Benchmark Year	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007
		b			t			р			R^2	
GILT 2.5 GILT 3 GILT 3.5 GILT 4 GILT 4.5	1.074* 0.432* 0.458** 0.515** 0.527**	$\begin{array}{c} 1.074^{*} \\ 0.445 \\ 0.471^{*} \\ 0.460 \\ 0.505^{*} \end{array}$	1.074* 0.473* 0.496* 0.532* 0.535*	6.495 3.548 3.771 3.344 3.570	6.495 2.266 2.463 2.110 2.511	6.495 2.685 2.799 2.363 2.505	$\begin{array}{c} 0.023 \\ 0.012 \\ 0.005 \\ 0.007 \\ 0.003 \end{array}$	$\begin{array}{c} 0.023 \\ 0.086 \\ 0.049 \\ 0.079 \\ 0.033 \end{array}$	$\begin{array}{c} 0.023 \\ 0.044 \\ 0.027 \\ 0.042 \\ 0.028 \end{array}$	$\begin{array}{c} 0.955 \\ 0.677 \\ 0.640 \\ 0.528 \\ 0.495 \end{array}$	$\begin{array}{c} 0.955\\ 0.562\\ 0.503\\ 0.426\\ 0.412\end{array}$	$\begin{array}{c} 0.955 \\ 0.590 \\ 0.528 \\ 0.383 \\ 0.343 \end{array}$
GILT 5 GILT 5.5 GILT 6 GILT 6.5 GILT 7.5 GILT 7.5 GILT 8 GILT 8.5 GILT 9 GILT 9.5	$\begin{array}{c} 0.526^{**} \\ 0.515^{*} \\ 0.499^{*} \\ 0.475^{*} \\ 0.453 \\ 0.436 \\ 0.420 \\ 0.402 \\ 0.387 \\ 0.372 \end{array}$	$\begin{array}{c} 0.485\\ 0.453\\ 0.422\\ 0.382\\ 0.346\\ 0.322\\ 0.302\\ 0.279\\ 0.260\\ 0.243\end{array}$	$\begin{array}{c} 0.503 \\ 0.460 \\ 0.412 \\ 0.356 \\ 0.307 \\ 0.266 \\ 0.232 \\ 0.194 \\ 0.157 \\ 0.122 \end{array}$	$\begin{array}{c} 3.218\\ 2.906\\ 2.630\\ 0.034\\ 2.157\\ 1.988\\ 1.861\\ 1.708\\ 1.559\\ 1.448\end{array}$	$\begin{array}{c} 2.174 \\ 1.883 \\ 1.654 \\ 0.186 \\ 1.254 \\ 1.131 \\ 1.028 \\ 0.921 \\ 0.828 \\ 0.750 \end{array}$	$\begin{array}{c} 2.145\\ 1.829\\ 1.553\\ 0.219\\ 1.083\\ 0.917\\ 0.785\\ 0.644\\ 0.507\\ 0.384 \end{array}$	$\begin{array}{c} 0.007\\ 0.012\\ 0.021\\ 0.034\\ 0.050\\ 0.068\\ 0.086\\ 0.111\\ 0.143\\ 0.171\\ \end{array}$	$\begin{array}{c} 0.058\\ 0.092\\ 0.132\\ 0.186\\ 0.242\\ 0.287\\ 0.331\\ 0.381\\ 0.429\\ 0.472\\ \end{array}$	$\begin{array}{c} 0.053\\ 0.092\\ 0.146\\ 0.219\\ 0.300\\ 0.377\\ 0.448\\ 0.532\\ 0.621\\ 0.708\\ \end{array}$	$\begin{array}{c} 0.443\\ 0.394\\ 0.347\\ 0.302\\ 0.264\\ 0.233\\ 0.210\\ 0.183\\ 0.158\\ 0.139\\ \end{array}$	$\begin{array}{c} 0.344\\ 0.283\\ 0.233\\ 0.185\\ 0.149\\ 0.124\\ 0.105\\ 0.086\\ 0.071\\ 0.059\\ \end{array}$	$\begin{array}{c} 0.277\\ 0.218\\ 0.167\\ 0.123\\ 0.089\\ 0.065\\ 0.049\\ 0.033\\ 0.021\\ 0.012\\ \end{array}$
GILT 10 GILT 10.5 GILT 11 GILT 11.5 GILT 12.5 GILT 12.5 GILT 13 GILT 13.5 GILT 14 GILT 14.5	$\begin{array}{c} 0.341 \\ 0.341 \\ 0.329 \\ 0.316 \\ 0.298 \\ 0.281 \\ 0.262 \\ 0.245 \\ 0.232 \\ 0.214 \end{array}$	$\begin{array}{c} 0.210\\ 0.210\\ 0.200\\ 0.184\\ 0.164\\ 0.141\\ 0.114\\ 0.093\\ 0.070\\ 0.043\\ \end{array}$	0.055 0.055 0.028 -0.006 -0.040 -0.078 -0.117 -0.152 -0.179 -0.214	$\begin{array}{c} 1.351 \\ 1.273 \\ 1.210 \\ 1.140 \\ 1.063 \\ 0.991 \\ 0.921 \\ 0.860 \\ 0.815 \\ 0.746 \end{array}$	$\begin{array}{c} 0.675\\ 0.613\\ 0.566\\ 0.505\\ 0.438\\ 0.370\\ 0.294\\ 0.234\\ 0.175\\ 0.105\\ \end{array}$	$\begin{array}{c} 0.265\\ 0.166\\ 0.082\\ -0.017\\ -0.117\\ -0.224\\ -0.333\\ -0.433\\ -0.510\\ -0.613\end{array}$	$\begin{array}{c} 0.200\\ 0.225\\ 0.248\\ 0.275\\ 0.307\\ 0.340\\ 0.374\\ 0.405\\ 0.430\\ 0.469\end{array}$	$\begin{array}{c} 0.517\\ 0.555\\ 0.585\\ 0.626\\ 0.672\\ 0.720\\ 0.775\\ 0.820\\ 0.865\\ 0.918\\ \end{array}$	$\begin{array}{c} 0.796\\ 0.871\\ 0.936\\ 0.987\\ 0.909\\ 0.827\\ 0.745\\ 0.673\\ 0.619\\ 0.551\\ \end{array}$	$\begin{array}{c} 0.123\\ 0.111\\ 0.101\\ 0.091\\ 0.080\\ 0.070\\ 0.061\\ 0.054\\ 0.049\\ 0.041\\ \end{array}$	$\begin{array}{c} 0.048\\ 0.040\\ 0.034\\ 0.028\\ 0.021\\ 0.015\\ 0.010\\ 0.006\\ 0.003\\ 0.001\\ \end{array}$	$\begin{array}{c} 0.006\\ 0.002\\ 0.001\\ 2.3E\text{-}05\\ 0.001\\ 0.004\\ 0.009\\ 0.015\\ 0.021\\ 0.030\\ \end{array}$

PANEL B1: Time Series Regression Analysis of the Control Group for GILT Rate Value-at-Risk Actual and Historical Value-at-Risk Actual from 2.5

	U	·			years to 25	Years						
						V Time	Periods					
GILT Rate	Total Period	Sub-l	Periods	Total Period	Sub-P	eriods	Total Period	Sub-P	eriods	Total Period	Sub-P	eriods
Benchmark Year	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007
		b			t			р			R^2	
GILT 15 GILT 15.5 GILT 16.5 GILT 16.5 GILT 17 GILT 17.5 GILT 18 GILT 18.5 GILT 19 GILT 19.5	$\begin{array}{c} 0.195\\ 0.180\\ 0.161\\ 0.148\\ 0.135\\ 0.124\\ 0.114\\ 0.103\\ 0.092\\ 0.080\\ \end{array}$	0.016 -0.008 -0.032 -0.052 -0.074 -0.098 -0.121 -0.138 -0.164 -0.185	-0.246 -0.273 -0.301 -0.323 -0.345 -0.366 -0.383 -0.397 -0.411 -0.422	$\begin{array}{c} 0.676 \\ 0.624 \\ 0.557 \\ 0.516 \\ 0.477 \\ 0.443 \\ 0.412 \\ 0.376 \\ 0.342 \\ 0.302 \end{array}$	$\begin{array}{c} 0.037 \\ -0.020 \\ -0.076 \\ -0.124 \\ -0.175 \\ -0.232 \\ -0.286 \\ -0.328 \\ -0.390 \\ -0.446 \end{array}$	-0.706 -0.789 -0.878 -0.954 -1.032 -1.111 -1.183 -1.247 -1.314 -1.379	$\begin{array}{c} 0.511\\ 0.544\\ 0.587\\ 0.615\\ 0.641\\ 0.665\\ 0.687\\ 0.713\\ 0.738\\ 0.768\\ \end{array}$	$\begin{array}{c} 0.971\\ 0.985\\ 0.941\\ 0.904\\ 0.865\\ 0.822\\ 0.782\\ 0.750\\ 0.705\\ 0.666\end{array}$	$\begin{array}{c} 0.493 \\ 0.446 \\ 0.397 \\ 0.359 \\ 0.322 \\ 0.288 \\ 0.260 \\ 0.236 \\ 0.214 \\ 0.193 \end{array}$	$\begin{array}{c} 0.034\\ 0.029\\ 0.023\\ 0.020\\ 0.017\\ 0.015\\ 0.013\\ 0.011\\ 0.009\\ 0.007\\ \end{array}$	$\begin{array}{c} 1.6E\text{-}04\\ 4.4E\text{-}05\\ 0.001\\ 0.002\\ 0.003\\ 0.006\\ 0.009\\ 0.012\\ 0.017\\ 0.022 \end{array}$	$\begin{array}{c} 0.040\\ 0.049\\ 0.060\\ 0.070\\ 0.082\\ 0.093\\ 0.104\\ 0.115\\ 0.126\\ 0.137\\ \end{array}$
GILT 20 GILT 20.5 GILT 21 GILT 21.5 GILT 22.5 GILT 23 GILT 23.5 GILT 24 GILT 24.5 GILT 25	-3.4E-04 -0.010 -0.033 -0.046 -0.067 -0.076 -0.082 -0.089 -0.062 -0.068 -0.073	-0.501 -0.525 -0.537 -0.558 -1.999 -1.966 -1.930 -1.884* -2.116 -2.068 -2.021	-0.629 -0.642 -0.662 -0.671 -0.969* -0.963* -0.956* -0.948* -0.970 -0.957 -0.945	$\begin{array}{c} -0.001\\ -0.032\\ -0.102\\ -0.143\\ -0.181\\ -0.207\\ -0.228\\ -0.252\\ -0.144\\ -0.161\\ -0.176\end{array}$	$\begin{array}{c} -1.047\\ -1.116\\ -1.110\\ -1.178\\ -2.593\\ -2.808\\ -3.033\\ -3.252\\ -3.785\\ -4.049\\ -4.319\end{array}$	-1.798 -1.880 -1.894 -1.972 -2.754 -2.847 -2.928 -3.013 -2.558 -2.609 -2.658	$\begin{array}{c} 0.999\\ 0.975\\ 0.921\\ 0.889\\ 0.862\\ 0.842\\ 0.826\\ 0.808\\ 0.891\\ 0.878\\ 0.867\end{array}$	$\begin{array}{c} 0.335\\ 0.307\\ 0.317\\ 0.292\\ 0.081\\ 0.067\\ 0.056\\ 0.047\\ 0.164\\ 0.154\\ 0.145\end{array}$	$\begin{array}{c} 0.106\\ 0.093\\ 0.095\\ 0.084\\ 0.033\\ 0.029\\ 0.026\\ 0.024\\ 0.063\\ 0.059\\ 0.057\\ \end{array}$	$\begin{array}{c} 1.2E\text{-}07\\ 1.0E\text{-}04\\ 0.001\\ 0.002\\ 0.005\\ 0.006\\ 0.007\\ 0.009\\ 0.004\\ 0.005\\ 0.006\end{array}$	$\begin{array}{c} 0.155\\ 0.172\\ 0.198\\ 0.217\\ 0.691\\ 0.724\\ 0.754\\ 0.779\\ 0.935\\ 0.943\\ 0.949\end{array}$	$\begin{array}{c} 0.264\\ 0.282\\ 0.309\\ 0.327\\ 0.558\\ 0.575\\ 0.588\\ 0.602\\ 0.621\\ 0.630\\ 0.638\\ \end{array}$

PANEL B2: Time Series Regression Analysis of the Control Group for GILT Rate Value-at-Risk Actual and Historical Value-at-Risk Actual from 15 years to 25 Years

Table P.16 (Continued)

				У	ears to 14.	5 Years						
						V Time	Periods					
GILT Rate	Total Period	Sub-	Periods	Total Period	Sub-P	Periods	Total Period	Sub-P	eriods	Total Period	Sub-F	Periods
Benchmark Year	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007
		b			t			р			R^2	
GILT 2.5 GILT 3 GILT 3.5 GILT 4 GILT 4.5	-0.049 0.634* 0.705* 0.767* 0.865**	-0.049 0.183 0.182 0.131 0.305	-0.049 0.176 0.164 0.163 0.288	-0.131 2.707 2.922 3.072 3.595	-0.131 1.057 0.875 0.577 1.130	-0.131 1.173 0.863 0.732 1.140	$\begin{array}{c} 0.908 \\ 0.035 \\ 0.019 \\ 0.012 \\ 0.003 \end{array}$	$\begin{array}{c} 0.908 \\ 0.350 \\ 0.415 \\ 0.585 \\ 0.288 \end{array}$	$\begin{array}{c} 0.908 \\ 0.294 \\ 0.416 \\ 0.483 \\ 0.277 \end{array}$	0.008 0.550 0.516 0.486 0.499	$\begin{array}{c} 0.008 \\ 0.218 \\ 0.113 \\ 0.053 \\ 0.124 \end{array}$	$\begin{array}{c} 0.008\\ 0.216\\ 0.096\\ 0.056\\ 0.098\end{array}$
GILT 5 GILT 5.5 GILT 6 GILT 6.5 GILT 7 GILT 7.5 GILT 8 GILT 8.5 GILT 9 GILT 9.5	0.898** 0.916** 0.924** 0.919** 0.915* 0.911* 0.907* 0.895* 0.895* 0.903*	$\begin{array}{c} 0.319\\ 0.322\\ 0.322\\ 0.312\\ 0.295\\ 0.295\\ 0.291\\ 0.280\\ 0.273\\ 0.269\\ \end{array}$	$\begin{array}{c} 0.284\\ 0.271\\ 0.253\\ 0.227\\ 0.205\\ 0.181\\ 0.159\\ 0.130\\ 0.109\\ 0.091 \end{array}$	3.492 3.368 3.233 0.009 2.955 2.828 2.743 2.586 2.451 2.397	$\begin{array}{c} 1.121 \\ 1.095 \\ 1.063 \\ 0.339 \\ 0.961 \\ 0.920 \\ 0.884 \\ 0.831 \\ 0.789 \\ 0.752 \end{array}$	$\begin{array}{c} 1.069\\ 0.983\\ 0.890\\ 0.448\\ 0.699\\ 0.607\\ 0.523\\ 0.423\\ 0.348\\ 0.282\end{array}$	$\begin{array}{c} 0.004 \\ 0.005 \\ 0.007 \\ 0.009 \\ 0.011 \\ 0.014 \\ 0.017 \\ 0.023 \\ 0.029 \\ 0.032 \end{array}$	$\begin{array}{c} 0.291 \\ 0.302 \\ 0.316 \\ 0.339 \\ 0.362 \\ 0.381 \\ 0.399 \\ 0.427 \\ 0.451 \\ 0.471 \end{array}$	$\begin{array}{c} 0.306\\ 0.345\\ 0.391\\ 0.448\\ 0.498\\ 0.555\\ 0.610\\ 0.680\\ 0.734\\ 0.783\\ \end{array}$	$\begin{array}{c} 0.484 \\ 0.466 \\ 0.423 \\ 0.402 \\ 0.381 \\ 0.367 \\ 0.340 \\ 0.316 \\ 0.307 \end{array}$	$\begin{array}{c} 0.122\\ 0.117\\ 0.112\\ 0.093\\ 0.086\\ 0.080\\ 0.071\\ 0.065\\ 0.059\\ \end{array}$	$\begin{array}{c} 0.087\\ 0.075\\ 0.062\\ 0.049\\ 0.039\\ 0.030\\ 0.022\\ 0.015\\ 0.010\\ 0.007\\ \end{array}$
GILT 10 GILT 10.5 GILT 11 GILT 11.5 GILT 12.5 GILT 13 GILT 13.5 GILT 14 GILT 14.5	0.91* 0.912* 0.912* 0.912* 0.912* 0.912* 0.905* 0.905* 0.905* 0.99* 0.892* 0.884	$\begin{array}{c} 0.257\\ 0.257\\ 0.253\\ 0.244\\ 0.232\\ 0.218\\ 0.200\\ 0.185\\ 0.171\\ 0.154 \end{array}$	$\begin{array}{c} 0.051 \\ 0.051 \\ 0.029 \\ 0.006 \\ -0.018 \\ -0.039 \\ -0.064 \\ -0.088 \\ -0.109 \\ -0.129 \end{array}$	2.368 2.337 2.305 2.285 2.247 2.232 2.215 2.199 2.184 2.155	$\begin{array}{c} 0.714\\ 0.683\\ 0.652\\ 0.611\\ 0.568\\ 0.523\\ 0.470\\ 0.427\\ 0.388\\ 0.345\end{array}$	0.219 0.153 0.085 0.018 -0.051 -0.111 -0.181 -0.248 -0.306 -0.364	$\begin{array}{c} 0.034\\ 0.036\\ 0.038\\ 0.040\\ 0.043\\ 0.044\\ 0.045\\ 0.047\\ 0.048\\ 0.050\\ \end{array}$	$\begin{array}{c} 0.493\\ 0.512\\ 0.530\\ 0.556\\ 0.584\\ 0.614\\ 0.649\\ 0.680\\ 0.707\\ 0.738\\ \end{array}$	$\begin{array}{c} 0.830\\ 0.881\\ 0.934\\ 0.986\\ 0.960\\ 0.913\\ 0.859\\ 0.809\\ 0.765\\ 0.722\\ \end{array}$	$\begin{array}{c} 0.301 \\ 0.296 \\ 0.290 \\ 0.287 \\ 0.280 \\ 0.277 \\ 0.274 \\ 0.271 \\ 0.268 \\ 0.263 \end{array}$	$\begin{array}{c} 0.054\\ 0.049\\ 0.045\\ 0.040\\ 0.035\\ 0.029\\ 0.024\\ 0.020\\ 0.016\\ 0.013\\ \end{array}$	$\begin{array}{c} 0.004\\ 0.002\\ 0.001\\ 2.7E\text{-}05\\ 2.2E\text{-}04\\ 0.001\\ 0.003\\ 0.005\\ 0.008\\ 0.011\\ \end{array}$

Table P.16 (Continued)

PANEL C1: Time Series Regression Analysis of the Primary Sample for GILT Rate Value-at-Risk Actual and Historical Value-at-Risk Actual from 2.5 years to 14.5 Years

						V Time	Periods					
GILT Rate	Total Period	Sub-I	Periods	Total Period	Sub-P	eriods	Total Period	Sub-P	eriods	Total Period	Sub-P	Periods
Benchmark Year	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007
		b			t			р			R^2	
GILT 15 GILT 15.5 GILT 16 GILT 16.5 GILT 17 GILT 17.5 GILT 18.5 GILT 18.5 GILT 19 GILT 19.5	$\begin{array}{c} 0.876\\ 0.867\\ 0.856\\ 0.847\\ 0.838\\ 0.827\\ 0.817\\ 0.803\\ 0.788\\ 0.770\\ \end{array}$	$\begin{array}{c} 0.145\\ 0.134\\ 0.127\\ 0.119\\ 0.110\\ 0.098\\ 0.086\\ 0.077\\ 0.065\\ 0.054\\ \end{array}$	-0.142 -0.154 -0.165 -0.174 -0.183 -0.193 -0.200 -0.206 -0.214 -0.220	$\begin{array}{c} 2.119\\ 2.103\\ 2.066\\ 2.062\\ 2.065\\ 2.067\\ 2.074\\ 2.060\\ 2.047\\ 2.023\\ \end{array}$	$\begin{array}{c} 0.319\\ 0.292\\ 0.274\\ 0.257\\ 0.236\\ 0.209\\ 0.184\\ 0.165\\ 0.140\\ 0.117\\ \end{array}$	$\begin{array}{c} -0.400\\ -0.435\\ -0.468\\ -0.496\\ -0.525\\ -0.561\\ -0.587\\ -0.612\\ -0.646\\ -0.675\end{array}$	$\begin{array}{c} 0.054\\ 0.056\\ 0.059\\ 0.060\\ 0.059\\ 0.059\\ 0.059\\ 0.058\\ 0.060\\ 0.061\\ 0.064 \end{array}$	$\begin{array}{c} 0.757\\ 0.777\\ 0.790\\ 0.803\\ 0.819\\ 0.839\\ 0.839\\ 0.858\\ 0.872\\ 0.892\\ 0.910\\ \end{array}$	$\begin{array}{c} 0.696\\ 0.671\\ 0.648\\ 0.629\\ 0.609\\ 0.585\\ 0.568\\ 0.552\\ 0.530\\ 0.513\\ \end{array}$	$\begin{array}{c} 0.257\\ 0.254\\ 0.247\\ 0.246\\ 0.247\\ 0.247\\ 0.249\\ 0.246\\ 0.244\\ 0.239\\ \end{array}$	$\begin{array}{c} 0.011\\ 0.009\\ 0.008\\ 0.007\\ 0.006\\ 0.005\\ 0.004\\ 0.003\\ 0.002\\ 0.002\\ 0.002 \end{array}$	$\begin{array}{c} 0.013\\ 0.016\\ 0.018\\ 0.020\\ 0.022\\ 0.026\\ 0.028\\ 0.030\\ 0.034\\ 0.037\\ \end{array}$
GILT 20 GILT 20.5 GILT 21.5 GILT 22.5 GILT 22.5 GILT 23.5 GILT 23.5 GILT 24 GILT 24.5 GILT 25	$\begin{array}{c} 0.678 \\ 0.669 \\ 0.626 \\ 0.611 \\ 0.690 \\ 0.675 \\ 0.662 \\ 0.648 \\ 0.677 \\ 0.660 \\ 0.644 \end{array}$	$\begin{array}{c} -0.315\\ -0.317\\ -0.302\\ -0.304\\ -0.706\\ -0.675\\ -0.641\\ -0.605\\ -1.573\\ -1.537\\ -1.503\end{array}$	-0.514 -0.511 -0.529 -0.527 -0.695* -0.663* -0.663* -0.647* -0.778* -0.765* -0.752*	$\begin{array}{c} 1.518\\ 1.519\\ 1.394\\ 1.369\\ 1.396\\ 1.383\\ 1.377\\ 1.366\\ 1.124\\ 1.108\\ 1.095\end{array}$	-0.738 -0.748 -0.794 -0.809 -0.861 -0.856 -0.843 -0.825 -3.579 -3.817 -4.057	-1.748 -1.763 -1.940 -1.968 -2.532 -2.524 -2.498 -2.473 -3.070 -3.105 -3.138	$\begin{array}{c} 0.160\\ 0.160\\ 0.197\\ 0.204\\ 0.205\\ 0.209\\ 0.211\\ 0.214\\ 0.312\\ 0.318\\ 0.324 \end{array}$	$\begin{array}{c} 0.489\\ 0.483\\ 0.463\\ 0.455\\ 0.455\\ 0.455\\ 0.461\\ 0.470\\ 0.173\\ 0.163\\ 0.154\end{array}$	$\begin{array}{c} 0.114\\ 0.112\\ 0.088\\ 0.085\\ 0.045\\ 0.045\\ 0.047\\ 0.048\\ 0.037\\ 0.036\\ 0.035\\ \end{array}$	$\begin{array}{c} 0.187\\ 0.188\\ 0.178\\ 0.172\\ 0.218\\ 0.215\\ 0.213\\ 0.211\\ 0.202\\ 0.197\\ 0.193\\ \end{array}$	$\begin{array}{c} 0.083\\ 0.085\\ 0.112\\ 0.116\\ 0.198\\ 0.196\\ 0.192\\ 0.185\\ 0.928\\ 0.936\\ 0.943\\ \end{array}$	$\begin{array}{c} 0.253\\ 0.257\\ 0.320\\ 0.326\\ 0.517\\ 0.515\\ 0.510\\ 0.505\\ 0.702\\ 0.707\\ 0.711 \end{array}$

PANEL C2: Time Series Regression Analysis of the Primary and Secondary Sample for GILT Rate Value-at-Risk Actual and Historical Value-at-Risk

Table P.16 (Continued)

The Table P.16 time series regression slopes show that some of the selected GILT Government bond benchmark rate Historical Value-at-Risk Actual variable exhibits some statistically significant explanatory power to the samples and the control group Historical Value-at-Risk Actual variable for all three time periods. However, a greater level of significance is evident for the shorter maturities. These significances are evident for the time periods 1994 to 2008 and 1994 to 2007.

This suggests that the GILT Government bond benchmark rate Historical Value-at-Risk Actual variable reacted with the samples and the control group Historical Value-at-Risk Actual variable similarly before and after the 2005 accounting change.

P.14 Change in Foreign Currency Rates and Market Price Return

The regression analysis for the change in selected foreign exchange currency rates and the market price return is presented for the time range 1994 to 2008 in Table P.17. In these regressions, the market price return variable is tested as the dependent variable, and the selected foreign exchange currency rate variables for European Euro (EUR) to Great British Pound (GBP), Unites Stated Dollar (USD) to GBP and Japanese Yen (JPY) to GBP are individually tested as the independent variable.

Table P.17Time Series Regression Analysis for Change in Foreign Exchange
Currency Rates and Market Price Return

Time Series Regression

Table P.17 Panels A to C show the results for the time series regressions that test the change in selected foreign exchange currency rates and the market price return by applying the regression specified in equation (O.13). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; and, Panel C presents results for the Primary sample.

Time Series Regression Table Description

The Table P.17 columns represent the following:

The Foreign Exchange Rate Currency Pair column represents the samples and the control group result panels for the selected rates.

The Rates and Returns Time Periods column represents the following:

Total Period column represents the time series regression results for the year range 1994 to 2008, with a single year represented by the number of days from 1st January to 31st December. The Sub-Periods column represents the time series regression results for the year range 1994 to 2004, and for 1994 to 2007, with a single year represented by the number of days from 1st January to 31st December.

The total and sub-period columns present the coefficients and related statistics for the regression specified in equation (O.13), and follows the model:

 $dM_{N,t} = a + b \ dCURRENCY_t + e_{N,t}$

Where: $dM_{N,t}$ is the log Market Price return $(M_{t-1,t})$ for N firms at time t, $dCURRENCY_t$ is the change in the selected currency pair rate $(dCURRENCY_{t-1,t})$ at time t, and $e_{N,t}$ is the regression error term.

				Tal	ble P.17 (C	ontinued)						
					Rates	and Return	ns Time Peric	ods				
Foreign Exchange	Total Period	Sub-I	Periods	Total Period	Sub-P	Periods	Total Period	Sub-P	Periods	Total Period	Sub-P	Periods
Rate Currency Pair	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007
		b			t			р			R^2	
	PAN	IEL A:	Foreig	gn Exchange	e Rates vs.	Primary	and Second	lary Samj	ple Retur	n		
EURO and GBP USD and GPB YEN and GBP	2.103* 0.984 0.934*	0.882 -1.075 -0.271	1.132 -0.883 -0.071	3.006 1.713 2.670	0.540 -1.205 -0.410	1.003 -1.282 -0.127	$0.017 \\ 0.110 \\ 0.019$	0.618 0.259 0.692	0.349 0.224 0.901	0.530 0.184 0.354	0.068 0.139 0.018	0.126 0.120 0.001
		PAN	EL B:	Foreign E	Exchange	Rates vs.	Control Gro	oup Retur	'n			
EURO and GBP USD and GPB YEN and GBP	1.637* 0.569 0.581	1.133 -0.516 -0.321	1.148 -0.784 -0.156	2.394 1.098 1.738	0.711 -0.560 -0.499	0.981 -1.086 -0.270	0.044 0.292 0.106	0.516 0.589 0.630	0.359 0.299 0.792	0.417 0.085 0.189	0.112 0.034 0.027	0.121 0.090 0.006
		PANE	LC:	Foreign E	xchange F	Rates vs. P	rimary San	nple Retu	rn			
EURO and GBP USD and GPB YEN and GBP	3.328** 1.885* 1.63**	1.387 -0.855 0.247	1.660 -0.588 0.337	5.139 2.784 0.001	1.184 -0.991 0.702	2.075 -0.841 0.545	0.001 0.015 0.001	$0.302 \\ 0.348 \\ 0.702$	$0.077 \\ 0.417 \\ 0.545$	0.768 0.374 0.596	0.259 0.098 0.017	0.381 0.056 0.031

The Table P.17 time series regression slopes show that some selected foreign exchange currency pair rate changes exhibit statistically significant explanatory power to the samples and the control group market price returns for the time period 1994 to 2008. However, the Primary sample slopes show statistical significance for all selected rates for the 1994 to 2008 time period.

The Primary and Secondary sample slopes show statistical significance for the EURO and GBP, and YEN and GBP rates. The Control group shows statistical significance for the EURO and GBP rate.

Examining the significant regression slopes, the Primary and Secondary sample exhibits greater magnitude, statistical significance, and model strength when compared to the Control group.

The statistically significant results suggest that the change in selected foreign exchange rates reacted in relation to the Primary and Secondary sample market price return variable after the 2005 accounting change, than the Control group reactions.

P.15 Foreign Currency Rate Value-at-Risk Actual and Historical Value-at-Risk Actual

The regression analysis for selected foreign exchange currency rate Historical Value-at-Risk Actual and market price return Historical Value-at-Risk Actual is presented for the time range 1994 to 2008 in Table P.18. In these regressions, the Historical Value-at-Risk Actual variable is tested as the dependent variable, and the selected foreign exchange currency rate Value-at-Risk Actual variables for the currency pairs European Euro (EUR) and Great British Pound (GBP), Unites Stated Dollar (USD) and GBP and Japanese Yen (JPY) and GBP are individually tested as the independent variable.

Table P.18Time Series Regression Analysis for Foreign Exchange Currency
Rates Historical Value-at-Risk Actual and Historical Value-at-Risk Actual

Time Series Regression

Table P.18 Panels A to C show the results for the time series regressions that test the selected foreign exchange currency rate Historical Value-at-Risk Actual and Historical Value-at-Risk Actual by applying the regression specified in equation (O.14). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; and, Panel C presents results for the Primary sample.

Time Series Regression Table Description

The Table P.18 columns represent the following:

The Foreign Exchange Rate Currency Pair column represents the samples and the control group result panels for the selected rates.

The V Time Periods column represents the following:

The Total Period column represents the time series regression results for the year range 1994 to 2008, with a single year represented by the number of days from 1st January to 31st December. The Sub-Periods column represents the time series regression results for the year range 1994 to 2004, and for 1994 to 2007, with a single year represented by the number of days from 1st January to 31st December.

The total and sub-period columns present the coefficients and related statistics for the regression specified in equation (O.14), and follows the model:

$V_{N,t} = a + b V CURRENCY_t + e_{N,t}$

Where: $V_{N,t}$ is the Value-at-Risk (V_t) for N firms at time t, VCURRENCY_t is the Value-at-Risk for the selected currency pair rate at time t, and $e_{N,t}$ is the regression error term.

				Tak	ole P.18 (C	ontinued)						
						V Time	Periods					
Foreign Exchange	Total Period	Sub-F	Periods	Total Period	Sub-P	eriods	Total Period	Sub-F	Periods	Total Period	Sub-P	Periods
Rate Currency Pair	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007	1994- 2008	1994- 2004	1994- 2007
		b			t			р			R^2	
	PA	NEL A:	Fore	ign Exchan	ige Rate V	vs. Prim	ary and Sec	ondary S	ample V			
EURO and GBP USD and GPB YEN and GBP	4.498** 1.087 0.815*	1.938 -0.226 -0.207	5.223 -0.567 0.028	4.135 1.725 2.202	0.416 -0.218 -0.373	2.372 -0.691 0.056	$0.006 \\ 0.108 \\ 0.046$	0.718 0.833 0.718	0.064 0.503 0.956	0.740 0.186 0.272	$0.080 \\ 0.005 \\ 0.015$	0.529 0.038 2.6E-04
		PA	NEL B:	Foreign	n Exchang	e Rate V	vs. Control	Group V				
EURO and GBP USD and GPB YEN and GBP	3.023 0.534 0.505	1.042 -0.181 -0.133	4.526 -0.472 0.146	2.403 1.021 1.645	0.202 -0.202 -0.277	1.840 -0.616 0.312	0.053 0.326 0.124	0.859 0.844 0.788	0.125 0.549 0.760	0.490 0.074 0.172	$0.020 \\ 0.005 \\ 0.008$	$0.404 \\ 0.031 \\ 0.008$
		PAN	NEL C:	Foreign	Exchange	e Rate V v	vs. Primary	Sample V	7			
EURO and GBP USD and GPB YEN and GBP	1.928* 1.928* 1.458**	0.229 0.229 0.366	-0.363 -0.363 0.453	5.639 2.718 3.891	0.417 0.232 0.707	1.994 -0.467 0.993	0.001 0.018 0.002	0.717 0.822 0.497	0.103 0.649 0.340	0.841 0.362 0.538	$0.080 \\ 0.006 \\ 0.053$	0.443 0.018 0.076

The Table P.18 time series regression slopes show that for some selected foreign exchange currency pair rate Historical Value-at-Risk Actuals exhibit statistically significant explanatory power to the Primary and Secondary sample Historical Value-at-Risk Actual for the 1994 to 2008 time period. The Primary sample slopes show statistical significance for all selected rates for the same time period.

The Primary and Secondary sample slopes show statistical significance for the EURO and GBP, and YEN and GBP rates.

This suggests that the EURO and GBP, and YEN and GBP foreign exchange rate Historical Value-at-Risk Actual variables reacted in relation to the Primary and Secondary sample market price return Historical Value-at-Risk Actual variable significantly after the 2005 accounting change, than the Control group reactions.

APPENDIX Q THE FAMA AND FRENCH BASELINE REGRESSIONS

This appendix presents analysis and results for the Fama and French (2008) baseline regression. This baseline regression tests the relation between the market value and the book-to-market ratio to expected returns.

Q.1 The Fama and French Baseline Regressions that Use Market Value and Book-to-Market Ratio to Predict Market Price Returns

The Fama and French (2008) baseline multiple regression analysis approach, for the market value and book-to-market ratio to predict future market price returns, is presented in Table Q.1. This analysis is presented for the time range 1994 to 2007. In the regressions, the 1 year forward market price return variable is tested as the dependent variable, and the market value and the book-to-market ratio variables are tested as independent variables.

Table Q.1 **Regressions for Market Value (MVt) and Book-to-Market Ratio** (BM_t) to Predict Yearly Market Price Returns (dM_{t+1}) using the Fama and McBeth t-statistic

Yearly Regression

Table Q.1 Panel A to Panel D show the results for the multiple regressions that test the market value and the book-to-market ratio to 1 year forward market price return. The regression results presented in Table Q.1 for each time period are produced from the average cross-sectional regressions specified in equation (0.15) and presented in Table Q.2. Table Q.1 Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.

Yearly Regression Table Description

The Table Q.1 columns represent the following:

The Total Period column represents the average regression results for the year range 1994 to 2007, with a single year represented by the variable value at 31st December. The Sub-Period column represents average regression results from 1994 to 2004, with a single year represented by the variable value at 31st December.

The Measure column represents, for the two independent variable data panels, the average regression results for the total and sub-time periods. This column presents the averaged coefficients, related statistics and the Fama and McBeth (1973) t-statistic, calculated using equation (0.16), for the regression specified in equation (0.15). Regressions are estimated on a yearly basis for t = 1, 2, ..., T and represent the time range 1992 to 2007 (See Table Q.2). The results from the series of regressions are applied to produce the time period averages and the Fama and McBeth *t-statistic*. The one time period regression follows the model:

 $dM_{i,t+1} = a + b_1 MV_{i,t} + b_2 BM_{i,t} + e_{i,t+1}$ Where: $dM_{i,t+1}$ is the one year forward log Market Price Return $(M_{t,t+1})$ for the *i*th firm at time t + 1, $MV_{i,t}$ is the market value (MV_t) for the *i*th firm at time t, $BM_{i,t}$ is the book-to-market ratio (BM_t) for the *i*th firm at time t, and $e_{i,t+1}$ is the regression error term.

	Measure		Total Period	Sub-Period	
			1992-2007	1992-2004	
	Ν	/larket Valu	ue (MV _t)		
	Slope	b SD(4)	-0.014	-0.005	
		SD(b) Vears(T)	0.044	0.038	
		t (b)	-1.303	-0.443	
Regression					
	Intercept	а	0.185	0.176	
	1	SD(a)	0.360	0.400	
		Years(T)	16	13	
		t (a)	2.059	1.586	
Model	R^2	?	0.085	0.055	
Model	Adjuste	$ed R^2$	0.066	0.054	

Table Q.1 (Continued)

PANEL A Regression for Market Value (MV_t) and Book-to-Market Ratio (BM_t) to Predict Yearly Market Price Returns (dM_{t+1}) for the Primary and Secondary Sample for 1992 to 2007 using the Fama and McBeth *t-statistic*

Book-to-Market Ratio (BM_t)

	Slope	b SD(b)	0.667 1.055	0.728 1.079
		Years(T)	16	13
		t (b)	2.529	2.432
Regression		·		
	Intercept	а	0.185	0.176
	*	SD(a)	0.360	0.400
		Years(T)	16	13
		t (a)	2.059	1.586
		2	0.095	0.055
Model	R	2	0.085	0.055
mouel	Adjuste	ed R ²	0.066	0.054

Panel notes:

The sample consists of 16 firms in the LSE (5 banks and 11 banking related firms) that adopted the IFRS standards in 2005. Observations are based on yearly cross-sectional regressions presented in Table Q.2 from 1992 to 2007.

	Measure		Total Period	Sub-Period
	liteusure		1992-2007	1992-2004
	Ν	Market Val	ue (MV _t)	
	Slope	b SD(b) Years(T) t (b)	0.001 0.063 16 0.076	0.002 0.069 13 0.113
Regression	Intercept	a SD(a) Years(T) t (a)	-0.084 0.511 16 -0.655	-0.074 0.556 13 -0.478
Model	R ² Adjust	2 ed R^{2}	0.358 0.212	0.394 0.256

PANEL B Regression for Market Value (MV_t) and Book-to-Market Ratio (BM_t) to Predict Yearly Market Price Returns (dM_{t+1}) for the Control Group for 1992 to 2007 using the Fama and McBeth *t-statistic*

Table Q.1 (Continued)

Book-to-Market Ratio (BM_t)

	Slope	b SD(b) Years(T) t (b)	0.710 1.002 16 2.834	0.767 0.976 13 2.835
Regression				
	Intercept	а	-0.084	-0.074
	1	SD(a)	0.511	0.556
		Years(T)	16	13
		t (a)	-0.655	-0.478
		,		
Model	R^2	;	0.358	0.394
	Adjusted R ²		0.212	0.256

Panel notes:

The sample consists of 16 firms in the LSE (5 banks and 11 banking related firms) that adopted the IFRS standards in 2005. Observations are based on yearly cross-sectional regressions presented in Table Q.2 from 1992 to 2007.

	Measure		Total Period	Sub-Period
Weasure		1992-2007	1992-2004	
	Ν	Aarket Val	ue (MV _t)	
	Slope	b SD(b) Years(T) t (b)	0.005 0.300 16 0.061	-0.048 0.167 13 -1.029
Regression	Intercept	a SD(a) Years(T) t (a)	-0.008 3.633 16 -0.009	0.688 1.873 13 1.325
Model	R ² Adjusted R ²		0.692 0.354	0.723 0.408

PANEL C Regression for Market Value (MV_t) and Book-to-Market Ratio (BM_t) to Predict Yearly Market Price Returns (dM_{t+1}) for the Primary Sample for 1992 to 2007 using the Fama and McBeth *t-statistic*

Table Q.1 (Continued)

Book-to-Market Ratio (BM_t)

	Slope	b SD(b) Years(T) t (b)	0.067 1.008 16 0.267	0.190 1.034 13 0.664
Regression				
	Intercept	а	-0.008	0.688
	1	SD(a)	3.633	1.873
		Years(T)	16	13
		t (a)	-0.009	1.325
		2	0.602	0.722
Model	ĸ	2	0.092	0.725
	Adjusted R ²		0.354	0.408

Panel notes:

The sample consists of 16 firms in the LSE (5 banks and 11 banking related firms) that adopted the IFRS standards in 2005. Observations are based on yearly cross-sectional regressions presented in Table Q.2 from 1992 to 2007.

edict Yearly	92 to 2007 us	e Returns (c ing the Fam	a and McBeth <i>t-std</i>	ndary Sample i <i>itistic</i>
	Measure		Total Period	Sub-Period
Wieasure		1992-2007	1992-2004	
	Ν	/larket Valu	ue (MV _t)	
	Slope	b SD(b) Years(T) t (b)	0.001 0.063 16 0.076	0.002 0.069 13 0.113
Regression	Intercept	a SD(a) Years(T) t (a)	-0.084 0.511 16 -0.655	-0.074 0.556 13 -0.478
Model	R^2 Adjusted R^2		0.358 0.212	0.394 0.256

Table Q.1 (Continued)

PANEL D Regression for Market Value (MV_t) and Book-to-Market Ratio (BM_t) to Predict Yearly Market Price Returns (dM_{t+1}) for the Secondary Sample for 1992 to 2007 using the Fama and McBeth *t-statistic*

Book-to-Market Ratio (BM_t)

	Slope	b SD(b) Years(T)	0.667 1.055 16	0.728 1.079 13
Regression		t (b)	2.529	2.432
	Intercept	а	-0.084	-0.074
	*	SD(a)	0.511	0.556
		Years(T)	16	13
		t (a)	-0.655	-0.478
		2	0.358	0 394
Model	Adjusted R^2		0.212	0.256

Panel notes:

The sample consists of 16 firms in the LSE (5 banks and 11 banking related firms) that adopted the IFRS standards in 2005. Observations are based on yearly cross-sectional regressions presented in Table Q.2 from 1992 to 2007.

Table Q.2Regressions for Market Value (MVt) and Book-to-Market Ratio
(BMt) to Predict Yearly Market Price Returns (dMt+1)

Cross-Sectional Regression

Table Q.2 Panels A1 to D2 show the results for the cross-sectional multiple regressions that test the market value and book-to-market variables to the 1 year forward market price return by applying the regression specified in equation (O.15). In the regressions, the 1 year forward market price return variable is tested as the dependent variable, and the market value and the book-to-market ratio variables are tested as independent variables.

Table Q.2 Panels A1 and A2 present results for the Primary and Secondary sample; Panels B1 and B2 present results for the Control group; Panels C1 and C2 present results for the Primary sample; and, Panels D1 and D2 present results for the Secondary sample.

In Table Q.2 for the 1 year forward market price return dependent variable: Panels A1; B1; C1; and, D1 represent the multiple regression results for the independent variable, market value; and, Panels A2; B2; C2; and, D2 represent the multiple regression results for the independent variable, the book-to-market ratio.

Cross-Sectional Regression Table Description

The Table Q.2 columns represent the following:

Year is the *panel data* year represented by the number of days from 1st January to 31st December, and also represents the 31st December variable record date for each year.

The Regression column presents the coefficients and related statistics for the regression specified in equation (0.15) and follows the model:

$dM_{i,t+1} = a + b_1 M V_{i,t} + b_2 B M_{i,t} + e_{i,t+1}$

Where: $dM_{i,t+1}$ is the one year forward log Market Price Return $(M_{t,t+t})$ for the i^{th} firm at time t + 1, $MV_{i,t}$ is the market value (MV_t) for the i^{th} firm at time t, $BM_{i,t}$ is the book-to-market ratio (BM_t) for the i^{th} firm at time t, and $e_{i,t+1}$ is the regression error term. The Lower and Upper are the corresponding coefficient levels predicted at a 99% confidence level.

The Correlation column presents the correlation statistics for the variable pairs. The Descriptive Statistics column presents for the independent variable, the sample mean calculated using equation (O.17) and the sample standard deviation, *SD*, using equation (O.18). The *Obs.* column presents the number of sample firms observed for the year.
							Pri	mary	and Sec	condary :	sampie	10F 195	2 10 2	00/							
							Regressi	on								Corr	elation		Descr	iptive	
Year			Slope				I	ntercept				Mo	odel		dM _{t+1} ,	MVt	MV _t , E	BMt	Stati	stics	Obs.
	Ь	t(b)	p(b)	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	Adj. R ²	s(e)	df(e)	r	<i>p(r)</i>	r	<i>p(r)</i>	Mean	SD	
1992	-0.023	-0.68	0.51	-0.13	0.08	0.698**	9.09	0.00	0.38	0.76	0.07	0.03	0.24	11	-0.276	0.17	-0.251	0.19	-0.07	0.67	14
1993	-0.027	-0.82	0.43	-0.13	0.08	0.203	0.29	0.78	-0.19	0.23	0.02	-0.09	0.19	10	-0.238	0.22	0.048	0.44	-0.53	0.62	13
1994	0.015	0.53	0.61	-0.07	0.10	0.114**	3.84	0.00	0.04	0.39	0.03	0.01	0.16	10	0.116	0.35	0.088	0.39	-0.48	0.55	13
1995	0.072**	6.27	0.00	0.04	0.11	-0.399	1.83	0.09	-0.06	0.25	0.09	0.74	0.07	11	0.884**	0.00	0.045	0.44	-0.64	0.63	14
1996	-0.022	-0.78	0.45	-0.11	0.06	0.375**	3.44	0.00	0.03	0.43	0.02	-0.08	0.18	12	-0.193	0.25	-0.131	0.32	-0.60	0.62	15
1997	-0.038	-1.00	0.34	-0.15	0.08	0.294	0.53	0.60	-0.20	0.29	0.03	-0.07	0.23	12	-0.245	0.19	-0.279	0.16	-0.67	0.73	15
1998	0.001	0.02	0.98	-0.13	0.13	0.325**	4.00	0.00	0.09	0.59	0.02	-0.11	0.25	12	-0.044	0.44	-0.215	0.22	-0.75	0.80	15
1999	-0.053	-1.01	0.33	-0.21	0.10	0.493	0.90	0.38	-0.28	0.52	0.08	-0.05	0.35	13	-0.202	0.23	-0.333	0.10	-0.84	0.77	16
2000	0.025	0.75	0.47	-0.08	0.13	-0.249	-0.70	0.50	-0.32	0.20	0.08	0.06	0.24	13	0.086	0.38	-0.252	0.17	-0.85	0.82	16
2001	0.054	1.26	0.23	-0.08	0.18	-0.678*	-2.50	0.03	-0.64	0.06	0.12	0.03	0.31	13	0.222	0.20	-0.330	0.11	-0.65	0.63	16
2002	-0.015	-0.51	0.62	-0.10	0.07	0.357**	4.69	0.00	0.09	0.42	0.02	-0.08	0.19	13	-0.238	0.19	-0.598**	0.01	-0.32	0.53	16
2003	-0.003	-0.28	0.79	-0.04	0.03	0.139**	4.15	0.00	0.03	0.20	0.01	0.02	0.09	13	-0.263	0.16	-0.525*	0.02	-0.46	0.53	16
2004	-0.045	-1.78	0.10	-0.12	0.03	0.616**	4.37	0.00	0.10	0.52	0.13	0.28	0.18	13	-0.564*	0.01	-0.458*	0.04	-0.51	0.49	16
2005	-0.027*	-2.40	0.03	-0.06	0.01	0.322**	3.40	0.00	0.01	0.22	0.03	0.29	0.08	13	-0.411	0.06	-0.275	0.15	-0.55	0.50	16
2006	-0.025	-0.67	0.52	-0.14	0.09	0.214	0.21	0.84	-0.27	0.31	0.08	0.01	0.27	13	-0.249	0.18	-0.246	0.18	-0.58	0.55	16
2007	-0.115	-1.35	0.20	-0.37	0.14	0.141**	-3.67	0.00	-1.40	-0.15	0.54	0.05	0.62	13	-0.320	0.11	-0.070	0.40	-0.41	0.49	16

PANEL A1: Regression for Market Value (MV_t) from the regression with Book-to-Market Ratio (BM_t) to Predict Yearly Market Price Returns (dM_{t+1}) for the Primary and Secondary Sample for 1992 to 2007

Table Q.2 (Continued)

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 16 observations per yearly regression from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

							Regressio	n								Corr	elation		Descri	iptive	
			Slope				In	tercept				Mo	odel		dM _{t+1} ,	BMt	MV _t , I	BM _t	Statis	stics	Obs.
Year	b	t(b)	p(b)	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	Adj. R ²	s(e)	df(e)	r	<i>p(r)</i>	r	<i>p(r)</i>	Mean	SD	•
1992	0.119	1.17	0.27	-0.20	0.44	0.698**	3.45	0.01	0.07	1.33	0.07	0.03	0.24	11	0.380	0.09	-0.251	0.07	-0.07	0.67	14
1993	0.054	0.62	0.55	-0.22	0.33	0.203	0.87	0.41	-0.54	0.94	0.02	-0.09	0.19	10	0.176	0.28	0.048*	0.03	-0.53	0.62	13
1994	-0.113	-1.39	0.20	-0.37	0.14	0.114	0.57	0.58	-0.52	0.75	0.03	0.01	0.16	10	-0.388	0.10	0.088	0.42	-0.48	0.55	13
1995	-0.004	-0.15	0.88	-0.10	0.09	-0.399**	-4.82	0.00	-0.66	-0.14	0.09	0.74	0.07	11	0.019	0.47	0.045	0.09	-0.62	0.65	14
1996	-0.057	-0.72	0.49	-0.30	0.18	0.375	1.87	0.09	-0.24	0.99	0.02	-0.08	0.18	12	-0.172	0.27	-0.131	0.19	-0.60	0.62	15
1997	-0.047	-0.54	0.60	-0.31	0.22	0.294	1.12	0.29	-0.51	1.10	0.03	-0.07	0.23	12	-0.074	0.40	-0.279	0.49	-0.67	0.73	15
1998	0.071	0.80	0.44	-0.20	0.34	0.325	1.04	0.32	-0.63	1.28	0.02	-0.11	0.25	12	0.230	0.21	-0.215	0.06	-0.68	0.77	15
1999	-0.109	-0.86	0.40	-0.49	0.27	0.493	1.25	0.23	-0.69	1.68	0.08	-0.05	0.35	13	-0.148	0.29	-0.333*	0.03	-0.84	0.77	16
2000	0.131	1.66	0.12	-0.11	0.37	-0.249	-0.94	0.37	-1.05	0.55	0.08	0.06	0.24	13	0.382	0.07	-0.252	0.22	-0.85	0.82	16
2001	0.179	1.31	0.21	-0.23	0.59	-0.678	-2.07	0.06	-1.66	0.31	0.12	0.03	0.31	13	0.241	0.18	-0.330	0.11	-0.65	0.63	16
2002	0.039	0.33	0.74	-0.31	0.39	0.357	1.74	0.11	-0.26	0.98	0.02	-0.08	0.19	13	0.214	0.21	-0.598	0.27	-0.32	0.53	16
2003	0.056	1.15	0.27	-0.09	0.20	0.139	1.55	0.15	-0.13	0.41	0.01	0.02	0.09	13	0.387	0.07	-0.525	0.27	-0.46	0.53	16
2004	0.118	1.11	0.29	-0.20	0.44	0.616**	3.32	0.01	0.06	1.17	0.13	0.28	0.18	13	0.475*	0.03	-0.458	0.15	-0.51	0.49	16
2005	-0.092	-2.13	0.05	-0.22	0.04	0.322**	3.59	0.00	0.05	0.59	0.03	0.29	0.08	13	-0.332	0.10	-0.275**	0.01	-0.55	0.50	16
2006	0.145	1.11	0.29	-0.25	0.54	0.214	0.70	0.50	-0.71	1.14	0.08	0.01	0.27	13	0.337	0.10	-0.246	0.45	-0.58	0.55	16
2007	-0.361	-1.11	0.29	-1.34	0.62	0.141	0.20	0.85	-1.99	2.27	0.54	0.05	0.62	13	-0.255	0.17	-0.070	0.23	-0.41	0.49	16

PANEL A2: Regression for Book-to-Market Ratio (BM_t) with Market Value (MV_t) to Predict Yearly Market Price Returns (dM_{t+1}) for the Primary and Secondary Sample for 1992 to 2007

Table Q.2 (Continued)

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 16 observations per yearly regression from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

						I	Regressio	n		_						Corr	elation		Descr	iptive	
Year		:	Slope				I	ntercept				Mo	odel		dM _{t+1} ,	MVt	MV _t , E	BM _t	Stati	stics	Obs.
	Ь	<i>t(b)</i>	p(b)	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	Adj. R ²	s(e)	df(e)	r	<i>p(r)</i>	r	<i>p(r)</i>	Mean	SD	•
1992	-0.072**	-3.71	0.00	-0.14	-0.01	0.597**	4.84	0.00	0.20	1.00	0.68	0.61	0.10	9	-0.73**	0.00	-0.080	0.40	5.24	1.56	12
1993	-0.002	-0.05	0.96	-0.11	0.11	-0.140	-0.76	0.47	-0.76	0.48	0.35	0.19	0.11	8	0.261	0.22	0.466	0.07	5.84	1.19	11
1994	0.043*	2.74	0.03	-0.01	0.10	-0.138	-1.46	0.18	-0.46	0.18	0.49	0.36	0.06	8	0.687**	0.01	0.076	0.41	5.80	1.11	11
1995	0.081	1.98	0.08	-0.06	0.22	-0.584*	-2.33	0.05	-1.43	0.26	0.46	0.33	0.15	8	0.548*	0.04	0.083	0.40	5.90	1.16	11
1996	0.085**	3.47	0.01	0.01	0.16	-0.428*	-2.92	0.02	-0.90	0.05	0.60	0.52	0.11	9	0.76**	0.00	0.145	0.33	5.73	1.37	12
1997	0.108*	2.43	0.04	-0.04	0.25	-1.030	-2.12	0.06	-2.61	0.55	0.40	0.26	0.20	9	0.583*	0.02	-0.385	0.11	5.85	1.49	12
1998	-0.118	-2.03	0.07	-0.31	0.07	0.692	1.22	0.25	-1.15	2.54	0.47	0.35	0.29	9	-0.639*	0.01	-0.392	0.10	5.73	1.64	12
1999	0.034	0.63	0.54	-0.14	0.21	-0.807	-1.53	0.16	-2.52	0.91	0.33	0.18	0.23	9	-0.064	0.42	-0.407	0.09	6.04	1.41	12
2000	0.031	1.35	0.21	-0.04	0.11	-0.404	-1.70	0.12	-1.17	0.37	0.22	0.04	0.08	9	0.458	0.07	-0.669**	0.01	6.01	1.40	12
2001	-0.075	-1.81	0.10	-0.21	0.06	0.110	0.30	0.77	-1.07	1.29	0.38	0.24	0.16	9	-0.614*	0.02	-0.602*	0.02	5.79	1.43	12
2002	-0.020	-0.61	0.56	-0.13	0.09	0.299	1.35	0.21	-0.42	1.02	0.08	-0.13	0.14	9	-0.232	0.23	-0.198	0.27	5.47	1.32	12
2003	-0.044*	-2.99	0.02	-0.09	0.00	0.331**	3.36	0.01	0.01	0.65	0.52	0.41	0.06	9	-0.69**	0.01	0.007	0.49	5.71	1.30	12
2004	-0.025	-0.76	0.46	-0.13	0.08	0.543*	3.05	0.01	-0.04	1.12	0.16	-0.03	0.12	9	-0.346	0.14	0.409	0.09	5.82	1.24	12
2005	-0.037	-0.95	0.37	-0.17	0.09	0.108	0.49	0.64	-0.61	0.82	0.52	0.42	0.14	9	0.137	0.34	0.477	0.06	6.11	1.19	12
2006	0.011	0.15	0.89	-0.24	0.26	-0.008	-0.02	0.99	-1.38	1.36	0.00	-0.22	0.27	9	0.025	0.47	0.462	0.07	6.14	1.20	12
2007	0.017	0.32	0.76	-0.16	0.19	-0.480	-1.37	0.20	-1.61	0.65	0.07	-0.14	0.22	9	0.149	0.32	-0.203	0.26	6.06	1.25	12

PANEL B1: Regression for Market Value (MV_t) from the regression with Book-to-Market Ratio (BM_t) to Predict Yearly Market Price Returns (dM_{t+1}) for the Control Group for 1992 to 2007

Table Q.2 (Continued)

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The Control group consists of a maximum of 12 observations per yearly regression from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

555

							Regressi	on								Corr	elation		Descr	iptive	
Year			Slope				I	Intercept				Mo	odel		dM _{t+1} ,	BMt	MV _t , E	BMt	Stati	stics	Obs.
	Ь	t(b)	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	Adj. R ²	s(e)	df(e)	r	<i>p(r)</i>	r	<i>p(r)</i>	Mean	SD	•
1992	0.804	2.07	0.07	-0.46	2.06	0.597**	4.84	0.00	0.20	1.00	0.68	0.61	0.10	9	0.446	0.07	-0.080	0.40	0.14	0.08	12
1993	0.858	1.85	0.10	-0.69	2.41	-0.140	-0.76	0.47	-0.76	0.48	0.35	0.19	0.11	8	0.59*	0.03	0.466	0.07	0.07	0.09	11
1994	-0.143	-0.49	0.64	-1.12	0.83	-0.138	-1.46	0.18	-0.46	0.18	0.49	0.36	0.06	8	-0.072	0.42	0.076	0.41	0.08	0.06	11
1995	1.105	1.56	0.16	-1.27	3.48	-0.584*	-2.33	0.05	-1.43	0.26	0.46	0.33	0.15	8	0.449	0.08	0.083	0.40	0.11	0.07	11
1996	0.259	0.79	0.45	-0.81	1.33	-0.428*	-2.92	0.02	-0.90	0.05	0.60	0.52	0.11	9	0.274	0.19	0.145	0.33	0.17	0.10	12
1997	1.956	0.91	0.39	-5.02	8.93	-1.030	-2.12	0.06	-2.61	0.55	0.40	0.26	0.20	9	-0.007	0.49	-0.385	0.11	0.15	0.03	12
1998	1.607	0.99	0.35	-3.69	6.91	0.692	1.22	0.25	-1.15	2.54	0.47	0.35	0.29	9	0.471	0.06	-0.392	0.10	0.21	0.06	12
1999	3.095	2.08	0.07	-1.74	7.93	-0.807	-1.53	0.16	-2.52	0.91	0.33	0.18	0.23	9	0.545*	0.03	-0.407	0.09	0.20	0.05	12
2000	0.214	0.29	0.78	-2.19	2.62	-0.404	-1.70	0.12	-1.17	0.37	0.22	0.04	0.08	9	-0.243	0.22	-0.669**	0.01	0.16	0.04	12
2001	0.082	0.09	0.93	-2.98	3.14	0.110	0.30	0.77	-1.07	1.29	0.38	0.24	0.16	9	0.388	0.11	-0.602*	0.02	0.17	0.06	12
2002	0.294	0.50	0.63	-1.64	2.22	0.299	1.35	0.21	-0.42	1.02	0.08	-0.13	0.14	9	0.201	0.27	-0.198	0.27	0.16	0.07	12
2003	0.269	0.89	0.40	-0.72	1.26	0.331**	3.36	0.01	0.01	0.65	0.52	0.41	0.06	9	0.200	0.27	0.007	0.49	0.16	0.06	12
2004	-0.428	-0.66	0.53	-2.55	1.69	0.543*	3.05	0.01	-0.04	1.12	0.16	-0.03	0.12	9	-0.324	0.15	0.409	0.09	0.16	0.06	12
2005	1.971	3.09	0.01	-0.10	4.04	0.108	0.49	0.64	-0.61	0.82	0.52	0.42	0.14	9	0.69**	0.01	0.477	0.06	0.11	0.07	12
2006	-0.344	-0.17	0.87	-6.82	6.14	-0.008	-0.02	0.99	-1.38	1.36	0.00	-0.22	0.27	9	-0.040	0.45	0.462	0.07	0.12	0.05	12
2007	-0.239	-0.65	0.53	-1.43	0.95	-0.480	-1.37	0.20	-1.61	0.65	0.07	- 0.14	0.22	9	-0.236	0.23	-0.203	0.26	0.19	0.18	12

PANEL B2: Regression for Book-to-Market Ratio (BMt) with Market Value (MVt) to Predict Yearly Market Price Returns (dMt+1) for the Control Group for 1992 to 2007

Table Q.2 (Continued)

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The Control group consists of a maximum of 12 observations per yearly regression from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

]	Regression	L								Corr	relation		Descri	iptive	
Year			Slope				In	tercept				Mo	odel		dM _{t+1} ,	MVt	MV _t ,	BMt	Stati	stics	Obs.
	Ь	t(b)	p(b)	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	Adj. R ²	s(e)	df(e)	r	<i>p(r)</i>	r	<i>p(r)</i>	Mean	SD	
1992	-0.152	-3.73	0.17	-2.75	2.45	1.837	5.82	0.11	-18.25	21.93	0.93	0.80	0.05	1	-0.844	0.08	-0.474	0.26	7.85	0.79	4
1993	-0.085	-1.12	0.46	-4.91	4.74	1.135	1.86	0.31	-37.66	39.93	0.96	0.87	0.08	1	-0.589	0.21	-0.394	0.30	8.52	0.69	4
1994	-0.484*	-42.25	0.02	-1.21	0.25	5.502*	48.15	0.01	-1.77	12.78	1.00	1.00	0.01	1	-0.439	0.28	0.603	0.20	8.48	0.59	4
1995	0.276**	90.49	0.01	0.08	0.47	-2.479**	-79.72	0.01	-4.46	-0.50	1.00	1.00	0.00	1	0.833	0.08	0.754	0.12	8.81	0.53	4
1996	0.044	0.29	0.82	-9.62	9.71	0.782	0.53	0.69	-93.27	94.83	0.88	0.63	0.15	1	0.244	0.38	0.153	0.42	9.03	0.59	4
1997	-0.048	-0.37	0.78	-8.32	8.22	1.237	1.42	0.39	-54.06	56.53	0.77	0.32	0.11	1	-0.726	0.14	-0.696	0.15	9.27	0.67	4
1998	-0.051	-1.00	0.50	-3.26	3.16	-1.601	-1.99	0.30	-52.74	49.54	0.97	0.92	0.05	1	-0.536	0.23	0.397	0.30	9.25	0.58	4
1999	-0.029	-0.24	0.83	-1.19	1.13	0.568	0.48	0.68	-11.15	12.28	0.09	-0.82	0.19	2	-0.229	0.36	-0.287	0.32	10.13	0.84	5
2000	0.008	0.06	0.96	-1.28	1.29	-0.083	-0.06	0.96	-13.56	13.39	0.00	-0.99	0.19	2	0.035	0.48	-0.175	0.39	10.48	0.76	5
2001	0.067	1.20	0.35	-0.48	0.62	-0.346	-0.65	0.58	-5.67	4.97	0.90	0.80	0.08	2	-0.145	0.41	-0.427	0.24	10.46	0.79	5
2002	-0.039	-0.58	0.62	-0.70	0.62	0.847	1.19	0.35	- 6.19	7.88	0.62	0.24	0.10	2	-0.198	0.37	0.076	0.45	10.21	0.77	5
2003	-0.060	-1.66	0.24	-0.42	0.30	0.997	2.06	0.18	-3.81	5.80	0.68	0.36	0.05	2	-0.258	0.34	0.583	0.15	10.42	0.81	5
2004	-0.067	-0.55	0.64	-1.27	1.14	0.553	0.31	0.79	-17.31	18.42	0.60	0.20	0.10	2	-0.752	0.07	0.846*	0.04	10.50	0.81	5
2005	-0.090	-1.47	0.28	-0.70	0.52	1.130	1.48	0.28	-6.47	8.73	0.59	0.18	0.06	2	-0.755	0.07	0.663	0.11	10.61	0.70	5
2006	-0.190	-0.62	0.60	-3.23	2.85	1.833	0.49	0.67	-35.21	38.87	0.26	-0.48	0.30	2	-0.510	0.19	0.627	0.13	10.74	0.63	5
2007	0.971	2.39	0.14	-3.07	5.01	-12.045	-2.72	0.11	-56.03	31.94	0.82	0.65	0.40	2	0.381	0.26	0.442	0.23	10.60	0.55	5

PANEL C1: Regression for Market Value (MV_t) from the regression with Book-to-Market Ratio (BM_t) to Predict Yearly Market Price Returns (dM_{t+1}) for the Primary Sample for 1992 to 2007

Table Q.2 (Continued)

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 5 observations per yearly regression from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

											2007											
								Regressio	n								Corr	relation		Descri	iptive	
Yea	ır		Sloj	pe				1	ntercept	t			Мо	del		$dM_{t\!+\!1},$	BM _t	MV _t , I	BM _t	Statis	stics	Obs.
	b	t(l	b) p(b)	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	Adj. R ²	s(e)	df(e)	r	<i>p(r)</i>	r	<i>p(r)</i>	Mean	SD	
199	-0.17	2 -1.8	1 0.	32	-6.22	5.87	1.837	5.82	0.11	-18.25	21.93	0.93	0.80	0.05	1	0.380	0.49	-0.014	0.26	-0.14	0.34	4
199	0.657	3.71	l 0.	17	-10.61	11.92	1.135	1.86	0.31	-37.66	39.93	0.96	0.87	0.08	1	0.176*	0.03	0.949	0.30	-0.71	0.30	4
199	4 1.758	* 42.5	55 0.	01	-0.87	4.39	5.502*	48.15	0.01	-1.77	12.78	1.00	1.00	0.01	1	-0.388	0.27	0.452	0.20	-0.61	0.16	4
199	-0.34	5* -51.	89 0.	01	-0.77	0.08	-2.479**	-79.72	0.01	-4.46	-0.50	1.00	1.00	2E-03	1	0.019	0.37	0.264	0.12	-0.79	0.24	4
199	1.085	2.56	5 0.	24	-25.93	28.10	0.782	0.53	0.69	-93.27	94.83	0.88	0.63	0.15	1	-0.172*	0.04	0.930	0.42	-0.88	0.21	4
ر <mark>م</mark> 199	0.814	1.05	5 0.	49	-48.71	50.34	1.237	1.42	0.39	-54.06	56.53	0.77	0.32	0.11	1	-0.074	0.07	0.862	0.15	-1.02	0.11	4
∞ 199	-2.63	7 -5.0	7 0.	12	-35.71	30.44	-1.601	-1.99	0.30	-52.74	49.54	0.97	0.92	0.05	1	0.23*	0.01	-0.973	0.30	-0.95	0.06	4
199	9 0.143	0.29	0.	80	-4.79	5.08	0.568	0.48	0.68	-11.15	12.28	0.09	-0.82	0.19	2	-0.148	0.34	0.251	0.32	-1.30	0.20	5
200	0.033	0.07	7 0.	95	-4.42	4.49	-0.083	-0.06	0.96	-13.56	13.39	4E-03	-0.99	0.19	2	0.382	0.47	0.044	0.39	-1.05	0.22	5
200	0.672	4.17	7 0.	05	-0.93	2.27	-0.346	-0.65	0.58	-5.67	4.97	0.90	0.80	0.08	2	0.241*	0.02	0.909	0.24	-0.92	0.27	5
200	0.376	1.74	t 0.	22	-1.77	2.52	0.847	1.19	0.35	-6.19	7.88	0.62	0.24	0.10	2	0.214	0.07	0.744	0.45	-0.73	0.24	5
200	0.370	1.96	5 0.	19	-1.51	2.25	0.997	2.06	0.18	-3.81	5.80	0.68	0.36	0.05	2	0.387	0.20	0.486	0.15	-0.83	0.15	5
200	-0.27	8 -0.4	0 0.	73	-7.11	6.56	0.553	0.31	0.79	-17.31	18.42	0.60	0.20	0.10	2	0.475	0.08	-0.732*	0.04	-0.85	0.14	5
200	0.069	0.33	3 0.	77	-1.99	2.13	1.130	1.48	0.28	-6.47	8.73	0.59	0.18	0.06	2	-0.332	0.26	-0.388	0.11	-0.77	0.21	5
200	-0.04	5 -0.0	6 0.	96	-8.17	8.08	1.833	0.49	0.67	-35.21	38.87	0.26	-0.48	0.30	2	0.337	0.28	-0.345	0.13	-0.79	0.24	5
200	-1.42	4 -2.7	8 0.	11	-6.50	3.65	-12.045	-2.72	0.11	-56.03	31.94	0.82	0.65	0.40	2	-0.255	0.16	-0.570	0.23	-0.48	0.44	5

PANEL C2: Regression for Book-to-Market Ratio (BM_t) with Market Value (MV_t) to Predict Yearly Market Price Returns (dM_{t+1}) for the Primary Sample for 1992 to 2007

Table Q.2 (Continued)

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 5 observations per yearly regression from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

								Sec	ondary	Sample	for 1992	2 to 200) 7								
						I	Regression	n								Corr	elation		Descri	ptive	
Year		5	Slope				I	ntercept				Mo	odel		dM _{t+1} , 1	MVt	MV _t , B	Mt	Statis	stics	Obs.
	Ь	t(b)	p(b)	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	Adj. R ²	s(e)	df(e)	r	<i>p(r)</i>	r	<i>p(r)</i>	Mean	SD	•
1992	-0.072**	-3.71	5E-03	-0.14	-0.01	0.597**	4.84	9E-04	0.20	1.00	0.68	0.61	0.10	9	-0.73**	4E-03	-0.080	0.40	5.24	1.56	12
1993	-0.002	-0.05	0.96	-0.11	0.11	-0.140	-0.76	0.47	-0.76	0.48	0.35	0.19	0.11	8	0.261	0.22	0.466	0.07	5.84	1.19	11
1994	0.043*	2.74	0.03	-0.01	0.10	-0.138	-1.46	0.18	-0.46	0.18	0.49	0.36	0.06	8	0.687**	0.01	0.076	0.41	5.80	1.11	11
1995	0.081	1.98	0.08	-0.06	0.22	-0.584*	-2.33	0.05	-1.43	0.26	0.46	0.33	0.15	8	0.548*	0.04	0.083	0.40	5.90	1.16	11
1996	0.085**	3.47	0.01	0.01	0.16	-0.428*	-2.92	0.02	-0.90	0.05	0.60	0.52	0.11	9	0.76**	2E-03	0.145	0.33	5.73	1.37	12
1997	0.108*	2.43	0.04	-0.04	0.25	-1.030	-2.12	0.06	-2.61	0.55	0.40	0.26	0.20	9	0.583*	0.02	-0.385	0.11	5.85	1.49	12
1998	-0.118	-2.03	0.07	-0.31	0.07	0.692	1.22	0.25	-1.15	2.54	0.47	0.35	0.29	9	-0.639*	0.01	-0.392	0.10	5.73	1.64	12
1999	0.034	0.63	0.54	-0.14	0.21	-0.807	-1.53	0.16	-2.52	0.91	0.33	0.18	0.23	9	-0.064	0.42	-0.407	0.09	6.04	1.41	12
2000	0.031	1.35	0.21	-0.04	0.11	-0.404	-1.70	0.12	-1.17	0.37	0.22	0.04	0.08	9	0.458	0.07	-0.669**	0.01	6.01	1.40	12
2001	-0.075	-1.81	0.10	-0.21	0.06	0.110	0.30	0.77	-1.07	1.29	0.38	0.24	0.16	9	-0.614*	0.02	-0.602*	0.02	5.79	1.43	12
2002	-0.020	-0.61	0.56	-0.13	0.09	0.299	1.35	0.21	-0.42	1.02	0.08	-0.13	0.14	9	-0.232	0.23	-0.198	0.27	5.47	1.32	12
2003	-0.044*	-2.99	0.02	-0.09	4E-03	0.331**	3.36	0.01	0.01	0.65	0.52	0.41	0.06	9	-0.69**	0.01	0.007	0.49	5.71	1.30	12
2004	-0.025	-0.76	0.46	-0.13	0.08	0.543*	3.05	0.01	-0.04	1.12	0.16	-0.03	0.12	9	-0.346	0.14	0.409	0.09	5.82	1.24	12
2005	-0.037	-0.95	0.37	-0.17	0.09	0.108	0.49	0.64	-0.61	0.82	0.52	0.42	0.14	9	0.137	0.34	0.477	0.06	6.11	1.19	12
2006	0.011	0.15	0.89	-0.24	0.26	-0.008	-0.02	0.99	-1.38	1.36	4E-03	-0.22	0.27	9	0.025	0.47	0.462	0.07	6.14	1.20	12
2007	0.017	0.32	0.76	-0.16	0.19	-0.480	-1.37	0.20	-1.61	0.65	0.07	-0.14	0.22	9	0.149	0.32	-0.203	0.26	6.06	1.25	12

PANEL D1: Regression for Market Value (MVt) from the regression with Book-to-Market Ratio (BMt) to Predict Yearly Market Price Returns (dMt+1) for the Secondary Sample for 1992 to 2007

Table Q.2 (Continued)

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 11 observations per yearly regression from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

							Regressi	on								Corr	elation		Descr	iptive	
Year			Slope				I	ntercept				Mo	del		dM _{t+1} ,	BMt	MV _t , E	BMt	Stati	stics	Obs.
	Ь	t(b)	р(b)	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	Adj. R ²	s(e)	df(e)	r	<i>p(r)</i>	r	<i>p(r)</i>	Mean	SD	
1992	0.119	2.07	0.07	-0.46	2.06	0.597**	4.84	9E-04	0.20	1.00	0.68	0.61	0.10	9	0.446	0.07	-0.080	0.40	0.14	0.08	12
1993	0.054	1.85	0.10	-0.69	2.41	-0.140	-0.76	0.47	-0.76	0.48	0.35	0.19	0.11	8	0.59*	0.03	0.466	0.07	0.07	0.09	11
1994	-0.113	-0.49	0.64	-1.12	0.83	-0.138	-1.46	0.18	-0.46	0.18	0.49	0.36	0.06	8	-0.072	0.42	0.076	0.41	0.08	0.06	11
1995	-0.004	1.56	0.16	-1.27	3.48	-0.584*	-2.33	0.05	-1.43	0.26	0.46	0.33	0.15	8	0.449	0.08	0.083	0.40	0.11	0.07	11
1996	-0.057	0.79	0.45	-0.81	1.33	-0.428*	-2.92	0.02	-0.90	0.05	0.60	0.52	0.11	9	0.274	0.19	0.145	0.33	0.17	0.10	12
1997	-0.047	0.91	0.39	-5.02	8.93	-1.030	-2.12	0.06	-2.61	0.55	0.40	0.26	0.20	9	-0.007	0.49	-0.385	0.11	0.15	0.03	12
1998	0.071	0.99	0.35	-3.69	6.91	0.692	1.22	0.25	-1.15	2.54	0.47	0.35	0.29	9	0.471	0.06	-0.392	0.10	0.21	0.06	12
1999	-0.109	2.08	0.07	-1.74	7.93	-0.807	-1.53	0.16	-2.52	0.91	0.33	0.18	0.23	9	0.545*	0.03	-0.407	0.09	0.20	0.05	12
2000	0.131	0.29	0.78	-2.19	2.62	-0.404	-1.70	0.12	-1.17	0.37	0.22	0.04	0.08	9	-0.243	0.22	-0.669**	0.01	0.16	0.04	12
2001	0.179	0.09	0.93	-2.98	3.14	0.110	0.30	0.77	-1.07	1.29	0.38	0.24	0.16	9	0.388	0.11	-0.602*	0.02	0.17	0.06	12
2002	0.039	0.50	0.63	-1.64	2.22	0.299	1.35	0.21	-0.42	1.02	0.08	-0.13	0.14	9	0.201	0.27	-0.198	0.27	0.16	0.07	12
2003	0.056	0.89	0.40	-0.72	1.26	0.331**	3.36	0.01	0.01	0.65	0.52	0.41	0.06	9	0.200	0.27	0.007	0.49	0.16	0.06	12
2004	0.118	-0.66	0.53	-2.55	1.69	0.543*	3.05	0.01	-0.04	1.12	0.16	-0.03	0.12	9	-0.324	0.15	0.409	0.09	0.16	0.06	12
2005	-0.092	3.09	0.01	-0.10	4.04	0.108	0.49	0.64	-0.61	0.82	0.52	0.42	0.14	9	0.69**	0.01	0.477	0.06	0.11	0.07	12
2006	0.145	-0.17	0.87	-6.82	6.14	-0.008	-0.02	0.99	-1.38	1.36	4E-03	-0.22	0.27	9	-0.040	0.45	0.462	0.07	0.12	0.05	12
2007	-0.361	-0.65	0.53	-1.43	0.95	-0.480	-1.37	0.20	-1.61	0.65	0.07	-0.14	0.22	9	-0.236	0.23	-0.203	0.26	0.19	0.18	12

PANEL D2: Regression for Book-to-Market Ratio (BMt) with Market Value (MVt) to Predict Yearly Market Price Returns (dMt+1) for the Secondary Sample for 1992 to 2007

Table Q.2 (Continued)

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 11 observations per yearly regression from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

Analysis

Applying the Fama and McBeth (1973) *t-statistic*, presented in equation (O.16), the Table Q.1 average yearly regression slopes, for the samples and the control group show that the book-to-market variable exhibits statistically significant explanatory power to expected market price returns. This explanatory power is evident for the 1992 to 2007 and 1992 to 2004 time periods. Examining the Secondary sample results, it is evident that the Primary and Secondary sample significance is attributed to the Secondary sample firms.

Examining the significant regression slopes, the Control group exhibits greater magnitude and statistical significance when compared to the Primary and Secondary sample slopes.

This suggests that the Control group book-to-market ratio has greater predictive *value* for market price returns before and after the 2005 accounting change than the Primary and Secondary sample. The significant slopes for the book-to-market ratio variable and the *one-period forward* market price return show a similar pattern to the Fama and French (2008) base line regressions. For the samples and the control group, however, the yearly slopes do not show significance for the market value variable. In addition, their average yearly regression intercepts, for both the market value and book-to-market ratio to the 1 year forward average returns, do not exhibit statistical significance.

This reveals that the Control group's book-to-market ratio variables reacted in relation to its expected market price return variable before and after the 2005 accounting change. In general, these reactions exhibit greater significance when compared to the Primary and Secondary sample.

APPENDIX R ANALYSIS AND RESULTS FOR BOOK-TO-MARKET RATIO, KEY MARKET TOTALS, KEY ACCOUNTING TOTALS AND HISTORICAL VALUE-AT-RISK ACTUAL

The analysis and results for descriptive statistics and correlations applied to selected variables are presented in this appendix. The selected variables include market price, Historical Value-at-Risk Actual, book-to-market ratio, market value, and the key accounting totals. These key accounting total variables are total equity, total assets, total liabilities and net income. The selected variables are represented by fiscal year-end total levels.

R.1 Descriptive Statistics for Book-to-Market Ratio, Key Accounting and Market Total Variables, and Historical Value-at-Risk Actual Variable

The descriptive statistics for the variables book-to-market ratio, accounting totals on a consolidated basis, market totals that include market price and Historical Value-at-Risk Actual is presented on a yearly basis for the time range 1992 to 2008 in Table R.4⁶¹. In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.⁶²

Analysis

Table R.4 presents the yearly descriptive statistics for the sample average and the standard deviation of sample means for *total variables* that include the key accounting total variables and the market price variable. In addition, Table R.4 presents descriptive statistics for the Historical Value-at-Risk Actual variable.

⁶¹ Information regarding calculated and missing data for these variables is presented in Appendix Z.

⁶² In general, the analysis of this section's results is presented prior to the presentation of the actual results tables.

Table R.4 Panel A shows that the Primary and Secondary variables tested, generally exhibit materially significant increases for both statistical mean and the standard deviation of sample means after 2005. However, the market price mean and its standard deviation levels exhibit little difference before and after 2005. The Panel B Control group variables exhibit less significant material changes for before and after 2005 when compared to the Primary and Secondary sample changes. Both the Primary and Secondary sample and the Control group exhibited significant material increase in average Historical Value-at-Risk Actual after 2005 compared to before.

To determine the changes in the descriptive statistics for the samples and the control group *total* variables for before and after 2005, the percentage changes for the yearly mean and standard deviation of sample means are presented in Table R.1 and Table R.2 respectively. Table R.1 and Table R.2 present percentage changes for the time periods 1992 to 2004, to provide changes in levels for before 2005; and, 1992 to 2007 and 1992 to 2008 to provide changes for after 2005. For the Historical Value-at-Risk Actual variable, the percentage changes are calculated for the time periods: 1994 to 2004; 1994 to 2008.

Table R.1 and Table R.2 Percentage Changes for Mean and Standard Deviation of Sample Means for Total and Historical Value-at-Risk Actual Variable Results from Table R.4

Percentage Change Analysis Table Description

In Table R.1 and Table R.2, the Variable column represents the Description and Symbol columns that present the descriptions and *codes* for the selected variables. The Year column represents the beginning (From) and ending (To) year applied to calculate the percentage changes presented in the Percentage Change (%) column. The Percentage Change column in Table R.1 represents the Mean column. This Mean column presents the mean variable percentage changes rounded to the nearest whole number. The Percentage Change column in Table R.2 represents the Standard Deviation column. This Standard Deviation column presents the percentage changes of the standard deviation of sample means rounded to the nearest whole number. In Table R.1 and Table R.2, the PS column presents the percentage changes for the Control group. The P column presents the percentage changes for the Scolumn presents the per

Table description note:

Refer to this table description for Table R.1 and Table R.2 table details.

Variable		Year	Perc	entage	Change	(%)
Description	Symbol	From - To		М	ean	
F			PS	С	Р	S
Book-to-Market Ratio	TBM	1992 - 2004 1992 - 2007 1992 - 2008	186 195 254	1 5 69	-53 -26 56	156 155 179
Market-to-Book Ratio	TMB	1992 - 2004 1992 - 2007 1992 - 2008	43 31 -7	-1 -3 -32	98 44 -23	22 27 0
Market Price	Р	1992 - 2004 1992 - 2007 1992 - 2008	336 489 287	84 183 92	499 606 246	292 462 307
Market Value	TMV	1992 - 2004 1992 - 2007 1992 - 2008	1,293 1,318 773	52 96 30	1,334 1,337 774	192 308 205
Volume	VOL	1992 - 2004 1992 - 2007 1992 - 2008	-30 12 97	-23 -23 -39	-29 -7 99	192 817 725
Total Shares Outstanding	TS	1992 - 2004 1992 - 2007 1992 - 2008	63 70 181	-15 -32 -33	67 74 192	7 22 40
Total Equity	TE	1992 - 2004 1992 - 2007 1992 - 2008	423 682 753	58 94 28	478 769 859	119 209 173
Total Assets	TA	1992 - 2004 1992 - 2007 1992 - 2008	372 970 1,435	46 76 22	380 990 1,468	74 216 208
Total Liabilities	TL	1992 - 2004 1992 - 2007 1992 - 2008	369 979 1,471	11 18 8	375 994 1,496	55 216 221
Net Income	TI	1992 - 2004 1992 - 2007 1992 - 2008	484 784 -1,520	-29 -10 -12	478 759 -1,555	834 2,343 673
Historical Value-At-Risk Actual	V	1994 - 2004 1994 - 2007 1994 - 2008	-325 -375 -1,925	90 50 -360	-150 -1,050 -5,300	-283 -200 -1,050

Table R.1Percentage Changes for Mean Total and Historical Value-at-Risk
Actual Variable Results from Table R.4

Variable		Year	Per	centage	Change ('	%)
Description	Symbol	From - To	S	tandard	Deviatior	1
	~) 0		PS	С	Р	S
Book-to-Market Ratio	TBM	1992 - 2004 1992 - 2007 1992 - 2008	-95 -95 -90	-22 167 1,111	-82 -6 150	-96 -95 -93
Market-to-Book Ratio	TMB	1992 - 2004 1992 - 2007 1992 - 2008	-14 -17 -10	-29 100 171	-6 92 36	-21 -21 -11
Market Price	Р	1992 - 2004 1992 - 2007 1992 - 2008	290 624 481	104 200 104	353 1,081 710	304 571 444
Market Value	TMV	1992 - 2004 1992 - 2007 1992 - 2008	1,493 1,442 1,051	53 98 34	1,322 1,215 1,150	101 188 106
Volume	VOL	1992 - 2004 1992 - 2007 1992 - 2008	-45 -28 53	4 -17 -44	-50 -56 26	116 824 820
Total Shares Outstanding	TS	1992 - 2004 1992 - 2007 1992 - 2008	63 69 254	-15 -35 -36	50 55 357	9 56 60
Total Equity	TE	1992 - 2004 1992 - 2007 1992 - 2008	494 787 846	64 100 36	480 757 758	65 152 94
Total Assets	TA	1992 - 2004 1992 - 2007 1992 - 2008	339 983 1,425	55 90 33	264 959 1,336	-16 99 100
Total Liabilities	TL	1992 - 2004 1992 - 2007 1992 - 2008	333 986 1,451	0.39 40 11	256 963 1,366	-10 111 103
Net Income	TI	1992 - 2004 1992 - 2007 1992 - 2008	164 299 1,658	-30 -10 -17	49 141 1,724	-17 -7 -58
Historical Value-At-Risk Actual	V	1994 - 2004 1994 - 2007 1994 - 2008	10 -10 210	-13 100 125	-67 -19 210	27 -18 177

Table R.2Percentage Changes for Standard Deviation of Sample Means for
Total and Historical Value-at-Risk Actual Variable Results from Table R.4

Further analysis of the detailed results presented in Table R.4 using relative percentage changes exhibited in Table R.1 and Table R.2 show corroborative evidence for the general observations stated earlier in this section. Percentage change analysis of the key accounting variables suggest that the Primary and Secondary sample firms experienced more than a 100% change for total assets, total liabilities and net income, and approximately a 50% increase for total equity, after 2005 compared to before. However, the same material significance was not exhibited for the Control group firms. As noted earlier, the market price variable for the Primary and Secondary sample and the Control group did not vary significantly when compared to before and after 2005.

In addition, noted earlier, the Historical Value-at-Risk Actual variable for both the samples and the control group firms exhibited significant material changes from before to after the 2005. The Primary and Secondary sample firms registered a significant material increase, and the Control group firms registered a less materially significant increase.

Further analysis of the Table R.4 and Table R.1 results show that the Primary and Secondary sample firms may have experienced materially significant increases if examining the difference between the change in accounting total variables and the change in the market price variable after the 2005 accounting change; with the earlier stated Historical Value-at-Risk Actual exhibiting significant increases after 2005. This supposition is not evident when examining the Control group firms. They seem to report the same levels if examining the difference between the same accounting and market price change variables after 2005 compared to before; with a less significant material increase in Historical Value-at-Risk Actual after 2005.

The next section, Appendix R.2, further examines the reactions of these total variables by comparing them with the Historical Value-at-Risk Actual using correlation analysis. This examination is then followed in Appendix S with a direct examination of the earlier alluded difference relationship between accounting and market price change variables. This difference relationship, examined in Appendix S, provides the basis of the relative delta measurement approach detailed in Section 3.2.4 and Appendix E.

R.1.2 Descriptive Statistics and Analysis of the Net Income Accounting Total Variable

The convention applied in this study in its examination of the net income variable is to apply yearly figures from 1992 to 2001 and *second half-year* figures from 2002 to 2008. Table R.3 is presented with the aim to analyse the net income variable levels for the full year, first half-year and second half-year. This table presents descriptive statistics and percentage changes on a yearly basis for the Primary and Secondary sample net income accounting total variable for the time range 1992 to 2008.

Analysis

Table R.3 presents the Primary and Secondary sample yearly descriptive statistics for the sample average and the standard deviation of sample means for the consolidated net income total variable.

From 1992 to 2001 inclusive, this study applies the full year net income figures shown in the column *Full Year*. Due to ease of data collection, from 2002 to 2008 this study applies the second half-year net income figures shown in column *Second Half* to represent the net income variable. It is considered in this study that the application of either the full or the applied half-year figures after 2002, for its net income variable, does not significantly affect the findings reported in this research.

Table R.3Descriptive Statistics for the Primary and Secondary Sample Net
Income Total Variable

Table Description

The Table R.3 columns represent the following:

The Statistic column, the same as Table R.4, represent the panels: Mean calculated using equation (0.17); and, Standard Deviation of sample means calculated using equation (0.18), for the sample firms observed (Obs.) for year (Year).

The Year column represents the *panel data* year. This Year column, same as Table R.4, represents the fiscal accounting year applied to group variables on a yearly basis. In addition, the Year column represents the reporting year and specifies the variable record date applied for data collection. For banks in the UK banking sector the variable record date is generally the 31st December each year. For firms with a different financial end of year reporting date, the variable record date adheres to the convention specified in Section 3.6, criterion 7.

For the *Total Net Income Variable (TI)* column: *Full Year* represents consolidated net income totals on a fiscal year basis; *First Half* represents the grouping for the consolidated net income totals reported for the first half-year: from 1st January to 30th June; *Second Half* represents the grouping for the consolidated net income totals reported for the second half-year: from 1st July to 31st December; *Percentage Change* represents the relative percentage change of the figures from the first half-year to the second half-year.

			Total Net In	ncome Variable (T	T)	
Statistic	Year	Full Year	First Half	Second Half	Percentage Change (%)	Obs.
	1992	88.77	-	_	_	15
	1993	207.34	-	-	-	15
	1994	310.40	-	-	-	15
	1995	458.84	-	-	-	16
	1996	595.74	-	-	-	16
	1997	517.24	-	-	-	16
	1998	475.11	-	-	-	16
Maan	1999	579.80	-	-	-	16
Mean	2000	771.62	-	-	-	16
	2001	702.40	-	-	-	16
	2002	665.99	388.24	277.83	-28.44	16
	2003	892.17	453.08	438.37	-3.25	16
	2004	1,080.21	562.43	518.28	-7.85	16
	2005	1,349.75	662.92	686.14	3.50	16
	2006	1,474.22	759.03	715.24	-5.77	16
	2007	1,668.96	948.72	784.85	-17.27	16
	2008	-852.37	405.82	-1,260.71	-410.66	16
	1000	225.50				
	1992	337.58	-	-	-	
	1993	457.12	-	-	-	
	1994	573.53	-	-	-	
	1995	997.02	-	-	-	
	1996	1,265.84	-	-	-	
	1997	9/5.66	-	-	-	
0, 1, 1	1998	818.52	-	-	-	
Standard	1999	1,035.38	-	-	-	
Deviation	2000	1,326.50	-	-	-	
	2001	1,209.03	-	-	-	
	2002	1,166.52	6/4.38	499.26	-25.97	
	2003	1,560.87	/86.68	822.27	4.52	
	2004	1,914.42	1,029.61	892.05	-13.36	
	2005	2,516.04	1,223.60	1,294.94	5.83	
	2006	2,597.55	1,406.16	1,209.70	-13.97	
	2007	2,967.91	1,673.31	1,348.44	-19.41	
	2008	6,339.85	1,064.35	5,936.21	457.73	

Table R.3 (Continued)

Table notes: The Total Net Income variable means are presented using sample averages. The sample consists of a maximum of 16 observations per year from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.

Table R.4Descriptive Statistics for the Totals and Historical Value-at-Risk
Actual Variables

Table Description

The Table R.4 columns represent the following:

Statistic column represents the panels: Mean calculated using equation (O.17); and, Standard Deviation of sample means calculated using equation (O.18), for the sample firms observed (Obs.) for year (Year).

The Year column represents the *panel data* year. This Year column also represents the fiscal accounting year applied to group variables on a yearly basis. In addition, the Year column represents the source data reporting year and specifies the variable record date applied for data collection. For banks in the UK banking sector, the variable record date is generally the 31st December each year. For firms with a different financial end of year reporting date, the variable record date adheres to the convention specified in Section 3.6, criterion 7.

For the market based variable columns:

TBM represents the total book-to-market ratio; *TMB* represents the total market-to-book ratio; *P* represents the market price, scale^a: 0.01GBP^b equivalent to 1GBX^c; *TMV* represents the total market value, scale: 1,000,000 GBP; *VOL*^d represents the volume, scale: 1,000,000 *share units*; *TS*^e represents the total number of shares outstanding, scale: 1,000,000 GBP.

For the key financial statement accounting total based variable columns:

TE represents total shareholders' equity, scale: 1,000,000 GBP;

TA represents total assets, scale: 1,000,000 GBP;

TL represents total liabilities scale: 1,000,000 GBP;

TI represents net income, scale: 1,000,000 GBP (Appendix R.1.2 presents analysis and the data collection specification of the net income variable applied in this study).

The V column represents the Historical Value-at-Risk Actual.

The Obs. column represents the number of sample firms observed for the year.

Table description notes:

^a The term scale refers to the scaling applied to the respective variable prior to the variable's presentation in the specified table.

^b GBP is the ISO code for the Great British Pound sterling.

^c GBX is the ISO code for the Great British Pound pence sterling.

^e Data for the total number of outstanding variable TS contains for the samples and the control group the following missing data percentages: 5.27% for the Primary Sample, 5.9% for the Secondary Sample and 3% for the Control group.

^d Data for the volume variable VOL contains for the samples and the control group the following missing data percentages: 1.9% for the Primary Sample, 7.6% for the Secondary Sample and 4.4% for the Control group.

							Table H	R.4 (Continued)					
							P	PANEL A					
		Descripti	ve Stati	istics for	the Prima	ry and S	Secondary	Sample Totals a	nd Historical	Value-at-Risk	Actual Vari	ables	
Statistic	Year	TBM	TMB	Р	TMV	VOL	TS	TE	TA	TL	TI	V	Obs.
Mean	1992 1993 1994 1995 1996 1997 1998 1999 2000 2001 2002 2003 2004 2005 2006 2007 2008	-0.78 -0.31 0.17 0.61 0.63 0.62 0.62 0.57 0.57 0.62 0.82 0.72 0.67 0.65 0.64 0.74 1.20	1.29 1.88 1.74 2.36 2.32 2.76 2.86 2.98 3.20 2.28 1.58 1.80 1.85 1.94 2.06 1.69 1.20	130.68 220.25 226.30 290.00 331.26 434.52 474.79 581.21 686.13 557.60 405.72 524.64 569.95 685.75 794.36 769.14 505.13	$\begin{array}{c} 1.072.94\\ 1.944.95\\ 1.829.57\\ 3.347.13\\ 4.303.56\\ 6.459.71\\ 6.518.68\\ 10.976.54\\ 14.455.96\\ 13.864.12\\ 10.869.04\\ 13.908.85\\ 14.944.69\\ 16.083.28\\ 17.820.91\\ 15.210.53\\ 9.368.03\\ \end{array}$	2.13 2.06 1.70 1.19 1.00 3.53 1.48 1.58 1.84 1.23 1.77 1.70 1.49 3.07 7.11 2.39 4.19	1,753.24 1,772.18 2.041.57 2,325.91 2,444.05 2.058.24 2,067.17 2,100.31 2,604.30 2,669.90 2,669.47 2,787.90 2,849.75 2,892.34 2.920.13 2,981.56 4,919.64	$\begin{array}{c} 1.389.43\\ 1.576.58\\ 1.772.17\\ 2.581.65\\ 3.061.60\\ 2.669.03\\ 2.814.29\\ 3.363.36\\ 5.526.56\\ 6.016.42\\ 5.819.95\\ 6.688.74\\ 7.266.53\\ \textbf{8.298.72}\\ 9.089.28\\ 10.869.92\\ 11.850.49\end{array}$	28,380,13 33,472,93 33,956,28 48,659,97 53,767,61 51,191,13 51,851,11 59,488,25 87,004,45 95,731,27 101,645,62 113,229,02 133,943,09 189,681,79 207,964,15 303,685,55 435,711,99	$\begin{array}{c} 26.829.30\\ 31,717.56\\ 32.018.81\\ 45,814.31\\ 50,443.63\\ 48.346.95\\ 48.884.30\\ 55.944.79\\ 80,978.72\\ 89.167.92\\ 95.387.98\\ 105.903.51\\ 125.879.94\\ 180,557.01\\ 197.821.21\\ 289.564.60\\ 421,490.90\\ \end{array}$	88.77 207.34 310.40 458.84 595.74 517.24 475.11 579.80 771.62 702.40 277.83 438.37 518.28 686.14 715.24 784.85 -1,260.71	- 0.04 -0.19 0.01 0.01 -0.11 -0.13 -0.30 -0.41 -0.52 -0.09 0.01 0.06 -0.11 -0.73	15 15 16 16 16 16 16 16 16 16 16 16 16 16 16
Standard Deviation	1992 1993 1994 1995 1996 1997 1998 1999 2000 2001 2002 2003 2004 2005 2006 2007	6.71 3.88 2.05 0.40 0.44 0.44 0.44 0.43 0.41 0.38 0.40 0.37 0.34 0.33 0.35 0.34	0.98 1.28 1.05 1.33 1.37 2.09 2.48 2.00 2.73 1.38 0.89 0.89 0.89 0.84 0.96 1.11 0.81	64.55 101.84 103.84 159.42 194.81 255.38 284.48 276.09 357.62 220.78 180.86 231.47 251.65 317.19 370.84 467.16	1,732.91 3,091.50 2,786.37 4,983.43 6,728.48 11,612.36 12,029.15 20,271.54 25,289.78 23,302.64 19,224.03 26,291.29 27,607.71 29,202.14 30,797.16 26,719.79	4.20 3.21 3.75 1.68 1.59 6.45 2.25 2.85 2.80 2.18 2.59 2.54 4.35 12.02 3.03	2.903.18 2.930.45 3.549.35 4.278.31 4,624.74 3.556.28 3.571.28 3.646.71 4.258.53 4.362.88 4.374.76 4.623.95 4.741.27 4.773.74 4.797.41 4.918.01	2.233.45 2.497.14 2.905.92 5.197.04 6.437.43 4.287.30 4.381.25 5.392.37 9.128.87 10.055.87 10.156.46 12.084.07 13.256.75 15.317.66 16.264.49 19.805.21	54,216,66 63,685,58 62,450,93 96,868,80 109,443,89 92,271,09 91,813,28 107,974,78 149,765,64 164,472,18 174,579,19 199,758,35 238,080,95 342,063,02 373,862,35 587,245,72	51.678.82 60,854.66 59.243.38 91,116.61 102,408.91 87,790.00 87,222.88 102,356.97 139,903.42 153,763.22 164,107.37 186,947.16 223,866.63 326,884.90 357,143.36 561,050.50	337.58 457.12 573.53 997.02 1,265.84 975.66 818.52 1,035.38 1,326.50 1,209.03 499.26 822.27 892.05 1,294.94 1,209.70 1,348.44	0.21 0.20 0.18 0.12 0.26 0.13 0.31 0.19 0.25 0.47 0.23 0.08 0.13 0.19	

Panel notes: The total and Value-at-Risk Actual variable means are sample averages. The sample consists of a maximum of 16 observations per year from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.

							Table R.4	(Continued)							
	PANEL B Descriptive Statistics for the Control Group Totals and Historical Value-at-Risk Actual Variables														
Statistic	Year	TBM	TMB	Р	TMV	VOL	TS	TE	TA	TL	TI	V	Obs.		
Mean	1992 1993 1994 1995 1996 1997 1998 1999 2000 2001 2002 2003 2004 2005 2006 2007	1.16 1.08 1.08 1.12 1.19 1.16 1.23 1.22 1.18 1.19 1.18 1.18 1.17 1.12 1.12 1.12	0.87 0.94 0.93 0.90 0.85 0.86 0.81 0.82 0.85 0.84 0.85 0.85 0.85 0.86 0.86 0.85 0.86	135.91 183.92 164.58 184.80 196.20 227.18 218.66 292.05 288.77 246.16 172.00 223.50 250.40 347.01 390.19 384.77	373.48 491.85 458.32 527.19 554.67 673.25 664.29 797.54 763.89 624.85 429.54 531.60 567.69 745.11 768.95 731.30	0.44 0.38 0.54 0.43 0.56 0.35 0.36 1.07 0.33 0.30 0.20 0.23 0.34 0.39 0.69 0.34	291.36 292.86 293.66 295.63 299.04 299.05 298.74 284.00 268.41 258.00 254.30 253.45 248.03 233.27 214.73 197.96	428.01 534.29 492.71 589.75 646.16 776.30 804.36 952.41 886.69 719.40 494.14 623.86 677.71 849.71 872.91 831.33 612.02	548.47 648.99 626.25 711.19 752.96 877.20 914.10 1.079.52 1.018.32 857.07 628.58 754.71 802.53 969.66 995.08 995.08 963.05	112.09 102.53 121.60 107.64 106.80 100.90 109.74 127.11 131.63 137.68 134.45 130.85 124.82 119.95 122.16 131.72 121.25	9.90 11.67 12.08 13.38 14.88 15.14 15.15 13.26 15.27 11.57 6.19 6.65 7.03 7.88 8.76 8.95 8.25	-0.10 -0.19 -0.03 -0.05 -0.24 -0.17 -0.02 -0.30 -0.38 -0.46 -0.01 0.06 0.05 -0.05 -0.05	12 12 12 12 12 12 12 12 12 12 12 12 12 1		
Standard Deviation	2008 1992 1993 1994 1995 1996 1997 1998 1999 2000 2001 2002 2003 2004 2005 2006 2007	1.96 0.09 0.09 0.06 0.07 0.13 0.04 0.07 0.06 0.05 0.08 0.08 0.07 0.07 0.07 0.07 0.07 0.08 0.05 0.05 0.05 0.24	0.59 0.07 0.08 0.05 0.06 0.08 0.03 0.03 0.05 0.04 0.04 0.05 0.05 0.05 0.07 0.04 0.14	260.42 91.95 116.35 96.79 113.68 133.49 160.84 193.03 193.96 172.09 115.17 166.00 187.68 251.90 324.01 275.79	486.30 332.14 436.25 407.71 492.97 483.33 605.08 629.78 760.47 740.44 605.18 422.46 500.85 507.06 647.89 674.87 657.51	0.27 0.52 0.27 0.94 0.67 0.61 0.34 0.36 2.63 0.31 0.36 0.21 0.23 0.54 0.45 1.36 0.43	195.61 274.80 275.51 274.57 274.59 271.32 271.31 271.20 261.97 252.28 243.05 243.20 243.19 233.30 210.41 191.77 178.00	547.83 372.15 461.11 421.53 512.14 546.50 692.26 754.41 893.74 826.57 675.10 468.83 580.06 610.46 742.65 764.84 744.49	669.17 448.39 535.41 513.45 579.50 618.30 771.53 832.62 975.68 905.65 757.67 551.06 661.57 693.77 805.92 850.25 851.58	121.35 93.22 90.57 100.26 87.33 92.10 96.12 83.47 95.26 90.78 96.84 97.53 99.13 93.58 85.90 95.59 130.94	8.76 8.37 9.34 9.44 10.72 11.96 12.67 11.56 10.21 12.83 11.40 5.24 5.63 5.82 6.60 6.83 7.55	-0.46 0.08 0.11 0.14 0.19 0.22 0.25 0.23 0.13 0.08 0.09 0.07 0.07 0.07 0.16 0.16			

Panel notes: The total and Value-at-Risk Actual variable means are yearly averages from the Control group firms. The Control group consists of a maximum of 12 observations per year from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.

	Table R.4 (Continued)														
	PANEL C														
		De	escripti	ve Statis	tics for the	Prima	ry Sample	Totals and Hist	torical Value-	at-Risk Actua	l Variables				
Statistic	Year	TBM	TMB	Р	TMV	VOL	TS	TE	TA	TL	TI	V	Obs.		
Mean	1992 1993 1994 1995 1996 1997 1998 1999 2000 2001 2002 2003 2004 2005 2006 2007 2008	0.91 0.51 0.55 0.42 0.38 0.32 0.34 0.28 0.36 0.41 0.49 0.44 0.43 0.47 0.47 0.67 1.42	1.20 2.11 1.86 2.62 2.88 3.59 3.32 3.73 2.91 2.59 2.13 2.32 2.37 2.19 2.24 1.73 0.93	100.96 198.50 176.64 263.00 339.86 436.13 441.73 588.11 635.22 599.30 467.75 570.21 605.17 679.09 762.06 712.42 349.71	3,200.92 5,970.36 5,506.48 9,283.47 12,179.65 18,629.99 18,651.52 32,763.13 43,543.03 42,374.50 33,383.12 42,720.20 45,898.95 49,106.25 54,210.60 45,989.09 27,971.04	$\begin{array}{c} 5.64\\ 5.64\\ 4.98\\ 3.05\\ 2.58\\ 5.36\\ 3.91\\ 4.31\\ 4.89\\ 1.95\\ 3.58\\ 4.65\\ 4.00\\ 6.64\\ 21.50\\ 5.22\\ 11.22\\ \end{array}$	5,207.10 5,261.65 6,122.47 7,028.31 7,399.77 6,146.26 6,172.58 6,294.64 7,918.08 8,099.05 8,123.69 8,503.12 8,687.57 8,786.19 8,845.93 9,049.43 15,179.60	$\begin{array}{c} 3,760.80\\ 4,180.60\\ 4,815.60\\ 7,221.40\\ 8,629.60\\ 7,166.40\\ 7,515.40\\ 9,085.80\\ 16,079.40\\ 17,828.40\\ 17,388.00\\ 20,001.80\\ 21,754.00\\ 24,763.60\\ 32,665.60\\ 36,049.80\end{array}$	$\begin{array}{c} 88,472.00\\ 103,989.80\\ 104,955.60\\ 151,778.00\\ 168,308.60\\ 159,759.80\\ 161,044.00\\ 185,353.60\\ 274,638.20\\ 302,590.20\\ 321,684.60\\ 358,736.00\\ 424,548.80\\ 601,687.40\\ 659,130.20\\ 964,391.00\\ 1,387,056.00\\ \end{array}$	$\begin{array}{c} 84,197.40\\ 99,240.20\\ 99,616.20\\ 143,716.40\\ 158,844.00\\ 152,037.60\\ 153,045.00\\ 175,694.40\\ 256,965.20\\ 283,014.60\\ 302,899.20\\ 336,703.80\\ 400,253.40\\ 574,286.80\\ 628,730.60\\ 921,364.60\\ 1,343,462.20\\ \end{array}$	$\begin{array}{c} 279.60\\ 618.40\\ 936.40\\ 1,402.60\\ 1,825.20\\ 1,569.00\\ 1,426.80\\ 1.746.40\\ 2,354.60\\ 2,196.20\\ 855.40\\ 1.373.60\\ 1.616.80\\ 2,162.40\\ 2,214.80\\ 2,402.40\\ -4,068.80\\ \end{array}$	-0.02 -0.22 0.12 0.06 -0.33 -0.12 -0.31 -0.16 -0.32 -0.38 -0.05 -0.01 0.02 -0.23 -1.08	4 4 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5		
Standard Deviation	1992 1993 1994 1995 1996 1997 1998 1999 2000 2001 2002 2003 2004 2005 2006 2007 2008	0.34 0.16 0.08 0.13 0.11 0.10 0.05 0.07 0.10 0.07 0.06 0.10 0.11 0.32 0.85	0.36 0.58 0.33 0.93 1.07 1.83 1.63 0.77 0.68 0.77 0.57 0.35 0.34 0.43 0.51 0.69 0.49	44.59 95.59 61.03 138.31 187.50 164.88 197.14 211.04 199.81 107.47 131.64 205.97 202.04 302.76 337.76 526.49 361.15	2.347.87 3.801.74 3.311.76 5.302.52 7.470.64 15.309.10 16.529.75 26.005.51 29.292.98 23.612.06 21.534.65 32.896.39 33.395.55 34.845.38 33.878.14 30.874.54 29.353.78	5.72 3.43 5.48 1.54 1.82 5.67 2.22 3.43 2.73 0.85 2.68 2.68 2.68 2.74 2.87 3.95 12.77 2.53 7.18	3,140.77 3,163.51 4,111.70 5,327.26 5,958.10 4,121.69 4,138.95 4,222.41 4,075.99 4,189.70 4,198.23 4,553.48 4,719.37 4,712.15 4,725.78 4,860.90 14 338.59	2.870.33 3.264.07 3.805.39 7.851.63 9.921.71 5.599.08 5.549.41 6.938.86 10.435.00 11.169.69 11.964.16 14.993.54 16.635.96 19.645.72 20.127.26 24.609.92 24.634.31	66.706.34 78,458.76 73,715.01 125,808.33 145,060.38 102,344.30 99,483.38 121,988.17 141,751.68 153,736.61 162,038.99 200,115.08 242,910.81 360,742.77 392,479.81 706,746.98 958,124,65	63,435.59 74,739.06 69,492.00 116,966.83 133,958.01 96,609.44 93,649.88 114,826.57 130,697.96 142,603.82 150,655.63 184,942.78 225,955.60 344,720.79 375,092.54 674,253.86 929.841 65	595.15 686.53 719.21 1.450.54 1.803.83 1.245.77 927.02 1.237.99 1.425.09 1.191.78 571.86 970.94 886.11 1.523.86 1.181.27 1.433.68 10.853.87	0.21 0.14 0.06 0.17 0.33 0.13 0.19 0.18 0.10 0.24 0.07 0.03 0.04 0.17 0.65			

Panel notes: The total and Value-at-Risk Actual variable means are sample averages. The sample consists of a maximum of 5 observations per year from the 5 banks in the LSE that adopted the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.

							Table R.4	(Continucu)					
		De	escrinți	ve Statist	ics for the	Seconda	PAI rv Sample T	NEL D Totals and Hist	orical Value-a	t-Risk Actual	Variables		
Statistic	Year	TBM	TMB	P	TMV	VOL	TS	TE	TA	TL	TI	V	Obs.
Mean	1992 1993 1994 1995 1996 1997 1998 1999 2000 2001 2002 2003 2004	-1.40 -0.61 0.03 0.70 0.74 0.75 0.74 0.70 0.67 0.72 0.97 0.84 0.79	1.32 1.80 1.70 2.25 2.07 2.38 2.65 2.64 3.32 2.15 1.33 1.56 1.61	141.49 228.16 244.36 302.28 327.35 433.78 489.82 578.08 709.27 538.64 377.53 503.93 553.04	299.13 481.16 492.52 648.79 723.53 927.77 1.003.75 1.073.55 1.234.57 904.86 635.37 812.78 874.56	0.12 0.27 0.06 0.15 0.12 2.52 0.13 0.06 0.15 0.90 0.94 0.35	183.31 186.06 186.61 188.45 191.44 200.05 201.07 193.80 188.95 189.01 190.28 190.07 196.19	311.54 392.94 388.80 472.67 530.69 624.77 677.42 762.24 762.24 762.24 7647.34 561.74 637.35 681.31	1,065.65 1,419.81 1,683.86 1,788.14 1,703.52 1,841.74 2,217.97 2,276.73 1,716.38 1,704.49 1,627.90 1,634.94 1,840.50	752.89 1.025.45 1.292.73 1.313.36 1.170.74 1.214.84 1.538.52 1.513.15 984.86 1.055.79 1.064.70 994.29 1.164.74	2.03 20.49 25.86 29.86 36.90 39.17 42.52 49.52 52.08 23.40 15.30 13.27 18.96	0.06 -0.17 -0.02 -0.01 -0.01 -0.01 -0.05 -0.36 -0.46 -0.58 0.11	11 11 11 11 11 11 11 11 11 11 11
	2004 2005 2006 2007 2008	0.73 0.72 0.77 1.10	1.01 1.82 1.97 1.67 1.32	688.78 809.05 794.92 575.77	1,072.84 1,280.14 1,220.27 912.11	1.44 0.56 1.10 0.99	213.31 226.58 223.43 256.02	814.68 934.04 962.79 850.80	2.406.51 2.888.67 3.364.89 3.282.90	1,588.92 1,953.30 2,382.79 2,413.04	15.11 33.63 49.60 15.69	0.02 0.08 -0.06 -0.57	11 11 11 11 11
Standard Deviation	1992 1993 1994 1995 1996 1997 1998 1999 2000 2001 2002 2003 2004 2005 2006 2007	7.84 4.55 2.41 0.45 0.44 0.47 0.48 0.46 0.46 0.46 0.46 0.42 0.40 0.39 0.35 0.36	1.14 1.47 1.23 1.50 1.46 2.17 2.83 2.31 3.31 1.60 0.91 0.97 0.90 1.13 1.32 0.90	68.96 107.34 112.40 173.03 206.90 294.87 324.11 310.63 417.12 259.29 198.34 248.77 278.86 337.95 399.87 462.76	283.68 453.65 452.51 613.17 685.50 895.82 835.99 827.44 902.08 591.83 453.51 523.38 570.20 603.69 737.78 815.92	0.25 0.43 0.05 0.12 0.13 6.96 0.23 0.05 0.16 2.54 2.19 0.61 0.54 3.58 0.94 2.31	145.45 150.52 150.33 148.61 146.78 145.73 144.99 147.62 147.33 150.48 149.53 148.83 157.85 193.66 226.35 226.68	$\begin{array}{c} 311.81\\ 400.43\\ 371.25\\ 445.37\\ 487.17\\ 566.93\\ 633.68\\ 736.13\\ 676.62\\ 552.20\\ 387.90\\ 463.18\\ 514.96\\ 637.55\\ 806.26\\ 786.14\end{array}$	$\begin{array}{c} 1.751.77\\ 2.238.80\\ 3.241.29\\ 3.304.88\\ 2.896.32\\ 2.951.68\\ 3.795.44\\ 3.724.01\\ 1.591.12\\ 1.490.41\\ 1.309.63\\ 1.129.94\\ 1.464.17\\ 1.725.79\\ 2.544.65\\ 3.491.24\\ \end{array}$	$\begin{array}{c} 1.701.51\\ 2.169.02\\ 3.131.51\\ 3.175.44\\ 2.749.14\\ 2.796.22\\ 3.621.51\\ 3.507.07\\ 1.459.14\\ 1.400.95\\ 1.254.26\\ 1.184.03\\ 1.535.61\\ 1.878.40\\ 2.681.10\\ 3.582.51\end{array}$	52.66 44.93 38.60 40.78 51.57 50.73 50.19 71.95 68.08 40.22 21.72 31.52 43.85 38.09 38.64 48.80	0.22 0.22 0.19 0.10 0.14 0.14 0.33 0.16 0.28 0.54 0.28 0.54 0.28 0.15 0.15 0.18	

Table R.4 (Continued)

Panel notes: The change and Value-at-Risk Actual variable means are sample averages. The sample consists of a maximum of 11 observations per year from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.

R.2 Correlation Analysis for Book-to-Market Ratio, Key Accounting and Market Total Variables to Historical Value-at-Risk Actual Variable

The correlation analysis between the total variables and the Historical Value-at-Risk variable in Table R.4 is presented on a yearly basis for the time range 1994 to 2008 in Table R.5. In Table R.5, Panel A presents results for the Primary and Secondary sample and the Control group; Panel B presents results for the Primary sample and the Secondary sample.

Table R.5Correlation Analysis for the Totals and Historical Value-at-Risk
Actual Variables

Table Description

The Table R.5 columns represent the following:

Correlation Variables column represents the variable pairs correlated.

Measure column represents the data panels:

r panel represents the correlation coefficient calculated using the approach specified in equation (O.20); p(r) panel represents the probability significance level for correlation coefficient, *r*, calculated by applying equation (M.15).

For the samples and the control group columns:

1994 - 2008 column represents the correlation results for the total time period range in years. A single year represents the fiscal accounting year, with the 31st December selected as the variable record date.

For the samples and the control group columns:

1994 - 2004 and 1994 - 2007 columns represent the correlation results for the sub-time period year ranges. A single year represents the fiscal accounting year, with the 31st December selected as the variable record date.

Table R.5 (Continued)

PANEL A

Correlation for the Primary and Secondary Sample and the Control Group Total Variables to the Historical Value-at-Risk Actual Variable

Correlation		Primary	and Secondary	Sample		Control Group	
Variables	Measure	1994-2008	1994-2004	1994-2007	1994-2008	1994-2004	1994-2007
TBM vs. V	r	-0.739**	-0.614*	-0.483	-0.525*	-0.301	-0.301
	p(r)	1.7E-03	4.4E-02	8.0E-02	4.5E-02	3.0E-01	3.0E-01
TMB vs. V	r	0.464	0.393	0.223	0.562*	0.344	0.344
	p(r)	8.2E-02	2.3E-01	4.4E-01	2.9E-02	2.3E-01	2.3E-01
P vs. V	r	0.077	-0.284	0.079	0.463	0.537*	0.537*
	p(r)	7.8E-01	4.0E-01	7.9E-01	8.2E-02	4.8E-02	4.8E-02
TMV vs. V	r	-0.083	-0.560	-0.179	0.581*	0.520	0.520
	p(r)	0.768	0.073	0.540	0.023	0.057	0.057
VOL vs. V	r	0.074	0.179	0.416	0.367	0.317	0.317
	p(r)	7.9E-01	6.0E-01	1.4E-01	1.8E-01	2.7E-01	2.7E-01
TS vs. V	r	-0.677**	-0.638*	-0.231	0.080	-0.200	-0.200
	p(r)	5.6E-03	3.5E-02	4.3E-01	7.8E-01	4.9E-01	4.9E-01
TE vs. V	r	-0.424	-0.664*	-0.111	0.538*	0.464	0.464
	p(r)	1.1E-01	2.6E-02	7.1E-01	3.9E-02	9.4E-02	9.4E-02
TA vs. V	r	-0.482	-0.625*	0.059	0.522*	0.442	0.442
	p(r)	6.9E-02	4.0E-02	8.4E-01	4.6E-02	1.1E-01	1.1E-01
TL vs. V	r	-0.483	-0.623*	0.065	-0.288	-0.336	-0.336
	p(r)	6.8E-02	4.1E-02	8.2E-01	3.0E-01	2.4E-01	2.4E-01
TI vs. V	r	0.717**	0.169	0.315	0.259	0.200	0.200
	p(r)	2.6E-03	6.2E-01	2.7E-01	3.5E-01	4.9E-01	4.9E-01
Years	(T)	15	11	14	15	14	14

Panel notes: **, * Correlation coefficients significant at two-tailed levels.

Table R.5 (Continued)

PANEL B

Correlation for the Primary Sample and Secondary Sample Total Variables to the Historical Value-at-Risk Actual Variable

Correlation		F	rimary Sample		Se	condary Sample	•
Variables	Measure	1994-2008	1994-2004	1994-2007	1994-2008	1994-2004	1994-2007
TBM vs. V	г	-0.801**	-0.094	-0.075	-0.599*	-0.558	-0.478
	р(г)	3.4E-04	7.8E-01	8.0E-01	1.8E-02	7.4E-02	8.4E-02
TMB vs. V	r	0.534*	0.174	0.084	0.465	0.501	0.354
	p(r)	4.0E-02	6.1E-01	7.7E-01	8.0E-02	1.2E-01	2.1E-01
P vs. V	r	0.157	-0.359	-0.110	0.134	-0.124	0.201
	p(r)	5.8E-01	2.8E-01	7.1E-01	6.3E-01	7.2E-01	4.9E-01
TMV vs. V	r	-0.027	-0.429	-0.165	0.294	0.141	0.336
	p(r)	0.923	0.188	0.574	0.287	0.679	0.241
VOL vs. V	r	-0.110	-0.044	0.290	-0.042	-0.081	0.032
	p(r)	7.0E-01	9.0E-01	3.2E-01	8.8E-01	8.1E-01	9.1E-01
TS vs. V	r	-0.768**	-0.329	-0.119	-0.133	0.325	0.453
	p(r)	8.3E-04	3.2E-01	6.9E-01	6.4E-01	3.3E-01	1.0E-01
TE vs. V	r	-0.496	-0.408	-0.113	0.043	-0.111	0.227
	p(r)	6.0E-02	2.1E-01	7.0E-01	8.8E-01	7.5E-01	4.4E-01
TA vs. V	r	-0.640*	-0.363	-0.023	0.050	0.414	0.439
	p(r)	1.0E-02	2.7E-01	9.4E-01	8.6E-01	2.1E-01	1.2E-01
TL vs. V	r	-0.644**	-0.360	-0.019	0.053	0.550	0.490
	p(r)	9.6E-03	2.8E-01	9.5E-01	8.5E-01	8.0E-02	7.6E-02
TI vs. V	r	0.828**	0.087	0.166	0.573*	0.676*	0.511
	p(r)	1.4E-04	8.0E-01	5.7E-01	2.5E-02	2.2E-02	6.2E-02
Years	(T)	15	11	14	15	14	14

Analysis

The Table R.5 correlation analysis results show that the samples and the control group total variables to the market price return Historical Value-at-Risk Actual variable exhibit some statistically significant relationships. The Primary and Secondary sample shows significant correlations for the book-to-market ratio and total shares outstanding for the time periods 1994 to 2008, and 1994 to 2004. Total equity, total assets and total liabilities exhibit significant correlations for the 1994 to 2004 time period, and net income for the 1994 to 2008 time period.

The Control group shows significant correlations for the book-to-market ratio, and its reciprocal market-to-book ratio, market value, total equity and total assets for the 1993 to 2008 time period. Market price exhibit significant correlations for the 1994 to 2007, and 1994 to 2004 time periods.

Examining Table R.5, the Primary and Secondary sample correlation coefficients show that the significant total variables generally varied in the opposite direction to the Historical Value-at-Risk Actual before and after 2005. The net income variable, however, moved in the same direction as Historical Value-at-Risk Actual for the 1994 to 2008 time period. Examining the Control group results, the significant total variables generally varied in the same direction to the Historical Value-at-Risk Actual before and after 2005. The Control group book-to-market ratio, however, moved in the opposite direction to the Historical Value-at-Risk Actual before and after 2005. The Control group book-to-market ratio, however, moved in the opposite direction to the Historical Value-at-Risk Actual before.

Further examination that focuses on the correlation between the book-to-market ratio and Historical Value-at-Risk for the time period 1994 to 2008 for the Primary and Secondary sample exhibits a correlation coefficient of -0.793 with a significance level at 0.01. The Control group for the same variables and time period exhibit a correlation coefficient of -0.525 with significance at 0.05. This result suggests that the Primary and Secondary sample book-to-market ratio variable reacted to the Historical Value-at-Risk Actual variable with a greater magnitude after 2005 than the Control group's reaction.

Such a relationship may suggest that firms in the Primary and Secondary sample, compared to the firms in the Control group, would have experienced a greater average increase in total equity relative the market value after 2005.

APPENDIX S ANALYSIS AND RESULTS FOR CHANGES IN -BOOK-TO-MARKET RATIO, KEY MARKET TOTALS AND KEY ACCOUNTING TOTALS

The analysis and results for descriptive statistics and correlations applied to selected change variables are presented in this appendix. The selected change variables are the yearly changes for the totals analysed in Appendix R. These change variables include change in market price (market price return), change in book-to-market ratio, change in market value and the change in key accounting totals. These key accounting total variables are total equity, total assets, total liabilities and net income. The Historical Value-at-Risk Actual is analysed with reference to the change variables. The selected change variables are calculated using fiscal year-end total levels.

S.1 Descriptive Statistics for the Change in - Book-to-Market Ratio, Key Accounting and Market Total Variables, and Historical Value-at-Risk Actual Variable

Calculated from the variables presented in Table R.4, the change in book-to-market ratio, the change in key accounting totals, the change in market totals, and also the Historical Value-at-Risk Actual are presented on a yearly basis for the time range 1994 to 2008 in Table S.3. In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.⁶³

Analysis

Table S.3 shows, for selected samples and control group total variables and the Historical Value-at-Risk variable presented in Table R.4, descriptive statistics results for *change in the total variables*. The Historical Value-at-Risk Actual variable results in Table R.4 are also re-tabulated in Table S.3. Table S.3 shows that for the samples and

⁶³ In general, the analysis of this section's results is presented prior to the presentation of the actual results tables.

the control group, both statistical mean and yearly standard deviation of sample means increase after 2005 compared to before; with some of these increases exhibiting material significance.

To determine the changes in the descriptive statistics for the change variables for before and after 2005, the percentage changes for the yearly mean and standard deviation of sample means are presented in Table S.1 and Table S.2 respectively. Table S.1 and Table S.2 present percentage changes for the time periods 1993 to 2004, to provide changes in levels for before 2005; and, 1993 to 2007 and 1993 to 2008 to provide changes for after 2005.

Table S.1 and Table S.2 Percentage Changes for Mean and Standard Deviation ofSample Means for Change in Total and Historical Value-at-Risk Actual VariableResults from Table S.3

Percentage Change Analysis Table Description

In Table S.1 and Table S.2, the Variable column represents the Description and Symbol columns that present the descriptions and *codes* for the selected *change* variables. The Year column represents the beginning (From) and ending (To) year applied to calculate the percentage changes presented in the Percentage Change column. The Percentage Change (%) column in Table S.1 represents the Mean column. This Mean column presents the mean variable percentage changes rounded to the nearest whole number. The Percentage Change column in Table S.2 represents the Standard Deviation column. This Standard Deviation column presents the percentage changes of the standard deviation of sample means rounded to the nearest whole number. In Table S.1 and Table S.2, the PS column presents the percentage changes for the Primary and Secondary sample. The C column presents the percentage changes for the Control group. The P column presents the percentage changes for the Secondary sample. The S column presents the percentage changes for the Secondary sample.

Table description note:

Refer to this table description for Table S.1 and Table S.2 table details.

Variable		Year	Perc	entage Cl	nange (%	b)
Description	Symbol	From To		Mean	ı	
Description	Symbol	FI0III - 10	PS	С	Р	S
Log Book-to-Market Ratio	dln(BM)	1993 - 2004 1993 - 2007 1993 - 2008	-87 -150 -226	-96 -195 -629	-96 -153 -219	-78 -142 -222
Book-to-Market Ratio	dBM	1993 - 2004 1993 - 2007 1993 - 2008	-95 -140 -201	-97 -237 -1,018	-95 -191 -345	-95 -126 -160
Log Market-To-Book Ratio	dln(MB)	1993 - 2004 1993 - 2007 1993 - 2008	-87 -150 -226	-96 -195 -629	-96 -153 -219	-78 -142 -222
Market-To-Book Ratio	dMB	1993 - 2004 1993 - 2007 1993 - 2008	-81 -135 -190	-95 -166 -447	-97 -131 -160	-56 -134 -224
Market Price	dM	1993 - 2004 1993 - 2007 1993 - 2008	-85 -114 -214	-63 -94 -225	-90 -126 -261	-82 -107 -184
Market Value	dMV	1993 - 2004 1993 - 2007 1993 - 2008	-85 -118 -171	-68 -122 -237	-88 -122 -199	-84 -116 -155
Shares Outstanding	dS	1993 - 2004 1993 - 2007 1993 - 2008	-104 -79 -1,396	-700 -4,900 -600	33 67 2,708	-84 -55 -887
Log Total Equity	dln (E)	1993 - 2004 1993 - 2007 1993 - 2008	-75 -51 -99	-60 -124 -266	-46 45 -5	-83 -77 -125
Total Equity	dE	1993 - 2004 1993 - 2007 1993 - 2008	-166 -225 -158	-63 -110 -207	-48 46 14	-128 -136 -99
Log Total Equity Per Share	dln(E/S)	1993 - 2004 1993 - 2007 1993 - 2008	-79 -55 -252	-55 -89 -263	-55 43 -306	-83 -73 -247
Total Equity Per Share	d(E/S)	1993 - 2004 1993 - 2007 1993 - 2008	-348 -548 613	-58 -84 -206	-58 42 -204	-181 -200 146
Total Assets	dA	1993 - 2004 1993 - 2007 1993 - 2008	-58 -39 -71	-58 -128 -283	20 129 213	-73 -73 -131
Total Liabilities	dL	1993 - 2004 1993 - 2007 1993 - 2008	-69 -16 -50	-125 92 -31	23 128 221	-88 -44 -108
Log Net Income	dln(I)	1993 - 2004 1993 - 2007 1993 - 2008	-75 -75 -206	132 -395 -1	-75 -93 -121	-81 -45 -355
Net Income	dI	1993 - 2004 1993 - 2007 1993 - 2008	-83 -60 -159	-16 -79 537	-86 -98 -140	-81 41 -193

Table S.1Percentage Changes for Mean Change in Total Variables and
Historical Value-at-Risk Actual Variable Results from Table S.3

Variable		Year	Perce	ntage C	hange (%))
Description	Crown by a 1	T.	Sta	ndard D	eviation	
Description	Symbol	FIGH - TO	PS	С	Р	S
Log Book-to-Market Ratio	dln(BM)	1993 - 2004 1993 - 2007 1993 - 2008	-10 36 38	-63 75 264	-24 291 406	24 57 12
Book-to-Market Ratio	dBM	1993 - 2004 1993 - 2007 1993 - 2008	-88 -66 -58	-57 147 833	34 855 1,587	-87 -70 -81
Log Market-To-Book Ratio	dln(MB)	1993 - 2004 1993 - 2007 1993 - 2008	-10 36 38	-63 75 264	-24 291 406	24 57 12
Market-To-Book Ratio	dMB	1993 - 2004 1993 - 2007 1993 - 2008	-47 -56 -61	-69 25 74	-57 64 57	-37 -56 -66
Market Price	dM	1993 - 2004 1993 - 2007 1993 - 2008	-63 17 174	-49 51 26	-46 125 521	-62 10 107
Market Value	dMV	1993 - 2004 1993 - 2007 1993 - 2008	-37 30 24	-45 84 104	-52 113 184	-32 28 -38
Shares Outstanding	dS	1993 - 2004 1993 - 2007 1993 - 2008	-36 -75 365	-40 468 -24	700 425 12,025	-34 -77 364
Log Total Equity	dln (E)	1993 - 2004 1993 - 2007 1993 - 2008	4 13 161	-28 169 202	-61 -36 143	26 15 174
Total Equity	dE	1993 - 2004 1993 - 2007 1993 - 2008	-90 -88 -70	-40 99 19	-64 -33 134	-90 -90 -73
Log Total Equity Per Share	dln(E/S)	1993 - 2004 1993 - 2007 1993 - 2008	-22 -48 201	-31 124 212	-77 -53 385	-14 -48 202
Total Equity Per Share	d(E/S)	1993 - 2004 1993 - 2007 1993 - 2008	-88 -90 -76	-41 76 23	-79 -51 193	-88 -91 -81
Total Assets	dA	1993 - 2004 1993 - 2007 1993 - 2008	-29 63 74	-23 97 137	2 377 188	-28 36 0
Total Liabilities	dL	1993 - 2004 1993 - 2007 1993 - 2008	-31 41 2	-55 19 -50	5 380 185	-31 34 -49
Log Net Income	dln(I)	1993 - 2004 1993 - 2007 1993 - 2008	-16 -80 1	-46 32 -34	-42 -84 -10	18 -73 25
Net Income	dI	1993 - 2004 1993 - 2007 1993 - 2008	-63 -35 -39	-44 -50 491	-75 -97 -37	31 186 -43

Table S.2Percentage Changes for Standard Deviation of Sample Means for
Change in Total Variables and Historical Value-at-Risk Actual Variable Results
from Table S.3

Examination of Table S.3, and the relative percentage change results presented in Table S.1 and Table S.2, suggests that for the Primary and Secondary sample, the *change in accounting total variables* did not exhibit materially significant changes after 2005 compared to before. However, the change in the market price variable exhibited a significant material decrease after 2005 compared to before. The Control group results suggest that the accounting and the market price change variables both exhibited materially significant decreases after 2005 compared to before, other than the net income variable that exhibits a materially significant increase in 2008.

The general inference from analysing the Table S.3 results is that the Primary and Secondary sample may have become exposed to materially significant *difference component* levels (that is, the levels of *difference* between the *change in accounting total variables and the market price return variable*) after 2005 compared to before. The Control group results, however, generally suggest a lower *difference component* after 2005 when compared to the Primary and Secondary sample levels.

Table S.3Descriptive Statistics for the Change in Totals and Historical Value-
at-Risk Actual Variables

Table Description

The Table S.3 columns represent the following:

Statistic column represents the panels: Mean calculated using equation (0.21); and, Standard Deviation of sample means calculated using equation (0.22), for the sample firms observed (Obs.) for year (Year).

The Year column represents the *panel data* year. This Year column also represents the fiscal accounting year applied to group variables on a yearly basis. In addition, the Year column represents the reporting year and specifies the variable record date applied for data collection. For banks in the UK banking sector the variable record date is generally the 31st December each year. For firms with a different financial end of year reporting date, the variable record date adheres to the convention specified in Section 3.6, criterion 7.

Representing yearly changes, the market derived variable columns are:

dln(BM) represents the log change in book-to-market ratio calculated by applying the approach specified in equation (3.33);

dBM represents the relative change in book-to-market ratio calculated by applying the approach specified in equation (3.34);

dln(MB) represents the log change in market-to-book ratio calculated by applying the approach specified in equation (3.33);

dMB represents the relative change in market-to-book ratio calculated by applying the approach specified in equation (3.34);

dM represents the log change in market price calculated by applying the approach specified in equation (3.35), also termed the market price return;

dMV represents the log change in market value calculated by applying the approach specified in equation (3.33);

dS represents the log change in total shares outstanding calculated by applying the approach specified in equation (3.33).

Representing yearly change variables derived from the key financial statement accounting total variables, the variable columns:

dln(E) represents the log change in total shareholders' equity calculated by applying equation (3.33);

dE represents the relative change in total shareholders' equity calculated by applying equation (3.34);

dln(E/S) represents the log change in total shareholders' equity per share calculated by applying equation (3.33);

d(E/S) represents the relative change in total shareholders' equity per share calculated by applying equation (3.34);

dA represents the log change in total assets calculated by applying equation (3.33);

dL represents the log change in total liabilities calculated by applying equation (3.33);

dln(I) represents the log change in net income calculated by applying equation (3.33);

dI represents the relative change in net income calculated by applying equation (3.34).

The *V* column represents the Historical Value-at-Risk Actual. The *Obs.* column represents the number of sample firms observed for the year.

Table S.3 (Continued)																		
								Р	ANEL A									
		Descriptive	e Statist	ics for the	Prima	ry and S	Seconda	ry Samp	le Change in	n Totals	and Hist	orical V	alue-at-	Risk Ac	ctual Va	ariables		
Statistic	Year	dln(BM)	dBM	dln(MB)	dMB	dM	dMV	dS	dln(E)	dE	dln(E/S)	d(E/S)	dA	dL	dln(I)	dI	V	Obs.
Mean	1993 1994 1995 1996 1997 1998 1999 2000 2001 2002 2003 2004 2005 2006 2007 2008	$\begin{array}{c} -0.344\\ -0.005\\ -0.117\\ 0.037\\ -0.065\\ -0.011\\ -0.093\\ -0.004\\ 0.199\\ 0.327\\ -0.140\\ -0.046\\ \hline -0.037\\ -0.039\\ 0.172\\ 0.433\\ \end{array}$	-0.605 0.038 -0.176 0.046 -0.019 0.025 -0.066 0.064 0.289 0.497 -0.118 -0.029 -0.027 -0.034 0.242 0.611	$\begin{array}{c} 0.344\\ 0.005\\ 0.117\\ -0.037\\ 0.065\\ 0.011\\ 0.093\\ 0.004\\ -0.199\\ -0.327\\ 0.140\\ 0.046\\ \hline 0.037\\ 0.039\\ -0.172\\ -0.433\\ \end{array}$	0.359 0.055 -1.055 -0.028 0.105 0.051 0.126 0.098 -0.146 -0.240 0.168 0.070 0.048 0.045 -0.124 -0.323	0.559 0.034 0.213 0.091 0.256 0.058 0.259 0.177 -0.160 -0.372 0.233 0.086 0.203 0.154 -0.077 -0.636	0.591 0.042 0.382 0.119 0.261 0.115 0.305 0.216 -0.166 -0.358 0.247 0.087 0.227 0.164 -0.109 -0.419	-0.024 0.007 0.200 0.032 0.006 0.028 -0.025 0.063 -0.006 0.013 0.012 0.001 0.026 0.007 -0.005 0.311	$\begin{array}{c} 0.178\\ 0.036\\ 0.190\\ 0.155\\ 0.159\\ 0.079\\ 0.154\\ 0.219\\ 0.043\\ -0.037\\ 0.097\\ 0.044\\ \hline 0.181\\ 0.115\\ 0.087\\ 0.002\\ \end{array}$	-0.079 0.092 0.409 0.174 0.223 0.095 0.174 0.420 0.084 -0.018 0.110 0.052 0.270 0.127 0.099 0.046	0.203 0.029 0.142 0.123 0.153 0.052 0.179 0.157 0.049 -0.050 0.084 0.043 0.155 0.107 0.092 -0.309	-0.023 0.084 0.241 0.132 0.172 0.059 0.201 0.189 0.076 -0.034 0.096 0.057 0.196 0.117 0.103 -0.164	0.225 0.020 0.206 0.103 0.119 0.095 0.094 0.157 0.063 -0.009 0.055 0.095 0.307 0.114 0.138 0.066	0.227 0.101 0.216 0.067 0.062 0.093 -0.004 0.201 0.130 0.040 0.060 0.070 0.355 0.053 0.191 0.113	0.445 0.260 0.214 0.269 -0.109 0.225 0.197 0.309 -0.376 -0.866 0.258 0.110 0.274 0.190 0.110 -0.472	1.537 0.394 0.508 0.337 -0.034 0.369 0.207 0.172 -0.239 -0.133 -2.762 0.258 4.809 1.134 0.613 -0.903	0.04 -0.19 0.01 -0.11 -0.11 -0.13 -0.30 -0.41 -0.52 -0.09 0.01 0.06 -0.11 -0.73	15 15 16 1
Standard Deviation	1993 1994 1995 1996 1997 1998 1999 2000 2001 2002 2003 2004 2005 2006 2007 2008	0.221 0.313 0.162 0.135 0.291 0.283 0.232 0.402 0.317 0.365 0.177 0.199 0.141 0.300 0.300 0.305	$\begin{array}{c} 1.211\\ 0.297\\ 0.287\\ 0.140\\ 0.359\\ 0.277\\ 0.219\\ 0.365\\ 0.525\\ 0.755\\ 0.154\\ 0.151\\ 0.136\\ 0.092\\ 0.414\\ 0.513\\ \end{array}$	$\begin{array}{c} 0.221\\ 0.313\\ 0.162\\ 0.135\\ 0.291\\ 0.283\\ 0.232\\ 0.402\\ 0.317\\ 0.365\\ 0.177\\ 0.199\\ 0.141\\ 0.101\\ 0.300\\ 0.305\\ \end{array}$	0.532 0.374 4.511 0.132 0.276 0.326 0.592 0.592 0.223 0.220 0.210 0.283 0.151 0.151 0.12 0.236 0.208	0.232 0.230 0.253 0.131 0.177 0.222 0.249 0.346 0.250 0.318 0.183 0.086 0.211 0.096 0.271 0.635	0.218 0.221 0.398 0.126 0.177 0.231 0.436 0.455 0.267 0.316 0.187 0.138 0.309 0.115 0.283 0.270	$\begin{array}{c} 0.181\\ 0.011\\ 0.572\\ 0.070\\ 0.199\\ 0.067\\ 0.137\\ 0.289\\ 0.070\\ 0.026\\ 0.036\\ 0.116\\ 0.136\\ 0.089\\ 0.045\\ 0.841\\ \end{array}$	$\begin{array}{c} 0.112\\ 0.154\\ 0.152\\ 0.103\\ 0.284\\ 0.152\\ 0.121\\ 0.439\\ 0.274\\ 0.200\\ 0.127\\ 0.117\\ 0.327\\ 0.102\\ 0.126\\ 0.292\\ \end{array}$	$\begin{array}{c} 1.153\\ 0.120\\ 0.664\\ 0.133\\ 0.433\\ 0.181\\ 0.137\\ 1.110\\ 0.343\\ 0.200\\ 0.144\\ 0.111\\ 0.543\\ 0.115\\ 0.133\\ 0.344 \end{array}$	$\begin{array}{c} 0.224\\ 0.149\\ 0.080\\ 0.052\\ 0.111\\ 0.112\\ 0.096\\ 0.185\\ 0.218\\ 0.184\\ 0.123\\ 0.175\\ 0.235\\ 0.082\\ 0.116\\ 0.675\\ \end{array}$	1.215 0.117 0.284 0.060 0.142 0.123 0.177 0.232 0.274 0.179 0.138 0.145 0.267 0.091 0.121 0.294	0.163 0.191 0.297 0.156 0.159 0.119 0.087 0.418 0.276 0.173 0.213 0.115 0.373 0.115 0.373 0.266 0.283	$\begin{array}{c} 0.245\\ 0.287\\ 0.497\\ 0.202\\ 0.161\\ 0.186\\ 0.306\\ 0.478\\ 0.230\\ 0.168\\ 0.378\\ 0.170\\ 0.422\\ 0.304\\ 0.346\\ 0.250\\ \end{array}$	0.819 0.396 0.388 0.206 0.459 0.405 0.311 0.386 0.848 0.945 0.360 0.685 0.551 0.794 0.164 0.828	$\begin{array}{c} 2.926\\ 0.934\\ 0.896\\ 0.314\\ 0.304\\ 0.710\\ 0.626\\ 0.814\\ 0.384\\ 0.982\\ 11.647\\ 1.092\\ \textbf{20.664}\\ 3.527\\ 1.903\\ 1.783\\ \end{array}$	0.21 0.20 0.18 0.12 0.26 0.13 0.31 0.19 0.25 0.47 0.23 0.08 0.13 0.19 0.65	

Panel notes: The change and Value-at-Risk Actual variable means are sample averages. The sample consists of a maximum of 16 observations per year from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.

	Table S.3 (Continued)																	
								P	PANEL B									
		De	escriptiv	ve Statistic	es for th	e Conti	ol Gro	up Chan	ge in Totals	and His	torical Va	lue-at-H	Risk Act	ual Var	iables			
Statistic	Year	dln(BM)	dBM	dln(MB)	dMB	dM	dMV	dS	dln(E)	dE	dln(E/S)	d(E/S)	dA	dL	dln(I)	dI	V	Obs.
	1993	-0.075	-0.067	0.075	0.083	0.336	0.345	0.002	0.273	0.320	0.271	0.316	0.215	-0.122	0.073	0.183	-	12
	1994	0.010	0.012	-0.010	-0.008	-0.092	-0.039	0.050	-0.032	-0.020	-0.082	-0.0//	0.008	0.268	0.038	-0.028	-0.10	12
	1995	0.033	0.035	-0.033	-0.032	0.103	0.097	0.007	0.150	0.164	0.144	0.157	0.127	0.313	0.044	0.081	-0.19	12
	1997	-0.021	-0.017	0.021	0.025	0.020	0.005	3 5E-05	0.120	0.138	0.112	0.070	0.007	0.020	0.008	0.055	-0.05	12
	1998	0.061	0.064	-0.061	-0.058	-0.106	-0.121	-0.001	-0.061	-0.040	-0.060	-0.039	-0.040	0.214	0.098	0.248	-0.24	12
	1999	-0.011	-0.009	0.011	0.012	0.355	0.311	-0.044	0.298	0.446	0.342	0.497	0.285	0.243	-0.220	-0.903	-0.17	12
Mean	2000	-0.036	-0.034	0.036	0.038	0.015	-0.035	-0.051	-0.057	-0.032	-0.006	0.018	-0.040	0.092	0.044	0.063	-0.02	12
Weah	2001	0.012	0.014	-0.012	-0.010	-0.183	-0.222	-0.039	-0.222	-0.196	-0.183	-0.164	-0.196	-0.197	-0.299	0.001	-0.30	12
	2002	-0.014	-0.010	0.014	0.019	-0.30/	-0.311	-0.018	-0.328	-0.273	-0.310	-0.259	-0.2/5	-0.066	-0.559	-0.360	-0.38	12
	2003	0.001	0.002	-0.001	-0.001	0.238	0.234	-0.004	0.236	0.277	0.240	0.282	0.188	-0.020	0.059	0.103	-0.46	12
	2004	-0.003	-0.002	0.003	0.004	0.124	0.293	-0.038	0.242	0.283	0.122	0.132	0.091	0.030	0.109	0.104	0.01	12
	2005	0.007	0.008	-0.007	-0.006	0.094	0.026	-0.071	0.032	0.044	0.103	0.118	0.027	0.105	0.043	0.182	0.05	12
	2007	0.071	0.092	-0.071	-0.055	0.020	-0.076	-0.096	-0.065	-0.032	0.031	0.052	-0.060	-0.234	-0.215	0.039	-0.05	12
	2008	0.397	0.615	-0.397	-0.288	-0.421	-0.474	-0.010	-0.453	-0.341	-0.442	-0.336	-0.394	-0.084	0.072	1.165	-0.46	12
	1993	0.106	0.090	0.106	0.126	0.161	0.164	0.025	0.099	0.134	0.093	0.125	0.118	0.606	0.448	0.608	-	
	1994	0.060	0.063	0.060	0.057	0.122	0.124	0.176	0.153	0.175	0.069	0.063	0.159	0.608	0.195	0.355	0.08	
	1995	0.043	0.044	0.043	0.042	0.069	0.115	0.035	0.068	0.078	0.076	0.087	0.065	1.340	0.175	0.170	0.11	
	1996	0.080	0.088	0.080	0.074	0.175	0.253	0.189 1.2E.04	0.232	0.309	0.160	0.141	0.238	0.255	0.194	0.348	0.14	
	1998	0.092	0.085	0.092	0.053	0.138	0.175	0.003	0.229	0.145	0.230	0.145	0.149	0.363	0.140	0.120	0.19	
	1999	0.060	0.059	0.060	0.061	0.359	0.381	0.042	0.349	0.717	0.325	0.694	0.316	0.318	0.423	2.598	0.25	
Standard	2000	0.053	0.049	0.053	0.057	0.255	0.253	0.042	0.238	0.212	0.238	0.212	0.251	0.480	0.268	0.288	0.23	
Deviation	2001	0.068	0.073	0.068	0.064	0.081	0.097	0.049	0.103	0.080	0.086	0.071	0.097	0.840	0.599	0.685	0.13	
	2002	0.101	0.094	0.101	0.110	0.178	0.187	0.019	0.137	0.108	0.142	0.114	0.114	0.126	0.404	0.282	0.08	
	2003	0.037	0.037	0.037	0.037	0.131	0.131	0.007	0.133	0.175	0.134	0.176	0.120	0.168	0.121	0.089	0.09	
	2004	0.039	0.039	0.039	0.039	0.082	0.090	0.015	0.071	0.081	0.064	0.074	0.091	0.273	0.240	0.339	0.07	
	2005	0.037	0.035	0.037	0.039	0.120	0.153	0.067	0.125	0.161	0.093	0.128	0.127	0.340	0.115	0.169	0.07	
	2006	0.040	0.041	0.040	0.039	0.179	0.188	0.104	0.159	0.101	0.146	0.148	0.100	0.300	0.529	0.204	0.10	
	2007	0.180	0.222	0.180	0.138	0.243	0.301	0.142	0.200	0.200	0.208	0.220	0.232	0.724	0.393	3 591	0.18	
	-000	0.500	0.010	5.500	0.217	0.200	0.001	0.017	0.299	0.100	5.270	0.101	0.200	0.500	0.270	0.071	5.10	

Panel notes: The change and Value-at-Risk Actual variable means are yearly averages from the Control group firms. The Control group consists of a maximum of 12 observations per year from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.

								1 4010 0										
								P.	ANEL C									
		De	scriptiv	e Statistic	s for the	e Prima	ry Sam	ple Chan	ge in Totals	and Hi	storical V	'alue-at-	Risk Ac	tual Va	riables			
Statistic	Year	dln(BM)	dBM	dln(MB)	dMB	dM	dMV	dS	dln(E)	dE	dln(E/S)	d(E/S)	dA	dL	dln(I)	dI	V	Obs.
	1993	-0.576	-0.437	0.576	0.781	0.666	0.676	0.012	0.120	0.132	0.108	0.118	0.125	0.126	1.036	4.099		4
	1994	0.104	0.182	-0.104	-0.024	-0.056	-0.041	0.013	0.120	0.129	0.107	0.115	0.062	0.060	0.666	1.173	-0.02	4
	1995	-0.184	-0.152	0.184	0.229	0.324	0.327	0.114	0.288	0.368	0.174	0.191	0.211	0.207	0.292	0.374	-0.22	5
	1996	-0.087	-0.077	0.087	0.098	0.230	0.226	0.019	0.144	0.158	0.125	0.134	0.124	0.122	0.224	0.251	0.12	5
	1997	-0.136	-0.106	0.136	0.173	0.226	0.241	-0.119	-0.034	-0.005	0.085	0.089	0.043	0.047	-0.181	-0.150	0.06	5
	1998	0.065	0.077	-0.065	-0.055	-0.035	-0.029	0.006	0.028	0.029	0.022	0.024	0.014	0.013	0.001	0.034	-0.33	5
	1999	-0.173	-0.137	0.173	0.214	0.334	0.581	0.025	0.196	0.222	0.172	0.192	0.125	0.121	0.118	0.150	-0.12	5
Mean	2000	0.253	0.310	-0.253	-0.212	0.094	0.351	0.272	0.574	1.179	0.302	0.374	0.438	0.427	0.409	0.546	-0.31	5
	2001	0.129	0.148	-0.129	-0.114	-0.032	-0.013	0.019	0.11/	0.125	0.098	0.104	0.095	0.092	-0.091	-0.066	-0.16	5
	2002	0.185	0.218	-0.185	-0.160	-0.266	-0.254	0.008	-0.0/1	-0.061	-0.078	-0.068	0.047	0.055	-0.943	-0.600	-0.32	2
	2003	-0.098	-0.080	0.098	0.118	0.1/5	0.207	0.033	0.101	0.113	0.068	0.075	0.070	0.068	0.390	0.623	-0.38	5
	2004	-0.023	-0.022	0.023	0.025	0.000	0.081	0.016	0.065	0.101	0.049	0.050	0.150	0.155	0.256	0.369	-0.05	5
	2005	0.087	0.098	-0.087	-0.078	0.085	0.110	0.028	0.132	0.129	0.124	0.141	0.002	0.302	0.204	0.550	-0.01	5
	2000	-0.018	-0.017	0.018	0.019	0.120	0.137	0.014	0.174	0.128	0.104	0.112	0.092	0.090	0.074	0.100	0.02	5
	2007	0.508	1 072	-0.508	-0.244	-0.171	-0.149	0.020	0.174	0.195	-0.223	-0.123	0.280	0.287	-0.219	-1 647	-0.23	5
	2008	0.085	1.072	-0.085	-0.472	-1.071	-0.009	0.557	0.114	0.151	-0.225	-0.123	0.391	0.405	-0.219	-1.047	-1.08	
	1993	0.067	0.038	0.067	0.120	0.110	0.113	0.004	0.109	0.125	0.108	0.122	0.060	0.060	1.121	5.089	-	
	1994	0.439	0.422	0.439	0.494	0.230	0.220	0.012	0.067	0.073	0.066	0.072	0.115	0.123	0.539	1.190	0.21	
	1995	0.236	0.180	0.236	0.317	0.256	0.259	0.225	0.253	0.380	0.062	0.073	0.235	0.233	0.256	0.372	0.14	
	1996	0.136	0.130	0.136	0.143	0.099	0.088	0.066	0.080	0.091	0.033	0.037	0.045	0.046	0.032	0.039	0.06	
	1997	0.254	0.231	0.254	0.290	0.250	0.253	0.277	0.290	0.252	0.038	0.042	0.264	0.262	0.225	0.190	0.17	
	1998	0.157	0.174	0.157	0.143	0.132	0.140	0.014	0.056	0.058	0.063	0.064	0.065	0.070	0.295	0.306	0.33	
0, 1, 1	1999	0.241	0.234	0.241	0.256	0.280	0.645	0.029	0.104	0.130	0.097	0.121	0.055	0.057	0.243	0.239	0.13	
Standard	2000	0.200	0.295	0.200	0.138	0.140	0.545	0.464	0.645	1.864	0.191	0.279	0.471	0.463	0.260	0.398	0.19	
Deviation	2001	0.149	0.177	0.149	0.126	0.138	0.149	0.026	0.060	0.067	0.052	0.058	0.035	0.036	0.240	0.223	0.18	
	2002	0.171	0.211	0.171	0.142	0.176	0.186	0.020	0.142	0.125	0.140	0.125	0.077	0.078	0.264	0.100	0.10	
	2003	0.185	0.177	0.185	0.198	0.117	0.147	0.062	0.123	0.137	0.093	0.100	0.097	0.097	0.462	0.876	0.24	
	2004	0.051	0.051	0.051	0.052	0.059	0.054	0.032	0.042	0.045	0.025	0.026	0.061	0.063	0.653	1.253	0.07	
	2005	0.121	0.134	0.121	0.110	0.117	0.158	0.047	0.183	0.234	0.139	0.166	0.195	0.196	0.257	0.335	0.03	
	2006	0.044	0.043	0.044	0.045	0.071	0.074	0.023	0.069	0.079	0.062	0.068	0.017	0.017	0.232	0.244	0.04	
	2007	0.262	0.363	0.262	0.197	0.248	0.241	0.021	0.070	0.084	0.051	0.060	0.286	0.288	0.178	0.173	0.17	
	2008	0.339	0.641	0.339	0.188	0.683	0.321	0.485	0.265	0.293	0.524	0.358	0.173	0.171	1.006	3.225	0.65	

Table S.3 (Continued)

Panel notes: The change and Value-at-Risk Actual variable means are sample averages. The sample consists of a maximum of 5 observations per year from the 5 banks in the LSE that adopted the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.

	Table S.3 (Continued)																	
									PANEL D									
Descriptive Statistics for the Secondary Sample Change in Totals and Historical Value-at-Risk Actual Variables																		
Statistic	Year	dln(BM)	dBM	dln(MB)	dMB	dM	dMV	dS	dln(E)	dE	dln(E/S)	d(E/S)	dA	dL	dln(I)	dI	V	Obs.
	1993	-0.260	-0.666	0.260	0.206	0.520	0.561	-0.038	0.200	-0.156	0.237	-0.074	0.261	0.264	0.230	0.605	-	11
	1994	-0.049	-0.020	0.049	0.087	0.070	0.075	0.005	0.002	0.077	-0.003	0.071	0.003	0.118	0.098	0.083	0.06	11
	1995	-0.090	-0.186	0.090	-1.569	0.169	0.403	0.234	0.150	0.425	0.130	0.261	0.204	0.219	0.183	0.561	-0.17	11
	1996	0.082	0.091	-0.082	-0.0/4	0.040	0.080	0.037	0.159	0.1/9	0.122	0.131	0.096	0.046	0.286	0.369	-0.02	11
	1997	-0.039	0.012	0.039	0.080	0.207	0.208	0.032	0.230	0.307	0.1//	0.202	0.14/	0.008	-0.085	0.008	-0.01	11
	1998	-0.058	-0.033	0.058	0.089	0.092	0.108	-0.030	0.098	0.119	0.003	0.072	0.124	-0.061	0.300	0.491	-0.01	11
	2000	-0.121	-0.033	0.037	0.080	0.223	0.155	-0.048	0.058	0.152	0.182	0.205	0.079	0.001	0.252	0.233	-0.10	11
Mean	2000	0.230	0.352	-0.230	-0.161	-0.217	-0.235	-0.017	0.009	0.065	0.026	0.063	0.048	0.147	-0.506	-0.318	-0.36	11
	2002	0.392	0.624	-0.392	-0.277	-0.420	-0.406	0.016	-0.021	0.002	-0.037	-0.018	-0.034	0.034	-0.831	0.080	-0.46	11
	2003	-0.159	-0.135	0.159	0.190	0.259	0.265	0.003	0.095	0.108	0.092	0.106	0.048	0.057	0.197	-4.300	-0.58	11
	2004	-0.056	-0.033	0.056	0.091	0.095	0.089	-0.006	0.035	0.044	0.041	0.060	0.070	0.032	0.043	0.116	-0.11	11
	2005	-0.093	-0.084	0.093	0.105	0.257	0.280	0.025	0.194	0.310	0.169	0.221	0.287	0.352	0.279	6.843	0.02	11
	2006	-0.048	-0.041	0.048	0.056	0.170	0.177	0.004	0.113	0.127	0.109	0.119	0.124	0.036	0.243	1.604	0.08	11
	2007	0.110	0.172	-0.110	-0.070	-0.034	-0.091	-0.017	0.047	0.056	0.064	0.074	0.071	0.147	0.127	0.851	-0.06	11
	2008	0.318	0.402	-0.318	-0.255	-0.439	-0.306	0.299	-0.049	-0.002	-0.348	-0.182	-0.082	-0.020	-0.587	-0.564	-0.57	11
	1993	0.195	1.428	0.195	0.543	0.256	0.242	0.212	0.110	1.353	0.249	1.433	0.176	0.278	0.610	0.800	-	
	1994	0.264	0.234	0.264	0.341	0.232	0.224	0.010	0.168	0.135	0.164	0.132	0.217	0.336	0.166	0.641	0.22	
	1995	0.129	0.328	0.129	5.525	0.251	0.452	0.672	0.078	0.700	0.086	0.330	0.330	0.582	0.438	1.050	0.22	
	1990	0.107	0.119	0.107	0.097	0.105	0.117	0.074	0.113	0.149	0.059	0.009	0.185	0.234	0.240	0.300	0.19	
	1997	0.311	0.400	0.318	0.281	0.150	0.150	0.134	0.200	0.404	0.119	0.135	0.107	0.120	0.320	0.334	0.10	
	1999	0.230	0.215	0.230	0.262	0.249	0.240	0.162	0.175	0.140	0.120	0.120	0.097	0.200	0.420	0.750	0.14	
Standard	2000	0.423	0.348	0.423	0.670	0.409	0.423	0.084	0.176	0.193	0.146	0.157	0.340	0.470	0.435	0.910	0.33	
Deviation	2001	0.372	0.622	0.372	0.260	0.272	0.285	0.082	0.327	0.417	0.262	0.332	0.336	0.279	0.998	0.423	0.16	
Dernanon	2002	0.417	0.883	0.417	0.244	0.362	0.358	0.029	0.226	0.229	0.206	0.203	0.201	0.199	1.143	1.132	0.28	
	2003	0.179	0.148	0.179	0.221	0.206	0.206	0.012	0.135	0.153	0.138	0.156	0.253	0.459	0.311	13.95	0.54	
	2004	0.241	0.183	0.241	0.343	0.098	0.165	0.140	0.139	0.133	0.214	0.176	0.126	0.192	0.719	1.044	0.28	
	2005	0.113	0.097	0.113	0.133	0.227	0.351	0.164	0.383	0.644	0.272	0.306	0.439	0.501	0.655	25.01	0.10	
	2006	0.119	0.109	0.119	0.132	0.105	0.131	0.107	0.117	0.132	0.092	0.103	0.235	0.371	0.956	4.226	0.15	
	2007	0.307	0.433	0.307	0.240	0.281	0.309	0.049	0.127	0.131	0.129	0.133	0.240	0.373	0.164	2.285	0.18	
	2008	0.218	0.274	0.218	0.185	0.530	0.150	0.983	0.301	0.367	0.753	0.278	0.176	0.142	0.760	0.453	0.61	

Panel notes: The change and Value-at-Risk Actual variable means are sample averages. The sample consists of a maximum of 11 observations per year from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. The highlighted record identifies the samples IFRS accounting standards adoption year.
S.2 Correlation Analysis for the Change in - Book-to-Market Ratio, Key Accounting and Market Total Variables to Historical Value-at-Risk Actual Variable

The correlation analysis between the change in total variables and the Historical Valueat-Risk variable in Table S.3 is presented on a yearly basis for the time range 1994 to 2008 in Table S.4. In Table S.4, Panel A presents results for the Primary and Secondary sample and the Control group; Panel B presents results for the Primary sample and the Secondary sample.

Table S.4Correlation Analysis for the Change in Totals and Historical
Value-at-Risk Actual Variables

Table Description

The Table S.4 columns represent the following:

Correlation Variables column represents the variable pairs correlated.

Measure column represents the data panels:

r panel represents the correlation coefficient calculated using the approach specified in equation (O.24); p(r) panel represents the probability significance level for correlation coefficient, *r*, calculated by applying equation (M.15).

For the samples and the control group columns:

1994 - 2008 column represents the correlation results for the total time period year range. A single year represents the fiscal accounting year, with the 31st December selected as the variable record date.

For the samples and the control group columns:

1994 - 2004 and 1994 - 2007 columns represent the correlation results for the sub-time period year ranges. A single year represents the fiscal accounting year, with the 31st December selected as the variable record date.

Table S.4 (Continued)

PANEL A

Correlation for the Primary and Secondary Sample and the Control Group Change in Total Variables to the Historical Value-at-Risk Actual Variable

Correlation		Primary a	and Secondary	Sample	(Control Group	
Variables	Measure	1994-2008	1994-2004	1994-2007	1994-2008	1994-2004	1994-2007
dln(BM) vs. V	r	-0.597*	-0.284	-0.281	-0.487	-0.052	-0.113
	p(r)	1.9E-02	4.0E-01	3.3E-01	6.6E-02	8.8E-01	7.0E-01
dBM vs. V	r	-0.622*	-0.316	-0.328	-0.486	-0.048	-0.090
	p(r)	1.3E-02	3.4E-01	2.5E-01	6.6E-02	8.9E-01	7.6E-01
dln(MB) vs. V	r	0.597*	0.284	0.281	0.487	0.052	0.113
	p(r)	1.9E-02	4.0E-01	3.3E-01	6.6E-02	8.8E-01	7.0E-01
dMB vs. V	r	0.280	0.138	0.180	0.485	0.054	0.131
	p(r)	3.1E-01	6.9E-01	5.4E-01	6.7E-02	8.8E-01	6.6E-01
dM vs. V	r	0.684**	0.353	0.375	0.514*	0.202	0.336
	p(r)	4.9E-03	2.9E-01	1.9E-01	5.0E-02	5.5E-01	2.4E-01
dMV vs. V	r	0.585*	0.326	0.325	0.502	0.248	0.306
	p(r)	2.2E-02	3.3E-01	2.6E-01	5.6E-02	4.6E-01	2.9E-01
dS vs. V	r	-0.578*	-0.012	-0.057	-0.161	0.224	-0.158
	p(r)	2.4E-02	9.7E-01	8.5E-01	5.7E-01	5.1E-01	5.9E-01
dln(E) vs. V	r	0.512	0.350	0.383	0.483	0.245	0.281
	p(r)	5.1E-02	2.9E-01	1.8E-01	6.8E-02	4.7E-01	3.3E-01
dE vs. V	r	0.337	0.252	0.241	0.406	0.185	0.218
	p(r)	2.2E-01	4.6E-01	4.1E-01	1.3E-01	5.9E-01	4.5E-01
dln(E/S) vs. V	r	0.742**	0.375	0.420	0.513	0.205	0.320
	p(r)	1.6E-03	2.6E-01	1.3E-01	5.0E-02	5.5E-01	2.6E-01
d(E/S) vs. V	r	0.686**	0.366	0.387	0.428	0.136	0.247
	p(r)	4.7E-03	2.7E-01	1.7E-01	1.1E-01	6.9E-01	3.9E-01
dA vs. V	r	0.395	0.318	0.400	0.491	0.263	0.288
	p(r)	1.5E-01	3.4E-01	1.6E-01	6.3E-02	4.3E-01	3.2E-01
dL vs. V	r	0.133	0.042	0.171	0.213	0.310	0.122
	p(r)	6.4E-01	9.0E-01	5.6E-01	4.4E-01	3.5E-01	6.8E-01
dln(I) vs. V	r	0.601*	0.449	0.482	0.385	0.558	0.516
	p(r)	1.8E-02	1.7E-01	8.1E-02	1.6E-01	7.4E-02	5.9E-02
dI vs. V	r	0.597*	0.743**	0.625*	-0.199	0.162	0.259
	p(r)	1.9E-02	8.8E-03	1.7E-02	4.8E-01	6.3E-01	3.7E-01
Years(1)	15	11	14	15	11	14

Panel notes: **, * Correlation coefficients significant at two-tailed levels.

Table S.4 (Continued)

PANEL B

Correlation for the Primary Sample and Secondary Sample Change in Total Variables to the Historical Value-at-Risk Actual Variable

Correlation		P	rimary Sample		Sec	ondary Sample	:
Variables	Measure	1994-2008	1994-2004	1994-2007	1994-2008	1994-2004	1994-2007
dln(BM) vs. V	r	-0.752**	-0.372	-0.345	-0.523*	-0.359	-0.366
	p(r)	1.2E-03	2.6E-01	2.3E-01	4.5E-02	2.8E-01	2.0E-01
dBM vs. V	r	-0.811**	-0.353	-0.349	-0.528*	-0.392	-0.407
	p(r)	2.5E-04	2.9E-01	2.2E-01	4.3E-02	2.3E-01	1.5E-01
dln(MB) vs. V	r	0.752**	0.372	0.345	0.523*	0.359	0.366
	p(r)	1.2E-03	2.6E-01	2.3E-01	4.5E-02	2.8E-01	2.0E-01
dMB vs. V	r	0.674**	0.387	0.335	0.189	0.111	0.160
	p(r)	5.8E-03	2.4E-01	2.4E-01	5.0E-01	7.5E-01	5.9E-01
dM vs. V	r	0.820**	0.321	0.337	0.592*	0.396	0.423
	p(r)	1.8E-04	3.4E-01	2.4E-01	2.0E-02	2.3E-01	1.3E-01
dMV vs. V	r	0.658**	0.169	0.183	0.521*	0.375	0.379
	p(r)	7.6E-03	6.2E-01	5.3E-01	4.6E-02	2.6E-01	1.8E-01
dS vs. V	r	-0.760**	-0.465	-0.444	-0.373	0.056	0.009
	p(r)	1.0E-03	1.5E-01	1.1E-01	1.7E-01	8.7E-01	9.8E-01
dln(E) vs. V	r	-0.065	-0.197	-0.194	0.519*	0.336	0.376
	p(r)	8.2E-01	5.6E-01	5.1E-01	4.8E-02	3.1E-01	1.9E-01
dE vs. V	r	-0.103	-0.258	-0.261	0.378	0.261	0.267
	p(r)	7.2E-01	4.4E-01	3.7E-01	1.6E-01	4.4E-01	3.6E-01
dln(E/S) vs. V	r	0.633*	0.096	0.080	0.596*	0.299	0.358
	p(r)	1.1E-02	7.8E-01	7.9E-01	1.9E-02	3.7E-01	2.1E-01
d(E/S) vs. V	r	0.468	0.029	0.018	0.566*	0.311	0.331
	p(r)	7.9E-02	9.3E-01	9.5E-01	2.8E-02	3.5E-01	2.5E-01
dA vs. V	r	-0.421	-0.166	-0.083	0.563*	0.338	0.416
	p(r)	1.2E-01	6.3E-01	7.8E-01	2.9E-02	3.1E-01	1.4E-01
dL vs. V	r	-0.435	-0.163	-0.074	0.299	0.057	0.172
	p(r)	1.0E-01	6.3E-01	8.0E-01	2.8E-01	8.7E-01	5.6E-01
dln(l) vs. V	r	0.296	0.188	0.196	0.671**	0.527	0.571*
	p(r)	2.8E-01	5.8E-01	5.0E-01	6.2E-03	9.6E-02	3.3E-02
dI vs. V	r	0.692**	0.143	0.129	0.571*	0.697*	0.596*
	p(r)	4.2E-03	6.7E-01	6.6E-01	2.6E-02	1.7E-02	2.5E-02
Years(1	()	15	11	14	15	11	14
Panel notes:	**, * Cor	relation coeff	icients signi	ficant at two	o-tailed levels.		

Analysis

The Table S.4 correlation analysis results show that the samples and the control group *change in total variables* to the Historical Value-at-Risk Actual variable exhibit some statistically significant relationships. Majority of the correlations are observed for the Primary and Secondary sample for the 1994 to 2008 time period. With the Historical Value-at-Risk exhibiting statistically significant correlations with the variables change in book-to-market ratio, change in market price, change in total shares outstanding, change in equity per share and change in net income. The change in the net income variable also shows significance for the 1994 to 2004, and 1994 to 2007 time periods. The Control group exhibits statistical significance for the relationship between the change in the market price and the Historical Value-at-Risk Actual for the 1994 to 2008 time period.

A further examination shows that the correlation between the change in the market price variable and the Historical Value-at-Risk Actual for the time period 1994 to 2008, exhibits for the Primary and Secondary sample, a correlation coefficient of 0.684 at a 0.01 significance level. The Control group for the same variables and time period exhibit a correlation of 0.154 at a 0.05 significance level. The correlation between the change in the net income variable and the Historical Value-at-Risk Actual, exhibits for the Primary and Secondary sample a correlation of 0.597 at a 0.05 significance level for the 1994 to 2008 time period, 0.625 at a 0.05 significance level for the 1994 to 2007 time period, and a strong relationship is shown for the 1994 to 2004 time period with a correlation of 0.743 at a 0.01 significance level. For the other significantly related change variables to the Historical Value-at-Risk Actual variable, the correlation coefficient report levels between 0.742 to -0.622 at a significance level below 0.05. The significant correlations between market price return and Historical Value-at-Risk Actual for the samples and the control group from 1994 to 2008, suggests that the market price return reacted at similar levels and direction to the Historical Value-at-Risk Actual variable, after 2005 than before. This may suggest that the samples and the control group market price returns would have varied at the lower levels of its distribution after the 2005 accounting change.

APPENDIX T ANALYSIS AND RESULTS FOR CHANGE IN MARKET VALUE AND MARKET PRICE RETURN REGRESSIONS

This appendix presents the analysis and results for the relation before and after the 2005 accounting change between yearly changes in the market value variable to the market price variable (market price return variable). This analysis is conducted using cross-sectional and time series regressions.

T.1 Regressions for Change in Market Value and Market Price Return

The regression analysis for the change in market value and market price return is presented for the time range 1994 to 2008 in Table T.1 and Table T.2. In the regressions, the market price return variable is tested as the dependent variable, and the change in the market value variable is tested as the independent variable.

Table T.1Cross-Sectional Regression Analysis for Change in Market Value
and Market Price Return

Cross-Sectional Regression

Table T.1 Panel A to Panel D show the results for the cross-sectional regressions that test the change in market value and market price return by applying the regression specified in equation (O.25). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.

Cross-Sectional Regression Table Description

The Table T.1 columns represent the following:

Year is the *panel data* year represented by the number of days from 1st January to 31st December, and also represents the 31st December variable record date for each year.

The Regression column presents the coefficients and related statistics for the regression specified in equation (0.25), and follows the model:

$$dM_{i,t} = a + b \ dMV_{i,t} + e_{i,t}$$

Where: $dM_{i,t}$ is the log Market Price return $(M_{t-1,t})$ for the i^{th} firm at time t, $dMV_{i,t}$ is the change in Market Value $(MV_{t-1,t})$ for the i^{th} firm at time t, and $e_{N,t}$ is the regression error term. The *Lower* and *Upper* are the corresponding coefficient levels predicted at a 99% confidence level.

The Correlation column presents the correlation statistics for the relationship between the dependent and independent variables. The Descriptive Statistics column presents for the independent variable, the sample mean calculated using equation (O.17) and the sample standard deviation, SD, using equation (O.18). The *Obs.* column presents the number of sample firms observed for the year.

								Table T.1	(Conti	nued)								
		Cross-S	Sectional I	Regressi	on Analy	sis for the P	rimary	PAI and Seco	NEL A Indary S	ample Cl	nange in	Marl	ket Valu	e and Mark	et Price Re	eturn		
						Regre	ssion							Com	lation	Descri	ptive	
Year			Slope					Intercept				Model		Corre	lation	Statis	tics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	1.038**	60.41	2.8E-16	0.99	1.09	-0.01*	-2.66	2.1E-02	-0.02	1.E-03	1.00	0.01	12	0.998**	1.4E-16	0.04	0.22	14
1995	-0.322	-2.03	6.5E-02	-0.81	0.16	0.336**	3.92	2.0E-03	0.07	0.60	0.26	0.23	12	-0.506*	3.2E-02	0.38	0.40	14
1996	0.896**	6.19	3.2E-05	0.46	1.33	-0.015	-0.63	5.4E-01	-0.09	0.06	0.75	0.07	13	0.864**	1.6E-05	0.12	0.13	15
1997	0.993**	29.53	2.7E-13	0.89	1.09	-0.003	-0.29	7.8E-01	-0.03	0.03	0.99	0.02	13	0.993**	1.3E-13	0.26	0.18	15
1998	0.775**	4.94	2.7E-04	0.30	1.25	-0.031	-0.78	4.5E-01	-0.15	0.09	0.65	0.14	13	0.808**	1.4E-04	0.12	0.23	15
1999	0.381**	3.35	4.8E-03	0.04	0.72	0.143*	2.41	3.0E-02	-0.03	0.32	0.44	0.19	14	0.667**	2.4E-03	0.31	0.44	16
2000	0.594**	4.68	3.6E-04	0.22	0.97	0.048	0.77	4.5E-01	-0.14	0.23	0.61	0.22	14	0.781**	1.8E-04	0.22	0.46	16
2001	0.903**	13.90	1.4E-09	0.71	1.10	-0.010	-0.50	6.3E-01	-0.07	0.05	0.93	0.07	14	0.966**	7.0E-10	-0.17	0.27	16
2002	1.004**	49.98	3.5E-17	0.94	1.06	-0.012	-1.28	2.2E-01	-0.04	0.02	0.99	0.02	14	0.997**	1.8E-17	-0.36	0.32	16
2003	0.961**	17.88	4.9E-11	0.80	1.12	-0.005	-0.28	7.8E-01	-0.05	0.04	0.96	0.04	14	0.979**	2.4E-11	0.25	0.19	16
2004	0.344*	2.46	2.8E-02	-0.07	0.76	0.056*	2.52	2.4E-02	-0.01	0.12	0.30	0.07	14	0.549*	1.4E-02	0.09	0.14	16
2005	0.637**	9.63	1.5E-07	0.44	0.83	0.059*	2.37	3.3E-02	-0.02	0.13	0.87	0.08	14	0.932**	7.5E-08	0.23	0.31	16
2006	0.553**	3.31	5.1E-03	0.06	1.05	0.063	1.91	7.7E-02	-0.04	0.16	0.44	0.07	14	0.663**	2.6E-03	0.16	0.12	16
2007	0.882**	8.84	4.2E-07	0.58	1.18	0.019	0.66	5.2E-01	-0.07	0.11	0.85	0.11	14	0.921**	2.1E-07	-0.11	0.28	16
2008	1.917**	5.24	1.3E-04	0.83	3.01	0.168	0.93	3.7E-01	-0.37	0.71	0.66	0.38	14	0.814**	6.3E-05	-0.42	0.27	16

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 16 observations per yearly regression from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

							I	Table T.1	(Contin	nued)								
								PA	NEL B									
			Cross-Se	ectional	Regressio	on Analysis	for the	Control (Group C	hange in	Market	Value	e and M	arket Price	Return			
						Regre	ssion							C	1. (Descri	ptive	
Year			Slope					Intercept				Model		Corre	lation	Statis	tics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	a	t(a)	<i>p(a)</i>	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	-0.014	-0.04	9.7E-01	-1.08	1.05	-0.093*	-2.27	5.0E-02	-0.23	0.04	2E-04	0.13	9	-0.014	4.8E-01	-0.04	0.12	11
1995	0.526**	5.39	4.4E-04	0.21	0.84	0.052**	3.63	5.5E-03	0.01	0.10	0.76	0.04	9	0.874**	2.2E-04	0.10	0.11	11
1996	0.46*	2.81	1.8E-02	-0.06	0.98	-0.018	-0.43	6.8E-01	-0.15	0.11	0.44	0.14	10	0.665**	9.2E-03	0.08	0.25	12
1997	0.865**	10.76	8.1E-07	0.61	1.12	0.001	0.08	9.4E-01	-0.05	0.05	0.92	0.05	10	0.959**	4.0E-07	0.12	0.17	12
1998	0.974**	14.69	4.3E-08	0.76	1.18	0.012	0.71	4.9E-01	-0.04	0.07	0.96	0.05	10	0.978**	2.1E-08	-0.12	0.24	12
1999	0.937**	31.95	2.1E-11	0.84	1.03	0.063**	4.48	1.2E-03	0.02	0.11	0.99	0.04	10	0.995**	1.1E-11	0.31	0.38	12
2000	0.997**	18.92	3.7E-09	0.83	1.16	0.05**	3.90	3.0E-03	0.01	0.09	0.97	0.04	10	0.986**	1.8E-09	-0.03	0.25	12
2001	0.72**	5.29	3.5E-04	0.29	1.15	-0.023	-0.71	4.9E-01	-0.13	0.08	0.74	0.04	10	0.858**	1.8E-04	-0.22	0.10	12
2002	0.914**	10.76	8.1E-07	0.64	1.18	-0.023	-0.75	4.7E-01	-0.12	0.07	0.92	0.05	10	0.959**	4.1E-07	-0.31	0.19	12
2003	1**	55.97	8.0E-14	0.94	1.06	0.004	0.84	4.2E-01	-0.01	0.02	1.00	0.01	10	0.998**	4.0E-14	0.23	0.13	12
2004	0.904**	21.88	8.9E-10	0.77	1.04	0.024**	4.06	2.3E-03	0.01	0.04	0.98	0.01	10	0.99**	4.5E-10	0.11	0.09	12
2005	0.714**	6.87	4.3E-05	0.39	1.04	0.122**	3.58	5.0E-03	0.01	0.23	0.83	0.05	10	0.908**	2.2E-05	0.29	0.15	12
2006	0.799**	4.82	7.0E-04	0.27	1.33	0.073*	2.44	3.5E-02	-0.02	0.17	0.70	0.10	10	0.836**	3.5E-04	0.03	0.19	12
2007	0.713**	6.02	1.3E-04	0.34	1.09	0.074	2.10	6.2E-02	-0.04	0.19	0.78	0.12	10	0.885**	6.4E-05	-0.08	0.30	12
2008	0.582**	10.36	1.2E-06	0.40	0.76	-0.145**	-4.52	1.1E-03	-0.25	-0.04	0.91	0.06	10	0.956**	5.8E-07	-0.47	0.33	12

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The Control group consists of a maximum of 12 observations per yearly regression from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

								Table T.1	(Conti	nued)								
			C C		· ·			PA	NEL C	· ·		4 3 7 1	1.54	() (D •	D (
<u>.</u>			Cross-Se	ctional f	(kegressio)	n Analysis I Regre	ssion	rimary S	sample (_nange in	Marke	t valu	e and w	larket Price	Return			
Year			Slope					Intercept				Mode		Corre	lation	Descri Statis	ptive tics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	1.046**	31.90	9.8E-04	0.72	1.37	-0.013	-2.06	1.8E-01	-0.08	0.05	1.00	0.01	2	0.999**	4.9E-04	-0.04	0.22	4
1995	0.99**	48.71	4.2E-04	0.79	1.19	-2.3E-04	-0.03	9.8E-01	-0.08	0.08	1.00	0.01	2	1**	2.1E-04	0.33	0.26	4
1996	1.074*	4.34	4.9E-02	-1.38	3.53	-0.013	-0.21	8.5E-01	-0.60	0.57	0.90	0.04	2	0.951*	2.5E-02	0.23	0.09	4
1997	0.984**	11.81	7.1E-03	0.16	1.81	-0.012	-0.43	7.1E-01	-0.28	0.26	0.99	0.04	2	0.993**	3.5E-03	0.24	0.25	4
1998	0.94**	45.66	4.8E-04	0.74	1.14	-0.007	-2.78	1.1E-01	-0.03	0.02	1.00	5E-03	2	1**	2.4E-04	-0.03	0.14	4
1999	0.280	1.46	2.4E-01	-0.84	1.40	0.171	1.09	3.5E-01	-0.75	1.09	0.42	0.25	3	0.644	1.2E-01	0.58	0.65	5
2000	0.222	2.98	5.8E-02	-0.21	0.66	0.016	0.35	7.5E-01	-0.25	0.28	0.75	0.08	3	0.865*	2.9E-02	0.35	0.54	5
2001	0.915**	10.86	1.7E-03	0.42	1.41	-0.020	-1.78	1.7E-01	-0.09	0.05	0.98	0.03	3	0.988**	8.3E-04	-0.01	0.15	5
2002	0.943**	34.84	5.2E-05	0.79	1.10	-0.027*	-3.26	4.7E-02	-0.07	0.02	1.00	0.01	3	0.999**	2.6E-05	-0.25	0.19	5
2003	0.718*	3.60	3.7E-02	-0.45	1.88	0.027	0.54	6.2E-01	-0.26	0.31	0.81	0.06	3	0.901*	1.8E-02	0.21	0.15	5
2004	0.911	2.68	7.5E-02	-1.08	2.90	-0.008	-0.24	8.2E-01	-0.20	0.18	0.70	0.04	3	0.84*	3.8E-02	0.08	0.05	5
2005	0.738**	48.07	2.0E-05	0.65	0.83	0.004	1.47	2.4E-01	-0.01	0.02	1.00	5E-03	3	0.999**	9.9E-06	0.11	0.16	5
2006	0.889*	4.27	2.4E-02	-0.33	2.11	-0.002	-0.05	9.6E-01	-0.19	0.18	0.86	0.03	3	0.927*	1.2E-02	0.14	0.07	5
2007	1.026**	20.94	2.4E-04	0.74	1.31	-0.018	-1.41	2.5E-01	-0.09	0.06	0.99	0.02	3	0.997**	1.2E-04	-0.15	0.24	5
2008	1.717	2.37	9.9E-02	-2.52	5.96	0.078	0.15	8.9E-01	-3.01	3.17	0.65	0.47	3	0.807*	4.9E-02	-0.67	0.32	5

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 5 observations per yearly regression from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

									Contin	lucu)								
								PAN	NEL D									
		0	Cross-Sect	ional Re	gression	Analysis fo	r the Se	condary S	Sample	Change ir	n Marke	t Valu	ie and N	Iarket Price	e Return			
						Regres	ssion							_		Descri	ntive	
Year			Slope					Intercept				Model		Corre	elation	Statis	stics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	1.032**	45.68	5.8E-11	0.96	1.11	-0.008	-1.57	1.5E-01	-0.03	0.01	1.00	0.02	8	0.998**	1.4E-16	0.08	0.22	10
1995	-0.449**	-3.91	4.5E-03	-0.83	-0.06	0.35**	5.17	8.5E-04	0.12	0.58	0.66	0.16	8	-0.506*	3.2E-02	0.40	0.45	10
1996	0.684**	3.76	4.5E-03	0.09	1.28	-0.014	-0.57	5.8E-01	-0.10	0.07	0.61	0.07	9	0.864**	1.6E-05	0.08	0.12	11
1997	0.995**	26.34	7.9E-10	0.87	1.12	2.7E-04	0.02	9.8E-01	-0.04	0.04	0.99	0.02	9	0.993**	1.3E-13	0.27	0.16	11
1998	0.783**	3.67	5.1E-03	0.09	1.48	-0.039	-0.65	5.3E-01	-0.24	0.16	0.60	0.16	9	0.808**	1.4E-04	0.17	0.24	11
1999	0.74**	3.72	4.8E-03	0.09	1.39	0.092	1.53	1.6E-01	-0.10	0.29	0.61	0.16	9	0.667**	2.4E-03	0.18	0.25	11
2000	0.928**	10.53	2.3E-06	0.64	1.21	0.070	1.84	9.9E-02	-0.05	0.19	0.92	0.12	9	0.781**	1.8E-04	0.16	0.42	11
2001	0.916**	10.08	3.4E-06	0.62	1.21	-0.002	-0.07	9.5E-01	-0.11	0.10	0.92	0.08	9	0.966**	7.0E-10	-0.23	0.29	11
2002	1.01**	39.34	2.2E-11	0.93	1.09	-0.010	-0.73	4.8E-01	-0.05	0.03	0.99	0.03	9	0.997**	1.8E-17	-0.41	0.36	11
2003	0.998**	39.71	2.0E-11	0.92	1.08	-0.006	-0.68	5.1E-01	-0.03	0.02	0.99	0.02	9	0.979**	2.4E-11	0.26	0.21	11
2004	0.317	1.91	8.9E-02	-0.22	0.86	0.067	2.23	5.3E-02	-0.03	0.16	0.29	0.09	9	0.549*	1.4E-02	0.09	0.17	11
2005	0.598**	7.48	3.8E-05	0.34	0.86	0.089*	2.56	3.1E-02	-0.02	0.20	0.86	0.09	9	0.932**	7.5E-08	0.28	0.35	11
2006	0.488*	2.31	4.6E-02	-0.20	1.18	0.083	1.82	1.0E-01	-0.07	0.23	0.37	0.09	9	0.663**	2.6E-03	0.18	0.13	11
2007	0.829**	6.67	9.1E-05	0.43	1.23	0.041	1.08	3.1E-01	-0.08	0.17	0.83	0.12	9	0.921**	2.1E-07	-0.09	0.31	11
2008	2.644**	3.37	8.3E-03	0.09	5.19	0.370	1.40	2.0E-01	-0.49	1.23	0.56	0.37	9	0.814**	6.3E-05	-0.31	0.15	11

Table T 1 (Continued)

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 11 observations per yearly regression from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

Table T.2Time Series Regression Analysis for Change in Market Value and
Market Price Return

Time Series Regression

Table T.2 Panel A to Panel D show the results for the time series regressions that test the change in market value and market price return by applying the regression specified in equation (O.26). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.

Time Series Regression Table Description

The Table T.2 columns represent the following:

The Total Period column represents the time series regression results for the year range 1994 to 2008, with a single year represented from 1st January to 31st December. The Sub-Periods column represents the time series regression results for the year range 1994 to 2004, and for 1994 to 2007, with a single year represented from 1st January to 31st December.

The Measure column represents the time series regression results for the total and sub-time periods and is represented by the data panels:

The *Regression* panel presents the coefficients and related statistics for the regression specified in equation (O.26), and follows the model:

$dM_{N,t} = a + b \ dMV_{N,t} + e_{N,t}$

Where: $dM_{N,t}$ is the log Market Price return $(M_{t-1,t})$ for N firms at time t, $dE_{N,t}$ is the log change in Market Value $(MV_{t-1,t})$ for N firms at time t, and $e_{N,t}$ is the regression error term. The *Lower* and *Upper* are the corresponding coefficient levels predicted at a 99% confidence level.

The *Correlation* panel presents the correlation statistics for the relationship between the dependent and independent variables. The *Descriptive Statistics* panel presents for the variables tested, the average sample means calculated using equation (M.6); the standard deviation, *SD*, of sample means for the time series using equation (M.7); and, the standard error of means, *SE*, for the time series. The *Observations* panel presents: *Firms (n)* (term *N* in the regression) that represents the average number of sample firms in the time series regression (for the Primary and Secondary sample, for the time period 1994 to 2004, the number of firms are rounded down); *Years (T)* represents the number of time series years; *Total (n × T)* represents the number of averaged pooled time series regression observations.

	M		Total Period	Sub-P	Periods
	Measure		1994-2008	1994-2004	1994-2007
	Slope	Ь	1.033**	0.88**	0.872**
	STOP 5	<i>t</i> (<i>b</i>)	13.397	13.717	15.701
		p (b)	5.5E-09	2.4E-07	2.3E-09
		SE(b)	0.077	0.064	0.056
		Lower	0.801	0.672	0.703
		Upper	1.265	1.089	1.042
Regression Correlat Descriptive S Observat	Intercept	а	-0.042*	-0.020	-0.013
Regression	1	t (a)	-2.257	-1.357	-1.040
U		p (a)	4.2E-02	2.1E-01	3.2E-01
		SE(a)	0.019	0.015	0.012
		Lower	-0.098	-0.069	-0.051
		Upper	0.014	0.029	0.025
	Model	R^2	0.932	0.954	0.954
		s(e)	0.069	0.044	0.040
		df(e)	13	9	12
			0.066**	0.077**	0.077**
Correla	tion	r p(r)	2.8E-09	1.2E-07	1.2E-09
		MV	0.074	0.114	0.109
		SD(MV)	0.238	0.216	0.202
Degeninting	Ctatiation	SE(M)	0.061	0.065	0.054
Descriptive	SIGUISTICS	М	0.035	0.080	0.083
		SD (M)	0.254	0.194	0.180
		SE(M)	0.066	0.059	0.048
		Firms (n)	16	15	16
Observat	tions	Years (T)	15	11	14

Table T.2 (Continued)
PANEL A

Time Series Regression Analysis for the Primary and Secondary Sample Change in

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The sample consists of a maximum of 233 observations from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. Observations are based on yearly averages calculated using daily levels from 1994 to 2008.

See main table notes for the regression model applied.

Regression coefficients and related statistics are calculated using OLS regression.

	Magner		Total Period	Sub-F	Periods
	Measure		1994-2008	1994-2004	1994-2007
	Slope	b	0.968**	0.992**	0.99**
		r(0)	1 3E-10	9 0E-08	5 9E-09
		SE(b)	0.053	0.064	0.068
		Lower	0.808	0.782	0.781
		Upper	1.129	1.201	1.199
	Intercept	а	0.020	0.004	0.018
Regression	<u>^</u>	t (a)	-0.936	-0.401	-0.095
		p (a)	3.7E-01	7.0E-01	9.3E-01
		SE(a)	0.021	0.025	0.020
		Lower	-0.014	-0.034	-0.019
		Upper	0.054	0.042	0.055
	Model	R^2	0.962	0.963	0.946
		s(e)	0.044	0.038	0.045
		df(e)	13	9	12
~ I		r	0.981**	0.982**	0.973**
Correla	tion	p(r)	6.3E-11	4.5E-08	2.9E-09
		MV	-3.6E-04	0.021	0.033
		SD (MV)	0.219	0.188	0.183
Descriptive	Statistics	SE(M)	0.057	0.057	0.049
Descriptive	Giunstics	Μ	0.020	0.025	0.051
		SD(M)	0.217	0.190	0.186
		SE(M)	0.056	0.057	0.050
		Firms (n)	12	12	12
Observa	tions	Years (T)	15	11	14
		$T_{otal}(n \times T)$	178	120	166

Table T.2 (Continued) PANEL B

Time Series Regression Analysis for the Control Group Change in Market Value and

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The control consists of a maximum of 178 observations from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. Observations are based on yearly averages calculated using daily levels from 1994 to 2008.

See main table notes for the regression model applied.

Regression coefficients and the related statistics are calculated using OLS regression.

	Maggire		Total Period	Sub-F	Periods
	Measure		1994-2008	1994-2004	1994-2007
	Slope	b	1.087**	0.727**	0.748**
	I I	t (b)	9.540	6.856	8.479
		p (b)	3.1E-07	7.4E-05	2.1E-06
		SE(b)	0.114	0.106	0.088
		Lower	0.744	0.382	0.479
		Upper	1.431	1.071	1.017
Regression Correla	Intercent	a	-0.079*	-0.014	-0.017
Regression		t(a)	-2.348	-0.509	-0.770
-		p (a)	3.5E-02	6.2E-01	4.6E-01
Regression Correla Descriptive Observa		SE(a)	0.033	0.028	0.022
		Lower	-0.179	-0.107	-0.083
		Upper	0.022	0.078	0.050
legression Correl	Model	R^2	0.875	0.839	0.857
		s(e)	0.125	0.078	0.070
		df(e)	13	9	12
			0.025**	0.016**	0.026**
Correl	ation	r p(r)	1.5E-07	3.7E-05	1.0E-06
		MV SD (1997)	0.074	0.152	0.127
		SD(MV)	0.294	0.232	0.219
Descriptive	Statistics	SE(M)	0.076	0.070	0.058
*		М	0.001	0.096	0.078
		SD(M)	0.342	0.184	0.177
		SE(M)	0.088	0.055	0.047
		Firms (n)	5	5	5
Observa	ations	Years (T)	15	11	14

Table T.2 (Continued) PANEL C

Time Series Regression Analysis for the Primary Sample Change in Market Value and Market Price Return

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The sample consists of a maximum of 70 observations from the 5 banks in the LSE that adopted the IFRS standards in 2005. Observations are based on yearly averages calculated using daily levels from 1994 to 2008.

See main table notes for the regression model applied.

Regression coefficients and the related statistics are calculated using OLS regression.

		Mark	et Price Return		
	N (Total Period	Sub-F	Periods
	Measure		1994-2008	1994-2004	1994-2007
	Slope	b	0.948**	0.868**	0.861**
	1	t (b)	10.573	8.199	9.723
		p (b)	9.4E-08	1.8E-05	4.8E-07
		SE(b)	0.090	0.106	0.089
		Lower	0.678	0.524	0.591
		Upper	1.218	1.213	1.132
	Intercent	a	-0.020	-0.010	-0.002
Regression		t (a)	-0.936	-0.401	-0.095
-		p (a)	3.7E-01	7.0E-01	9.3E-01
		SE(a)	0.021	0.025	0.020
		Lower	-0.084	-0.093	-0.064
		Upper	0.044	0.072	0.061
	Model	R^2	0.896	0.882	0.887
		s(e)	0.078	0.077	0.069
		df(e)	13	9	12
		14	0.946**	0 030**	0 0/2**
Correla	ntion	p(r)	4.7E-08	9.1E-06	2.4E-07
		MV	0 074	0.095	0 101
		SD(MV)	0.233	0.230	0.216
Deservinting	Ctatistics.	SE(M)	0.060	0.069	0.058
Descriptive	SIGUISTICS	М	0.050	0.072	0.085
		SD (M)	0.233	0.213	0.197
		SE(M)	0.060	0.064	0.053
		Firms (n)			11
Ohserva	tions	Years (T)	15	11	14
2 0.001 10		Total ($n \times T$)	163	119	152

Table T.2 (Continued)
PANEL D

Time Series Regression Analysis for the Secondary Sample Change in Market Value and

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The sample consists of a maximum of 163 observations from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. Observations are based on yearly averages calculated using daily levels from 1994 to 2008.

See main table notes for the regression model applied.

Regression coefficients and the related statistics are calculated using OLS regression.

Analysis

The Table T.1 and Table T.2 cross-sectional and time series regression slopes show that the change in the market value variable exhibits statistically significant explanatory power to the market price return variable for the samples and the control group for the time periods 1994 to 2008, 1994 to 2004 and 1994 to 2007.

The Table T.2 Panel C time series regression slope for the Primary sample shows that before 2005, a 1% variation in the change in the market value variable related to less than a 1% change in the market price return variable. After 2005, the slope shows that for a 1% variation in the change in market value the market price return changed more than 1%.

APPENDIX U ANALYSIS AND RESULTS FOR CHANGE IN ACCOUNTING AND CHANGE IN MARKET PRICE CROSS-SECTIONAL REGRESSIONS

This appendix present analysis and results for the yearly key accounting totals and market price return cross-sectional regressions.

U.1 Regressions for Change in Total Equity and Market Price Return

The regression analysis for the change in total equity and market price return is presented for the time range 1994 to 2008 in Table U.1. In the regressions, the market price return variable is tested as the dependent variable, and the change in the total equity variable is tested as the independent variable.

Table U.1Cross-Sectional Regression Analysis for Change in Total Equity and
Market Price Return

Cross-Sectional Regression

Table U.1 Panel A to Panel D show the results for the cross sectional regressions that test the change in total equity and market price return by applying the regression specified in equation (O.27). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.

Cross-Sectional Regression Table Description

The Table U.1 columns represent the following:

Year is the *panel data* year represented by the number of days from 1st January to 31st December, and also represents the 31st December variable record date for each year.

The Regression column presents the coefficients and related statistics for the regression specified in equation (O.27), and follows the model:

$$dM_{i,t} = a + b \ dE_{i,t} + e_{i,t}$$

Where: $dM_{i,t}$ is the log Market Price return $(M_{t-1,t})$ for the i^{th} firm at time t, $dE_{i,t}$ is the change in Total Equity $(E_{t-1,t})$ for the i^{th} firm at time t, and $e_{N,t}$ is the regression error term. The *Lower* and *Upper* are the corresponding coefficient levels predicted at a 99% confidence level.

The Correlation column presents the correlation statistics for the relationship between the dependent and independent variables. The Descriptive Statistics column presents for the independent variable, the sample mean calculated using equation (O.17) and the sample standard deviation, *SD*, using equation (O.18). The *Obs.* column presents the number of sample firms observed for the year.

							,	Table U.1	(Conti	nued)								
	Cross	-Sectior	nal Regres	ssion An	alysis for	the Primar	y and S	PA1 econdary	NEL A Sample	e Change i	in Total	Equi	ty ($E_{t-1,t}$)	and Market	Price Retu	rn (<i>M</i> _{t-1,t})		
						Regres	sion									Descri	ptive	
Year			Slope					Intercept				Model		Correl	ation	Statis	tics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	0.640	1.23	2.4E-01	-0.95	2.23	0.210	-0.33	7.5E-01	-0.26	0.21	0.11	0.23	12	0.335	1.2E-01	0.09	0.12	14
1995	-0.269**	-3.46	4.7E-03	-0.51	-0.03	0.504**	5.47	1.4E-04	0.14	0.50	0.50	0.19	12	-0.707**	2.4E-03	0.41	0.66	14
1996	-0.007	-0.03	9.8E-01	-0.83	0.82	0.270	1.56	1.4E-01	-0.09	0.27	4.E-05	0.14	13	-0.007	4.9E-01	0.17	0.13	15
1997	0.001	0.01	9.9E-01	-0.34	0.34	0.417**	4.77	3.6E-04	0.09	0.42	1.E-05	0.18	13	0.003	5.0E-01	0.22	0.43	15
1998	0.189	0.56	5.8E-01	-0.82	1.20	0.241	0.61	5.6E-01	-0.16	0.24	0.02	0.23	13	0.155	2.9E-01	0.09	0.18	15
1999	0.374	0.79	4.5E-01	-1.04	1.79	0.504	1.87	8.3E-02	-0.12	0.50	0.04	0.25	14	0.205	2.2E-01	0.17	0.14	16
2000	0.023	0.28	7.9E-01	-0.23	0.27	0.453	1.74	1.0E-01	-0.12	0.45	0.01	0.36	14	0.074	3.9E-01	0.42	1.11	16
2001	-0.130	-0.68	5.1E-01	-0.70	0.44	0.047*	-2.27	4.0E-02	-0.34	0.05	0.03	0.25	14	-0.179	2.5E-01	0.08	0.34	16
2002	0.136	0.32	7.5E-01	-1.12	1.39	-0.125**	-4.49	5.1E-04	-0.62	-0.13	0.01	0.33	14	0.086	3.8E-01	-0.02	0.20	16
2003	0.384	1.18	2.6E-01	-0.59	1.35	0.362**	3.30	5.3E-03	0.02	0.36	0.09	0.18	14	0.301	1.3E-01	0.11	0.14	16
2004	-0.4*	-2.24	4.2E-02	-0.93	0.13	0.17**	5.01	1.9E-04	0.04	0.17	0.26	0.08	14	-0.514*	2.1E-02	0.05	0.11	16
2005	0.314**	5.10	1.6E-04	0.13	0.50	0.227**	3.27	5.6E-03	0.01	0.23	0.65	0.13	14	0.806**	8.1E-05	0.27	0.54	16
2006	-0.094	-0.43	6.8E-01	-0.75	0.57	0.278**	4.44	5.6E-04	0.06	0.28	0.01	0.10	14	-0.113	3.4E-01	0.13	0.12	16
2007	-0.332	-0.62	5.5E-01	-1.94	1.27	0.216	-0.50	6.2E-01	-0.30	0.22	0.03	0.28	14	-0.163	2.7E-01	0.10	0.13	16
2008	-1.077*	-2.68	1.8E-02	-2.27	0.12	-0.186**	-4.35	6.6E-04	-0.99	-0.19	0.34	0.53	14	-0.583**	8.9E-03	0.05	0.34	16

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 16 observations per yearly regression from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

							1	Table U.1	(Contin	nued)								
		~	~ .				~	PA	NEL B		-							
		Cros	s-Sectiona	al Regre	ssion Ana	alysis for the	e Contr	ol Group	Change	in Total	Equity ($E_{t-1,t}$)	and Ma	rket Price R	teturn (<i>M_{t-1}</i>	, <i>t</i>)		
						Regre	ssion							Corre	lation	Descri	ptive	
Year			Slope					Intercept				Model		Cone	lation	Statis	tics	Obs.
	в	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	-0.316	-1.51	1.6E-01	-0.99	0.36	-0.098*	-2.81	2.0E-02	-0.21	0.02	0.20	0.12	9	-0.451	8.2E-02	-0.02	0.17	11
1995	0.752**	4.80	9.7E-04	0.24	1.26	-0.021	-0.74	4.8E-01	-0.11	0.07	0.72	0.04	9	0.848**	4.8E-04	0.16	0.08	11
1996	0.256	1.60	1.4E-01	-0.25	0.76	-0.020	-0.37	7.2E-01	-0.19	0.15	0.21	0.16	10	0.452	7.0E-02	0.16	0.31	12
1997	1.052**	12.93	1.4E-07	0.79	1.31	-0.032	-2.08	6.4E-02	-0.08	0.02	0.94	0.04	10	0.971**	7.2E-08	0.13	0.15	12
1998	1.315**	12.79	1.6E-07	0.99	1.64	-0.054*	-3.03	1.3E-02	-0.11	2.5E-03	0.94	0.06	10	0.971**	8.0E-08	-0.04	0.17	12
1999	0.495**	20.92	1.4E-09	0.42	0.57	0.134**	6.93	4.1E-05	0.07	0.20	0.98	0.06	10	0.989**	6.9E-10	0.45	0.72	12
2000	1.116**	7.88	1.3E-05	0.67	1.57	0.051	1.76	1.1E-01	-0.04	0.14	0.86	0.10	10	0.928**	6.7E-06	-0.03	0.21	12
2001	0.8**	4.09	2.2E-03	0.18	1.42	-0.026	-0.64	5.4E-01	-0.16	0.10	0.63	0.05	10	0.791**	1.1E-03	-0.20	0.08	12
2002	1.58**	10.48	1.0E-06	1.10	2.06	0.124*	2.83	1.8E-02	-0.02	0.26	0.92	0.05	10	0.957**	5.1E-07	-0.27	0.11	12
2003	0.707**	8.83	4.9E-06	0.45	0.96	0.042	1.62	1.4E-01	-0.04	0.12	0.89	0.05	10	0.941**	2.5E-06	0.28	0.17	12
2004	0.963**	8.91	4.5E-06	0.62	1.31	0.009	0.59	5.7E-01	-0.04	0.06	0.89	0.03	10	0.942**	2.3E-06	0.12	0.08	12
2005	0.676**	6.79	4.8E-05	0.36	0.99	0.139**	4.34	1.5E-03	0.04	0.24	0.82	0.05	10	0.906**	2.4E-05	0.28	0.16	12
2006	0.835**	3.59	4.9E-03	0.10	1.57	0.057	1.53	1.6E-01	-0.06	0.18	0.56	0.12	10	0.75**	2.5E-03	0.04	0.16	12
2007	0.808**	6.06	1.2E-04	0.39	1.23	0.046	1.34	2.1E-01	-0.06	0.15	0.79	0.12	10	0.887**	6.1E-05	-0.03	0.27	12
2008	1.212**	10.47	1.0E-06	0.85	1.58	-0.008	-0.18	8.6E-01	-0.15	0.13	0.92	0.06	10	0.957**	5.2E-07	-0.34	0.16	12

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The Control group consists of a maximum of 12 observations per yearly regression from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

							,	Table U.1	(Contin	nued)								
								PA	NEL C						<i></i>			
		C	cross-Section	ional Re	gression A	Analysis for	the Pri	mary Ch	ange in	Total Equ	uity (<i>E_{t-1,}</i>	_t) and	Marke	t Price Retu	$rn(M_{t-1,t})$			
						Regre	ssion							Corre	lation	Descri	ptive	
Year			Slope					Intercept				Model		Colle	ation	Statis	tics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	-2.557	-1.96	1.9E-01	-15.52	10.41	0.273	1.46	2.8E-01	-1.59	2.14	0.66	0.17	2	-0.811	9.5E-02	0.13	0.07	4
1995	0.184	0.40	7.3E-01	-4.38	4.74	0.256	1.13	3.8E-01	-1.99	2.51	0.07	0.30	2	0.273	3.6E-01	0.37	0.38	4
1996	-0.207	-0.27	8.1E-01	-7.73	7.31	0.262	1.96	1.9E-01	-1.07	1.59	0.04	0.12	2	-0.189	4.1E-01	0.16	0.09	4
1997	0.126	0.18	8.7E-01	-6.80	7.05	0.226	1.49	2.8E-01	-1.28	1.74	0.02	0.30	2	0.127	4.4E-01	-0.01	0.25	4
1998	0.339	0.21	8.5E-01	-15.51	16.19	-0.045	-0.48	6.8E-01	-0.96	0.87	0.02	0.16	2	0.148	4.3E-01	0.03	0.06	4
1999	1.481	1.64	2.0E-01	-3.80	6.77	0.004	0.02	9.9E-01	-1.32	1.33	0.47	0.24	3	0.687	1.0E-01	0.22	0.13	5
2000	0.058	2.10	1.3E-01	-0.10	0.22	0.025	0.45	6.8E-01	-0.30	0.35	0.60	0.10	3	0.771	6.3E-02	1.18	1.86	5
2001	0.436	0.38	7.3E-01	-6.32	7.19	-0.087	-0.54	6.3E-01	-1.03	0.85	0.05	0.16	3	0.213	3.7E-01	0.13	0.07	5
2002	0.754	1.10	3.5E-01	-3.26	4.76	-0.220	-2.52	8.6E-02	-0.73	0.29	0.29	0.17	3	0.535	1.8E-01	-0.06	0.13	5
2003	-0.219	-0.46	6.8E-01	-3.01	2.57	0.200	2.51	8.7E-02	-0.27	0.67	0.07	0.13	3	-0.256	3.4E-01	0.11	0.14	5
2004	-0.064	-0.08	9.4E-01	-4.50	4.38	0.070	1.17	3.3E-01	-0.28	0.42	2E-03	0.07	3	-0.048	4.7E-01	0.07	0.04	5
2005	0.47*	4.84	1.7E-02	-0.10	1.04	2.4E-04	0.01	9.9E-01	-0.16	0.16	0.89	0.05	3	0.942**	8.4E-03	0.18	0.23	5
2006	0.565	1.39	2.6E-01	-1.81	2.94	0.048	0.81	4.8E-01	-0.30	0.39	0.39	0.06	3	0.626	1.3E-01	0.13	0.08	5
2007	-0.591	-0.36	7.5E-01	-10.30	9.12	-0.057	-0.17	8.8E-01	-2.07	1.95	0.04	0.28	3	-0.201	3.7E-01	0.19	0.08	5
2008	0.349	0.26	8.1E-01	-7.44	8.14	-1.124	-2.79	6.9E-02	-3.48	1.23	0.02	0.78	3	0.150	4.1E-01	0.15	0.29	5

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 5 observations per yearly regression from the 5 banks in the LSE that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

								Table U.1	(Conti	nued)								
								PAI	NEL D									
		Cross-	Sectional	Regress	ion Analy	sis for the S	Seconda	ry Sampl	e Chan	ge in Tota	l Equity	$V(E_{t-1,t})$	t) and M	arket Price	Return (<i>M_t</i> .	1,t)		
						Regres	sion									Descri	ptive	
Year			Slope					Intercept				Model		Corre	ation	Statis	tics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	a	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	1.093*	2.35	4.7E-02	-0.47	2.66	-0.014	-0.21	8.4E-01	-0.25	0.22	0.41	0.19	8	0.639*	2.3E-02	0.08	0.14	10
1995	-0.302**	-6.79	1.4E-04	-0.45	-0.15	0.297**	7.93	4.7E-05	0.17	0.42	0.85	0.10	8	-0.923**	7.0E-05	0.42	0.77	10
1996	0.069	0.30	7.7E-01	-0.68	0.81	0.028	0.54	6.1E-01	-0.14	0.20	0.01	0.11	9	0.100	3.8E-01	0.18	0.15	11
1997	-0.027	-0.24	8.1E-01	-0.39	0.34	0.276**	4.59	1.3E-03	0.08	0.47	0.01	0.16	9	-0.081	4.1E-01	0.31	0.46	11
1998	0.118	0.30	7.7E-01	-1.15	1.38	0.078	0.87	4.1E-01	-0.21	0.37	0.01	0.25	9	0.100	3.8E-01	0.12	0.21	11
1999	-0.111	-0.20	8.5E-01	-1.96	1.74	0.242	2.10	6.5E-02	-0.13	0.62	4E-03	0.25	9	-0.065	4.2E-01	0.15	0.14	11
2000	0.208	0.30	7.7E-01	-2.07	2.49	0.199	1.42	1.9E-01	-0.26	0.65	0.01	0.43	9	0.098	3.9E-01	0.08	0.19	11
2001	-0.159	-0.75	4.7E-01	-0.85	0.53	-0.207*	-2.43	3.8E-02	-0.48	0.07	0.06	0.28	9	-0.243	2.4E-01	0.07	0.42	11
2002	0.130	0.25	8.1E-01	-1.58	1.83	-0.42**	-3.66	5.2E-03	-0.79	-0.05	0.01	0.38	9	0.082	4.1E-01	2E-03	0.23	11
2003	0.584	1.44	1.8E-01	-0.73	1.90	0.195*	2.65	2.6E-02	-0.04	0.43	0.19	0.20	9	0.433	9.2E-02	0.11	0.15	11
2004	-0.406	-1.98	7.9E-02	-1.07	0.26	0.113**	4.13	2.6E-03	0.02	0.20	0.30	0.09	9	-0.551*	3.9E-02	0.04	0.13	11
2005	0.291**	4.43	1.6E-03	0.08	0.51	0.167**	3.69	5.0E-03	0.02	0.31	0.69	0.13	9	0.828**	8.2E-04	0.31	0.64	11
2006	-0.187	-0.73	4.9E-01	-1.02	0.65	0.193**	4.21	2.3E-03	0.04	0.34	0.06	0.11	9	-0.236	2.4E-01	0.13	0.13	11
2007	-0.039	-0.06	9.6E-01	-2.37	2.29	-0.032	-0.33	7.5E-01	-0.35	0.29	3E-04	0.30	9	-0.018	4.8E-01	0.06	0.13	11
2008	-1.256**	-5.31	4.9E-04	-2.03	-0.49	-0.441**	-5.32	4.8E-04	-0.71	-0.17	0.76	0.27	9	-0.871**	2.4E-04	-2E-03	0.37	11

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 11 observations per yearly regression from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

Analysis

The Table U.1 cross-sectional regression slopes show that the change in the total equity variable exhibits varying levels of statistically significant explanatory power to the market price return variable for the samples and the control group.

The Primary and Secondary sample shows statistical significance for years 1995, 2004, 2005 and 2008. The Control group shows extensive statistical significance other than for the two years: 1994 and 1996.

U.2 Regressions for Change in Total Equity per Share and Market Price Return

The regression analysis for the change in total equity per share and market price return is presented for the time range 1994 to 2008 in Table U.2. In the regressions, the market price return variable is tested as the dependent variable, and the change in the total equity per share variable is tested as the independent variable.

Table U.2Cross-Sectional Regression Analysis for Change in Total Equity per
Share and Market Price Return

Cross-Sectional Regression

Table U.2 Panel A to Panel D show the results for the cross-sectional regressions that test the change in total equity per share and market price return by applying the regression specified in equation (O.27). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.

Cross-Sectional Regression Table Description

The Table U.2 columns represent the following:

Year is the panel data year represented by the number of days from 1st January to 31st December, and also represents the 31st December variable record date for each year.

The Regression column presents the coefficients and related statistics for the regression specified in equation (O.27), and follows the model:

$$dM_{i,t} = a + b \ d(E/S)_{i,t} + e_{i,t}$$

Where: $dM_{i,t}$ is the log Market Price return $(M_{t-1,t})$ for the i^{th} firm at time t, $d(E/S)_{i,t}$ is the change in Total Equity per Share $(E_{t-1,t}/S_{t-1,t})$ for the i^{th} firm at time t, and $e_{N,t}$ is the regression error term. The *Lower* and *Upper* are the corresponding coefficient levels predicted at a 99% confidence level.

The Correlation column presents the correlation statistics for the relationship between the dependent and independent variables. The Descriptive Statistics column presents for the independent variable, the sample mean calculated using equation (O.17) and the sample standard deviation, *SD*, using equation (O.18). The *Obs.* column presents the number of sample firms observed for the year.

								,	Table U.2	2 (Conti	nued)								
_	C	ross-Section	191 Reor	ression An	alvsis fo	or the Pri	mary and S	econdai	PAI rv Sample	NEL A e Chang	e in Total	Equity	ner S	hare (E.	/S) and	Market Pri	ice Return	(<i>M</i>)	
-			iur regi		<i>uiy</i> 515 10	<u> </u>	Regres	sion	<u>y sumpr</u>	^c chung		Lquity	per si	iure (12).	<u>1,1,1,5,1,1) unu</u>	:	Descri	ptive	
	Year			Slope					Intercept				Model		Corre	lation	Statis	tics	Obs.
_		b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	a	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
	1994	-0.749	-1.77	1.0E-01	-2.04	0.54	0.244	1.06	3.1E-01	-0.12	0.24	0.21	0.21	12	-0.455	5.1E-02	0.04	0.14	14
	1995	0.567**	5.18	2.3E-04	0.23	0.90	0.294**	4.32	9.9E-04	0.05	0.29	0.69	0.15	12	0.831**	1.1E-04	0.07	0.37	14
	1996	0.430	0.73	4.8E-01	-1.34	2.20	0.290	0.40	6.9E-01	-0.22	0.29	0.04	0.13	13	0.199	2.4E-01	0.13	0.06	15
	1997	0.051	0.15	8.8E-01	-0.98	1.09	0.475**	3.27	6.1E-03	0.02	0.48	2E-03	0.18	13	0.041	4.4E-01	0.17	0.14	15
6	1998	0.450	0.93	3.7E-01	-1.01	1.91	0.225	0.49	6.3E-01	-0.16	0.23	0.06	0.22	13	0.249	1.9E-01	0.06	0.12	15
11	1999	0.854	1.74	1.1E-01	-0.63	2.34	0.457	0.94	3.6E-01	-0.24	0.46	0.19	0.22	13	0.434	5.3E-02	0.20	0.12	15
	2000	0.118	0.30	7.7E-01	-1.07	1.30	0.502	1.32	2.1E-01	-0.19	0.50	0.01	0.36	14	0.079	3.9E-01	0.19	0.23	16
	2001	-0.174	-0.73	4.8E-01	-0.89	0.54	0.05*	-2.22	4.4E-02	-0.34	0.05	0.04	0.25	14	-0.191	2.4E-01	0.08	0.27	16
	2002	0.173	0.37	7.2E-01	-1.23	1.58	-0.118**	-4.39	6.2E-04	-0.61	-0.12	0.01	0.33	14	0.098	3.6E-01	-0.03	0.18	16
	2003	0.397	1.17	2.6E-01	-0.61	1.40	0.36**	3.50	3.6E-03	0.03	0.36	0.09	0.18	14	0.299	1.3E-01	0.10	0.14	16
	2004	-0.146	-0.94	3.6E-01	-0.61	0.32	0.164**	4.03	1.2E-03	0.02	0.16	0.06	0.09	14	-0.245	1.8E-01	0.06	0.14	16
	2005	0.71**	7.62	2.4E-06	0.43	0.99	0.154	2.12	5.3E-02	-0.03	0.15	0.81	0.10	14	0.898**	1.2E-06	0.20	0.27	16
	2006	0.143	0.52	6.1E-01	-0.68	0.97	0.259**	3.37	4.6E-03	0.02	0.26	0.02	0.10	14	0.137	3.1E-01	0.12	0.09	16
	2007	-0.165	-0.28	7.9E-01	-1.94	1.61	0.217	-0.64	5.3E-01	-0.34	0.22	0.01	0.28	14	-0.074	3.9E-01	0.10	0.12	16
	2008	1.454**	3.41	4.3E-03	0.18	2.73	0.019*	-2.84	1.3E-02	-0.82	0.02	0.45	0.49	14	0.673**	2.1E-03	-0.16	0.29	16

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 16 observations per yearly regression from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

								Table U.2	2 (Conti	nued)								
	~	~						PA	NEL B		~		(~)					
	Cros	s-Sectio	onal Regre	ession Ai	nalysis fo	r the Contro	ol Grou	p Change	e in Tota	al Equity	per Sha	re (E_t .	$(1,t/S_{t-1,t})$	and Market	Price Retu	$\operatorname{Irn}\left(\boldsymbol{M}_{t-1,t}\right)$		
						Regre	ssion							Com	lation	Descri	ptive	
Year			Slope					Intercept				Model		Colle	lation	Statis	tics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	1.816**	7.75	2.8E-05	1.05	2.58	0.047	2.07	6.9E-02	-0.03	0.12	0.87	0.05	9	0.933**	1.4E-05	-0.08	0.06	11
1995	0.611**	3.61	5.7E-03	0.06	1.16	0.007	0.22	8.3E-01	-0.09	0.10	0.59	0.05	9	0.769**	2.8E-03	0.16	0.09	11
1996	1.133**	7.14	3.1E-05	0.63	1.64	-0.059*	-2.43	3.5E-02	-0.14	0.02	0.84	0.07	10	0.914**	1.6E-05	0.07	0.14	12
1997	1.052**	12.93	1.4E-07	0.79	1.31	-0.032	-2.08	6.4E-02	-0.08	0.02	0.94	0.04	10	0.971**	7.2E-08	0.13	0.15	12
1998	1.311**	12.94	1.4E-07	0.99	1.63	-0.055*	-3.14	1.1E-02	-0.11	5E-04	0.94	0.06	10	0.971**	7.2E-08	-0.04	0.18	12
1999	0.511**	20.29	1.9E-09	0.43	0.59	0.101**	4.82	7.0E-04	0.03	0.17	0.98	0.06	10	0.988**	9.3E-10	0.50	0.69	12
2000	1.171**	13.44	1.0E-07	0.89	1.45	-0.005	-0.30	7.7E-01	-0.06	0.05	0.95	0.06	10	0.973**	5.0E-08	0.02	0.21	12
2001	1.045**	6.81	4.7E-05	0.56	1.53	-0.011	-0.40	6.9E-01	-0.10	0.08	0.82	0.04	10	0.907**	2.3E-05	-0.16	0.07	12
2002	1.454**	8.31	8.4E-06	0.90	2.01	0.071	1.43	1.8E-01	-0.09	0.23	0.87	0.07	10	0.935**	4.2E-06	-0.26	0.11	12
2003	0.706**	9.67	2.2E-06	0.47	0.94	0.038	1.59	1.4E-01	-0.04	0.11	0.90	0.04	10	0.95**	1.1E-06	0.28	0.18	12
2004	1.007**	6.85	4.5E-05	0.54	1.47	-0.010	-0.43	6.7E-01	-0.08	0.06	0.82	0.04	10	0.908**	2.2E-05	0.13	0.07	12
2005	0.927**	16.92	1.1E-08	0.75	1.10	0.026	1.34	2.1E-01	-0.04	0.09	0.97	0.02	10	0.983**	5.5E-09	0.33	0.13	12
2006	1.192**	16.08	1.8E-08	0.96	1.43	-0.047**	-3.43	6.5E-03	-0.09	-4.E-03	0.96	0.04	10	0.981**	9.0E-09	0.12	0.15	12
2007	1.089**	19.06	3.4E-09	0.91	1.27	-0.036*	-2.93	1.5E-02	-0.08	3.E-03	0.97	0.04	10	0.987**	1.7E-09	0.05	0.22	12
2008	1.262**	10.57	9.5E-07	0.88	1.64	0.003	0.06	9.5E-01	-0.14	0.14	0.92	0.06	10	0.958**	4.8E-07	-0.34	0.15	12

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The Control group consists of a maximum of 12 observations per yearly regression from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

							r	Table U.2	(Conti	nued)								
	C	G	1.0		1 • 6	(1 D •	C	PA	NEL C	15 4				116 1	(D • D (
	Cros	s-Section	nal Regres	ssion An	alysis for	the Prima Regre	ry Samp ession	le Chang	e in Tot	al Equity	per Sha	re (E_t	$(1,t/S_{t-1,t})$	and Marke	t Price Reti	$\operatorname{Irn}\left(M_{t-1,t}\right)$		
Year			Slope					Intercept				Model		Corre	lation	Descri Statis	ptive stics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	-2.207	-1.35	3.1E-01	-18.47	14.06	0.197	0.92	4.5E-01	-1.93	2.32	0.48	0.20	2	-0.690	1.6E-01	0.11	0.07	4
1995	3.381*	5.46	3.2E-02	-2.77	9.53	-0.323	-2.59	1.2E-01	-1.56	0.92	0.94	0.08	2	0.968*	1.6E-02	0.19	0.07	4
1996	-0.961	-0.54	6.4E-01	-18.51	16.59	0.358	1.47	2.8E-01	-2.06	2.77	0.13	0.11	2	-0.359	3.2E-01	0.13	0.04	4
1997	0.149	0.04	9.7E-01	-41.46	41.75	0.212	0.52	6.5E-01	-3.81	4.23	6E-04	0.31	2	0.025	4.9E-01	0.09	0.04	4
1998	-0.038	-0.03	9.8E-01	-14.39	14.31	-0.034	-0.39	7.4E-01	-0.90	0.84	4E-04	0.16	2	-0.019	4.9E-01	0.02	0.06	4
1999	1.172*	9.55	1.1E-02	-0.05	2.39	0.205*	7.14	1.9E-02	-0.08	0.49	0.98	0.03	2	0.989**	5.4E-03	0.20	0.14	4
2000	0.465*	4.33	2.3E-02	-0.16	1.09	-0.080	-1.66	2.0E-01	-0.36	0.20	0.86	0.06	3	0.928*	1.1E-02	0.37	0.28	5
2001	0.162	0.12	9.1E-01	-7.87	8.19	-0.049	-0.31	7.8E-01	-0.98	0.89	5E-03	0.16	3	0.068	4.6E-01	0.10	0.06	5
2002	0.580	0.78	4.9E-01	-3.75	4.91	-0.227	-2.34	1.0E-01	-0.79	0.34	0.17	0.19	3	0.412	2.5E-01	-0.07	0.13	5
2003	-0.666	-1.19	3.2E-01	-3.94	2.61	0.225*	3.45	4.1E-02	-0.16	0.61	0.32	0.11	3	-0.566	1.6E-01	0.07	0.10	5
2004	1.111	0.99	4.0E-01	-5.47	7.70	0.010	0.16	8.9E-01	-0.36	0.38	0.24	0.06	3	0.494	2.0E-01	0.05	0.03	5
2005	0.634*	3.66	3.5E-02	-0.38	1.65	-0.004	-0.12	9.1E-01	-0.21	0.20	0.82	0.06	3	0.904*	1.8E-02	0.14	0.17	5
2006	0.791	2.00	1.4E-01	-1.52	3.10	0.032	0.63	5.7E-01	-0.26	0.33	0.57	0.05	3	0.756	6.9E-02	0.11	0.07	5
2007	-0.440	-0.19	8.6E-01	-14.29	13.41	-0.097	-0.23	8.3E-01	-2.53	2.34	0.01	0.28	3	-0.106	4.3E-01	0.17	0.06	5
2008	1.449	2.02	1.4E-01	-2.74	5.64	-0.893*	-3.63	3.6E-02	-2.33	0.54	0.58	0.51	3	0.759	6.8E-02	-0.12	0.36	5

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 5 observations per yearly regression from the 5 banks in the LSE that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

									Table U.2	(Conti	nued)								
		Cross-	Section	al Regress	ion Ana	lysis for 1	the Seconda	ry Sam	PA ple Chan	NEL D ge in To	tal Equity	y per Sh	are (E	$S_{t-1,t}/S_{t-1,t}$) and Marke	t Price Ret	urn (<i>M</i> _{t-1,t}))	
							Regres	sion									Descri	ntive	
	Year			Slope					Intercept				Model		Corre	lation	Statis	tics	Obs.
		b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
	1994	-0.570	-1.13	2.9E-01	-2.26	1.12	0.075	1.04	3.3E-01	-0.17	0.32	0.14	0.23	8	-0.372	1.4E-01	0.01	0.15	10
	1995	0.523**	6.00	3.2E-04	0.23	0.82	0.156**	4.35	2.4E-03	0.04	0.28	0.82	0.11	8	0.905**	1.6E-04	0.02	0.43	10
	1996	0.522	1.12	2.9E-01	-1.00	2.04	-0.028	-0.41	6.9E-01	-0.25	0.19	0.12	0.10	9	0.349	1.5E-01	0.13	0.07	11
	1997	-4.2E-04	-1E-03	9E-01	-1.09	1.09	0.267*	3.19	1.1E-02	-0.01	0.54	2E-07	0.16	9	-4.2E-04	5.0E-01	0.20	0.16	11
61	1998	0.403	0.71	5.0E-01	-1.45	2.26	0.063	0.73	4.8E-01	-0.22	0.34	0.05	0.25	9	0.230	2.5E-01	0.07	0.14	11
4	1999	0.744	1.22	2.5E-01	-1.24	2.72	0.073	0.51	6.2E-01	-0.39	0.54	0.14	0.23	9	0.377	1.3E-01	0.20	0.12	11
	2000	0.249	0.29	7.8E-01	-2.56	3.06	0.188	1.19	2.6E-01	-0.33	0.70	0.01	0.43	9	0.096	3.9E-01	0.11	0.16	11
	2001	-0.203	-0.77	4.6E-01	-1.06	0.66	-0.204*	-2.39	4.1E-02	-0.48	0.07	0.06	0.28	9	-0.248	2.3E-01	0.06	0.33	11
	2002	0.180	0.30	7.7E-01	-1.75	2.11	-0.417**	-3.62	5.5E-03	-0.79	-0.04	0.01	0.38	9	0.100	3.8E-01	-0.02	0.20	11
	2003	0.539	1.34	2.1E-01	-0.77	1.85	0.202*	2.75	2.2E-02	-0.04	0.44	0.17	0.20	9	0.408	1.1E-01	0.11	0.16	11
	2004	-0.161	-0.91	3.9E-01	-0.73	0.41	0.105**	3.33	8.8E-03	2E-03	0.21	0.08	0.10	9	-0.290	1.9E-01	0.06	0.18	11
	2005	0.685**	7.34	4.4E-05	0.38	0.99	0.105*	3.08	1.3E-02	-0.01	0.22	0.86	0.09	9	0.926**	2.2E-05	0.22	0.31	11
	2006	0.020	0.06	9.5E-01	-1.08	1.12	0.167*	3.20	1.1E-02	-3E-03	0.34	4E-04	0.11	9	0.020	4.8E-01	0.12	0.10	11
	2007	0.077	0.11	9.2E-01	-2.21	2.37	-0.040	-0.38	7.1E-01	-0.38	0.30	1E-03	0.30	9	0.036	4.6E-01	0.07	0.13	11
	2008	1.649**	5.16	6.0E-04	0.61	2.69	-0.138	-1.34	2.1E-01	-0.47	0.20	0.75	0.28	9	0.864**	3.0E-04	-0.18	0.28	11

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 11 observations per yearly regression from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

Analysis

Table U.2 exhibits similar results to Table U.1.

The Table U.2 cross-sectional regression slopes show that the change in the total equity per share variable exhibits varying levels of statistically significant explanatory power to the market price return variable for the samples and the control group.

The Primary and Secondary sample shows statistical significance for years 1995, 2005 and 2008. The Control group shows extensive statistical significance for the observation time range 1994 to 2008.

U.3 Regressions for Change in Total Assets and Market Price Return

The regression analysis for the change in total assets and market price return is presented for the time range 1994 to 2008 in Table U.3. In the regressions, the market price return variable is tested as the dependent variable, and the change in the total assets variable is tested as the independent variable.

Table U.3Cross-Sectional Regression Analysis for Change in Total Assets and
Market Price Return

Cross-Sectional Regression

Table U.3 Panel A to Panel D show the results for the cross-sectional regressions that test the change in total assets and market price return by applying the regression specified in equation (O.27). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.

Cross-Sectional Regression Table Description

The Table U.3 columns represent the following:

Year is the *panel data* year represented by the number of days from 1st January to 31st December, and also represents the 31st December variable record date for each year.

The Regression column presents the coefficients and related statistics for the regression specified in equation (O.27), and follows the model:

$dM_{i,t} = a + b \ dA_{i,t} + e_{i,t}$

Where: $dM_{i,t}$ is the log Market Price return $(M_{t-1,t})$ for the i^{th} firm at time t, $dA_{i,t}$ is the log change in Total Assets $(A_{t-1,t})$ for the i^{th} firm at time t, and $e_{N,t}$ is the regression error term. The *Lower* and *Upper* are the corresponding coefficient levels predicted at a 99% confidence level.

The Correlation column presents the correlation statistics for the relationship between the dependent and independent variables. The Descriptive Statistics column presents for the independent variable, the sample mean calculated using equation (O.17) and the sample standard deviation, *SD*, using equation (O.18). The *Obs*. column presents the number of sample firms observed for the year.

								,	Table U.3	(Conti	nued)								
		Cross	s-Section	nal Regree	ssion An	alysis for	the Primar	y and S	PA1 Secondary	NEL A 7 Sample	e Change	in Total	Asset	s (A _{t-1,t})	and Market	Price Retu	rn (<i>M_{t-1,t}</i>)		
							Regres	sion							Como	1	Descri	ptive	
	Year			Slope					Intercept				Model		Corre	lation	Statis	tics	Obs.
		b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
	1994	-0.023	-0.07	9.5E-01	-1.09	1.04	0.231	0.53	6.0E-01	-0.16	0.23	4E-04	0.24	12	-0.019	4.7E-01	0.02	0.19	14
	1995	0.283	1.22	2.5E-01	-0.43	0.99	0.405	1.89	8.3E-02	-0.10	0.41	0.11	0.25	12	0.331	1.2E-01	0.21	0.30	14
	1996	-0.223	-1.00	3.4E-01	-0.90	0.45	0.237*	2.78	1.6E-02	-0.01	0.24	0.07	0.13	13	-0.267	1.7E-01	0.10	0.16	15
	1997	-0.063	-0.21	8.4E-01	-0.99	0.86	0.443**	4.42	6.9E-04	0.08	0.44	3E-03	0.18	13	-0.057	4.2E-01	0.12	0.16	15
6	1998	0.637	1.32	2.1E-01	-0.82	2.09	0.215	-0.03	9.8E-01	-0.22	0.22	0.12	0.22	13	0.343	1.1E-01	0.09	0.12	15
17	1999	0.877	1.21	2.5E-01	-1.28	3.04	0.449	1.94	7.4E-02	-0.10	0.45	0.10	0.24	14	0.307	1.2E-01	0.09	0.09	16
	2000	-0.014	-0.06	9.5E-01	-0.67	0.65	0.465	1.86	8.4E-02	-0.11	0.47	3E-04	0.36	14	-0.016	4.8E-01	0.16	0.42	16
	2001	-0.047	-0.20	8.5E-01	-0.77	0.67	0.041*	-2.36	3.3E-02	-0.35	0.04	3E-03	0.26	14	-0.052	4.2E-01	0.06	0.28	16
	2002	-0.464	-0.97	3.5E-01	-1.88	0.95	-0.139**	-4.72	3.3E-04	-0.61	-0.14	0.06	0.32	14	-0.252	1.7E-01	-0.01	0.17	16
	2003	0.324	1.52	1.5E-01	-0.31	0.96	0.35**	4.73	3.2E-04	0.08	0.35	0.14	0.18	14	0.376	7.5E-02	0.05	0.21	16
	2004	-0.298	-1.61	1.3E-01	-0.85	0.25	0.195**	4.23	8.5E-04	0.03	0.20	0.16	0.08	14	-0.395	6.5E-02	0.10	0.11	16
	2005	0.381**	3.41	4.0E-03	0.05	0.71	0.244	1.63	1.3E-01	-0.07	0.24	0.45	0.16	14	0.673**	2.0E-03	0.31	0.37	16
	2006	-0.157	-1.24	2.4E-01	-0.53	0.22	0.254**	6.24	2.2E-05	0.09	0.25	0.10	0.09	14	-0.315	1.2E-01	0.11	0.19	16
	2007	-0.316	-1.22	2.4E-01	-1.09	0.45	0.192	-0.44	6.7E-01	-0.26	0.19	0.10	0.27	14	-0.310	1.2E-01	0.14	0.27	16
	2008	-0.617	-1.07	3.0E-01	-2.34	1.10	-0.112**	-3.67	2.5E-03	-1.08	-0.11	0.08	0.63	14	-0.275	1.5E-01	0.07	0.28	16

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 16 observations per yearly regression from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

								Table U.3	8 (Conti	nued)								
		C	G	.1.D		. .	Cart	PA	NEL B		A			.l 4 D	- 4 () (`		
		Cros	s-Section	ai Kegre	ssion Ana	Regree	ssion	ol Group	Cnange	e în Totai	Assets (.	$\mathbf{A}_{t-1,t}$	and Mai	rket Price R	aeturn (<i>M</i> _{t-1,}	t)		
Year			Slope					Intercept				Model		Corre	lation	Descri Statis	ptive tics	Obs.
	Ь	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	-0.203	-0.82	4.3E-01	-1.01	0.60	-0.091*	-2.41	3.9E-02	-0.21	0.03	0.07	0.12	9	-0.265	2.2E-01	0.01	0.16	11
1995	0.030	0.09	9.3E-01	-1.13	1.19	0.099	1.97	8.0E-02	-0.06	0.26	8E-04	0.07	9	0.028	4.7E-01	0.13	0.06	11
1996	0.360	1.78	1.1E-01	-0.28	1.00	-0.012	-0.24	8.2E-01	-0.17	0.15	0.24	0.16	10	0.490	5.3E-02	0.09	0.24	12
1997	0.72*	2.95	1.5E-02	-0.05	1.49	0.033	0.78	4.5E-01	-0.10	0.17	0.47	0.12	10	0.682**	7.3E-03	0.10	0.15	12
1998	1.062**	15.02	3.4E-08	0.84	1.29	-0.063**	-4.19	1.9E-03	-0.11	-0.02	0.96	0.05	10	0.979**	1.7E-08	-0.04	0.22	12
1999	1.118**	17.92	6.3E-09	0.92	1.32	0.036	1.38	2.0E-01	-0.05	0.12	0.97	0.07	10	0.985**	3.1E-09	0.29	0.32	12
2000	0.957**	8.90	4.6E-06	0.62	1.30	0.053	2.03	7.0E-02	-0.03	0.14	0.89	0.09	10	0.942**	2.3E-06	-0.04	0.25	12
2001	0.539*	2.65	2.5E-02	-0.11	1.19	-0.077	-1.74	1.1E-01	-0.22	0.06	0.41	0.07	10	0.642*	1.2E-02	-0.20	0.10	12
2002	1.482**	9.67	2.2E-06	1.00	1.97	0.100	2.22	5.1E-02	-0.04	0.24	0.90	0.06	10	0.95**	1.1E-06	-0.27	0.11	12
2003	1.006**	7.39	2.3E-05	0.57	1.44	0.049	1.62	1.4E-01	-0.05	0.14	0.85	0.05	10	0.919**	1.2E-05	0.19	0.12	12
2004	0.748**	4.63	9.4E-04	0.24	1.26	0.056*	2.76	2.0E-02	-0.01	0.12	0.68	0.05	10	0.826**	4.7E-04	0.09	0.09	12
2005	0.821**	5.44	2.8E-04	0.34	1.30	0.159**	4.34	1.5E-03	0.04	0.28	0.75	0.06	10	0.865**	1.4E-04	0.21	0.13	12
2006	0.778*	3.04	1.2E-02	-0.03	1.59	0.073	1.83	9.7E-02	-0.05	0.20	0.48	0.14	10	0.694**	6.2E-03	0.03	0.16	12
2007	0.863**	4.61	9.7E-04	0.27	1.46	0.072	1.67	1.3E-01	-0.07	0.21	0.68	0.14	10	0.825**	4.8E-04	-0.06	0.23	12
2008	0.702**	11.94	3.1E-07	0.52	0.89	-0.145**	-5.18	4.1E-04	-0.23	-0.06	0.93	0.05	10	0.967**	1.5E-07	-0.39	0.28	12

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The Control group consists of a maximum of 12 observations per yearly regression from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

							,	Table U.3	(Contin	nued)								
		~	~ .					PA	NEL C									
		Cros	s-Sectiona	l Regres	sion Ana	lysis for the	e Primai	ry Sample	e Chang	e in Total	Assets	$(A_{t-1,t})$	and Ma	rket Price I	Return (<i>M_t</i> .)	1, <i>t</i>)		
						Regre	ssion							Com	lation	Descri	ptive	
Year			Slope					Intercept				Model		Corre	elation	Statis	tics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	1.701	2.29	1.5E-01	-5.66	9.07	-0.162	-1.86	2.0E-01	-1.03	0.70	0.72	0.15	2	0.851	7.4E-02	0.06	0.12	4
1995	0.165	0.22	8.5E-01	-7.41	7.74	0.289	1.29	3.3E-01	-1.93	2.51	0.02	0.31	2	0.151	4.3E-01	0.21	0.23	4
1996	-1.884	-2.37	1.4E-01	-9.76	5.99	0.463*	4.49	4.6E-02	-0.56	1.49	0.74	0.06	2	-0.859	7.0E-02	0.12	0.05	4
1997	0.284	0.44	7.0E-01	-6.07	6.64	0.214	1.43	2.9E-01	-1.26	1.69	0.09	0.29	2	0.300	3.5E-01	0.04	0.26	4
1998	1.212	1.06	4.0E-01	-10.11	12.54	-0.051	-0.77	5.2E-01	-0.71	0.61	0.36	0.13	2	0.601	2.0E-01	0.01	0.07	4
1999	3.123	1.33	2.8E-01	-10.62	16.87	-0.056	-0.18	8.7E-01	-1.90	1.79	0.37	0.26	3	0.608	1.4E-01	0.12	0.05	5
2000	0.213	1.79	1.7E-01	-0.48	0.91	3.8E-04	0.01	9.96E-01	-0.42	0.42	0.52	0.11	3	0.718	8.6E-02	0.44	0.47	5
2001	3.130	2.27	1.1E-01	-4.91	11.17	-0.329	-2.39	9.6E-02	-1.13	0.47	0.63	0.10	3	0.795	5.4E-02	0.09	0.03	5
2002	-0.956	-0.79	4.9E-01	-8.00	6.08	-0.222	-2.21	1.1E-01	-0.81	0.36	0.17	0.18	3	-0.417	2.4E-01	0.05	0.08	5
2003	0.299	0.44	6.9E-01	-3.67	4.27	0.154	2.04	1.3E-01	-0.29	0.60	0.06	0.13	3	0.246	3.5E-01	0.07	0.10	5
2004	0.200	0.37	7.4E-01	-2.97	3.37	0.036	0.41	7.1E-01	-0.47	0.54	0.04	0.07	3	0.208	3.7E-01	0.15	0.06	5
2005	0.313	1.06	3.7E-01	-1.41	2.03	-0.025	-0.22	8.4E-01	-0.70	0.65	0.27	0.11	3	0.523	1.8E-01	0.35	0.20	5
2006	0.440	0.18	8.7E-01	-13.79	14.67	0.080	0.35	7.5E-01	-1.24	1.40	0.01	0.08	3	0.104	4.3E-01	0.09	0.02	5
2007	-0.404	-0.92	4.3E-01	-2.99	2.18	-0.055	-0.33	7.7E-01	-1.05	0.94	0.22	0.25	3	-0.467	2.1E-01	0.29	0.29	5
2008	2.401	1.32	2.8E-01	-8.22	13.02	-2.010	-2.63	7.8E-02	-6.47	2.45	0.37	0.63	3	0.606	1.4E-01	0.39	0.17	5

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels. The sample consists of a maximum of 5 observations per yearly regression from the 5 banks in the LSE that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

							,	Table U.3	(Conti	nued)								
		Cross-	Sectional	Regress	ion Analy	ysis for the S	Seconda	PA PA	NEL D le Chan	ge in Tota	al Assets	$(A_{t-1,t})$) and M	arket Price	Return (<i>M_t</i>	1,t)		
						Regres	sion									Descri	ntive	
Year			Slope					Intercept				Model		Corre	lation	Statis	tics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	-0.135	-0.36	7.3E-01	-1.39	1.12	0.070	0.91	3.9E-01	-0.19	0.33	0.02	0.24	8	-0.127	3.6E-01	2.8E-03	0.22	10
1995	0.299	1.21	2.6E-01	-0.53	1.13	0.108	1.17	2.8E-01	-0.20	0.42	0.16	0.24	8	0.393	1.3E-01	0.20	0.33	10
1996	-0.242	-1.43	1.9E-01	-0.79	0.31	0.064	1.89	9.1E-02	-0.05	0.17	0.19	0.10	9	-0.430	9.4E-02	0.10	0.18	11
1997	-0.826	-2.07	6.9E-02	-2.13	0.47	0.388**	5.44	4.1E-04	0.16	0.62	0.32	0.14	9	-0.567*	3.4E-02	0.15	0.11	11
1998	0.467	0.73	4.8E-01	-1.61	2.54	0.034	0.31	7.6E-01	-0.32	0.39	0.06	0.25	9	0.237	2.4E-01	0.12	0.12	11
1999	0.482	0.60	5.6E-01	-2.13	3.09	0.187	1.90	9.0E-02	-0.13	0.51	0.04	0.25	9	0.196	2.8E-01	0.08	0.10	11
2000	-0.048	-0.12	9.1E-01	-1.35	1.25	0.216	1.65	1.3E-01	-0.21	0.64	2E-03	0.43	9	-0.040	4.5E-01	0.03	0.34	11
2001	-0.088	-0.33	7.5E-01	-0.96	0.79	-0.213*	-2.45	3.7E-02	-0.50	0.07	0.01	0.29	9	-0.108	3.8E-01	0.05	0.34	11
2002	-0.567	-0.99	3.5E-01	-2.43	1.29	-0.439**	-3.96	3.3E-03	-0.80	-0.08	0.10	0.36	9	-0.314	1.7E-01	-0.03	0.20	11
2003	0.336	1.36	2.1E-01	-0.47	1.14	0.243**	3.99	3.2E-03	0.05	0.44	0.17	0.20	9	0.413	1.0E-01	0.05	0.25	11
2004	-0.336	-1.45	1.8E-01	-1.09	0.42	0.119**	3.68	5.1E-03	0.01	0.22	0.19	0.09	9	-0.435	9.1E-02	0.07	0.13	11
2005	0.409**	3.90	4.0E-03	0.07	0.75	0.14*	2.62	2.8E-02	-0.03	0.31	0.63	0.15	9	0.793**	2.0E-03	0.29	0.44	11
2006	-0.168	-1.23	2.5E-01	-0.61	0.28	0.19**	5.42	4.2E-04	0.08	0.31	0.14	0.10	9	-0.379	1.3E-01	0.12	0.24	11
2007	-0.177	-0.46	6.6E-01	-1.43	1.08	-0.022	-0.23	8.2E-01	-0.32	0.28	0.02	0.29	9	-0.151	3.3E-01	0.07	0.24	11
2008	0.010	0.01	9.9E-01	-3.26	3.28	-0.438*	-2.33	4.4E-02	-1.05	0.17	1E-05	0.56	9	0.003	5.0E-01	-0.08	0.18	11

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 11 observations per yearly regression from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

Analysis

The Table U.3 cross-sectional regression slopes show that the change in the total assets variable exhibits varying levels of statistically significant explanatory power to the market price return variable for the samples and the control group.

The Primary and Secondary sample only shows statistical significance for the year 2005. The Control group shows extensive statistical significance other than for the years: 1994, 1995 and 1996.

U.4 Regressions for Change in Total Liabilities and Market Price Return

The regression analysis for the change in total liabilities and market price return is presented for the time range 1994 to 2008 in Table U.4. In the regressions, the market price return variable is tested as the dependent variable, and the change in the total liabilities variable is tested as the independent variable.

Table U.4Cross-Sectional Regression Analysis for Change in Total Liabilities
and Market Price Return

Cross-Sectional Regression

Table U.4 Panel A to Panel D show the results for the cross-sectional regressions that test the change in total liabilities and market price return by applying the regression specified in equation (O.27). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.

Cross-Sectional Regression Table Description

The Table U.4 columns represent the following:

Year is the *panel data* year represented by the number of days from 1st January to 31st December, and also represents the 31st December variable record date for each year.

The Regression column presents the coefficients and related statistics for the regression specified in equation (O.27), and follows the model:

 $dM_{i,t} = a + b \ dL_{i,t} + e_{i,t}$

Where: $dM_{i,t}$ is the log Market Price return $(M_{t-1,t})$ for the i^{th} firm at time t, $dL_{i,t}$ is the log change in Total Liabilities $(E_{t-1,t})$ for the i^{th} firm at time t, and $e_{N,t}$ is the regression error term. The *Lower* and *Upper* are the corresponding coefficient levels predicted at a 99% confidence level.

The Correlation column presents the correlation statistics for the relationship between the dependent and independent variables. The Descriptive Statistics column presents for the independent variable, the sample mean calculated using equation (O.17) and the sample standard deviation, *SD*, using equation (O.18). The *Obs.* column presents the number of sample firms observed for the year.

							,	Table U.4	(Conti	nued)								
								PA	NEL A									
	Cross-S	Sectiona	I Regressi	ion Anal	ysis for tl	he Primary	and Sec	condary S	ample (Change in	Total L	iabili	ties (L_{t-1} ,	t) and Mark	et Price Ref	turn ($M_{t-1,t}$)	
						Regres	sion							C	1	Descri	ptive	
Year			Slope					Intercept				Model		Corre	lation	Statis	tics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	-0.183	-0.81	4.3E-01	-0.87	0.51	0.255	0.79	4.5E-01	-0.15	0.26	0.05	0.23	12	-0.228	2.2E-01	0.10	0.29	14
1995	0.128	0.90	3.9E-01	-0.31	0.56	0.414*	2.48	2.9E-02	-0.04	0.41	0.06	0.25	12	0.251	1.9E-01	0.22	0.50	14
1996	-0.156	-0.90	3.9E-01	-0.68	0.37	0.21*	2.82	1.5E-02	-0.01	0.21	0.06	0.13	13	-0.241	1.9E-01	0.07	0.20	15
1997	0.142	0.47	6.5E-01	-0.77	1.05	0.399**	4.90	2.9E-04	0.10	0.40	0.02	0.18	13	0.129	3.2E-01	0.06	0.16	15
1998	0.237	0.73	4.8E-01	-0.74	1.21	0.234	0.55	5.9E-01	-0.16	0.23	0.04	0.23	13	0.199	2.4E-01	0.09	0.19	15
1999	0.143	0.67	5.1E-01	-0.49	0.78	0.448**	4.10	1.1E-03	0.07	0.45	0.03	0.25	14	0.176	2.6E-01	-4E-03	0.31	16
2000	-0.066	-0.35	7.4E-01	-0.64	0.51	0.480	1.95	7.1E-02	-0.10	0.48	0.01	0.36	14	-0.092	3.7E-01	0.20	0.48	16
2001	-0.170	-0.59	5.6E-01	-1.02	0.68	0.082	-1.86	8.4E-02	-0.36	0.08	0.02	0.26	14	-0.156	2.8E-01	0.13	0.23	16
2002	-0.741	-1.59	1.3E-01	-2.13	0.65	-0.11**	-4.39	6.2E-04	-0.57	-0.11	0.15	0.30	14	-0.391	6.7E-02	0.04	0.17	16
2003	0.189	1.58	1.4E-01	-0.17	0.54	0.353**	5.00	2.0E-04	0.09	0.35	0.15	0.17	14	0.390	6.8E-02	0.06	0.38	16
2004	-0.054	-0.40	6.9E-01	-0.46	0.35	0.162**	3.71	2.3E-03	0.02	0.16	0.01	0.09	14	-0.107	3.5E-01	0.07	0.17	16
2005	0.321**	3.12	8.0E-03	0.01	0.63	0.255	1.61	1.3E-01	-0.08	0.26	0.41	0.17	14	0.64**	4.0E-03	0.36	0.42	16
2006	-0.104	-1.31	2.1E-01	-0.34	0.13	0.231**	6.71	1.0E-05	0.09	0.23	0.11	0.09	14	-0.330	1.1E-01	0.05	0.30	16
2007	-0.189	-0.93	3.7E-01	-0.79	0.42	0.192	-0.52	6.1E-01	-0.27	0.19	0.06	0.27	14	-0.241	1.8E-01	0.19	0.35	16
2008	-0.528	-0.80	4.4E-01	-2.50	1.45	-0.049**	-3.25	5.8E-03	-1.11	-0.05	0.04	0.64	14	-0.208	2.2E-01	0.11	0.25	16

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 16 observations per yearly regression from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

622

								Table U.4	l (Contin	nued)								
				_				PA	NEL B									
		Cross-	Sectional	Regress	ion Analy	ysis for the (Control	Group C	hange i	n Total Li	iabilities	$S(L_{t-1,t})$) and M	arket Price	Return (M	t-1,t)		
						Regres	ssion							G	1	Descri	ptive	
Year			Slope					Intercept				Model		Corre	lation	Statis	tics	Obs.
	Ь	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	0.089	1.48	1.7E-01	-0.11	0.28	-0.116*	-3.02	1.5E-02	-0.24	0.01	0.20	0.12	9	0.442	8.7E-02	0.27	0.61	11
1995	-0.028	-1.94	8.4E-02	-0.08	0.02	0.112**	5.88	2.4E-04	0.05	0.17	0.30	0.06	9	-0.544*	4.2E-02	0.31	1.34	11
1996	1.5E-04	6.9E-04	9E-01	-0.69	0.69	0.020	0.38	7.1E-01	-0.15	0.19	5E-08	0.18	10	2.2E-04	5.0E-01	0.03	0.25	12
1997	-0.017	-0.20	8.5E-01	-0.30	0.26	0.103	2.16	5.6E-02	-0.05	0.25	4E-03	0.16	10	-0.062	4.2E-01	0.03	0.56	12
1998	-0.385	-2.23	5.0E-02	-0.93	0.16	-0.023	-0.34	7.4E-01	-0.24	0.20	0.33	0.20	10	-0.576*	2.5E-02	0.21	0.35	12
1999	0.714*	2.58	2.7E-02	-0.16	1.59	0.181	1.68	1.2E-01	-0.16	0.52	0.40	0.29	10	0.632*	1.4E-02	0.24	0.32	12
2000	0.332*	2.53	3.0E-02	-0.08	0.75	-0.015	-0.24	8.1E-01	-0.21	0.18	0.39	0.21	10	0.624*	1.5E-02	0.09	0.48	12
2001	0.002	0.08	9.4E-01	-0.10	0.10	-0.182**	-7.20	2.9E-05	-0.26	-0.10	6E-04	0.09	10	0.024	4.7E-01	-0.20	0.84	12
2002	0.285	0.65	5.3E-01	-1.11	1.67	-0.288**	-4.78	7.4E-04	-0.48	-0.10	0.04	0.18	10	0.201	2.7E-01	-0.07	0.13	12
2003	0.257	1.10	3.0E-01	-0.48	1.00	0.243**	6.42	7.6E-05	0.12	0.36	0.11	0.13	10	0.328	1.5E-01	-0.02	0.17	12
2004	0.112	1.27	2.3E-01	-0.17	0.39	0.121**	5.17	4.2E-04	0.05	0.19	0.14	0.08	10	0.372	1.2E-01	0.03	0.27	12
2005	0.136	1.32	2.2E-01	-0.19	0.46	0.328**	9.72	2.1E-06	0.22	0.44	0.15	0.12	10	0.385	1.1E-01	0.02	0.34	12
2006	0.043	0.43	6.8E-01	-0.28	0.36	0.090	1.64	1.3E-01	-0.08	0.26	0.02	0.19	10	0.134	3.4E-01	0.11	0.56	12
2007	-0.216*	-2.67	2.4E-02	-0.47	0.04	-0.030	-0.51	6.2E-01	-0.22	0.16	0.42	0.19	10	-0.645*	1.2E-02	-0.23	0.72	12
2008	0.477**	3.27	9.0E-03	0.01	0.94	-0.381**	-8.58	6.4E-06	-0.52	-0.24	0.52	0.15	10	0.718**	4.0E-03	-0.08	0.31	12

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The Control group consists of a maximum of 12 observations per yearly regression from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

Table U.4 (Continued)																		
PANEL C Cross Sectional Degression Analysis for the Primary Samula Change in Total Liabilities (L_{-}) and Market Price Deturn (M_{-})																		
Cross-Sectional Regression Analysis for the Primary Sample Change in Total Liabilities $(L_{t-1,t})$ and Market Price Return $(M_{t-1,t})$																		
	Regression																ptive	
Year	Slope				Intercept					Model			Correlation		Statistics		Obs.	
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	1.612	2.41	1.4E-01	-5.03	8.26	-0.154	-1.87	2.0E-01	-0.97	0.66	0.74	0.14	2	0.862	6.9E-02	0.06	0.12	4
1995	0.156	0.20	8.6E-01	-7.48	7.79	0.292	1.31	3.2E-01	-1.92	2.50	0.02	0.31	2	0.142	4.3E-01	0.21	0.23	4
1996	-1.869	-2.36	1.4E-01	-9.74	6.00	0.457*	4.50	4.6E-02	-0.55	1.47	0.74	0.06	2	-0.858	7.1E-02	0.12	0.05	4
1997	0.294	0.46	6.9E-01	-6.10	6.69	0.212	1.42	2.9E-01	-1.27	1.69	0.09	0.29	2	0.307	3.5E-01	0.05	0.26	4
1998	1.088	0.99	4.3E-01	-9.79	11.96	-0.049	-0.73	5.4E-01	-0.72	0.62	0.33	0.13	2	0.575	2.1E-01	0.01	0.07	4
1999	2.719	1.15	3.3E-01	-11.08	16.51	0.005	0.02	9.9E-01	-1.81	1.82	0.31	0.27	3	0.554	1.7E-01	0.12	0.06	5
2000	0.214	1.74	1.8E-01	-0.51	0.93	0.002	0.03	9.8E-01	-0.43	0.43	0.50	0.11	3	0.708	9.0E-02	0.43	0.46	5
2001	3.100	2.35	1.0E-01	-4.61	10.81	-0.319	-2.47	9.0E-02	-1.07	0.44	0.65	0.09	3	0.805	5.0E-02	0.09	0.04	5
2002	-1.035	-0.90	4.4E-01	-7.77	5.69	-0.209	-2.04	1.3E-01	-0.81	0.39	0.21	0.18	3	-0.461	2.2E-01	0.05	0.08	5
2003	0.308	0.46	6.8E-01	-3.64	4.26	0.155	2.08	1.3E-01	-0.28	0.59	0.07	0.13	3	0.254	3.4E-01	0.07	0.10	5
2004	0.180	0.34	7.6E-01	-2.93	3.29	0.038	0.43	7.0E-01	-0.47	0.55	0.04	0.07	3	0.191	3.8E-01	0.15	0.06	5
2005	0.309	1.05	3.7E-01	-1.41	2.03	-0.027	-0.23	8.4E-01	-0.72	0.66	0.27	0.12	3	0.519	1.9E-01	0.36	0.20	5
2006	-0.191	-0.08	9.4E-01	-14.10	13.72	0.137	0.63	5.7E-01	-1.13	1.41	2E-03	0.08	3	-0.046	4.7E-01	0.09	0.02	5
2007	-0.401	-0.91	4.3E-01	-2.97	2.16	-0.055	-0.33	7.7E-01	-1.05	0.93	0.22	0.25	3	-0.466	2.1E-01	0.29	0.29	5
2008	2.484	1.37	2.6E-01	-8.07	13.04	-2.076	-2.66	7.7E-02	-6.64	2.49	0.39	0.62	3	0.622	1.3E-01	0.40	0.17	5

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels. The sample consists of a maximum of 5 observations per yearly regression from the 5 banks in the LSE that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.
								Table U.4	(Conti	nued)								
		Cross-Se	ectional R	egressio	n Analys	is for the Se	condar	PA y Sample	NEL D Change	e in Total	Liabiliti	ies (L_t	.1,t) and]	Market Price	e Return (M	I _{t-1,t})		
						Regres	sion									Descri	ntive	
Year			Slope					Intercept				Model		Corre	lation	Statis	tics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	-0.285	-1.29	2.4E-01	-1.03	0.46	0.103	1.37	2.1E-01	-0.15	0.36	0.17	0.22	8	-0.414	1.2E-01	0.12	0.34	10
1995	0.128	0.88	4.0E-01	-0.36	0.62	0.141	1.63	1.4E-01	-0.15	0.43	0.09	0.25	8	0.298	2.0E-01	0.22	0.58	10
1996	-0.218	-1.72	1.2E-01	-0.63	0.19	0.051	1.75	1.1E-01	-0.04	0.15	0.25	0.09	9	-0.497	6.0E-02	0.05	0.23	11
1997	-0.071	-0.17	8.7E-01	-1.41	1.27	0.272**	4.79	9.9E-04	0.09	0.46	3E-03	0.16	9	-0.057	4.3E-01	0.07	0.13	11
1998	0.134	0.35	7.4E-01	-1.12	1.39	0.076	0.84	4.2E-01	-0.22	0.37	0.01	0.25	9	0.115	3.7E-01	0.12	0.21	11
1999	0.077	0.35	7.4E-01	-0.65	0.80	0.23*	2.99	1.5E-02	-0.02	0.48	0.01	0.25	9	0.114	3.7E-01	-0.06	0.36	11
2000	-0.125	-0.44	6.7E-01	-1.06	0.81	0.226	1.72	1.2E-01	-0.20	0.65	0.02	0.43	9	-0.144	3.4E-01	0.10	0.47	11
2001	-0.149	-0.46	6.5E-01	-1.19	0.90	-0.195	-2.00	7.7E-02	-0.51	0.12	0.02	0.28	9	-0.153	3.3E-01	0.15	0.28	11
2002	-0.754	-1.37	2.1E-01	-2.55	1.04	-0.395**	-3.71	4.8E-03	-0.74	-0.05	0.17	0.35	9	-0.415	1.0E-01	0.03	0.20	11
2003	0.188	1.39	2.0E-01	-0.25	0.63	0.248**	4.13	2.5E-03	0.05	0.44	0.18	0.20	9	0.419	1.0E-01	0.06	0.46	11
2004	-0.038	-0.23	8.3E-01	-0.59	0.51	0.096*	3.07	1.3E-02	-0.01	0.20	0.01	0.10	9	-0.075	4.1E-01	0.03	0.19	11
2005	0.324*	3.08	1.3E-02	-0.02	0.67	0.143*	2.29	4.8E-02	-0.06	0.35	0.51	0.17	9	0.716**	7.0E-03	0.35	0.50	11
2006	-0.098	-1.11	3.0E-01	-0.39	0.19	0.173**	5.52	3.7E-04	0.07	0.28	0.12	0.10	9	-0.347	1.5E-01	0.04	0.37	11
2007	-0.101	-0.40	7.0E-01	-0.91	0.71	-0.019	-0.20	8.5E-01	-0.33	0.29	0.02	0.29	9	-0.133	3.5E-01	0.15	0.37	11
2008	0.669	0.55	6.0E-01	-3.30	4.63	-0.425*	-2.54	3.2E-02	-0.97	0.12	0.03	0.55	9	0.180	3.0E-01	-0.02	0.14	11

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 11 observations per yearly regression from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

Analysis

The Table U.4 cross-sectional regression slopes show that the change in the total liabilities variable exhibits varying levels of statistically significant explanatory power to the market price return variable for the samples and the control group.

The Primary and Secondary sample only shows statistical significance for the year 2005. The Control group shows statistical significance for the years: 1999, 2000, 2007 and 2008.

U.5 Regressions for Change in Net Income and Market Price Return

The regression analysis for the change in net income and market price return is presented for the time range 1994 to 2008 in Table U.5. In the regressions, the market price return variable is tested as the dependent variable, and the change in the net income variable is tested as the independent variable.

Table U.5Cross-Sectional Regression Analysis for Change in Net Income and
Market Price Return

Cross-Sectional Regression

Table U.5 Panel A to Panel D show the results for the cross-sectional regressions that test the change in net income and market price return by applying the regression specified in equation (O.27). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.

Cross-Sectional Regression Table Description

The Table U.5 columns represent the following:

Year is the *panel data* year represented by the number of days from 1st January to 31st December, and also represents the 31st December variable record date for each year.

The Regression column presents the coefficients and related statistics for the regression specified in equation (O.27), and follows the model:

$$dM_{i,t} = a + b \ dI_{i,t} + e_{i,t}$$

Where: $dM_{i,t}$ is the log Market Price return $(M_{t-1,t})$ for the i^{th} firm at time t, $dI_{i,t}$ is the change in Net Income $(E_{t-1,t})$ for the i^{th} firm at time t, and $e_{N,t}$ is the regression error term. The *Lower* and *Upper* are the corresponding coefficient levels predicted at a 99% confidence level.

The Correlation column presents the correlation statistics for the relationship between the dependent and independent variables. The Descriptive Statistics column presents for the independent variable, the sample mean calculated using equation (O.17) and the sample standard deviation, *SD*, using equation (O.18). The *Obs.* column presents the number of sample firms observed for the year.

									Table U.5	6 (Conti	nued)								
		Cros	ss-Sectio	onal Regre	ession A	nalysis fo	r the Prima	ry and	PA Secondar	NEL A y Samp ¹	le Change	e in Net l	[ncom	$e(I_{t-1,t})$	and Market	Price Retur	rn (<i>M</i> _{t-1,t})		
							Regres	sion							C	1-41-1	Descri	ptive	
	Year			Slope					Intercept				Model		Corre	lation	Statis	tics	Obs.
		b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
	1994	0.059	0.86	4.1E-01	-0.15	0.27	0.218	0.15	8.8E-01	-0.20	0.22	0.06	0.23	12	0.240	2.0E-01	0.39	0.93	14
	1995	-0.154*	-2.26	4.3E-02	-0.36	0.05	0.5**	4.26	1.1E-03	0.08	0.50	0.30	0.22	12	-0.547*	2.1E-02	0.51	0.90	14
	1996	-0.054	-0.47	6.5E-01	-0.40	0.29	0.266	2.10	5.6E-02	-0.05	0.27	0.02	0.13	13	-0.129	3.2E-01	0.34	0.31	15
	1997	1.3E-04	8.3E-04	9E-01	-0.48	0.48	0.4**	5.38	1.3E-04	0.11	0.40	5E-08	0.18	13	2.3E-04	5.0E-01	-0.03	0.30	15
6	1998	-0.008	-0.10	9.2E-01	-0.27	0.25	0.264	0.91	3.8E-01	-0.14	0.26	7E-04	0.23	13	-0.027	4.6E-01	0.37	0.71	15
27	1999	-0.008	-0.07	9.4E-01	-0.32	0.31	0.463**	3.83	1.8E-03	0.06	0.46	4E-04	0.26	14	-0.020	4.7E-01	0.21	0.63	16
	2000	0.125	1.15	2.7E-01	-0.20	0.45	0.416	1.77	9.9E-02	-0.11	0.42	0.09	0.34	14	0.295	1.3E-01	0.17	0.81	16
	2001	0.206	1.25	2.3E-01	-0.29	0.70	0.107	-1.51	1.5E-01	-0.33	0.11	0.10	0.25	14	0.316	1.2E-01	-0.24	0.38	16
	2002	0.073	0.87	4.0E-01	-0.18	0.32	-0.121**	-4.48	5.2E-04	-0.60	-0.12	0.05	0.32	14	0.226	2.0E-01	-0.13	0.98	16
	2003	0.007	1.76	1.0E-01	-4.6E-03	0.02	0.383**	5.68	5.6E-05	0.12	0.38	0.18	0.17	14	0.426	5.0E-02	-2.76	11.6	16
	2004	-0.010	-0.50	6.3E-01	-0.07	0.05	0.157**	3.88	1.7E-03	0.02	0.16	0.02	0.09	14	-0.131	3.1E-01	0.26	1.09	16
	2005	0.002	0.62	5.4E-01	-0.01	0.01	0.36**	3.52	3.4E-03	0.03	0.36	0.03	0.22	14	0.165	2.7E-01	4.81	20.6	16
	2006	0.017*	2.89	1.2E-02	-5.1E-04	0.03	0.197**	6.54	1.3E-05	0.07	0.20	0.37	0.08	14	0.611**	5.9E-03	1.13	3.53	16
	2007	0.036	0.99	3.4E-01	-0.07	0.15	0.114	-1.39	1.9E-01	-0.31	0.11	0.07	0.27	14	0.255	1.7E-01	0.61	1.90	16
	2008	0.177	2.13	5.1E-02	-0.07	0.42	0.003*	-2.96	1.0E-02	-0.96	2.6E-03	0.25	0.57	14	0.495*	2.6E-02	-0.90	1.78	16

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 16 observations per yearly regression from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

							,	Table U.5	6 (Conti	nued)								
		Cro	ss Santian	al Dogr	action An	alveis for th	o Cont	PA rol Crour	NEL B	a in Nat I	naomo (I) o	nd Mar	kat Driga Da	turn(M)			
			55-56CH01		cssion An	Regres	sion		Chang		ncome (I _{t-1,t}) a			turn (1/1 _{t-1,t})	Deseri	ntino	
Year			Slope					Intercept				Model		Corre	lation	Statis	tics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	0.180	1.84	9.8E-02	-0.14	0.50	-0.087*	-2.62	2.8E-02	-0.20	0.02	0.27	0.11	9	0.523*	4.9E-02	-0.03	0.36	11
1995	0.158	1.26	2.4E-01	-0.25	0.57	0.09**	3.97	3.2E-03	0.02	0.16	0.15	0.07	9	0.388	1.2E-01	0.08	0.17	11
1996	0.38**	3.64	4.5E-03	0.05	0.71	-0.016	-0.43	6.8E-01	-0.13	0.10	0.57	0.12	10	0.755**	2.3E-03	0.09	0.35	12
1997	-0.873*	-2.80	1.9E-02	-1.86	0.12	0.157**	3.86	3.2E-03	0.03	0.29	0.44	0.12	10	-0.662**	9.5E-03	0.06	0.12	12
1998	-0.287*	-2.33	4.2E-02	-0.68	0.10	-0.035	-0.53	6.1E-01	-0.24	0.17	0.35	0.20	10	-0.592*	2.1E-02	0.25	0.49	12
1999	-0.128**	-7.73	1.6E-05	-0.18	-0.08	0.239**	5.47	2.7E-04	0.10	0.38	0.86	0.14	10	-0.926**	7.9E-06	-0.90	2.60	12
2000	-0.195	-0.71	4.9E-01	-1.06	0.67	0.028	0.36	7.3E-01	-0.22	0.27	0.05	0.26	10	-0.220	2.5E-01	0.06	0.29	12
2001	-0.056	-1.68	1.2E-01	-0.16	0.05	-0.183**	-8.40	7.6E-06	-0.25	-0.11	0.22	0.08	10	-0.469	6.2E-02	1.4E-03	0.68	12
2002	-0.145	-0.75	4.7E-01	-0.76	0.47	-0.359**	-4.11	2.1E-03	-0.64	-0.08	0.05	0.18	10	-0.231	2.4E-01	-0.36	0.28	12
2003	0.486	1.11	2.9E-01	-0.90	1.87	0.188**	3.20	9.5E-03	1.9E-03	0.37	0.11	0.13	10	0.331	1.5E-01	0.10	0.09	12
2004	-0.034	-0.44	6.7E-01	-0.28	0.21	0.129**	4.71	8.3E-04	0.04	0.22	0.02	0.09	10	-0.138	3.3E-01	0.15	0.34	12
2005	-0.145	-0.66	5.2E-01	-0.84	0.55	0.346**	8.21	9.4E-06	0.21	0.48	0.04	0.12	10	-0.204	2.6E-01	0.10	0.17	12
2006	-0.332	-1.76	1.1E-01	-0.93	0.26	0.154*	2.64	2.5E-02	-0.03	0.34	0.24	0.16	10	-0.487	5.4E-02	0.18	0.26	12
2007	-0.457	-2.22	5.1E-02	-1.11	0.20	0.038	0.62	5.5E-01	-0.15	0.23	0.33	0.21	10	-0.574*	2.5E-02	0.04	0.30	12
2008	0.008	0.43	6.8E-01	-0.05	0.06	-0.43**	-6.69	5.4E-05	-0.63	-0.23	0.02	0.21	10	0.135	3.4E-01	1.16	3.59	12

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The Control group consists of a maximum of 12 observations per yearly regression from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

							,	Table U.5	6 (Contin	nued)								
		C	a	1.5				PA	NEL C			/ - \				<u>`</u>		
		Cros	ss-Section	al Regre	ssion Ana	Page	e Prima	ry Sampl	e Chang	ge in Net I	Income	(I _{t-1,t}) :	and Mar	ket Price R	leturn (<i>M_{t-1},</i>	<i>t</i>)		
Year			Slope			Kegi		Intercept				Model		Corre	elation	Descri Statis	ptive tics	Obs.
	Ь	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	0.093	0.77	5.2E-01	-1.10	1.28	-0.165	-0.88	4.7E-01	-2.03	1.69	0.23	0.25	2	0.480	2.6E-01	1.17	1.19	4
1995	0.292	0.66	5.8E-01	-4.08	4.67	0.215	0.99	4.3E-01	-1.95	2.37	0.18	0.28	2	0.424	2.9E-01	0.37	0.37	4
1996	-0.192	-0.11	9.2E-01	-17.85	17.46	0.278	0.62	6.0E-01	-4.20	4.75	0.01	0.12	2	-0.076	4.6E-01	0.25	0.04	4
1997	-0.506	-0.59	6.2E-01	-9.06	8.05	0.150	0.78	5.2E-01	-1.75	2.05	0.15	0.28	2	-0.384	3.1E-01	-0.15	0.19	4
1998	-0.257	-1.05	4.0E-01	-2.68	2.17	-0.026	-0.40	7.3E-01	-0.67	0.62	0.36	0.13	2	-0.597	2.0E-01	0.03	0.31	4
1999	0.122	0.18	8.7E-01	-3.82	4.06	0.316	1.80	1.7E-01	-0.71	1.34	0.01	0.32	3	0.104	4.3E-01	0.15	0.24	5
2000	0.142	0.77	5.0E-01	-0.94	1.23	0.016	0.13	9.0E-01	-0.69	0.72	0.16	0.15	3	0.404	2.5E-01	0.55	0.40	5
2001	0.440	1.77	1.8E-01	-1.02	1.90	-0.003	-0.07	9.5E-01	-0.31	0.30	0.51	0.11	3	0.714	8.8E-02	-0.07	0.22	5
2002	1.116	1.42	2.5E-01	-3.49	5.72	0.403	0.84	4.6E-01	-2.39	3.20	0.40	0.16	3	0.633	1.3E-01	-0.60	0.10	5
2003	-0.089	-1.54	2.2E-01	-0.43	0.25	0.231*	3.99	2.8E-02	-0.11	0.57	0.44	0.10	3	-0.664	1.1E-01	0.62	0.88	5
2004	1.3E-04	4.9E-03	9E-01	-0.16	0.16	0.066	1.92	1.5E-01	-0.13	0.27	8E-06	0.07	3	0.003	5.0E-01	0.57	1.25	5
2005	0.217	1.38	2.6E-01	-0.70	1.14	0.012	0.17	8.7E-01	-0.40	0.43	0.39	0.11	3	0.623	1.3E-01	0.34	0.34	5
2006	0.254	3.07	5.4E-02	-0.23	0.74	0.095*	4.80	1.7E-02	-0.02	0.21	0.76	0.04	3	0.871*	2.7E-02	0.10	0.24	5
2007	0.807	1.18	3.2E-01	-3.17	4.79	-0.242	-1.99	1.4E-01	-0.95	0.47	0.32	0.24	3	0.564	1.6E-01	0.09	0.17	5
2008	0.150	1.74	1.8E-01	-0.35	0.65	-0.823	-2.87	6.4E-02	-2.50	0.85	0.50	0.56	3	0.709	9.0E-02	-1.65	3.22	5

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 5 observations per yearly regression from the 5 banks in the LSE that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

							r	Table U.5	(Conti	nued)								
		_						PA	NEL D		_							
		Cross	-Sectional	Regres	sion Anal	ysis for the	Second	ary Samp	ole Char	ige in Net	Income	$e(I_{t-1,t})$	and Ma	rket Price F	Return (<i>M_{t-1},</i>	<i>t</i>)		
						Regres	sion							<i>c</i>		Descri	ntive	
Year			Slope					Intercept				Model		Corre	lation	Statis	tics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	0.181	1.64	1.4E-01	-0.19	0.55	0.055	0.81	4.4E-01	-0.17	0.28	0.25	0.21	8	0.501	7.0E-02	0.08	0.64	10
1995	-0.166*	-2.74	2.5E-02	-0.37	0.04	0.262**	3.78	5.4E-03	0.03	0.49	0.48	0.19	8	-0.696*	1.3E-02	0.56	1.05	10
1996	-0.006	-0.06	9.5E-01	-0.31	0.30	0.043	0.90	3.9E-01	-0.11	0.20	4E-04	0.11	9	-0.021	4.8E-01	0.37	0.37	11
1997	0.032	0.21	8.4E-01	-0.47	0.54	0.267**	5.40	4.3E-04	0.11	0.43	5E-03	0.16	9	0.068	4.2E-01	0.01	0.33	11
1998	-0.026	-0.25	8.1E-01	-0.36	0.31	0.105	1.14	2.8E-01	-0.19	0.40	0.01	0.25	9	-0.083	4.0E-01	0.49	0.78	11
1999	-0.008	-0.07	9.4E-01	-0.35	0.34	0.227*	2.84	2.0E-02	-0.03	0.49	6E-04	0.25	9	-0.024	4.7E-01	0.23	0.75	11
2000	0.167	1.20	2.6E-01	-0.29	0.62	0.214	1.77	1.1E-01	-0.18	0.61	0.14	0.40	9	0.371	1.3E-01	3E-03	0.91	11
2001	0.115	0.54	6.0E-01	-0.57	0.80	-0.181	-1.67	1.3E-01	-0.53	0.17	0.03	0.28	9	0.179	3.0E-01	-0.32	0.42	11
2002	0.107	1.07	3.1E-01	-0.22	0.43	-0.429**	-3.94	3.4E-03	-0.78	-0.08	0.11	0.36	9	0.335	1.6E-01	0.08	1.13	11
2003	0.008	1.89	9.2E-02	-0.01	0.02	0.292**	5.02	7.2E-04	0.10	0.48	0.28	0.18	9	0.532*	4.6E-02	-4.30	13.9	11
2004	-0.013	-0.42	6.9E-01	-0.11	0.09	0.097*	3.13	1.2E-02	-3.8E-03	0.20	0.02	0.10	9	-0.138	3.4E-01	0.12	1.04	11
2005	0.001	0.36	7.2E-01	-0.01	0.01	0.25**	3.36	8.4E-03	0.01	0.49	0.01	0.24	9	0.121	3.6E-01	6.84	25.0	11
2006	0.016*	2.43	3.8E-02	-0.01	0.04	0.145**	5.20	5.7E-04	0.05	0.24	0.40	0.09	9	0.63*	1.9E-02	1.60	4.23	11
2007	0.029	0.73	4.8E-01	-0.10	0.16	-0.059	-0.63	5.4E-01	-0.36	0.24	0.06	0.29	9	0.236	2.4E-01	0.85	2.28	11
2008	-0.093	-0.24	8.2E-01	-1.36	1.17	-0.491	-1.78	1.1E-01	-1.39	0.41	0.01	0.56	9	-0.079	4.1E-01	-0.56	0.45	11

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 11 observations per yearly regression from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

Analysis

The Table U.5 cross-sectional regression slopes show that the change in the net income variable exhibits varying levels of statistically significant explanatory power to the market price return variable for the samples and the control group.

The Primary and Secondary sample only shows statistical significance for the years 1995 and 2006. The Control group shows statistical significance for the years: 1996, 1997, 1998 and 1999.

APPENDIX V ANALYSIS AND RESULTS FOR ACCOUNTING TO MARKET PRICE RELATIVE DELTA AND VALUE-AT-RISK USING CROSS-SECTIONAL REGRESSIONS FOR SAMPLES AND CONTROL GROUP AND USING TIME SERIES REGRESSIONS FOR UK BANKS

This appendix presents analysis and results for the yearly *accounting to market price relative delta* and *Historical Value-at-Risk Actual* cross-sectional regressions. This appendix also presents analysis and results for the *accounting to market price relative delta* and *Historical Value-at-Risk Actual* time series regressions for the UK banks.

V.1 Regressions for Total Equity to Market Price Relative Delta and Historical Value-at-Risk Actual

The regression analysis for the total equity to market price relative delta and Historical Value-at-Risk Actual is presented for the time range 1994 to 2008 in Table V.1. In the regressions, the Historical Value-at-Risk Actual variable is tested as the dependent variable, and the total equity to market price relative delta variable is tested as the independent variable.

Table V.1Cross-Sectional Regression Analysis for Total Equity to MarketPrice Relative Delta and Historical Value-at-Risk Actual

Cross-Sectional Regression

Table V.1 Panel A to Panel D show the results for the cross-sectional regressions that test the total equity to market price relative delta and Historical Value-at-Risk Actual by applying the regression specified in equation (O.28). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.

Cross-Sectional Regression Table Description

The Table V.1 columns represent the following:

Year is the *panel data* year represented by the number of days from 1st January to 31st December, and also represents the 31st December variable record date for each year.

The Regression column presents the coefficients and related statistics for the regression specified in equation (O.28), and follows the model:

$$V_{i,t} = a + b (dE - dM)_{i,t} + e_{i,t}$$

Where: $V_{i,t}$ is the Value-at-Risk (V_t) for the i^{th} firm at time t, $(dE - dM)_{i,t}$ is the Total Equity to Market Price Relative Delta $(dE_{t-1,t} - dM_{t-1,t})$ for the i^{th} firm at time t, and $e_{i,t}$ is the regression error term. The *Lower* and *Upper* are the corresponding coefficient levels predicted at a 99% confidence level.

The Correlation column presents the correlation statistics for the relationship between the dependent and independent variables. The Descriptive Statistics column presents for the independent variable, the sample mean calculated using equation (O.17) and the sample standard deviation, SD, using equation (O.18). The *Obs.* column presents the number of sample firms observed for the year.

Table	V.1	(Continued)
	PA	NEL A

Cross-Sectional Regression Analysis for the Primary and Secondary Sample Total Equity to Market Price Relative Delta $(dE_{t-1,t} - dM_{t-1,t})$ and Historical Value-at-Risk Actual (V_t)

						Regres	sion							Correl	ation	Descri	ptive	
Year			Slope					Intercept				Model		Conci	ation	Statis	stics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	a	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	-0.787**	-4.87	3.8E-04	-1.28	-0.29	0.086*	2.41	3.3E-02	-0.02	0.20	0.66	0.13	12	-0.815**	1.9E-04	0.06	0.22	14
1995	-0.2**	-5.61	1.1E-04	-0.31	-0.09	-0.144**	-4.73	4.9E-04	-0.24	-0.05	0.72	0.11	12	-0.851**	5.7E-05	0.20	0.86	14
1996	-0.317	-1.27	2.3E-01	-1.07	0.44	0.040	0.81	4.4E-01	-0.11	0.19	0.11	0.17	13	-0.332	1.1E-01	0.08	0.19	15
1997	-0.199**	-4.40	7.2E-04	-0.33	-0.06	0.001	0.04	9.7E-01	-0.06	0.06	0.60	0.08	13	-0.773**	3.6E-04	-0.03	0.47	15
1998	-0.269	-1.00	3.4E-01	-1.08	0.54	-0.094	-1.36	2.0E-01	-0.30	0.11	0.07	0.27	13	-0.267	1.7E-01	0.04	0.26	15
1999	-0.224	-1.77	9.8E-02	-0.60	0.15	-0.126**	-3.78	2.0E-03	-0.23	-0.03	0.18	0.13	14	-0.428*	4.9E-02	-0.09	0.26	16
2000	-0.136*	-2.16	4.9E-02	-0.32	0.05	-0.102	-1.43	1.7E-01	-0.31	0.11	0.25	0.28	14	-0.499*	2.4E-02	0.24	1.14	16
2001	-0.185	-1.92	7.6E-02	-0.47	0.10	-0.251**	-5.14	1.5E-04	-0.40	-0.11	0.21	0.17	14	-0.456*	3.8E-02	0.24	0.46	16
2002	-0.598**	-6.92	7.1E-06	-0.86	-0.34	-0.202**	-4.70	3.4E-04	-0.33	-0.07	0.77	0.12	14	-0.88**	3.5E-06	0.35	0.36	16
2003	-0.135	-0.21	8.4E-01	-2.04	1.77	-0.534**	-3.69	2.4E-03	-0.96	-0.10	3E-03	0.49	14	-0.056	4.2E-01	-0.12	0.20	16
2004	0.632	2.00	6.5E-02	-0.31	1.57	-0.069	-1.27	2.2E-01	-0.23	0.09	0.22	0.21	14	0.471*	3.3E-02	-0.03	0.17	16
2005	0.014	0.24	8.1E-01	-0.15	0.18	0.013	0.58	5.7E-01	-0.05	0.08	4E-03	0.09	14	0.064	4.1E-01	0.07	0.39	16
2006	-0.279	-1.35	2.0E-01	-0.89	0.34	0.056	1.74	1.0E-01	-0.04	0.15	0.11	0.13	14	-0.338	1.0E-01	-0.03	0.16	16
2007	-0.536**	-8.78	4.6E-07	-0.72	-0.35	-0.018	-0.85	4.1E-01	-0.08	0.05	0.85	0.08	14	-0.92**	2.3E-07	0.18	0.32	16
2008	-0.713**	-15.1	4.6E-10	-0.85	-0.57	-0.243**	-4.72	3.3E-04	-0.40	-0.09	0.94	0.16	14	-0.971**	2.3E-10	0.68	0.88	16

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 16 observations per yearly regression from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

							Tal	ble V.1 (C	Continue	d)								
Cr	oss Sections	l Dograd	ssion Anal	vsis for	the Contr	ol Group T	otol Faui	PAN ity to Mar	EL B chot Priv	a Palativ	a Dalta I	(dF .	- dM - 1) and Histori	cal Valua a	t Dielz Ac	tual (V	7)
	JSS-Sectiona	i Kegi es		19515 101		Regre	ssion		KUTIN		C Della	(u D _t -1,t	- <i>awi</i> t-1,t) and mistori	cal valuc-a	Descri	ntive	<i>t</i>)
Year			Slope				I	Intercept				Model		Corre	lation	Statis	tics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	a	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	-0.039	-0.37	7.2E-01	-0.38	0.30	-0.095**	-3.56	6.1E-03	-0.18	-0.01	0.02	0.08	9	-0.123	3.6E-01	0.07	0.25	11
1995	0.944	1.07	3.1E-01	-1.92	3.80	-0.246**	-3.82	4.1E-03	-0.46	-0.04	0.11	0.12	9	0.337	1.6E-01	0.06	0.04	11
1996	-0.121	-0.78	4.5E-01	-0.62	0.37	-0.014	-0.30	7.7E-01	-0.16	0.13	0.06	0.14	10	-0.239	2.3E-01	0.14	0.28	12
1997	-2.482	-1.88	9.0E-02	-6.67	1.71	0.014	0.24	8.1E-01	-0.17	0.20	0.26	0.17	10	-0.511*	4.5E-02	0.03	0.04	12
1998	-2.502**	-6.28	9.1E-05	-3.76	-1.24	-0.075	-1.86	9.2E-02	-0.20	0.05	0.80	0.10	10	-0.893**	4.6E-05	0.07	0.08	12
1999	-0.017	-0.08	9.4E-01	-0.71	0.67	-0.171	-2.16	5.6E-02	-0.42	0.08	6E-04	0.26	10	-0.025	4.7E-01	0.09	0.37	12
2000	-1.578*	-2.85	1.7E-02	-3.33	0.18	-0.091	-1.57	1.5E-01	-0.28	0.09	0.45	0.18	10	-0.669**	8.6E-03	-0.05	0.10	12
2001	0.296	0.37	7.2E-01	-2.26	2.85	-0.299**	-7.19	3.0E-05	-0.43	-0.17	0.01	0.14	10	0.115	3.6E-01	-0.01	0.05	12
2002	-0.602*	-2.48	3.3E-02	-1.37	0.17	-0.362**	-17.64	7.3E-09	-0.43	-0.30	0.38	0.07	10	-0.617*	1.6E-02	0.03	0.08	12
2003	-0.059	-0.14	8.9E-01	-1.35	1.23	-0.454**	-14.68	4.3E-08	-0.55	-0.36	2E-03	0.09	10	-0.046	4.4E-01	0.04	0.07	12
2004	-1.253	-1.72	1.2E-01	-3.56	1.05	-0.013	-0.66	5.2E-01	-0.08	0.05	0.23	0.07	10	-0.478	5.8E-02	-5E-03	0.03	12
2005	-0.376	-1.25	2.4E-01	-1.33	0.58	0.039	1.53	1.6E-01	-0.04	0.12	0.13	0.07	10	-0.367	1.2E-01	-0.05	0.07	12
2006	-0.615	-1.65	1.3E-01	-1.79	0.56	0.024	0.50	6.3E-01	-0.13	0.17	0.21	0.15	10	-0.463	6.5E-02	-0.05	0.12	12
2007	-0.291	-0.75	4.7E-01	-1.52	0.94	-0.065	-1.30	2.2E-01	-0.22	0.09	0.05	0.16	10	-0.230	2.4E-01	-0.05	0.12	12
2008	-1.96**	-3.42	6.6E-03	-3.78	-0.14	-0.308**	-5.21	4.0E-04	-0.49	-0.12	0.54	0.13	10	-0.734**	3.3E-03	0.08	0.07	12

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The Control group consists of a maximum of 12 observations per yearly regression from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

							Tab	ole V.1 (C	ontinue	d)								
Cro	ss Soational	Dograss	ion Analy	sis for th	o Drimori	, Sampla Ta	tol Fau	PAN uity to Ma	EL C rkot Pr	ioo Dolatiy	vo Dolta	(dF	dM) and Histori	ool Voluo a	t Dielz Ao	tual (V	2
	ss-sectional	Regiess	Sion Analy	515 IUI UI	e i i iiiai y	Regress	sion				ve Della	(<i>uL_{t-1,}</i>	t - u l v1 t-1,t		cal value-a	Deceri		<i>t</i>)
Year			Slope					Intercept				Model		Correl	ation	Statis	tics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	-0.707**	-23.06	1.9E-03	-1.01	-0.40	0.112**	11.6	7.3E-03	0.02	0.21	1.00	0.02	2	-0.998**	9.4E-04	0.19	0.29	4
1995	-0.238	-1.23	3.4E-01	-2.15	1.68	-0.212	-3.17	8.7E-02	-0.87	0.45	0.43	0.13	2	-0.657	1.7E-01	0.04	0.40	4
1996	-0.057	-0.21	8.6E-01	-2.78	2.67	0.114	2.84	1.0E-01	-0.28	0.51	0.02	0.07	2	-0.144	4.3E-01	-0.07	0.15	4
1997	-0.397	-1.83	2.1E-01	-2.55	1.76	-0.036	-0.44	7.0E-01	-0.83	0.76	0.63	0.12	2	-0.791	1.0E-01	-0.23	0.33	4
1998	-0.200	-0.10	9.3E-01	-19.23	18.83	-0.359	-1.40	3.0E-01	-2.91	2.19	0.01	0.45	2	-0.073	4.6E-01	0.06	0.14	4
1999	-0.089	-0.26	8.1E-01	-2.07	1.89	-0.125	-1.68	1.9E-01	-0.56	0.31	0.02	0.14	3	-0.149	4.1E-01	-0.11	0.21	5
2000	-0.064	-1.23	3.0E-01	-0.37	0.24	-0.243	-2.47	9.0E-02	-0.82	0.33	0.34	0.18	3	-0.580	1.5E-01	1.09	1.76	5
2001	-1.146	-3.12	5.3E-02	-3.29	1.00	0.019	0.25	8.2E-01	-0.41	0.45	0.76	0.10	3	-0.874*	2.6E-02	0.16	0.14	5
2002	-0.346	-1.10	3.5E-01	-2.18	1.49	-0.252*	-3.26	4.7E-02	-0.70	0.20	0.29	0.10	3	-0.537	1.8E-01	0.21	0.15	5
2003	-0.764	-1.49	2.3E-01	-3.77	2.24	-0.429*	-4.37	2.2E-02	-1.00	0.14	0.42	0.21	3	-0.651	1.2E-01	-0.06	0.20	5
2004	-0.142	-0.28	8.0E-01	-3.14	2.85	-0.045	-1.30	2.8E-01	-0.25	0.16	0.02	0.08	3	-0.158	4.0E-01	1.9E-03	0.08	5
2005	0.048	0.39	7.2E-01	-0.67	0.76	-0.011	-0.60	5.9E-01	-0.12	0.10	0.05	0.03	3	0.220	3.6E-01	0.10	0.13	5
2006	0.302	0.95	4.1E-01	-1.55	2.16	0.015	0.81	4.8E-01	-0.09	0.12	0.23	0.04	3	0.481	2.1E-01	0.01	0.07	5
2007	-0.59**	-9.47	2.5E-03	-0.95	-0.23	-0.011	-0.40	7.2E-01	-0.17	0.15	0.97	0.03	3	-0.984**	1.2E-03	0.36	0.28	5
2008	-0.877*	-5.27	1.3E-02	-1.85	0.09	-0.011	-0.05	9.6E-01	-1.35	1.32	0.90	0.23	3	-0.95**	6.6E-03	1.22	0.70	5

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 5 observations per yearly regression from the 5 banks in the LSE that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

							Ta	ble V.1 (0	Continu	ed)								
Cross	s-Sectional F	Regressio	on Analysi	s for the	Seconda	ry Sample 7	Fotal Ec	PAN Juity to M	IEL D Iarket P	rice Rela	tive Delt	ta (<i>dE</i>	_{t-1,t} - dM	t-1,t) and Histe	orical Valu	e-at-Risk	Actual	(V_t)
						Regress	sion									Descri	ntive	
Year			Slope					Intercept				Model		Correl	ation	Statis	tics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	-0.961**	-3.44	8.8E-03	-1.90	-0.02	0.071	1.50	1.7E-01	-0.09	0.23	0.60	0.15	8	-0.773**	4.4E-03	0.01	0.18	10
1995	-0.205**	-5.99	3.3E-04	-0.32	-0.09	-0.115**	-3.42	9.1E-03	-0.23	-2.2E-03	0.82	0.10	8	-0.904**	1.6E-04	0.26	1.00	10
1996	-0.217	-0.59	5.7E-01	-1.42	0.99	0.006	0.07	9.4E-01	-0.25	0.26	0.04	0.20	9	-0.192	2.9E-01	0.14	0.17	11
1997	-0.168**	-4.34	1.9E-03	-0.29	-0.04	-0.004	-0.21	8.4E-01	-0.06	0.06	0.68	0.06	9	-0.823**	9.3E-04	0.04	0.50	11
1998	-0.231	-1.74	1.1E-01	-0.66	0.20	-0.001	-0.02	9.8E-01	-0.13	0.12	0.25	0.13	9	-0.503	5.7E-02	0.03	0.30	11
1999	-0.257	-1.77	1.1E-01	-0.73	0.22	-0.122*	-2.98	1.6E-02	-0.26	0.01	0.26	0.13	9	-0.508	5.5E-02	-0.07	0.28	11
2000	-0.407	-1.94	8.5E-02	-1.09	0.28	-0.111	-1.21	2.6E-01	-0.41	0.19	0.29	0.29	9	-0.542*	4.2E-02	-0.14	0.43	11
2001	-0.136	-1.59	1.5E-01	-0.41	0.14	-0.319**	-6.26	1.5E-04	-0.48	-0.15	0.22	0.15	9	-0.469	7.3E-02	0.28	0.55	11
2002	-0.611**	-5.78	2.7E-04	-0.95	-0.27	-0.198*	-3.25	1.0E-02	-0.40	2.3E-04	0.79	0.14	9	-0.887**	1.3E-04	0.42	0.41	11
2003	-0.034	-0.04	9.7E-01	-3.03	2.96	-0.584*	-2.64	2.7E-02	-1.30	0.14	1E-04	0.57	9	-0.012	4.9E-01	-0.15	0.20	11
2004	0.661	1.66	1.3E-01	-0.63	1.95	-0.077	-0.97	3.6E-01	-0.34	0.18	0.23	0.26	9	0.485	6.5E-02	-0.05	0.20	11
2005	0.014	0.21	8.4E-01	-0.21	0.24	0.022	0.70	5.0E-01	-0.08	0.12	5E-03	0.10	9	0.070	4.2E-01	0.05	0.47	11
2006	-0.280	-1.10	3.0E-01	-1.11	0.55	0.072	1.55	1.6E-01	-0.08	0.22	0.12	0.15	9	-0.344	1.5E-01	-0.04	0.19	11
2007	-0.502**	-5.48	3.9E-04	-0.80	-0.20	-0.016	-0.56	5.9E-01	-0.11	0.08	0.77	0.09	9	-0.877**	1.9E-04	0.09	0.31	11
2008	-0.687**	-15.46	8.7E-08	-0.83	-0.54	-0.269**	-6.47	1.2E-04	-0.40	-0.13	0.96	0.12	9	-0.982**	4.3E-08	0.44	0.87	11

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 11 observations per yearly regression from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

Analysis

The Table V.1 cross-sectional regression slopes show that the total equity to market price relative delta variable exhibits varying levels of statistically significant explanatory power to the Historical Value-at-Risk Actual variable for the samples and the control group.

The Primary and Secondary sample shows statistical significance for the years: 1994, 1995, 1997, 2000, 2002, 2007 and 2008. The Control group shows statistical significance for the years: 1998, 2000, 2002 and 2008.

V.2 Regressions for Total Assets to Market Price Relative Delta and Historical Value-at-Risk Actual

The regression analysis for the total assets to market price relative delta and Historical Value-at-Risk Actual is presented for the time range 1994 to 2008 in Table V.2. In the regressions, the Historical Value-at-Risk Actual variable is tested as the dependent variable, and the total assets to market price relative delta variable is tested as the independent variable.

Table V.2Cross-Sectional Regression Analysis for Total Assets to MarketPrice Relative Delta and Historical Value-at-Risk Actual

Cross-Sectional Regression

Table V.2 Panel A to Panel D show the results for the cross-sectional regressions that test the total assets to market price relative delta and Historical Value-at-Risk Actual by applying the regression specified in equation (O.28). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.

Cross-Sectional Regression Table Description

The Table V.2 columns represent the following:

Year is the *panel data* year represented by the number of days from 1st January to 31st December, and also represents the 31st December variable record date for each year.

The Regression column presents the coefficients and related statistics for the regression specified in equation (O.28), and follows the model:

$$V_{i,t} = a + b \left(dA - dM \right)_{i,t} + e_{i,t}$$

Where: $V_{i,t}$ is the Value-at-Risk (V_t) for the i^{th} firm at time t, $(dA - dM)_{i,t}$ is the Total Assets to Market Price Relative Delta $(dA_{t-1,t} - dM_{t-1,t})$ for the i^{th} firm at time t, and $e_{i,t}$ is the regression error term. The *Lower* and *Upper* are the corresponding coefficient levels predicted at a 99% confidence level.

The Correlation column presents the correlation statistics for the relationship between the dependent and independent variables. The Descriptive Statistics column presents for the independent variable, the sample mean calculated using equation (O.17) and the sample standard deviation, *SD*, using equation (O.18). The *Obs.* column presents the number of sample firms observed for the year.

T 1 1 1 1	r 🔨	(n	1)
I OBLO V		1 ontin	11001
таше у	- 4		սշա
		0.0	

Cross-Sectional Regression Analysis for the Primary and Secondary Sample Total Assets to Market Price Relative Delta $(dA_{t-1,t} - dM_{t-1,t})$ and Historical Value-at-Risk Actual (V_t)

PANEL A

						Regress	sion							Corrol	ation	Descri	ptive	
Year			Slope					Intercept				Model		Coller	ation	Statis	tics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	-0.544**	-4.16	1.3E-03	-0.94	-0.14	0.033	0.86	4.1E-01	-0.08	0.15	0.59	0.14	12	-0.769**	6.6E-04	-0.01	0.30	14
1995	-0.203	-1.17	2.6E-01	-0.73	0.33	-0.185**	-3.46	4.7E-03	-0.35	-0.02	0.10	0.20	12	-0.321	1.3E-01	-0.01	0.32	14
1996	-0.100	-0.47	6.5E-01	-0.75	0.55	0.015	0.32	7.6E-01	-0.13	0.16	0.02	0.18	13	-0.129	3.2E-01	0.01	0.23	15
1997	-0.232	-1.93	7.5E-02	-0.59	0.13	-0.025	-0.75	4.7E-01	-0.12	0.07	0.22	0.11	13	-0.473*	3.8E-02	-0.14	0.24	15
1998	-0.261	-0.77	4.6E-01	-1.28	0.76	-0.095	-1.34	2.0E-01	-0.31	0.12	0.04	0.27	13	-0.209	2.3E-01	0.04	0.21	15
1999	-0.259	-1.91	7.7E-02	-0.66	0.14	-0.15**	-3.91	1.6E-03	-0.26	-0.04	0.21	0.12	14	-0.455*	3.8E-02	-0.17	0.24	16
2000	-0.331*	-2.70	1.7E-02	-0.70	0.03	-0.141*	-2.17	4.8E-02	-0.34	0.05	0.34	0.26	14	-0.585**	8.7E-03	-0.02	0.55	16
2001	-0.253*	-2.27	4.0E-02	-0.59	0.08	-0.24**	-4.99	2.0E-04	-0.38	-0.10	0.27	0.16	14	-0.518*	2.0E-02	0.22	0.38	16
2002	-0.554**	-7.65	2.3E-06	-0.77	-0.34	-0.213**	-5.55	7.1E-05	-0.33	-0.10	0.81	0.11	14	-0.898**	1.2E-06	0.36	0.40	16
2003	0.049	0.09	9.3E-01	-1.63	1.73	-0.509**	-3.23	6.1E-03	-0.98	-0.04	5E-04	0.49	14	0.023	4.7E-01	-0.18	0.22	16
2004	0.318	0.89	3.9E-01	-0.74	1.38	-0.093	-1.60	1.3E-01	-0.27	0.08	0.05	0.23	14	0.232	1.9E-01	0.01	0.17	16
2005	-0.053	-0.68	5.1E-01	-0.28	0.18	0.019	0.85	4.1E-01	-0.05	0.09	0.03	0.08	14	-0.180	2.5E-01	0.10	0.28	16
2006	-0.168	-1.23	2.4E-01	-0.58	0.24	0.057	1.74	1.0E-01	-0.04	0.15	0.10	0.13	14	-0.311	1.2E-01	-0.04	0.24	16
2007	-0.37**	-6.37	1.7E-05	-0.54	-0.20	-0.033	-1.21	2.5E-01	-0.11	0.05	0.74	0.10	14	-0.862**	8.6E-06	0.21	0.43	16
2008	-0.777**	-8.51	6.6E-07	-1.05	-0.50	-0.184	-1.98	6.7E-02	-0.46	0.09	0.84	0.27	14	-0.915**	3.3E-07	0.70	0.76	16

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 16 observations per yearly regression from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

							Tab	le V.2 (C	ontinueo	l)								
C	oss-Section	al Roaros	sion Anal	vsis for	the Contr	ol Group T	ntal Ace	PAN ots to Mor	EL B ·kot Pric	a Rolativ	a Dolta 4	dA .	- dM - `) and Histori	cal Value a	t-Risk Ao	tual (V	2
	USS-Section	ai Regi es		y 515 101	the Contr	Regress	sion		KUTIK			(u/1 _t -1,t	- <i>ant</i> -1,t) and mistori	cal value-a	Descri		9
Year			Slope]	Intercept				Model		Correl	lation	Statis	tics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	-0.061	-0.52	6.2E-01	-0.45	0.32	-0.092**	-3.28	9.6E-03	-0.18	-7.1E-04	0.03	0.08	9	-0.171	3.1E-01	0.10	0.23	11
1995	-0.937**	-3.43	7.5E-03	-1.83	-0.05	-0.165**	-6.57	1.0E-04	-0.25	-0.08	0.57	0.08	9	-0.753**	3.8E-03	0.02	0.09	11
1996	-0.220	-1.13	2.9E-01	-0.84	0.40	-0.016	-0.37	7.2E-01	-0.15	0.12	0.11	0.14	10	-0.335	1.4E-01	0.07	0.22	12
1997	-0.517	-1.15	2.8E-01	-1.94	0.91	-0.052	-0.98	3.5E-01	-0.22	0.12	0.12	0.18	10	-0.341	1.4E-01	-0.01	0.12	12
1998	-2.752*	-2.56	2.8E-02	-6.16	0.66	-0.060	-0.68	5.1E-01	-0.34	0.22	0.40	0.18	10	-0.629*	1.4E-02	0.07	0.05	12
1999	-1.362	-1.35	2.1E-01	-4.57	1.84	-0.267*	-2.69	2.3E-02	-0.58	0.05	0.15	0.24	10	-0.392	1.0E-01	-0.07	0.07	12
2000	-0.737	-0.90	3.9E-01	-3.33	1.85	-0.057	-0.70	5.0E-01	-0.31	0.20	0.08	0.23	10	-0.274	1.9E-01	-0.06	0.09	12
2001	0.053	0.10	9.3E-01	-1.70	1.80	-0.302**	-7.33	2.5E-05	-0.43	-0.17	9E-04	0.14	10	0.030	4.6E-01	-0.01	0.08	12
2002	-0.456	-1.60	1.4E-01	-1.36	0.45	-0.368**	-15.85	2.1E-08	-0.44	-0.29	0.20	0.07	10	-0.451	7.1E-02	0.03	0.08	12
2003	-0.002	-3.5E-03	9E-01	-1.70	1.69	-0.456**	-12.17	2.6E-07	-0.57	-0.34	1E-06	0.09	10	-0.001	5.0E-01	-0.05	0.05	12
2004	-0.370	-0.87	4.1E-01	-1.72	0.99	-0.019	-0.76	4.6E-01	-0.10	0.06	0.07	0.07	10	-0.264	2.0E-01	-0.03	0.05	12
2005	0.079	0.22	8.3E-01	-1.08	1.24	0.066	1.34	2.1E-01	-0.09	0.22	5E-03	0.08	10	0.068	4.2E-01	-0.12	0.06	12
2006	-0.550	-1.62	1.4E-01	-1.62	0.52	0.018	0.36	7.2E-01	-0.14	0.17	0.21	0.15	10	-0.457	6.8E-02	-0.07	0.13	12
2007	-0.417	-1.29	2.3E-01	-1.44	0.61	-0.084	-1.65	1.3E-01	-0.24	0.08	0.14	0.15	10	-0.377	1.1E-01	-0.08	0.14	12
2008	0.651	1.19	2.6E-01	-1.08	2.38	-0.482**	-9.02	4.0E-06	-0.65	-0.31	0.12	0.18	10	0.353	1.3E-01	0.03	0.10	12

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The Control group consists of a maximum of 12 observations per yearly regression from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

641

_							Tab	ole V.2 (C	ontinue	d)								
Cro	oss-Sectional	Regress	ion Analy	sis for th	e Primar	v Samnle T	ntal Ass	PAN ets to Ma	EL C rket Pri	ce Relativ	ve Delta	(dA.,	. dM.	.) and Histor	ical Value-	at-Risk A	ctual ('	V)
		i itegi ess	ion muly	515 101 11	<u>e i i i i i i i i i i i i i i i i i i i</u>	Regress	ion			ce itelati	ve Denta	(42 11-1,	<i>t u</i> 1/1/-1,	<i>i)</i> und mistor		Desori	ntivo	(1)
Year			Slope					Intercept				Model		Corre	lation	Statis	tics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	-1.223	-2.37	1.4E-01	-6.35	3.90	0.126	1.41	2.9E-01	-0.76	1.01	0.74	0.13	2	-0.859	7.1E-02	0.12	0.15	4
1995	-0.201	-0.71	5.5E-01	-3.01	2.61	-0.245	-2.89	1.0E-01	-1.09	0.60	0.20	0.16	2	-0.449	2.8E-01	-0.11	0.32	4
1996	-0.288	-1.38	3.0E-01	-2.35	1.78	0.088	2.62	1.2E-01	-0.24	0.42	0.49	0.05	2	-0.699	1.5E-01	-0.11	0.14	4
1997	-0.320	-1.02	4.1E-01	-3.44	2.79	-0.002	-0.02	9.8E-01	-1.00	1.00	0.34	0.17	2	-0.585	2.1E-01	-0.18	0.30	4
1998	2.390	1.34	3.1E-01	-15.32	20.10	-0.487	-2.63	1.2E-01	-2.33	1.35	0.47	0.33	2	0.688	1.6E-01	0.05	0.11	4
1999	-0.322	-1.44	2.5E-01	-1.63	0.99	-0.183	-2.66	7.6E-02	-0.58	0.22	0.41	0.11	3	-0.638	1.2E-01	-0.21	0.25	5
2000	-0.378	-1.97	1.4E-01	-1.50	0.74	-0.182	-1.95	1.5E-01	-0.73	0.36	0.56	0.15	3	-0.751	7.2E-02	0.34	0.38	5
2001	-1.449*	-3.29	4.6E-02	-4.02	1.13	0.022	0.31	7.7E-01	-0.39	0.44	0.78	0.10	3	-0.885*	2.3E-02	0.13	0.11	5
2002	-0.337	-2.00	1.4E-01	-1.32	0.65	-0.217*	-3.48	4.0E-02	-0.58	0.15	0.57	0.07	3	-0.755	7.0E-02	0.31	0.22	5
2003	-0.712	-0.75	5.1E-01	-6.25	4.83	-0.457	-3.04	5.6E-02	-1.33	0.42	0.16	0.25	3	-0.398	2.5E-01	-0.11	0.13	5
2004	-0.173	-0.34	7.6E-01	-3.15	2.80	-0.031	-0.56	6.2E-01	-0.35	0.29	0.04	0.08	3	-0.193	3.8E-01	0.08	0.08	5
2005	-0.009	-0.09	9.3E-01	-0.58	0.56	-0.004	-0.14	9.0E-01	-0.18	0.17	3E-03	0.03	3	-0.051	4.7E-01	0.27	0.17	5
2006	-0.282	-0.98	4.0E-01	-1.97	1.40	0.009	0.46	6.8E-01	-0.11	0.13	0.24	0.04	3	-0.491	2.0E-01	-0.03	0.07	5
2007	-0.336*	-4.19	2.5E-02	-0.80	0.13	-0.072	-1.47	2.4E-01	-0.36	0.21	0.85	0.07	3	-0.924*	1.2E-02	0.46	0.46	5
2008	-1.079**	-12.77	1.0E-03	-1.57	-0.59	0.495*	3.77	3.3E-02	-0.27	1.26	0.98	0.10	3	-0.991**	5.2E-04	1.46	0.59	5

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 5 observations per yearly regression from the 5 banks in the LSE that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

							Ta	ble V.2 (0	Continu	ed)								
Cros	s-Sectional F	Regressio	on Analvsi	s for the	e Seconda	rv Sample '	Fotal As	PAN ssets to M	IEL D Iarket P	rice Relat	tive Delt	a (dA)	dM	and Histo	orical Value	e-at-Risk	Actual	(V.)
		8				Regress	ion						-1,11	-1,t)		Descri	ntive	(*1)
Year			Slope					Intercept				Model		Correl	ation	Statis	tics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	-0.511**	-3.47	8.5E-03	-1.01	-0.02	0.030	0.62	5.5E-01	-0.13	0.19	0.60	0.15	8	-0.775**	4.2E-03	-0.07	0.34	10
1995	-0.241	-1.05	3.3E-01	-1.02	0.53	-0.159	-2.21	5.8E-02	-0.40	0.08	0.12	0.23	8	-0.347	1.6E-01	0.04	0.33	10
1996	0.017	0.07	9.5E-01	-0.84	0.88	-0.025	-0.40	7.0E-01	-0.23	0.18	5E-04	0.20	9	0.022	4.7E-01	0.06	0.24	11
1997	-0.170	-1.26	2.4E-01	-0.61	0.27	-0.031	-0.91	3.9E-01	-0.14	0.08	0.15	0.10	9	-0.387	1.2E-01	-0.12	0.23	11
1998	-0.382*	-2.72	2.4E-02	-0.84	0.07	0.005	0.15	8.8E-01	-0.10	0.11	0.45	0.11	9	-0.671*	1.2E-02	0.03	0.24	11
1999	-0.243	-1.32	2.2E-01	-0.84	0.35	-0.138*	-2.78	2.1E-02	-0.30	0.02	0.16	0.14	9	-0.404	1.1E-01	-0.15	0.24	11
2000	-0.271	-1.51	1.6E-01	-0.85	0.31	-0.105	-1.06	3.2E-01	-0.42	0.22	0.20	0.31	9	-0.450	8.2E-02	-0.19	0.54	11
2001	-0.187	-1.89	9.1E-02	-0.51	0.13	-0.308**	-6.11	1.8E-04	-0.47	-0.14	0.28	0.14	9	-0.533*	4.6E-02	0.27	0.45	11
2002	-0.562**	-7.19	5.1E-05	-0.82	-0.31	-0.239**	-5.19	5.7E-04	-0.39	-0.09	0.85	0.12	9	-0.923**	2.6E-05	0.39	0.47	11
2003	0.023	0.03	9.8E-01	-2.31	2.36	-0.574*	-2.50	3.4E-02	-1.32	0.17	1E-04	0.57	9	0.011	4.9E-01	-0.21	0.25	11
2004	0.317	0.67	5.2E-01	-1.22	1.86	-0.103	-1.18	2.7E-01	-0.39	0.18	0.05	0.29	9	0.218	2.6E-01	-0.03	0.19	11
2005	-0.043	-0.39	7.0E-01	-0.40	0.31	0.024	0.77	4.6E-01	-0.08	0.12	0.02	0.10	9	-0.130	3.5E-01	0.03	0.29	11
2006	-0.161	-0.97	3.6E-01	-0.70	0.38	0.077	1.65	1.3E-01	-0.07	0.23	0.09	0.15	9	-0.308	1.8E-01	-0.05	0.29	11
2007	-0.363**	-4.06	2.8E-03	-0.65	-0.07	-0.023	-0.67	5.2E-01	-0.14	0.09	0.65	0.11	9	-0.804**	1.4E-03	0.10	0.40	11
2008	-1.059**	-12.52	5.4E-07	-1.33	-0.78	-0.191**	-3.53	6.4E-03	-0.37	-0.02	0.95	0.15	9	-0.972**	2.7E-07	0.36	0.56	11

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 11 observations per yearly regression from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

Analysis

The Table V.2 cross-sectional regression slopes show that the total assets to market price relative delta variable exhibits varying levels of statistically significant explanatory power to the Historical Value-at-Risk Actual variable for the samples and the control group.

The Primary and Secondary sample shows statistical significance for the years: 1994, 2000, 2001, 2002, 2007 and 2008. The Control group shows statistical significance for the years: 1995 and 1998.

V.3 Regressions for Total Liabilities to Market Price Relative Delta and Historical Value-at-Risk Actual

The regression analysis for the total liabilities to market price relative delta and Historical Value-at-Risk Actual is presented for the time range 1994 to 2008 in Table V.3. In the regressions, the Historical Value-at-Risk Actual variable is tested as the dependent variable, and the total liabilities to market price relative delta variable is tested as the independent variable.

Table V.3Cross-Sectional Regression Analysis for Total Liabilities to Market
Price Relative Delta and Historical Value-at-Risk Actual

Cross-Sectional Regression

Table V.3 Panel A to Panel D show the results for the cross-sectional regressions that test the total liabilities to market price relative delta and Historical Value-at-Risk Actual by applying the regression specified in equation (O.28). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.

Cross-Sectional Regression Table Description

The Table V.3 columns represent the following:

Year is the *panel data* year represented by the number of days from 1st January to 31st December, and also represents the 31st December variable record date for each year.

The Regression column presents the coefficients and related statistics for the regression specified in equation (O.28), and follows the model:

$$V_{i,t} = a + b \left(dL - dM \right)_{i,t} + e_{i,t}$$

Where: $V_{i,t}$ is the Value-at-Risk (V_t) for the i^{th} firm at time t, $(dL - dM)_{i,t}$ is the Total Liabilities to Market Price Relative Delta $(dL_{t-1,t} - dM_{t-1,t})$ for the i^{th} firm at time t, and $e_{i,t}$ is the regression error term. The *Lower* and *Upper* are the corresponding coefficient levels predicted at a 99% confidence level.

The Correlation column presents the correlation statistics for the relationship between the dependent and independent variables. The Descriptive Statistics column presents for the independent variable, the sample mean calculated using equation (O.17) and the sample standard deviation, *SD*, using equation (O.18). The *Obs.* column presents the number of sample firms observed for the year.

Table	V.3	(Continued)
	PAI	NEL A

Cross-Sectional Regression Analysis for the Primary and Secondary Sample Total Liabilities to Market Price Relative Delta $(dL_{t-1,t} - dM_{t-1,t})$ and Historical Value-at-Risk Actual (V_t)

						Regress	ion							Corrol	ation	Descri	ptive	
Year			Slope					Intercept				Model		Cone	auon	Statis	tics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	-0.404**	-4.19	1.3E-03	-0.70	-0.11	0.068	1.76	1.0E-01	-0.05	0.18	0.59	0.14	12	-0.771**	6.3E-04	0.07	0.41	14
1995	-0.070	-0.60	5.6E-01	-0.42	0.28	-0.183**	-3.29	6.4E-03	-0.35	-0.01	0.03	0.21	12	-0.171	2.8E-01	2.9E-03	0.50	14
1996	-0.110	-0.60	5.6E-01	-0.66	0.44	0.011	0.23	8.2E-01	-0.13	0.15	0.03	0.18	13	-0.163	2.8E-01	-0.02	0.27	15
1997	-0.177	-1.25	2.3E-01	-0.60	0.25	-0.027	-0.66	5.2E-01	-0.15	0.10	0.11	0.12	13	-0.329	1.2E-01	-0.19	0.22	15
1998	-0.162	-0.58	5.7E-01	-1.01	0.68	-0.099	-1.39	1.9E-01	-0.31	0.12	0.02	0.27	13	-0.158	2.9E-01	0.03	0.26	15
1999	-0.164	-1.82	9.1E-02	-0.43	0.11	-0.15**	-3.81	1.9E-03	-0.27	-0.03	0.19	0.13	14	-0.437*	4.5E-02	-0.26	0.36	16
2000	-0.238	-2.01	6.4E-02	-0.59	0.12	-0.129	-1.82	9.0E-02	-0.34	0.08	0.22	0.28	14	-0.473*	3.2E-02	0.02	0.62	16
2001	-0.26*	-2.21	4.4E-02	-0.61	0.09	-0.221**	-4.12	1.0E-03	-0.38	-0.06	0.26	0.17	14	-0.509*	2.2E-02	0.29	0.37	16
2002	-0.529**	-7.35	3.6E-06	-0.74	-0.32	-0.196**	-4.74	3.2E-04	-0.32	-0.07	0.79	0.12	14	-0.891**	1.8E-06	0.41	0.41	16
2003	0.168	0.47	6.5E-01	-0.89	1.23	-0.489**	-3.61	2.8E-03	-0.89	-0.09	0.02	0.48	14	0.125	3.2E-01	-0.17	0.35	16
2004	0.078	0.25	8.0E-01	-0.84	1.00	-0.089	-1.49	1.6E-01	-0.27	0.09	5E-03	0.24	14	0.068	4.0E-01	-0.02	0.20	16
2005	-0.022	-0.33	7.4E-01	-0.22	0.18	0.017	0.72	4.8E-01	-0.05	0.09	0.01	0.08	14	-0.089	3.7E-01	0.15	0.33	16
2006	-0.140	-1.51	1.5E-01	-0.42	0.14	0.049	1.51	1.5E-01	-0.05	0.15	0.14	0.12	14	-0.374	7.7E-02	-0.10	0.35	16
2007	-0.287**	-4.26	7.9E-04	-0.49	-0.09	-0.036	-0.97	3.5E-01	-0.14	0.07	0.56	0.13	14	-0.751**	4.0E-04	0.27	0.49	16
2008	-0.809**	-8.33	8.5E-07	-1.10	-0.52	-0.123	-1.23	2.4E-01	-0.42	0.17	0.83	0.27	14	-0.912**	4.2E-07	0.75	0.73	16

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 16 observations per yearly regression from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

646

Cro	ss-Sectional	Regressi	on Analys	is for th	e Control	Group Tot	al Liabil	PAN ities to M	EL B arket Pi	rice Relat	ive Delt:	a (dL.	dM.	and Histo	rical Value	-at-Risk A	octual ((V.)
			011111119			Regress	sion					. (<i>.,</i>	(,1) ***** 111000		Descri	ntive	
Year			Slope					Intercept				Model		Correl	ation	Statis	tics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	0.004	0.08	9.4E-01	-0.15	0.16	-0.099*	-3.21	1.1E-02	-0.20	1.4E-03	8E-04	0.09	9	0.028	4.7E-01	0.36	0.57	11
1995	-0.072**	-5.00	7.4E-04	-0.12	-0.03	-0.173**	-8.97	8.8E-06	-0.24	-0.11	0.74	0.06	9	-0.857**	3.7E-04	0.21	1.38	11
1996	-0.277*	-2.41	3.6E-02	-0.64	0.09	-0.029	-0.86	4.1E-01	-0.14	0.08	0.37	0.12	10	-0.607*	1.8E-02	0.01	0.31	12
1997	-0.061	-0.63	5.4E-01	-0.37	0.25	-0.053	-0.96	3.6E-01	-0.23	0.12	0.04	0.19	10	-0.196	2.7E-01	-0.08	0.59	12
1998	-0.281*	-2.85	1.7E-02	-0.59	0.03	-0.15*	-2.54	2.9E-02	-0.34	0.04	0.45	0.17	10	-0.669**	8.7E-03	0.32	0.53	12
1999	-0.354	-1.42	1.9E-01	-1.14	0.43	-0.212*	-2.82	1.8E-02	-0.45	0.03	0.17	0.24	10	-0.411	9.3E-02	-0.11	0.29	12
2000	0.056	0.29	7.8E-01	-0.56	0.67	-0.021	-0.29	7.8E-01	-0.25	0.21	0.01	0.24	10	0.091	3.9E-01	0.08	0.38	12
2001	-0.026	-0.52	6.2E-01	-0.18	0.13	-0.303**	-7.57	1.9E-05	-0.43	-0.18	0.03	0.14	10	-0.161	3.1E-01	-0.01	0.84	12
2002	-0.136	-1.14	2.8E-01	-0.52	0.24	-0.35**	-9.56	2.4E-06	-0.47	-0.23	0.11	0.08	10	-0.338	1.4E-01	0.24	0.20	12
2003	-0.052	-0.33	7.5E-01	-0.55	0.44	-0.469**	-9.76	2.0E-06	-0.62	-0.32	0.01	0.09	10	-0.104	3.7E-01	-0.26	0.18	12
2004	-0.002	-0.02	9.9E-01	-0.29	0.29	-0.007	-0.31	7.6E-01	-0.08	0.07	3E-05	0.08	10	-0.005	4.9E-01	-0.09	0.25	12
2005	0.049	0.67	5.2E-01	-0.18	0.28	0.072*	2.28	4.6E-02	-0.03	0.17	0.04	0.08	10	0.207	2.6E-01	-0.31	0.31	12
2006	-0.030	-0.34	7.4E-01	-0.31	0.25	0.055	1.13	2.9E-01	-0.10	0.21	0.01	0.17	10	-0.105	3.7E-01	0.01	0.56	12
2007	-0.090	-1.93	8.3E-02	-0.24	0.06	-0.073	-1.74	1.1E-01	-0.21	0.06	0.27	0.14	10	-0.521*	4.1E-02	-0.25	0.90	12
2008	0.035	0.13	9.0E-01	-0.82	0.89	-0.476**	-4.50	1.1E-03	-0.81	-0.14	2E-03	0.19	10	0.042	4.5E-01	0.34	0.21	12

Table V.3 (Continued)

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The Control group consists of a maximum of 12 observations per yearly regression from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

	~							PAN	EL C									
Cros	s-Sectional I	Regressio	n Analysis	s for the	Primary S	Sample Tot	al Liabi	lities to N	1arket F	Price Rela	tive Delt	$ta (dL_i)$	$_{t-1,t} - dM_t$.1,t) and Hist	orical Valu	e-at-Risk	Actual	(V_t)
						Regress	sion							~		Descri	ntive	
Year			Slope					Intercept				Model		Corre	lation	Statis	tics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	p(r)	Mean	SD	
1994	-1.251	-2.18	1.6E-01	-6.95	4.44	0.127	1.32	3.2E-01	-0.83	1.08	0.70	0.14	2	-0.839	8.1E-02	0.12	0.14	4
1995	-0.198	-0.70	5.6E-01	-3.01	2.61	-0.245	-2.88	1.0E-01	-1.09	0.60	0.20	0.16	2	-0.444	2.8E-01	-0.12	0.32	4
1996	-0.296	-1.47	2.8E-01	-2.30	1.71	0.086	2.63	1.2E-01	-0.24	0.41	0.52	0.05	2	-0.720	1.4E-01	-0.11	0.14	4
1997	-0.319	-1.00	4.2E-01	-3.49	2.85	-0.001	-0.01	9.9E-01	-1.00	1.00	0.33	0.17	2	-0.577	2.1E-01	-0.18	0.30	4
1998	2.450	1.45	2.8E-01	-14.29	19.19	-0.489	-2.76	1.1E-01	-2.25	1.27	0.51	0.32	2	0.717	1.4E-01	0.05	0.11	4
1999	-0.325	-1.48	2.3E-01	-1.61	0.96	-0.185	-2.71	7.3E-02	-0.58	0.21	0.42	0.11	3	-0.650	1.2E-01	-0.21	0.25	5
2000	-0.391	-2.05	1.3E-01	-1.51	0.73	-0.182	-2.01	1.4E-01	-0.71	0.35	0.58	0.14	3	-0.763	6.7E-02	0.33	0.38	5
2001	-1.441	-3.09	5.4E-02	-4.16	1.28	0.018	0.24	8.3E-01	-0.42	0.45	0.76	0.10	3	-0.872*	2.7E-02	0.12	0.11	5
2002	-0.334	-2.03	1.4E-01	-1.29	0.63	-0.215*	-3.47	4.0E-02	-0.58	0.15	0.58	0.07	3	-0.761	6.8E-02	0.32	0.22	5
2003	-0.696	-0.73	5.2E-01	-6.29	4.89	-0.457	-2.99	5.8E-02	-1.35	0.44	0.15	0.25	3	-0.387	2.6E-01	-0.11	0.13	5
2004	-0.171	-0.34	7.5E-01	-3.08	2.74	-0.030	-0.54	6.3E-01	-0.36	0.30	0.04	0.08	3	-0.195	3.8E-01	0.09	0.08	5
2005	-0.011	-0.11	9.2E-01	-0.58	0.55	-0.003	-0.11	9.2E-01	-0.18	0.17	4E-03	0.03	3	-0.066	4.6E-01	0.28	0.17	5
2006	-0.283	-1.03	3.8E-01	-1.89	1.32	0.009	0.43	6.9E-01	-0.11	0.12	0.26	0.04	3	-0.512	1.9E-01	-0.03	0.07	5
2007	-0.334*	-4.15	2.5E-02	-0.80	0.14	-0.072	-1.46	2.4E-01	-0.36	0.22	0.85	0.07	3	-0.923*	1.3E-02	0.46	0.46	5
2008	-1.085**	-13.43	8.9E-04	-1.56	-0.61	0.517*	4.09	2.6E-02	-0.22	1.26	0.98	0.10	3	-0.992**	4.5E-04	1.48	0.59	5

Table V.3 (Continued)

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 5 observations per yearly regression from the 5 banks in the LSE that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

							Ta	ble V.3 (C	Continue	ed)								
Cross-	Sectional R	egression	n Analysis	for the S	econdary	Sample To	tal Liab	PAN pilities to 1	EL D Market	Price Rel	ative De	lta (<i>d1</i>	$L_{t-1,t} - dN$	I _{t-1,t}) and His	torical Val	ue-at-Risk	: Actua	$l(V_t)$
						Regressi	ion									Descri	ntive	
Year			Slope					Intercept				Model		Correl	ation	Statis	tics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	-0.376**	-3.95	4.2E-03	-0.69	-0.06	0.082	1.88	9.7E-02	-0.06	0.23	0.66	0.14	8	-0.813**	2.1E-03	0.05	0.48	10
1995	-0.067	-0.47	6.5E-01	-0.54	0.41	-0.164	-2.18	6.1E-02	-0.42	0.09	0.03	0.24	8	-0.165	3.2E-01	0.05	0.56	10
1996	-0.049	-0.23	8.3E-01	-0.75	0.66	-0.024	-0.39	7.1E-01	-0.22	0.18	0.01	0.20	9	-0.075	4.1E-01	0.01	0.30	11
1997	-0.095	-0.58	5.7E-01	-0.63	0.44	-0.030	-0.65	5.3E-01	-0.18	0.12	0.04	0.11	9	-0.191	2.9E-01	-0.20	0.21	11
1998	-0.242	-1.84	9.9E-02	-0.67	0.19	1.3E-04	3E-03	9.97E-01	-0.12	0.12	0.27	0.13	9	-0.522*	5.0E-02	0.03	0.30	11
1999	-0.139	-1.28	2.3E-01	-0.49	0.21	-0.143*	-2.72	2.3E-02	-0.31	0.03	0.15	0.14	9	-0.393	1.2E-01	-0.29	0.41	11
2000	-0.166	-1.08	3.1E-01	-0.67	0.33	-0.074	-0.74	4.8E-01	-0.40	0.25	0.11	0.32	9	-0.338	1.5E-01	-0.12	0.67	11
2001	-0.164	-1.43	1.9E-01	-0.54	0.21	-0.298**	-4.80	9.8E-04	-0.50	-0.10	0.18	0.15	9	-0.430	9.3E-02	0.36	0.42	11
2002	-0.534**	-6.31	1.4E-04	-0.81	-0.26	-0.214**	-3.92	3.5E-03	-0.39	-0.04	0.82	0.13	9	-0.903**	7.0E-05	0.45	0.48	11
2003	0.168	0.39	7.1E-01	-1.23	1.57	-0.545*	-2.84	2.0E-02	-1.17	0.08	0.02	0.57	9	0.129	3.5E-01	-0.20	0.42	11
2004	0.034	0.08	9.4E-01	-1.32	1.39	-0.109	-1.18	2.7E-01	-0.41	0.19	7E-04	0.29	9	0.027	4.7E-01	-0.06	0.22	11
2005	-0.012	-0.14	8.9E-01	-0.30	0.27	0.024	0.74	4.8E-01	-0.08	0.13	2E-03	0.10	9	-0.046	4.5E-01	0.10	0.37	11
2006	-0.127	-1.12	2.9E-01	-0.50	0.24	0.067	1.41	1.9E-01	-0.09	0.22	0.12	0.15	9	-0.351	1.4E-01	-0.13	0.42	11
2007	-0.239*	-2.66	2.6E-02	-0.53	0.05	-0.018	-0.40	7.0E-01	-0.17	0.13	0.44	0.14	9	-0.663*	1.3E-02	0.18	0.50	11
2008	-1.12**	-11.01	1.6E-06	-1.45	-0.79	-0.100	-1.50	1.7E-01	-0.32	0.12	0.93	0.17	9	-0.965**	8.0E-07	0.42	0.52	11

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 11 observations per yearly regression from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

Analysis

The Table V.3 cross-sectional regression slopes show that the total liabilities to market price relative delta variable exhibits varying levels of statistically significant explanatory power to the Historical Value-at-Risk Actual variable for the samples and the control group.

The Primary and Secondary sample shows statistical significance for the years: 1994, 2000, 2002, 2007 and 2008. The Control group shows statistical significance for the years: 1995, 1996 and 1998.

V.4 Regressions for Net income to Market Price Relative Delta and Historical Value-at-Risk Actual

The regression analysis for the net income to market price relative delta and Historical Value-at-Risk Actual is presented for the time range 1994 to 2008 in Table V.4. In the regressions, the Historical Value-at-Risk Actual variable is tested as the dependent variable, and the net income to market price relative delta variable is tested as the independent variable.

Table V.4Cross-Sectional Regression Analysis for Net Income to Market Price
Relative Delta and Historical Value-at-Risk Actual

Cross-Sectional Regression

Table V.4 Panel A to Panel D show the results for the cross-sectional regressions that test the net income to market price relative delta and Historical Value-at-Risk Actual by applying the regression specified in equation (O.28). In this table, Panel A presents results for the Primary and Secondary sample; Panel B presents results for the Control group; Panel C presents results for the Primary sample; and, Panel D presents results for the Secondary sample.

Cross-Sectional Regression Table Description

The Table V.4 columns represent the following:

Year is the *panel data* year represented by the number of days from 1st January to 31st December, and also represents the 31st December variable record date for each year. The Regression column presents the coefficients and related statistics for the regression specified in equation (O.28), and follows the model:

$V_{i,t} = a + b (dI - dM)_{i,t} + e_{i,t}$

Where: $V_{i,t}$ is the Value-at-Risk (V_t) for the i^{th} firm at time t, $(dI - dM)_{i,t}$ is the Net Income to Market Price Relative Delta $(dI_{t-1,t} - dM_{t-1,t})$ for the i^{th} firm at time t, and $e_{i,t}$ is the regression error term. The *Lower* and *Upper* are the corresponding coefficient levels predicted at a 99% confidence level.

The Correlation column presents the correlation statistics for the relationship between the dependent and independent variables. The Descriptive Statistics column presents for the independent variable, the sample mean calculated using equation (O.17) and the sample standard deviation, *SD*, using equation (O.18). The *Obs.* column presents the number of sample firms observed for the year.

Table V.4 (Continued)

Cross-Sectional Regression Analysis for the Primary and Secondary Sample Net Income to Market Price Relative Delta $(dI_{t-1,t} - dM_{t-1,t})$ and Historical Value-at-Risk Actual (V_t)

PANEL A

						Regress	ion							Correl	lation	Descri	ptive	
Year			Slope					Intercept				Model		Cone	auon	Statis	tics	Obs.
	Ь	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	0.019	0.28	7.9E - 01	-0.19	0.23	0.033	0.52	6.1E-01	- 0.16	0.23	0.01	0.22	12	0.080	3.9E-01	0.36	0.91	14
1995	-0.13**	-3.20	7.6E-03	-0.25	-0.01	-0.145**	-3.36	5.7E-03	-0.28	-0.01	0.46	0.16	12	-0.679**	3.8E-03	0.29	1.06	14
1996	-0.089	-0.65	5.3E-01	-0.50	0.32	0.036	0.61	5.5E-01	-0.14	0.21	0.03	0.18	13	-0.176	2.7E-01	0.25	0.36	15
1997	-0.035	-0.37	7.1E-01	-0.32	0.25	-0.003	-0.07	9.4E-01	-0.13	0.12	0.01	0.12	13	-0.103	3.6E-01	-0.29	0.35	15
1998	0.063	0.65	5.3E-01	-0.23	0.35	-0.124	-1.62	1.3E-01	-0.35	0.11	0.03	0.27	13	0.177	2.6E-01	0.31	0.75	15
1999	-0.053	-1.04	3.2E-01	-0.21	0.10	-0.11**	-3.25	5.8E-03	-0.21	-0.01	0.07	0.13	14	-0.268	1.6E-01	-0.05	0.68	16
2000	-0.144	-1.46	1.7E-01	-0.44	0.15	-0.136	-1.81	9.1E-02	-0.36	0.09	0.13	0.30	14	-0.364	8.3E-02	-4E-03	0.78	16
2001	0.001	0.01	9.96E-01	-0.38	0.38	-0.296**	-6.01	3.2E-05	-0.44	-0.15	2E-06	0.19	14	0.001	5.0E-01	-0.08	0.39	16
2002	-0.016	-0.24	8.1E-01	-0.22	0.19	-0.41**	-6.26	2.1E-05	-0.61	-0.22	4E-03	0.25	14	-0.064	4.1E-01	0.24	0.96	16
2003	0.036**	6.87	7.7E-06	0.02	0.05	-0.411**	-6.82	8.3E-06	-0.59	-0.23	0.77	0.23	14	0.878**	3.9E-06	-2.99	11.5	16
2004	0.001	0.02	9.9E-01	-0.17	0.17	-0.090	-1.49	1.6E-01	-0.27	0.09	2E-05	0.24	14	0.005	4.9E-01	0.17	1.11	16
2005	0.002	1.86	8.4E-02	-1.1E-03	4.6E-03	0.005	0.27	7.9E-01	-0.05	0.06	0.20	0.08	14	0.445*	4.2E-02	4.61	20.6	16
2006	0.017	1.89	8.0E-02	-0.01	0.04	0.047	1.50	1.6E-01	-0.05	0.14	0.20	0.12	14	0.451*	4.0E-02	0.98	3.47	16
2007	-3.3E-04	-0.01	9.9E-01	-0.08	0.08	-0.112*	-2.17	4.8E-02	-0.27	0.04	1E-05	0.19	14	-0.003	5.0E-01	0.69	1.85	16
2008	0.037	0.33	7.4E-01	-0.29	0.36	-0.72**	-4.25	8.1E-04	-1.22	-0.22	0.01	0.67	14	0.089	3.7E-01	-0.27	1.57	16

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 16 observations per yearly regression from 16 firms in the LSE consisting of the 5 banks and 11 banking related firms that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

							Tab	le V.4 (C	ontinue	d)								
	mass Santian	al Dogra	ssion Ano	lucis for	the Cont	nal Chaup N	lat Incon	PAN no to Mor	EL B	o Dolotiv	o Dolto ((41		and Historia	al Valua at	- Dick A of	nol (V	<u> </u>
	1088-Section	ai Kegie	SSIUII AIIA	1ys1s 101	the Cont	Regres	sion		KetTIN		e Della	(<i>u</i> 1 _{<i>t</i>-1,<i>t</i>} -	• <i>uwi</i> t-1,t)		al value-al	Descri	ntive	,
Year			Slope]	Intercept				Model		Correl	lation	Statis	tics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	-0.119	-1.52	1.6E-01	-0.37	0.13	-0.09**	-3.84	4.0E-03	-0.17	-0.01	0.20	0.08	9	-0.453	8.1E-02	0.06	0.31	11
1995	-0.104	-0.42	6.8E-01	-0.90	0.69	-0.19**	-5.14	6.1E-04	-0.31	-0.07	0.02	0.12	9	-0.139	3.4E-01	-0.02	0.16	11
1996	0.152	0.86	4.1E-01	-0.41	0.71	-0.042	-0.97	3.5E-01	-0.18	0.10	0.07	0.14	10	0.263	2.0E-01	0.07	0.24	12
1997	-0.616**	-4.90	6.3E-04	-1.02	-0.22	-0.074*	-2.38	3.9E-02	-0.17	0.02	0.71	0.11	10	-0.84**	3.1E-04	-0.04	0.25	12
1998	-0.322**	-10.20	1.3E-06	-0.42	-0.22	-0.126**	-5.52	2.5E-04	-0.20	-0.05	0.91	0.07	10	-0.955**	6.6E-07	0.35	0.66	12
1999	-0.005	-0.17	8.7E-01	-0.09	0.08	-0.178	-2.13	5.9E-02	-0.44	0.09	3E-03	0.26	10	-0.054	4.3E-01	-1.26	2.93	12
2000	-0.369*	-2.91	1.6E-02	-0.77	0.03	0.001	0.02	9.8E-01	-0.16	0.17	0.46	0.18	10	-0.677**	7.8E-03	0.05	0.43	12
2001	-0.019	-0.32	7.5E-01	-0.20	0.17	-0.299**	-7.17	3.0E-05	-0.43	-0.17	0.01	0.14	10	-0.102	3.8E-01	0.18	0.73	12
2002	-0.099	-1.63	1.3E-01	-0.29	0.09	-0.388**	-18.04	5.9E-09	-0.46	-0.32	0.21	0.07	10	-0.458	6.7E-02	-0.05	0.37	12
2003	-0.084	-0.41	6.9E-01	-0.74	0.57	-0.467**	-12.19	2.5E-07	-0.59	-0.35	0.02	0.09	10	-0.127	3.5E-01	-0.13	0.13	12
2004	-0.078	-1.32	2.2E-01	-0.26	0.11	-0.005	-0.24	8.2E-01	-0.07	0.06	0.15	0.07	10	-0.386	1.1E-01	0.03	0.36	12
2005	0.132	1.38	2.0E-01	-0.17	0.44	0.087*	2.89	1.6E-02	-0.01	0.18	0.16	0.07	10	0.401	9.8E-02	-0.23	0.23	12
2006	-0.339**	-4.34	1.5E-03	-0.59	-0.09	0.084*	2.84	1.7E-02	-0.01	0.18	0.65	0.10	10	-0.808**	7.3E-04	0.09	0.38	12
2007	-0.301**	-8.81	5.0E-06	-0.41	-0.19	-0.045*	-2.80	1.9E-02	-0.10	0.01	0.89	0.06	10	-0.941**	2.5E-06	0.02	0.49	12
2008	-0.016	-1.03	3.3E-01	-0.06	0.03	-0.439**	-7.64	1.7E-05	-0.62	-0.26	0.10	0.18	10	-0.309	1.6E-01	1.59	3.57	12

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The Control group consists of a maximum of 12 observations per yearly regression from 12 banking related firms in the LSE that did not adopt the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

_							Tab	ole V.4 (C	ontinue	d)								
Cr	ass Section	al Dogras	sion Analy	rsis for th	na Primar	w Samnla N	ot Inco	PAN mo to Mo	EL C rkot Pri	ca Palativ	va Dalta	(d1 .	- dM - S	and Histor	ical Valua a	of Diely Ac	stual (L	7)
	USS-Section?	a Regies	sion r mary	<u>- 515 101 ti</u>		Regressi	on				ve Dena	(u1 _{t-1,t}	- uwi _{t-1,t}		icai vaiuc-a	Dogori	ntivo	<i>t)</i>
Year			Slope					Intercept				Model		Corre	lation	Statis	tics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	0.041	0.31	7.8E-01	-1.25	1.33	-0.069	-0.34	7.6E-01	-2.08	1.94	0.05	0.25	2	0.217	3.9E-01	1.23	1.10	4
1995	-0.282	-1.35	3.1E-01	-2.35	1.79	-0.208	-3.24	8.4E-02	-0.85	0.43	0.48	0.13	2	-0.691	1.5E-01	0.05	0.35	4
1996	-0.213	-0.63	6.0E-01	-3.59	3.16	0.123	3.70	6.6E-02	-0.21	0.45	0.16	0.06	2	-0.405	3.0E-01	0.02	0.11	4
1997	-0.383	-2.23	1.6E-01	-2.09	1.32	-0.087	-1.03	4.1E-01	-0.93	0.75	0.71	0.11	2	-0.844	7.8E-02	-0.38	0.37	4
1998	0.725	1.78	2.2E-01	-3.32	4.77	-0.421	-2.94	9.9E-02	-1.84	1.00	0.61	0.28	2	0.783	1.1E-01	0.07	0.40	4
1999	-0.201	-1.15	3.3E-01	-1.22	0.82	-0.153	-2.42	9.5E-02	-0.52	0.22	0.31	0.12	3	-0.554	1.7E-01	-0.18	0.35	5
2000	-0.076	-0.25	8.2E-01	-1.84	1.69	-0.278	-1.65	2.0E-01	-1.26	0.71	0.02	0.22	3	-0.143	4.1E-01	0.45	0.36	5
2001	0.493	0.81	4.8E-01	-3.05	4.04	-0.146	-1.65	2.0E-01	-0.66	0.37	0.18	0.19	3	0.424	2.4E-01	-0.03	0.16	5
2002	-0.648*	-3.70	3.4E-02	-1.67	0.37	-0.539**	-8.66	3.2E-03	-0.90	-0.18	0.82	0.05	3	-0.906*	1.7E-02	-0.33	0.14	5
2003	-0.224*	-3.68	3.5E-02	-0.58	0.13	-0.282*	-4.79	1.7E-02	-0.63	0.06	0.82	0.12	3	-0.905*	1.7E-02	0.45	0.96	5
2004	-0.008	-0.25	8.2E-01	-0.19	0.17	-0.042	-1.09	3.6E-01	-0.26	0.18	0.02	0.08	3	-0.142	4.1E-01	0.50	1.25	5
2005	-0.024	-0.42	7.0E-01	-0.36	0.31	-4.5E-04	-0.02	9.8E-01	-0.12	0.12	0.06	0.03	3	-0.237	3.5E-01	0.25	0.28	5
2006	-0.035	-0.28	8.0E-01	-0.77	0.70	0.016	0.79	4.9E-01	-0.11	0.14	0.02	0.05	3	-0.157	4.0E-01	-0.02	0.19	5
2007	-0.568	-1.74	1.8E-01	-2.48	1.34	-0.078	-0.75	5.1E-01	-0.69	0.53	0.50	0.14	3	-0.708	9.1E-02	0.26	0.21	5
2008	0.129	1.15	3.3E-01	-0.53	0.78	-1.009*	-3.53	3.9E-02	-2.68	0.66	0.31	0.62	3	0.553	1.7E-01	-0.58	2.78	5

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 5 observations per yearly regression from the 5 banks in the LSE that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

							Ta	ble V.4 (Continu	ed)								
	a a b							PAN	NEL D				- \				(** *)	
	Cross-Sect	ional Re	gression A	Analysis	for the Sa	ample Net I	ncome t	to Market	t Price F	Relative D	elta (<i>dI_t</i> .	$A_{1,t}$ - dN	<i>I_{t-1,t})</i> and	d Historical	Value-at-R	isk Actual	(V_t)	
						Regress	sion							Corre	lation	Descri	ptive	
Year			Slope					Intercept				Model		Cone	lation	Statis	tics	Obs.
	b	<i>t(b)</i>	<i>p(b)</i>	Lower	Upper	а	t(a)	p(a)	Lower	Upper	R^2	s(e)	df(e)	r	<i>p(r)</i>	Mean	SD	
1994	0.121	0.91	3.9E-01	-0.33	0.57	0.062	0.88	4.0E-01	-0.18	0.30	0.09	0.22	8	0.305	2.0E-01	0.01	0.56	10
1995	-0.133*	-2.99	1.7E-02	-0.28	0.02	-0.116	-2.09	7.0E-02	-0.30	0.07	0.53	0.17	8	-0.727**	8.6E-03	0.39	1.24	10
1996	-0.014	-0.08	9.3E-01	-0.57	0.54	-0.019	-0.23	8.2E-01	-0.29	0.25	8E-04	0.20	9	-0.028	4.7E-01	0.33	0.38	11
1997	0.091	1.01	3.4E-01	-0.20	0.38	0.013	0.34	7.5E-01	-0.11	0.14	0.10	0.10	9	0.318	1.7E-01	-0.26	0.36	11
1998	-0.029	-0.53	6.1E-01	-0.21	0.15	0.005	0.09	9.3E-01	-0.15	0.16	0.03	0.14	9	-0.175	3.0E-01	0.40	0.84	11
1999	-0.044	-0.75	4.7E-01	-0.24	0.15	-0.103*	-2.31	4.7E-02	-0.25	0.04	0.06	0.15	9	-0.243	2.4E-01	0.01	0.79	11
2000	-0.098	-0.79	4.5E-01	-0.50	0.31	-0.075	-0.72	4.9E-01	-0.41	0.26	0.06	0.33	9	-0.253	2.3E-01	-0.21	0.85	11
2001	-0.044	-0.38	7.1E-01	-0.42	0.33	-0.362**	-7.00	6.3E-05	-0.53	-0.19	0.02	0.17	9	-0.126	3.6E-01	-0.10	0.46	11
2002	0.018	0.20	8.4E-01	-0.27	0.31	-0.465**	-4.64	1.2E-03	-0.79	-0.14	5E-03	0.30	9	0.068	4.2E-01	0.50	1.07	11
2003	0.036**	6.90	7.0E-05	0.02	0.05	-0.415**	-5.70	3.0E-04	-0.65	-0.18	0.84	0.23	9	0.917**	3.5E-05	-4.56	13.8	11
2004	-0.004	-0.04	9.7E-01	-0.29	0.28	-0.111	-1.25	2.4E-01	-0.40	0.18	2E-04	0.29	9	-0.014	4.8E-01	0.02	1.06	11
2005	0.002	1.46	1.8E-01	5.5E-03	0.01	0.011	0.39	7.1E-01	-0.08	0.11	0.19	0.09	9	0.438	8.9E-02	6.59	24.9	11
2006	0.016	1.43	1.9E-01	-0.02	0.05	0.062	1.33	2.2E-01	-0.09	0.21	0.18	0.14	9	0.430	9.4E-02	1.43	4.16	11
2007	-0.005	-0.20	8.4E-01	-0.09	0.08	-0.056	-0.92	3.8E-01	-0.26	0.14	5E-03	0.19	9	-0.068	4.2E-01	0.89	2.24	11
2008	-0.655**	-3.74	4.6E-03	-1.22	-0.09	-0.651**	-5.30	4.9E-04	-1.05	-0.25	0.61	0.40	9	-0.78**	2.3E-03	-0.13	0.72	11

Panel notes: **, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels. The sample consists of a maximum of 11 observations per yearly regression from 11 banking related firms in the LSE that adopted the IFRS standards in 2005. The regressions are from 1994 to 2008. See main table notes for the regression model applied. Regression coefficients and related statistics are calculated using OLS regression.

Analysis

The Table V.4 cross-sectional regression slopes show that the net income to market price relative delta variable exhibits varying levels of statistically significant explanatory power to the Historical Value-at-Risk Actual variable for the samples and the control group.

The Primary and Secondary sample shows statistical significance for the years: 1995 and 2003. The Control group shows statistical significance for the years: 1997, 1998, 2000, 2006 and 2007.

V.5 Regressions for the UK Banks Relative Delta and Historical Value-at-Risk Actual

For the UK banks, the analysis and detailed results for the *accounting total to market price relative delta* and the *Historical Value-at-Risk Actual* time series regressions are presented in this section. The regression analysis is conducted within the time range 1994 to 2008.

V.5.1 Regressions for HSBC Holdings PLC Relative Delta and Historical Valueat-Risk Actual

The HSBC Holdings PLC regression analysis for the relative delta and Historical Value-at-Risk Actual is presented for the time range 1994 to 2008 in Table V.5. In the regressions, the Historical Value-at-Risk Actual variable is tested as the dependent variable, and the relative delta variable is tested as the independent variable.

Table V.5Time Series Regression Analysis for HSBC Holdings PLC Relative
Delta and Historical Value-at-Risk Actual

Time Series Regression

Table V.5 Panel A to Panel D show the results for the time series regressions that test the relative delta and Historical Value-at-Risk Actual by applying the regression specified in equation (3.56). In this table, Panel A presents results for the total equity to market price relative delta and Historical Value-at-Risk Actual regression; Panel B presents results for the total assets to market price relative delta and Historical Value-at-Risk Actual regression; Panel B presents results for the total assets results for the total liabilities to market price relative delta and Historical Value-at-Risk Actual regression; Panel C presents results for the total liabilities to market price relative delta and Historical Value-at-Risk Actual regression; and, Panel D presents results for the net income to market price relative delta and Historical Value-at-Risk Actual regression.

Time Series Regression Table Description

The Table V.5 columns represent the following:

The Total Period column represents the time series regression results for the year range 1994 to 2008, with a single year represented from 1st January to 31st December, and a 31st December variable record date. The Sub-Periods column represents the time series regression results for the year range 1994 to 2004, and for 1994 to 2007, with a single year represented from 1st January to 31st December, and a 31st December variable record date.

The Measure column represents the time series regression results for the total and sub-time periods and is represented by the following data panels:

The *Regression* panel presents the coefficients and related statistics for the regression specified in equation (3.56), and follows the model:

 $V_{\rm t} = a + b \ (dAC - dM)_t + e_t$

Where: V_t is the Value-at-Risk (V_t) at time t, $(dAC - dM)_t$ is the Accounting Total to Market Price Relative Delta $(dAC_{t-1,t} - dM_{t-1,t})$ at time t, and e_t is the regression error term. The Lower and Upper are the corresponding coefficient levels predicted at a 99% confidence level.

The *Correlation* panel presents the correlation statistics for the relationship between the dependent and independent variables. The *Descriptive Statistics* panel presents for the variables tested, the average sample means calculated using equation (M.6); the standard deviation, *SD*, of sample means for the time series using equation (M.7); and, the standard error of means, *SE*, for the time series. *The Observations* panel presents: *Firms (n)* that represents the one firm in the time series regression; *Years (T)* represents the number of time series years; *Total (n × T)* represents the number of pooled time series regression observations.

Correlation R^2 *Check* panel presents: r(dAC-dM, e) that represents the correlation test results for the coefficient of determination R^2 estimated by equation (M.22); p(r) represents the probability significance level for the correlation coefficient r(dAC-dM, e), and is calculated by applying equation (M.15); *Years (T)* represents the number of time series years.

Measure		Total Period	Sub-Periods	
		1994-2008	1994-2004	1994-2007
Slope	b	-0.350	-0.355	-0.345
	t (b)	-1.928	-1.691	-1.820
	p (b)	7.6E-02	1.3E-01	9.4E-02
	SE(b)	0.182	0.210	0.190
	Lower	-0.897	-1.038	-0.925
	Upper	0.197	0.328	0.234
Intercept	а	-0.110	-0.137	-0.108
Regression	t (a)	-2.110	-2.059	-1.951
	p (a)	5.5E-02	7.0E-02	7.5E-02
	SE(a)	0.052	0.067	0.055
Model	Lower	-0.268	-0.353	-0.277
	Upper	0.047	0.079	0.061
	R^2	0.222	0.241	0.216
	s(e)	0.184	0.209	0.192
	df(e)	13	9	12
	r	-0.471*	-0.491	-0.465*
Correlation	<i>p(r)</i>	3.8E-02	6.3E-02	4.7E-02
	dE-dM	0.119	0.102	0.111
	SD (dE-dM)	0.271	0.314	0.280
Descriptive Statistics	SE(dE-dM)	0.070	0.095	0.075
1	V 95	-0.152	-0.173	-0.146
	SD (V 95)	0.201	0.227	0.208
	SE(V 95)	0.052	0.069	0.056
	Firm Count (n)	1	1	1
Observations	Years (T)	15	11	14
	Total ($n \times T$)	15	11	14
	r (dAC-dM, e)	-2.5E-09	-2.1E-09	1.6E-10
Correlation R ² Check	p(r)	-	-	-
	Years (T)	15	11	14

PANEL A Time Series Regression Analysis for HSBC Holdings PLC Total Equity to Market Price Relative Delta (*dE*_{t+1} - *dM*_{t+1}) and Historical Value-at-Risk Actual (*V*_t)

Table V.5 (Continued)

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The model applied is $V_t = a + b (dE - dM)_t + e_t$ where: V_t is the Value-at-Risk (V_t) at time t, $(dE - dM)_t$ is the Total Equity to Market Price Relative Delta $(dE_{t-1,1} - dM_{t-1,1})$ at time t, and e_t is the regression error term. Regression coefficients and related statistics are calculated using OLS regression.

Measure		Total Period S		ub-Periods	
		1994-2008	1994-2004	1994-2007	
	Slope	b	-0.290	-0.491	-0.354
	1	t (b)	-1.539	-1.900	-1.521
		p(b)	1.5E-01	9.0E-02	1.5E-01
		SE(b)	0.189	0.259	0.233
		Lower	-0.858	-1.332	-1.064
		Upper	0.278	0.349	0.357
Inte	Intercept	а	-0.124*	-0.166*	-0.126*
Regression	1	t (a)	-2.354	-2.707	-2.316
0		p (a)	3.5E-02	2.4E-02	3.9E-02
		SE(a)	0.053	0.061	0.055
		Lower	-0.283	-0.364	-0.293
		Upper	0.035	0.033	0.040
Model	Model	R^2	0.154	0.286	0.162
		s(e)	0.192	0.202	0.198
		df(e)	13	9	12
Correlation		r	-0.393	-0.535*	-0.402
		p(r)	7.4E-02	4.5E-02	7.7E-02
		dA-dM	0.095	0.016	0.057
		SD (dA-dM)	0.272	0.248	0.236
Descriptive Statistics		SE(dA-dM)	0.070	0.075	0.063
		V 95	-0.152	-0.173	-0.146
		SD (V 95)	0.201	0.227	0.208
		SE(V 95)	0.052	0.069	0.056
Observations		Firm Count (n)	1	1	1
		Years (T)	15	11	14
		Total ($n \times T$)	15	11	14
Correlation R ² Check		r (dAC-dM, e)	1.1E-09	2.4E-09	6.0E-09
		p(r) Years (T)	- 15	- 11	- 14

PANEL B Time Series Regression Analysis for HSBC Holdings PLC Total Assets to Market Price Relative Delta (dA_{t-1}, - dM_{t-1}) and Historical Value-at-Risk Actual (V_t)

Table V.5 (Continued)

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The model applied is $V_t = a + b (dA - dM)_t + e_t$ where: V_t is the Value-at-Risk (V_t) at time t, $(dA - dM)_t$ is the Total Assets to Market Price Relative Delta $(dA_{t-1,1} - dM_{t-1,1})$ at time t, and e_t is the regression error term. Regression coefficients and related statistics are calculated using OLS regression.

Measure		Total Period Sub-Peri		Periods
		1994-2008	1994-2004	1994-2007
Slope	Ь	-0.282	-0.500	-0.349
	t (b)	-1.505	-1.917	-1.493
	p (b)	1.6E-01	8.7E-02	1.6E-01
	SE(b)	0.188	0.261	0.234
	Lower	-0.848	-1.349	-1.064
	Upper	0.283	0.348	0.365
Intercept	а	-0.125*	-0.167*	-0.127*
Regression	t (a)	-2.354	-2.734	-2.316
C	p (a)	3.5E-02	2.3E-02	3.9E-02
	SE(a)	0.053	0.061	0.055
	Lower	-0.284	-0.365	-0.294
	Upper	0.035	0.031	0.040
Model	R^2	0.148	0.290	0.157
	s(e)	0.193	0.202	0.199
	df(e)	13	9	12
	r	-0.385	-0.538*	-0.396
Correlation	p(r)	7.8E-02	4.4E-02	8.1E-02
	dL-dM	0.096	0.013	0.056
	SD (dL-dM)	0.275	0.245	0.235
Descripting Statistics	SE(dL-dM)	0.071	0.074	0.063
Descriptive statistics	V 95	-0.152	-0.173	-0.146
	SD (V 95)	0.201	0.227	0.208
	SE(V 95)	0.052	0.069	0.056
	Firm Count (n)	1	1	1
Observations	Years (T)	15	11	14
	Total $(n \times T)$	15	11	14
	r (dAC-dM, e)	-6.1E-09	-1.9E-09	-5.4E-09
Correlation R ² Check	p(r)	-	-	-
	Years (T)	15	11	14

PANEL C Time Series Regression Analysis for HSBC Holdings PLC Total Liabilities to Market Price Relative Delta ($dL_{t-1,t}$ - $dM_{t-1,t}$) and Historical Value-at-Risk Actual (V_t)

Table V.5 (Continued)

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The model applied is $V_t = a + b (dL - dM)_t + e_t$ where: V_t is the Value-at-Risk (V_t) at time t, $(dL - dM)_t$ is the Total Liabilities to Market Price Relative Delta $(dL_{t-1,1} - dM_{t-1,1})$ at time t, and e_t is the regression error term.

Regression coefficients and related statistics are calculated using OLS regression.
Measure		Total Period	Sub-Periods		
		1994-2008	1994-2004	1994-2007	
	Slope	b	-0.011	-0.150	-0.074
	I I I I	t(b)	-0.085	-0.662	-0.425
		p (b)	9.3E-01	5.2E-01	6.8E-01
		SE(b)	0.130	0.227	0.175
		Lower	-0.403	-0.887	-0.609
		Upper	0.381	0.587	0.460
	Intercept	а	-0.152*	-0.171*	-0.142*
Regression	1	t (a)	-2.816	-2.417	-2.433
0		p(a)	1.5E-02	3.9E-02	3.2E-02
		SE(a)	0.054	0.071	0.058
		Lower	-0.315	-0.400	-0.320
		Upper	0.011	0.059	0.036
	Model	R^2	0.001	0.046	0.015
		s(e)	0.209	0.234	0.215
		df(e)	13	9	12
		r	-0.023	-0.215	-0.122
Correl	ation	p(r)	4.7E-01	2.6E-01	3.4E-01
		dI-dM	-0.012	0.017	0.060
		SD (dI-dM)	0.430	0.327	0.340
	G	SE(dI-dM)	0.111	0.098	0.091
Descriptive	Statistics	V 95	-0.152	-0.173	-0.146
		SD (V 95)	0.201	0.227	0.208
		SE(V 95)	0.052	0.069	0.056
		Firm Count (n)	1	1	1
Observe	ations	Years (T)	15	11	14
		Total ($n \times T$)	15	11	14
	2	r (dAC-dM, e)	1.1E-09	2.4E-09	6.0E-09
Correlation	R ² Check	p(r)	-	-	-
		Years (T)	15	11	14

PANEL D Time Series Regression Analysis for HSBC Holdings PLC Net Income to Market Price Relative Delta (*dI*_{t-1,t} - *dM*_{t-1,t}) and Historical Value-at-Risk Actual (*V*_t)

 Table V.5 (Continued)

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels.

The model applied is $V_t = a + b (dI - dM)_t + e_t$ where: V_t is the Value-at-Risk (V_t) at time t, $(dI - dM)_t$ is the Net Income to Market Price Relative Delta $(dI_{t-1,1} - dM_{t-1,1})$ at time t, and e_t is the regression error term. Regression coefficients and related statistics are calculated using OLS regression.

Analysis

For HSBC Holdings PLC, the Table V.5 time series regression slopes show that the accounting totals to market price relative delta variable does not exhibit statistically significant explanatory power to the Historical Value-at-Risk Actual variable.

These time series slopes suggest that the HSBC Holdings PLC relative delta variable did not react significantly to the Historical Value-at-Risk Actual variable before and after the 2005 accounting change.

The *Correlation* R^2 *Check* results in this table exhibit levels close to zero. This study considers these levels to signify a valid measure for the reported R^2 . However, for this *Correlation* R^2 *Check* level it was not possible to decipher statistical levels of significance from the system used to produce this result.

V.5.2 Regressions for Barclays PLC Relative Delta and Historical Value-at-Risk Actual

The Barclays PLC regression analysis for the relative delta and Historical Value-at-Risk Actual is presented for the time range 1994 to 2008 in Table V.6. In the regressions, the Historical Value-at-Risk Actual variable is tested as the dependent variable, and the relative delta variable is tested as the independent variable.

Table V.6Time Series Regression Analysis for Barclays PLC Relative Delta
and Historical Value-at-Risk Actual

Time Series Regression

Table V.6 Panel A to Panel D show the results for the time series regressions that test the relative delta and Historical Value-at-Risk Actual by applying the regression specified in equation (3.56). In this table, Panel A presents results for the total equity to market price relative delta and Historical Value-at-Risk Actual regression; Panel B presents results for the total assets to market price relative delta and Historical Value-at-Risk Actual regression; Panel B presents results for the total assets to market price relative delta and Historical Value-at-Risk Actual regression; Panel C presents results for the total liabilities to market price relative delta and Historical Value-at-Risk Actual regression; and, Panel D presents results for the net income to market price relative delta and Historical Value-at-Risk Actual regression.

Time Series Regression Table Description

The Table V.6 columns represent the following:

The Total Period column represents the time series regression results for the year range 1994 to 2008, with a single year represented from 1st January to 31st December, and a 31st December variable record date. The Sub-Periods column represents the time series regression results for the year range 1994 to 2004, and for 1994 to 2007, with a single year represented from 1st January to 31st December, and a 31st December variable record date.

The Measure column represents the time series regression results for the total and sub-time periods and is represented by the following data panels:

The *Regression* panel presents the coefficients and related statistics for the regression specified in equation (3.56), and follows the model:

$V_{\rm t} = a + b \ (dAC - dM)_t + e_t$

Where: V_t is the Value-at-Risk (V_t) at time t, $(dAC - dM)_t$ is the Accounting Total to Market Price Relative Delta $(dAC_{t-1,t} - dM_{t-1,t})$ at time t, and e_t is the regression error term. The Lower and Upper are the corresponding coefficient levels predicted at a 99% confidence level.

The *Correlation* panel presents the correlation statistics for the relationship between the dependent and independent variables. The *Descriptive Statistics* panel presents for the variables tested, the average sample means calculated using equation (M.6); the standard deviation, *SD*, of sample means for the time series using equation (M.7); and, the standard error of means, *SE*, for the time series. *The Observations* panel presents: *Firms (n)* that represents the one firm in the time series regression; *Years (T)* represents the number of time series years; *Total (n × T)* represents the number of pooled time series regression observations.

Correlation R^2 *Check* panel presents: r(dAC-dM, e) that represents the correlation test results for the coefficient of determination R^2 estimated by equation (M.22); p(r) represents the probability significance level for the correlation coefficient r(dAC-dM, e), and is calculated by applying equation (M.15); *Years (T)* represents the number of time series years.

Measure		Total Period	Sub-Periods	
		1994-2008	1994-2004	1994-2007
Slope	b	-0.594**	-0.396	-0.415*
1	t (b)	-5.775	-1.757	-2.381
	p (b)	6.4E-05	1.1E-01	3.5E-02
	SE(b)	0.103	0.225	0.174
	Lower	-0.904	-1.127	-0.947
	Upper	-0.284	0.336	0.117
Intercept	а	-0.118*	-0.133	-0.115*
Regression	t (a)	-2.327	-2.158	-2.319
0	p(a)	3.7E-02	5.9E-02	3.9E-02
	SE(a)	0.051	0.062	0.049
	Lower	-0.270	-0.334	-0.266
	Upper	0.035	0.067	0.036
Model	R^2	0.720	0.255	0.321
	s(e)	0.188	0.205	0.184
	df(e)	13	9	12
	r	-0.848**	-0.505	-0.566*
Correlation	p(r)	3.2E-05	5.6E-02	1.7E-02
	dE-dM	0.138	0.003	0.035
	SD (dE-dM)	0.488	0.288	0.293
D	SE(dE-dM)	0.126	0.087	0.078
Descriptive Statistics	V 95	-0.200	-0.135	-0.129
	SD (V 95)	0.342	0.225	0.214
	SE(V 95)	0.088	0.068	0.057
	Firm Count (n)	1	1	1
Observations	Years (T)	15	11	14
	Total ($n \times T$)	15	11	14
-	r (dAC-dM, e)	4.3E-09	-1.4E-09	-5.3E-09
Correlation R ² Check	p(r)	-	-	-
	Years (T)	15	11	14

PANEL A Time Series Regression Analysis for Barclays PLC Total Equity to Market Price Relative Delta (dE_{t-Lt} - dM_{t-Lt}) and Historical Value-at-Risk Actual (V_t)

Table V.6 (Continued)

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The model applied is $V_t = a + b (dE - dM)_t + e_t$ where: V_t is the Value-at-Risk (V_t) at time t, $(dE - dM)_t$ is the Total Equity to Market Price Relative Delta $(dE_{t-1,1} - dM_{t-1,1})$ at time t, and e_t is the regression error term. Regression coefficients and related statistics are calculated using OLS regression.

Measure		Total Period	Sub-Periods	
		1994-2008	1994-2004	1994-2007
Slope	h	-0.545**	-0.494	-0.293
	t(h)	-4.731	-1.625	-1.461
	p (b)	3.9E-04	1.4E-01	1.7E-01
	SE(b)	0.115	0.304	0.201
	Lower	-0.892	-1.481	-0.906
	Upper	-0.198	0.494	0.320
Intercept	a	-0.108	-0.142	-0.112
Regression	t (a)	-1.841	-2.245	-1.988
0	p (a)	8.9E-02	5.1E-02	7.0E-02
	SE(a)	0.059	0.063	0.056
	Lower	-0.285	-0.347	-0.284
	Upper	0.069	0.063	0.060
Model	R^2	0.633	0.227	0.151
	s(e)	0.215	0.209	0.205
	df(e)	13	9	12
	r	-0.795**	-0.476	-0.389
Correlation	p(r)	2.0E-04	6.9E-02	8.5E-02
	dA-dM	0.168	-0.014	0.060
	SD (dA-dM)	0.499	0.217	0.284
_	SE(dA-dM)	0.129	0.066	0.076
Descriptive Statistics	V 95	-0.200	-0.135	-0.129
	SD (V 95)	0.342	0.225	0.214
	SE(V 95)	0.088	0.068	0.057
	Firm Count (n)	1	1	1
Observations	Years (T)	15	11	14
	Total ($n \times T$)	15	11	14
	r (dAC-dM, e)	-5.9E-09	-1.8E-08	1.1E-08
Correlation R ² Check	p(r)	-	-	-
	Years (T)	15	11	14

PANEL B Time Series Regression Analysis for Barclays PLC Total Assets to Market Price Relative Delta $(dA_{t-l,t} - dM_{t-l,t})$ and Historical Value-at-Risk Actual (V_t)

Table V.6 (Continued)

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The model applied is $V_t = a + b (dA - dM)_t + e_t$ where: V_t is the Value-at-Risk (V_t) at time t, $(dA - dM)_t$ is the Total Assets to Market Price Relative Delta $(dA_{t-1,1} - dM_{t-1,1})$ at time t, and e_t is the regression error term. Regression coefficients and related statistics are calculated using OLS regression.

М			Sub-Periods	
Measure		1994-2008	1994-2004	1994-2007
Slope	b	-0.541**	-0.487	-0.287
1	t (b)	-4.693	-1.599	-1.433
	p (b)	4.2E-04	1.4E-01	1.8E-01
	SE(b)	0.115	0.304	0.200
	Lower	-0.888	-1.476	-0.898
	Upper	-0.194	0.503	0.325
Intercept	a	-0.108	-0.141	-0.112
Regression	t (a)	-1.832	-2.234	-1.985
0	p (a)	9.0E-02	5.2E-02	7.0E-02
	SE(a)	0.059	0.063	0.056
	Lower	-0.286	-0.347	-0.284
	Upper	0.070	0.064	0.060
Model	R^2	0.629	0.221	0.146
	s(e)	0.216	0.210	0.206
	df(e)	13	9	12
	r	-0.793**	-0.470	-0.382
Correlation	p(r)	2.1E-04	7.2E-02	8.9E-02
	dL-dM	0.169	-0.014	0.061
	SD (dL-dM)	0.501	0.218	0.285
Description Statistics	SE(dL-dM)	0.129	0.066	0.076
Descriptive Statistics	V 05	-0 200	-0.135	-0 129
	V 95 SD (V 05)	0.342	0.225	0.214
	SE(V 95) SE(V 95)	0.088	0.068	0.057
	Firm Count (n)	1	1	1
Observations	Years (T)	15	11	14
	Total ($n \times T$)	15	11	14
	r (dAC-dM, e)	2.1E-09	2.2E-09	9.8E-09
Correlation R ² Check	p(r)	-	-	-
	Years (T)	15	11	14

PANEL C Time Series Regression Analysis for Barclays PLC Total Liabilities to Market Price Relative Delta (*dL*_{t-1}, - *dM*_{t-1}) and Historical Value-at-Risk Actual (*V*_t)

Table V.6 (Continued)

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The model applied is $V_t = a + b (dL - dM)_t + e_t$ where: V_t is the Value-at-Risk (V_t) at time t, $(dL - dM)_t$ is the Total Liabilities to Market Price Relative Delta $(dL_{t-1,1} - dM_{t-1,1})$ at time t, and e_t is the regression error term.

Regression coefficients and related statistics are calculated using OLS regression.

Measure		Total Period	Sub-Periods		
		1994-2008	1994-2004	1994-2007	
	Slope	h	-0.159	-0.023	-0.024
	lorop -	t(b)	-1.572	-0.279	-0.318
		p (b)	1.4E-01	7.9E-01	7.6E-01
		SE(b)	0.101	0.082	0.077
		Lower	-0.463	-0.289	-0.259
		Upper	0.145	0.243	0.210
	Intercept	a	-0.154	-0.130	-0.125
Regression	*	t (a)	-1.728	-1.768	-2.040
U		p (a)	1.1E-01	1.1E-01	6.4E-02
		SE(a)	0.089	0.073	0.061
		Lower	-0.422	-0.368	-0.312
		Upper	0.114	0.109	0.062
	Model	R^2	0.160	0.009	0.008
		s(e)	0.325	0.236	0.222
		df(e)	13	9	12
		r	-0.400	-0.093	-0.091
Correl	ation	p(r)	7.0E-02	3.9E-01	3.8E-01
		dI-dM	0.291	0.211	0.193
		SD (dI-dM)	0.861	0.912	0.802
Densisting	C	SE(dI-dM)	0.222	0.275	0.214
Descriptive	e Statistics	V 95	-0.200	-0.135	-0.129
		SD (V 95)	0.342	0.225	0.214
		SE(V 95)	0.088	0.068	0.057
		Firm Count (n)	1	1	1
Observ	ations	Years (T)	15	11	14
		Total ($n \times T$)	15	11	14
	2	r (dAC-dM, e)	2.8E-08	2.1E-08	-1.3E-08
Correlation	n R ² Check	p(r)	-	-	-
		Years (T)	15	11	14

PANEL D Time Series Regression Analysis for Barclays PLC Net Income to Market Price Relative Delta $(dI_{t-1,t} - dM_{t-1,t})$ and Historical Value-at-Risk Actual (V_t)

Table V.6 (Continued)

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels.

The model applied is $V_t = a + b (dI - dM)_t + e_t$ where: V_t is the Value-at-Risk (V_t) at time t, $(dI - dM)_t$ is the Net Income to Market Price Relative Delta $(dI_{t-1,1} - dM_{t-1,1})$ at time t, and e_t is the regression error term. Regression coefficients and related statistics are calculated using OLS regression.

Analysis

For Barclays PLC, the Table V.6 time series regression slopes show that the accounting totals to market price relative delta variable exhibits varying levels of statistically significant explanatory power to the Historical Value-at-Risk Actual variable.

The Table V.6 Panel A time series regression slope shows that before 2005, for the 1994 to 2004 time period, the total equity to market price relative delta variable does not show a statistically significant reaction to the Historical Value-at-Risk Actual variable. After 2005, the regression slope shows that for the 1994 to 2007 time period, a 1% change in the total equity to market price relative delta variable translates to nearly a 0.5% (-0.415) increase in the Historical Value-at-Risk Actual variable. The regression slope shows that for the 1994 to 2008 time period, a 1% change in the total equity to market price at 1% change in the total equity to 2008 time period, a 1% change in the total equity to 2008 time period, a 1% change in the total equity to market price relative delta variable. The regression slope shows that for the 1994 to 2008 time period, a 1% change in the total equity to market price relative delta variable. The regression slope shows that for the 1994 to 2008 time period, a 1% change in the total equity to market price relative delta variable translates to nearly a 0.6% (-0.594) increase in the Historical Value-at-Risk Actual variable.

The Panel B time series regression slope shows that before 2005, for the 1994 to 2004 time period, the total assets to market price relative delta variable does not show a statistically significant reaction to the Historical Value-at-Risk Actual variable. After 2005, the regression slope shows that for the 1994 to 2008 time period, a 1% change in the total assets to market price relative delta variable translates to more than a 0.5% (-0.545) increase in the Historical Value-at-Risk Actual variable.

The Panel C time series regression slope shows that before 2005, for the 1994 to 2004 time period, the total liabilities to market price relative delta variable does not show a statistically significant reaction to the Historical Value-at-Risk Actual variable. After 2005, the regression slope shows that for the 1994 to 2008 time period, a 1% change in the total liabilities to market price relative delta variable translates to more than a 0.5% (-0.541) increase in the Historical Value-at-Risk Actual variable.

The Panel D time series regression slope shows that before and after 2005, the net income to market price relative delta variable does not show a statistically significant reaction to the Historical Value-at-Risk Actual variable.

These time series slopes suggest that the Barclays PLC relative delta variable reacted significantly to the Historical Value-at-Risk Actual variable after the 2005 accounting change. However, the net income to market price relative delta variable did not show a statistically significant reaction to the Historical Value-at-Risk Actual variable before and after 2005.

The *Correlation* R^2 *Check* results in this table exhibit levels close to zero. This study consider these levels to signify a valid measure for the reported R^2 . However, for this *Correlation* R^2 *Check* level it was not possible to decipher statistical levels of significance from the system used to produce this result.

V.5.3 Regressions for The Royal Bank of Scotland Group PLC Relative Delta and Historical Value-at-Risk Actual

The Royal Bank of Scotland Group PLC regression analysis for the relative delta and Historical Value-at-Risk Actual is presented for the time range 1994 to 2008 in Table V.7. In the regressions, the Historical Value-at-Risk Actual variable is tested as the dependent variable, and the relative delta variable is tested as the independent variable.

Table V.7Time Series Regression Analysis for The Royal Bank of Scotland
Group PLC Relative Delta and Historical Value-at-Risk Actual

Time Series Regression

Table V.7 Panel A to Panel D show the results for the time series regressions that test the relative delta and Historical Value-at-Risk Actual by applying the regression specified in equation (3.56). In this table, Panel A presents results for the total equity to market price relative delta and Historical Value-at-Risk Actual regression; Panel B presents results for the total assets to market price relative delta and Historical Value-at-Risk Actual regression; Panel B presents results for the total assets to market price relative delta and Historical Value-at-Risk Actual regression; Panel C presents results for the total liabilities to market price relative delta and Historical Value-at-Risk Actual regression; and, Panel D presents results for the net income to market price relative delta and Historical Value-at-Risk Actual regression.

Time Series Regression Table Description

The Table V.7 columns represent the following:

The Total Period column represents the time series regression results for the year range 1994 to 2008, with a single year represented from 1st January to 31st December, and a 31st December variable record date. The Sub-Periods column represents the time series regression results for the year range 1994 to 2004, and for 1994 to 2007, with a single year represented from 1st January to 31st December, and a 31st December variable record date.

The Measure column represents the time series regression results for the total and sub-time periods and is represented by the following data panels:

The *Regression* panel presents the coefficients and related statistics for the regression specified in equation (3.56), and follows the model:

 $V_{\rm t} = a + b \ (dAC - dM)_t + e_t$

Where: V_t is the Value-at-Risk (V_t) at time t, $(dAC - dM)_t$ is the Accounting Total to Market Price Relative Delta $(dAC_{t-1,t} - dM_{t-1,t})$ at time t, and e_t is the regression error term. The Lower and Upper are the corresponding coefficient levels predicted at a 99% confidence level.

The *Correlation* panel presents the correlation statistics for the relationship between the dependent and independent variables. The *Descriptive Statistics* panel presents for the variables tested, the average sample means calculated using equation (M.6); the standard deviation, *SD*, of sample means for the time series using equation (M.7); and, the standard error of means, *SE*, for the time series. *The Observations* panel presents: *Firms (n)* that represents the one firm in the time series regression; *Years (T)* represents the number of time series years; *Total (n × T)* represents the number of pooled time series regression observations.

Correlation R^2 *Check* panel presents: r(dAC-dM, e) that represents the correlation test results for the coefficient of determination R^2 estimated by equation (M.22); p(r) represents the probability significance level for the correlation coefficient r(dAC-dM, e), and is calculated by applying equation (M.15); *Years (T)* represents the number of time series years.

Measure		Total Period	Sub-Periods		
		1994-2008	1994-2004	1994-2007	
	Slope	b	-0.265*	-0.118*	-0.125*
	1	t (b)	-2.665	-2.887	-3.054
		p (b)	1.9E-02	1.8E-02	1.0E-02
		SE(b)	0.099	0.041	0.041
		Lower	-0.564	-0.251	-0.250
		Upper	0.034	0.015	2.9E-05
	Intercept	а	-0.101	-0.035	-0.054
Regression	1	t (a)	-0.816	-0.666	-1.148
		p (a)	4.3E-01	5.2E-01	2.7E-01
		SE(a)	0.123	0.052	0.047
		Lower	-0.472	-0.205	-0.198
		Upper	0.271	0.135	0.090
	Model	R^2	0.353	0.481	0.437
		s(e)	0.443	0.167	0.169
		df(e)	13	9	12
		r	-0.594**	-0.693**	-0.661**
Correl	ation	<i>p(r)</i>	9.7E-03	9.0E-03	5.0E-03
		dE-dM	0.464	0.363	0.345
		SD (dE-dM)	1.192	1.289	1.141
Density	C	SE(dE-dM)	0.308	0.389	0.305
Descriptive	e Statistics	V 95	-0.224	-0.078	-0.097
		SD (V 95)	0.531	0.219	0.216
		SE(V 95)	0.137	0.066	0.058
		Firm Count (n)	1	1	1
Observ	ations	Years (T)	15	11	14
		Total ($n \times T$)	15	11	14
		r (dAC-dM, e)	-2.5E-09	4.7E-09	-4.7E-09
Correlation	n R ² Check	p(r)	-	-	-
		Years (T)	15	11	14

PANEL A Time Series Regression Analysis for The Royal Bank of Scotland Group PLC Total Equity to Market Price Relative Delta (*dE*_{t-Lt} - *dM*_{t-Lt}) and Historical Value-at-Risk Actual (*V*_t)

Table V.7 (Continued)

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The model applied is $V_t = a + b (dE - dM)_t + e_t$ where: V_t is the Value-at-Risk (V_t) at time t, $(dE - dM)_t$ is the Total Equity to Market Price Relative Delta $(dE_{t-1,1} - dM_{t-1,1})$ at time t, and e_t is the regression error term. Regression coefficients and related statistics are calculated using OLS regression.

Measure		Total Period	Sub-Periods		
		1994-2008	1994-2004	1994-2007	
	Slope	b	-0.694**	-0.365*	-0.345**
		t (b)	-/.134	-2.363	-3.431
		<i>p</i> (<i>b</i>)	/./E-06	4.2E-02	5.0E-03
		SE(b)	0.097	0.154	0.101
		Lower	-0.986	-0.866	-0.653
		Upper	-0.401	0.137	-0.038
	Intercent	a	-0.005	-0.044	-0.036
Regression	intercept	t(a)	-0.071	-0.775	-0.787
Regression		r(a)	9 4E-01	4 6E-01	4 5E-01
		p(a)	0.071	0.057	0.046
		SL(u)	0.071	0.007	0.010
		Lower	-0.219	-0.228	-0.178
		Upper	0.209	0.140	0.105
	Model	D ²	0 797	0 383	0.495
	mouei	Λ	0.248	0.182	0.160
		df(e)	13	9	12
		uj(c)			
Comp	lation	r	-0.892**	-0.619*	-0.704**
Correi	allon	p(r)	3.8E-06	2.1E-02	2.5E-03
		dA-dM	0.315	0.093	0.177
		SD (dA-dM)	0.683	0.372	0.440
		SE(dA-dM)	0.176	0.112	0.118
Descriptive	e Statistics				
		V 95	-0.224	-0.078	-0.097
		SD (V 95)	0.531	0.219	0.216
		SE(V 95)	0.137	0.066	0.058
		Firm Count (n)	1	1	1
Observ	ations	Years (T)	15	11	14
		Total $(n \times T)$	15	11	14
		r (dAC-dM, e)	-5.0E-09	-2.0E-09	-3.7E-09
Correlation	R^2 Check	p(r)	-	-	-
		Years (T)	15	11	14

Time Series Regression Analysis for The Royal Bank of Scotland Group PLC Total Assets to Market Price Relative Delta ($dA_{t-1,t} - dM_{t-1,t}$) and Historical Value-at-Risk Actual (V_t)

Table V.7 (Continued) PANEL B

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The model applied is $V_t = a + b (dA - dM)_t + e_t$ where: V_t is the Value-at-Risk (V_t) at time t, $(dA - dM)_t$ is the Total Assets to Market Price Relative Delta $(dA_{t-1,1} - dM_{t-1,1})$ at time t, and e_t is the regression error term. Regression coefficients and related statistics are calculated using OLS regression.

		Total Period	Sub-Periods	
Meas	measure		1994-2004	1994-2007
Slope	b	-0.691**	-0.363*	-0.343**
	t (b)	-7.132	-2.303	-3.358
	p (b)	7.7E-06	4.7E-02	5.7E-03
	SE(b)	0.097	0.158	0.102
	Lower	-0.983	-0.876	-0.654
	Upper	-0.399	0.149	-0.031
Intercept Regression	a t (a) p (a) SE(a)	-0.005 -0.067 9.5E-01 0.071	-0.044 -0.775 4.6E-01 0.057	-0.037 -0.786 4.5E-01 0.047
	Lower	-0.219	-0.230	-0.180
	Upper	0.209	0.141	0.106
Model	R ²	0.796	0.371	0.484
	s(e)	0.249	0.183	0.161
	df(e)	13	9	12
Correlation	r	-0.892**	-0.609*	-0.696**
	p(r)	3.8E-06	2.3E-02	2.8E-03
Descripting Statistics	dL-dM	0.316	0.092	0.177
	SD (dL-dM)	0.685	0.368	0.439
	SE(dL-dM)	0.177	0.111	0.117
Descriptive statistics	V 95	-0.224	-0.078	-0.097
	SD (V 95)	0.531	0.219	0.216
	SE(V 95)	0.137	0.066	0.058
Observations	Firm Count (n)	1	1	1
	Years (T)	15	11	14
Correlation R ² Check	Total $(n \times T)$	15	11	14
	r (dAC-dM, e)	1.5E-08	-1.4E-08	5.8E-09
	p(r)	-	-	-
	Years (T)	15	11	14

Time Series Regression Analysis for The Royal Bank of Scotland Group PLC Total Liabilities to Market Price Relative Delta ($dL_{t-1,t} - dM_{t-1,t}$) and Historical Value-at-Risk Actual (V_t)

Table V.7 (Continued) PANEL C

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The model applied is $V_t = a + b (dL - dM)_t + e_t$ where: V_t is the Value-at-Risk (V_t) at time t, $(dL - dM)_t$ is the Total Liabilities to Market Price Relative Delta $(dL_{t-1,1} - dM_{t-1,1})$ at time t, and e_t is the regression error term.

Regression coefficients and related statistics are calculated using OLS regression.

Measure		Total Period	Sub-Periods		
		1994-2008	1994-2004	1994-2007	
	Slope	h	0.266**	0.033	0.021
	Stope	t (h)	5.169	0.430	0.284
		p(b)	1.8E-04	6.8E-01	7.8E-01
		SE(b)	0.051	0.078	0.075
		Lower	0.111	-0.220	-0.209
		Upper	0.421	0.286	0.252
	Intercept	а	-0.214*	-0.089	-0.105
Regression		t (a)	-2.627	-1.206	-1.611
C		p (a)	2.1E-02	2.6E-01	1.3E-01
		SE(a)	0.081	0.074	0.065
		Lower	-0.459	-0.330	-0.303
		Upper	0.031	0.151	0.094
	Model	R^2	0.673	0.020	0.007
		s(e)	0.315	0.229	0.224
		df(e)	13	9	12
C 1		r	0.82**	0.142	0.082
Correla	tion	p(r)	9.0E-05	3.4E-01	3.9E-01
		dI-dM	-0.036	0.342	0.333
		SD (dI-dM)	1.636	0.929	0.823
Descriptions		SE(dI-dM)	0.422	0.280	0.220
Descriptive .	Statistics	V 95	-0.224	-0.078	-0.097
		SD (V 95)	0.531	0.219	0.216
		SE(V 95)	0.137	0.066	0.058
		Firm Count (n)	1	1	1
Observa	tions	Years (T)	15	11	14
		Total $(n \times T)$	15	11	14
		r (dAC-dM, e)	-1.1E-08	-4.8E-10	-4.4E-09
Correlation I	R ² Check	p(r)	-	-	-
		Years (T)	15	11	14

PANEL D Time Series Regression Analysis for The Royal Bank of Scotland Group PLC Net Income to Market Price Relative Delta $(dI_{t-1,t} - dM_{t-1,t})$ and Historical Value-at-Risk Actual (V_t)

Table V.7 (Continued)

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The model applied is $V_t = a + b (dI - dM)_t + e_t$ where: V_t is the Value-at-Risk (V_t) at time t, $(dI - dM)_t$ is the Net Income to Market Price Relative Delta $(dI_{t-1,1} - dM_{t-1,1})$ at time t, and e_t is the regression error term. Regression coefficients and related statistics are calculated using OLS regression.

Analysis

For the The Royal Bank of Scotland Group PLC, the Table V.7 time series regression slopes show that the accounting totals to market price relative delta variable exhibits varying levels of statistically significant explanatory power to the Historical Value-at-Risk Actual variable.

The Table V.7 Panel A time series regression slope shows that before 2005, for the 1994 to 2004 time period, a 1% change in the total equity to market price relative delta variable translates to just above a 0.1% (-0.118) increase in the Historical Value-at-Risk Actual variable. After 2005, the regression slope shows that for the 1994 to 2007 time period, a 1% change in the total equity to market price relative delta variable translates to just above a 0.12% (-0.125) increase in the Historical Value-at-Risk Actual variable. The regression slope shows that for the 1994 to 2008 time period, a 1% change in the total equity delta variable translates to just above a 0.12% (-0.125) increase in the Historical Value-at-Risk Actual variable. The regression slope shows that for the 1994 to 2008 time period, a 1% change in the total equity to market price relative delta variable.

The Panel B time series regression slope shows that before 2005, for the 1994 to 2004 time period, a 1% change in the total assets to market price relative delta variable translates to just above a 0.35% (-0.365) increase in the Historical Value-at-Risk Actual variable. After 2005, the regression slope shows that for the 1994 to 2007 time period, a 1% change in the total assets to market price relative delta variable translates to just above a 0.34% (-0.345) increase in the Historical Value-at-Risk Actual variable. The regression slope shows that for the 1994 to 2008 time period, a 1% change in the total assets to market price relative delta variable. The regression slope shows that for the 1994 to 2008 time period, a 1% change in the total assets to market price relative delta variable. The regression slope shows that for the 1994 to 2008 time period, a 1% change in the total assets to market price relative delta variable translates to nearly a 0.7% (-0.694) increase in the Historical Value-at-Risk Actual variable.

The Panel C time series regression slope shows that before 2005, for the 1994 to 2004 time period, a 1% change in the total liabilities to market price relative delta variable translates to just above a 0.35% (-0.363) increase in the Historical Value-at-Risk Actual variable. After 2005, the regression slope shows that for the 1994 to 2007 time period, a 1% change in the total liabilities to market price relative delta variable translates to just above a 0.34% (-0.343) increase in the Historical Value-at-Risk Actual variable above a 0.34% (-0.343) increase in the Historical Value-at-Risk Actual variable. The regression slope shows that for the 1994 to 2008 time period, a 1% change in the total

liabilities to market price relative delta variable translates to nearly a 0.7% (-0.691) increase in the Historical Value-at-Risk Actual variable.

The Panel D time series regression slope shows that before 2005, for the 1994 to 2004 time period, the net income to market price relative delta variable does not show a statistically significant reaction to the Historical Value-at-Risk Actual variable. After 2005, the regression slope shows that for the 1994 to 2008 time period, a 1% change in the net income to market price relative delta variable translates to nearly a 0.27% (-0.266) increase in the Historical Value-at-Risk Actual variable.

These time series slopes suggest that The Royal Bank of Scotland Group PLC relative delta variable reacted significantly to the Historical Value-at-Risk Actual variable after the 2005 accounting change. These reactions exhibit an increase in the magnitude of the difference between the variables after 2005.

The *Correlation* R^2 *Check* results in this table exhibit levels close to zero. This study considers these levels to signify a valid measure for the reported R^2 . However, for this *Correlation* R^2 *Check* level it was not possible to decipher statistical levels of significance from the system used to produce this result.

V.5.4 Regressions for Lloyds Banking Group PLC Relative Delta and Historical Value-at-Risk Actual

The Lloyds Banking Group PLC regression analysis for the relative delta and Historical Value-at-Risk Actual is presented for the time range 1994 to 2008 in Table V.8. In the regressions, the Historical Value-at-Risk Actual variable is tested as the dependent variable, and the relative delta variable is tested as the independent variable.

Table V.8Time Series Regression Analysis for Lloyds Banking Group PLC
Relative Delta and Historical Value-at-Risk Actual

Time Series Regression

Table V.8 Panel A to Panel D show the results for the time series regressions that test the relative delta and Historical Value-at-Risk Actual by applying the regression specified in equation (3.56). In this table, Panel A presents results for the total equity to market price relative delta and Historical Value-at-Risk Actual regression; Panel B presents results for the total assets to market price relative delta and Historical Value-at-Risk Actual regression; Panel B presents results for the total assets to market price relative delta and Historical Value-at-Risk Actual regression; Panel C presents results for the total liabilities to market price relative delta and Historical Value-at-Risk Actual regression; and, Panel D presents results for the net income to market price relative delta and Historical Value-at-Risk Actual regression.

Time Series Regression Table Description

The Table V.8 columns represent the following:

The Total Period column represents the time series regression results for the year range 1994 to 2008, with a single year represented from 1st January to 31st December, and a 31st December variable record date. The Sub-Periods column represents the time series regression results for the year range 1994 to 2004, and for 1994 to 2007, with a single year represented from 1st January to 31st December, and a 31st December variable record date.

The Measure column represents the time series regression results for the total and sub-time periods and is represented by the following data panels:

The *Regression* panel presents the coefficients and related statistics for the regression specified in equation (3.56), and follows the model:

 $V_{\rm t} = a + b \ (dAC - dM)_t + e_t$

Where: V_t is the Value-at-Risk (V_t) at time t, $(dAC - dM)_t$ is the Accounting Total to Market Price Relative Delta $(dAC_{t-1,t} - dM_{t-1,t})$ at time t, and e_t is the regression error term. The Lower and Upper are the corresponding coefficient levels predicted at a 99% confidence level.

The *Correlation* panel presents the correlation statistics for the relationship between the dependent and independent variables. The *Descriptive Statistics* panel presents for the variables tested, the average sample means calculated using equation (M.6); the standard deviation, *SD*, of sample means for the time series using equation (M.7); and, the standard error of means, *SE*, for the time series. *The Observations* panel presents: *Firms (n)* that represents the one firm in the time series regression; *Years (T)* represents the number of time series years; *Total (n × T)* represents the number of pooled time series regression observations.

Correlation R^2 *Check* panel presents: r(dAC-dM, e) that represents the correlation test results for the coefficient of determination R^2 estimated by equation (M.22); p(r) represents the probability significance level for the correlation coefficient r(dAC-dM, e), and is calculated by applying equation (M.15); *Years (T)* represents the number of time series years.

Measure		Total Period	Sub-Periods	
		1999-2008	1999-2004	1999-2007
Slope	b	-1.042**	-1.181	-1.174
1	t (b)	-4.964	-1.218	-2.060
	p (b)	1.1E-03	2.9E-01	7.8E-02
	SE(b)	0.210	0.970	0.570
	Lower	-1.747	-5.647	-3.169
	Upper	-0.338	3.284	0.820
Intercept	а	-0.118	-0.167	-0.104
Regression	t (a)	-1.455	-0.893	-0.986
	p (a)	1.8E-01	4.2E-01	3.6E-01
	SE(a)	0.081	0.187	0.105
	Lower	-0.392	-1.028	-0.471
	Upper	0.155	0.694	0.264
Model	R^2	0.755	0.270	0.378
	s(e)	0.209	0.250	0.222
	df(e)	8	4	7
	r	-0.869**	-0.520	-0.614*
Correlation	p(r)	5.5E-04	1.5E-01	3.9E-02
	dF-dM	0.227	0.161	0.131
	SD (dE-dM)	0.331	0.115	0.138
_	SE(dE-dM)	0.105	0.047	0.046
Descriptive Statistics	V95	-0.355	-0.358	-0.257
	SD (V 95)	0.397	0.262	0.263
	SE(V 95)	0.126	0.107	0.088
	Firm Count (n)	1	1	1
Observations	Years (T)	10	6	9
	Total ($n \times T$)	10	6	9
	r (dAC-dM, e)	-1.000**	-1.000**	-1.000**
Correlation R ² Check	p(r)	1.1E-62	4.6E-31	2.0E-51
	Years (T)	10	6	9

Table V.8 (Continued)
PANEL A

Time Series Regression Analysis for Lloyds Banking Group PLC Total Equity to Market Price Relative Delta $(dE_{t-1,t} - dM_{t-1,t})$ and Historical Value-at-Risk Actual (V_t)

Panel notes:

_

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The model applied is $V_t = a + b (dE - dM)_t + e_t$ where: V_t is the Value-at-Risk (V_t) at time t, $(dE - dM)_t$ is the Total Equity to Market Price Relative Delta $(dE_{t-1,1} - dM_{t-1,1})$ at time t, and e_t is the regression error term. Regression coefficients and related statistics are calculated using OLS regression.

Measure		Total Period	Sub-Periods	
		1999-2008	1999-2004	1999-2007
Slope	b	-0.675**	-0.213	-0.473
	t (b)	-3.840	-0.370	-1.002
	p (b)	4.9E-03	7.3E-01	3.5E-01
	SE(b)	0.176	0.575	0.472
	Lower	-1.264	-2.861	-2.126
	Upper	-0.085	2.435	1.179
Intercept Regression	a t (a) p (a) SE(a)	-0.162 -1.724 1.2E-01 0.094	-0.318 -2.013 1.1E-01 0.158	-0.186 -1.660 1.4E-01 0.112
	Lower	-0.476	-1.047	-0.580
	Upper	0.153	0.410	0.207
Model	R^{2} $s(e)$ $df(e)$	0.648 0.250 8	0.033 0.288 4	0.125 0.263 7
Correlation	r	-0.805**	-0.182	-0.354
	p(r)	2.5E-03	3.7E-01	1.7E-01
Descripting Statistics	dA-dM	0.287	0.184	0.149
	SD (dA-dM)	0.474	0.224	0.197
	SE(dA-dM)	0.150	0.092	0.066
Descriptive Statistics	V 95	-0.355	-0.358	-0.257
	SD (V 95)	0.397	0.262	0.263
	SE(V 95)	0.126	0.107	0.088
Observations	Firm Count (n)	1	1	1
	Years (T)	10	6	9
	Total $(n \times T)$	10	6	9
Correlation R ² Check	r (dAC-dM, e)	-1.000**	-1.000**	-1.000**
	p(r)	1.4E-59	2.7E-28	3.7E-51
	Years (T)	10	6	9

Time Series Regression Analysis for Lloyds Banking Group PLC Total Assets to Market Price Relative Delta $(dA_{t-I,t} - dM_{t-I,t})$ and Historical Value-at-Risk Actual (V_t)

Table V.8 (Continued) PANEL B

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The model applied is $V_t = a + b (dA - dM)_t + e_t$ where: V_t is the Value-at-Risk (V_t) at time t, $(dA - dM)_t$ is the Total Assets to Market Price Relative Delta $(dA_{t-1,1} - dM_{t-1,1})$ at time t, and e_t is the regression error term. Regression coefficients and related statistics are calculated using OLS regression.

Total Period Sub-Periods Measure 1999-2008 1999-2004 1999-2007 -0.665** -0.200 -0.449 Slope b -3.801 -0.359 -0.972 t (b) 7.4E-01 3.6E-01 5.2E-03 p (b) 0.175 0.558 0.462 SE(b)-1.251 -2.767-2.067Lower -0.0782.367 1.168 Upper Intercept -0.163 -0.321-0.190а -2.047 -1.692 -1.728 Regression t (a) 1.2E-01 1.1E-01 1.3E-01 p (a) 0.094 0.157 0.112 SE(a)-0.479-1.042-0.582 Lower 0.400 0.203 0.153 Upper R^2 Model 0.644 0.031 0.119 0.252 0.289 0.264 s(e)8 4 7 df(e) -0.802** -0.177 -0.345 r Correlation p(r)2.6E-03 3.7E-01 1.8E-01 0.289 0.185 0.150 dL-dM0.480 0.231 0.202 SD (dL-dM)0.095 SE(dL-dM)0.152 0.067 **Descriptive Statistics** -0.355 -0.358 -0.257 V95 0.397 0.262 0.263 SD (V 95) 0.126 0.107 0.088 SE(V 95) 1 1 1 Firm Count (n) 10 6 9 Years (T) **Observations** 10 6 9 Total $(n \times T)$ -1.000** -1.000** -1.000** r (dAC-dM, e)Correlation R² Check 1.6E-60 2.7E-27 7.9E-51 p(r)Years (T) 10 6 9

Time Series Regression Analysis for Lloyds Banking Group PLC Total Liabilities to Market Price Relative Delta $(dL_{t-1,t} - dM_{t-1,t})$ and Historical Value-at-Risk Actual (V_t)

Table V.8 (Continued) PANEL C

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The model applied is $V_t = a + b (dL - dM)_t + e_t$ where: V_t is the Value-at-Risk (V_t) at time t, $(dL - dM)_t$ is the Total Liabilities to Market Price Relative Delta $(dL_{t-1,1} - dM_{t-1,1})$ at time t, and e_t is the regression error term.

Regression coefficients and related statistics are calculated using OLS regression.

Measure		Total Period		Sub-Periods	
		1999-2008	1999-2004	1999-2007	
Slope	Ь	-0.285	-0.216	-0.244*	
*	t (b)	-1.682	-2.397	-2.574	
	p(b)	1.3E-01	7.5E-02	3.7E-02	
	SE(b)	0.170	0.090	0.095	
	Lower	-0.855	-0.630	-0.577	
	Upper	0.284	0.199	0.088	
Intercept	а	-0.281	-0.292*	-0.199*	
Regression	t (a)	-2.293	-3.585	-2.810	
8	p (a)	5.1E-02	2.3E-02	2.6E-02	
	SE(a)	0.123	0.081	0.071	
	Lower	-0.693	-0.667	-0.447	
	Upper	0.130	0.083	0.049	
Model	R^2	0.261	0.589	0.486	
	s(e)	0.362	0.188	0.202	
	df(e)	8	4	7	
Correlation	r	-0.511	-0.768*	-0.697*	
Correlation	<i>p(r)</i>	6.6E-02	3.7E-02	1.8E-02	
	dI-dM	0.258	0.304	0.236	
	SD (dI-dM)	0.712	0.934	0.751	
Descriptive Statistics	SE(dI-dM)	0.225	0.381	0.250	
Descriptive Statistics	V 95	-0.355	-0.358	-0.257	
	SD (V 95)	0.397	0.262	0.263	
	SE(V 95)	0.126	0.107	0.088	
	Firm Count (n)	1	1	1	
Observations	Years (T)	10	6	9	
	Total ($n \times T$)	10	6	9	
	r (dAC-dM, e)	-1.0**	-1.0**	-1.0**	
Correlation R ² Check	p(r)	8.6E-61	6.7E-31	1.3E-53	
	Years (T)	10	6	9	

Time Series Regression Analysis for Lloyds Banking Group PLC Net Income to Market Price Relative Delta $(dI_{t-1,t} - dM_{t-1,t})$ and Historical Value-at-Risk Actual (V_t)

Table V.8 (Continued) PANEL D

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The model applied is $V_t = a + b (dI - dM)_t + e_t$ where: V_t is the Value-at-Risk (V_t) at time t, $(dI - dM)_t$ is the Net Income to Market Price Relative Delta $(dI_{t-1,1} - dM_{t-1,1})$ at time t, and e_t is the regression error term. Regression coefficients and related statistics are calculated using OLS regression.

Analysis

For Lloyds Banking Group PLC, the Table V.8 time series regression slopes show that the accounting totals to market price relative delta variable exhibits varying levels of statistically significant explanatory power to the Historical Value-at-Risk Actual variable.

The Table V.8 Panel A time series regression slope shows that before 2005, for the 1994 to 2004 time period, the total equity to market price relative delta variable does not show a statistically significant reaction to the Historical Value-at-Risk Actual variable. After 2005, the regression slope shows that for the 1994 to 2008 time period, a 1% change in the total equity to market price relative delta variable translates to above a 1% (-1.042) increase in the Historical Value-at-Risk Actual variable.

The Panel B time series regression slope shows that before 2005, for the 1994 to 2004 time period, the total assets to market price relative delta variable does not show a statistically significant reaction to the Historical Value-at-Risk Actual variable. After 2005, the regression slope shows that for the 1994 to 2008 time period, a 1% change in the total assets to market price relative delta variable translates to above a 0.6% (-0.675) increase in the Historical Value-at-Risk Actual variable.

The Panel C time series regression slope shows that before 2005, for the 1994 to 2004 time period, the total liabilities to market price relative delta variable does not show a statistically significant reaction to the Historical Value-at-Risk Actual variable. After 2005, the regression slope shows that for the 1994 to 2008 time period, a 1% change in the total liabilities to market price relative delta variable translates to above a 0.6% (-0.665) increase in the Historical Value-at-Risk Actual variable.

The Panel D time series regression slope shows that before 2005, for the 1994 to 2004 time period, the net income to market price relative delta variable does not show a statistically significant reaction to the Historical Value-at-Risk Actual variable. After 2005, the regression slope shows that for the 1994 to 2007 time period, a 1% change in the net income to market price relative delta variable translates to above a 0.2% (-0.244) increase in the Historical Value-at-Risk Actual variable.

These time series slopes suggest that the Lloyds Banking Group PLC relative delta variable reacted significantly to the Historical Value-at-Risk Actual variable after the 2005 accounting change. However, the relative delta variable did not react significantly to the Historical Value-at-Risk Actual variable before 2005.

The results for the *Correlation* R^2 *Check* in this table exhibit a correlation level of -1. This study considers this level to signify further analysis of the reported R^2 . A *Correlation* R^2 *Check* parameter that exhibits such a high correlation measurement may indicate a possible limitation of the system used to produce this result.

V.5.5 Regressions for Standard Chartered PLC Relative Delta and Historical Value-at-Risk Actual

The Standard Chartered PLC regression analysis for the relative delta and Historical Value-at-Risk Actual is presented for the time range 1994 to 2008 in Table V.9. In the regressions, the Historical Value-at-Risk Actual variable is tested as the dependent variable, and the relative delta variable is tested as the independent variable.

Table V.9Time Series Regression Analysis for Standard Chartered PLCRelative Delta and Historical Value-at-Risk Actual

Time Series Regression

Table V.9 Panel A to Panel D show the results for the time series regressions that test the relative delta and Historical Value-at-Risk Actual by applying the regression specified in equation (3.56). In this table, Panel A presents results for the total equity to market price relative delta and Historical Value-at-Risk Actual regression; Panel B presents results for the total assets to market price relative delta and Historical Value-at-Risk Actual regression; Panel B presents results for the total assets to market price relative delta and Historical Value-at-Risk Actual regression; Panel C presents results for the total liabilities to market price relative delta and Historical Value-at-Risk Actual regression; and, Panel D presents results for the net income to market price relative delta and Historical Value-at-Risk Actual regression.

Time Series Regression Table Description

The Table V.9 columns represent the following:

The Total Period column represents the time series regression results for the year range 1994 to 2008, with a single year represented from 1st January to 31st December, and a 31st December variable record date. The Sub-Periods column represents the time series regression results for the year range 1994 to 2004, and for 1994 to 2007, with a single year represented from 1st January to 31st December, and a 31st December variable record date.

The Measure column represents the time series regression results for the total and sub-time periods and is represented by the following data panels:

The *Regression* panel presents the coefficients and related statistics for the regression specified in equation (3.56), and follows the model:

$V_{\rm t} = a + b \ (dAC - dM)_t + e_t$

Where: V_t is the Value-at-Risk (V_t) at time t, $(dAC - dM)_t$ is the Accounting Total to Market Price Relative Delta $(dAC_{t-1,t} - dM_{t-1,t})$ at time t, and e_t is the regression error term. The Lower and Upper are the corresponding coefficient levels predicted at a 99% confidence level.

The *Correlation* panel presents the correlation statistics for the relationship between the dependent and independent variables. The *Descriptive Statistics* panel presents for the variables tested, the average sample means calculated using equation (M.6); the standard deviation, *SD*, of sample means for the time series using equation (M.7); and, the standard error of means, *SE*, for the time series. *The Observations* panel presents: *Firms (n)* that represents the one firm in the time series regression; *Years (T)* represents the number of time series years; *Total (n × T)* represents the number of pooled time series regression observations.

Correlation R^2 *Check* panel presents: r(dAC-dM, e) that represents the correlation test results for the coefficient of determination R^2 estimated by equation (M.22); p(r) represents the probability significance level for the correlation coefficient r(dAC-dM, e), and is calculated by applying equation (M.15); *Years (T)* represents the number of time series years.

	150	Total Period	Sub-F	eriods
Measu	ure	1994-2008	1994-2004	1994-2007
Slope	Ь	-0.383	-0.145	-0.035
*	t (b)	-2.083	-0.482	-0.126
	p(b)	5.8E-02	6.4E-01	9.0E-01
	SE(b)	0.184	0.301	0.277
	Lower	-0.937	-1.123	-0.881
	Upper	0.171	0.833	0.811
Intercept	а	-0.141*	-0.185*	-0.141*
Regression	t (a)	-2.179	-2.663	-2.306
C	p (a)	4.8E-02	2.6E-02	4.0E-02
	SE(a)	0.065	0.069	0.061
	Lower	-0.335	-0.411	-0.327
	Upper	0.054	0.041	0.046
Model	R^2	0.250	0.025	0.001
	s(e)	0.237	0.229	0.223
	df(e)	13	9	12
Completion	r	-0.5*	-0.159	-0.036
Correlation	<i>p(r)</i>	2.9E-02	3.2E-01	4.5E-01
	dE-dM	0.113	0.021	0.044
	SD (dE-dM)	0.344	0.241	0.224
	SE(dE-dM)	0.089	0.073	0.060
Descriptive Statistics	V 95	-0.184	-0.188	-0.142
	SD (V 95)	0.263	0.220	0.215
	SE(V 95)	0.068	0.066	0.057
	Firm Count (n)	1	1	1
Observations	Years (T)	15	11	14
	Total $(n \times T)$	15	11	14
	r (dAC-dM, e)	5.3E-09	1.6E-09	6.6E-09
<i>Correlation</i> R^2 <i>Check</i>	p(r)	-	-	-
	Years (T)	15	11	14

PANEL A Time Series Regression Analysis for Standard Chartered PLC Total Equity to Market Price Relative Delta ($dE_{t,1,t}$ - $dM_{t,1,t}$) and Historical Value-at-Risk Actual (V_t)

Table V.9 (Continued)

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The model applied is $V_t = a + b (dE - dM)_t + e_t$ where: V_t is the Value-at-Risk (V_t) at time t, $(dE - dM)_t$ is the Total Equity to Market Price Relative Delta $(dE_{t-1,1} - dM_{t-1,1})$ at time t, and e_t is the regression error term. Regression coefficients and related statistics are calculated using OLS regression.

		Total Period	Sub-F	Periods
Meas	ure	1994-2008	1994-2004	1994-2007
Slope	h	-0.348*	-0.086	-0.024
	t (h)	-2.163	-0.300	-0.090
	p (b)	5.0E-02	7.7E-01	9.3E-01
	SE(b)	0.161	0.285	0.263
	Lower	-0.832	-1.012	-0.825
	Upper	0.136	0.841	0.778
Intercept	а	-0.159*	-0.19*	-0.142*
Regression	t (a)	-2.583	-2.712	-2.381
0	p (a)	2.3E-02	2.4E-02	3.5E-02
	SE(a)	0.062	0.070	0.060
	Lower	-0.345	-0.418	-0.325
	Upper	0.026	0.038	0.040
Model	R^2	0.265	0.010	0.001
	s(e)	0.234	0.231	0.223
	df(e)	13	9	12
	r	-0.514*	-0.100	-0.026
Correlation	p(r)	2.5E-02	3.9E-01	4.7E-01
	dA-dM	0.072	-0.027	-0.010
	SD (dA-dM)	0.390	0.256	0.236
	SE(dA-dM)	0.101	0.077	0.063
Descriptive Statistics	V 95	-0.184	-0.188	-0.142
	SD (V 95)	0.263	0.220	0.215
	SE(V 95)	0.068	0.066	0.057
	Firm Count (n)	1	1	1
Observations	Years (T)	15	11	14
	Total ($n \times T$)	15	11	14
2	r (dAC-dM, e)	3.6E-09	7.6E-09	6.0E-09
Correlation R ² Check	p(r)	-	-	-
	Years (T)	15	11	14

Time Series Regression Analysis for Standard Chartered PLC Total Assets to Market Price Relative Delta ($dA_{t-1,t} - dM_{t-1,t}$) and Historical Value-at-Risk Actual (V_t)

Table V.9 (Continued) PANEL B

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The model applied is $V_t = a + b (dA - dM)_t + e_t$ where: V_t is the Value-at-Risk (V_t) at time t, $(dA - dM)_t$ is the Total Assets to Market Price Relative Delta $(dA_{t-1,1} - dM_{t-1,1})$ at time t, and e_t is the regression error term. Regression coefficients and related statistics are calculated using OLS regression.

	м		Total Period	Sub-F	eriods
	Measu	ire	1994-2008	1994-2004	1994-2007
	Slope	b	-0.343	-0.081	-0.019
	1	t (b)	-2.154	-0.287	-0.072
		p(b)	5.1E-02	7.8E-01	9.4E-01
		SE(b)	0.159	0.284	0.260
		Lower	-0.822	-1.005	-0.814
		Upper	0.137	0.842	0.776
	Intercept	а	-0.16*	-0.19*	-0.142*
Regression		t (a)	-2.592	-2.709	-2.380
č		p(a)	2.2E-02	2.4E-02	3.5E-02
		SE(a)	0.062	0.070	0.060
		Lower	-0.345	-0.419	-0.325
		Upper	0.026	0.038	0.040
	Model	R^2	0.263	0.009	4.3E-04
		s(e)	0.235	0.231	0.223
		df(e)	13	9	12
<i>a</i> 1		r	-0.513*	-0.095	-0.021
Correl	ation	p(r)	2.5E-02	3.9E-01	4.7E-01
		dL-dM	0.071	-0.030	-0.012
		SD (dL-dM)	0.394	0.257	0.238
	G	SE(dL-dM)	0.102	0.078	0.064
Descriptive	e Statistics	V 95	-0.184	-0.188	-0.142
		SD (V 95)	0.263	0.220	0.215
		SE(V 95)	0.068	0.066	0.057
		Firm Count (n)	1	1	1
Observ	ations	Years (T)	15	11	14
		Total ($n \times T$)	15	11	14
	2	r (dAC-dM, e)	-4.6E-09	2.4E-09	6.4E-09
Correlation	n R ² Check	p(r)	-	-	-
		Years (T)	15	11	14

Time Series Regression Analysis for Standard Chartered PLC Total Liabilities to Market Price Relative Delta $(dL_{t-l,t} - dM_{t-l,t})$ and Historical Value-at-Risk Actual (V_t)

Table V.9 (Continued) PANEL C

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels; correlation coefficients significant at the 0.01 and 0.05 one-tailed levels.

The model applied is $V_t = a + b (dL - dM)_t + e_t$ where: V_t is the Value-at-Risk (V_t) at time t, $(dL - dM)_t$ is the Total Liabilities to Market Price Relative Delta $(dL_{t-1,1} - dM_{t-1,1})$ at time t, and e_t is the regression error term.

Regression coefficients and related statistics are calculated using OLS regression.

			Total Period	Sub-F	Periods
	Measu	ure	1994-2008	1994-2004	1994-2007
	Slope	Ь	-0.133	0.110	0.112
	1	t (b)	-0.942	0.682	0.752
		p (b)	3.6E-01	5.1E-01	4.7E-01
		SE(b)	0.141	0.161	0.149
		Lower	-0.558	-0.413	-0.342
		Upper	0.292	0.633	0.566
	Intercept	а	-0.17*	-0.189*	-0.145*
Regression	1	t (a)	-2.433	-2.769	-2.478
8		p (a)	3.0E-02	2.2E-02	2.9E-02
		SE(a)	0.070	0.068	0.059
		Lower	-0.381	-0.411	-0.324
		Upper	0.041	0.033	0.034
	Model	R^2	0.064	0.049	0.045
		s(e)	0.265	0.227	0.218
		df(e)	13	9	12
		r	-0.253	0.222	0.212
Correl	ation	p(r)	1.8E-01	2.6E-01	2.3E-01
		dI-dM	0.106	0.012	0.026
		SD (dI-dM)	0.501	0.445	0.408
	G	SE(dI-dM)	0.129	0.134	0.109
Descriptive	Statistics	V 95	-0.184	-0.188	-0.142
		SD (V 95)	0.263	0.220	0.215
		SE(V 95)	0.068	0.066	0.057
		Firm Count (n)	1	1	1
Observa	ations	Years (T)	15	11	14
		Total ($n \times T$)	15	11	14
	2	r (dAC-dM, e)	-1.5E-08	-4.3E-09	-2.1E-09
Correlation	R ^₄ Check	p(r)	-	-	-
		Years (T)	15	11	14

Time Series Regression Analysis for Standard Chartered PLC Net Income to Market Price Relative Delta ($dI_{t-1,t}$ - $dM_{t-1,t}$) and Historical Value-at-Risk Actual (V_t)

Table V.9 (Continued) PANEL D

Panel notes:

**, * Regression coefficients significant at the 0.01 and 0.05 two-tailed levels.

The model applied is $V_t = a + b (dI - dM)_t + e_t$ where: V_t is the Value-at-Risk (V_t) at time t, $(dI - dM)_t$ is the Net Income to Market Price Relative Delta $(dI_{t-1,1} - dM_{t-1,1})$ at time t, and e_t is the regression error term. Regression coefficients and related statistics are calculated using OLS regression.

Analysis

For Standard Chartered PLC, the Table V.9 time series regression slopes show that the accounting totals to market price relative delta variable exhibits marginal levels of statistically significant explanatory power to the Historical Value-at-Risk Actual variable.

The Table V.9 Panel B time series regression slope shows that before 2005, for the 1994 to 2004 time period, the total assets to market price relative delta variable does not show a statistically significant reaction to the Historical Value-at-Risk Actual variable. After 2005, the regression slope shows that for the 1994 to 2008 time period, a 1% change in the total assets to market price relative delta variable translates to above a 0.3% (-0.348) increase in the Historical Value-at-Risk Actual variable.

These time series slopes suggest that the Standard Chartered PLC total assets to market price relative delta variable reacted significantly to the Historical Value-at-Risk Actual variable after the 2005 accounting change. Other than this, the time series slopes suggest that in general the accounting to market price relative delta variables do not react significantly to the Historical Value-at-Risk Actual variable before and after 2005.

The *Correlation* R^2 *Check* results in this table exhibit levels close to zero. This study considers these levels to signify a valid measure for the reported R^2 . However, for this *Correlation* R^2 *Check* level it was not possible to decipher statistical levels of significance from the system used to produce this result.

APPENDIX W PRIMARY SAMPLE AND SECONDARY SAMPLE DATA

W.1 Primary Sample and Secondary Sample Data

This appendix presents in Table W.1 to Table W.5 the data collected for the banks registered with the LSE's UK banking sector. These banks represent the Primary sample group of firms tested in this research. Summary information for these banks is presented in Table 3.6, labeled 'Primary' in the 'Sample' column.

Table W.6 to Table W.16 presents the data collected for the banking related firms registered with the LSE's UK financial services sector. These firms represent the Secondary sample group of firms tested in this research. Summary information for these banks is presented in Table 3.6, labeled 'Secondary' in the 'Sample' column.

For each table presented the following superscript letters represent the corresponding details:

^a The fiscal year-end financial statement reporting date.

^b The year used to group the variable.

^c Reported at the scale 1,000,000 GBP (1 Million GBP).

^d Net income amounts presented from 1992 to 2001 are the full year figures, and from 2002 to 2008 are the second half-year figures. This in accordance with the convention applied in this study to test the net income variable.

^e The end of day market price of the firm represented using the scale: 0.01GBP (Great British Pound), equivalent to 1GBX.

^f Market Value to Book Value ratio calculated by dividing the variables Market Value by Total Shareholders' Equity.

^g Market Value to Book Value ratio calculated by dividing the variables Common Shares Outstanding and Market Price by Total Shareholders' Equity.

				Market Variable								
Date ^a	Year ^b	Total Shareholders' Equity		Total A	Assets	Total Lia	abilities	Net Ir	come ^d	Market	Market	Market To
		Original	Restated	Original	Restated	Original	Restated	Original	Restated	Price	Value	Book Value
31 Dec 1992	1992	8,011	-	170,414	-	161,104	-	1,221	-	143.06	4,018.80	1.54
31 Dec 1993	1993	9,334	-	206,007	-	195,107	-	1,806	-	282.93	7,996.54	2.64
31 Dec 1994	1994	10,790	-	201,484	-	189,021	-	2,053	-	205.22	6,016.06	1.71
31 Dec 1995	1995	20,787	-	352,090	-	328,132	-	3,892	-	292.22	8,651.60	1.98
31 Dec 1996	1996	26,005	-	404,967	-	375,324	-	4,897	-	379.36	11,350.63	2.28
31 Dec 1997	1997	16,442	-	286,333	-	267,704	-	3,355	-	453.15	13,632.77	2.54
31 Dec 1998	1998	16,470	-	290,259	-	271,875	-	2,595	-	473.48	14,390.20	2.66^{f}
31 Dec 1999	1999	20,725	-	352,940	-	329,948	-	3,355	-	752.05	72,983.81	3.52 ^g
31 Dec 2000	2000	30,129	-	450,469	-	415,447	-	4,437	-	858.36	91,081.81	3.02
31 Dec 2001	2001	31,211	-	477,252	-	441,582	-	3,714	-	702.38	75,335.19	2.41
31 Dec 2002	2002	32,233	-	470,622	-	434,283	-	1,753	-	598.24	64,995.39	2.02
31 Dec 2003	2003	41,602	-	576,091	-	528,403	-	2,558	-	765.12	96,228.88	2.31
31 Dec 2004	2004	45,118	44,545	663,689	665,038	611,688	613,370	2,696	2,932	765.99	98,202.44	2.18
31 Dec 2005	2005	53,841	-	873,339	-	816,123	-	4,459	-	813.05	105,742.40	1.97
31 Dec 2006	2006	55,362	-	949,093	-	890,371	-	3,426	-	811.31	107,736.90	1.95
31 Dec 2007	2007	64,383	-	1,180,037	-	1,112,009	-	4,039	-	733.75	99,600.50	1.54
31 Dec 2008	2008	63,612	-	1,753,054	-	1,684,826	-	-1,035	-	576.89	80,136.81	1.24

Table W.1Original and Restated Data for Accounting Variables and Market Variables for HSBC HOLDINGS PLC Surveyed on a Yearly
Basis from 1992 to 2008

				Market Variable								
Date ^a	Year ^b	Total Sha Eq	areholders' juity	Total A	Assets	Total Lia	abilities	Net In	ncome ^d	Market	Market	Market To
		Original	Restated	Original	Restated	Original	Restated	Original	Restated	Price	Value	Book Value
31 Dec 1992	1992	5,279	-	149,118	-	143,156	-	-343	-	92.88	6,157.41	1.17
31 Dec 1993	1993	5,312	-	166,008	-	160,019	-	313	-	154.85	10,315.92	1.94
31 Dec 1994	1994	6,161	-	162,403	-	155,913	-	1,179	-	148.64	9,981.67	1.62
31 Dec 1995	1995	7,027	-	168,826	-	161,456	-	1,364	-	179.92	11,986.58	1.71
31 Dec 1996	1996	7,267	-	186,002	-	178,415	-	1,639	-	243.59	15,417.70	2.12
31 Dec 1997	1997	7,577	-	234,614	-	226,711	-	1,130	-	393.93	24,637.71	3.25
31 Dec 1998	1998	7,880	-	219,451	-	211,257	-	1,335	-	315.54	19,569.58	2.48 ^g
31 Dec 1999	1999	8,478	-	254,788	-	245,958	-	1,759	-	433.86	26,623.05	3.14 ^g
31 Dec 2000	2000	13,182	-	316,185	-	301,403	-	2,473	-	504.47	34,470.62	2.62
31 Dec 2001	2001	14,502	-	356,643	-	340,135	-	2,465	-	553.89	38,006.09	2.62
31 Dec 2002	2002	15,150	-	403,011	-	387,705	-	997	-	374.94	25,709.00	1.70
31 Dec 2003	2003	16,374	-	443,262	-	426,605	-	1,361	-	485.23	32,698.64	2.00
31 Dec 2004	2004	17,417	15,870	522,089	536,793	503,771	520,029	1,552	1,456	570.69	37,818.09	2.17
31 Dec 2005	2005	17,426	-	923,671	-	899,241	-	1,606	-	595.04	39,650.84	2.28
31 Dec 2006	2006	19,799	-	996,023	-	968,633	-	2,264	-	710.93	47,706.98	2.41
31 Dec 2007	2007	23,291	-	1,225,898	-	1,193,422	-	1,783	-	490.83	33,289.40	1.43
31 Dec 2008	2008	32,966	-	2,050,312	-	2,006,553	-	2,664	-	153.40	12,842.38	0.55

Table W.2Original and Restated Data for Accounting Variables and Market Variables for BARCLAYS PLC Surveyed on a Yearly Basis
from 1992 to 2008

			Accounting Variable ^c									
Date ^a	Year ^b	Total Shareholders' Equity		Total Assets		Total Li	Total Liabilities		come ^d	Market	Market	Market To
		Original	Restated	Original	Restated	Original	Restated	Original	Restated	Price ^e	Value	Book Value
30 Sep 1992	1992	1,780	2	34,498		32,702	-	11	-	41.83	1,265.22	0.71
30 Sep 1993	1993	1,897	-	36,294	-	34,378	-	139	-	84.44	2,596.38	1.37
30 Sep 1994	1994	1,936		45,320	-	43,368	(1) (1) (1) (1) (1) (1) (1) (1) (1) (1)	331		107.16	3,324.15	2.34
30 Sep 1995	1995	2,165		51,044		48,861		352	-	116.20	3,628.08	2.36
30 Sep 1996	1996	2,510	-	61,116	-	58,456	-	439	-	126.27	3,990.24	2.13
30 Sep 1997	1997	3,042	-	72,601	-	69,369	-	457	-	178.31	5,972.82	2.59
30 Sep 1998	1998	2,953	-	79,676	-	76,631	20	637	a de la companya de	173.01	5,863.79	2.77
30 Sep 1999	1999	4,202	-	88,852	1 3 - 11	84,504	-	776	-	336.47	11,617.26	4.07
31 Dec 2000	2000	23,116	-	320,004	-	296,342	-	1,478	-	441.58	42,357.62	2.22
31 Dec 2001	2001	27,620		368,782	-	340,577		1,868	03-r	466.70	47,683.47	2.13
31 Dec 2002	2002	27,052	-	412,000	-	383,109	-	635		415.34	43,053.00	1.83
31 Dec 2003	2003	28,099		455,275	-	424,463	-	570		459.44	48,776.46	1.94
31 Dec 2004	2004	31,865	33,905	583,467	588,075	547,773	550,678	2,150	2,455	489.03	55,584.00	2.03
31 Dec 2005	2005	35,435	1	776,671		739,127	-	2,858	2,866	489.87	55,925.58	1.58
31 Dec 2006	2006	40,227	1 m - 1	871,276	-	825,786	-	3,224		556.30	62,757.95	1.56
31 Dec 2007	2007	53,038		1,897,575	14.1	1,806,149	4	3,748	-	371.80	44,443.56	0.84
31 Dec 2008	2008	58,879	-	2,394,570		2,314,072		-23,335	-	49.40	19,491.70	0.37

Table W.3Original and Restated Data for Accounting Variables and Market Variables for THE ROYAL BANK OF SCOTLAND GROUP
PLC Surveyed on a Yearly Basis from 1992 to 2008

				Accounting Variable ^c							Market Variable			
Date ^a	Year ^b	Year ^b Total Shar Equ		Total Shareholders' Equity Total Assets		Total Li	abilities	Net In	come ^d	Market	Market	Market To		
		Original	Restated	Original	Restated	Original	Restated	Original	Restated	Price ^e	Value	Book Value		
31 Dec 1992	1992	2,730	12	61,004	-	57,739	-	441	-		-	1.2.		
31 Dec 1993	1993	3,063	-	79,757	-	76,152	-	605	-	-		- ÷		
31 Dec 1994	1994	3,661	-	81,357	-7	77,129		797		÷.				
31 Dec 1995	1995	4,158	-	147,996	-	143,203	- 2	961	1 C-1	246.96	16,835.50	4.05		
31 Dec 1996	1996	5,027	-	147,320	-	142,259	-	1,575		320.72	23,079.10	4.59		
31 Dec 1997	1997	6,218	-	158,070	-	151,812	-	2,335	-	586.31	42,458.64	6.83		
31 Dec 1998	1998	7,454	-	167,976	-	160,480		2,120	-	636.97	46,469.21	6.23 ^g		
31 Dec 1999	1999	8,658		176,056		167,365		2,514		577.00	42,403.87	4.9 ^g		
31 Dec 2000	2000	9,709	-	217,954	-	207,693		2,724	-	527.45	38,933.37	4.01		
31 Dec 2001	2001	10,737	-	236,516	-	224,782	200	2,500		555.76	41,576.21	3.87		
31 Dec 2002	2002	7,954	-	252,740		244,055		668		332.27	24,900.17	3.13		
31 Dec 2003	2003	9,624		252,012	-	241,661	-	2,099		333.76	25,059.32	2.60		
31 Dec 2004	2004	9,977	11,047	279,843	284,422	269,270	272,744	1,338	1,298	352.38	26,470.95	2.65		
31 Dec 2005	2005	10,195		309,754	-	299,124		1,301	-	363.93	27,361.48	2.68		
31 Dec 2006	2006	11,155	1 A .	343,598		332,091		1,589	-	425.76	32,191.01	2.89		
31 Dec 2007	2007	12,141		353,346	1.5	340,921	-	1,749		351.64	26,657.14	2.39		
31 Dec 2008	2008	9,393	-	435,200	-	425,501		243	-	93.87	10,797.24	0.89		

Table W.4Original and Restated Data for Accounting Variables and Market Variables for LLOYDS BANKING GROUP PLC Surveyed
on a Yearly Basis from 1992 to 2008

				Accounting Variable ^c							Market Variable		
Date ^a	Year ^b	Total Sha Eq	Total Shareholders' Equity		Total Assets		abilities	Net Income ^d		Market	Market	Market To	
		Original	Restated	Original	Restated	Original	Restated	Original	Restated	Price	Value	Book Value	
31 Dec 1992	1992	1,004		27,326	4.	26,286		68	1	126.05	1,362.24	1.36	
31 Dec 1993	1993	1,297		31,883		30,545	C 1,4 ii	229	-	271.79	2,972.60	2.48	
31 Dec 1994	1994	1,530	-	34,214	<u>-</u>	32,650		322		245.53	2,704.02	1.77	
31 Dec 1995	1995	1,970	144	38,934	÷.	36,930	14	444	-	479.68	5,315.59	3.00	
31 Dec 1996	1996	2,339		42,138	4	39,766		576	-	629.36	7,060.57	3.30	
31 Dec 1997	1997	2,553		47,181	-	44,592	÷ .	568	-	568.96	6,448.00	2.74	
31 Dec 1998	1998	2,820	-	47,858	Carlos and	44,982	-	447	140	609.66	6,964.82	2.47 ^g	
31 Dec 1999	1999	3,366		54,132		50,697	-	328		841.18	10,187.67	3.03 ^g	
31 Dec 2000	2000	4,261	-	68,579	-	63,941	-	661	÷.	844.25	10,871.71	2.68	
31 Dec 2001	2001	5,072		73,758	-	67,997	14	434	-	717.76	9,271.55	1.90	
31 Dec 2002	2002	4,551		70,050	÷	65,344	-	224		617.98	8,258.04	1.99	
31 Dec 2003	2003	4,310		67,040	-	62,387		280	-	807.48	10,837.68	2.74	
31 Dec 2004	2004	4,393	4,742	73,656	76,465	68,765	71,221	348	394	847.75	11,419.28	2.82	
31 Dec 2005	2005	6,921	-	125,002	-	117,819	-	588	588	1,133.54	16,850.93	2.44	
31 Dec 2006	2006	8,611		135,661	135,702	126,772	126,815	571		1,305.98	20,660.14	2.40	
31 Dec 2007	2007	10,475	10 1 1	165,099	165,416	154,322	154,640	693	0	1,614.09	25,954.85	2.47	
31 Dec 2008	2008	15,399	- 0÷	302,144		286,359		1,119	-	875.00	16,587.09	1.58	

Table W.5Original and Restated Data for Accounting Variables and Market Variables for STANDARD CHARTERED PLC Surveyed on
a Yearly Basis from 1992 to 2008

			Accounting Variable ^c									
Date ^a	Year ^b	Year ^b Total Shareholders' Equity		Total	Assets	Total Li	abilities	Net Ir	ncome ^d	Market	Market	Market To
		Original	Restated	Original	Restated	Original	Restated	Original	Restated	Price	Value	Book Value
31 Dec 1992	1992	462		6,209	1.1	5,738	1	64	1.4	219.34	670.53	1.85
31 Dec 1993	1993	675	-	7,803	1.1	7,118	1. A 1	140	-	449.52	1,374.21	2.61
31 Dec 1994	1994	731	-	11,339	2.1	10,595	-	132	÷	481.39	1,471.65	2.59
31 Dec 1995	1995	849	1.2	11,652		10,789	5 <u>2</u>	139	4	674.27	2,061.31	3.14
31 Dec 1996	1996	941	-	10,329	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	9,373	-	179	-	746.73	2,282.82	3.14
31 Dec 1997	1997	1,046	1.1-1	10,608	-	9,548	-	171		942.90	2,882.52	3.58
31 Dec 1998	1998	1,171		13,519	0.00	12,335		168	- in 1	811.05	2,479.46	2.12 ^g
31 Dec 1999	1999	1,370		13,327	1.0	11,954		245	-	921.21	2,816.23	2.06 ^g
31 Dec 2000	2000	1,161	1.4	5,922	-	4,760	2	221	÷	1,321.00	2,985.46	3.38
31 Dec 2001	2001	1,113		5,544		4,431	-	-21	- (-)	845.00	1,909.89	2.27
31 Dec 2002	2002	1,044	10 1 2	4,535	÷	3,491	-	13		511.00	1,154.97	1.44
31 Dec 2003	2003	1,029	-	2,583	-	1,553	-	27	-	632.00	1,428.46	1.82
31 Dec 2004	2004	1,103	1,119	2,668	2,676	1,554	1,546	88	99	751.00	1,697.43	2.02
31 Dec 2005	2005	1,343	-	3,287	-	1,943		101	-	950.00	2,147.21	2.11
31 Dec 2006	2006	1,443	-	5,067	÷.	3,623	1.5	123	÷	1,116.00	2,522.41	2.27
31 Dec 2007	2007	1,696	19 4 9	6,834	-	5,138	1.0	160	-	1,302.00	2,942.81	2.26
31 Dec 2008	2008	1,632		7,551	4	5,919		-24	-	858.00	1,939.27	1.49

Table W.6Original and Restated Data for Accounting Variables and Market Variables for SCHRODERS PLC Surveyed on a Yearly Basis
from 1992 to 2008
				А	ccounting Varial	ole ^c					Market Variable	
Date ^a	Year ^b	Total Sha Eq	areholders' uity	Total	Assets	Total Li	abilities	Net In	ncome ^d	Market	Market	Market To
		Original	Restated	Original	Restated	Original	Restated	Original	Restated	Price	Value	Book Value
31 Dec 1992	1992	154		542	1.2.	387	-	31	12	174.31	355.77	2.31
31 Dec 1993	1993	176	-	601	1.4	425	- 1. 2 .1	42	-	293.53	606.66	3.44
31 Dec 1994	1994	200	- C	746	5.	546	4	54	2	353.66	731.34	3.65
31 Dec 1995	1995	232	10 4	826	- Al-	594	12	68	4	529.52	1,105.65	4.76
31 Dec 1996	1996	270		851		582	-	79	-	649.77	1,365.57	5.07
31 Dec 1997	1997	279		905	-	626	-	94		1,030.58	2,125.60	7.62
31 Dec 1998	1998	233		961	Cherry Cherry	728	-	105		1,143.41	2,320.86	9.97 ^g
31 Dec 1999	1999	247	1. A.	1,038	1. . .	791	-	112	-	904.39	1,803.90	7.32 ^g
31 Dec 2000	2000	252		1,217		965	-	118	÷	1,277.78	2,455.24	9.74
31 Dec 2001	2001	284	÷ .	1,418	(i)	1,134	1.2	124	4	833.33	1,585.69	5.59
31 Dec 2002	2002	392		1,825	1.14	1,433	-	76		767.44	1,458.31	3.72
31 Dec 2003	2003	449	-	1,966	-	1,517	-	85	-	840.44	1,654.51	3.68
31 Dec 2004	2004	526	415	2,090	1,996	1,565	1,581	92	85	868.22	1,714.28	3.26
31 Dec 2005	2005	317	-	1,915	-	1,597	-	-59	-	707.36	1,398.77	4.41
31 Dec 2006	2006	354		1,870	1,913	1,516	1,559	78	- A.	906.33	1,796.37	5.07
31 Dec 2007	2007	296	140	1,087	1,075	791	779	110	112	833.50	1,093.39	3.09
31 Dec 2008	2008	278	+	1,227		949		55		860.00	1,130.13	3.82

Table W.7Original and Restated Data for Accounting Variables and Market Variables for PROVIDENT FINANCIAL PLC Surveyed on
a Yearly Basis from 1992 to 2008

				А	ccounting Variab	ole ^c					Market Variable	
Date ^a	Year ^b	Total Sha Eq	areholders' juity	Total	Assets	Total Li	abilities	Net Ir	ncome ^d	Market	Market	Market To
		Original	Restated	Original	Restated	Original	Restated	Original	Restated	Price	Value	Book Value
31 Jul 1992	1992	66.42	-	416.53	-	347.48	-	8.47	-	103.07	78.20	1.18
31 Jul 1993	1993	81.45	-	574.84	-	489.18	-	11.15	-	185.04	165.73	2.03
31 Jul 1994	1994	98.22	-	714.60	-	611.92	-	19.86	-	233.63	214.87	2.19
31 Jul 1995	1995	112.07	-	923.82	-	809.40	-	21.09	-	301.36	292.33	2.61
31 Jul 1996	1996	178.55	-	1,072.63	-	892.04	-	28.67	-	330.00	397.75	2.23
31 Jul 1997	1997	195.50	-	1,371.36	-	1,173.54	-	36.13	-	422.50	515.12	2.63
31 Jul 1998	1998	206.35	-	1,645.03	-	1,436.63	-	46.24	-	847.50	1,034.80	5.01
31 Jul 1999	1999	266.16	-	1,701.72	-	1,433.45	-	50.86	-	847.50	1,089.88	4.09
31 Jul 2000	2000	367.84	-	2,637.88	-	2,262.06	-	95.24	-	1,045.00	1,410.75	3.84
31 Jul 2001	2001	407.89	-	2,755.18	-	2,342.23	-	59.10	-	755.00	1,027.82	2.52
31 Jul 2002	2002	471.92	-	3,043.29	-	2,565.28	-	22.58	-	452.50	649.94	1.38
31 Jul 2003	2003	486.53	-	3,560.95	-	3,068.30	-	26.83	-	712.50	1,026.09	2.11
31 Jul 2004	2004	509.26	-	3,866.07	-	3,352.13	-	29.74	-	667.00	962.24	1.89
31 Jul 2005	2005	540.32	572.51	4,517.56	4,726.34	3,971.27	4,147.96	27.39	27.67	757.50	1,095.61	2.03
31 Jul 2006	2006	654.99	-	4,787.53	-	4,125.17	-	55.79	-	864.50	1,265.12	1.93
31 Jul 2007	2007	745.37	-	5,347.17	-	4,594.51	-	64.10	-	814.50	1,200.02	1.61
31 Jul 2008	2008	715.40	-	5,723.70	-	5,003.30	-	43.30	-	601.00	887.56	1.24 ^g

Table W.8Original and Restated Data for Accounting Variables and Market Variables for CLOSE BROTHERS GROUP PLC Surveyed
on a Yearly Basis from 1992 to 2008

				А	ccounting Variab	ble ^c					Market Variable	1
Date ^a	Year ^b	Total Sha Eq	areholders' juity	Total	Assets	Total Li	abilities	Net Ir	ncome ^d	Market	Market	Market To
		Original	Restated	Original	Restated	Original	Restated	Original	Restated	Price	Value	Book Value
30 Sep 1992	1992	20.28	12	37.51		17.22	1.20	1.28	-	19.95	26.10	1.29
30 Sep 1993	1993	21.29	1	38.24	-	16.95		2.07	-	38.29	50.10	2.35
30 Sep 1994	1994	24.74	-	38.56	-	13.83	1.1	3.39		48.58	63.57	2.57
30 Sep 1995	1995	26.78	-	113.58		86.79	-22	2.72		64.03	84.49	3.16
30 Sep 1996	1996	29.67	-	152.18	-	122.47		4.37	-	54.05	73.40	2.47
30 Sep 1997	1997	79.72	-	211.85	-	132.14	-	2.31	- 1	66.92	91.08	1.15
30 Sep 1998	1998	76.16	-	200.72	-	124,56	47	6.02	-	59.84	135.91	1.79
30 Sep 1999	1999	80.90	-	235.92	-	155.03	1.4.0	10.49	-	99.74	234.20	2.91
30 Sep 2000	2000	98.82		273.54	-	174.72	-	24.13	-	367.75	884.31	8.99
30 Sep 2001	2001	217.42	-	643.29		425.66	201	13.70	C-	205.91	554.26	2.98
30 Sep 2002	2002	207.15	-	810.08	-	602.47		1.02	-	49.87	135.98	0.73
30 Sep 2003	2003	178.60	-	707.03	-	527.68	-	-45.89	-	45.37	124.72	0.74
30 Sep 2004	2004	132.92	-	690.57	-	557.37	2	-68.95		55.98	205.49	1.55
30 Sep 2005	2005	419.85	430.30	3,282.82	3,326.70	2,862.97	2,896.40	1.73	12.48	114.00	689.72	1.74
30 Sep 2006	2006	450.10	-	2,214,29		1,764.19	-	26.18	25.84	163.75	1,025.21	2.32
30 Sep 2007	2007	422.10		2,254.20	14	1,634.29	-2-	37.98	-	183.00	1,150.77	2.78
30 Sep 2008	2008	545.36		2,346.92		1,603.62		10.53	-	126.00	903.40	2.18

Table W.9Original and Restated Data for Accounting Variables and Market Variables for ABERDEEN ASSET MANAGEMENT PLC
Surveyed on a Yearly Basis from 1992 to 2008

				А	ccounting Varial	ble ^c					Market Variable	
Date ^a	Year ^b	Total Sha Eq	areholders' juity	Total A	Assets	Total Liz	abilities	Net Ir	acome ^d	Market	Market	Market To
		Original	Restated	Original	Restated	Original	Restated	Original	Restated	Price	Value	Book Value
30 Sep 1992	1992	11.20	-	1,409.90	-	1,398.70	-	-146.80	-	45.87	3.10	0.28 ^g
30 Sep 1993	1993	-36.10	-	2,698.30	-	2,734.40	-	-44.59	-	141.43	9.56	-0.07
30 Sep 1994	1994	-24.80	-	1,831.20	-	1,856.00	-	3.91	-	248.45	16.79	-0.14
30 Sep 1995	1995	39.60	-	1,555.20	-	1,515.60	-	15.10	-	151.37	86.94	2.20
30 Sep 1996	1996	47.30	-	1,299.40	-	1,252.10	-	18.10	-	145.25	83.43	1.70
30 Sep 1997	1997	61.30	-	1,232.10	-	1,170.80	-	21.60	-	275.21	158.08	2.51
30 Sep 1998	1998	84.00	-	1,602.70	-	1,518.70	-	23.90	-	238.32	172.33	1.99
30 Sep 1999	1999	110.70	-	1,670.60	-	1,559.90	-	30.30	-	334.76	243.87	2.15
30 Sep 2000	2000	132.90	-	1,866.50	-	1,733.60	-	28.50	-	240.71	175.68	1.28
30 Sep 2001	2001	161.30	-	2,397.80	-	2,236.50	-	32.30	-	353.89	258.85	1.56
30 Sep 2002	2002	191.50	-	2,784.80	-	2,593.30	-	20.20	-	274.98	203.63	1.01
30 Sep 2003	2003	224.10	-	3,439.00	-	3,196.10	-	22.00	-	532.43	397.80	1.57
30 Sep 2004	2004	268.40	-	4,948.00	-	4,665.60	-	29.20	-	543.59	408.83	1.45
30 Sep 2005	2005	308.00	312.80	5,998.20	7,086.80	5,680.30	6,774.00	32.20	29.90	846.47	631.74	1.99
30 Sep 2006	2006	279.00	-	9,083.10	-	8,804.10	-	36.30	-	1,074.42	783.24	2.46
30 Sep 2007	2007	313.30	-	12,081.10	-	11,767.80	-	32.10	-	478.63	344.68	1.08
30 Sep 2008	2008	621.50	-	11,499.40	-	10,877.90	-	18.70	-	65.00	194.02	0.62

Table W.10Original and Restated Data for Accounting Variables and Market Variables for PARAGON GROUP OF COMPANIES PLC
Surveyed on a Yearly Basis from 1992 to 2008

				А	ccounting Varial	ble ^c]	Market Variable	
Date ^a	Year ^b	Total Sha Eq	areholders' uity	Total A	Assets	Total Li	abilities	Net Ir	ncome ^d	Market	Market	Market To
		Original	Restated	Original	Restated	Original	Restated	Original	Restated	Price	Value	Book Value
31 Dec 1992	1992	9.20	-	52.04	-	42.84	-	2.76	-	246.00	38.57	4.19
31 Dec 1993	1993	11.79	-	80.47	-	68.67	-	3.46	-	316.00	58.69	4.98
31 Dec 1994	1994	14.20	-	63.72	-	49.51	-	3.95	-	245.00	47.89	3.37
31 Dec 1995	1995	17.40	-	72.92	-	55.46	-	4.87	-	329.00	77.63	4.46
31 Dec 1996	1996	21.81	-	117.57	-	95.77	-	6.31	-	307.50	84.41	3.87
31 Dec 1997	1997	25.20	-	142.79	-	117.59	-	7.05	-	400.00	114.72	4.55
31 Dec 1998	1998	40.41	-	200.32	-	159.90	-	12.22	-	540.00	170.73	4.22 ^g
31 Dec 1999	1999	51.44	-	253.14	-	201.70	-	15.53	-	920.00	318.68	6.2 ^g
31 Dec 2000	2000	70.68	-	405.63	-	334.95	-	16.87	-	956.00	342.26	4.84
31 Dec 2001	2001	75.43	-	521.40	-	445.97	-	11.68	-	892.50	323.80	4.29
31 Dec 2002	2002	100.43	-	542.36	-	441.93	-	2.09	-	555.00	218.54	2.18
31 Dec 2003	2003	105.90	-	513.92	-	408.02	-	5.11	-	740.50	301.15	2.84
31 Dec 2004	2004	110.59	117.44	586.62	600.98	476.03	483.54	6.69	10.48	852.50	348.41	3.15
31 Dec 2005	2005	130.42	-	687.67	-	557.25	-	14.26	-	992.00	409.36	3.14
31 Dec 2006	2006	159.15	-	907.30	-	748.15	-	16.05	-	1,198.00	506.48	3.18
31 Dec 2007	2007	184.75	-	1,223.10	-	1,038.35	-	18.54	-	1,050.00	448.24	2.82
31 Dec 2008	2008	184.63	-	1,307.96	-	1,123.33	-	1.50	-	833.50	356.81	1.93

Table W.11Original and Restated Data for Accounting Variables and Market Variables for RATHBONE BROTHERS PLC Surveyed on a
Yearly Basis from 1992 to 2008

				A	ecounting Variab	le ^c				1	Market Variable	
Date ^a	Year ^b	Total Sha Eq	reholders' uity	Total	Assets	Total Li	abilities	Net Ir	ncome ^d	Market	Market	Market To
		Original	Restated	Original	Restated	Original	Restated	Original	Restated	Price	Value	Book Value
30 Sep 1992	1992	470.79		566.10	4.	94.74	-	11.99	-	169.00	292.11	0.61
30 Sep 1993	1993	580.61	-	713.20	-	131.16	-	13.05	-	281.00	485.70	0.82
30 Sep 1994	1994	643.12	1.0	790.79	-	145.16		12.98		318.50	551.22	0.84
30 Sep 1995	1995	747.79	1.1	897.37		149.59		10.84	-	368.00	636.98	0.84
30 Sep 1996	1996	886.21		1,051.41	-	165.20	-	18.49		412.50	714.04	0.79
30 Sep 1997	1997	1,082.80	· -	1,244.68	-	161.87	-	24.55		483.00	836.07	0.75
30 Sep 1998	1998	1,145,32	-	1,377.20	4	231.88	140	27.75		512.00	886.25	0.76
30 Sep 1999	1999	987.46	-	1,284.74	÷	297.28		-5.73	-	836.00	868.24	0.88
30 Sep 2000	2000	874.04		1,137.32	-	263.27	-	-19.46		1,034.00	832.99	0.95
30 Sep 2001	2001	541.11		787.92	-	246.81	-	-17.53		651.00	424.66	0.78
30 Sep 2002	2002	498.33	1. Contraction	737.50	-	239.17		-1.29		462.50	301.68	0.61
30 Sep 2003	2003	495.50		722.25	-	226.75	-	0.29	-	633.50	413.24	0.83
30 Sep 2004	2004	426.72	-	616.52	-	189.79	-	-0.34	-	793.50	370.93	0.87
30 Sep 2005	2005	512.18	520.88	711.35	÷	199.17	190.47	27.52	27.05	1,113.00	484.24	0.95
30 Sep 2006	2006	598.29	12	791.59	->	193.30	-	7.34	0-0	1,371.00	530.89	0.89
30 Sep 2007	2007	745.51	1.4.1	965.89		220.38		7.19	- 	1,680.00	625.85	1.05
30 Sep 2008	2008	640.95	-	837.55		196.60	-	-2.94		1,235.00	439.61	0.59

Table W.12Original and Restated Data for Accounting Variables and Market Variables for ELECTRA PRIVATE EQUITY PLC Surveyed
on a Yearly Basis from 1992 to 2008

				А	ccounting Variab	ole ^c					Market Variable	5
Date ^a	Year ^b	Total Sha Eq	areholders' uity	Total	Assets	Total Li	abilities	Net In	come ^d	Market	Market	Market To
		Original	Restated	Original	Restated	Original	Restated	Original	Restated	Price	Value	Book Value
31 Oct 1992	1992	208.32	1	243.42		35.09		5.64		118.00	182.76	0.88
31 Oct 1993	1993	285.61	-	327.84	-	42.23	-	6.56	-	185.00	286.53	1.01
31 Oct 1994	1994	279.60	-	329.37	-	49.78	1.1.2	6.61	-	185.00	286.97	1.03
31 Oct 1995	1995	311.13	-	365.29		54.16		7,72	04 mm	206.50	320.32	1.03
31 Oct 1996	1996	349.25	-	405.41	-	56.16	C-COL	8.97	-	217.00	336.61	0.97
31 Oct 1997	1997	406.91		454.10	-	47.20	Cert I	9.52	-	243.50	377.72	0.93
31 Oct 1998	1998	422.42	-	479.93	-	57.50	19 C - 19	10.23	-	234.25	363.37	0.86
31 Oct 1999	1999	487.46	-	537.43	-	49.97	0.800	9.58	-	273.50	421.79	0.87
31 Oct 2000	2000	506.91	-	558.08	-	51.17		10.59	-	309.50	435.68	0.86
31 Oct 2001	2001	377.76	-	423.39	-	45.63	0-mil	8.86	0.00	280.50	353.90	0.94
31 Oct 2002	2002	314.59	-	366.13	-	51.54	÷	4.47		233.75	294.92	0.94
31 Oct 2003	2003	357.15	-	409.20	-	52.05		4.57	-	265.50	334.98	0.94
31 Oct 2004	2004	375.56	-	410.85	-	35.29	-	5.38	0-1	262.00	323.65	0.86
31 Oct 2005	2005	430.67	435.62	482.05	481.82	51.37	46.19	5.83	-	313.25	371.39	0.86
31 Oct 2006	2006	515.36	-	541.61	-	26.25	-	5.99	-	385.50	448.96	0.87
31 Oct 2007	2007	571.02		598.96	-	27.94	-	7.43		439.00	501.23	0.97
31 Oct 2008	2008	385.66	-	412.73	-	27.07		8.09	1-1	305.00	345.14	0.60

Table W.13 Original and Restated Data for Accounting Variables and Market Variables for BANKERS INVESTMENT TRUST PLCSurveyed on a Yearly Basis from 1992 to 2008

				А	ccounting Variab	le ^c					Market Variable	
Date ^a	Year ^b	Total Sha Eq	areholders' uity	Total	Assets	Total Li	abilities	Net Ir	ncome ^d	Market	Market	Market To
		Original	Restated	Original	Restated	Original	Restated	Original	Restated	Price	Value	Book Value
31 Dec 1992	1992	775.98	-	839.35	-	63.38	-	19.53	-	185.00	632.88	0.82
31 Dec 1993	1993	1,022.27	-	1,145.16	-	122.90	-	22.45	-	245.00	919.49	0.90
31 Dec 1994	1994	942.63	-	1,079.72	-	137.09	-	23.26	-	230.00	863.20	0.92
31 Dec 1995	1995	1,093.96	-	1,217.76	-	123.80	-	25.36	-	265.00	994.56	0.91
31 Dec 1996	1996	1,182.71	-	1,294.23	-	111.52	-	26.68	-	266.50	1,027.42	0.87
31 Dec 1997	1997	1,406.37	-	1,531.29	-	124.92	-	30.38	-	319.50	1,199.10	0.85
31 Dec 1998	1998	1,593.62	-	1,725.60	-	131.98	-	30.23	-	367.00	1,377.36	0.86 ^g
31 Dec 1999	1999	2,085.47	-	2,149.76	-	64.29	-	28.19	-	477.50	1,773.93	0.85 ^g
31 Dec 2000	2000	1,836.68	-	1,998.75	-	162.07	-	27.06	-	478.00	1,681.05	0.92
31 Dec 2001	2001	1,512.22	-	1,676.00	-	163.77	-	29.55	-	391.00	1,375.07	0.91
31 Dec 2002	2002	1,080.26	-	1,242.70	-	162.44	-	13.04	-	261.50	916.13	0.85
31 Dec 2003	2003	1,232.94	-	1,395.94	-	163.01	-	13.58	-	303.00	1,040.87	0.85
31 Dec 2004	2004	1,309.48	1,325.19	1,486.97	1,481.86	177.49	156.67	11.96	11.93	331.50	1,123.60	0.86
31 Dec 2005	2005	1,387.47	-	1,536.54	-	149.07	-	11.38	-	414.00	1,221.25	0.88
31 Dec 2006	2006	1,351.09	-	1,503.53	-	152.44	-	11.08	-	454.50	1,185.21	0.88
31 Dec 2007	2007	1,221.86	-	1,371.49	-	149.63	-	10.45	-	478.50	1,069.09	0.79
31 Dec 2008	2008	881.80	-	1,024.23	-	142.42	-	9.48	-	351.00	752.94	0.62

Table W.14Original and Restated Data for Accounting Variables and Market Variables for WITAN INVESTMENT TRUST PLC
Surveyed on a Yearly Basis from 1992 to 2008

				A	ccounting Variab	le ^c					Market Variable	
Date ^a	Year ^b	Total Sha Eq	areholders' uity	Total A	Assets	Total Li	abilities	Net Ir	ncome ^d	Market	Market	Market To
		Original	Restated	Original	Restated	Original	Restated	Original	Restated	Price	Value	Book Value
31 Mar 1993	1992	349.79	-	485.58	-	135.63	-	1.44	-	117.00	209.60	0.59
31 Mar 1994	1993	427.08	-	531.61	-	104.53	-	5.46	-	171.00	308.44	0.72
31 Mar 1995	1994	419.94	-	585.96	-	166.02	-	-2.36	-	174.00	314.41	0.75
31 Mar 1996	1995	543.67	-	759.67	-	216.00	-	2.72	-	223.00	402.96	0.74
31 Mar 1997	1996	573.96	-	743.93	-	169.97	-	6.13	-	242.50	439.17	0.77 ^g
31 Mar 1998	1997	726.68	-	924.31	-	197.63	-	1.39	-	327.00	592.50	0.82
31 Mar 1999	1998	751.07	-	888.77	-	137.71	-	4.67	-	341.00	620.48	0.76
31 Mar 2000	1999	811.39	-	875.83	-	64.44	-	13.81	-	439.00	699.87	0.86
31 Mar 2001	2000	753.19	-	803.42	-	50.23	-	16.86	-	436.50	684.64	0.91
31 Mar 2002	2001	758.28	-	814.11	-	55.83	-	0.57	-	424.50	665.82	0.88 ^g
31 Mar 2003	2002	674.71	-	713.84	-	39.14	-	2.04	-	371.50	582.69	0.86
31 Mar 2004	2003	977.47	-	1,103.80	-	126.29	-	-9.46	-	577.50	905.80	0.93
31 Mar 2005	2004	1,109.21	1,113.12	1,239.60	1,230.03	130.25	116.77	-1.49	-2.11	694.00	1,083.88	0.98
31 Mar 2006	2005	1,534.70	-	1,846.19	-	311.50	-	-12.51	-	1,020.00	1,593.01	1.04
31 Mar 2007	2006	1,635.60	-	2,008.60	-	373.00	-	-11.00	-	1,000.00	1,561.78	0.95
31 Mar 2008	2007	1,690.00	-	2,141.40	-	451.40	-	73.90	-	1,147.00	1,775.82	1.09
31 Mar 2009	2008	1,350.50	-	1,767.00	-	416.50	-	34.40	-	831.00	1,283.61	0.76

Table W.15Original and Restated Data for Accounting Variables and Market Variables for RIT CAPITAL PARTNERS PLC Surveyed on
a Yearly Basis from 1992 to 2008

				А	ccounting Variab	le ^c					Market Variable	
Date ^a	Year ^b	Total Sha Eq	areholders' uity	Total	Assets	Total Li	abilities	Net Ir	acome ^d	Market	Market	Market To
		Original	Restated	Original	Restated	Original	Restated	Original	Restated	Price	Value	Book Value
31 Jan 1993	1992	898.96	-	920.72	-	21.76	-	23.03	-	158.90	800.86	0.89
31 Jan 1994	1993	1,077.29	-	1,104.27	-	26.98	-	23.83	-	203.90	1,027.66	0.96
31 Jan 1995	1994	948.13	-	1,003.52	-	49.71	-	26.83	-	169.80	855.79	0.90
31 Jan 1996	1995	1,226.00	-	1,285.98	-	53.20	-	30.99	-	213.00	1,073.52	0.88
31 Jan 1997	1996	1,357.14	-	1,421.97	-	57.87	-	30.16	-	229.00	1,154.16	0.85
31 Jan 1998	1997	1,562.97	-	1,633.65	-	63.58	-	32.90	-	260.50	1,312.92	0.84
31 Jan 1999	1998	1,728.24	-	1,797.44	-	61.87	-	33.48	-	293.60	1,479.74	0.86
31 Jan 2000	1999	1,886.71	-	1,969.87	-	73.58	-	34.71	-	305.25	1,538.46	0.82
31 Jan 2001	2000	1,973.93	-	2,060.07	-	76.40	-	34.11	-	335.75	1,692.18	0.86
31 Jan 2002	2001	1,672.32	-	1,768.27	-	86.28	-	16.16	-	292.40	1,473.70	0.88
31 Jan 2003	2002	1,204.26	-	1,306.24	-	92.41	-	15.12	-	212.75	1,072.26	0.89
31 Jan 2004	2003	1,474.62	-	1,583.20	-	99.03	-	16.95	-	260.50	1,312.92	0.89
31 Jan 2005	2004	1,623.29	1,634.69	1,742.28	1,737.75	109.15	93.23	16.33	-	274.10	1,381.46	0.85
31 Jan 2006	2005	2,037.92	-	2,207.19	-	155.20	-	16.38	-	349.00	1,758.96	0.86
31 Jan 2007	2006	2,833.87	-	3,000.85	-	160.69	-	21.18	-	365.50	2,455.83	0.86
31 Jan 2008	2007	2,704.79	-	3,109.51	-	397.34	-	23.94	-	338.00	2,271.06	0.80
31 Jan 2009	2008	2,122.98	-	2,414.36	-	284.65	-	18.52	-	268.00	1,800.72	0.66

Table W.16Original and Restated Data for Accounting Variables and Market Variables for ALLIANCE TRUST PLC Surveyed on a
Yearly Basis from 1992 to 2008

APPENDIX X CONTROL GROUP DATA

X.1 Control Group Data

This appendix presents in Table X.1 to Table X.12 the data collected for banking related firms registered with the LSE's UK financial services sector. These firms represent the Control group in this research. Summary information for these firms is presented in Table 3.7, labeled 'Control' in the 'Group' column.

For each table presented the following superscript letters represent the corresponding details:

^a The fiscal year-end financial statement reporting date.

^b The year used to group the variable.

^c Reported at the scale 1,000,000 GBP (1 Million GBP).

^d Net income amounts presented from 1992 to 2001 are the full year figures, and from 2002 to 2008 are the second half-year figures. This in accordance with the convention applied in this study to test the net income variable.

^e The end of day market price of the firm represented using the scale: 0.01GBP (Great British Pound), equivalent to 1GBX.

^f Market Value to Book Value ratio is calculated is dividing the variables Market Value by Total Shareholders' Equity.

^g Market Value to Book Value ratio is calculated is dividing the variables Common Shares Outstanding and Market Price by Total Shareholders' Equity.

				А	ccounting Variab	ole ^c					Market Variable	
Date ^a	Year ^b	Total Sha Eq	areholders' uity	Total	Assets	Total Li	abilities	Net In	come ^d	Market	Market	Market To
		Original	Restated	Original	Restated	Original	Restated	Original	Restated	Price ^e	Value	Book Value
31 Jul 1992	1992	10.74		11.44		0.70		0.10	-	24.00	9.81	0.91
31 Jul 1993	1993	17.07	-	17.27	-	0.20		0.04	-	38.00	15.54	0.91
31 Jul 1994	1994	20.30	-	20.44	-	0.14		0.11	-	45.50	18.60	0.92
31 Jul 1995	1995	19.56		19.83	Constant of the	0.26	2	0.20	ciére i la	44.50	18.20	0.93
31 Jul 1996	1996	40.27	-	40.74		0.47	-	0.35	-	48.00	37.84	0.94
31 Jul 1997	1997	46.89		47.39	-	0.50	-	0.40	-	51.00	40.21	0.86
31 Jul 1998	1998	22.59	-	24.02	· -	1.43	4	0.89	- E (23.25	18.33	0.81
31 Jul 1999	1999	34.95		38.44	-	3.49	-	0.23	-	37.00	29.17	0.83
31 Jul 2000	2000	43.64	-	54.18	-	10.54		0.18	-	46.75	36.86	0.84
31 Jul 2001	2001	37.71	-	38.33		0.62	0.00	0.51	i circi	38.50	30.35	0.80
31 Jul 2002	2002	38.73	-	39.39	-	0.66	-	0.27	-	45.50	34.80	0.90
31 Jul 2003	2003	40.80	-	41.43	-	0.63	-	0.30	-	48.25	36.90	0.90
31 Jul 2004	2004	47.31	-	48.27	-	0.96		0.53	-	56.75	43.41	0.92
31 Jul 2005	2005	66.80	67.37	68.30	68.03	1.50	0.65	0.71	-	85.50	65.40	0.98
31 Jul 2006	2006	81.39		89.78	-	8.39	14.	0.73	-	101.25	77.44	0.95
31 Jul 2007	2007	130.08		130.85		0.77	10.0	0.68		171.75	133.86	1.03
31 Jul 2008	2008	105.17	-	105.79	4	0.62		1.04	-	128.00	100.21	0.95 ^g

Table X.1Original and Restated Data for Accounting Variables and Market Variables for PACIFIC HORIZON INVESTMENT TRUST
PLC Surveyed on a Yearly Basis from 1992 to 2008

				А	ccounting Variab	ole ^c					Market Variable	
Date ^a	Year ^b	Total Sha Eq	areholders' juity	Total	Assets	Total Li	abilities	Net Ir	come ^d	Market	Market	Market To
		Original	Restated	Original	Restated	Original	Restated	Original	Restated	Price	Value	Book Value
31 Jan 1993	1992	19.76	-	19.84	-	0.08	-	-0.13	-	92.00	14.75	0.75
31 Jan 1994	1993	28.82	-	29.08	-	0.26	-	-0.13	-	195.00	31.26	1.08
31 Jan 1995	1994	42.61	-	42.75	-	0.14	-	-0.25	-	137.00	39.33	0.92
31 Jan 1996	1995	43.03	-	50.94	-	7.91	-	-0.22	-	137.50	39.66	0.92
31 Jan 1997	1996	30.86	-	38.90	-	8.04	-	-0.40	-	84.50	26.88	0.87
31 Jan 1998	1997	23.88	-	31.76	-	7.88	-	-0.29	-	63.25	20.12	0.84
31 Jan 1999	1998	25.47	-	34.38	-	8.90	-	0.09	-	60.00	19.09	0.75
31 Jan 2000	1999	92.39	-	112.33	-	19.94	-	-0.73	-	246.00	78.25	0.85
31 Jan 2001	2000	50.00	-	61.82	-	11.82	-	-0.80	-	130.00	40.88	0.82
31 Jan 2002	2001	38.18	-	48.58	-	10.40	-	-0.28	-	100.50	31.05	0.81
31 Jan 2003	2002	28.29	-	35.68	-	7.39	-	-0.24	-	79.50	24.57	0.87
31 Jan 2004	2003	44.11	-	52.01	-	7.90	-	-0.19	-	126.75	39.17	0.89
31 Jan 2005	2004	54.24	-	68.90	-	14.67	-	-0.26	-	165.75	50.72	0.94
31 Jan 2006	2005	87.08	-	102.10	-	15.02	-	-0.33	-	304.00	93.02	1.07
31 Jan 2007	2006	65.18	-	77.77	-	12.59	-	-0.15	-	205.00	62.73	0.96
31 Jan 2008	2007	45.16	-	56.79	-	11.63	-	-0.02	-	129.25	40.20	0.62
31 Jan 2009	2008	38.26	-	49.34	-	11.08	-	0.23	-	95.75	29.78	0.66

Table X.2Original and Restated Data for Accounting Variables and Market Variables for BAILLIE GIFFORD SHIN NIPPON PLC
Surveyed on a Yearly Basis from 1992 to 2008

				A	ecounting Variab	ole ^c					Market Variable	
Date ^a	Year ^b	Total Sha Eq	areholders' juity	Total	Assets	Total Li	abilities	Net In	ncome ^d	Market	Market	Market To
		Original	Restated	Original	Restated	Original	Restated	Original	Restated	Price ^e	Value	Book Value
31 Oct 1992	1992	37.52	1	38.98		1.46		1.82		43.80	36.79	0.99
31 Oct 1993	1993	51.39	1	53.07	-	1.68		1.47	-	58.00	48.72	0.96
31 Oct 1994	1994	52.93		59.61	-	6.67		1.51	4	61.20	51.41	0.97
31 Oct 1995	1995	58.85	-	65.18	-	6.33	- 22	1.31	1	65.20	54.77	0.93
31 Oct 1996	1996	67.61		74.16	-	6.55		1.60	-	69.20	58.13	0.86
31 Oct 1997	1997	70.38	-	87.71	-	17.33		1.98	-	69.30	58.21	0.83
31 Oct 1998	1998	60.14	1.1	77.23	-	17.10	2	2.46	-	58.40	49.06	0.82
31 Oct 1999	1999	85.51		103.17	-	17.66		2.58	-	76.70	64.43	0.75
31 Oct 2000	2000	112.71	-	130.92	-	18.21	-	2.51	-	104.00	86.84	0.77
31 Oct 2001	2001	80.23	-	102.94	-	22.71		2.24	- Art	78.20	65.30	0.82
31 Oct 2002	2002	57.22	-	74.41	-	17.18	÷.	0.82	3	52.80	44.09	0.77
31 Oct 2003	2003	74.75	-	98.77	-	24.02	-	0.99	-	68.30	56.51	0.76
31 Oct 2004	2004	83.63	-	101.63	-	18.00	4	1.25		80.60	65.58	0.79
31 Oct 2005	2005	98.77	100.52	116.66	116.62	17.89	16.10	1.45	-	102.60	83.49	0.85
31 Oct 2006	2006	131.40	-	147.46	-	16.06	-	1,70	-	144.50	117.58	0.89
31 Oct 2007	2007	82.36	-	92.94	1.60	10.57	-	1.34	100	142.50	70.29	0.85
31 Oct 2008	2008	43.17	-	51.38		8.21		1.43	-	73.75	35.34	0.43

Table X.3Original and Restated Data for Accounting Variables and Market Variables for DUNEDIN SMALLER COMPANIES
INVESTMENT TRUST Surveyed on a Yearly Basis from 1992 to 2008

				А	ccounting Variab	ole ^c					Market Variable	
Date ^a	Year ^b	Total Sha Eq	areholders' juity	Total	Assets	Total Liabilities		Net Income ^d		Market	Market	Market To
		Original	Restated	Original	Restated	Original	Restated	Original	Restated	Price	Value	Book Value
31 Dec 1992	1992	353.29	-	478.16	-	124.87	-	12.90	-	278.00	324.46	0.92
31 Dec 1993	1993	460.58	-	596.93	-	136.35	-	13.14	-	392.00	464.11	1.03
31 Dec 1994	1994	397.32	-	533.17	-	135.85	-	15.96	-	332.00	393.07	1.01
31 Dec 1995	1995	477.04	-	606.35	-	129.31	-	16.20	-	381.50	451.97	0.96
31 Dec 1996	1996	512.34	-	596.91	-	84.57	-	17.13	-	406.50	482.11	0.95
31 Dec 1997	1997	580.69	-	672.07	-	91.39	-	17.92	-	445.00	528.18	0.92
31 Dec 1998	1998	626.86	-	730.09	-	103.23	-	18.38	-	436.00	517.83	0.83 ^g
31 Dec 1999	1999	653.62	-	821.09	-	167.47	-	16.70	-	489.00	540.48	0.83 ^g
31 Dec 2000	2000	601.16	-	739.20	-	138.03	-	14.31	-	500.00	518.08	0.87
31 Dec 2001	2001	419.48	-	539.91	-	120.43	-	12.26	-	403.00	356.30	0.86
31 Dec 2002	2002	308.75	-	414.71	-	105.96	-	4.70	-	298.00	257.38	0.84
31 Dec 2003	2003	371.39	-	481.13	-	109.74	-	4.33	-	388.00	335.28	0.90
31 Dec 2004	2004	408.82	413.32	522.08	521.43	113.26	108.11	6.10	-	436.50	377.54	0.94
31 Dec 2005	2005	523.63	-	614.19	-	90.56	-	6.58	-	562.00	486.08	0.93
31 Dec 2006	2006	579.27	-	663.20	-	83.93	-	7.14	-	618.00	534.92	0.94
31 Dec 2007	2007	646.24	-	728.64	-	82.40	-	8.05	-	662.00	573.19	1.00
31 Dec 2008	2008	568.83	-	739.72	-	170.89	-	9.69	-	589.00	527.11	0.83

Table X.4Original and Restated Data for Accounting Variables and Market Variables for MURRAY INTERNATIONAL TRUST PLC
Surveyed on a Yearly Basis from 1992 to 2008

				А	ccounting Varial	ole ^c					Market Variable	
Date ^a	Year ^b	Total Sha Eq	areholders' juity	Total Assets		Total Liabilities		Net Income ^d		Market	Market	Market To
		Original	Restated	Original	Restated	Original	Restated	Original	Restated	Price ^e	Value	Book Value
30 Nov 1992	1992	149.30	2	280.68		130.92	4	3.67		46.50	131.20	0.78
30 Nov 1993	1993	221.03	-	314.58	-	93.11		10.81	-	73.00	221.77	0.91
30 Nov 1994	1994	215.53		280.86	-	64.82	1.1	7.19	2	74.50	226.33	0.94
30 Nov 1995	1995	232.66	-	296.14		62.98	- 27	5.45	- in the second s	80.00	243.04	0.95
30 Nov 1996	1996	275.13	-	332.18	-	57.05	-	6.36	-	77.75	236.20	0.86
30 Nov 1997	1997	290.35	-	350.86	1 - C	60.51		6.88	-	84.50	256.69	0.89
30 Nov 1998	1998	252.99	-	312.56	-	59.57		7.44		67.75	205.81	0.82
30 Nov 1999	1999	328.01	-	388.90		60.89	1.00	7.45	-	89.25	254.15	0.77
30 Nov 2000	2000	283.72	-	337.32	(-)	53.61		4.47	-	94.25	233.99	0.82
30 Nov 2001	2001	245.21		295.80	-	50.60	0400	3.36	- Circle	81.00	192.54	0.79
30 Nov 2002	2002	183.83	-	233.55	-	49.73		1.84	-	59.00	139.98	0.76
30 Nov 2003	2003	243.44	-	285.14	-	41.71	-	2.17	-	84.00	194.88	0.80
30 Nov 2004	2004	311.03	-	349.15		38.12	-	2.42	0-0	110.25	253.31	0.81
30 Nov 2005	2005	316.26	313.38	357.04	351.88	40.78	38.49	2.19	-	142.00	259.30	0.82
30 Nov 2006	2006	326.21	-	362.89	-	36.68	-	2,26		164.25	268.68	0.82
30 Nov 2007	2007	272.51	-	311.80	-	39.29	-	1.54	01	152.00	212.89	0.78
30 Nov 2008	2008	77.03	-	102.18	+	25.15		3.36	-	62.75	51.68	0.19

Table X.5Original and Restated Data for Accounting Variables and Market Variables for THROGMORTON TRUST PLC Surveyed on
a Yearly Basis from 1992 to 2008

				А	ccounting Varial	ble ^c					Market Variable	:
Date ^a	Year ^b	Total Sha Eq	ureholders' uity	Total	Assets	Total Li	abilities	Net In	come ^d	Market	Market	Market To
		Original	Restated	Original	Restated	Original	Restated	Original	Restated	Price ^e	Value	Book Value
30 Sep 1992	1992	350.28	2	649.68		207.66	4	15.24	-	79.50	305.28	0.88
30 Sep 1993	1993	424.25	-	781.28	-	236.05		16.12	-	105.50	405.12	0.97
30 Sep 1994	1994	396.99	-	748.59	-7	234.28	121	16.12	0.2	94.00	360.96	0.92
30 Sep 1995	1995	445.21	-	841.79		263.24		15.49		92.50	298.37	0.84
30 Sep 1996	1996	469.07	-	704.91	-	235.84		18.53	-	97.50	314.50	0.68
30 Sep 1997	1997	548.97	-	617.29	-	68.32	1.0	17.28	-	119.50	458.88	0.84
30 Sep 1998	1998	520.17	-	617.47	-	97.30	4	18.77		113.75	366.91	0.71
30 Sep 1999	1999	578.06	-	681.46	-	103.40	-	19.12	-	135.25	430.84	0.75
30 Sep 2000	2000	642.23	-	761.72	-	119.49		18.21	-	158.50	486.63	0.89
30 Sep 2001	2001	492.42	-	648.88	-	156.46	0400	21.66	10 ÷ 1	118.50	359.59	0.73
30 Sep 2002	2002	334.31	-	486.67	-	152.36	-	7.63		90.25	318.91	0.95
30 Sep 2003	2003	370.82	-	501.91	-	131.08	-	7.91	-	98.75	348.94	0.94
30 Sep 2004	2004	386.18	-	517.03	-	130.85	4	7.14	0.00	101.00	341.19	0.88
30 Sep 2005	2005	452.84	460.86	587.75	586.98	134.91	126.12	8.04	-	126.50	408.99	0.90
30 Sep 2006	2006	470.39	12	595.65	-	125.26	-	12.08	÷.	132.50	408.78	0.87
30 Sep 2007	2007	485.77	-	612.01	1.00	126.24		8.98	100	140.75	423.67	0.87
30 Sep 2008	2008	333.52	-	424.06	-	90.55		9.99	-	100.75	294.81	0.61

Table X.6Original and Restated Data for Accounting Variables and Market Variables for BRITISH ASSETS TRUST PLC Surveyed on a
Yearly Basis from 1992 to 2008

				А	ccounting Variab	le ^c					Market Variable	
Date ^a	Year ^b	Total Sha Eq	areholders' uity	Total	Assets	Total Li	abilities	Net II	ncome ^d	Market	Market	Market To
		Original	Restated	Original	Restated	Original	Restated	Original	Restated	Price	Value	Book Value
31 Mar 1993	1992	885.62		1,122.65	4	237.03	4.	27.16		264.00	756.45	0.86
31 Mar 1994	1993	1,006.85	-	1,239.09	-	232.24	4.1	29.90	-	306.50	900.56	0.90
31 Mar 1995	1994	941.60		1,180.77	2	239.17		30.46	-	292.50	859.42	0.91
31 Mar 1996	1995	1,126.63	2	1,286.43		159.81		33.88	-	330.50	971.07	0.86
31 Mar 1997	1996	1,235.96	-	1,368.26	-	132.30	-	36.91	-	364.50	1,070.97	0.87 ^g
31 Mar 1998	1997	1,653.00	-	1,900.26	-	247.26	-	42.53	-	500.50	1,470.55	0.89
31 Mar 1999	1998	1,644.26	-	1,869.42	-	225.16	2 P. 1	34.42	2	464.50	1,364.77	0.83
31 Mar 2000	1999	1,636.94	-	1,870.53		233.59	-	25.98	-	534.50	1,410.22	0.86
31 Mar 2001	2000	1,373.28	-	1,603.42	-	230.14	4	30.96	-	475.50	1,202.37	0.88
31 Mar 2002	2001	1,201.58	-	1,443.12	2	241.54	re l o	14.40	-	422.50	1,043.56	0.87 ^g
31 Mar 2003	2002	681.32	-	901.80	-	220.48	-	12.51	-	231.00	560.86	0.82
31 Mar 2004	2003	883.64	-	1.107.21		223.56		14.36	-	300.50	729.60	0.83
31 Mar 2005	2004	981.77	1,003.13	1,214.32	-	232.55	211.19	15.62		329.75	789.58	0.81
31 Mar 2006	2005	1,214.60	-	1,443.57		228.97	÷.	18.51	.÷.	429.50	1,004.80	0.83
31 Mar 2007	2006	1,205.23	-	1,416.05	-	210.82	-	19.51	4	481.00	1,036.54	0.86
31 Mar 2008	2007	945.04	-	1,153.42	- 2	208.39		21.98	-	403.25	799.62	0.66
31 Mar 2009	2008	641.40	-	845.17		203.77		17.63	-	292.50	570.72	0.60

Table X.7Original and Restated Data for Accounting Variables and Market Variables for THE EDINBURGH INVESTMENT TRUST
PLC Surveyed on a Yearly Basis from 1992 to 2008

				А	ccounting Variab	le ^c					Market Variable	
Date ^a	Year ^b	Total Sha Eq	reholders' uity	Total	Assets	Total Liabilities		Net Income ^d		Market	Market	Market To
		Original	Restated	Original	Restated	Original	Restated	Original	Restated	Price ^e	Value	Book Value
31 Oct 1992	1992	556.17	-	654.95	-	98.78	-	12.26	-	176.00	447.53	0.81
31 Oct 1993	1993	716.27	-	845.22	-	128.95	-	13.94	-	225.00	572.49	0.80
31 Oct 1994	1994	675.37	-	804.72	-	129.35	-	14.01	-	216.00	551.31	0.82
31 Oct 1995	1995	801.15	-	962.86	-	161.71	-	16.37	-	242.50	687.35	0.86
31 Oct 1996	1996	916.08	-	1,071.78	-	155.70	-	17.47	-	274.50	778.05	0.85
31 Oct 1997	1997	1,024.18	-	1,115.13	-	90.95	-	17.82	-	306.00	867.34	0.85
31 Oct 1998	1998	1,099.19	-	1,198.51	-	99.33	-	18.18	-	344.00	975.04	0.89
31 Oct 1999	1999	1,287.09	-	1,379.94	-	92.85	-	18.51	-	393.50	1,085.87	0.84
31 Oct 2000	2000	1,356.86	-	1,597.79	-	240.93	-	20.67	-	457.00	1,150.74	0.85
31 Oct 2001	2001	908.07	-	1,146.07	-	238.01	-	21.71	-	359.00	810.65	0.89
31 Oct 2002	2002	671.44	-	912.41	-	240.97	-	9.51	-	259.00	552.46	0.82
31 Oct 2003	2003	719.52	-	955.04	-	235.53	-	10.27	-	281.00	590.95	0.82
31 Oct 2004	2004	733.47	-	891.84	-	158.37	-	9.64	-	298.75	624.12	0.85
31 Oct 2005	2005	889.00	894.41	1,050.90	1,049.15	161.90	0.00	9.44	-	377.00	787.59	0.89
31 Oct 2006	2006	730.59	-	840.79	-	110.20	-	7.55	-	451.00	645.60	0.88
31 Oct 2007	2007	802.35	-	911.71	-	109.36	-	7.83	-	529.00	710.28	0.89
31 Oct 2008	2008	525.68	-	634.71	-	109.03	-	5.86	-	372.00	482.74	0.60

Table X.8Original and Restated Data for Accounting Variables and Market Variables for SCOTTISH INVESTMENT TRUST PLC
Surveyed on a Yearly Basis from 1992 to 2008

				A	ccounting Variab	le ^c					Market Variable	5
Date ^a	Year ^b	Total Sha Eq	reholders' uity	Total Assets Tc		Total Li	Total Liabilities		come ^d	Market	Market	Market To
		Original	Restated	Original	Restated	Original	Restated	Original	Restated	Price ^e	Value	Book Value
30 Apr 1993	1992	366.41	.2	457.81	4	91.41	4	6.13	1	81.60	316.58	0.86
30 Apr 1994	1993	459.01	-	507.02	÷	48.01	4.1	5.90	-	107.00	415.12	0.90
30 Apr 1995	1994	442.39	1.1	594.38	2	151.98	1.201	8.38	-	104.20	404.25	0.91
30 Apr 1996	1995	556.53	1	611.12	1.2	54.60	1.1	10.12	4	120.60	467.88	0.84
30 Apr 1997	1996	599.23	-	650.22		50.98	-	9.58		129.20	501.24	0.70
30 Apr 1998	1997	719.68	-	790.84	-	71.16	-	9.49	-	157.90	612.59	0.85
30 Apr 1999	1998	776.07	-	885.25	-	109.19		10.14	-	171.00	656.98	0.85
30 Apr 2000	1999	818.08	- 21	953.89		135.81	-	5.07	-	201.90	684.64	0.84
30 Apr 2001	2000	756.49	1	876.69	-	120.20		5.89	-	219.20	668.82	0.88
30 Apr 2002	2001	676.39	-	798.49		122.10	Co l or C	2.38	Conta P	195.25	595.75	0.88
30 Apr 2003	2002	493.69	-	612.22	-	118.53		1.86	-	143.00	420.93	0.85
30 Apr 2004	2003	612.77		696.76	-	83.99	-	2.12	-	173.00	509.21	0.83
30 Apr 2005	2004	649.15	651.88	738.90	738.14	89.75	86.26	3.60	-	184.50	537.55	0.83
30 Apr 2006	2005	935.20	-	1,125.61		190.41	-	4.71	÷.	290.00	833.34	0.89
30 Apr 2007	2006	963.44		1,115.80	-	152.36	1.4	7.75		300.25	842.66	0.88
30 Apr 2008	2007	1,030.85	-	1,116.32	2	85.46	4	8.08	+	348.00	919.35	0.95
30 Apr 2009	2008	680.76	-	773.02		92.27	÷	7.38	-	236.50	623.52	0.60

Table X.9Original and Restated Data for Accounting Variables and Market Variables for MONKS INVESTMENT TRUST PLC
Surveyed on a Yearly Basis from 1992 to 2008

				A	ccounting Variab	le ^c					Market Variable	
Date ^a	Year ^b	Total Sha Eq	areholders' uity	Total A	Assets	Total Liabilities		Net Income ^d		Market	Market	Market To
		Original	Restated	Original	Restated	Original	Restated	Original	Restated	Price	Value	Book Value
31 Jan 1993	1992	424.91	-	443.87	-	18.96	-	6.53	-	256.00	369.94	0.88
31 Jan 1994	1993	538.92	-	550.05	-	11.13	-	9.21	-	341.00	492.77	0.92
31 Jan 1995	1994	437.95	-	460.35	-	22.40	-	9.97	-	265.00	382.95	0.88
31 Jan 1996	1995	515.78	-	583.86	-	68.09	-	13.60	-	292.00	421.96	0.82
31 Jan 1997	1996	618.46	-	696.89	-	78.43	-	15.87	-	362.00	523.12	0.85
31 Jan 1998	1997	642.91	-	737.38	-	94.47	-	15.94	-	369.50	533.96	0.84
31 Jan 1999	1998	585.82	-	719.50	-	133.69	-	16.54	-	321.50	464.60	0.79
31 Jan 2000	1999	964.43	-	1,120.23	-	155.80	-	19.18	-	560.50	795.48	0.82
31 Jan 2001	2000	1,026.52	-	1,213.08	-	186.57	-	24.12	-	635.50	863.16	0.84
31 Jan 2002	2001	875.68	-	1,098.58	-	222.90	-	11.44	-	567.00	770.14	0.88
31 Jan 2003	2002	590.30	-	814.51	-	224.21	-	9.97	-	408.50	554.85	0.94
31 Jan 2004	2003	955.51	-	1,200.00	-	244.49	-	10.42	-	620.00	842.13	0.88
31 Jan 2005	2004	1,135.03	1,144.53	1,358.13	-	223.10	213.60	10.34	32.61	713.50	969.13	0.85
31 Jan 2006	2005	1,444.99	-	1,644.48	-	199.49	-	12.53	-	965.00	1,278.57	0.88
31 Jan 2007	2006	1,743.85	-	1,939.35	-	195.50	-	16.28	-	1,258.00	1,573.24	0.90
31 Jan 2008	2007	1,208.15	-	1,441.46	-	233.31	-	18.69	-	1,020.00	1,063.89	0.61
31 Jan 2009	2008	697.30	-	890.07	-	192.77	-	18.53	-	592.50	606.26	0.50

Table X.10Original and Restated Data for Accounting Variables and Market Variables for THE MERCANTILE INVESTMENT TRUSTPLC Surveyed on a Yearly Basis from 1992 to 2008

				А	ccounting Variab	le ^c					Market Variable	
Date ^a	Year ^b	Total Sha Eq	areholders' juity	Total	Assets	Total Li	abilities	Net Income ^d		Market	Market	Market To
		Original	Restated	Original	Restated	Original	Restated	Original	Restated	Price ^e	Value	Book Value
31 Dec 1992	1992	1,187.57	-	1,380.74	-	184.92	-	19.21	-	105.00	1,103.38	0.93
31 Dec 1993	1993	1,558.64	-	1,784.28	-	201.05	-	24.73	-	147.00	1,544.73	0.99
31 Dec 1994	1994	1,418.21	-	1,694.38	-	250.81	-	22.75	-	136.00	1,429.42	1.01
31 Dec 1995	1995	1,687.37	-	1,873.89	-	154.79	-	26.80	-	162.00	1,702.62	1.01
31 Dec 1996	1996	1,772.87	-	1,974.61	-	201.78	-	30.57	-	150.25	1,579.13	0.89
31 Dec 1997	1997	2,140.61	-	2,278.68	-	138.07	-	30.06	-	175.75	1,847.13	0.86
31 Dec 1998	1998	2,392.88	-	2,619.34	-	226.46	-	34.18	-	188.25	1,978.51	0.83 ^g
31 Dec 1999	1999	3,022.31	-	3,259.22	-	236.91	-	30.38	-	247.50	2,525.31	0.84 ^g
31 Dec 2000	2000	2,929.41	-	3,165.77	-	236.36	-	37.54	-	267.50	2,639.04	0.90
31 Dec 2001	2001	2,354.86	-	2,595.99	-	241.13	-	38.29	-	221.50	2,102.04	0.89
31 Dec 2002	2002	1,697.45	-	1,940.62	-	243.18	-	14.35	-	163.00	1,546.88	0.91
31 Dec 2003	2003	2,041.75	-	2,261.78	-	220.03	-	14.27	-	188.50	1,788.87	0.88
31 Dec 2004	2004	2,112.46	2,131.30	2,349.85	2,347.68	237.38	216.38	17.27	-	194.50	1,775.81	0.84
31 Dec 2005	2005	2,414.00	-	2,536.20	-	122.20	-	18.95	-	258.50	2,138.45	0.89
31 Dec 2006	2006	2,408.14	-	2,616.05	-	207.91	-	17.12	-	284.50	2,133.68	0.89
31 Dec 2007	2007	2,490.95	-	2,708.07	-	217.12	-	18.42	-	318.75	2,184.45	0.91
31 Dec 2008	2008	1,782.45	-	2,007.80	-	225.35	-	20.25	-	228.50	1,551.56	0.62

Table X.11Original and Restated Data for Accounting Variables and Market Variables for THE FOREIGN & COLONIAL
INVESTMENT TRUST PLC Surveyed on a Yearly Basis from 1992 to 2008

				А	ccounting Variab	le ^c					Market Variable	
Date ^a	Year ^b	Total Sha Eq	ureholders' uity	Total	Assets	Total Li	abilities	Net Ir	ncome ^d	Market	Market	Market To
		Original	Restated	Original	Restated	Original	Restated	Original	Restated	Price	Value	Book Value
31 Mar 1993	1992	794.53		1,042.82	4	248.29	4.1	13.90		184.50	665.53	0.84
31 Mar 1994	1993	928.65		1,070.01	-	141.35		14.85	-	219.00	789.98	0.85
31 Mar 1995	1994	871.34	1.2	1,094.93	2	223.59	1.4	18.80		217.00	782.76	0.90
31 Mar 1996	1995	1,113.14	-	1,335.84	- 2	222.70		21.39	-	268.75	969.43	0.87
31 Mar 1997	1996	1,216.07	-	1,485.96	-	269.89	-	24.60	-	290.50	1,047.89	0.86 ^g
31 Mar 1998	1997	1,574.10	-	1,897.04	-	322.94	-	21.69	-	384.00	1,385.16	0.88
31 Mar 1999	1998	1,645.90		1,881.37	-	235.47	24	20.26	-	375.50	1,354.50	0.82
31 Mar 2000	1999	1,927.58	-	2,225.06		297.48		14.62	-	493.50	1,671.58	0.87
31 Mar 2001	2000	1,464.29		1,677.93	-	213.64	4	25.21	2	377.00	1,239.27	0.85
31 Mar 2002	2001	1,302.99	-	1,528.19	- 2	225.21	r e l a r	10.81	-	369.00	1,140.89	0.88 ^g
31 Mar 2003	2002	844.32	-	1,077.02	-	232.70		11.01		234.50	698.82	0.82
31 Mar 2004	2003	1,127.78	-	1,375.29		247.51		12.79	-	305.00	906.78	0.80
31 Mar 2005	2004	1,229.44	1,242.62	1,470.28	1,468,54	240.84	225.92	10.72		333.00	984.36	0.80
31 Mar 2006	2005	1,753.35	-	1,989.15	-	235.80	-	11.80	.÷.	521.50	1,502.31	0.86
31 Mar 2007	2006	1,769.87	12	2,076.11	-	306.24		13.11		542.00	1,525.52	0.86
31 Mar 2008	2007	1,836.44	-	2,292.53		456.09	1	11.78	4	600.00	1,643.94	0.93
31 Mar 2009	2008	1,080.34	-	1,406.83	-	326.50		9.77	-	353.00	961.89	0.52

Table X.12Original and Restated Data for Accounting Variables and Market Variables for SCOTTISH MORTGAGE INVESTMENT
TRUST PLC Surveyed on a Yearly Basis from 1992 to 2008

APPENDIX Y CAPITAL EVENTS THAT ADJUST MARKET PRICE

This appendix provides events that are applied by Thomson Reuters to adjust potentially levels of the market price variable used in this study. The events presented are adapted from information sourced from Thomson Reuters in the month of November 2011 and screen prints from the Thomson Reuters Datastream database interface.

Y.1 Capital Events that Adjust Market Price

The events identified in this study to effect levels of the market price variable are:

Bonus/Scrip Issue in same stock Bonus/Scrip Issue into different stock Rights Issue or in same stock Rights Issue into different stock Stock Dividend Stock Split/Subdivision Consolidation/Reverse Split Open/Public offer Capital write-up Capital write-down Capital Repayment Exchange Into Multiple Issues Demerger Merger

Call Payments Registrations Redemptions Capital Reorganisation Conversions Issues Pending Distributions Miscellaneous

The following sections provide descriptions for some of these stated events. The definitions presented are adapted from information sourced from Thomson Reuters in the month of November 2011.

Y.1.1 Bonus/Scrip Issue in Same Stock

The "free" issue of new shares to existing shareholders, on a new for old basis. Shareholders are sent definitive certificates indicating their new number of shares on the ex-dividend date of the bonus issue. The issue is free in so much as the shareholder does not pay for the shares. However, because of the increased number of shares in the market, the price of the firm's single share is adjusted downward accordingly.

Y.1.2 Bonus/Scrip Issue Into Different Stock

The "free" issue of new *different class* of shares to existing shareholders, on a new for old basis. This is the same as the event *Bonus/Scrip Issue in same stock*, the difference being that the shareholders are provided a different class of shares. This event does not adjust a firm's share price.

Y.1.3 Stock Dividend

A stock dividend is where profits are turned into share capital and then issued to shareholders as a dividend. Thomson Reuters does not adjust the market price for stock dividends explicitly. However, they are adjusted at the exchange level. Therefore, when recorded by Thomson Reuters a firm's share price has been adjusted. The price of a firm's shares generally adjust downwards for this stock dividend event.

Y.1.4 Stock Split/Sub Division

This event is specified as the splitting of the nominal value of a stock to increase the number of shares in issue without issuing new shares. This event has the effect of lowering the share's market price. After this event, shareholders are issued with new share certificates, indicating the new nominal value of the stock and the new number of shares held on the effective date of the subdivision.

Y.1.5 Consolidation/Reverse Split

Increasing the nominal value of a share to reduce the number of shares in issue. In this event, several shares of a lower nominal value are consolidated into one share of a higher nominal value. This event has the effect of increasing the share's market price to adjust for the decrease in the number of shares. Shareholders are issued with new share certificates on the effective date of the consolidation.

Y.1.6 Rights Issue Into Same Class

The issue of "new" shares to existing shareholders in order to raise additional capital for the firm. The shares are issued on a "new" for "old" basis at a favorable market price. This event does not explicitly adjust the market price of shares.

Y.1.7 Capital Write-Up

The increase of the nominal value of the stock without reducing the number of shares in issue. Capital write-ups are financed by the firm's reserves and can only be implemented after a court sanction. There is no change in the number of shares or to the share price. The firm would do this to take money out of its capital reserves by increasing the nominal value of its shares.

Y.1.8 Capital Write-Down

The decrease of the nominal value of a stock without increasing the number of shares in issue. Capital write-downs are used to pull money back into the firm's reserves and can only be implemented by a court sanction. This event does not cause a change to the number of shares or to the share price.

Y.1.9 Capital Repayment

An event where a firm makes a payment to reduce an amount of capital liability. Shareholders retain the same number of shares as before an issue became effective and there will be a decrease in the price of the shares.

Y.1.10 Exchange Into

Thomson Reuters determines that there is no formal definition for the *Exchange into* event. However, it is stated that based on the terms (type of issue) there is an adjustment for the *Exchange into* capital issues. Generally, this would reduce the market price per share.

Y.1.11 Multiple Issues

Thomson Reuters determines that there is no formal definition for the *Multiple issues* event. However, it is stated that based on the terms (type of issue) there is an adjustment for the *multiple* capital issues. Generally, this would reduce the market price per share.

Y.1.12 Demerger

The splitting of one firm into two or more firms by spinning off one or more of the firm's subsidiaries. Shareholders receive documents informing them of the demerger and the share certificates in the new firm are received on the effective date of the issue. The share price of the original firm is "split" based on the share price of the new firm. The original firms share price is reflected by factors that include the terms of the demerger and the size of the new firm.

Y.1.13 Merger

When two or more firms combine activities to form a new firm, they are said to merge. The directors of the firms involved meet, discuss and agree the merger. The shareholders of each firm involved are sent documents detailing the terms and conditions the directors propose. Shareholders have no vote on the proposal. A merger can take place between two listed firms or between a listed firm and a non-listed firm. The merged firm's share price is reflected by factors that include the terms of the merger and the size of the new firm.

APPENDIX Z CALCULATED AND OMITTED DATA

Z.1 Calculated and Omitted Data

This appendix presents in Table Z.1 variable items that are calculated or omitted from analysis. Some items are calculated due to calculation conventions applied in this study. Some items are omitted due to minor data irregularities and missing data. In addition, this table discloses items that are collected after *1*st *January 1991*. This date relates to the starting time range group year 1992 (see Section 3.7.7). This study considers items that are calculated, omitted, or collected after the *start date* does not significantly affect the findings presented in this research.

Table Z.1 Calculated and Part Omitted Data Items

The Data Changed column specifies if the item presented in the Data Item column is calculated or omitted for a specified date or date range(s) presented in the Notable Dates column. The Description column provides further details for the data item's omission or calculation.

Data Changed	Data Item	Notable Dates	Description
Calculated	Market-to-Book Ratio	31st December 1998 and 31st December 1999	For HSBC HOLDINGS PLC the 31st December 1998 and the 31st December 1999 market-to-book ratio values are calculated using the ratio of the Total Number of Shares Outstanding multiplied by the Adjusted Share Price divided by the Total Shareholders' Equity.
			The remaining market-to-book values are calculated using the ratio of the Market Value divided by the Total Shareholders' Equity.
Calculated	Book-to-Market Ratio	General	The Book-to-Market ratio is calculated as the reciprocal of the Market-to-Book value. The Market-to-Book value is stated to maintain consistency with the data providers published data.
Calculated	Value-at-Risk Calculation	30th December 1994	For PROVIDENT FINANCIAL PLC from 30th December 1994 the Value-at- Risk values were calculated using the Excel PERCENTILE function.
Calculated	Value-at-Risk Calculation	General	The market price returns for all time horizons for the sample and control Value- at-Risk calculations were performed on an absolute day basis, for example, for a 500-day time horizon, the return was calculated on a 500 day basis and not a 501 day basis. However, the return for the 1-day time horizon Value-at-Risk was calculated on a 2 day basis.
Calculated	Value-at-Risk Calculation	General	The <i>returns</i> and percentage changes for all time horizons for the GDP and selected indices Value-at-Risk calculations were performed on an absolute day basis, i.e. for a 250-day time horizon, the <i>returns</i> and percentage changes were calculated on a 250-day basis and not a 251 day basis

Table Z.1 (Continued)									
Data Changed	Data Item	Notable Dates	Description						
Part omitted	Adjusted Close Price	31st December 1992, 31st December 1993	For LLOYDS BANKING GROUP PLC the 31st December 1992, 31st December 1993 and 31st December 1994 the Adjusted Close Prices are unavailable and are replaced by 0.00.						
Part omitted	Adjusted Close Price	19th October 1993 and 31st October 1994	For BANKERS' INVESTMENT TRUST PLC (THE), ticker symbol BNKR has the same market close price for 19th October 1993 and 31st October 1994. In order to avoid divide by zero errors the market close price for 19th October 1993 has been kept at 185.00 GBX and the market close price for 31st October 1994 has been set from 185.00 GBX to 185.01GBX.						
Part omitted	Adjusted Close Price	31st July 1998 and 30th July 1999	For CLOSE BROTHERS GROUP PLC, ticker symbol CBG has the same market close price for 31st July 1998 and 30th July 1999. In order to avoid divide by zero errors the market close price for 31st July 1998 has been kept at 847.50 GBX and the market close price for 30th July 1999 has been set from 847.50 GBX to 847.51 GBX.						
Part omitted	Market Value	31st December 1992, 31st December 1993	For LLOYDS BANKING GROUP PLC the 31st December 1992, 31st December 1993 and 31st December 1994 the Market Values are unavailable and is replaced by 0.00.						
Part omitted	Market-to-Book Ratio	31st December 1992, 31st December 1993	For LLOYDS BANKING GROUP PLC the 31st December 1992, 31st December 1993 and 31st December 1994 the Market-to-Book ratios are unavailable and are replaced are 0.00.						
Part omitted	Value-at-Risk	30th December 1994 and 29th December 1995	For LLOYDS BANKING GROUP PLC the 30th December 1994 and 29th December 1995 Value-at-Risk for the actual 250-day time horizon that should be calculated with 300 days of historical returns is not calculated due to the Adjusted Close Price data being unavailable within the 300						

		Table Z.1 (Continued)		
-	Data Changed	Data Item	Notable Dates	Description
-				historical day range.
	Part omitted	Value-at-Risk	29th July 1994 (1995 year)	For CLOSE BROTHERS GROUP PLC the 29th July 1994 Value-at-Risk for the actual 250-day time horizon calculated with 300 days of historical returns is not included in the 1995 sample due to the Value-at-Risk calculation being outside of the 300 historical day range.
	Part omitted	Value-at-Risk	29th July 1994 (1995 year)	For PACIFIC HORIZON INVESTMENT TRUST PLC the 29th July 1994 Value-at-Risk for the 250-day time horizon calculated with 300 days of historical returns is not included in the 1995 sample due to the Value-at-Risk calculation being outside of the 300 historical day range.
	Starts	RTS Stock Index	Starts from 9th January 1995	The Russia RTS INDEX - PRICE INDEX, Datastream identity RSRTSIN(PI), starts from 9th January 1995.
	Starts	Bovespa FTSE Stock Index	Starts from 11th January 1994	The Brazil Bovespa FTSE BRAZIL -PRICE INDEX, Datastream identity WIBRAZL(PI), starts from 11th January 1994.
	Starts	S&P/ASX 200 Stock Index	Starts from 29th May 1992	The Australia S&P/ASX 200 - PRICE INDEX, Datastream identity ASX200I(PI), starts from 29th May 1992.
	Starts	MSCI Stock Index	Starts from 31st December 1992	The South Africa MSCI SOUTH AFRICA - PRICE INDEX, Datastream identity MSSARFL(PI), starts from 31st December 1992.
	Starts	EUR and GBP Currency Pair	Starts from 4th January 1999	The exchange rate for the European Euro (EUR) to the Great British Pound (GBP), starts from 4th January 1999. This is the first day that the European Euro was issued

Table Z.1 (Continued)					
Data Changed	Data Item	Notable Dates	Description		
Part omitted	GILT 2.5 year	See description	Data unavailable for the 2.5 Year GILT benchmark rate from 1st January 1991 to 27th March 1991, and from 21st December 1992 to 29th July 1993, and from 2nd December 1996 to 30th July 1998, and from 27th May 2003 to 5th June 2003, and from 25th February 2005 to 6th April 2006, and from 27th December 2007 to 2nd July 2008.		
Part omitted	GILT 3 year	See description	Data unavailable for the 3 Year GILT benchmark rate from 21st December 1992 to 28th January 1993, and from 2nd December 1996 to 29th January 1998, and from 25th February 2005 to 20th September 2005.		
Part omitted	GILT 3.5 year	See description	Data unavailable for the 3.5 Year from 2nd December 1996 to 31st July 1997, and from 25th February 2005 to 8th April 2005.		
Part omitted	GILT 4 year	2nd December 1996 to 30th January 1997	Data unavailable for the 4 Year GILT benchmark rate from 2nd December 1996 to 30th January 1997.		
Part omitted	GILT 17 year	4th October 1991 to 10th February 2008	Data unavailable for the 17 Year GILT benchmark rate from 4th October 1991 to 10th February 1992.		
Part omitted	GILT 17.5 year	5th April 1991 to 10th February 2008	Data unavailable for the 17 Year GILT benchmark rate from 5th April 1991 to 10th February 1992.		
Part omitted	GILT 18 year	1st January 1992 to 10th February 2008	Data unavailable for the 18 Year GILT benchmark rate from 1st January 1992 to 10th February 2008.		
Part omitted	GILT 18.5 year	1st January 1992 to 10th February 2008	Data unavailable for the 18.5 Year GILT benchmark rate from 1st January 1992 to 10th February 2008.		

Table Z.1 (Continued)					
Data Changed	Data Item	Notable Dates	Description		
Part omitted	GILT 19 year	1st January 1992 to 29th April 1992	Data unavailable for the 19 Year GILT benchmark rate from 1st January 1992 to 29th April 1992.		
Part omitted	GILT 19.5 year	1st January 1992 to 29th April 1992	Data unavailable for the 19.5 Year GILT benchmark rate from 1st January 1992 to 29th April 1992.		
Part omitted	GILT 20 year	See description	Data unavailable for the 20 Year GILT benchmark rate from 1st January 1992 to 29th April 1992, and from 5th December 1995 to 27th February.		
Part omitted	GILT 20.5 year	See description	Data unavailable for the 20.5 Year GILT benchmark rate from 1st January 1992 to 29th April 1992, and from 5th June 1995 to 27th February 1996.		
Part omitted	GILT 21 year	See description	Data unavailable for the 21 Year GILT benchmark rate from 1st January 1992 to 29th April 1992, and from 5th December 1994 to 27th February 1996.		
Part omitted	GILT 21.5 year	See description	Data unavailable for the 21.5 Year GILT benchmark rate from 1st January 1992 to 29th April 1992, and from 3rd June 1994 to 27th February 1996.		
Part omitted	GILT 22 year	See description	Data unavailable for the 22 Year GILT benchmark rate from 1st January 1992 to 29th April 1992, and from 3rd December 2003 to 27th February 1996, and from 5th September 1997 to 28th January 1998.		
Part omitted	GILT 22.5 year	See description	Data unavailable for the 22.5 Year from 1st January 1992 to 29th April 1992, and from 4th June 1993 to 27th February 1996, and from 6th March 1997 to 28th January 1998.		
Part omitted	GILT 23 year	See description	Data unavailable for the 23 Year GILT benchmark rate from 1st January 1992 to 29th April 1992, and from 4th December 2003 to 27th February		

Data Changed	Data Item	Notable Dates	Description
			1996, and from 5th September 1996 to 28th January 1998.
Part omitted	GILT 23.5 year	See description	Data unavailable for the 23.5 Year GILT benchmark rate from 1st Januar 1992 to 29th April 1992, and from 5th June 1992 to 27th February 199 and 6th March 1996 to 28th January 1998.
Part omitted	GILT 24 year	See description	Data unavailable for the 24 Year GILT benchmark rate from 1st Januar 1992 to 29th January 1998, and from 24th September 1999 to 23rd Ma 2000.
Part omitted	GILT 24.5 year	See description	Data unavailable for the 24.5 Year GILT benchmark rate from 1st Januar 1992 to 29th January 1998, and from 8th July 1999 to 23rd May 2000.
Part omitted	GILT 25 year	See description	Data unavailable for the 25 Year GILT benchmark rate from 1st Januar 1992 to 29th January 1998, and from 30th October 1998 to 6th Januar 1999, and from 19th February 1999 to 23rd May 2000.